Philip G. Janicak · Stephen R. Marder Rajiv Tandon · Morris Goldman *Editors*

Schizophrenia

Recent Advances in Diagnosis and Treatment



Schizophrenia

Philip G. Janicak • Stephen R. MarderRajiv Tandon • Morris GoldmanEditors

Schizophrenia

Recent Advances in Diagnosis and Treatment



Editors Philip G. Janicak Department of Psychiatry Rush University Chicago, IL, USA

Rajiv Tandon College of Medicine University of Florida Gainesville, FL, USA Stephen R. Marder UCLA Semel Institute Los Angeles, CA, USA

Morris Goldman Department of Psychiatry and Behavioral Sciences Northwestern University Chicago, IL, USA

ISBN 978-1-4939-0655-0 ISBN 978-1-4939-0656-7 (eBook) DOI 10.1007/978-1-4939-0656-7 Springer New York Heidelberg Dordrecht London

Library of Congress Control Number: 2014936416

© Springer Science+Business Media New York 2014

This work is subject to copyright. All rights are reserved by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed. Exempted from this legal reservation are brief excerpts in connection with reviews or scholarly analysis or material supplied specifically for the purpose of being entered and executed on a computer system, for exclusive use by the purchaser of the work. Duplication of this publication or parts thereof is permitted only under the provisions of the Copyright Law of the Publisher's location, in its current version, and permission for use must always be obtained from Springer. Permissions for use may be obtained through RightsLink at the Copyright Clearance Center. Violations are liable to prosecution under the respective Copyright Law.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

While the advice and information in this book are believed to be true and accurate at the date of publication, neither the authors nor the editors nor the publisher can accept any legal responsibility for any errors or omissions that may be made. The publisher makes no warranty, express or implied, with respect to the material contained herein.

Printed on acid-free paper

Springer is part of Springer Science+Business Media (www.springer.com)

For Rodney and Jude who taught me so much Phil

To Paula for her support and her patience

Steve

For Gitanjali, Anisha, Neeraj, and Nammi for their love and support

Rajiv

For Hilarie, Frank DeHaan, and Paul Tolpin to whom I am forever grateful

Morrie

Preface

Over 60 years ago, chlorpromazine dramatically changed our understanding and treatment of patients suffering from schizophrenia. Clozapine reemerged over 20 years ago, representing both our most effective drug for refractory schizophrenia and most vexing in terms of its disadvantages. While some would argue that there has been little progress since, an enormous amount of research into the underlying pathoetiology of schizophrenia has been conducted. This body of knowledge sets the stage for future progress in:

- Parsing out the heterogeneity associated with the umbrella term "schizophrenia," allowing for subgroups to emerge with major implications for more effective treatment approaches.
- Early identification of at-risk individuals who may be prophylactically treated to prevent or attenuate the disease process.
- Developing therapies which are more targeted and less deleterious in terms of unwanted effects.
- Translating symptom reduction into improved functionality and quality of life.

At present, the most promising initiatives to facilitate these goals are in the areas of genetics, neuroimaging, and psychosocial rehabilitation approaches.

This book is divided into four sections. Part I summarizes our present state of knowledge about the diagnosis and treatment of schizophrenia. Part II considers recent discoveries into its pathoetiology, including the status of biological markers, genetics, and neuroimaging as they relate to diagnosis and potential novel therapeutic approaches. Part III explores the optimization of our present therapeutic approaches, novel treatments, and management of the substantial risks associated with both the illness and its present therapies. Part IV discusses progress in the long-term management of schizophrenia, focusing on biological and psychotherapeutic strategies to improve functioning and facilitate recovery. Finally, Part V considers future directions and predictions of how diagnosis and treatment of schizophrenia will change.

While the sense of being in a "holding pattern" is inevitable given the pace of progress in the clinical arena, there is also a growing recognition that the significant strides taken in basic research will usher in a second renaissance in the care of patients suffering from schizophrenia.

Chicago, IL, USA Los Angeles, CA, USA Gainesville, FL, USA Chicago, IL, USA Philip G. Janicak, M.D. Stephen R. Marder, M.D. Rajiv Tandon, M.D. Morris Goldman, M.D.

Contents

Part I Overview

1	Introduction Philip G. Janicak, Stephen R. Marder, Rajiv Tandon, and Morris Goldman	3
2	The Evolving Nosology of Schizophrenia: Relevance for Treatment Rajiv Tandon and Dawn Bruijnzeel	13
Par	t II Recent Research into the Pathophysiology of Schizophrenia	
3	Overview of Neurobiology Rajiv Tandon and Morris Goldman	27
4	Pathophysiology of Schizophrenia Jaya Padmanabhan and Matcheri S. Keshavan	35
5	Genetics of Schizophrenia Alan R. Sanders	59
6	Recent Advances in Neuroimaging Biomarkers of Schizophrenia Lei Wang and John G. Csernansky	71
Par	t III Medical Management of Schizophrenia	
7	Acute Management of Schizophrenia Philip G. Janicak	107
8	Therapeutic Neuromodulation for Treatment of Schizophrenia Jeffrey T. Rado and Edgar I. Hernandez	139

9	Pharmacogenetics in the Treatment of Schizophrenia Bill W. Massey, Jiang Li, and Herbert Y. Meltzer	161		
10	Management of Comorbid Medical Conditions in Schizophrenia Jeffrey T. Rado	175		
11	Management of Water Imbalance in Schizophrenia Morris Goldman and Pichai Ittasakul	205		
12	Management of Medication-Related Adverse Effects Wanlop Atsariyasing and Morris Goldman	225		
Part IV Management of Schizophrenia and Its Long-Term Complications				
13	Long-Term Pharmacological Management of Schizophrenia Stephen R. Marder	265		
14	Psychosocial Rehabilitation and Psychotherapy Approaches Robert S. Kern, William P. Horan, Shirley M. Glynn, L. Felice Reddy, Jason Holden, Eric Granholm, Luana R. Turner, Mary E. Sullivan, and William D. Spaulding	275		
Part V Future Directions and Implications				
15	Conclusion Philip G. Janicak, Stephen R. Marder, Rajiv Tandon, and Morris Goldman	315		
Ind		321		

Contributors

Wanlop Atsariyasing, M.D. Department of Psychiatry, Faculty of Medicine Siriraj Hospital, Mahidol University, Bangkok, Thailand

Dawn Bruijnzeel, M.D. Department of Psychiatry, University of Florida College of Medicine, Gainesville, FL, USA

John G. Csernansky, M.D. Department of Psychiatry and Behavioral Sciences, Northwestern University Feinberg School of Medicine, Chicago, IL, USA

Shirley M. Glynn, Ph.D. VA Greater Los Angeles Healthcare System, University of California - Los Angeles, Semel Institute for Neuroscience and Human Behavior, David Geffen School of Medicine, Los Angeles, CA, USA

Morris Goldman, M.D. Department of Psychiatry and Behavioral Sciences, Northwestern University, Chicago, IL, USA

Eric Granholm, Ph.D. VA San Diego Healthcare System and University of California, San Diego, CA, USA

Edgar I. Hernandez, M.D. Department of Psychiatry, Rush University Medical Center, Chicago, IL, USA

Jason Holden, Ph.D. Research Service, VA San Diego Healthcare System, San Diego, CA, USA

William P. Horan, Ph.D. VA Greater Los Angeles Healthcare System, University of California - Los Angeles, Semel Institute for Neuroscience and Human Behavior, David Geffen School of Medicine, Los Angeles, CA, USA

Pichai Ittasakul, M.D. Department of Psychiatry, Ramathibodi Hospital, Mahidol University, Bangkok, Thailand

Philip G. Janicak, M.D. Department of Psychiatry, Rush University Medical Center, Chicago, IL, USA

Robert S. Kern, Ph.D. VA Greater Los Angeles Healthcare System, University of California - Los Angeles, Semel Institute for Neuroscience and Human Behavior, David Geffen School of Medicine, Los Angeles, CA, USA

Matcheri S. Keshavan, M.D. Department of Psychiatry, Beth Israel Deaconess Medical Center and Massachusetts Mental Health Center, Harvard Medical School, Boston, MA, USA

Jiang Li, Ph.D. Department of Psychiatry and Behavioral Sciences, Northwestern University, Feinberg School of Medicine, Chicago, IL, USA

Stephen R. Marder, M.D. UCLA Semel Institute, Los Angeles, CA, USA

Bill W. Massey, Ph.D. Department of Psychiatry and Behavioral Sciences, Northwestern University, Feinberg School of Medicine, Chicago, IL, USA

Herbert Y. Meltzer, M.D. Department of Psychiatry and Behavioral Sciences, Northwestern University, Feinberg School of Medicine, Chicago, IL, USA

Jaya Padmanabhan, M.D. Department of Psychiatry, Beth Israel Deaconess Medical Center and Massachusetts Mental Health Center, Boston, MA, USA

Jeffrey T. Rado, M.D., M.P.H. Department of Psychiatry, Rush University Medical Center, Chicago, IL, USA

Department of Internal Medicine, Rush University Medical Center, Chicago, IL, USA

L. Felice Reddy, Ph.D. VA Greater Los Angeles Healthcare System, USA

Alan R. Sanders, M.D. Department of Psychiatry and Behavioral Sciences, NorthShore University HealthSystem, Evanston, IL, USA

University of Chicago, Evanston, IL, USA

William D. Spaulding, Ph.D. Department of Psychology, University of Nebraska - Lincoln, Lincoln, NE, USA

Mary E. Sullivan, M.S.W. Department of Psychology, University of Nebraska - Lincoln, Lincoln, NE, USA

Rajiv Tandon, M.D. College of Medicine, University of Florida, Gainesville, FL, USA

Luana R. Turner, Psy.D. University of California - Los Angeles, Semel Institute for Neuroscience and Human Behavior, David Geffen School of Medicine, Los Angeles, CA, USA

Lei Wang, Ph.D. Department of Psychiatry and Behavioral Sciences and Radiology, Northwestern University Feinberg School of Medicine, Chicago, IL, USA

Part I Overview

Chapter 1 Introduction

Philip G. Janicak, Stephen R. Marder, Rajiv Tandon, and Morris Goldman

Clinical Features

Schizophrenia is a heterogeneous disorder with a broad range of symptoms and a variable course. It typically afflicts younger individuals, often leading to lifelong disability. The most dramatic aspect of this disease is the periodic episodes of psychotic symptoms (e.g., auditory hallucinations, paranoid delusions) which at times require emergent interventions to protect the individual and others. This illness, however, also includes more persistent impairment in multiple areas of cognition, substantial deficits in interpersonal relationships, and depressed mood, at times predisposing individuals to a heightened risk of suicide. The culmination of these symptoms may be a marked diminution in the ability to function adequately (e.g., socially, occupationally), as well as a greatly diminished quality of life.

The concept of schizophrenia continues to evolve with our expanding ability to detect genetic and environmental influences, as well as their interactions, advances in neuroimaging, and recognition of other biological factors that better inform our

P.G. Janicak (🖂)

S.R. Marder UCLA Semel Institute, Los Angeles, CA, USA e-mail: marder@ucla.edu

R. Tandon College of Medicine, University of Florida, P.O. Box 100256, Gainesville, FL 32610-0256, USA e-mail: tandon@ufl.edu

M. Goldman

Department of Psychiatry, Rush University Medical Center, 2150 West Harrison Street, Chicago, IL 60612, USA e-mail: pjanicak@rush.edu

Department of Psychiatry and Behavioral Sciences, Northwestern University, East Ontario 446, Chicago, IL 60611, USA e-mail: m-goldman@northwestern.edu

understanding of this condition. While the search for biomarkers has been frustratingly slow, it is important to the development of more targeted treatments and to better predict clinical outcomes.

In parallel, treatments for schizophrenia are evolving to accommodate this new information. Unfortunately, the pace of these initiatives leaves patients and clinicians with therapeutic approaches that presently fall far short of ideal. For example, functional impairment across multiple psychopathological dimensions continues to occur in most patients and life expectancies are dramatically shortened due to modifiable factors such as suicidality, substance abuse, and cardiovascular disease [1, 2].

Schizophrenia is characterized by multiple psychopathological dimensions (positive, negative, cognitive, mood, disorganization, and motor symptoms) whose severity varies across patients and course of illness. Each of these dimensions contributes to the significant distress and disability associated with this disorder. Additionally, these symptom dimensions appear to have different underlying neurobiological substrates and exhibit distinctive patterns of response to the various treatments utilized. The course of schizophrenia is characterized by a sequential trajectory that involves a premorbid phase with subtle and nonspecific cognitive, motor, and/or social dysfunction. The prodromal phase may include attenuated positive, negative, and cognitive symptoms and declining function. The first psychotic episode heralds the formal onset of schizophrenia, and often a lifelong course with the initial decade of illness sometimes marked by repeated episodes of psychosis with partial and variable degrees and duration of inter-episode remission and accrual of disability with each episode of illness. Finally, a stable phase or plateau may include psychotic symptoms that are less prominent while negative symptoms and cognitive deficits may increasingly predominate [3]. Recovery of varying degrees can occur at any stage of the illness and, in contrast to the original perspective of unalterably progressive deterioration, a significant proportion of patients do exhibit substantial improvement.

In Chap. 2, the description and definition of our current clinical understanding of schizophrenia in DSM-5 are summarized and its treatment implications discussed [4, 5].

Epidemiology and Genetic and Environmental Risk Factors

A comprehensive review of 150 studies from 33 countries indicated that between 1965 and 2001 the *mean incidence* of schizophrenia was 15.2 per 100,000 (range=7.3–43 per 100,000) [6]. Further, the incidence was higher (i.e., a ratio of 1.4) and the onset of illness earlier (i.e., 2–3 years) in men than women [7]. A second review estimated that the *lifetime prevalence* was 4.0 per 1,000 [8].

As emphasized in Chap. 5, the central role of genetic factors in the etiology of schizophrenia has long been recognized. The role of environment in the development of schizophrenia, however, is now being increasingly appreciated. For example, existing data indicate that being raised in an urban versus rural setting confers an up

to twofold greater risk for developing this disorder and birth during winter and early spring increases the risk by 5-15 % [9, 10]. In addition, culture plays an important role in symptom presentation, course, and outcome leading to a decreased or increased chance of being diagnosed with schizophrenia [11].

Longitudinal and birth cohort studies are also providing a growing list of potential risk factors such as prenatal environmental (e.g., in utero infections, micronutrient deficiencies, obstetric complications such as preeclampsia or brain damage) or postnatal environmental (e.g., stress, inflammation, cannabis use) effects in the etiopathology of schizophrenia [12]. These factors appear to interact in complex ways with genetic, psychological, sociocultural, and economic issues to increase the risk of developing this illness. For example, four candidate genes for schizophrenia may play a role in obstetric-related hypoxic events (i.e., AKT1, BDNF, GRM3, DTNBP1) [13]. In the context of cannabis use, one meta-analysis of prospective studies found a pooled odds ratio of 2.3 (95 % CI=1.7-2.9) for an association between cannabis use and increased risk for schizophrenia [14]. Another meta-analysis provided evidence that cannabis may play a role in an earlier onset of psychosis [15]. Yet a third study found an association between cannabis use and a more adverse course of schizophrenia over a 10-year follow-up period [16]. Further, as an example of gene x environment interaction, it appears the possession of the catechol-O-methyltransferase (COMT) valine 158 allele which increases vulnerability to psychosis may be exacerbated by cannabis use [17]. These studies offer the hope that environmental factors can be addressed with existing and future interventions, ultimately reducing the risk of developing or favorably moderating the course of schizophrenia.

Overview of Neurobiology

A range of neurobiological differences between individuals with and without schizophrenia have been defined and their relevance to the pathophysiology of schizophrenia is reviewed in Chaps. 3-6 [18]. These chapters provide summaries of our current knowledge regarding the neurobiology of schizophrenia, but are largely devoted to understanding some of the limitations of previous research approaches, as well as how current approaches address these challenges. All four chapters underscore the need to identify biomarkers and intermediate phenotypes in order to focus on future research efforts. Chapter 3 emphasizes the general need to characterize mechanisms of brain function to better distinguish adaptive from pathologic processes. The failure to consider the likelihood that many findings are a reaction to pathology rather than contributing to the pathology may in itself account for a failure to better integrate the massive amount of evidence supporting the view that schizophrenia reflects a diseased brain. Chapter 4 lays out some of the barriers which are beginning to be addressed, as well as providing a framework for integrating diverse findings by providing examples of how knowledge from several different domains can be integrated yielding more powerful models to investigate the illness(es). Chapter 5 provides an historical review of genetic findings in schizophrenia and at the same time summarizes the logic, strengths, and weakness of these evolving approaches. It also attempts to provide a roadmap of how this field will incorporate multimodal approaches in the coming years. Chapter 6 addresses neuroimaging and focuses more on recent advances, again illustrating efforts to incorporate multiple modalities in the service of better characterizing the functional significance of findings and providing biomarkers of the illnesses. What follows next is background information about the different research domains which characterize current neurobiologic efforts. We anticipate this will help contextualize the chapters themselves.

Neurotransmission

Dopamine (DA) is a critical neurotransmitter in our conceptualization of schizophrenia. While existing antipsychotics all impact this system (i.e., primarily through antagonism of the DA₂ receptor), their greatest clinical benefit is in controlling DA mesolimbic hyperactivity and ameliorating positive symptoms. More recent data also indicates an ability to improve brain reward mechanisms [19]. Other neurotransmitters (e.g., serotonin, glutamate, GABA acetylcholine) also appear to play important roles, particularly for more difficult-to-treat symptoms (i.e., negative, cognitive). Indeed, these systems may alter DA neurotransmission and in turn be impacted by DA. In this context, specific genes related to these neurotransmitters may be associated with the pathoetiology of schizophrenia. Examples include glutamate (e.g., GRIN2B, SAP97) and serotonin (e.g., 5-HTT variants) [20-22]. Associations with immune system dysregulation (e.g., increased levels of proinflammatory cytokines, high levels of autoantibodies) and schizophrenia have also identified potential genetic links, including the IL-3 receptor (IL3RB) which may interact with polymorphisms of neuregulin (NRG1) to increase the risk of schizophrenia [23]. Another example is the observation of increased antibodies for the α 7 nicotinic cholinergic receptor in patients with schizophrenia [24]. These issues are considered in greater detail in Chap. 4.

Genetics

Several genetic and nongenetic factors (i.e., multifactorial mode of inheritance) contribute to the risk of developing schizophrenia [25]. Family, twin, adoption, and candidate gene studies all support a genetic component for psychotic disorders. Further, genome-wide association studies (GWAS) suggest that submicroscopic chromosomal abnormalities (i.e., *copy number variants* (CNVs)) involving deletions and/or duplications may confer a higher risk for developing schizophrenia [26]. These studies provide evidence of a role for both common and rare, large CNVs in schizophrenia [27]. These abnormalities may occur de novo, be inherited, or involve both processes. Early findings of CNVs at the 22 q 11.2 locus indicate

they may play an important role in the susceptibility to complex disorders such as schizophrenia [28]. Such changes may interact with a polygenic liability and/or environmental factors to increase this risk [29]. There is also a signal for an association with genetic markers in the *major histocompatibility complex* (6 p 22.1) containing genes. Finally, more recent evidence from a genome-wide analysis indicates a greater overlap in specific *single-nucleotide polymorphisms* (SNPs) for five major psychiatric disorders (i.e., schizophrenia, bipolar disorder, autism spectrum, attention-deficit hyperactivity, and major depressive disorder) [30]. Despite these advances, most of the tentative heritability of schizophrenia is still unclear.

In this context, the exploration of *epigenetic variations* with recent technological advances may better inform us about the impact of environment on genetic expression associated with neuropsychiatric disorders [31]. Epigenetics involves the modification of chromatin (i.e., the DNA–histone complex in the nucleus). These phenomena can dynamically moderate gene expression separate from DNA sequence variation and may regulate important CNS neurobiological and cognitive processes [32]. Various external factors can promote lability in the epigenome (e.g., altered DNA methylation and histone modifications) predisposing an individual to develop schizophrenia. Certain antipsychotics (e.g., clozapine) may also alter the epigenome [33]. Although our understanding of this process is in its infancy, the dynamic and reversible qualities of the epigenome may facilitate the development of novel diagnostic and therapeutic approaches for schizophrenia. A more complete discussion of these issues is provided in Chap. 5.

Neuroimaging

The use of emerging *neuroimaging technology* to determine the pathoetiology of schizophrenia has expanded dramatically over the last 30 years. Techniques such as positron emission tomography (PET); single-photon emission computed tomography (SPECT); structural, real-time functional and spectroscopic magnetic resonance imaging (MRI); and diffusion tensor imaging (DTI) are all contributing to a better understanding of neurodevelopmental and neurodegenerative aspects of this illness. For example, there is a growing appreciation of dysregulated neural connectivity which moves beyond identification of associated structural and functional abnormalities [34]. The field of "connectomics" has recently identified such impairments in first episodes of schizophrenia [35]. A recent meta-analysis of 44 studies (618 patients; 606 controls) using PET or SPECT imaging indicated that the largest dopamine abnormalities in patients with schizophrenia may be presynaptic (i.e., DA synthesis capacity, baseline DA synaptic levels, DA release) [36]. Ironically, the authors note that all presently available antipsychotic agents target *postsynaptic* activity. Another meta-analysis of 27 studies (928 patients; 867 controls) found that over time (up to 10 years) patients with schizophrenia had significantly larger decreases in whole brain volume; whole brain and frontal gray matter; and frontal, parietal, and temporal white matter [37]. A recent high resonance MRI study (n=156) detected disruption in white matter tracts in patients with deficit (i.e., negative

symptoms) schizophrenia (i.e., right inferior, right arcuate, and uncinate fasciculi) in contrast to nondeficit patients and healthy controls [38]. In addition, there is evidence for accelerated aging in white matter tracts in schizophrenia using DTI-derived fractional anisotropy [39]. These issues are discussed in greater detail in Chap. 6.

Finally, combining various imaging approaches with molecular genetics provides a powerful tool to understand the origins of and the ability to develop more targeted therapeutic interventions for managing schizophrenia [40, 41]. These issues are touched on in Chap. 9.

Endophenotypes

A related area is the identification of *endophenotypes* which may serve as biomarkers for schizophrenia. These are molecular, neuropsychological, neuroimaging, and electrophysiological observations closely associated with the genetic basis of an illness [42]. They represent intermediate links between genetic variation and clinical presentation which can clarify genetically transmitted, brain-based deficits across patients and relatives [43, 44]. To qualify, these biomarkers should meet certain criteria, including heritability, trait stability, test–retest reliability, diagnostic stability, cosegregation of illness within families, and simplicity [45].

One potential example is the dysregulation of GABA inhibitory interneurons during brain maturation in schizophrenia [46, 47]. Another example is neuroimaging in schizophrenia which has identified both structural and functional dysregulation that may serve as endophenotypes for this disease [48]. These include prediction of transition to psychosis in high risk individuals, linking these abnormalities to specific symptoms of schizophrenia, and associating structural changes with clinical outcome.

Recently, the Consortium on Genetics for Schizophrenia reported on 12 regions meeting GWAS significant (e.g., antisaccade task; 3p14) and suggestive (e.g., emotion recognition; 1p36) criteria for potential endophenotypes related to schizophrenia [49]. Yet another area of inquiry has linked abnormal sleep with COMT mutations. Such disturbances during adolescence may impair cortical development leading to deficits in cognitive performance and sleep-related cognitive deficits and may also serve as an endophenotype for schizophrenia [50]. Chapter 5 contains a more complete discussion of these issues.

Management of Schizophrenia

The recent research regarding various environmental influences, neurotransmitters, genetics, neuroimaging, and other biological markers is providing important clues to the pathophysiology of schizophrenia with the hope of better informing our

treatment approaches. Thus, we increasingly recognize the need to employ creative strategies with a variety of therapies to alleviate the multiplicity of symptoms that characterize schizophrenia in both its acute and long-term management. These include the development of novel antipsychotic agents, adjuncts to existing antipsychotics (e.g., glutamatergic), a variety of novel psychotherapeutic interventions (e.g., cognitive remediation), and therapeutic neuromodulation (e.g., transcranial magnetic stimulation).

As our appreciation of the implications regarding clinical response, remission, and recovery also grows, the emphasis is shifting to earlier disease detection and interventions. The study of high risk (HR) states for psychosis focuses on those individuals with prepsychotic or potential prodromal symptoms [51]. In this context, a meta-analysis of 19 studies (1,188 HR patients; 1,029 controls) found evidence for impaired neurocognition and social cognition in this population [52]. In particular, deficits in verbal fluency and memory functioning appeared to increase the risk of transitioning to psychosis.

Such approaches offer hope for a better understanding about the origins of this disorder and the provision of more appropriate services to favorably alter its course. For example, a recent open-trial (n=31) considered the role of low-dose aripiprazole (i.e., up to 7.5 mg/d) in drug-naive or short-exposure patients with ultrahigh risk states or first-episode psychosis [53]. In this study, the use of lower doses was efficacious, but side effects were still prevalent. In another small trial (n=31), the authors found evidence for reduced efficacy with a second exposure to antipsychotics after a first episode, suggesting relapse may be biologically harmful to some patients [54].

Complications Associated with Schizophrenia and Its Treatment

Long-Term Management

Until recently, long-term management in schizophrenia focused on preventing psychotic relapse and helping patients remain out of the hospital. This approach has been reevaluated based on the observation from clinicians and patients that many patients were unable to meet their personal goals. As a result, interventions—both psychosocial and pharmacological—are increasingly focused on improving an individual's vocational and social outcomes as well as the quality of his or her life. New approaches to long-term management also address improving physical health. Individuals with psychotic illness are more vulnerable to premature death due to cardiovascular disease. In addition, obesity and type 2 diabetes are highly prevalent and can interfere with one's health and social adjustment. Decision-making about the selection of a pharmacological treatment should be guided by these health concerns. Moreover, regular interactions between clinicians and patients should focus on modifiable risk factors associated with poor health such as diet, exercise, and smoking. Improving functioning and quality of life almost always requires combining an antipsychotic medication with a psychosocial intervention. As noted in Chap. 14 a number of psychosocial interventions have a strong evidence base. These include symptom-based interventions such as cognitive behavioral therapy for psychosis and treatments which directly affect functioning such as skills training and supported employment. Unfortunately, these treatments are often underutilized and suggest that implementing these practices can potentially improve outcomes.

It is also important to point out that there is strong evidence that psychosocial interventions are only effective when patients are stabilized on an antipsychotic medication [55, 56]. Assuring that individuals with schizophrenia remain on their medications is a continuing challenge during long-term treatment. As noted in Chap. 13, there are a number of approaches to improve adherence, including educating individuals as to why they should remain on medications even when they are feeling well. The recent introduction of new long-acting antipsychotic medications provides additional tools for assisting patients to continue their treatment.

References

- 1. Harvey PD, Heaton RK, Carpenter Jr WT, Green MF, Gold JM, Schoenbaum M. Functional impairment in people with schizophrenia: focus on employability and eligibility for disability compensation. Schizophr Res. 2012;140(1–3):1–8.
- 2. Laursen TM, Munk-Olsen T, Vestergaard M. Life expectancy and cardiovascular mortality in persons with schizophrenia. Curr Opin Psychiatry. 2012;25(2):83–8.
- Tandon R, Nasrallah HA, Keshavan MS. Schizophrenia, "Just the Facts". Part 4: clinical features and concept. Schizophr Res. 2009;110:1–23.
- Tandon R, Carpenter WT. DSM-5 status of psychotic disorders: 1-year pre-publication. Schizophr Bull. 2012;38:369–70.
- American Psychiatric Association. Diagnostic and statistical manual of mental disorders. 5 (DSM-5) th ed. Washington, DC: American Psychiatric Association; 2013.
- McGrath J, Saha S, Welham J, El Saadi O, MacCauley C, Chant D. A systematic review of the incidence of schizophrenia: the distribution of rates and the influence of sex, urbanicity, migrant status and methodology. BMC Med. 2004;2:13.
- 7. Tandon R, Keshavan MS, Nasrallah HA. Schizophrenia, "Just the Facts". What we know in 2008. Part 2: epidemiology and etiology. Schizophr Res. 2008;102:1–18.
- Saha S, Chant D, Welham J, McGrath J. A systematic review of the prevalence of schizophrenia. PLoS Med. 2005;2(5):e141.
- March D, Hatch SL, Morgan C, Kirkbride JB, Bresnahan M, Fearon P, Sussner E. Psychosis and place. Epidemiol Rev. 2008;30:84–100.
- Davies G, Welham J, Chant D, Torey EF, McGrath J. A systematic review and meta-analysis of Northern Hemisphere season of birth studies in schizophrenia. Schizophr Bull. 2003; 29:587–93.
- 11. Myers NL. Update: schizophrenia across cultures. Curr Psychiatry Rep. 2011;13:305–11.
- 12. Brown AS. The environment and susceptibility to schizophrenia. Prog Neurobiol. 2011; 93(1):23–58.
- Nicodemus KK, Marenco S, Batten AJ, Vakkalanka R, Egan MF, Straub RE, Weinberger DR. Serious obstetric complications interact with hypoxia-regulated/vascular expression genes to influence schizophrenia risk. Mol Psychiatry. 2008;13(9):873–7.
- Arseneault L, Cannon M, Poulton R, Murray R, Caspi A, Moffitt TE. Cannabis use in adolescence and risk for adult psychosis: longitudinal prospective study. BMJ. 2002;325:1212–3.

- Large M, Sharma S, Compton MT, Slade T, Nielssen O. Cannabis use and earlier onset of psychosis: a systematic meta-analysis. Arch Gen Psychiatry. 2011;68(6):555–61.
- Foti DJ, Kotov R, Guey LT, Bromet EJ. Cannabis use and the course of schizophrenia: 10-year follow-up after first hospitalization. Am J Psychiatry. 2010;167:987–93.
- 17. Caspi A, Moffitt TE, Cannon M, McClay J, Murray R, Harrington H, Taylor A, Arseneault L, Williams B, Braithwaite A, Poulton R, Craig IW. Moderation of the effect of adolescent-onset cannabis use on adult psychosis by a functional polymorphism in the catechol-O-methyltransferase gene: longitudinal evidence of a gene X environment interaction. Biol Psychiatry. 2005;57(10):1117–27.
- Keshavan MS, Nasrallah HA, Tandon R. Schizophrenia, "Just the Facts" 3. Neurobiology. Schizophr Res. 2008;106:89–107.
- Nielsen MO, Rostrup E, Wulff S, Bak N, Broberg BV, Lublin H, Kapur S, Glenthoj B. Improvement of brain reward abnormalities by antipsychotic monotherapy in schizophrenia. Arch Gen Psychiatry. 2012;69(12):1195–204.
- 20. Li D, He L. Association study between the NMDA receptor 2B subunit gene (GRIN2B) and schizophrenia: a HuGE review and meta-analysis. Genet Med. 2007;9(1):4–8.
- Sato J, Shimazu D, Yamamoto N, Nishikawa T. An association analysis of synapse-associated protein 97 (SAP97) gene in schizophrenia. J Neural Transm. 2008;115(9):1355–65.
- Bosia M, Anselmetti S, Pirovano A, Ermoli E, Marino E, Bramanti P, Smeraldi E, Cavallaro R. HTTLPR functional polymorphisms in schizophrenia: executive functions vs. sustained attention dissociation. Prog Neuropsychopharmacol Biol Psychiatry. 2010;34(1):81–5.
- 23. Hanninen K, Katila H, Saarela M, Rontu R, Mattila KM, Fan M, Hurme M, Lehtimaki T. Interleukin-1 beta gene polymorphism and its interactions with neuregulin-1 gene polymorphism are associated with schizophrenia. Eur Arch Psychiatry Clin Neurosci. 2008;258(1):10–5.
- 24. Chandley MJ, Miller MN, Kwasigroch CN, Wilson TD, Miller BE. Increased antibodies for the α 7 subunit of the nicotinic receptor in schizophrenia. Schizophr Res. 2009;109(1–3):98–101.
- Roth TL, Lubin FD, Sodhi M, Kleinman JE. Epigenetic mechanisms in schizophrenia. Biochim Biophys Acta. 2009;1790(9):869–77.
- Doherty JL, O'Donovan MC, Owen MJ. Recent genomic advances in schizophrenia. Clin Genet. 2012;81(2):103–9.
- Tiwari AK, Zai CC, Muller DJ, Kennedy JL. Genetics in schizophrenia: where are we and what next? Dialogues Clin Neurosci. 2010;12:289–303.
- Levy RJ, Xu B, Gogos JA, Karayiorgou M. Copy number variation and psychiatric disease risk. Methods Mol Biol. 2012;838:97–113.
- 29. Maric NP, Svrakic DM. Why schizophrenia genetics needs epigenetics: a review. Psychiatr Danub. 2012;24(1):2–18.
- Smoller JW, Craddock N, Kendler K, Lee PH, Neale BM, Nurnberger JL, Ripke S, Santangelo S, Sullivan PF. Identification of risk loci with shared effects on five major psychiatric disorders: a genome-wide analysis. Lancet. 2013;381(9875):1371–9.
- 31. Gebicke-Haerter PJ. Epigenetics of schizophrenia. Pharmacopsychiatry. 2012;45 Suppl 1:S42-8.
- Dempster E, Viana J, Pidsley R, Mill J. Epigenetic studies of schizophrenia: progress, predicaments, and promised for the future. Schizophr Bull. 2013;39(1):11–6.
- 33. Boks MP, de Jong NM, Kas MJ, Vinkers CH, Fernandes C, Kahn RS, Mill J, Ophoff RA. Current status and future prospects for epigenetic psychopharmacology. Epigenetics. 2012;7(1):20–8.
- Ruiz S, Birbaumer N, Sitaram R. Abnormal neural connectivity in schizophrenia and fMRIbrain-computer-interface as a potential therapeutic approach. Front Psychiatry. 2013;4:17.
- Fornito A, Yoon J, Zalesky A, Bullmore ET, Carter CS. General and specific functional connectivity disturbances in first-episode schizophrenia during cognitive control performance. Biol Psychiatry. 2011;70:64–72.
- 36. Howes OD, Kambeitz J, Kim E, Stahl D, Slifstein M, et al. The nature of dopamine dysfunction in schizophrenia and what this means for treatment: meta-analysis of imaging studies. Arch Gen Psychiatry. 2012;69(8):776–86.

- Olabi B, Ellison-Wright I, McIntosh AM, Wood SJ, Bullmore E, Lawrie SM. Are there progressive brain changes in schizophrenia? A meta-analysis of structural magnetic resonance imaging studies. Biol Psychiatry. 2011;70:88–96.
- Voineskos AN, Foussias G, Lerch J, Felsky D, Remington G, Rajji TK, Lobaugh N, Pollock BG, Mulsant BH. Neuroimaging evidence for the deficit subtype of schizophrenia. JAMA Psychiatry. 2013;70(5):472–80.
- Kochunov P, Glahn DC, Rowland LM, Olvera RL, Winkler A, et al. Testing the hypothesis of accelerated cerebral white matter aging in schizophrenia and major depression. Biol Psychiatry. 2013;73:482–91.
- Andreasen NC, Nopoulos P, Magnotta V, Pierson R, Ziebell S, Ho BC. Progressive brain change in schizophrenia: a prospective longitudinal study of first-episode schizophrenia. Biol Psychiatry. 2011;70:672–9.
- Iritani S. What happens in the brain of schizophrenia patients? An investigation from the viewpoint of neuropathology. Nagoya J Med Sci. 2013;75(1–2):11–28.
- 42. Ferrarelli F. Endophenotypes and biological markers and of schizophrenia: from biological signs of illness to novel treatment targets. Curr Pharm Des. 2013;19(36):6462–79.
- 43. Light GA, Swerdlow NR, Rissling AJ, Radant A, Sugar CA, Srock J, Pela M, Geyer MA, Braff DL. Characterization of neurophysiologic and neurocognitive biomarkers for use in genomic and clinical outcome studies of schizophrenia. PLoS One. 2012;7(7):e39434.
- 44. Braff DL, Freedman R, Schork NJ, Gottesman II. Deconstructing schizophrenia: an overview of the use of endophenotypes in order to understand a complex disorder. Schizophr Bull. 2007;33(1):21–32.
- 45. Daskalakis ZJ. On a quest for the elusive schizophrenia biomarker. Biol Psychiatry. 2012; 72:714–5.
- Lewis DA, Hashimoto T, Volk DW. Cortical inhibitory neurons and schizophrenia. Nat Rev Neurosci. 2005;6(4):312–24.
- 47. Hasan A, Wobrock T, Grefkes C, et al. Deficient inhibitory cortical networks in antipsychoticnaïve subjects at risk of developing first-episode psychosis and first-episode schizophrenia patients: a cross-sectional study. Biol Psychiatry. 2012;72(9):744–51.
- 48. Ahmed AO, Buckley PF, Hanna M. Neuroimaging schizophrenia: a picture is worth a thousand words, but is saying it anything important? Curr Psychiatry Rep. 2013;15:345.
- 49. Greenwood TA, Swerdlow NR, Gur NR, Cadenhead KS, Calkins ME, Dobie DJ, et al. Genome-wide linkage analyses of 12 Endophenotypes for schizophrenia from the consortium on the genetics of schizophrenia. Am J Psychiatry. 2013;170(5):521–32.
- 50. Tucci V, Lassi G, Kas MJ. Current understanding of the interplay between catechol-Omethyltransferase genetic variants, sleep, brain development and cognitive performance in schizophrenia. CNS Neurol Disord Drug Targets. 2012;11(3):292–8.
- Fusar-Poli P, Borgwardt S, Bechdolf A, Addington J, Riecher-Rossler A, et al. The psychosis high-risk state: a comprehensive state-of-the-art review. JAMA Psychiatry. 2013;70(1): 107–20.
- 52. Fusar-Poli P, Deste G, Smieskova R, Barlati S, Yung AR, et al. Cognitive functioning in prodromal psychosis: a meta-analysis. Arch Gen Psychiatry. 2012;69(6):562–71.
- 53. Liu CC, Chien YL, Hsieh MH, Hwang TJ, Hwu HG, Liu CM. Aripiprazole for drug-naïve or antipsychotic-short-exposure subjects with ultra-high risk state and first-episode psychosis: an open-label study. J Clin Psychopharmacol. 2013;33:18–23.
- Emsley R, Oosthuizen P, Koen L, Niehaus D, Martinez L. Comparison of treatment response in second-episode versus first-episode schizophrenia. J Clin Psychopharmacol. 2013;33:80–3.
- 55. Hogarty GE, Schooler NR, Ulrich R, Mussare F, Ferro P, Herron E. Fluphenazine and social therapy in the aftercare of schizophrenic patients. Relapse analyses of a two-year controlled study of fluphenazine decanoate and fluphenazine hydrochloride. Arch Gen Psychiatry. 1979;36(12):1283–94.
- Hogarty GE, Ulrich RF. The limitations of antipsychotic medication on schizophrenia relapse and adjustment and the contributions of psychosocial treatment. J Psychiatr Res. 1998; 32(3–4):243–50.

Chapter 2 The Evolving Nosology of Schizophrenia: Relevance for Treatment

Rajiv Tandon and Dawn Bruijnzeel

Introduction

The formulation of our current nosology for psychiatric disorders began in the late nineteenth century and culminated in publication of a section related to mental disorders (Section V) in the sixth revision of the International Classification of Disease (ICD-6) in 1949 and in the first edition of the American Diagnostic and Statistical Manual of Mental Disorders (DSM-I) in 1952 [1, 2]. In subsequent revisions of these texts (ICD-7–ICD-10 and DSM-II–DSM-5), substantial changes in diagnostic criteria were made but the basic structure was preserved. Differences between the two systems narrowed and there is now marked concordance between DSM-5 and ICD-10 [3, 4]. DSM-5 became operational in 2013, but ICD-10 is currently being revised and ICD-11 will likely be released in 2016. Schizophrenia is one of the major diagnostic categories in all versions of both manuals.

Although schizophrenia has been studied as a specific disease entity for the past century, its precise nature (core definition, boundaries, causes, and pathogenesis) remains undefined [5, 6]. Since its designation as dementia praecox by Kraepelin (1856–1926) and schizophrenia by Eugen Bleuler (1857–1939), its definitions have varied and its boundaries expanded and receded over the past century (Fig. 2.1) [7, 8]. Thus, it is instructive to examine the varying definitions of schizophrenia over the past 150 years and trace its evolution to the present DSM-5 and soon-to-be released ICD-11 [9, 10].

D. Bruijnzeel

R. Tandon (🖂)

College of Medicine, University of Florida, P.O. Box 100256, Gainesville, FL 32610-0256, USA e-mail: tandon@ufl.edu

Department of Psychiatry, University of Florida College of Medicine, 1600 SW Archer Road, Gainesville, FL 32608, USA e-mail: dheron@ufl.edu



Fig. 2.1 Dimensions of schizophrenia (Adapted with permission from Tandon et al. [29])

The Concept of Schizophrenia from the Nineteenth Century to DSM-IV and ICD-10

The current concept of schizophrenia derives from Emil Kraepelin's formulation of dementia praecox in the late nineteenth century and his elaboration of this idea in the early part of the twentieth century [7, 11]. Until that time, there were two broad prevailing constructs of major psychiatric illness or psychosis. Griesinger postulated that there was only one basic form of psychosis with diverse manifestations attributable to endogenous and environmental factors (einheitpsychose), while others suggested that there were several distinct psychotic disorders (e.g., catatonia, hebephrenia, folie circulaire, dementia paranoides, melancholia) [12–15].

Kraepelin discerned two distinct patterns for the course of illness among the various psychotic disorders and used them to define and classify mental conditions. His approach was influenced by the contemporaneous classification of general paresis of insanity (neurosyphilis or tertiary syphilis). The distinctive course and outcome among patients with this illness in psychiatric hospitals led to its delineation, etiological identification, development of the diagnostic Wasserman test, and later to definitive treatment. Kraepelin concluded that course and outcome of illness are the best disease characteristics to use in defining and demarcating psychiatric disease entities. Based on his study of several hundred cases of hospitalized patients, he delineated two distinct disorders: dementia praecox, which included catatonia, hebephrenia, and paranoid states; and *manic-depressive insanity*, which comprised folie circulaire and melancholia [16]. Kraepelin identified schizophrenia on the basis of its onset (in adolescence or early adulthood), course (deteriorating), and outcome (demence or "mental dullness") [7, 11]. He distinguished dementia praecox from manic-depressive insanity on the basis of the chronicity and poor outcome of the former contrasted with the episodic nature and better outcome of the latter.

During the same time period, Eugen Bleuler renamed this condition "schizophrenia" because he considered the splitting of different psychic functions to be its defining characteristic [8]. He postulated that schizophrenia was defined by a set of basic or fundamental symptoms which were unique and always present in those with this group of diseases, whereas its course and outcome were variable [8, 11]. He called delusions and hallucinations accessory symptoms and considered them to be variable and nonspecific. He described fundamental symptoms (now identified as negative symptoms) which included autism, ambivalence, flat affect, and loosening of associations. He further believed that many mild cases existed and considerably broadened the scope of the disease construct. He included latent and simple forms as part of this entity which he called the group of schizophrenias.

Influenced by the thinking of Karl Jaspers, Kurt Schneider (1887–1967) believed that impairment of empathic communication was the fundamental defect in schizophrenia [17]. He considered "un-understandability" of the personal experience as pathognomonic. Based on this premise, he defined 11 first-rank symptoms which he considered diagnostic of schizophrenia [11, 17]. Given central importance in the Present State Examination, these symptoms were incorporated into both ICD-9 and DSM-III definitions of schizophrenia [18].

Definitions of schizophrenia over the past half-century from DSM-I-DSM-IV and ICD-6-ICD-10 all incorporated Kraepelinian chronicity, Bleulerian negative symptoms, and Schneiderian positive (first-rank) symptoms as part of their definition. The relative emphasis paid to these three roots, however, has varied. In the 1960s and 1970s, there was significant discordance between the DSM (I and II) and ICD (7 and 8) systems with regard to this issue. Whereas DSM-I and DSM-II highlighted the Bleulerian perspective (emphasis on negative symptoms and very broad definition of the schizophrenias, including latent, pseudoneurotic, pseudopsychopathic, and residual subtypes), ICD-7 and ICD-8 stressed Kraepelinian chronicity [1, 19]. This overly broad definition of schizophrenia in the DSM system led to poor reliability in diagnosing schizophrenia and a marked discrepancy between its diagnosis in the USA and the rest of the world which utilized the ICD system [20]. In reaction to these anomalies, the operationalized criteria of DSM-III provided the narrowest definition of schizophrenia with an emphasis on Kraepelinian chronicity and Schneiderian positive (first-rank) symptoms [21]. In fact, the imperative of improved reliability and the belief that positive symptoms could be more reliably diagnosed resulted in positive symptoms becoming the defining characteristic of schizophrenia in DSM-III and ICD-9. Further, certain positive symptoms such as Schneiderian first-rank symptoms were accorded special importance and the presence of any single bizarre delusion or special auditory hallucinations (voices arguing or commenting) was considered sufficient for a diagnosis of schizophrenia in DSM-III and DSM-IV and ICD-9 and ICD-10 [22, 23]. Although there were modest changes in the definition of schizophrenia from DSM-III to DSM-IV, the importance of positive symptoms with the special significance placed on bizarre delusions and special hallucinations in the diagnostic criteria remained.

The Present: DSM-5 and Towards ICD-11

The reliability of diagnosing schizophrenia improved substantially after the introduction of DSM-III in 1980, but several limitations in its definition and description became apparent over the past three decades [24]. Although the significant heterogeneity of schizophrenia was always recognized, with multiple etiological factors and pathophysiological processes, it has been treated as a singular entity [6, 25, 26]. Now, however, it is almost certain that our construct of schizophrenia encompasses not one but several diseases [27]. In the past, we tried to explain the heterogeneity of the illness by defining distinct subtypes-paranoid, disorganized, catatonic, simple, and undifferentiated. These classic subtypes, however, provided a very poor description of the enormous heterogeneity of this condition. Subtypes' stability is low, reliability of diagnosing them weak, their validity questionable, and only the paranoid and undifferentiated subtypes were utilized with some frequency [28, 29]. Because these subtypes have virtually no clinical or research utility, they are being eliminated from the classification system in DSM-5 and ICD-11 [10, 30, 31]. Instead, the heterogeneity of schizophrenia is now formally described in terms of psychopathological dimensions (discussed below) which will be measured in all patients throughout the course of the illness [32]. Since the severity of different symptom domains varies in each patient over the illness course and in response to treatment, such dimensional assessments will be an invaluable tool, allowing clinicians to provide individualized, measurement-based care [33].

Another significant limitation in definitions of schizophrenia until DSM-IV and ICD-10 was the inability to characterize the distinct stages of the illness [34]. The classification system did not allow clinicians to adequately describe the course of their patient's illness (e.g., first episode, chronic) or denote the patient's current clinical status (e.g., active symptoms, in partial or full remission). The provision of modified course specifiers in both DSM-5 and ICD-11 helps to redress this anomaly [10]. Additionally, there was no provision in DSM-IV or ICD-10 to describe the early or late prodromal phases of the illness when there is the possibility of arresting active pathology and preventing disease progression [35, 36]. The addition of Attenuated Psychosis Syndrome as a condition for further study in Section 3 of DSM-5 will partly address this deficiency.

Revisions in DSM-5 and currently proposed changes in ICD-11 will correct the major failure in properly characterizing the clinical heterogeneity of schizophrenia and denoting stages in the evolution of the disease. Additionally, revisions in DSM-5 and ICD-11 correct other discrepancies in the DSM-IV and ICD-10 description of schizophrenia. The four major changes made in DSM-5 (and likely to be made in ICD-11) are separately summarized below.

Changes in Characteristic Symptoms of Schizophrenia in DSM-5 and ICD-11

The changes in the diagnostic criteria of schizophrenia in DSM-5 were modest and continuity with DSM-IV is broadly maintained. No changes were made in the

DSM-IV-TR criterion A	DSM-5 criterion A
Schizophrenia	
Criterion A. Characteristic symptoms:	Criterion A. Characteristic symptoms:
≥2 of the following, each present for a significant portion of time during a 1-month period (or less if successfully treated)	≥2 of the following, each present for a significant portion of time during a 1-month period (or less if successfully treated). At least one of these should include 1–3
1. Delusions	1. Delusions
2. Hallucinations	2. Hallucinations
3. Disorganized speech	3. Disorganized speech
4. Grossly disorganized or catatonic behavior	4. Grossly disorganized or catatonic behavior
5. Negative symptoms, i.e., affective flattening, alogia, or avolition	5. Negative symptoms, i.e., restricted emotional expression or avolition
Note: Only one criterion. A symptom is required if delusions are bizarre or hallucinations consist of a voice keeping up a running commentary on the person's behavior or thoughts, or ≥2 voices conversing with each other	Note: Deleted

Table 2.1 Characteristic symptoms of schizophrenia (Criterion A) in DSM-IV and DSM-5

Source: Reprinted with permission from Tandon R, et al. [51]

minimum duration (6 months), functional impairment (necessary), and exclusion (psychotic mood disorder, schizoaffective disorder, substance-induced psychotic disorder, psychosis secondary to a general medical condition) criteria. Two key changes, however, were made in the definition of characteristic symptoms required in the active phase of the illness (Table 2.1):

- Elimination of special treatment of *bizarre delusions and other Schneiderian first-rank symptoms*. In DSM-IV, only one characteristic symptom was required if it was a bizarre delusion or special hallucination (voices arguing or commenting). Since Schneiderian first-rank symptoms are not pathognomonic for schizophrenia, do not have diagnostic specificity or prognostic significance, and the reliability of distinguishing bizarre from non-bizarre delusions is poor, they will be treated like any other "positive symptom" with regard to their diagnostic implication in DSM-5 and ICD-11 [37–40]. Thus, two characteristic symptoms will be necessary to make a diagnosis of schizophrenia even if one of them is a bizarre delusion or "special hallucination." This change affects a very small number of patients, who will now more appropriately receive a diagnosis of delusional disorder [41].
- Requirement in both DSM-5 and ICD-11 that at least one of the two required symptoms to meet characteristic criterion A be delusions, hallucinations, or disorganized thinking. These are core "positive symptoms" diagnosed with high reliability and might reasonably be considered necessary for a diagnosis of schizophrenia. Less than 1 % of patients with DSM-IV schizophrenia will be affected by this change and in such patients a necessary exploration of other causes of catatonia will be required [41–43].

Elimination of Classic Subtypes of Schizophrenia in DSM-5 and ICD-11

As noted previously, the schizophrenia subtypes had limited diagnostic stability, low reliability, poor validity, and little clinical or research utility. Further, except for the paranoid and undifferentiated subtypes, the others were rarely utilized in most mental healthcare systems. Consequently, these subtypes are eliminated in DSM-5 and ICD-11. This change should simplify clinical description of schizophrenia.

Addition of Psychopathological Dimensions of Schizophrenia

Schizophrenia is characterized by several psychopathological domains which have distinctive courses, patterns of treatment response, and prognostic implications (Box 2.1 and Fig. 2.1). The relative severity of these symptom dimensions varies across patients, as well as within patients at different stages of their illness. Measuring their relative severity through the course of illness and in the context of treatment can provide useful information to the clinician about the nature of the illness in a particular patient and the specific impact of treatment on different aspects of the illness (e.g., analogous to measuring pulse, temperature, blood pressure, respiratory rate). As a simple rating scale (akin to a thermometer or a sphygmomanometer), it should encourage clinicians to explicitly assess and track changes in the severity of these dimensions and to use this information to guide treatment. Six symptom domains with a total of eight items are described in DSM-5 and each item is scored on an anchored 0-4 scale (see Box 2.1) [32]. In addition to their clinical utility, dimensional measurement should prove useful from a research perspective and thereby permit studies on etiology and pathogenesis which cut across current diagnostic categories [44, 45]. Such approaches are consistent with recent findings in genetics and neuroscience and the recent Research Domain Criteria (RDoC) project initiated by the National Institutes of Mental Health [46, 47].

Box 2.1 Dimensions of Psychotic Disorders: Assessment of Severity

All dimensions will be assessed on a 0–4 scale cross-sectionally, with severity assessment based on symptoms in the past 7 days.

- 1. Delusions
- 2. Hallucinations
- 3. Disorganization
- 4. Negative symptoms
- 5. Impaired cognition
- 6. Depression
- 7. Mania
- 8. Psychomotor symptoms including catatonia

Attenuated Psychosis Syndrome

It is believed that the still unsatisfactory outcome of schizophrenia in a significant proportion of individuals with the disorder is due to the identification of the illness and initiation of treatment late in its course after a substantial amount of damage has occurred. The introduction of Attenuated Psychosis Syndrome (Box 2.2) in Section 3 of DSM-5 will support the efforts of clinicians to recognize and monitor psychotic symptoms early in the course of their evolution, and, if necessary, intervene during these crucial stages [35, 36, 48]. Early recognition and intervention are important in other branches of medicine and these changes in DSM-5 should stimulate the development of a similar practice in psychiatry. Although recognition of Attenuated Psychosis Syndrome is important, the ability to reliably diagnose the condition in routine practice is not demonstrated and its nosologic relationship to other psychiatric diagnostic entities is not precisely defined [36, 49]. Consequently, it is not in the main body (Section 2) of the DSM-5 but in Section 3 as a condition needing further study [35]. It should be emphasized that its diagnosis is not an indication for routine antipsychotic treatment but should instead prompt a careful search for comorbidities (e.g., anxiety, depression, substance use disorder) and their appropriate treatment along with close monitoring for a possible transition to psychosis.

Box 2.2 Criteria for Attenuated Psychosis Syndrome in DSM-5 [3]

All six of the following:

- 1. *Characteristic symptoms*: at least one of the following in attenuated form with intact reality testing, but of sufficient severity and/or frequency that it is not discounted or ignored.
 - (a) Delusions
 - (b) Hallucinations
 - (c) Disorganized speech
- 2. *Frequency/currency*: symptoms meeting criterion A must be present in the past month and occur at an average frequency of at least once per week in past month.
- 3. *Progression*: symptoms meeting criterion A must have begun in or significantly worsened in the past year.
- 4. *Distress/disability/treatment seeking*: symptoms meeting criterion A are sufficiently distressing and disabling to the patient and/or parent/guardian to lead them to seek help.
- Exclude other DSM-5 disorder: symptoms meeting criterion A are not better explained by any DSM-5 diagnosis, including substance-related disorder.
- 6. *Exclude prior psychotic disorder*: clinical criteria for any DSM-5 psychotic disorder have never been met.

Treatment Implications of Current Conceptualization of Schizophrenia: 2014 and Beyond

For the past century, schizophrenia was conceptualized as a singular disorder with a distinctive pathology, albeit with multiple etiological factors and diverse manifestations. This formulation dictated a single specific treatment directed at the common pathology (i.e., antipsychotic medications). These agents, however, are ineffective against some core features of the illness, such as negative symptoms and cognitive deficits [50]. Although researchers assiduously searched for an alternative core pathology in schizophrenia, no single mechanism has been found and no single treatment is effective against all aspects of the illness. With DSM-5, schizophrenia is explicitly conceptualized as a multidimensional illness with each dimension potentially explained by a distinct pathophysiology and the target of a unique treatment [51].

In DSM-5, six distinct dimensions of schizophrenia are defined (Fig. 2.1). Whereas positive symptoms and mood symptoms have two items (delusions and hallucinations for positive symptoms, and depression and mania for mood symptoms), the other four dimensions have one item (Box 2.1). Each item is scored on an anchored 0–4 scale (Section 3 of DSM-5) [3]. Different schizophrenia patients exhibit varying admixtures of symptom severity across the six dimensions. Additionally, these dimensions respond differently to various treatments and different patients show dissimilar patterns of treatment response. Unique measurement of how each of these distinct dimensions responds to specific treatments in individual patients enables monitoring of response and appropriate treatment adjustments. The availability of a simple scale which provides measures of symptom severity across these distinct dimensions enables measurement-based, individualized treatment.

Antipsychotic medications are particularly effective for treating positive symptoms and disorganization in schizophrenia. Specific treatments are in development to address negative symptoms and cognitive deficits in schizophrenia. Hopefully, some of these will be effective and safe, becoming available for clinical practice in the near future. Antidepressants and mood stabilizers are moderately effective for the mood symptoms in schizophrenia. Antipsychotic medications, benzodiazepines, and electroconvulsive therapy all demonstrate efficacy for motor symptoms (including catatonia) in schizophrenia. As better treatments for various symptom dimensions are developed, more effective, "evidence-based" pharmacotherapy will be possible.

It is clear that schizophrenia is not one but several disorders. The delineation of unique psychopathological dimensions facilitates the identification of distinct etiopathophysiological pathways to schizophrenia; development of specific biological tests for the "different schizophrenias"; elaboration of specific treatments for the "different schizophrenias"; construction of an etio-pathophysiological nosology of the schizophrenias; and most importantly, better outcomes for our patients [47, 52–54]. As DSM-5 provides a better bridge to this anticipated future, it also provides tools for more effective treatment of schizophrenia today [24, 45, 55].

References

- 1. World Health Organization. Manual of the international statistical classification of diseases, injuries and causes of death. Sixth Revision (ICD-6)th ed. Geneva: World Health Organization; 1949.
- 2. American Psychiatric Association. Diagnostic and statistical manual of mental disorders. 1 (DSM-I)th ed. Washington, DC: American Psychiatric Association; 1952.
- American Psychiatric Association. Diagnostic and statistical manual of mental disorders. 5 (DSM-5)th ed. Washington, DC: American Psychiatric Association; 2013.
- World Health Organization. The ICD-10 classification of mental and behavioral disorders: clinical descriptions and diagnostic guidelines (CDDG). Geneva: World Health Organization; 1992.
- Tandon R, Maj M. Nosological status and definition of schizophrenia. Some considerations for DSM-V and ICD-11. Asian J Psychiatr. 2008;1:22–7.
- 6. Tandon R, Keshavan MS, Nasrallah HA. Schizophrenia, "just the facts". What we know in 2008. Part 1: overview. Schizophr Res. 2008;100:4–19.
- 7. Kraepelin E. Dementia Praecox and Paraphrenia, 1919. In: Robertson GM, editor. New York: Krieger; 1971.
- Bleuler E. Dementia Praecox, or the group of Schizophrenias, 1911, translated by J. Zinkin. New York: International University Press; 1950.
- 9. Tandon R, Carpenter WT. DSM-5 status of psychotic disorders: 1-year pre-publication. Schizophr Bull. 2012;38:369–70.
- Gaebel W, Zielasek J, Cleveland H-R. Psychotic disorders in ICD-11. Psychiatry. 2013;10:11–7.
- 11. Hoenig J. The concept of schizophrenia. Kraepelin-Bleuler-Schneider. Br J Psychiatry. 1983;142:547–56.
- 12. Berrios GE, Beer M. The notion of unitary psychosis: a conceptual history. Hist Psychiatry. 1994;5:13–36.
- Barnes MP, Saunders M, Walls TJ, Saunders I, Kirk CA. The syndrome of Karl Ludwig Kahlbaum. J Neurol Neurosurg Psychiatry. 1986;41:991–6.
- 14. Sedler MJ. Falret's discovery (1854): the origin of the concept of bipolar affective illness, translated by MJ Sedler and Eric C Dessain. Am J Psychiatry. 1983;140:1127–33.
- 15. Sedler MJ. The legacy of Ewald Hecker, a new translation of Die Hebephrenie, translated by Mary Louise Schoelly. Am J Psychiatry. 1985;142:1265–71.
- Jablensky A, Hugler H, von Cranach M, Kalinov K. Kraepelin revisited: a reassessment and statistical analysis of dementia praecox and manic-depressive insanity in 1908. Psychol Med. 1993;23:843–58.
- 17. Schneider K. Clinical psychopathology, Translated by Hamilton MW. New York: Grune and Stratton; 1959.
- Wing JK, Cooper JE, Sartorius N. The measurement and classification of psychiatric symptoms. Cambridge: Cambridge University Press; 1974.
- 19. American Psychiatric Association. Diagnostic and statistical manual of mental disorders. 2 (DSM-II)th ed. Washington, DC: American Psychiatric Association; 1968.
- Kendell RD, Cooper JR, Gourlay AJ, et al. Diagnostic criteria of American and British psychiatrists. Arch Gen Psychiatry. 1971;25:123–30.
- American Psychiatric Association. Diagnostic and statistical manual of mental disorders.
 3 (DSM-III)th ed. Washington, DC: American Psychiatric Association; 1980.
- 22. Mellor CS. First rank symptoms of schizophrenia. Br J Psychiatry. 1970;117:15-23.
- 23. Wing JK, Nixon J. Discriminating symptoms in schizophrenia: a report from the International Pilot Study of Schizophrenia. Arch Gen Psychiatry. 1975;32:953–9.
- 24. Tandon R. The nosology of schizophrenia: towards DSM-5 and ICD-11. Psychiatr Clin North Am. 2012;35:555–69.

- 25. Wyatt RJ, Alexander RC, Egan MF, Kirch DG. Schizophrenia, just the facts. what do we know, how well do we know it. Schizophr Res. 1988;1:3–18.
- Tandon R. Moving beyond findings: concepts and model-building in schizophrenia. J Psychiatr Res. 1999;29:255–60.
- 27. Keshavan MS, Nasrallah HA, Tandon R. Schizophrenia, "Just the Facts" 6. Moving ahead with the schizophrenia concept: from the elephant to the mouse. Schizophr Res. 2011;127:3–13.
- Helmes E, Landmark J. Subtypes of schizophrenia: a cluster analytic approach. Can J Psychiatry. 2003;48:702–8.
- Tandon R, Nasrallah HA, Keshavan MS. Schizophrenia, "Just the Facts". Part 4: clinical features and concept. Schizophr Res. 2009;110:1–23.
- Braff DL, Ryan J, Rissling AJ, Carpenter WT. Lack of use in the literature in the last 20 years supports dropping traditional schizophrenia subtypes from DSM-5 and ICD-11. Schizophr Bull. 2013;39(4):751–3.
- Tandon R, Gaebel W, Barch DM, et al. Definition and description of schizophrenia in DSM-5. Schizophr Res. 2013;150:3–10.
- 32. Barch DM, Bustillo J, Gaebel W, et al. Logic and justification for dimensional assessment of symptoms and related phenomena in psychosis: relevance to DSM-5. Schizophr Res. 2013;150:15–20.
- Tandon R. Schizophrenia spectrum and other psychotic disorders: DSM-5 revisions and their clinical implications. Psychopharm Rev. 2013;48:33–40.
- McGorry PD. Risk syndromes, clinical staging, and DSM V: new diagnostic infrastructure for early intervention in psychiatry. Schizophr Res. 2010;120:49–53.
- Tsuang MT, van Os J, Tandon R, et al. Attenuated psychosis syndrome in DSM-5. Schizophr Res. 2013;150:31–5.
- 36. Tandon N, Shah J, Keshavan MS, Tandon R. Attenuated psychosis and the schizophrenia prodrome: current status of risk identification and psychosis prevention. Neuropsychiatry. 2012;2:345–53.
- Carpenter WT, Strauss JS, Muleh S. Are there pathognomonic symptoms in schizophrenia. An empiric investigation of Schneider's first-rank symptoms. Arch Gen Psychiatry. 1973; 28:847–52.
- Nordgard J, Arnfred SM, Handest P, Parnas J. The diagnostic status of first-rank symptoms. Schizophr Bull. 2008;34:137–54.
- 39. Ihara K, Morgan C, Fearon P, et al. The prevalence, diagnostic significance, and demographic characteristics of Schneiderian first-rank symptoms in an epidemiological sample of first-episode psychosis. Psychopathology. 2009;42:81–91.
- Bell V, Halligan PW, Ellis HD. Diagnosing delusions: a review of inter-rater reliability. Schizophr Res. 2006;86:76–9.
- 41. Tandon R, Bruijnzeel D, Rankupalli B. Does change in definition of psychotic symptoms in diagnosis of schizophrenia in DSM-5 affect caseness? Asian J Psychiatr. 2013;6:330–2.
- Heckers S, Tandon R, Bustillo J. Catatonia in the DSM. Shall we move or not? Schizophr Bull. 2010;36:205–7.
- 43. Tandon R, Heckers S, Bustillo J, et al. Catatonia in DSM-5. Schizophr Res. 2013;150:26–30.
- 44. Tandon R, Keshavan MS, Nasrallah HA. Schizophrenia, "Just the Facts". What we know in 2008. Part 5: treatment and prevention. Schizophr Res. 2010;122:1–23.
- 45. Tandon R, Targum SD, Nasrallah HA, Ross R. Strategies for maximizing clinical effectiveness in the treatment of schizophrenia. J Psychiatr Pract. 2006;12:348–63.
- 46. Insel T, Cuthbert B, Garvey M, et al. Research Domain Criteria (RDoC): toward a new classification framework in mental disorders. Am J Psychiatry. 2010;167:748–51.
- 47. Insel TR. Rethinking schizophrenia. Nature. 2010;468:187-93.
- Woods SW, Walsh BC, Saksa JR, McGlashan TH. The case for including attenuated psychotic symptoms syndrome in DSM-5 as a psychosis risk syndrome. Schizophr Res. 2010;123: 199–207.
- Regier DA, Narrow WE, Clarke DE, et al. DSM-5 field trials in the United States and Canada, Part II: test-retest reliability of selected categorical diagnoses. Am J Psychiatry. 2013;170:59–70.

- 50. Tandon R. Antipsychotics in the treatment of schizophrenia: an overview. J Clin Psychiatry. 2011;72 Suppl 1:4–8.
- 51. Tandon R, Carpenter WT. Psychotic disorders in DSM-5. Psychiatry. 2013;10:5-9.
- 52. Kapur S, Phillips AG, Insel TR. Why has it taken so long for biological psychiatry to develop clinical tests and what to do about it? Mol Psychiatry. 2012;17:1174–9.
- 53. Hyman SE. The diagnosis of mental disorders: the problem of reification. Annu Rev Clin Psychol. 2010;6:155–79.
- Kupfer DJ, Regier DA. Neuroscience, clinical evidence, and the future of psychiatric classification in DSM-5. Am J Psychiatry. 2011;168:672–4.
- 55. Tandon R. Definition of schizophrenia in the DSM-5: too radical, too conservative, or just right. Schizophr Res. 2013;150:1–2.

Part II Recent Research into the Pathophysiology of Schizophrenia
Chapter 3 Overview of Neurobiology

Rajiv Tandon and Morris Goldman

we must strive to divide things by classes, where the natural joints are, and not try to break them apart after the manner of a bad carver

Plato, Phaedrus 265e

Introduction

In a *Nature* Perspective called "Rethinking Schizophrenia," the head of the National Institutes of Mental Health, Thomas R. Insel, acknowledged, "For schizophrenia our knowledge base in 2010 is mostly based on clinical observation" [1]. He observed "we still do not have a basic understanding of the pathophysiology of the disorder and therefore lack the tools for curative treatment." How, despite the several thousand scientific articles per year on schizophrenia reporting myriad evidence of brain dysfunction at multiple levels of analysis, has this situation arisen? A recent review noted that the situation is quite analogous to the blind men probing different parts of an elephant, each coming up with their own provisional assessment based on what they perceive [2]. In a subsequent article, Insel and others go on to identify some of the issues which in part are reflected in the revisions to the nosology of psychiatric disorders (i.e., DSM-5, ICD-10); the renewed effort to integrate pathologic findings into disordered brain functioning; and current priorities of funding agencies (i.e., R-Doc) [3]. These include:

R. Tandon (🖂)

College of Medicine, University of Florida, P.O. Box 100256, Gainesville, FL 32610-0256, USA e-mail: tandon@ufl.edu

M. Goldman

Department of Psychiatry and Behavioral Sciences, Northwestern University, East Ontario 446, Chicago, IL 60611, USA e-mail: m-goldman@northwestern.edu

- 1. Categorizing illnesses by reliably identifying collections of symptoms based on longitudinal observations (e.g., DSM-III, DSM IV) may be counterproductive. Current evidence suggests that important biologic findings cross boundaries to the extent that persons with different disorders may share a common pathophysiology while the pathophysiology of different persons with the same disorder may be completely different. This should not be so surprising, given the complexity of the human mind and the many steps from genes to molecular mechanisms to neural circuits to behavior and psychopathology.
- 2. Very few neurobiological findings in schizophrenia have been truly replicated. For example, over 1,000 genes and 8,000 polymorphisms, as well as 25 imaging measures, are linked to schizophrenia. Non-replications and approximate replications are common. A major problem is the lack of an adequate sample size to account for the multiple comparisons intrinsic to studies of genes and brain structure when not guided by a specific hypothesis tested by a specific outcome measure. This provides opportunities for "approximate replications" which appear to (pseudo-) confirm a previous finding because so many different measures related to the original finding are obtained in the follow-up study.
- 3. Schizophrenia is a heterogeneous disorder. While this view has been asserted for decades and generally accepted by the neuroscience community, it does not have a major impact on research beyond leading to larger patient samples and multi-site studies. Among the reasons for not acknowledging this reality are that previous efforts to categorize the disorder have proven to be of no utility, and that it is much more difficult to balance groups of schizophrenia patients (compared with balancing groups of healthy controls). Over time, a general consensus developed that schizophrenia must be reconceptualized and investigators must "deconstruct the concept of schizophrenia, and then reconstruct the defined elements." Specifically, heterogeneity in clinical expression must be resolved by mapping phenotypic dimensions to markers of biological processes (intermediate phenotypes or endophenotypes) and eventually to causative etiological factors [4]. Precisely how this will be accomplished is not yet established, though several possible routes are proposed.
- 4. *The vast majority of studies compare a group of patients with a group of healthy individuals.* One does not need a biological test in most cases to make this distinction. These "extreme" comparisons are of little use to clinicians and do not advance our understanding of the illness(es). Furthermore, finding a biological difference between patients and controls does not tell us whether that difference reflects pathology, a restitutive process, or an epiphenomenon (i.e., *a finding is not necessarily pathology*).

In the remainder of this chapter, we discuss some of the implications which arise from these four critical issues. In particular, we point out how this recognition challenges basic assumptions guiding neuroscience research. When possible, we provide examples demonstrating the potential relevance to studying the neurobiology of schizophrenia.

Identifying Findings Is Not the Same as Identifying an Abnormality

Part of the logic of using healthy persons as controls is the belief that findings that cannot be accounted for by controlling for demographic (e.g., age, sex, race) and treatment factors (drug exposure, years of illness) are abnormalities. Surprisingly, with few exceptions, this assumption is rarely challenged, and one can easily identify examples by scanning the content of most clinic neuroscience journals [2]. The brain is a remarkably adaptive organ, and most activity is likely compensatory, perhaps designed around maintaining a specific psychological state(s) [5, 6]. It is likely that the vast majority of findings at all levels of analysis in schizophrenia are not abnormalities but instead reflect adaptive efforts which are quite distant from the initial insult(s).

To Determine If a Finding Reflects Pathology, Learn About Its Regulation

To determine whether a "finding" is adaptive (e.g., physiological response) or pathological, one needs to understand what normally regulates it. Often, this information is ignored even if available. For instance, many studies were interested in the possible association between plasma levels of the neurophysiologic peptides arginine vasopressin and oxytocin and severe mental illness [7]. Vasopressin serves primarily as the antidiuretic hormone maintaining plasma tonicity in a narrow range and plasma levels are largely determined by plasma tonicity (i.e., sodium concentration, osmolality). Of the 15 psychiatric studies published from 2000 to 2006 in which arginine vasopressin was the major outcome variable, only six provided a measure of plasma tonicity. Only two reported any of the other easily obtainable measures which modulate vasopressin levels (e.g., blood pressure, smoking, cortisol) and only one acknowledged that any of this was of possible relevance. Indeed, one asserted that "plasma osmolality was not measured because it is not associated with plasma vasopressin in depression" [8]. It seems quite possible that most biological functioning, even that related to severe mental illness, is only subtly disrupted, and that it may be very difficult to isolate even fundamental abnormalities if one does not concurrently measure factors known to modulate them. Thus, clinical researchers are well advised to measure confounding factors outside of the demographic and treatment-related variables usually assessed in current studies, especially if they are known to modulate the outcome variable of interest. It is quite possible, as discussed below, that follow-up studies assessing these modulatory influences might lead to a delineation of mechanisms and a better understanding of pathogenesis.

Some Abnormalities May Be Epiphenomena But Still Provide Clues About Mechanisms

While most of what goes on in the brain likely reflects compensatory processes, there still may be considerable "collateral damage" which has little to do with the mental illness. Such information may, however, provide clues to help identify the pathophysiology. This approach seems particularly promising for limbic functions which cover multiple domains, some of which can be reliably assessed (e.g., neurophysiologic, behavioral) and others which are more obscure (e.g., affect, cognition). For instance, a sequence of studies identified that polydipsic schizophrenia patients have subtle impairments in peripheral vasopressin regulation which are attributable to hippocampal volume loss but are unrelated to the mental illness [9, 10]. Concurrent work, however, suggests that this abnormality is closely associated with a generalized stress diathesis, as well as other neuroendocrine findings which contribute to these patients' social deficits [11]. Thus, while an abnormal finding per se may not contribute to a mental illness, it could direct you to the right "neighborhood."

Heterogeneity Obscures Findings by Mixing Patients with Deficient and Adaptive Function

The schizophrenia literature is full of studies showing patients are both higher and lower on a given neurobiologic measure, or that groups of patients exhibit a larger standard deviation than controls. While these results are often attributable to either small sample sizes, a failure to control for confounding variables or nonspecific factors associated with the mental illness; they may also reflect real differences which warrant further exploration. Measures of cortisol responses to psychological stress provide a good example of this. It is widely appreciated that patients with schizophrenia exhibit both hypo- and hypercortisolemia, and specifically elevated and blunted responses to psychological stress [12–14]. While some findings may reflect acuity of the illness or confounding effects of antipsychotic medications on the hypothalamo-pituitary axis, others may reflect real differences critical to understanding the mental disorder [15]. Schizophrenia patients experience more stressful events than controls which in healthy persons would blunt their responses to future stressors [16–18]. Furthermore, this blunting is proportional to indices of psychological resilience and the integrity of the hippocampus [19, 20]. Indeed, Brenner found that most schizophrenia patients with blunted responses exhibit evidence of greater resilience consistent with the response in healthy persons. Others, however, exhibited enhanced responses unaffected by the subject's psychological resilience [21]. These observations may be particularly relevant to understanding differences between polydipsic and nonpolydipsic schizophrenia patients. Polydipsic patients exhibit enhanced responses and nonpolydipsic patients exhibit blunted cortisol responses to psychological stress [22]. The polydipsic enhanced responses are proportional to deformations in hippocampal structure not seen in those without polydipsia, while the blunted responses in nonpolydipsic patients may be linked to their higher level of functioning [10, 14]. Thus, this example demonstrates how failing to distinguish groups of schizophrenia patients whose findings "cancel each other out" potentially obscures data critical to understanding the pathophysiology of the illness for one group (i.e., hippocampal-modulated stress diathesis) and a potential treatment intervention for another (i.e., teaching coping skills to enhance resilience).

Compensatory Responses May Identify Potential Treatment Interventions for Other Patients

The identification of a particular "finding" in schizophrenia as a compensatory response to some known or unknown pathological insult might lead to a potentially effective treatment. For example, compensatory increased cholinergic activity during an acute psychotic phase of schizophrenia provides some of the impetus for the current interest in the evaluation of muscarinic and nicotinic cholinergic agents to treat schizophrenia (see Chap. 7) [23, 24].

How Does One Relate Underlying Neurobiology to Treatment of Schizophrenia

Despite the note of caution emphasized in this chapter, even the limited current understanding of the neurobiology of schizophrenia is still of value to the clinician. The range of etiological and pathophysiological factors considered relevant to the disorder is reviewed in the next three chapters. Many of the most consistent neurobiological findings are summarized in Chap. 4. Genetic findings in schizophrenia and the best way to understand their relevance are reviewed in Chap. 5. Neuroimaging findings in schizophrenia and their current and future clinical application are considered in Chap. 6. Although very little of this information has provided useful clinical tools to-date, changes in our current research paradigm warrant optimism for the future which is likely to expand clinical applicability because of both methodological and conceptual advances. Neurobiological research is also likely to benefit from improvements in nosology, with further refinements in the DSM criteria and incorporation of objective measures [25]. As neuroimaging and electrophysiological technologies mature, dysfunctions are more likely to become apparent. Better integration of neuroimaging, neurochemical and electrophysiological data, as well as the combined use of biomarkers may be more useful than single markers used in isolation. Multimodal imaging (e.g., combining fMRI with ERP data) may also be more useful than individual modalities alone. Better delineation of endophenotypic markers and susceptibility genes are likely to yield more valid animal models for further hypothesis testing. Larger, multisite studies should improve statistical power to confirm or refute the tantalizing observations currently limited by small sample sizes.

Conclusion

The past three decades of research in the neurobiology of schizophrenia point to brain structural, functional, and neurochemical alterations in multiple regions and their connecting circuitries. These alterations appear to begin early in development and evolve during the course of the illness, suggesting a sequential derailment of developmental processes. While the precise meaning of these brain changes in schizophrenia currently remains unclear, conceptual and methodological refinements augur well for a better understanding in the near future.

References

- 1. Insel TR. Rethinking schizophrenia. Nature. 2010;468(7321):187-93.
- 2. Keshavan MS, Nasrallah HA, Tandon R. Moving ahead with the schizophrenia concept: from the elephant to the mouse. Schizophr Res. 2011;127(1–3):3–13.
- 3. Kapur S, Phillips AG, Insel TR. Why has it taken so long for biological psychiatry to develop clinical tests and what to do about it? Mol Psychiatry. 2012;17(12):1174–9.
- 4. Tandon R, Nasrallah HA, Keshavan MS. Schizophrenia, "just the facts" 4. Clinical features and conceptualization. Schizophr Res. 2009;110(1–3):1–23.
- 5. Nava E, Röder B. Adaptation and maladaptation insights from brain plasticity. Prog Brain Res. 2011;191:177–94.
- Karatsoreos IN, McEwen BS. Psychobiological allostasis: resistance, resilience and vulnerability. Trends Cogn Sci. 2011;15:576–84.
- Meyer-Lindenberg A, Domes G, Kirsch P, Heinrichs M. Oxytocin and vasopressin in the human brain: social neuropeptides for translational medicine. Nat Rev Neurosci. 2011;12(9):524–38.
- 8. Goldman MB. The mechanism of life-threatening water imbalance in schizophrenia and its relationship to the underlying psychiatric illness. Brain Res Rev. 2009;61(2):210–20.
- Goldman MB, Luchins DJ, Robertson GL. Mechanisms of altered water metabolism in psychotic patients with polydipsia and hyponatremia. N Engl J Med. 1988;318(7):397–403.
- Goldman MB, Wang L, Wachi C, Daudi S, Csernansky JG, Marlow-O'Connor M, et al. Structural pathology underlying neuroendocrine dysfunction in schizophrenia. Behav Brain Res. 2011;218(1):106–13.
- 11. Goldman MB, Gomes AM, Carter CS, Lee R. Divergent effects of two different doses of intranasal oxytocin on facial affect discrimination in schizophrenic patients with and without polydipsia. Psychopharmacology (Berl). 2011;216(1):101–10.
- 12. Bradley AJ, Dinan TG. A systematic review of hypothalamic-pituitary-adrenal axis function in schizophrenia: implications for mortality. J Psychopharmacol. 2010;24 Suppl 4:91–118.
- Tandon R, Mazzara C, DeQuardo JR, Craig KA, Meador-Woodruff JH, Goldman R, et al. Dexamethasone suppression test in schizophrenia: relationship to symptomatology, ventricular enlargement, and outcome. Biol Psychiatry. 1991;29:953–64.

3 Overview of Neurobiology

- Brenner K, St-Hilaire A, Liu A, Laplante DP, King S. Cortisol response and coping style predict quality of life in schizophrenia. Schizophr Res. 2011;128:23–9.
- 15. Tandon R, Halbreich U. The second-generation "atypical" antipsychotics: similar efficacy but different neuroendocrine side-effects. Psychoneuroendocrinology. 2003;28(1):1–7.
- Myin-Germeys I, Krabbendam L, Delespaul P, van Os J. Do life events have their effect on psychosis by influencing the emotional reactivity to daily life stress? Psychol Med. 2003; 33:327–33.
- Elzinga BM, Roelofs K, Tollenaar MS, Bakvis P, van Pelt J, Spinhoven P. Diminished cortisol responses to psychosocial stress associated with lifetime adverse events: a study among healthy young subjects. Psychoneuroendocrinology. 2008;33:227–37.
- Miller GE, Chen E, Zhou ES. If it goes up, must it come down? Chronic stress and the hypothalamic—pituitary—adrenocortical axis in humans. Psychol Bull. 2007;133:25–45.
- Kudielka BM, Hellhammer DH, Wüst S. Why do we respond so differently? Reviewing determinants of human salivary cortisol responses to challenge. Psychoneuroendocrinology. 2009;34:2–18.
- Delgado y Palacios R, Campo A, Henningsen K, Verhoye M, Poot D, Dijkstra J, et al. Magnetic resonance imaging and spectroscopy reveal differential hippocampal changes in anhedonic and resilient subtypes of the chronic mild stress rat model. Biol Psychiatry. 2011;70:449–57.
- Brenner K, Liu A, Laplante DP, Lupien S, Pruessner J, Ciampi A, et al. Cortisol response to a psychosocial stressor in schizophrenia: blunted, delayed, or normal?Psychoneuroendocrinology. 2009;34(6):859–68.
- Goldman MB, Gnerlich J, Hussain N. Neuroendocrine responses to a cold pressor stimulus in polydipsic hyponatremic and in matched schizophrenic patients. Neuropsychopharmacology. 2007;32(7):1611–21.
- Tandon R, Taylor SF, DeQuardo JR, et al. The cholinergic system in schizophrenia reconsidered: anticholinergic modulation of sleep and symptom profiles. Neuropsychopharmacology. 1999;21 Suppl 2:189–201.
- Tandon R, DeQuardo JR, Taylor SF, et al. Phasic and enduring negative symptoms in schizophrenia: biological markers and relationship to outcome. Schizophr Res. 2000;45:191–201.
- Tandon R, Carpenter WT. DSM-5 status of psychotic disorders. Schizophr Bull. 2012; 38(3):369–70.

Chapter 4 Pathophysiology of Schizophrenia

Jaya Padmanabhan and Matcheri S. Keshavan

Introduction

The neurobiological basis of schizophrenia has been suspected for over a century, beginning with Kraepelin's description of *dementia praecox* [1]. Unlike other neurological disorders such as Alzheimer's disease, however, few clear-cut neuropathological observations emerged. This led to an eminent neurologist's comment that "schizophrenia is the graveyard of neuropathologists" [2]. Scientific research in this area stagnated for much of the early twentieth century as psychodynamic theories dominated our understanding of mental illness. Beginning in the 1960s, however, neurobiological approaches to schizophrenia reemerged, starting with the landmark finding of ventricular enlargement [3].

As Weinberger stated, the challenge of finding verifiable brain changes in schizophrenia no longer exists given the advent of sophisticated neuroimaging, electrophysiological, and neuropathological techniques over the past three decades which led to an improved, albeit incomplete, understanding of the pathophysiology of this illness [4]. Thus, we review the current understanding of the neurobiology of schizophrenia with a focus on the substantive body of literature largely accumulated over the past 30 years. We conclude with a summary of the prevailing theoretical models of its pathophysiology, gaps in our knowledge, and promising directions for future research (Box 4.1).

M.S. Keshavan

J. Padmanabhan (🖂)

Department of Psychiatry, Beth Israel Deaconess Medical Center and Massachusetts Mental Health Center, 75 Fenwood Road, Boston, MA 02115, USA e-mail: jpadmana@bidmc.harvard.edu

Department of Psychiatry, Beth Israel Deaconess Medical Center and Massachusetts Mental Health Center, Harvard Medical School, 330 Brookline Avenue, Boston, MA 02215, USA e-mail: keshavams@gmail.com; mkeshava@bidmc.harvard.edu

Box 4.1 Key Facts in the Pathophysiology of Schizophrenia

- Structural brain findings:
 - Ventricular enlargement [3]
 - Subtle reductions in total gray matter volume [6]
 - Reductions in gray matter volume of the hippocampus and other medial temporal and limbic regions [7]
- Functional brain findings:
 - Decreased activation of prefrontal cortex (hypofrontality) [27]
 - Increased activation of temporal regions during hallucinations [31]
- Electrophysiological findings:
 - Diminished prepulse inhibition of startle response (PPI) [39] and diminished P50 suppression [46]
 - Decreased amplitudes of the P300 response [34] and mismatch negativity [54]
 - Abnormalities in gamma oscillations [58]
- Neuroendocrine, Oxidative, and Immunological
 - Elevated markers of oxidative stress, varying by clinical status [62]
 - Dysfunction of the hypothalamic–pituitary–adrenal axis (abnormal dexamethasone suppression) [74, 131]
 - Abnormal levels of inflammatory cytokines [68]
- Neuropathology
 - Reductions in dendritic spines and size of pyramidal neurons [98]
 - Relative preservation of total number of neurons [97]
 - The absence of gliosis and other neurodegenerative features [89, 97]
 - Reduced expression of GAD-67 in the dorsolateral prefrontal cortex [88]
- Neurochemical
 - Reduced N-acetyl aspartate in frontal and temporal regions [77]
 - Reduced PME (marker of membrane phospholipid synthesis) in prefrontal regions [80]
 - Elevated presynaptic dopamine function [83]

Brain Structure

Gray Matter

There are a number of well-established structural brain alterations in schizophrenia. The most consistent findings include enlargement of the third and lateral ventricles and slight reductions in total brain volume and total gray matter volume [5–7].

Additionally, regional reductions are reported consistently in the hippocampus, amygdala, parahippocampus, superior temporal gyrus, anterior cingulate, insula, and inferior and medial frontal gyri [7, 8].

Patients with first-episode psychosis also exhibit volumetric reductions in whole brain and hippocampus; enlargement in the third and lateral ventricles; and longitudinal loss of total cortical gray matter and gray matter in frontal, temporal, and parietal regions [9, 10]. Although further study is required, first-generation antipsychotic (FGA) exposure appears to correlate with enlargement of the caudate over time, while cumulative antipsychotic exposure may be associated with gray matter loss over time [11, 12].

Alterations are observed in other aspects of structural morphology. These include reductions in cortical thickness of the temporal and frontal lobes and reduced cortical surface area in multiple regions [13, 14]. Cerebral asymmetry is also reduced in both affected individuals and healthy relatives, particularly in the planum temporale which shows diminished left–right asymmetry in patients with schizophrenia compared with healthy controls [15].

Structural measures show some correlations with severity of symptoms and cognitive deficits. The most consistent finding is the association between gray matter reductions of the superior temporal gyrus and positive symptom severity, specifically hallucinations [16]. Hippocampal volume reduction is correlated with greater severity of positive and negative symptoms and poorer social function [17]. Prefrontal alterations are associated with impaired executive function, while temporal and hippocampal structural abnormalities correlate with deficits in performance speed, working memory, and abstraction [18].

High-risk individuals (variously defined as individuals with prodromal symptoms or a family history of schizophrenia) also exhibit reduced volume in structures such as the insula, superior temporal gyrus, and cingulate [19, 20]. Conversion to psychosis among high-risk individuals is associated with relative gray matter deficits in frontal, temporal, and parahippocampal regions, although this finding is not consistent [20, 21]. Nonpsychotic relatives of individuals with schizophrenia also demonstrate volumetric reductions in the anterior parahippocampus and hippocampus and enlargement of the third ventricle when compared with healthy controls [22, 23].

White Matter

White matter changes are documented in schizophrenia through use of diffusion tensor imaging (DTI), a form of magnetic resonance imaging (MRI) that evaluates its structure by measuring characteristics of water diffusion in the brain. DTI studies in schizophrenia have identified numerous regions with decreased fractional anisotropy (FA), a measure that reflects axonal diameter and myelination in white matter [24]. Meta-analyses of DTI studies in schizophrenia report FA reductions of the left deep frontal lobe, right deep frontal lobe (including the right cingulum), and left deep temporal lobe [24, 25]. FA is also generally reduced in the corpus callosum, suggesting deficits in interhemispheric communication [24]. These white matter

changes were used to propose a model of schizophrenia as a syndrome of "functional disconnection," resulting from abnormalities in the connectivity between brain regions which may ultimately be relevant in understanding the cognitive deficits in this disorder.

Summary

Although structural imaging has produced a rich data set uncovering a number of consistent brain alterations, several questions merit further study. The neurodevelopmental timeline of observed alterations is not fully understood. Though there has been an increasing focus on high-risk and first-episode patients, further longitudinal data is needed to determine which brain alterations mark the conversion to psychosis. Additionally, it is unclear whether structural changes result from intrinsic disease pathology or reflect adaptations to the disease state. Continued research with healthy relatives and first-episode populations may disentangle the effects of environmental factors (e.g., antipsychotic exposure) from intrinsic disease effects on brain structure. Lastly, the current knowledge base is limited regarding the relationship between structural brain changes and long-range network connectivity. Ongoing integration of structural data with functional imaging may provide insight into these issues.

Brain Function

Functional and Molecular Imaging

Over the past several decades, functional MRI (fMRI) has been used extensively to examine brain activation in schizophrenia during specific cognitive and emotional processes (e.g., tasks of working memory, attention, and decision-making) [26]. One of the most consistent findings is diminished activation of frontal regions during cognitive tasks (i.e., "hypofrontality"), with meta-analyses of both first-episode and multiepisode patients finding reduced activation in the dorsolateral prefrontal cortex during these tasks [27–29].

Further, functional studies of social cognition and emotional processing suggest dysfunction of the amygdala and hippocampus. Individuals with schizophrenia demonstrate decreased limbic activation during tasks, but abnormally increased limbic activation when presented with fear-inducing stimuli such as pictures of angry faces [30]. Functional studies also focus on neural correlates of hallucinations and other symptoms, demonstrating that real-time auditory hallucinations correlate with increased activation of fronto-temporal regions such as Broca's area and the middle and superior temporal gyri [31]. Another area of research focuses on the

"default mode network" which is active during rest. While activity in these regions is normally suppressed during cognitive tasks, it is under-suppressed in schizophrenia, perhaps reflecting attentional and other cognitive impairments [32].

Summary

Schizophrenia is associated with abnormal limbic activation in response to emotional stimuli and frontal dysfunction during cognitive tasks. Despite its extensive research use, functional imaging is far from achieving clinical utility in diagnosis or treatment for several reasons. fMRI signals are subtle and require specialized analysis, making it difficult to detect robust signals at the level of the individual patient [33]. Lack of standardized image acquisition and analysis techniques also impedes efforts to establish the clinical utility of fMRI [33]. As with structural alterations, the pathogenesis of functional alterations is poorly understood. Two lines of research which may address these issues are multimodal studies integrating functional imaging with structural imaging, DTI, or electrophysiology and functional connectivity analyses evaluating temporal correlations in networks of brain regions [26]. These developments may ultimately yield clinically relevant biomarkers which are robust enough to identify high-risk individuals and guide treatment choices.

Neurophysiology

Neurophysiological research in schizophrenia largely focuses on event-related potentials (ERPs), which are electrophysiological responses observed in an EEG after presentation of a stimulus. ERP data is collected by applying electrodes to the scalp and recording brain electrical activity during tasks or stimuli. Electrophysiological markers are based on variations in latency, amplitude, and scalp localization of waveforms generated by various auditory stimuli. These markers differ in their anatomical substrate, their association with cognitive and clinical measures, and their putative genetic influences. Electrophysiology has several important advantages as a clinical and research tool: it is noninvasive, presents fewer technical challenges than functional and molecular imaging methods, and has a long history of use with established paradigms and analysis methods.

P300

P300 is one of the most studied electrophysiological markers. It is a large, positive voltage response evoked about 300 ms after presentation of a salient stimulus.

Its amplitude is consistently lower in individuals with schizophrenia compared with healthy controls [34]. A subset of the P300, the P3b, localizes to the inferior parietal cortex, while the other component of the P300, the P3a, localizes to frontal regions [35]. P300 abnormalities may correlate with negative symptoms, cognitive dysfunction, and gray matter reductions in temporal regions [36]. A meta-analysis of 46 studies of P300 alterations in schizophrenia found a pooled effect size of 0.85 for the amplitude of the P300 measure but did not find any impact of antipsychotic medication on this measure [37]. P300 reduction is also observable in relatives of patients with schizophrenia and in other psychiatric disorders [38].

Prepulse Inhibition of Startle

Abnormal prepulse inhibition of a startle response (PPI) is a well-established electrophysiological finding in schizophrenia. The startle response is typically measured as the amplitude of electromyographic response of the orbicularis oculi muscle when it contracts after an unexpected stimulus. PPI is the reduction in the amplitude of this wave which occurs when the startling stimulus is preceded by a weak stimulus [39]. Individuals with schizophrenia do not exhibit the same degree of reduction in the startle response as healthy controls, suggesting defects in the brain's ability to selectively filter stimuli [39]. This abnormality correlates with thought disorder and disruption in global functioning but may normalize with second-generation antipsychotic (SGA) treatment [40, 41]. Abnormal PPI is highly heritable and present in unaffected first-degree relatives [42, 43]. PPI also has a direct association with gray matter volume in the right superior parietal cortex of first-episode patients and healthy controls [44].

P50 Auditory-Evoked Potential Suppression

The P50 wave is an evoked potential which occurs 50 ms after a stimulus. When two auditory clicks are presented 500 ms apart, generating two P50 waves, the amplitude of the second wave is reduced in comparison to the first. This phenomenon is known as P50 suppression and is abnormal in patients with schizophrenia who demonstrate a smaller reduction in amplitude of the second P50 wave compared with normal controls [45, 46]. Cholinergic neurotransmitter pathways may mediate P50 suppression, as indicated by the apparent ability of nicotine to briefly normalize suppression in people with schizophrenia [47]. P50 suppression localizes to the hippocampus, is estimated to have a heritability around 68 %, and is observed in first-degree relatives of patients with schizophrenia [48–50]. Treatment with clozapine, but not other antipsychotics, may normalize P50 suppression [51].

Mismatch Negativity

Mismatch negativity (MMN) refers to an ERP component which is measured when a series of repetitive auditory stimuli are punctuated by deviant or "oddball" stimuli. MMN is believed to reflect pre-attentive sensory processing and the response of the brain to changes in consecutive stimuli [52]. It localizes to the primary and secondary auditory cortices, as well as the dorsolateral prefrontal cortices [53]. MMN abnormalities appear specific to schizophrenia among psychiatric disorders and are consistently associated with cognitive activity and the ability to function independently [54, 55].

Cortical Oscillations and Neural Synchrony

Neural oscillations allow networks of brain regions to coordinate activity essential for cognitive processes. Gamma band oscillation, which is normally in the range of 30–80 Hz, reflects coordination of neuronal activity and appears highly heritable [56, 57]. Compared to healthy controls, individuals with schizophrenia demonstrate reduced power of gamma band oscillations and are less able to modulate these oscillations in frontal regions during tests of cognitive function [58, 59]. Because cortical synchrony matures during adolescence, gamma band abnormalities may reflect a failure of normal neurodevelopment. A number of studies support a correlation between positive symptoms, particularly auditory and visual hallucinations, and changes in gamma band activity [60].

Summary

The existence of electrophysiological abnormalities in schizophrenia is well established but their clinical and research implications require further study. Electrophysiological markers hold the potential to serve as "endophenotypes" (i.e., heritable, objective disease characteristics that bridge the gap between genetic factors and clinical phenotype). These markers fulfill some of the criteria for endophenotypes as they appear to be highly heritable and present in unaffected relatives. They do not, however, consistently fulfill other endophenotype criteria such as stability over time. In addition, the clinical utility of electrophysiological markers is currently limited because of the expense of EEG equipment, the need for subjects to cooperate with complex task instructions (for some markers), and the lack of diagnostic specificity or treatment implications for most existing markers [61]. Nevertheless, electrophysiology is highly relevant for future study, as markers may correlate with elements of early information processing disrupted in schizophrenia. Electrophysiology can also be applied to the study of simple neural processes in other mammals, thus expanding the role of animal models in schizophrenia research.

Oxidative, Immunological, and Neuroendocrine Abnormalities

Schizophrenia may involve abnormalities in the oxidative stress response. Oxidative stress results from the body's inability to fully neutralize free radicals generated by normal metabolic processes [62]. This can ultimately lead to cell membrane damage and impaired neurotransmission. Markers of elevated oxidative stress are observed in schizophrenia (e.g., diminished levels of the antioxidant glutathione in the brain and cerebrospinal fluid) [63]. A meta-analysis found decreased levels of the antioxidant red blood cell superoxide dismutase across the illness course, implying it may be a trait marker for schizophrenia [62]. Other antioxidants in this study varied by patients' clinical status or stage of disease. Antioxidant deficits may affect interneuron function and cortical synchrony. For example, in a mouse model, impaired synthesis of glutathione led to reductions in high-frequency gamma oscillations [64]. Thus, oxidative stress may play a critical role in the neurodevelopmental pathways leading to schizophrenia, while enhancement of antioxidant function may represent a novel therapeutic pathway. Further investigation is needed to explore whether antioxidant abnormalities explain the link between schizophrenia and certain environmental risk factors (e.g., psychosocial stress or viral infections) [65]. Additional trials also are necessary to validate the clinical potential of antioxidant treatment, although studies using N-acetyl cysteine show promise in the treatment of negative symptoms (see Chap. 7) [66].

Immunological theories propose that autoimmune dysfunction or infections may contribute to the etiology of schizophrenia. Epidemiological studies observe a correlation between prenatal exposure to infections and later development of schizophrenia [67]. Studies report a number of immunological findings, including changes in the levels of cytokines (signaling molecules that coordinate the inflammatory response), elevated levels of autoantibodies, and associations between autoimmune disorders and risk for schizophrenia [68-70]. A meta-analysis found that some cytokines (such as IL-6 and TGF- β) were associated with psychotic exacerbations, while others (IL-12, TNF- α) may be trait markers for schizophrenia [68]. In addition, anti-inflammatory medications may augment antipsychotic response and improve psychotic symptoms in randomized controlled trials (see Chap. 7) [71]. These findings lend support for immunological theories of schizophrenia and fit well with known prenatal risk factors. The current literature on immunological markers is often limited by inadequate control for confounding factors such as clinical status, body mass index, and smoking. Thus, many studies were unable to determine if abnormalities represented state or trait markers [68]. Additional research is necessary to clarify the role of cytokines in the pathogenesis of schizophrenia and whether they hold potential as therapeutic targets in addition to biomarkers.

Dysfunction in the hypothalamus–pituitary–adrenal (HPA) axis may mediate interactions between stress and psychosis. The stress response appears blunted in individuals with schizophrenia, as reflected in studies showing decreased cortisol response to psychological and physical stress [72, 73]. Multiple studies also demonstrate a relatively high rate of dexamethasone non-suppression (i.e., the absence of

cortisol suppression after a dexamethasone suppression test) in chronic schizophrenia, indicating failure of the HPA negative feedback mechanism [74]. Subgroups of patients (e.g., those with primary polydipsia) have higher rates of dexamethasone resistance and HPA overactivity which may result from abnormal hippocampal regulation of the stress response (see Chap. 11) [75]. Function of the HPA system is likely preserved overall, despite some abnormal input from the limbic system. One major challenge for neuroendocrine research in schizophrenia is distinguishing adaptive responses from intrinsic disease pathophysiology. To place neuroendocrine findings in an appropriate context, future studies will need to identify where abnormalities occur in the regulatory pathway and connect observed abnormalities with clinical symptoms.

Neurotransmitter Systems

Neurochemical Imaging

Proton magnetic resonance spectroscopy (¹H MRS) is a noninvasive imaging method which assesses the chemical composition of brain tissue in vivo by measuring magnetic resonance signals produced by atomic nuclei within molecules [76]. This form of imaging can estimate concentrations of several biologically relevant compounds including N-acetyl aspartate (NAA) which is a marker of neuronal integrity; glutamate and glutamine (Glu+Gln), which correlate with glutamatergic neurotransmission; and choline metabolites (Cho) which are an indicator of cellular turnover [76]. One of the most consistent findings in this area is a reduction in NAA in frontal, temporal, and thalamic regions. This is seen in both first-episode and multiplisode patients, indicating neuronal abnormalities in these regions [77]. Another consistent finding, which was reported in a meta-analysis of studies on Glu+Gln, is a reduction in frontal glutamate and increase in glutamine [78]. Possible reasons for reduced glutamatergic function include hypo-activation of N-methyl-D-aspartate (NMDA) receptors and abnormal expression of glutamate transporters [78]. These hypotheses require clarification, however, perhaps through animal models.

The current MRS literature is also limited by the paucity of longitudinal studies, making it difficult to assess the relationship between neurochemical alterations and disease course [78]. Because of long scanning times, studies typically limit themselves to small samples of cooperative subjects [79]. Scanner resolution is constrained by the size and strength of spectroscopy magnets [79]. Studies also differ in their choice of MRS techniques, and many did not control for variables such as antipsychotic medication status or duration of illness [76, 78]. Future studies will need to develop and use more standardized methods. Newer forms of MRS (e.g., proton echo-planar spectroscopic imaging (PEPSI)) are able to scan multiple brain regions in a short amount of time. These advances allow for the imaging of

agitated or uncooperative subjects, enhancing the potential clinical utility of MRS. Recent studies are also using more powerful magnets which permit greater spatial and temporal resolution [79].

MRS with ³¹P (phosphorus-31) is also used to examine cell membrane phospholipids (i.e., PME (phosphomonoester), a phospholipid precursor; and PDE (phosphodiester), a metabolite of phospholipid breakdown). PME and PDE can provide information about membrane phospholipid synthesis and turnover which in turn may reflect the condition of neuronal membranes [80]. Studies utilizing ³¹P have reported reductions in PME in prefrontal and medial temporal regions in schizophrenia, implying reduced production of membrane phospholipids [80]. As with ¹H MRS, studies with ³¹P are limited by small sample sizes, inconsistencies in imaging and analysis techniques, and differences in subject populations [79]. ³¹P MRS, however, continues to be relevant given its unique ability to examine processes of lipid metabolism and cell membrane turnover.

Two forms of molecular imaging are used to explore neurotransmitter systems in schizophrenia: positron-emission tomography (PET) and single-photon emissioncomputed tomography (SPECT). In both methods, radioactive-labeled tracers are injected into the bloodstream and emitted rays are measured. PET demonstrates better spatial resolution and more sensitivity to subtle brain changes than SPECT, and presently appears to be a more powerful brain imaging technique [81]. Two meta-analyses of SPECT and PET studies found elevations in striatal dopamine synthesis capacity but found no significant difference in dopamine transporter availability [82, 83]. Additionally, prefrontal hypo-activation is correlated with increased striatal dopamine function [84]. These findings support a modified version of the "dopamine hypothesis" (to be discussed below) in schizophrenia. Altered striatal dopamine synthesis may hold potential as a biomarker for risk of psychosis, but additional research is needed to determine how it changes over the illness course and whether it is specific to schizophrenia rather than affective psychosis [82]. In addition to clarifying these issues, future studies could integrate PET and SPECT methods with electrophysiology and structural imaging to identify dysfunctional networks [85]. As with MRS, PET and SPECT research is limited by small sample sizes, differences in technical methods, and variations in subject characteristics.

Dopamine

The dopamine hypothesis has persisted as a pathophysiological theory of schizophrenia for several decades. Various lines of evidence support this theory. These include the efficacy of D2 receptor antagonists in treating the acute symptoms of psychosis, as well as the ability of amphetamines to stimulate psychosis by increasing extracellular dopamine. Initial models theorized simply that increased dopamine transmission led to psychosis, and later, that defects in mesocortical dopamine activity resulted in overactivity of the mesolimbic system [86]. Further, depletion in prefrontal dopamine was believed to disinhibit subcortical regions, resulting in dopaminergic excess in striatal regions [87]. More recent models proposed that presynaptic dopamine dysfunction is the primary path to psychosis, in accordance with recent PET studies [82, 83]. These more recent findings point to presynaptic dopamine synthesis as a potential therapeutic focus (see Chap. 7) [83].

The ultimate place of dopaminergic dysfunction in the pathogenesis of schizophrenia is unclear with the literature increasingly implying that dopamine dysfunction is a downstream consequence of other deficits. One weakness of initial dopamine models was their relative difficulty in accounting for cognitive deficits and negative symptoms in schizophrenia. Altered D1-mediated transmission in the prefrontal areas or changes in glutamatergic neurotransmission may influence striatal dopamine systems and could underlie cognitive and negative symptoms [83, 84]. However, these theories await further confirmation.

Gamma Aminobutyric Acid

Gamma aminobutyric acid (GABA) is the primary inhibitory neurotransmitter in the brain and an increasing focus of attention in the pathophysiology of schizophrenia. Inhibitory control by GABA interneurons is believed to coordinate subsets of pyramidal neurons which comprise the majority of cortical neurons [56]. Thus, GABA interneurons are crucial for synchronized neural activity [56]. While the overall number of GABA interneurons is not reduced in schizophrenia, those which contain parvalbumin demonstrate reduced production and uptake of GABA [88]. In particular, levels of glutamic acid decarboxylase 67 (GAD-67), an essential enzyme for GABA synthesis, are diminished in several cortical regions in parvalbumin-containing GABA interneurons [88]. Deficiencies in GAD-67 and decreased expression of parvalbumin appear to correlate with alterations in gamma band oscillations, thus linking electrophysiological phenomena and neurotransmitter systems. Synaptic alterations in the dorsolateral prefrontal cortex may disrupt the balance of inhibitory control by GABA interneurons, leading to deficits in working memory [89]. It is not known whether the observed abnormalities in the GABA-ergic system represent primary deficits in schizophrenia or compensatory mechanisms for other deficits [56]. For example, increases in postsynaptic GABA-A receptors and reductions in presynaptic parvalbumin may compensate for reduced GABA release from chandelier neurons which serve to inhibit pyramidal neurons [90]. This viewpoint is supported by evidence from animal models showing that such alterations tend to improve GABA transmission [91].

Glutamate

Unlike dopamine, glutamate is present throughout the entire nervous system and is the main excitatory neurotransmitter in mammals [92]. The potential relevance of glutamate to the pathophysiology of schizophrenia was discovered through research on NMDA receptor antagonists such as phencyclidine (PCP) and ketamine. NMDA receptors are normally activated through the binding of glutamate or aspartate, but synthetic NMDA antagonists bind specifically to the NMDA receptor. When administered to healthy subjects, NMDA antagonists can induce psychotic symptoms, cognitive deficits, and defects in MMN which mimic those observed in schizophrenia [93, 94]. From this and other data, it was initially proposed that schizophrenia involved diminished function or density of NMDA receptors caused by abnormalities in glutamate neurotransmission [92]. Studies on postmortem brains report reductions in the NMDA receptor density in the prefrontal cortex and hippocampus [95]. Reduction in the density of dendritic spines, which receive excitatory glutamatergic synapses, may also reflect inadequate glutamate neurotransmission and resulting defects in connectivity [96]. Subsequent research, however, suggests that glutamatergic excess may actually be a driving force in the disease process. One promising pharmacotherapeutic application is the use of metabotropic glutamate 2/3 receptor agonists, which could balance excitatory and inhibitory systems by normalizing presynaptic glutamate levels (see Chap. 7) [92].

Summary

Dopaminergic dysfunction remains a consistent finding in schizophrenia and may be due primarily to alterations in presynaptic synthesis and transmission. While the relative place of dopaminergic dysfunction in the pathogenesis of schizophrenia remains unclear, it may play an early role in neurodevelopmental processes or it may reflect downstream consequences of other abnormalities. Recent investigations focusing on glutamatergic and GABA-ergic neurotransmitter systems may generate novel therapeutic targets.

Neuropathology

Schizophrenia is historically a challenging area in neuropathology, initially yielding few definite findings. As stated earlier, it is associated with subtle reductions in total brain volume and total gray matter volume [6]. Notably, these reductions are not attributable to loss of neurons, but to reductions in the size of neuronal cell bodies and in cortical neuropil [97]. Specifically, pyramidal cell bodies are about 10 % smaller in layer three of the dorsolateral prefrontal cortex and dendritic shafts and spines are reduced, resulting in diminished dendritic arborization [98]. Dendritic spines receive excitatory synapses throughout the central nervous system and reductions in their density may underlie connectivity defects between regions [96]. In addition, reductions and altered gene expression in the microglia, support cells of the nervous system which include astrocytes and oligodendrocytes, may contribute to white matter abnormalities [99].

Another consistent finding is the lack of gliosis. This is the proliferation of glial cells in response to nervous system damage which is observed in many neurodegenerative disorders [89]. Its absence in schizophrenia suggests a neurodevelopmental rather than a neurodegenerative process at work [97]. Postmortem studies observe higher densities of cortical neurons in deep rather than superficial layers of the limbic and prefrontal regions. This indicates that there may be early failures in the migration of neuronal precursor cells from subcortical to cortical regions during gestation [100]. Neuropathological studies have a number of limitations including the effects of variable postmortem intervals, heterogeneity within the schizophrenia diagnostic spectrum, and the confounding effects of antipsychotic medication [97].

Neuroplasticity

The presence of widespread gray matter deficits and synaptic alterations without apparent neuronal loss raises the question of whether these abnormalities are remediable. Neuroplasticity refers to the ability of the brain to adapt to the environment and reorganize aspects of neuronal circuits, such as synaptic density [101]. Neuroplasticity may be abnormally reduced in schizophrenia, as suggested by studies showing diminished long-term potentiation in response to transcranial magnetic stimulation and lowered stimulus-specific plasticity in response to electrophysio-logical tetanic stimulation [101, 102]. Recent findings indicate that levels of brain-derived neurotrophic factor (BDNF), a protein responsible for neuronal development and synaptic plasticity, may be altered in schizophrenia [103]. BDNF may closely interact with glutamatergic, dopaminergic, and GABA-ergic neurotransmitter systems and may mediate the gray matter improvements observed following cognitive remediation [104, 105].

Models of Pathogenesis

Although various neurobiological abnormalities are now well established in the schizophrenia literature, it is important to synthesize these findings to develop models of pathogenesis. The majority of evidence supports the conceptualization of schizophrenia as a neurodevelopmental disorder. This evidence includes the onset of schizophrenia in adolescence and the presence of premorbid cognitive deficits, minor physical anomalies, and neuromotor abnormalities during childhood [106–108]. The timeline of neurodevelopmental alterations, however, remains unclear.

One model posits that the pathogenesis of schizophrenia begins pre- or perinatally, perhaps through early environmental insults or genetically mediated defects in neuronal migration [86]. As mentioned earlier, postmortem studies imply alterations in neuronal migration from subcortical to cortical regions, a process which occurs during the second trimester [100]. Studies of high-risk individuals find a high prevalence of motor coordination issues in childhood, further supporting an early timeline of pathogenesis [108]. An exclusive focus on perinatal or prenatal abnormalities, however, does not explain why the characteristic symptoms of schizophrenia emerge in late adolescence.

Another model addresses this issue by focusing on neurodevelopmental abnormalities in late adolescence. Excessive synaptic or axonal pruning may underlie longitudinal changes reported in young high-risk and first-episode patients. These include gray matter reductions and reductions in NAA observed in MRI studies [10, 77]. Findings from neuropathology, such as reductions in dendritic arborization and synaptic density, are also consistent with this model [98].

Schizophrenia does not appear to be a neurodegenerative disorder, as demonstrated by lack of gliosis or other neurodegenerative signs in postmortem studies [97]. There still remains a question of whether a subgroup of individuals experience progressive deterioration. The first few years of illness in particular are marked by a decline in function and some individuals recover incompletely or take longer to recover after each successive psychotic episode [109].

Diminished neuroplasticity may fit well with existing neurodevelopmental models of pathogenesis. Dysfunction in GABA-ergic systems could reduce plasticity in cortical areas, while changes in NMDA receptor-mediated neurotransmission may disrupt long-term potentiation, a crucial process for learning and memory [101, 110]. Altered neuroplasticity may serve as a biological mechanism for environmentally mediated longitudinal changes in schizophrenia.

Summary

Although divergent in their timelines, these models could be integrated with the current understanding of neurotransmitter systems to create a "three-hit" model that longitudinally describes the pathogenesis of schizophrenia [111]. Early genetic or environmental "hits" may disturb glutamate-mediated processes of neuronal migration and survival during gestation. This could lead to cell death and loss of glutamatergic neurons, manifesting as premorbid cognitive and neuromotor deficits. In adolescence, hypofunction of NMDA receptors, whose sustained activity is necessary for synaptic survival, could result in excessive synaptic pruning and reduced neuroplasticity. Diminished neuronal connectivity could worsen cognitive and social function during this time period. Additionally, glutamatergic dysfunction could disrupt phasic and tonic dopamine release and upregulate subcortical dopaminergic neurons, ultimately precipitating the onset of psychotic symptoms [111, 112]. During the years following the first episode of psychosis, glutamatergic dysfunction could lead to increased phasic dopamine release during psychotic episodes, which could then increase glutamate release, resulting in oxidative stress and neuronal damage. Neurotoxicity caused by glutamatergic excess could then account for disease progression during the first years of illness. In addition, deficits in neuroplasticity may help explain chronic cognitive impairments. This model draws support from genetic

studies showing association of schizophrenia with glutamate- and neuroplasticityrelated genes (see Chap. 5). Further research using animal models and longitudinal brain imaging is likely to refine neurodevelopmental theories of pathogenesis.

Conclusion

Knowledge Gaps and Future Steps

As this review indicates, many facts about the pathophysiology of schizophrenia are being discovered at a rapid pace. This progress, however, is not matched by their translation to advances in treatment of schizophrenia. In general, treatments for schizophrenia were largely discovered by astute, yet serendipitous, observations. Thus far, few treatments are based on knowledge of pathophysiology (Fig. 4.1). Yet, some trends may be delineated.

Toward neuroscience-based classification of psychoses. While successive revisions of the Diagnostic and Statistical Manual of Mental Illness (DSM) serve to improve the reliability and clinical utility of a symptom-based classification of psychiatric disorders, there remains a critical lack of *validity* in the current categorization of psychoses. The century-old distinction between schizophrenia and affective psychoses remains in the DSM-5 despite the considerable overlap in symptoms, cognition, neurobiology, genetics, treatment response, and outcome characteristics across these disorders [113–117]. Emerging new data on pathophysiology of the psychotic disorders spectrum can eventually help move the field



Fig. 4.1 Pathways to treatment discovery. Treatments for schizophrenia have traditionally been developed by clinical observation and serendipity (*dashed lines*) but need to be increasingly informed by etiopathological observations and hypothesis testing (*solid lines*) (Adapted from Tandon et al. [132])

toward a neuroscience-informed nosology. Such progress is only possible if disease dimensions of what we now call schizophrenia, spanning molecular to behavioral domains across the psychosis spectrum, are deconstructed [118, 119]. The cross-cutting pathophysiological dimensions can then be used to identify new, perhaps more valid categories of the psychosis spectrum. Relevant to the pursuit of this goal are the NIMH Research Domain Criteria (R-DoC) which seek to map translationally relevant behavioral phenotypes to biomarkers in physiological and molecular domains [120]. The introduction of dimensional measures which cut across diagnoses in DSM-5 is a good step in this direction. It is critical for the field, however, to utilize such measures in large populations, agnostic to DSM diagnoses, and examine their relationships to biology [121].

Toward diagnostic and predictive biomarkers. Despite accumulating data on several altered biological processes, few diagnostic or predictive biomarkers exist for psychiatric disorders, other than those to rule out "other" medical disorders (e.g., testing to rule out hypothyroidism). Even "proxy" biomarkers such as cognitive deficits, which are widely prevalent, pervasive across multiple domains, persistent, map onto biology, and may predict outcome in schizophrenia, are not yet incorporated into standard psychiatric assessments [122]. At least in part, this impasse stems from a continued reliance on symptom-based categories as the gold standard for developing diagnostic tests [121, 123]. In striking contrast, there is unprecedented progress in imaging, genomics, and computational abilities which could deliver clinically useful tests in the near future. Given the lack of validity of symptom-based classifications in psychiatric disorders as discussed earlier, simple comparisons of imaging or other biomarker data between diseases and between disease and healthy subjects are unlikely to yield much. In contrast, examining etiological differences across biologically defined subgroups of disease may be much more valuable. For example, using data-driven approaches on phenotypically diverse subjects, it may be possible to derive subgroups characterized by distinctive biological features, quantifiable through neuroimaging and electrophysiology. These subgroups could then be compared in etiology and pathophysiology, ultimately generating targeted therapeutic approaches.

Multimodal approaches and automated machine learning algorithms (e.g., support vector machines (SVMs) which utilize multivariate pattern recognition methods) can robustly distinguish early course of schizophrenia and its progression [124]. Such classification approaches in the future may also incorporate information from other biomarker domains, including electrophysiology, metabolomic, proteomic, genomic, and gene expression profiles. Cellular markers derived from induced pluripotent stem cells are another exciting direction in the not-too-distant future [125].

Toward theory-driven therapeutic interventions. Stratifying psychosis spectrum disorders into neurobiologically separable entities will help develop more targeted interventions [119]. Molecular stratification of disease is already standard practice in the rest of medicine. Thus, the presence of "actionable" mutations such as BRCA1 gene in breast cancer can lead to prevention efforts [126]. In another example,

specific treatments can lead to substantial clinical benefit for cystic fibrosis for individuals with CFTR mutations [127]. Several examples of such theory-driven interventions in schizophrenia may now be mentioned. It is important to look beyond simply modulating the dopamine receptor. Recent thinking on the role of glutamatergic and GABA pathways discussed earlier suggests trials of several novel pharmacological agents which may impact these systems (see Chap. 7) [128]. One needs to look beyond neurotransmitter systems as well to treatments which address increased oxidative stress, such as *N*-acetyl cysteine [66]. Anti-inflammatory agents such as aspirin may be of value in treating psychosis [129]. Improvement in cognitive deficits in schizophrenia patients positive for herpes simplex antibody titers is reported with the antiviral agent valacyclovir [130]. Treatments that directly upregulate BDNF or improve neural plasticity, such as computer-based cognitive enhancement therapies, also hold promise as novel therapeutic approaches [105].

In conclusion, our understanding of the pathophysiology of schizophrenia has made remarkable progress, but much work remains to translate these observations into real differences for managing and potentially preventing this devastating illness. Future researchers will benefit from both a creative application of cutting-edge neuroscience knowledge and having an open mind to look beyond current conceptual models of this disease.

References

- 1. Kraepelin E. Dementia Praecox and Paraphrenia, 1919. Robertson G, editor. New York: RE Krieger; 1971.
- 2. Plum F. Prospects for research on schizophrenia. 3. Neurophysiology. Neuropathological findings. Neurosci Res Program Bull. 1972;10(4):384–8.
- Johnstone EC, Crow TJ, Frith CD, Husband J, Kreel L. Cerebral ventricular size and cognitive impairment in chronic schizophrenia. Lancet. 1976;2(7992):924–6.
- 4. Weinberger DR. From neuropathology to neurodevelopment. Lancet. 1995;346(8974): 552–7.
- Lawrie SM, Abukmeil SS. Brain abnormality in schizophrenia. A systematic and quantitative review of volumetric magnetic resonance imaging studies. Br J Psychiatry. 1998;172: 110–20.
- Wright IC, Rabe-Hesketh S, Woodruff PW, David AS, Murray RM, Bullmore ET. Metaanalysis of regional brain volumes in schizophrenia. Am J Psychiatry. 2000;157(1):16–25.
- Shepherd AM, Laurens KR, Matheson SL, Carr VJ, Green MJ. Systematic meta-review and quality assessment of the structural brain alterations in schizophrenia. Neurosci Biobehav Rev. 2012;36(4):1342–56.
- Honea R, Crow TJ, Passingham D, Mackay CE. Regional deficits in brain volume in schizophrenia: a meta-analysis of voxel-based morphometry studies. Am J Psychiatry. 2005; 162(12):2233–45.
- Steen RG, Mull C, McClure R, Hamer RM, Lieberman JA. Brain volume in first-episode schizophrenia: systematic review and meta-analysis of magnetic resonance imaging studies. Br J Psychiatry. 2006;188:510–8.
- Vita A, De Peri L, Deste G, Sacchetti E. Progressive loss of cortical gray matter in schizophrenia: a meta-analysis and meta-regression of longitudinal MRI studies. Trans Psychiatry. 2012;2:e190.

- 11. Keshavan MS, Bagwell WW, Haas GL, Sweeney JA, Schooler NR, Pettegrew JW. Changes in caudate volume with neuroleptic treatment. Lancet. 1994;344(8934):1434.
- Fusar-Poli P, Smieskova R, Kempton MJ, Ho BC, Andreasen NC, Borgwardt S. Progressive brain changes in schizophrenia related to antipsychotic treatment? A meta-analysis of longitudinal MRI studies. Neurosci Biobehav Rev. 2013;37(8):1680–91.
- Rimol LM, Hartberg CB, Nesvåg R, Fennema-Notestine C, Hagler DJ, Pung CJ, et al. Cortical thickness and subcortical volumes in schizophrenia and bipolar disorder. Biol Psychiatry. 2010;68(1):41–50.
- Rimol LM, Nesvåg R, Hagler DJ, Bergmann O, Fennema-Notestine C, Hartberg CB, et al. Cortical volume, surface area, and thickness in schizophrenia and bipolar disorder. Biol Psychiatry. 2012;71(6):552–60.
- Shapleske J, Rossell SL, Woodruff PW, David AS. The planum temporale: a systematic, quantitative review of its structural, functional and clinical significance. Brain Res Brain Res Rev. 1999;29(1):26–49.
- Sun J, Maller JJ, Guo L, Fitzgerald PB. Superior temporal gyrus volume change in schizophrenia: a review on region of interest volumetric studies. Brain Res Rev. 2009;61(1):14–32. Research Support, Non-U.S. Gov't Review.
- Brambilla P, Perlini C, Rajagopalan P, Saharan P, Rambaldelli G, Bellani M, et al. Schizophrenia severity, social functioning and hippocampal neuroanatomy: three-dimensional mapping study. Br J Psychiatry. 2013;202(1):50–5.
- Antonova E, Sharma T, Morris R, Kumari V. The relationship between brain structure and neurocognition in schizophrenia: a selective review. Schizophr Res. 2004;70(2–3):117–45.
- Palaniyappan L, Balain V, Liddle PF. The neuroanatomy of psychotic diathesis: a metaanalytic review. J Psychiatr Res. 2012;46(10):1249–56.
- Borgwardt SJ, Riecher-Rössler A, Dazzan P, Chitnis X, Aston J, Drewe M, et al. Regional gray matter volume abnormalities in the at risk mental state. Biol Psychiatry. 2007;61(10):1148–56.
- Mechelli A, Riecher-Rössler A, Meisenzahl EM, Tognin S, Wood SJ, Borgwardt SJ, et al. Neuroanatomical abnormalities that predate the onset of psychosis: a multicenter study. Arch Gen Psychiatry. 2011;68(5):489–95.
- 22. Seidman LJ, Pantelis C, Keshavan MS, Faraone SV, Goldstein JM, Horton NJ, et al. A review and new report of medial temporal lobe dysfunction as a vulnerability indicator for schizophrenia: a magnetic resonance imaging morphometric family study of the parahippocampal gyrus. Schizophr Bull. 2003;29(4):803–30.
- Boos HB, Aleman A, Cahn W, Hulshoff Pol H, Kahn RS. Brain volumes in relatives of patients with schizophrenia: a meta-analysis. Arch Gen Psychiatry. 2007;64(3):297–304.
- Yao L, Lui S, Liao Y, Du MY, Hu N, Thomas JA, et al. White matter deficits in first episode schizophrenia: an activation likelihood estimation meta-analysis. Prog Neuropsychopharmacol Biol Psychiatry. 2013;45C:100–6.
- Ellison-Wright I, Bullmore E. Meta-analysis of diffusion tensor imaging studies in schizophrenia. Schizophr Res. 2009;108(1–3):3–10.
- 26. Gur RE, Gur RC. Functional magnetic resonance imaging in schizophrenia. Dialogues Clin Neurosci. 2010;12(3):333–43.
- Berman KF, Meyer-Lindenberg A. Functional brain imaging studies in schizophrenia. In: Charney D, Nestler E, editors. Neurobiology of mental illness. 2nd ed. Oxford, MA: Oxford University Press; 2004.
- Fusar-Poli P, Perez J, Broome M, Borgwardt S, Placentino A, Caverzasi E, et al. Neurofunctional correlates of vulnerability to psychosis: a systematic review and metaanalysis. Neurosci Biobehav Rev. 2007;31(4):465–84.
- 29. Minzenberg MJ, Laird AR, Thelen S, Carter CS, Glahn DC. Meta-analysis of 41 functional neuroimaging studies of executive function in schizophrenia. Arch Gen Psychiatry. 2009; 66(8):811–22.
- Gur RE, Loughead J, Kohler CG, Elliott MA, Lesko K, Ruparel K, et al. Limbic activation associated with misidentification of fearful faces and flat affect in schizophrenia. Arch Gen Psychiatry. 2007;64(12):1356–66.

- 4 Pathophysiology of Schizophrenia
 - Jardri R, Pouchet A, Pins D, Thomas P. Cortical activations during auditory verbal hallucinations in schizophrenia: a coordinate-based meta-analysis. Am J Psychiatry. 2011;168(1): 73–81.
 - Whitfield-Gabrieli S, Ford JM. Default mode network activity and connectivity in psychopathology. Annu Rev Clin Psychol. 2012;8:49–76.
 - Jezzard P, Buxton RB. The clinical potential of functional magnetic resonance imaging. J Magn Reson Imaging. 2006;23(6):787–93.
 - Roth WT, Cannon EH. Some features of the auditory evoked response in schizophrenics. Arch Gen Psychiatry. 1972;27(4):466–71.
 - 35. Linden DE. The p300: where in the brain is it produced and what does it tell us? Neuroscientist. 2005;11(6):563–76.
 - 36. Ford JM. Schizophrenia: the broken P300 and beyond. Psychophysiology. 1999;36(6):667-82.
 - 37. Bramon E, Rabe-Hesketh S, Sham P, Murray RM, Frangou S. Meta-analysis of the P300 and P50 waveforms in schizophrenia. Schizophr Res. 2004;70(2–3):315–29.
 - Winterer G, Egan MF, Raedler T, Sanchez C, Jones DW, Coppola R, et al. P300 and genetic risk for schizophrenia. Arch Gen Psychiatry. 2003;60(11):1158–67.
 - Braff DL, Geyer MA, Swerdlow NR. Human studies of prepulse inhibition of startle: normal subjects, patient groups, and pharmacological studies. Psychopharmacology (Berl). 2001;156(2–3):234–58.
 - Perry W, Geyer MA, Braff DL. Sensorimotor gating and thought disturbance measured in close temporal proximity in schizophrenic patients. Arch Gen Psychiatry. 1999;56(3):277–81.
 - 41. Weike AI, Bauer U, Hamm AO. Effective neuroleptic medication removes prepulse inhibition deficits in schizophrenia patients. Biol Psychiatry. 2000;47(1):61–70.
 - 42. Hasenkamp W, Epstein MP, Green A, Wilcox L, Boshoven W, Lewison B, et al. Heritability of acoustic startle magnitude, prepulse inhibition, and startle latency in schizophrenia and control families. Psychiatry Res. 2010;178(2):236–43.
 - 43. Cadenhead KS, Swerdlow NR, Shafer KM, Diaz M, Braff DL. Modulation of the startle response and startle laterality in relatives of schizophrenic patients and in subjects with schizotypal personality disorder: evidence of inhibitory deficits. Am J Psychiatry. 2000;157(10):1660–8.
 - 44. Hammer TB, Oranje B, Skimminge A, Aggernæs B, Ebdrup BH, Glenthøj B, et al. Structural brain correlates of sensorimotor gating in antipsychotic-naive men with first-episode schizophrenia. J Psychiatry Neurosci. 2013;38(1):34–42.
 - 45. Griffith J, Hoffer LD, Adler LE, Zerbe GO, Freedman R. Effects of sound intensity on a midlatency evoked response to repeated auditory stimuli in schizophrenic and normal subjects. Psychophysiology. 1995;32(5):460–6.
 - 46. Adler LE, Pachtman E, Franks RD, Pecevich M, Waldo MC, Freedman R. Neurophysiological evidence for a defect in neuronal mechanisms involved in sensory gating in schizophrenia. Biol Psychiatry. 1982;17(6):639–54.
 - Adler LE, Hoffer LD, Wiser A, Freedman R. Normalization of auditory physiology by cigarette smoking in schizophrenic patients. Am J Psychiatry. 1993;150(12):1856–61.
 - Waldo MC, Cawthra E, Adler LE, Dubester S, Staunton M, Nagamoto H, et al. Auditory sensory gating, hippocampal volume, and catecholamine metabolism in schizophrenics and their siblings. Schizophr Res. 1994;12(2):93–106.
 - Hall MH, Schulze K, Rijsdijk F, Picchioni M, Ettinger U, Bramon E, et al. Heritability and reliability of P300, P50 and duration mismatch negativity. Behav Genet. 2006;36(6):845–57.
 - 50. Clementz BA, Geyer MA, Braff DL. Poor P50 suppression among schizophrenia patients and their first-degree biological relatives. Am J Psychiatry. 1998;155(12):1691–4.
 - Becker J, Gomes I, Ghisolfi ES, Schuch A, Ramos FL, Ehlers JA, et al. Clozapine, but not typical antipsychotics, correct P50 suppression deficit in patients with schizophrenia. Clin Neurophysiol. 2004;115(2):396–401.
 - Thaker GK. Neurophysiological endophenotypes across bipolar and schizophrenia psychosis. Schizophr Bull. 2008;34(4):760–73.

- Alho K, Woods DL, Algazi A, Knight RT, Näätänen R. Lesions of frontal cortex diminish the auditory mismatch negativity. Electroencephalogr Clin Neurophysiol. 1994;91(5):353–62.
- Umbricht D, Koller R, Schmid L, Skrabo A, Grübel C, Huber T, et al. How specific are deficits in mismatch negativity generation to schizophrenia? Biol Psychiatry. 2003;53(12): 1120–31.
- Light GA, Braff DL. Mismatch negativity deficits are associated with poor functioning in schizophrenia patients. Arch Gen Psychiatry. 2005;62(2):127–36.
- Uhlhaas PJ, Singer W. Abnormal neural oscillations and synchrony in schizophrenia. Nat Rev Neurosci. 2010;11(2):100–13.
- Linkenkaer-Hansen K, Smit DJ, Barkil A, van Beijsterveldt TE, Brussaard AB, Boomsma DI, et al. Genetic contributions to long-range temporal correlations in ongoing oscillations. J Neurosci. 2007;27(50):13882–9.
- Kwon JS, O'Donnell BF, Wallenstein GV, Greene RW, Hirayasu Y, Nestor PG, et al. Gamma frequency-range abnormalities to auditory stimulation in schizophrenia. Arch Gen Psychiatry. 1999;56(11):1001–5.
- 59. Cho RY, Konecky RO, Carter CS. Impairments in frontal cortical gamma synchrony and cognitive control in schizophrenia. Proc Natl Acad Sci U S A. 2006;103(52):19878–83.
- Spencer KM, Niznikiewicz MA, Nestor PG, Shenton ME, McCarley RW. Left auditory cortex gamma synchronization and auditory hallucination symptoms in schizophrenia. BMC Neurosci. 2009;10:85.
- Turetsky BI, Calkins ME, Light GA, Olincy A, Radant AD, Swerdlow NR. Neurophysiological endophenotypes of schizophrenia: the viability of selected candidate measures. Schizophr Bull. 2007;33(1):69–94.
- 62. Flatow J, Buckley P, Miller BJ. Meta-analysis of oxidative stress in schizophrenia. Biol Psychiatry. 2013;74(6):400–9.
- 63. Do KQ, Trabesinger AH, Kirsten-Krüger M, Lauer CJ, Dydak U, Hell D, et al. Schizophrenia: glutathione deficit in cerebrospinal fluid and prefrontal cortex in vivo. Eur J Neurosci. 2000;12(10):3721–8.
- 64. Steullet P, Cabungcal JH, Kulak A, Kraftsik R, Chen Y, Dalton TP, et al. Redox dysregulation affects the ventral but not dorsal hippocampus: impairment of parvalbumin neurons, gamma oscillations, and related behaviors. J Neurosci. 2010;30(7):2547–58.
- O'Donnell P. Cortical interneurons, immune factors and oxidative stress as early targets for schizophrenia. Eur J Neurosci. 2012;35(12):1866–70.
- Berk M, Malhi GS, Gray LJ, Dean OM. The promise of *N*-acetyl cysteine in neuropsychiatry. Trends Pharmacol Sci. 2013;34(3):167–77.
- 67. Mednick SA, Machon RA, Huttunen MO, Bonett D. Adult schizophrenia following prenatal exposure to an influenza epidemic. Arch Gen Psychiatry. 1988;45(2):189–92.
- Miller BJ, Buckley P, Seabolt W, Mellor A, Kirkpatrick B. Meta-analysis of cytokine alterations in schizophrenia: clinical status and antipsychotic effects. Biol Psychiatry. 2011;70(7): 663–71.
- Tanaka S, Matsunaga H, Kimura M, Tatsumi K, Hidaka Y, Takano T, et al. Autoantibodies against four kinds of neurotransmitter receptors in psychiatric disorders. J Neuroimmunol. 2003;141(1–2):155–64.
- Eaton WW, Pedersen MG, Nielsen PR, Mortensen PB. Autoimmune diseases, bipolar disorder, and non-affective psychosis. Bipolar Disord. 2010;12(6):638–46.
- Laan W, Grobbee DE, Selten JP, Heijnen CJ, Kahn RS, Burger H. Adjuvant aspirin therapy reduces symptoms of schizophrenia spectrum disorders: results from a randomized, doubleblind, placebo-controlled trial. J Clin Psychiatry. 2010;71(5):520–7.
- Albus M, Ackenheil M, Engel RR, Müller F. Situational reactivity of autonomic functions in schizophrenic patients. Psychiatry Res. 1982;6(3):361–70.
- Brenner K, Liu A, Laplante DP, Lupien S, Pruessner JC, Ciampi A, et al. Cortisol response to a psychosocial stressor in schizophrenia: blunted, delayed, or normal? Psychoneuroendocrinology. 2009;34(6):859–68.

- 74. Yeragani VK. The incidence of abnormal dexamethasone suppression in schizophrenia: a review and a meta-analytic comparison with the incidence in normal controls. Can J Psychiatry. 1990;35(2):128–32.
- Goldman MB, Blake L, Marks RC, Hedeker D, Luchins DJ. Association of nonsuppression of cortisol on the DST with primary polydipsia in chronic schizophrenia. Am J Psychiatry. 1993;150(4):653–5.
- Kraguljac NV, Reid M, White D, Jones R, den Hollander J, Lowman D, et al. Neurometabolites in schizophrenia and bipolar disorder—a systematic review and meta-analysis. Psychiatry Res. 2012;203(2–3):111–25.
- Brugger S, Davis JM, Leucht S, Stone JM. Proton magnetic resonance spectroscopy and illness stage in schizophrenia—a systematic review and meta-analysis. Biol Psychiatry. 2011;69(5):495–503.
- Marsman A, van den Heuvel MP, Klomp DW, Kahn RS, Luijten PR, Hulshoff Pol HE. Glutamate in schizophrenia: a focused review and meta-analysis of ¹H-MRS studies. Schizophr Bull. 2013;39(1):120–9.
- Dager SR, Corrigan NM, Richards TL, Posse S. Research applications of magnetic resonance spectroscopy to investigate psychiatric disorders. Top Magn Reson Imaging. 2008;19(2): 81–96.
- Smesny S, Rosburg T, Nenadic I, Fenk KP, Kunstmann S, Rzanny R, et al. Metabolic mapping using 2D 31P-MR spectroscopy reveals frontal and thalamic metabolic abnormalities in schizophrenia. Neuroimage. 2007;35(2):729–37.
- Silverman DH. Brain 18 F-FDG PET in the diagnosis of neurodegenerative dementias: comparison with perfusion SPECT and with clinical evaluations lacking nuclear imaging. J Nucl Med. 2004;45(4):594–607.
- 82. Fusar-Poli P, Meyer-Lindenberg A. Striatal presynaptic dopamine in schizophrenia, part II: meta-analysis of [(18)F/(11)C]-DOPA PET studies. Schizophr Bull. 2013;39(1):33–42.
- Howes OD, Kambeitz J, Kim E, Stahl D, Slifstein M, Abi-Dargham A, et al. The nature of dopamine dysfunction in schizophrenia and what this means for treatment. Arch Gen Psychiatry. 2012;69(8):776–86.
- Meyer-Lindenberg A, Miletich RS, Kohn PD, Esposito G, Carson RE, Quarantelli M, et al. Reduced prefrontal activity predicts exaggerated striatal dopaminergic function in schizophrenia. Nat Neurosci. 2002;5(3):267–71.
- Gallinat J, Heinz A. Combination of multimodal imaging and molecular genetic information to investigate complex psychiatric disorders. Pharmacopsychiatry. 2006;39 Suppl 1:S76–9.
- Weinberger DR. Implications of normal brain development for the pathogenesis of schizophrenia. Arch Gen Psychiatry. 1987;44(7):660–9.
- Pycock CJ, Kerwin RW, Carter CJ. Effect of lesion of cortical dopamine terminals on subcortical dopamine receptors in rats. Nature. 1980;286(5768):74–6.
- Akbarian S, Kim JJ, Potkin SG, Hagman JO, Tafazzoli A, Bunney WE, et al. Gene expression for glutamic acid decarboxylase is reduced without loss of neurons in prefrontal cortex of schizophrenics. Arch Gen Psychiatry. 1995;52(4):258–66.
- Piper M, Beneyto M, Burne TH, Eyles DW, Lewis DA, McGrath JJ. The neurodevelopmental hypothesis of schizophrenia: convergent clues from epidemiology and neuropathology. Psychiatr Clin North Am. 2012;35(3):571–84.
- 90. Stan AD, Lewis DA. Altered cortical GABA neurotransmission in schizophrenia: insights into novel therapeutic strategies. Curr Pharm Biotechnol. 2012;13(8):1557–62.
- Vreugdenhil M, Jefferys JG, Celio MR, Schwaller B. Parvalbumin-deficiency facilitates repetitive IPSCs and gamma oscillations in the hippocampus. J Neurophysiol. 2003; 89(3):1414–22.
- Moghaddam B, Javitt D. From revolution to evolution: the glutamate hypothesis of schizophrenia and its implication for treatment. Neuropsychopharmacology. 2012;37(1):4–15.
- Javitt DC, Zukin SR. Recent advances in the phencyclidine model of schizophrenia. Am J Psychiatry. 1991;148(10):1301–8.

- 94. Javitt DC, Steinschneider M, Schroeder CE, Arezzo JC. Role of cortical *N*-methyl-D-aspartate receptors in auditory sensory memory and mismatch negativity generation: implications for schizophrenia. Proc Natl Acad Sci U S A. 1996;93(21):11962–7.
- 95. Harrison PJ, Law AJ, Eastwood SL. Glutamate receptors and transporters in the hippocampus in schizophrenia. Ann N Y Acad Sci. 2003;1003:94–101.
- Glausier JR, Lewis DA. Dendritic spine pathology in schizophrenia. Neuroscience. 2013; 251:90–107.
- 97. Harrison PJ. The neuropathology of schizophrenia. A critical review of the data and their interpretation. Brain. 1999;122(Pt 4):593–624.
- Glantz LA, Lewis DA. Decreased dendritic spine density on prefrontal cortical pyramidal neurons in schizophrenia. Arch Gen Psychiatry. 2000;57(1):65–73.
- Höistad M, Segal D, Takahashi N, Sakurai T, Buxbaum JD, Hof PR. Linking white and grey matter in schizophrenia: oligodendrocyte and neuron pathology in the prefrontal cortex. Front Neuroanat. 2009;3:9.
- 100. Kovalenko S, Bergmann A, Schneider-Axmann T, Ovary I, Majtenyi K, Havas L, et al. Regio entorhinalis in schizophrenia: more evidence for migrational disturbances and suggestions for a new biological hypothesis. Pharmacopsychiatry. 2003;36 Suppl 3:S158–61.
- 101. Daskalakis ZJ, Christensen BK, Fitzgerald PB, Chen R. Dysfunctional neural plasticity in patients with schizophrenia. Arch Gen Psychiatry. 2008;65(4):378–85.
- Mears RP, Spencer KM. Electrophysiological assessment of auditory stimulus-specific plasticity in schizophrenia. Biol Psychiatry. 2012;71(6):503–11.
- 103. Hashimoto T, Bergen SE, Nguyen QL, Xu B, Monteggia LM, Pierri JN, et al. Relationship of brain-derived neurotrophic factor and its receptor TrkB to altered inhibitory prefrontal circuitry in schizophrenia. J Neurosci. 2005;25(2):372–83.
- 104. Favalli G, Li J, Belmonte-de-Abreu P, Wong AH, Daskalakis ZJ. The role of BDNF in the pathophysiology and treatment of schizophrenia. J Psychiatr Res. 2012;46(1):1–11.
- 105. Vinogradov S, Fisher M, Holland C, Shelly W, Wolkowitz O, Mellon SH. Is serum brainderived neurotrophic factor a biomarker for cognitive enhancement in schizophrenia? Biol Psychiatry. 2009;66(6):549–53.
- Erlenmeyer-Kimling L. Neurobehavioral deficits in offspring of schizophrenic parents: liability indicators and predictors of illness. Am J Med Genet. 2000;97(1):65–71.
- 107. Xu T, Chan RC, Compton MT. Minor physical anomalies in patients with schizophrenia, unaffected first-degree relatives, and healthy controls: a meta-analysis. PLoS One. 2011;6(9):e24129.
- Walker EF, Savoie T, Davis D. Neuromotor precursors of schizophrenia. Schizophr Bull. 1994;20(3):441–51.
- 109. Lieberman JA, Koreen AR, Chakos M, Sheitman B, Woerner M, Alvir JM, et al. Factors influencing treatment response and outcome of first-episode schizophrenia: implications for understanding the pathophysiology of schizophrenia. J Clin Psychiatry. 1996;57 Suppl 9:5–9.
- Voineskos D, Rogasch NC, Rajji TK, Fitzgerald PB, Daskalakis ZJ. A review of evidence linking disrupted neural plasticity to schizophrenia. Can J Psychiatry. 2013;58(2):86–92.
- Keshavan MS. Development, disease and degeneration in schizophrenia: a unitary pathophysiological model. J Psychiatr Res. 1999;33(6):513–21.
- 112. Grace AA. Cortical regulation of subcortical dopamine systems and its possible relevance to schizophrenia. J Neural Transm Gen Sect. 1993;91(2–3):111–34.
- 113. Keshavan MS, Morris DW, Sweeney JA, Pearlson G, Thaker G, Seidman LJ, et al. A dimensional approach to the psychosis spectrum between bipolar disorder and schizophrenia: the schizo-bipolar scale. Schizophr Res. 2011;133(1–3):250–4.
- 114. Hill SK, Reilly JL, Keefe RS, Gold JM, Bishop JR, Gershon ES, et al. Neuropsychological impairments in schizophrenia and psychotic bipolar disorder: findings from the bipolar and schizophrenia network on intermediate phenotypes (B-SNIP) study. Am J Psychiatry. 2013;170(11):1275–84.
- 115. Ivleva EI, Morris DW, Moates AF, Suppes T, Thaker GK, Tamminga CA. Genetics and intermediate phenotypes of the schizophrenia—bipolar disorder boundary. Neurosci Biobehav Rev. 2010;34(6):897–921.

- Purcell SM, Wray NR, Stone JL, Visscher PM, O'Donovan MC, Sullivan PF, et al. Common polygenic variation contributes to risk of schizophrenia and bipolar disorder. Nature. 2009;460(7256):748–52.
- 117. Smoller JW, Craddock N, Kendler K, Lee PH, Neale BM, Nurnberger JI, et al. Identification of risk loci with shared effects on five major psychiatric disorders: a genome-wide analysis. Lancet. 2013;381(9875):1371–9.
- 118. Insel TR. Rethinking schizophrenia. Nature. 2010;468(7321):187-93.
- 119. Keshavan MS. DSM-5 and incremental progress in psychiatric nosology. Asian J Psychiatr. 2013;6(2):97–8.
- 120. Cuthbert B, Insel T. The data of diagnosis: new approaches to psychiatric classification. Psychiatry. 2010;73(4):311–4.
- Keshavan MS, Clementz BA, Pearlson GD, Sweeney JA, Tamminga CA. Reimagining psychoses: an agnostic approach to diagnosis. Schizophr Res. 2013;146(1–3):10–6.
- 122. Keefe RS. Should cognitive impairment be included in the diagnostic criteria for schizophrenia? World Psychiatry. 2008;7(1):22–8.
- 123. Kapur S, Phillips AG, Insel TR. Why has it taken so long for biological psychiatry to develop clinical tests and what to do about it? Mol Psychiatry. 2012;17(12):1174–9.
- 124. Borgwardt S, Fusar-Poli P. Third-generation neuroimaging in early schizophrenia: translating research evidence into clinical utility. Br J Psychiatry. 2012;200(4):270–2.
- 125. Vaccarino FM, Stevens HE, Kocabas A, Palejev D, Szekely A, Grigorenko EL, et al. Induced pluripotent stem cells: a new tool to confront the challenge of neuropsychiatric disorders. Neuropharmacology. 2011;60(7–8):1355–63.
- 126. Maxwell KN, Domchek SM. Cancer treatment according to BRCA1 and BRCA2 mutations. Nat Rev Clin Oncol. 2012;9(9):520–8.
- 127. Boyle MP, De Boeck K. A new era in the treatment of cystic fibrosis: correction of the underlying CFTR defect. Lancet. 2013;1(2):158–63.
- 128. Tandon R, Nasrallah HA, Keshavan MS. Schizophrenia, "just the facts" 5. Treatment and prevention. Past, present, and future. Schizophr Res. 2010;122(1–3):1–23.
- 129. Keller WR, Kum LM, Wehring HJ, Koola MM, Buchanan RW, Kelly DL. A review of antiinflammatory agents for symptoms of schizophrenia. J Psychopharmacol. 2013;27(4):337–42.
- 130. Prasad KM, Eack SM, Keshavan MS, Yolken RH, Iyengar S, Nimgaonkar VL. Antiherpes virus-specific treatment and cognition in schizophrenia: A test-of-concept randomized double-blind placebo-controlled trial. Schizophr Bull. 2013;39(4):857–66.
- 131. Phillips LJ, McGorry PD, Garner B, Thompson KN, Pantelis C, Wood SJ, et al. Stress, the hippocampus and the hypothalamic-pituitary-adrenal axis: implications for the development of psychotic disorders. Aust N Z J Psychiatry. 2006;40(9):725–41.
- 132. Tandon R, Keshavan MS, Nasrallah HA. Schizophrenia, "Just the Facts": what we know in 2008 part 1: overview. Schizophr Res. 2008;100(1–3):4–19.

Chapter 5 Genetics of Schizophrenia

Alan R. Sanders

Abbreviations

22q11.21DS	22q11.21 deletion syndrome
ANK3	Ankyrin 3
ASD	Autism spectrum disorders
BP	Bipolar disorder
CACNA1C	Alpha subunit of the L-type calcium channel
CNV	Copy number variant
DISC1	Disrupted in schizophrenia 1
DSM	Diagnostic and Statistical Manual of Mental Disorders
eQTNs	Expression quantitative trait nucleotides
GWAS	Genome-wide association study
GWLS	Genome-wide linkage scan
GWS	Genome-wide significant
ISC	International Schizophrenia Consortium
ITIH	Inter-alpha-trypsin inhibitor heavy chains
LD	Linkage disequilibrium
LOD	Logarithm of the odds ratio
MGS	Molecular Genetics of Schizophrenia
NRGN	Neurogranin (protein kinase C substrate, RC3)
NRXN1	Neurexin 1
SGENE	Schizophrenia Genetics Consortium
SNP	Single nucleotide polymorphism
MAF	Minor allele frequency
NRXN1 SGENE SNP MAF	Neurexin 1 Schizophrenia Genetics Consortium Single nucleotide polymorphism Minor allele frequency

A.R. Sanders (🖂)

Department of Psychiatry and Behavioral Sciences, NorthShore University HealthSystem

University of Chicago, 1001 University Place, Evanston, IL 60201, USA e-mail: alan.sanders.md@gmail.com

PGC-SZ	Psychiatric Genomics Consortium for Schizophrenia
TCF4	Transcription factor 4
VCFS	Velocardiofacial syndrome
VIPR2	Vasoactive intestinal peptide receptor 2
WGS	Whole-genome sequencing
xMHC	Extended major histocompatibility complex

Introduction

The broad goals of psychiatric genetics are to locate susceptibility genes, understand how implicated genetic variants contribute to illness, and generally increase knowledge of pathophysiology. Researchers hope such advances will contribute to improved diagnostics, perhaps enabling prevention efforts and the development of new treatments. The general inaccessibility of the organ of interest (brain), the relative lack of specific well-defined neuropathology (e.g., compared with Parkinson's disease), and the immense complexity of the brain further heighten the appeal of genetic approaches to schizophrenia research which do not require substantial pathophysiological knowledge beforehand. Many human neuropsychiatric disorders, as well as behavioral traits, have substantial genetic components. These include diseases as diverse as alcoholism, Alzheimer's dementia, antisocial personality disorder, attention-deficit hyperactivity disorder, autism spectrum disorders (ASD), bipolar disorder (BP), depression, substance abuse/ dependence, anorexia nervosa, bulimia nervosa, Huntington's chorea, obsessivecompulsive disorder, panic disorder, schizophrenia, intellectual disability (ID), and Tourette's syndrome. These conditions (and many common nonpsychiatric disorders with partial genetic causation) are mostly considered as complex (multifactorial) genetic disorders. This means many and diverse genetic contributions are present along with significant environmental contributions and perhaps interactions among these components. Research approaches used to study the genetics of schizophrenia are also similarly employed with other complex genetic disorders and all benefit from the knowledge and technology derived from big science efforts, exemplified by the Human Genome Project (www.genome.gov/10001772). These approaches are roughly divided into the earlier (still ongoing) approaches which depend only on reliable phenotyping (diagnosing) and accurate characterization of familial relationships, and later approaches which also require working with biomaterials, most typically, DNA. The earlier approaches are often referred to as genetic epidemiology, while the later approaches start with linkage and progress through and beyond DNA re-sequencing. This chapter discusses highlights of each of the main approaches, summarizes current knowledge, and comments on future directions.

Genetic Epidemiology

Genetic epidemiological approaches include family studies, twin studies, adoption studies, and segregation analyses. Family studies compare the risk for illness in first-degree relatives of probands (first identified ill individual) to the risk for illness in family members of controls. Family history of schizophrenia is the strongest common predictor of illness elevating risk (~tenfold in first-degree relatives); however, most patients have no first-degree relatives with the disorder (called sporadic cases) [1, 2]. An early clue to schizophrenia being a complex genetic disorder (as opposed to a simpler Mendelian single gene disorder) was that family studies showed the risk for illness diminished much more quickly than the degree of biological relationship did. Indeed, the risk for third-degree relatives approaches the prevalence found in the general population. Twin studies compare the diagnostic concordance (agreement) rate by twin type (i.e., monozygotic or identical twins) where phenotypic differences are attributed to the environment versus dizygotic or fraternal twins (the latter typically being same-sex twin pairs) where genetic differences are also in play. Many twin studies were conducted (see examples), which consistently showed high heritability of around 80 %; however, there are many examples of identical twin pairs where one twin has schizophrenia and the co-twin does not-another clue to complex genetics [3-8]. Adoption studies compare the risk for illness in biological versus adoptive relatives of probands. Though there are fewer studies as one progresses from family to twin to adoption studies, schizophrenia has a substantial number of the latter. The general conclusions are that risk for illness travels with the biological relationship and not the adoptive relationship. Further, this risk does not depend on the rearing environment, including whether or not a parent with schizophrenia is present [9]. Segregation analyses attempted to fit observed families with known modes of inheritance (e.g., autosomal versus sex-linked; dominant versus recessive). While such analyses were much more successful for Mendelian disorders, they did not bear fruit for schizophrenia genetics.

In summary, over many decades using various methods of ascertainment, assessment, and phenotyping/diagnosis (i.e., Kraepelin to the Diagnostic and Statistical Manual of Mental Disorders), epidemiological studies of schizophrenia consistently found genetic contributions to be of primary importance [10, 11].

Molecular Approaches: Linkage

Molecular approaches burgeoned over recent decades as technological capabilities grew exponentially and costs decreased. While earlier molecular approaches (e.g., linkage and especially association) continue to be employed, more recent molecular approaches, such as exome sequencing and whole-genome sequencing (WGS), and a variety of other "omics" (transcriptomics, methylomics, proteomics, metabolomics), in addition to genomics and pharmacogenetics, are beginning to make larger contributions to our knowledge base. Linkage studies seek to find a linked genetic marker which is inherited with illness within a pedigree. Association studies seek to find an associated DNA sequence which occurs more frequently in patients than in controls. Both approaches assay for the degree of crossing over (or meiotic recombination) between homologous chromosomes which occurs between a polymorphic genetic marker and a putative risk variant contributing to illness. If these two loci are very close to each other on a chromosome, they would likely remain linked if looking at one or two generations (as in a linkage study) or in linkage disequilibrium (LD) over many generations if very close to each other (as in an association study). Earlier linkage studies used parametric logarithm of the odds ratio (LOD) methods to study one or a few large pedigrees with a high density of illness in the hopes this would reduce genetic heterogeneity. Later linkage studies employed nonparametric allele sharing methods, most typically focusing on affected sibling pairs from many more families. Linkage mapping studies are geared to search for one or a few common major gene effects. Over the years, numerous genome-wide linkage scans (GWLS) were performed, but genome-wide significant (GWS) linkage loci remain elusive. The strongest linkage evidence from the largest study, which was a meta-analysis of GWLS, suggested chromosome 8p as being linked to schizophrenia; this was previously supported by some individual dataset GWLS, meta-analyses, and a combined (jointly genotyped) GWLS [12-18]. Many other regions in the genome had lower and less consistent levels of support from various GWLS. Thus, the dearth of significant findings from GWLS supports the notion that common major gene effects are not present to any large extent, even though there may be very rare families with a major gene effect (e.g., DISC1, disrupted in schizophrenia 1) [19].

Molecular Approaches: Candidate Gene Association

Association studies were initially mostly population-based in which a group of unrelated cases were compared to a group of unrelated controls otherwise matched to the case group for differing allele frequencies at the studied genetic markers. Particularly for ancestral differences (race, ethnicity, population history), however, it was difficult to fully match the two groups. Thus, family-based association approaches compared alleles transmitted to affected offspring versus alleles not transmitted by parents in many trios of parents and child, a situation where population history is very well matched. Subsequently, much better methods of controlling for this population stratification were developed and most association studies were done using the population-based approach which enabled much larger sample sizes (case–control pairs being easier to collect than parent–child trios, especially for disorders where most of the cases are adults), resulting in better statistical power. For association studies, often an hypothesis (e.g., dopamine hypothesis) would nominate candidate genes to be tested (akin to a gene having a motive). At other times candidate genes would be tested because they were in the right position (e.g., under a linkage peak; akin to being at the scene of the crime). Since LD acts over much shorter distance than linkage (in a linkage study a few hundred highly informative genetic markers can assay the entire genome), multiple testing becomes a much larger statistical concern in an association study (since many more independent markers can be tested). Metaanalyses of some candidate gene association studies were performed, though overall results were limited. This is especially because the frequently underpowered sample sizes (which due to instability of findings in small samples would sometimes suggest large effects or odds ratios) and often inadequate or loose statistical thresholds contributed greatly to frequent non-replications. The largest unified candidate gene testing study of that era which focused on thoroughly testing 14 of the leading candidate genes of the time by analyzing 789 single nucleotide polymorphisms (SNPs) found no evidence of association [20]. With some rare exceptions, classical candidate genes for schizophrenia have not been upheld in genome-wide association studies (GWAS). Of note, this is a general pattern seen in GWAS of other complex disorders and bolsters the need for adequately powered samples and stringent statistical thresholds for declaring an association [21].

Molecular Approaches: Genome-Wide Association Studies

Continuing technological developments (especially microarrays) enabled hundreds of thousands of genetic markers, mostly tag SNPs (i.e., "tagging" much of the common variation by means of LD, but not necessarily themselves being "causative" SNPs), to be assayed, thus paving the way to search for common variation on a genome-wide scale, resulting in the advent of GWAS [22]. These commercial GWAS SNP arrays are further complemented by imputation, the computational prediction of non-genotyped SNP genotypes from the patterns of genotyped ones [23, 24]. The first combined GWAS in schizophrenia (2009) involved ~27,000 subjects from the Molecular Genetics of Schizophrenia (MGS) collaboration, International Schizophrenia Consortium (ISC), and the Schizophrenia Genetics Consortium (SGENE) [25–27]. This approach provided the first GWS evidence for association [25–27], typically defined as $p < 5 \times 10^{-8}$ [28]. The strongest result was for SNPs at the extended major histocompatibility complex (xMHC) region on chromosome 6p21.3-p22.1 where genes with immune functions and histone genes predominate [25-27]. Two other GWS loci which emerged from this meta-analysis were near NRGN (neurogranin [protein kinase C substrate, RC3]) on chromosome 11q24.2 and at TCF4 (transcription factor 4) on chromosome 18q21.2 [27]. All of these individually detectable common variant GWS loci were of low effect (i.e., odds ratios around 1.1–1.3). Besides these common risk variants, even lower effect variants which are undetectable individually but with observable effects en masse, (i.e., polygenic contributions) were also highlighted in these studies [26, 29]. GWAS also greatly enabled the detection of a large and increasing number of schizophrenia risk copy number variants (CNVs). These structural variants are submicroscopic

deletions or duplications detectable by a number of adjacent SNPs showing alterations in signal intensities. These included 1q21.1, 2p16.3 (NRXN1, neurexin 1), 3q29, 7q36.3 (VIPR2, vasoactive intestinal peptide receptor 2), 7q11.23, 15q13.3, 16p11.2, 17q12, and 22q11.21. Many of these CNVs span multiple and even dozens of genes with some exceptions such as NRXN1 and VIPR2 (see reviews) [30, 31]. These schizophrenia risk CNVs are often de novo (though some were inherited) and rare, but showed higher penetrance (higher odds ratios) than common loci (i.e., intermediate between Mendelian diseases and complex disorders) [32]. Genotype:phenotype correlations, however, are modest for these CNVs (e.g., the same CNV might lead to different phenotypes such as schizophrenia, ID, ASD, and epilepsy). This is referred to as pleiotropy where variation in a gene may contribute to different though often somehow related phenotypes—with the neurodevelopmental hypothesis of schizophrenia being echoed in many of these pleiotropy examples [33, 34]. CNVs can be more straightforward pointers to pathophysiology than GWAS associations to common SNPs because of the "indirect association" with GWAS. Subsequently, additional samples were reported and an even larger metaanalysis of schizophrenia GWAS (i.e., the Psychiatric Genomics Consortium for Schizophrenia, Part 1 [PGC-SZ-1]) found five more GWS loci and further confirmed the xMHC locus and TCF4 [35]. It also provided further support for substantial polygenic contributions to schizophrenia risk [35, 36]. A large multistage GWAS starting with a Swedish national sample provided support for previously detected GWS loci and also yielded 13 additional new GWS loci, some of which implicate calcium signaling in schizophrenia risk [37]. PGC-SZ-2 is currently in progress and reports from meetings indicate it is yielding many more GWS loci. It seems clear for many complex genetic disorders such as schizophrenia that increasing the sample sizes of GWAS meta-analyses will yield even more common low effect loci. Such implicated loci can then be fed into downstream analyses (e.g., network and pathway analyses) aimed at groups of genes, as well as studied on an individual gene level when appropriate. The well-established association of schizophrenia to variation at the xMHC locus is intriguing, given the many immune-related genes located there and previously known associations of schizophrenia to various autoimmune disorders [38, 39].

Still, our knowledge of the specific genes and the biology underlying statistical associations from GWAS remains tentative. For example, the high gene density, strong long-range LD, and the many haplotypes of the xMHC region make classical fine-mapping difficult. The case can be made that gene regulation may underlie some GWAS susceptibility loci. Most common SNP GWAS hits lie outside of genes or are not in LD with polymorphisms affecting the amino acid sequence of a protein; CNVs and trait-associated SNPs from GWAS are usually enriched for expression quantitative trait nucleotides (eQTNs); and GWAS SNPs are enriched in regulatory sequences (DNase I hypersensitive sites, ENCODE). Thus, transcriptomics (mRNA expression) analysis provides data which is closer to function than GWAS and less limited by diffuse boundaries typical of association data [40]. Other "omics" approaches (methylomics, proteomics, metabolomics) may also yield additional insight (see recent reviews) [41–44]. Despite these issues, it is worthwhile to note
that GWAS are far more successful than any previous approach to finding new susceptibility loci for complex disorders, including schizophrenia [21, 45, 46]. One of the main limitations of GWAS, however, is that their statistical power to detect an association with rare alleles (minor allele frequency, MAF<1 %) is very low. To assay for such rare variation, deep re-sequencing (i.e., exome sequencing and WGS) is much more useful. While some initial deep re-sequencing findings indicate that exonic de novo mutations are enriched in schizophrenia, larger samples are needed [47–49].

Genetic Relationships to Bipolar Disorder

There are diagnostic/symptom overlaps between schizophrenia and BP (especially psychosis) which are posited as having some genes in common, with schizoaffective disorder serving as a "bridge" [50–52]. Further, family studies show that schizoaffective disorder is enriched in families of schizophrenia and BP probands [53–58]. Later, larger studies found familial coaggregation of schizophrenia and BP mostly derived from additive genetic effects in common [59, 60]. GWAS for BP found several loci (i.e., detected individual common variants within/near genes), some of which were implicated in a joint GWAS meta-analysis of schizophrenia and BP, including *ANK3 (ankyrin 3), CACNA1C (alpha subunit of the L-type calcium channel*), and the region of *ITIH1-ITIH3 (inter-alpha-trypsin inhibitor heavy chains*) [35, 61]. GWAS for BP also show a polygenic contribution and some overlap with polygenes for schizophrenia [26, 62, 63]. Thus, there is some overlap of susceptibility between BP and schizophrenia for both some individually detectable risk alleles and for polygenic risk. By contrast, CNVs seem to play a smaller part in BP than in schizophrenia.

Clinical Applications and Future Directions

Upcoming clinical applications using knowledge derived from genetic research bear mentioning. Testing for CNVs is increasingly used for ASD and ID and is of some utility for schizophrenia. This is most notable in the setting of symptoms suggestive of 22q11.21del (e.g., psychosis, developmental delay, learning disability, facial and palatal dysmorphology). This is also known as 22q11.21 deletion syndrome (22q11.21DS) or velocardiofacial syndrome (VCFS) due to its comorbidities (especially cardiac) [64]. Identification of individual genes or groups of genes (from network or pathway analyses) can both inform pathophysiology and guide pharmaceutical industry development. Pharmacogenomics and pharmacogenetics should increasingly provide some clinical guidance (medication efficacy, specificity, metabolism), though as of now this is rather limited (see Chap. 9) [65]. Due to the multitude of genetic variants contributing to risk, from individually detectable ones

(now shading into the oligogenic area with PGC-SZ-2) to polygenes, it is likely that any future genetic tests for schizophrenia risk prediction will largely be composite tests assaying for many variants at once. Finally, as we acquire more knowledge about genetic contributions to schizophrenia, we may be able to improve prediction and prevention efforts by tying some of them into environmental risk factors for schizophrenia (e.g., obstetrical complications; urban birth/residence and migrant status; famine; seasonal effects via prenatal infections such as influenza; adolescent cannabis use; and advanced paternal age) [66–74].

Conclusion

With the relatively recent advent of the GWAS era, schizophrenia genetics research has yielded an unprecedented accumulation of new risk loci. Further, our knowledge is aided greatly by technological developments and improved and more rigorous statistical analyses. This is further supported by the large scale cooperation of the scientific community via rapid data sharing and very large meta-analyses. Future progress looks promising as these trends continue and are complemented by the addition of next-generation sequencing (WGS and exomic) in much deeper samples; functional study of statistically implicated variants; and eventually the study of human cellular models such as via neurons differentiated from induced pluripotent stem cells [75].

Glossary

- **ENCODE** Refers to the *Encyclopedia Of DNA Elements* project to identify all functional elements in the human genome sequence
- **Exome sequencing** The DNA sequencing of the exons in the genome, most often referring to the protein coding exons, i.e., not including all untranslated regions (UTRs)
- **Expression quantitative trait nucleotides (eQTNs)** Are SNPs that regulate expression levels of mRNAs or proteins
- **Genomics** Refers to mapping or sequencing of whole genomes (as opposed to focusing on an individual gene)
- Linkage disequilibrium Abbreviated LD, the nonrandom association of alleles (alternative forms) at two or more loci, which descended from an ancestral chromosome
- **Metabolomics** The study of the whole set of metabolites (small molecule end products of cellular processes, i.e., metabolism) of a biological unit (cell, tissue, organ, individual)
- **Methylomics** The study of DNA methylation (which affects gene expression) on a genome-wide scale

- **Network or pathway analyses** The study of a biological system with subunits connected into a larger patterns (networks, pathways), such as a protein–protein interaction network or a gene regulatory (e.g., DNA–protein interaction) network
- **Nonparametric linkage analysis** A type of linkage analysis where no assumptions or specifications are made regarding the genetic model for the disorder
- **Parametric linkage analysis** A type of linkage analysis where the genetic model for the disorder must be specified, i.e., allele frequency and penetrance (likelihood a risk genotype will be phenotypically expressed) parameters
- **Pharmacogenetics** The study of genetic differences affecting metabolic pathways important for drug response (including both therapeutic and side effects)
- **Polygenic contributions** The situation when variants at many genes contribute to a phenotype, as opposed to the phenotype arising from variation at one (monogenic) or a few genes (oligogenic)
- **Polymorphic** Having more than one form, as in a genetic maker with more than one allele (alternative form)
- Proteomics The study of an entire complement of proteins of a biological unit
- **Transcriptomics** The study of the set of all RNA (mRNA, rRNA, tRNA, and other noncoding RNA) of a biological unit (typically a collection of cells or a tissue), though in many instances limited to the study of mRNA
- **Whole-genome sequencing** The DNA sequencing of the whole genome of an individual, i.e., all chromosomes and also mitochondrial DNA

References

- 1. Gottesman II, Shields J. Schizophrenia: the epigenetic puzzle. Cambridge, MA: Cambridge University Press; 1982.
- 2. Yang J, Visscher PM, Wray NR. Sporadic cases are the norm for complex disease. Eur J Hum Genet. 2010;18(9):1039–43.
- Klaning U, Mortensen PB, Kyvik KO. Increased occurrence of schizophrenia and other psychiatric illnesses among twins. Br J Psychiatry. 1996;168(6):688–92.
- 4. Cannon TD, Kaprio J, Lonnqvist J, Huttunen M, Koskenvuo M. The genetic epidemiology of schizophrenia in a Finnish twin cohort. A population-based modeling study. Arch Gen Psychiatry. 1998;55(1):67–74.
- 5. Franzek E, Beckmann H. Different genetic background of schizophrenia spectrum psychoses: a twin study. Am J Psychiatry. 1998;155(1):76–83.
- Cardno AG, Marshall EJ, Coid B, Macdonald AM, Ribchester TR, Davies NJ, et al. Heritability estimates for psychotic disorders: the Maudsley twin psychosis series. Arch Gen Psychiatry. 1999;56(2):162–8.
- Cardno AG, Gottesman I. Twin studies of schizophrenia: from bow-and-arrow concordances to Star Wars Mx and functional genomics. Am J Med Genet. 2000;97(1):12–7.
- Sullivan PF, Kendler KS, Neale MC. Schizophrenia as a complex trait: evidence from a metaanalysis of twin studies. Arch Gen Psychiatry. 2003;60(12):1187–92.
- 9. Ingraham LJ, Kety SS. Adoption studies of schizophrenia. Am J Med Genet. 2000;97(1):18-22.
- Kraepelin E. Manic-depressive insanity and Paranoia. Edinburgh: E. & S. Livingstone (English translation 1921); 1899.
- 11. APA. Diagnostic and statistical manual of mental disorders: DSM-IV-TR. 4th ed., text revision ed. Washington, DC: American Psychiatric Association; 2000.

- 12. Ng MY, Levinson DF, Faraone SV, Suarez BK, DeLisi LE, Arinami T, et al. Meta-analysis of 32 genome-wide linkage studies of schizophrenia. Mol Psychiatry. 2009;14(8):774–85.
- 13. Blouin JL, Dombroski BA, Nath SK, Lasseter VK, Wolyniec PS, Nestadt G, et al. Schizophrenia susceptibility loci on chromosomes 13q32 and 8p21. Nat Genet. 1998;20(1):70–3.
- 14. Stefansson H, Sigurdsson E, Steinthorsdottir V, Bjornsdottir S, Sigmundsson T, Ghosh S, et al. Neuregulin 1 and susceptibility to schizophrenia. Am J Hum Genet. 2002;71(4):877–92.
- 15. Suarez BK, Duan J, Sanders AR, Hinrichs AL, Jin CH, Hou C, et al. Genomewide linkage scan of 409 European-ancestry and African American families with schizophrenia: suggestive evidence of linkage at 8p23.3-p21.2 and 11p13.1-q14.1 in the combined sample. Am J Hum Genet. 2006;78(2):315–33.
- 16. Badner JA, Gershon ES. Meta-analysis of whole-genome linkage scans of bipolar disorder and schizophrenia. Mol Psychiatry. 2002;7(4):405–11.
- Lewis CM, Levinson DF, Wise LH, DeLisi LE, Straub RE, Hovatta I, et al. Genome scan metaanalysis of schizophrenia and bipolar disorder, part II: schizophrenia. Am J Hum Genet. 2003;73(1):34–48.
- Holmans PA, Riley B, Pulver AE, Owen MJ, Wildenauer DB, Gejman PV, et al. Genomewide linkage scan of schizophrenia in a large multicenter pedigree sample using single nucleotide polymorphisms. Mol Psychiatry. 2009;14(8):786–95.
- Blackwood DH, Fordyce A, Walker MT, St Clair DM, Porteous DJ, Muir WJ. Schizophrenia and affective disorders–cosegregation with a translocation at chromosome 1q42 that directly disrupts brain-expressed genes: clinical and P300 findings in a family. Am J Hum Genet. 2001;69(2):428–33.
- Sanders AR, Duan J, Levinson DF, Shi J, He D, Hou C, et al. No significant association of 14 candidate genes with schizophrenia in a large European ancestry sample: implications for psychiatric genetics. Am J Psychiatry. 2008;165(4):497–506.
- 21. Hindorff LA, Sethupathy P, Junkins HA, Ramos EM, Mehta JP, Collins FS, et al. Potential etiologic and functional implications of genome-wide association loci for human diseases and traits. Proc Natl Acad Sci U S A. 2009;106(23):9362–7.
- Li M, Li C, Guan W. Evaluation of coverage variation of SNP chips for genome-wide association studies. Eur J Hum Genet. 2008;16(5):635–43.
- 23. Halperin E, Stephan DA. SNP imputation in association studies. Nat Biotechnol. 2009; 27(4):349–51.
- Nothnagel M, Ellinghaus D, Schreiber S, Krawczak M, Franke A. A comprehensive evaluation of SNP genotype imputation. Hum Genet. 2009;125(2):163–71.
- 25. Shi J, Levinson DF, Duan J, Sanders AR, Zheng Y, Pe'er I, et al. Common variants on chromosome 6p22.1 are associated with schizophrenia. Nature. 2009;460(7256):753–7.
- Purcell SM, Wray NR, Stone JL, Visscher PM, O'Donovan MC, Sullivan PF, et al. Common polygenic variation contributes to risk of schizophrenia and bipolar disorder. Nature. 2009;460(7256):748–52.
- 27. Stefansson H, Ophoff RA, Steinberg S, Andreassen OA, Cichon S, Rujescu D, et al. Common variants conferring risk of schizophrenia. Nature. 2009;460(7256):744–7.
- Dudbridge F, Gusnanto A. Estimation of significance thresholds for genomewide association scans. Genet Epidemiol. 2008;32(3):227–34.
- 29. Gottesman II, Shields J. A polygenic theory of schizophrenia. Proc Natl Acad Sci U S A. 1967;58(1):199–205.
- Gill M. Developmental psychopathology: the role of structural variation in the genome. Dev Psychopathol. 2012;24(4):1319–34.
- Malhotra D, Sebat J. CNVs: harbingers of a rare variant revolution in psychiatric genetics. Cell. 2012;148(6):1223–41.
- 32. Veltman JA, Brunner HG. De novo mutations in human genetic disease. Nat Rev Genet. 2012;13(8):565–75.
- Harrison PJ. The neuropathology of schizophrenia. A critical review of the data and their interpretation. Brain. 1999;122(Pt 4):593–624.

5 Genetics of Schizophrenia

- Owen MJ, O'Donovan MC, Thapar A, Craddock N. Neurodevelopmental hypothesis of schizophrenia. Br J Psychiatry. 2011;198(3):173–5.
- Ripke S, Sanders AR, Kendler KS, Levinson DF, Sklar P, Holmans PA, et al. Genome-wide association study identifies five new schizophrenia loci. Nat Genet. 2011;43(10):969–76.
- Lee SH, DeCandia TR, Ripke S, Yang J, Sullivan PF, Goddard ME, et al. Estimating the proportion of variation in susceptibility to schizophrenia captured by common SNPs. Nat Genet. 2012;44(3):247–50.
- Ripke S, O'Dushlaine C, Chambert K, Moran JL, KŠhler AK, Akterin S, et al. Genome-wide association analysis identifies 13 new risk loci for schizophrenia. Nat Genet. 2013;45(10):1150–9. doi:10.1038/ng.2742.
- Horton R, Wilming L, Rand V, Lovering RC, Bruford EA, Khodiyar VK, et al. Gene map of the extended human MHC. Nat Rev Genet. 2004;5(12):889–99.
- 39. Eaton WW, Byrne M, Ewald H, Mors O, Chen CY, Agerbo E, et al. Association of schizophrenia and autoimmune diseases: linkage of Danish national registers. Am J Psychiatry. 2006;163(3):521–8.
- Sequeira PA, Martin MV, Vawter MP. The first decade and beyond of transcriptional profiling in schizophrenia. Neurobiol Dis. 2012;45(1):23–36.
- Dempster E, Viana J, Pidsley R, Mill J. Epigenetic studies of schizophrenia: progress, predicaments, and promises for the future. Schizophr Bull. 2013;39(1):11–6.
- Pich EM, Vargas G, Domenici E. Biomarkers for antipsychotic therapies. Handb Exp Pharmacol. 2012;212:339–60.
- 43. Grayson DR, Guidotti A. The dynamics of DNA methylation in schizophrenia and related psychiatric disorders. Neuropsychopharmacology. 2013;38(1):138–66.
- Labrie V, Pai S, Petronis A. Epigenetics of major psychosis: progress, problems and perspectives. Trends Genet. 2012;28(9):427–35.
- 45. Manolio TA, Collins FS, Cox NJ, Goldstein DB, Hindorff LA, Hunter DJ, et al. Finding the missing heritability of complex diseases. Nature. 2009;461(7265):747–53.
- 46. Sullivan P. Don't give up on GWAS. Mol Psychiatry. 2012;17(1):2–3.
- 47. Girard SL, Gauthier J, Noreau A, Xiong L, Zhou S, Jouan L, et al. Increased exonic de novo mutation rate in individuals with schizophrenia. Nat Genet. 2011;43(9):860–3.
- 48. Xu B, Roos JL, Dexheimer P, Boone B, Plummer B, Levy S, et al. Exome sequencing supports a de novo mutational paradigm for schizophrenia. Nat Genet. 2011;43(9):864–8.
- 49. Need AC, McEvoy JP, Gennarelli M, Heinzen EL, Ge D, Maia JM, et al. Exome sequencing followed by large-scale genotyping suggests a limited role for moderately rare risk factors of strong effect in schizophrenia. Am J Hum Genet. 2012;91(2):303–12.
- Schulze TG, Hedeker D, Zandi P, Rietschel M, McMahon FJ. What is familial about familial bipolar disorder? Resemblance among relatives across a broad spectrum of phenotypic characteristics. Arch Gen Psychiatry. 2006;63(12):1368–76.
- 51. Goes FS, Zandi PP, Miao K, McMahon FJ, Steele J, Willour VL, et al. Mood-incongruent psychotic features in bipolar disorder: familial aggregation and suggestive linkage to 2p11-q14 and 13q21-33. Am J Psychiatry. 2007;164(2):236–47.
- 52. Kasanin J. The acute schizoaffective psychoses. Am J Psychiatry. 1933;90(1):97-126.
- Maier W, Lichtermann D, Minges J, Hallmayer J, Heun R, Benkert O, et al. Continuity and discontinuity of affective disorders and schizophrenia. Results of a controlled family study. Arch Gen Psychiatry. 1993;50(11):871–83.
- Maier W, Lichtermann D, Franke P, Heun R, Falkai P, Rietschel M. The dichotomy of schizophrenia and affective disorders in extended pedigrees. Schizophr Res. 2002;57(2–3):259–66.
- 55. Gershon ES, DeLisi LE, Hamovit J, Nurnberger Jr JI, Maxwell ME, Schreiber J, et al. A controlled family study of chronic psychoses. Schizophrenia and schizoaffective disorder. Arch Gen Psychiatry. 1988;45(4):328–36.
- Gershon ES, Hamovit J, Guroff JJ, Dibble E, Leckman JF, Sceery W, et al. A family study of schizoaffective, bipolar I, bipolar II, unipolar, and normal control probands. Arch Gen Psychiatry. 1982;39(10):1157–67.

- Kendler KS, McGuire M, Gruenberg AM, O'Hare A, Spellman M, Walsh D. The Roscommon Family Study. I. Methods, diagnosis of probands, and risk of schizophrenia in relatives. Arch Gen Psychiatry. 1993;50(7):527–40.
- Valles V, Van Os J, Guillamat R, Gutierrez B, Campillo M, Gento P, et al. Increased morbid risk for schizophrenia in families of in-patients with bipolar illness. Schizophr Res. 2000; 42(2):83–90.
- 59. Van Snellenberg JX, de Candia T. Meta-analytic evidence for familial coaggregation of schizophrenia and bipolar disorder. Arch Gen Psychiatry. 2009;66(7):748–55.
- Lichtenstein P, Yip BH, Bjork C, Pawitan Y, Cannon TD, Sullivan PF, et al. Common genetic determinants of schizophrenia and bipolar disorder in Swedish families: a population-based study. Lancet. 2009;373(9659):234–9.
- 61. Craddock N, Sklar P. Genetics of bipolar disorder. Lancet. 2013;381(9878):1654-62.
- Sklar P, Smoller JW, Fan J, Ferreira MA, Perlis RH, Chambert K, et al. Whole-genome association study of bipolar disorder. Mol Psychiatry. 2008;13(6):558–69.
- 63. WTCCC. Wellcome Trust Case Control Consortium: Genome-wide association study of 14,000 cases of seven common diseases and 3,000 shared controls. Nature. 2007;447(7145):661–78.
- 64. Costain G, Bassett AS. Clinical applications of schizophrenia genetics: genetic diagnosis, risk, and counseling in the molecular era. Appl Clin Genet. 2012;5:1–18.
- Malhotra AK, Zhang JP, Lencz T. Pharmacogenetics in psychiatry: translating research into clinical practice. Mol Psychiatry. 2012;17(8):760–9.
- 66. Clarke MC, Kelleher I, Clancy M, Cannon M. Predicting risk and the emergence of schizophrenia. Psychiatr Clin North Am. 2012;35(3):585–612.
- Byrne M, Agerbo E, Bennedsen B, Eaton WW, Mortensen PB. Obstetric conditions and risk of first admission with schizophrenia: a Danish national register based study. Schizophr Res. 2007;97(1–3):51–9.
- Mittal VA, Ellman LM, Cannon TD. Gene-environment interaction and covariation in schizophrenia: the role of obstetric complications. Schizophr Bull. 2008;34(6):1083–94.
- 69. Susser ES, Lin SP. Schizophrenia after prenatal exposure to the Dutch Hunger Winter of 1944–1945. Arch Gen Psychiatry. 1992;49(12):983–8.
- 70. St Clair D, Xu M, Wang P, Yu Y, Fang Y, Zhang F, et al. Rates of adult schizophrenia following prenatal exposure to the Chinese famine of 1959–1961. JAMA. 2005;294(5):557–62.
- McGrath J, Saha S, Chant D, Welham J. Schizophrenia: a concise overview of incidence, prevalence, and mortality. Epidemiol Rev. 2008;30:67–76.
- Casadio P, Di Forti M, Murray RM. Cannabis use as a component cause of schizophrenia. In: Brown AS, Patterson PH, editors. The origins of schizophrenia. New York: Columbia University Press; 2011. p. 157–75.
- 73. Malaspina D, Harlap S, Fennig S, Heiman D, Nahon D, Feldman D, et al. Advancing paternal age and the risk of schizophrenia. Arch Gen Psychiatry. 2001;58(4):361–7.
- 74. Torrey EF, Buka S, Cannon TD, Goldstein JM, Seidman LJ, Liu T, et al. Paternal age as a risk factor for schizophrenia: how important is it? Schizophr Res. 2009;114(1–3):1–5.
- Brennand KJ, Simone A, Tran N, Gage FH. Modeling psychiatric disorders at the cellular and network levels. Mol Psychiatry. 2012;17(12):1239–53.

Chapter 6 Recent Advances in Neuroimaging Biomarkers of Schizophrenia

Lei Wang and John G. Csernansky

Abbreviations for fMRI Tasks

- CP Context processing
- CPT Continuous performance task
- DMS Delayed match to sample
- EF Executive function, including DMS, GNG, MA, NB, OB, SR, ST, WCS, and WG
- EL Emotion-labeling
- EME Episodic memory, encoding
- EMR Episodic memory, retrieval
- FF Fearful faces
- FN Fear and neutral faces
- GNG Go/No-Go
- MA Mental arithmetic
- NB N-Back
- NF Negative faces
- NI Negative images vs. neutral images
- NV Nonvisual speech
- OB Oddball

e-mail: leiwang1@northwestern.edu

J.G. Csernansky

L. Wang (🖂)

Department of Psychiatry and Behavioral Sciences and Radiology, Northwestern University Feinberg School of Medicine, 710 N. Lake Shore Drive, Abbott Hall 1322, Chicago, IL 60611, USA

Department of Psychiatry and Behavioral Sciences, Northwestern University Feinberg School of Medicine, 446 E. Ontario, Suite 7-200, Chicago, IL 60610, USA e-mail: jgc@northwestern.edu

- PC Pavlovian conditioning
- RI Response inhibition
- RP Reward prediction
- SA Speech appraisal
- SL Listening to speech
- SP Speech/non-speech
- SR Sequence recall
- ST Stroop
- VB Verbal fluency
- WCS Wisconsin card sorting
- WG Word generation
- WM Working memory

Abbreviations for Gray Matter ROIs

Anterior cingulate cortex
Amygdala
Basal ganglia
Caudate
Cingulate gyrus
Dorsolateral prefrontal cortex
Entorhinal cortex
Globus pallidus
Heschl's gyrus
Hippocampus
Inferior frontal gyrus
Inferior parietal lobule
Medial prefrontal cortex
Medial superior frontal gyrus
Middle temporal gyrus
Medial temporal lobe
Nucleus accumbens
Orbitofrontal gyrus
Posterior cingulate cortex
Prefrontal cortex
Parahippocampal gyrus
Posterior parietal cortex
Planum temporale
Putamen
Superior temporal gyrus
Thalamus
Ventrolateral prefrontal cortex
Ventral striatum

Abbreviations for White Matter ROI

- AF Arcuate fasciculus, connecting STG and IPL with inferior frontal gyrus. AF is important in language processing
- ALIC Anterior limb of the internal capsule
- CB Cingulum bundle, connecting paralimbic-neocortical brain regions, also connecting limbic structures including DLPFC, CG, PHG, and AG. CB is involved in a number of functions, including pain perception, emotion, self-monitoring, and spatial orientation and memory
- CC Corpus callosum
- FX Fornix
- ILF Inferior longitudinal fasciculus, connecting the anterior temporal and occipital regions
- IOF Inferior occipitofrontal fasciculus, connecting the frontal with occipital and temporal lobes
- PLIC Posterior limb of the internal capsule
- SLF Superior longitudinal fasciculus, connecting the frontal, occipital, parietal, and temporal lobes
- UF Uncinate fasciculus, connecting OFG and IFG with the anterior pole and the AG. UF is involved functionally in decision making, autobiographical and episodic memory, as well as in social behavior

Abbreviations for White Matter

- FA Fractional anisotropy
- MD Mean diffusivity

Abbreviations for Other

DMN Default mode network

Overview

Since Kraepelin (1917) first described "dementia praecox" and Bleuler (1911) coined the term "schizophrenia," and until Johnstone and colleagues (1976) conducted the first neuroimaging study in schizophrenia with computerized tomography (CT), the search for the cause(s) and cure of schizophrenia has been primarily conducted in the dark [1–4]. This is because investigators were unable to image the living brain. In the ensuing three decades, numerous advances in neuroimaging were introduced. These included positron emission tomography (PET) for brain energy (glucose)

consumption (1979); structural magnetic resonance imaging (MRI, sMRI) for brain structure (1980); functional MRI (BOLD, or blood oxygen level dependent, fMRI) for brain activity (1990); and diffusion-weighted imaging (DWI) for white matter tracts that connect different regions of the brain (1990) [5–15]. The development of other neuroimaging technologies, such as magnetic resonance spectroscopy (MRS) and arterial spin labeling (ASL), combined with evolving data analysis methods and software tools, pushed open a window to the brain [16, 17]. Since the first MRI study of schizophrenia, neuroimaging research has provided insights into the neural underpinning and heterogeneity of the illness [18].

Decades before modern technology discerned anatomic and functional connections in the living brain, Geschwind put forth the visionary suggestion that connections between regions that subserved specific behavioral functions are critical to overall mental functioning [19]. The latest developments in modern neuroimaging technology and analytic methods are revealing the overall organization of the functional brain network, and Geschwind's inference on the "disconnexion syndromes in animals and man" has evolved into "the disconnection hypothesis" that schizophrenia is a syndrome of disconnection of neural networks [20–23].

Many excellent reviews have been published on the topic of schizophrenia and neuroimaging (PubMed search revealed close to 700 reviews since 1988 and over 175 since 2010), starting with early comprehensive reviews of brain structure by Pearlson and Marsh, McCarley et al., Shenton et al., and Buckley [4, 24–26]. More recent reviews include Shenton et al. on gray matter; Karlsgodt et al. and Thomason and Thompson on white matter; Waddington, McGuire et al., and Mueller et al. on function; and Fornito et al. and Pettersson-Yeo et al. on connectivity [23, 27–33]. There are now close to 3,000 papers reporting schizophrenia neuroimaging studies in the last 25 years and close to 1,000 since 2010.

In this chapter, we describe more recent developments of neuroimaging measures of regions of interest (ROIs) and functional networks, as well as multimodal integration and pattern analysis in schizophrenia.

Distributed Patterns of Widespread Abnormal Structure and Function

Volumetric measurement of manually traced ROIs in structural MRIs was the predominant method in the first decade of schizophrenia neuroimaging research. Andreasen et al. published the first quantitative MRI study of schizophrenia (e.g., see reviews by Lawrie and Abukmeil, McCarley et al., and Pearlson and Marsh) [24, 25, 34, 35]. These early studies unequivocally demonstrated that schizophrenia is associated with patterns of fundamental anatomic abnormalities almost everywhere in the brain. These patterns are subtle, but widespread, and include enlarged ventricular spaces and gray matter losses predominantly in the medial temporal and frontal lobes, but also in the midline limbic structures and the deep thalamic and striatal nuclei [24–26, 35]. Chapter 4 of this book

provides a more comprehensive summary of brain structural abnormalities and their clinical correlates.

In the ensuing decades, structural MRI research benefited from increases in scanner field strength from 1.5 T to 3 T and subsequent increased resolution down to 1 mm. In addition to measuring volume, the emerging field of computational anatomy (Box 6.1) also allowed for automated segmentation and quantification of structural shape and cortical thickness. Studies employing these new technologies confirmed earlier observed patterns of distributed gray matter loss in the frontal, temporal, limbic, and striatal regions [27, 36–44]. Visualization of these overall patterns became possible with the development of meta-analysis tools such as ALE (Activation Likelihood Estimation, http://www.brainmap.org/ale/) and SDM (Signed Differential Mapping, http://www.sdmproject.com/software/). Figure 6.1 shows visualization of the spatial patterns of gray matter loss in schizophrenia across prefrontal/dorsolateral prefrontal cortex, temporal/medial temporal lobe, anterior cingulate cortex, and thalamus, based on meta-analysis using these tools [44].

Box 6.1 Computational Anatomy

Although manual methods are still considered the gold standard for structural delineation today, they are unsatisfactory for a number of reasons:

- *Manual tracing* is time-consuming and quite often suffers from rater bias, drift, and attrition.
- Anatomic abnormalities appear to be very subtle in schizophrenia, often not exceeding 5 % in ROI volume reductions.
- *Structural size* is not precise or specific enough to provide biological insights to disease pathophysiology.

To overcome these limitations, computational anatomy was developed by several labs in the late 1990s to provide automatic or semiautomatic, computerized approaches for the understanding of biological shape variation, cortical thickness, and cortical folding patterns.

The origin of computational anatomy can be traced back to 1917 when Sir D'Arcy Wentworth Thompson emphasized the roles of physical laws and mechanics as determinants of the form and structure of living organisms [111]. Grenander and Miller coined the term computational anatomy (and later computational functional anatomy for brain function) which focuses on the quantitative analysis of shape variability in biological structures [112, 113]. Its basic principles are: building atlases and computing large-deformation maps to represent biological shape space; computing empirical probability laws on that space to represent biological shape average and variation; and making inferences based on the probability laws for hypothesis testing and classification. An early review of several of the computational anatomy methods is found in

Box 6.1 (continued)

Ashburner et al. and a more complete collection of these methods in a NeuroImage Special Issue on Mathematics in Brain Imaging [114, 115].

The field of computational anatomy has flourished with an abundance of methods including deformation-based morphometry (DBM), voxel-based morphometry (VBM), tensor-based morphometry (TBM), surface-based morphometry (SBM), and pattern-based morphometry (PBM). For example, a 2003 review presented three nonlinear deformation methods and a 2009 review listed and compared 14 [114, 116]. Freely available software packages that utilize these methods are now easily accessible and user-friendly. These packages, including FreeSurfer (http://surfer.nmr.mgh.harvard.edu/), Slicer (http://www.slicer.org), Brain Voyager (http://www.brainvoyager.com/), BRAINS (http://www.nitrc.org/projects/brains), and SPM (http://www.fil.ion.ucl.ac.uk/spm/), offer automatic registration, ROI delineation, cortical measurements, white matter fiber tract tracing, and statistical hypothesis testing.

These methods and software tools represent more than 15 years of research combining mathematics, engineering, and computer vision. They provide unprecedented precision and repeatability, thereby alleviating rater bias and drift. Additionally, they are increasingly automated and are able to evaluate ever larger datasets to allow for the detection of subtle biological differences. They employ sophisticated state-of-the art algorithms to identify shape or localized changes in all areas of the brain and provide biologically meaningful information about the underlying disease process [117].

The next advance in neuroimaging research in schizophrenia included functional imaging and diffusion imaging techniques, which allowed for the quantification of brain activities in these regions and the integrity of the white matter that connects them. The first set of (fMRI) studies in individuals with schizophrenia reported elevated responses to photic stimulation in the primary visual cortex; reduced activation in the sensorimotor cortex and supplementary motor area during a finger-to-thumb task; and reduced frontal activation and increased temporal activation during a word fluency task [45–49]. The first set of diffusion tensor imaging (DTI) studies reported reduced white matter anisotropy extending from frontal to occipital regions, along with a lack of normal asymmetry in the uncinate fasciculus which connects the frontal and temporal regions [27, 50–53].

Since then, a wealth of fMRI studies revealed patterns of widespread abnormal task-induced activity that are similar to patterns of gray matter abnormalities. Most consistent are decreases in the dorsolateral prefrontal cortex during working memory and executive functioning tasks, followed by altered activity in the lateral and medial temporal lobe during retrieval tasks, and in the cingulate gyrus during cued verbal recall and Stroop tasks [54]. However, some studies have also shown



Fig. 6.1 Distributed patterns of gray and white matter abnormalities in schizophrenia from metaanalyses (Reproduced with permission from Bora et al. [44]). Meta-analysis of voxel-based morphometry studies indicates reduced gray matter density in bilateral medial frontal and temporal regions, dorsal and rostral anterior cingulate cortex, insula, and thalamus; left lateral prefrontal areas, superior frontal gyrus, and orbitofrontal and fusiform regions. Meta-analysis of DTI studies indicates reduced white matter volume in the anterior limb of the internal capsule (ALIC), right inferior frontal occipital fasciculi, and the inferior longitudinal fasciculi (ILF), and increased white matter volume in the left superior longitudinal fasciculi (SLF). Meta-analysis of DTI studies also indicates reduced fractional anisotropy (FA) in the bilateral genu of the corpus callosum, anterior cingulate cortex/medial frontal white matter, right ALIC, right external capsule/corona radiate, left temporal white matter and retrolenticular internal capsule, external capsule including fibers from the left inferior frontal occipital fasciculi, ILF and fornix/stria terminalis, and right temporal WM including fibers from right inferior frontal occipital fasciculi and ILF. *Red*: gray matter, *Blue*: DTI, and *Green*: WM volume (Color figure online)

increased activities in these regions (see Table 6.1 for a more detailed description of increased and decreased activities found in different studies). Similarly, DTI studies revealed consistently decreased white matter integrity in these regions, as well as in the regions that connect them, including frontotemporal, fronto-occipital, temporo-occipital, thalamocortical, and interhemispheric connections [27]. However,

-		fMDI	DTI	
		Maagura	Maagura	
DOI	Destaura	Channes of stalling	Channen and attalling	Nutri
	Review	Change, no. of studies	Change, no. of studies	Notes
Frontal/pr	efrontal	PP.		
PFC	[45, 57, 139,	EF	MD	↑MD
	140]	↓, 1 /	Τ, 2	
		T, /		
		↓ 1, 15 EME_EMD		
		EME, EMR		
MDEC	[141]	↓, IL EL EE NE NI		
WII I'C	[1+1]	\perp 1 2R		
		• 1L 1R		
DLPFC	[45 55 141]	NB		
DEFIC	[10, 00, 111]	4		
		1, 2		
		Language, VB, WM, RI, CP		
		↓, 7L, 4R		
		1, 2L, 1R		
VLPFC	[141]	WM, CP		
		↓, 1L, 2R		
		↑, 1L, 1R		
OFG				
White	[57, 142]		FA, MD	
matter			↓, 11	
			$\leftrightarrow, 1$	
Temporal/	MTL			
AG	[141]	EL, FN, FF, NI		
		↓, 6L, 2R		
		↑, 1L		
HP/PHG	[141]	EL, FN, FF, NI		
		↓, 3L, 2R		
		1, 2R		
HP	[57, 140]	EME, EMR	FA	
		↓, 4	↓, 1	
PHG	[57]		FA, MD ↓, 1	↑ MD
			↑, 1	
ERC				
STG	[45, 57]	Language, VB	FA	
		SA, SL, SP, NV	↓, 1	
		↓, 4L, 1K		
		î, IL, 2R		

 Table 6.1 Distributed and heterogeneous pattern of widespread abnormal structure and function in schizophrenia

		fMRI:	DTI:	
		Measure	Measure	
ROI	Review	Change, no. of studies	Change, no. of studies	Notes
MTG	[57]	SA, SL, SP, NV	FA	
		↓, 3L, 2R	↓, 1	
		↑, 2R		
White	[57, 142]		FA	
matter			↓, 5	
			\leftrightarrow , 1	
Parietal				
IPL	[57]		FA	
			↓, 1	
White	[142]		FA	
matter			↓, 2	
			$\leftrightarrow, 1$	
Occipital				
White	[57, 142]		FA	
matter			↓, 3	
			\leftrightarrow , 1	
Limbic				
CG	[55, 57]	NB	MD	↑ MD
		↑, 1	↑, 1	
ACC	[57, 143]	CPT	FA	
		↓, 1	↓, 1	
Deep nucle	ei			
TH	[57, 140]	EMR	MD	↑ MD
		↓, 2	↑, 1	
VSTR	[141]	RP, PC		
		↓, 2L		
		↑, 1L, 1R		
Connection	ns			
CC	[57, 142,		FA, MD	
	144]		↓, 7	
			↔, 3	
FX	[57]		FA, MD	↑ MD
			↓, 1	
	100		↑, 1 TA	
ALIC	[57, 142]		FA	
			↓, 3	
			\leftrightarrow , 1	
DLIC	[57 140]		I, I EA	
r LIC	[37, 142]			
			*, ~ ~ 1	
			<, ⊥ ↑ 1	
			1, 1	

Table 6.1 (continued)

		fMRI:	DTI:	
		Measure	Measure	
ROI	Review	Change, no. of studies	Change, no. of studies	Notes
Frontal-te	mporal			
AF	[27]		FA ↓, 3	Reduced left- lateralized
СВ	[27, 57, 142]		FA, MD ↓, 10 ↔, 3 ↑, 1	↑, MD
UF	[27, 57, 142]		FA, MD ↓, 4	FA decrease in left>right asymmetry
			↔, 1 ↑, 1	↑, MD
Frontal-od	ccipital–pariet	al–temporal		
SLF	[57]		FA, MD ↓, 5 ↔, 1 ↑, 1	↑, MD
Fronto-oco	cipital			
IOF	[57]		FA, MD \downarrow , 2 \leftrightarrow , 1	↑, MD
Town and	navistal		1, 1	
White matter	[57, 142]		FA, MD ↓. 2	
Temporal_	-occinital		,	
ILF	[27, 57, 144]		FA ↓, 4	
Parietal–o	ccipital			
White matter	[142]		FA, MD ↓, 1	

 Table 6.1 (continued)

Structurally, reduced volume, altered asymmetry, and abnormal shape of gray matter are predominantly found in the medial and lateral temporal lobe, followed by frontal regions. Functionally, task-induced abnormal activities are seen in prefrontal regions most often, followed by the medial and lateral temporal lobe. These structural and functional changes are accompanied by similar changes in white matter integrity, in these particular regions and in the regions that connect thems Symbols

↓: reduced when compared with controls

1: increased when compared with controls

 \downarrow \uparrow : both reduced and increased when compared with controls

 \leftrightarrow : no difference when compared with controls

N/A: not available/not reported



Fig. 6.2 Distributed patterns of fMRI activation abnormalities in schizophrenia from metaanalyses (Reproduced with permission from Glahn et al. [55]). Meta-analysis of N-Back studies indicates reduced activation in bilateral dorsolateral prefrontal cortex, rostral prefrontal cortex, and right ventrolateral/insular cortex. Brain regions showing greater patient activation include left frontal pole, right dorsomedial prefrontal cortex (BA9), and anterior cingulate

some studies have observed increased white matter integrity measures, particularly in interregional connections.

Table 6.1 provides a detailed summary of the patterns of abnormal structure and function in studies of schizophrenia. The table is arranged by brain regions commonly found to have abnormal gray matter structure in sMRI studies. Findings from recent reviews of fMRI and DTI studies in these regions are then summarized in separate columns. Findings of abnormal interregional anatomic connections are listed in the bottom portion of the table. Similar to sMRI studies, advanced metaanalysis tools now allow for the visualization of spatial patterns of functional and anatomic connection alterations (Fig. 6.2: reduced activation in bilateral dorsolateral prefrontal cortex, rostral prefrontal cortex, and right ventrolateral/insular cortex during N-Back working memory paradigm [55]; reduced activation in the right superior frontal gyrus, left and right inferior frontal gyri, right inferior parietal gyrus, right lingual gyrus, and right posterior cingulate gyrus during episodic memory encoding; and reduced activation in the left inferior frontal gyrus, left precentral and middle frontal gyri, right anterior cingulate gyrus, left middle temporal gyrus, right cuneus, bilateral thalamus, right posterior cingulate gyrus, and bilateral cerebellum during episodic memory retrieval [not shown] [56]; Figs. 6.1 and 6.3: reduced anisotropy and white matter volume in the anterior thalamic radiation, inferior longitudinal fasciculi, inferior frontal occipital fasciculi, cingulum and fornix, cingulate, corpus callosum, frontal, temporal, parietal, and occipital lobes; as well as intralobar and interlobar connections of the superior longitudinal fasciculus,



Fig. 6.3 Distributed patterns of DTI abnormalities in schizophrenia from meta-analyses (Reproduced with permission from White et al. [57]). Meta-analysis of DTI studies indicate reduced anisotropy and white matter volume in the anterior thalamic radiation, inferior longitudinal fasciculi, inferior frontal occipital fasciculi, cingulum and fornix, cingulate, corpus callosum, frontal, temporal, parietal, and occipital lobes, as well as intralobar and interlobar connections of the superior longitudinal fasciculus, fronto-occipital longitudinal fasciculi, uncinate fasciculi, frontal longitudinal fasciculus, and the arcuate fasciculus

fronto-occipital longitudinal fasciculi, uncinate fasciculi, frontal longitudinal fasciculus, and arcuate fasciculus [44, 57]).

Disorganization of Brain Networks

Individuals with schizophrenia exhibit a wide range of clinical symptoms such as delusions, hallucinations, loss of emotion, and disorganized thoughts. These symptoms are accompanied by profound difficulties in a broad spectrum of cognitive functioning including working memory, episodic memory, and attention to stimuli, as well as the ability to control these cognitive processes [58]. Despite demonstrable abnormalities, however, studies of individual brain regions are unable to answer questions related to the pathophysiology of schizophrenia. This is because the methods employed by these studies are limited to the traditional view that cognitive functions are subserved by individual or separate brain regions. For example, studies of the dorsolateral prefrontal cortex could not completely account for the cognitive deficits exhibited in individuals with schizophrenia [59]. Recently, this traditional view has given way to a new paradigm in which cognitive constructs are viewed as being subserved by complex, interconnected, large-scale brain networks that segregate and then reintegrate to work together [60-64]. In this light, schizophrenia may be a disorder of neural and cognitive integration and that multifocal abnormalities in the brain, although subtle, likely involve brain networks or systems rather than a simple collection of disparate brain regions [21, 65–68].

A brain network is a set of brain systems that together subserve a specific or a set of specific cognitive behaviors such as working memory, attention, and executive function. A brain system is a set of cytoarchitectonically or functionally distinct brain regions along with the connections between these regions. Components and their links are defined based on measurement methods. (A broader, graph-theoretical concept of global, large-scale brain networks is described later in this chapter.) Accordingly, the schizophrenia neuroimaging research field is moving in the direction of network, multimodal, and pattern analyses (Box 6.2).

Below we summarize neuroimaging evidence that brain regional structural and functional impairments, as well as disruptions in the structural and functional connections between these regions, occur in a coordinated, network-wide fashion (i.e., schizophrenia is a disorder of disruptions in neural circuitry) [20, 21, 65, 69–71].

Box 6.2 Emerging Technologies and Characterization of Brain Networks

The theoretical prediction that healthy mental functioning depends on the existence of interregional connections and cooperation was set forward by Norman Geschwind (1965) in his seminal paper "Disconnexion syndromes in animals and man. I" [19]. In vivo neuroimaging support for this prediction started to gather only after technology advanced to the point where this was possible. Nora Volkow (1988) and colleagues provided the first neuroimaging evidence in a study of 12 healthy volunteers using PET scans [65]. She discovered covariations in the amount of regional deoxy [¹¹C] glucose uptake in the frontal and occipital regions, thalamus and frontal regions, and thalamus and occipital regions, and she proposed meaningful interactions in the forms of fronto-occipital, thalamo-frontal, and thalamo-occipital brain organization. The concept of a "functional connectivity" was solidified 5 years later when Friston et al. coined this term in a ^{[15}O] PET study of six healthy male volunteers [118]. Functional connectivity was subsequently defined as "temporal correlation between two neurophysiological (functional) measurements made in different brain areas". In vivo neuroimaging investigation of functional connectivity made its biggest breakthrough with the invention of fMRI by Seiji Ogawa (1990) and the discovery of resting state connectivity by Bharat Biswal (1995), seven years after the initial PET study by Volkow et al. [12, 119].

While correlation-based characterization of functional connectivity requires a priori definition of pairs of brain ROIs, a recently adapted multivariate approach, independent component analysis (ICA), does not have such a requirement. It allows for the automatic discovery of functional connectivity in the entire brain. ICA was originally used in signal processing in 1995 to solve "cocktail party" problems separating individual voices from recordings of many people speaking at once [120]. It was first applied to neuroimaging on groups of subjects to identify temporally coherent intrinsic brain networks by Vince Calhoun in a series of studies [121–123]. ICA extensions were later developed for functional network connectivity (FNC) analysis and

Box 6.2 (continued)

structure–function joint analysis (jICA) [124, 125]. Using both correlational analysis and ICA methods, fMRI studies made great contributions to the discovery of the brain's many resting state networks (RSN) which are temporally coherent, are spatially distributed, and correspond to functional behaviors. These networks include the default mode network (DMN), somatomotor network, visual network, language network, dorsal attention network, ventral attention network, and fronto-parietal control network [78, 126, 127]. In Fig. 6.4 we show examples of these networks as obtained from an ICA analysis of resting state data [126].



Fig. 6.4 Representation of different ICA components from resting state fMRI analysis. Resting state networks projected onto the cerebral surface. (a) Default mode network. (b) Somatomotor network. (c) Visual network. (d) Language network. (e) Dorsal attention network. (f) Ventral attention network. (g) Fronto-parietal control network. Reproduced with permission from Lee et al. [126]

Disrupted Regional Networks

Relationships between functional networks in schizophrenia were first interrogated by Volkow et al. in a PET study [65]. Compared with controls, individuals with schizophrenia showed reductions in what was later termed functional connectivity between frontal and occipital cortices and between the thalamus and frontal, occipital cortices. Using PET and word generation paradigms, Friston later reported a disruption of prefronto-temporal functional connectivity in individuals with schizophrenia [72]. In individuals with schizophrenia, the normal correlations between prefrontal and middle temporal cortices were absent, and the normal correlations between prefrontal and middle temporal cortices were increased. These prefrontal–temporal connectivity disruptions were later replicated in functional connectivity studies using PET and fMRI [68, 71, 73–76]. Further, dorsolateral prefrontal cortex–superior temporal gyrus/parahippocampal connectivity was reduced while ventrolateral prefrontal cortex–superior temporal gyrus/parahippocampal connectivity was increased during verbal encoding [77].

The default mode network (DMN), consisting of the medial temporal lobe, medial prefrontal cortex, posterior cingulate cortex, precuneus, and medial, lateral, and inferior parietal cortices, is among the most studied networks in schizophrenia [78]. Individuals with schizophrenia show disruption in the localized spatial patterns of activation during task, i.e., reduced activation in the precuneus, both reduced and increased activation in different parts of the posterior cingulate, and the inability to properly suppress medial prefrontal cortex activity [79, 80]. They also show disrupted connectivity among the DMN regions while at rest, although some inconsistencies exist. For example, Zhou et al. and Whitfield-Gabrieli et al. found increased connectivity, while Bluhm et al. found the opposite [80–82]. Despite these discrepancies, which may be due to inherent disease heterogeneity and differences in patient cohorts and analysis methods, hyper-connectivity within the DMN regions is the most consistent finding [83, 84].

More recent fMRI studies of schizophrenia extend these findings to the interconnections between the rest of the brain regions. Connectivity between the DMN and the frontal, parietal, and fronto-parietal networks is increased, while DMN connectivity with dorsolateral prefrontal cortex and orbitofrontal gyrus networks is decreased [80, 85, 86]. Also, a decrease in functional connectivity exists between the frontal/ prefrontal areas (including the dorsolateral prefrontal cortex) and other networks; and between the temporal network and other networks [73, 80, 81, 85–90]. These disruptions are thought to weaken frontal–temporal connectivity [71, 73–76]. They also corroborate the notion that the frontal–temporal white matter connections are weakened (i.e., the finding of "hypofrontality"), as well as the shifting of frontal to nonfrontal hubs (described in the next section).

The literature is not without inconsistencies. For example, Jafri et al. and Sakoglu et al. both reported changes between the frontal–visual networks but in opposite directions [85, 86]. While the DMN–frontal connectivity was increased in one study, the DMN–dorsolateral prefrontal cortex connectivity was decreased in another [80, 86]. Table 6.2 provides a summary of findings on the disruption in between-network functional connectivity from these studies.

Connection	Study	Correlation/ anticorrelation (+/-)	Increase/ decrease (↑/.L)	Notes
Frontal to other areas				
Frontal_DMN	[86]	+	↑	
Frontal-visual	[86]	+	` ↑	
Frontal-visual	[85]	+	Ļ	Medial visual
Frontal-MTL	[85]	+	ţ	Task modulated
Frontal-other lobes	[89]	+	Ļ	Graph theory
Frontal-temporal	[71, 74–76]	+	Ļ	
Prefrontal/DLPFC to ot	her areas			
Prefrontal-temporal	[73]	+	Ļ	
Prefrontal-cingulate	[87]	+	Ļ	Task modulated, MSFG-ACC
Prefrontal-cerebellum	[73]	+	Ļ	
DLPFC-parietal	[81]	+	Ļ	Bilateral
DLPFC-PCC	[81]	-	Ļ	Right
DLPFC-PPC	[90]	+	Ļ	C C
DLPFC-DMN	[<mark>80</mark>]	-	Ļ	
DLPFC-other brain area	as[88]	+	Ļ	Absence of normal connectivity
DLPFC-OFG, insula	[81]	-	↑.	Left DLPFC
VLPFC-PPC	[<mark>90</mark>]	+	↑.	
Fronto-parietal to other	areas			
Fronto-parietal—other a	ireas			
Fronto-parietal—DMN	[86]	+	↑.	
Fronto-parietal-visual	[86]	+	1	
Fronto-parietal—OFG	[85]	+	↑	Right lateral fronto-parietal
Fronto-parietal—MTL	[85]	+	↑	Task modulated, right lateral fronto-parietal
Temporal to other areas				
Temporal-occipital	[73]	+	Ļ	
Temporal-cerebellum	[73]	+	Ļ	
Temporal-parietal	[86]	+	Ļ	
Temporal-parietal	[85]	+	Ļ	Medial temporal
Temporal-visual	[85]	+	Ļ	Medial temporal, medial visual
Medial temporal-anterio temporal	or[85]	+	Ļ	
Other connectivities				
OFG-DMN	[85]	+	Ļ	Task modulated, posterior DMN
Parietal–DMN	[85]	+	↑	Task modulated, posterior DMN
Parietal-visual	[85]	+	Ļ	Medial visual

 Table 6.2 Disrupted between-network functional connectivity in schizophrenia

Connection	Study	Correlation/ anticorrelation (+/-)	Increase/ decrease (\uparrow/\downarrow)	Notes
ACC-other brain areas	[145]	+	ţ	Normal connectivities absent
Cerebellum–other brain areas	[73]	+	1	

Table 6.2	(continued)
Tuble 0.2	(continueu)

We note: (1) reduction of connectivity is most acute in the pairings of frontal to other areas and temporal to other areas, and this corroborates the weakening of the frontal-temporal connection as mentioned previously in DTI studies; (2) relatively few studies reported changes other than the frontal and temporal networks, and this corroborates the "hypofrontality" mentioned earlier and the shifting of frontal to nonfrontal hubs below; (3) changes in inter-network connectivities involving the DMN specifically depend on what the other network is: increases with frontal, parietal, and fronto-parietal, and decreases with DLPFC and OFG; (4) some inconsistent findings are also reported. For example, Jafri et al. and Sakoglu et al. both reported changes between the frontal–visual networks but in opposite directions [85, 86]

Symbols

↓: reduced when compared with controls

1: increased when compared with controls

N/A: not available/not reported

Disrupted Large-Scale Global Networks

Overwhelming evidence exists that schizophrenia is associated with widespread disruptions in normal structural and functional coherence, notably in the prefrontal, temporal, parietal, and limbic regions. Additionally, connections between these regions, both structurally via white matter pathways and functionally via synchronized neural oscillations, are also disrupted. Recent developments in large-scale brain network analysis methods, particularly those based on graph theory approaches (Box 6.3), allow for the examination of these disruptions in a coordinated, global fashion.

Box 6.3 Development of Large-Scale Brain Network Methods

Adapted from the graph-theoretic framework, interregional brain networks were first characterized in cats and macaques where connections between brain regions could be anatomically verified [128–130]. Organizations of mammalian brain networks use a combination of high levels of local clustering (i.e., cliques) and overall short path lengths to link all nodes of the

Box 6.3 (continued)

network [91]. This type of network is said to have "small-world properties," a concept originally described in social networks [131]. A network with small-world properties balances local segregation (modular processing) and global integration (distributed processing) with high signal-propagation speed, computational efficiency, and information synchronizability [132]. Virtually all complex, real-world networks have small-world properties (e.g., social networks, road maps, gene and neuron networks).

The characterization of brain organization in humans greatly improved with recent developments in neuroimaging techniques, especially white matter tract tracing and fMRI connectivity methods which quantify structural and functional connections. The application of graph-theoretical approach demonstrates that human brain structural and functional networks also exhibit small-world properties [132]. This approach usually includes the following steps of analysis: network nodes are defined as anatomic ROIs, such as subcortical or cortical structures in MRI or electrode sensor locations in the electroencephalogram (EEG); associations between nodes are then defined by structural or functional connectivity (i.e., correlations in cortical thickness or volume, DTI tractography-derived connection probability, correlations between fMRI BOLD time series, or spectral coherence between sensors); an adjacency matrix or undirected graph is then generated by compiling all pairwise associations (e.g., correlation); and finally, network parameters of interest are calculated [60, 132, 133]. The most commonly used network parameters are the small-world parameters (i.e., clustering coefficient, characteristic path length, and degree of connectivity). Other parameters include hierarchy, assortativity, connection distance, centrality, and identification of network hubs. A formal definition of "small-worldness" is that a network's level of clustering should be greater than that of a comparable random network, and the clustering coefficient to characteristic path length ratio is also greater than that of a comparable random network, where clustering coefficient is a measure of how much neighbors of each network node are also neighbors of each other and characteristic path length is the average length of the shortest path between any two nodes [91, 128, 134]. In Figs. 6.5, 6.6, and 6.7 we show illustrations of network analysis pipeline and outcomes using graph theory approaches [33, 91].

The first graph-theoretic application of network analysis in schizophrenia was performed by Micheloyannis et al. using EEG [135]. The first structural network analysis was performed by Bassett et al. using regional volumes measured in structural MRI, while Liu et al. were the first group using fMRI [89, 91]. Earlier EEG and magnetoencephalography (MEG) studies showed discrepant findings regarding small-worldness (decreased; preserved) and may be related



Fig. 6.5 Representations of graph theory network analysis pipeline and outcomes: Illustration of the main steps involved in graph analysis of human neuroimaging data (Reproduced with permission from Fornito et al. [33]). Top row: imaging modalities. Second row: imaging data are parcellated into distinct network nodes-e.g., anatomic parcellation (left), random parcellations of 500 (middle left) and 2,000 (middle right) regions, and functionally defined parcellation (right). Third row: interconnecting edges are defined—tractography for DTI (left), cross correlations for T1 (middle); time course dependencies for EPI (right). Fourth row: connectivity is represented as a continuously weighted matrix-time course correlations for resting state fMRI (left), thresholded to create either a weighted (*middle*) or binary (*right*) adjacency matrix. Bottom row: the adjacency matrix is used to construct a brain graph—e.g., a weighted, undirected graph where nodes are represented as purple circles and their interconnecting edges as green lines sized in proportion to edge weight (left), connectivity measures (middle): edge strength, regional strength, etc., and network topology (right) measure of connectivity: nodal degree, clustering coefficient, characteristic path length. Note that the nodes are grouped into three distinct modules, defined by the large colored circles. Nodes within modules have higher connectivity with each other than with the rest of the network (Color figure online)



Fig. 6.6 Representations of graph theory network analysis pipeline and outcomes: Illustration of the relationship between thresholding and connectivity weight (Reproduced with permission from Fornito et al. [33]). (a) A representative functional connectivity matrix from a single patient (*left*) and control (*right*). (b) The distribution of connectivity weights is shifted towards lower values in the patient (*red*) relative to the control (*blue*); the area shaded in red highlights the excess number of low weighted values in the patient's connectivity matrix. (c) Applying weightbased threshold τ (*solid lines*) to the patient and control results in different connection densities; applying connection density-based threshold κ (*broken lines*) results in a different minimum correlation weight threshold. (d) Correlation matrices after κ -matched thresholding have different minimum and mean weights (Color figure online)

to different thresholds across studies [97, 135–137]. In the quantification of network links and path lengths, various threshold values are used and they affect the group differences in overall path lengths more drastically than differences in clustering coefficients [100]. Since individuals with schizophrenia exhibit increased interdependence on long-distance nodes, both from functional connectivity (above) and from electrophysiology studies, it is reasonable to deduce that larger, but looser network clusters may have survived thresholding, contributing to decreased overall path length [138]. Since the early days of EEG, recent neuroimaging modalities of choice for investigating brain network organizations in schizophrenia shifted to DTI or fMRI, where connections (i.e., edges in a graph) can be explicitly quantified using measures of fiber tracts or functional connectivity throughout the brain, as opposed to being limited to the cerebral surface as measured in an EEG. In a



Fig. 6.7 Representations of graph theory network analysis pipeline and outcomes. Graphical visualization of multimodal network hierarchy (Reproduced with permission from Bassett et al. [91]). (a) Normal control subjects; (b) Individuals with schizophrenia. Nodes are ordered according to their degree (*y*-axis). Size of nodes indicates greater than (large) or less than (small) average clustering. Color of the nodes indicates lobe location: frontal (*blue*), temporal (*green*), parietal (*black*), or occipital (*red*). Lettering indicates approximate Brodmann area, and the overbar denotes left-sided regions. Note that highly clustered nodes are concentrated at the bottom of the normal hierarchy, which is dominated by highly connected nodes (many of them frontal) with low clustering; conversely, in individuals with schizophrenia, highly clustered nodes are more evenly distributed in terms of their degree, and frontal hubs are less prominent (Color figure online)

thorough review of the graph-theoretical analysis of disturbances in the structural and functional networks in schizophrenia, Fornito et al. provide a summary of MRI, DTI, and fMRI studies up to 2011 [33]. MRI studies based on sMRI, DTI, or fMRI and network analysis methods show that despite being "a putative neurodevelopmental disorder with profound effects on complex brain functions," the overall global network architecture in schizophrenia appears to be preserved. This suggests that key aspects of brain organization are highly conserved in schizophrenia [91]. Interestingly, however, virtually all of the local measures of this global network architecture differ from healthy controls. Reduced hierarchy, increased regional path length, reduced local efficiency, and loss of hub properties are all reported in individuals with schizophrenia [91–99]. Most notably, these patients show diminished hub properties in the frontal lobe and a shift of hubs from frontal to nonfrontal areas [91]. Table 6.3 summarizes findings from studies on network analysis in schizophrenia, including fMRI, electroencephalogram (EEG), and magnetoencephalography (MEG) studies. Based on these observations, the theory that schizophrenia is a "fragmentation of brain pathologies" was put forth to understand its pathophysiology as a breakdown in local network organization rather than a loss of global coherence [100].

Multimodal and Pattern-Analytical Neuroimaging Biomarkers

The observation that schizophrenia is a syndrome of dysfunctional neural networks created a framework for understanding how brain structural attributes relate to functional attributes, and how brain dynamics relate to behavioral manifestations of the syndrome. Within this framework, analytic approaches that combine different neuroimaging modalities and examine patterns of disease-related alternations in these modalities are providing a deeper understanding of the neural mechanisms of, and improved biomarkers for disease pathophysiology [101].

Multimodal analysis primarily involves two approaches:

- Measures collected in ROIs or finer-grained image elements (e.g., voxels or surface vertices) from different imaging modalities (e.g., sMRI and fMRI) are used to find correlations between the modalities.
- Data are fused first before statistical analysis.

Recent studies demonstrate a significant overlap between spatial distribution patterns of individual modalities, and a positive association between stronger alterations in one imaging modality and stronger disturbances in another. For example, increased disturbance in white matter connectivity of the superior temporal cortex is associated with thinning of the posterior cingulate cortex. This reflects a complex disruption of gray and white matter integrity within a cingulo-temporal network in schizophrenia [102]. Reduced functional activity in prefrontal and parietal cortices is associated with thinning of the same regions during a problem solving Tower of London task [103]. Reduced functional activity in the dorsal anterior cingulate cortex is associated with thinning of prefrontal-temporal cortices during a working

Studies in	Small-world properties				
chronological order	Modality	Small- worldness	Clustering coefficient	Characteristic path length	Other findings
[135]	EEG Resting state and working memory	Decreased	Ļ	↑ ↑	Differences were observed in different frequency bands
[136]	EEG Working memory	Decreased trend	Ļ	↑	• Differences were observed in different frequency bands
[89]	fMRI Resting state	Decreased	Ļ	↑	• Decreased regionally and globally
[91]	sMRI	Preserved globally but regionally disrupted	↔	↔	 Multimodal cortex shows reduced hierarchy, increased connection distance Multimodal cortex shows shifting of hubs from frontal to nonfrontal areas including premotor, prefrontal, orbitofrontal, inferior temporal, medial temporal, cingulate, and insular cortex
[97]	EEG Resting state	Preserved subtle trend towards decrease	Ļ	Ļ	Lowered centrality of major hubs
[93]	fMRI	Preserved	N/A	N/A	 Reduced local efficiency Reduced gray matter volume in hub regions Loss of hub properties in medial superior frontal gyrus and bilateral dorsal cingulate gyrus
	Episodic memory				Different hubs than controls

 Table 6.3
 Brain network organization findings in schizophrenia

Studies in	in Small-world properties				
chronological		Small-	Clustering	Characteristic	-
order	Modality	worldness	coefficient	path length	Other findings
[92]	DTI	Preserved	↔	÷	 Increased path length in olfactory, medial, and superior frontal regions, anterior cingulate, medial temporal pole, and superior occipital regions Indicating a less strongly integrated structural network and reduced central role of key frontal hubs
[94]	fMRI Resting state	Decreased	Ţ	ţ	 Increased characteristic path length only at lower cost points Global efficiency showed lower value only at lower cost points Local efficiencies were increased only at the higher cost points
[98]	fMRI Resting state	N/A	Ţ	N/A	More and smaller modulesDifferent hubs than controls
[95]	DTI	Preserved	\leftrightarrow	\leftrightarrow	Reduced global network efficiency, probably driven by reduced nodal degree in frontal, parietal, and occipital regions
[96]	DTI	N/A	N/A	N/A	 Increased global efficiency Reduced regional efficiency of the prefrontal cortex, the paralimbic/limbic regions, and subcortical structures

 Table 6.3 (continued)

Studies in		Small-world p	properties		
chronological order	Modality	Small- worldness	Clustering coefficient	Characteristic path length	Other findings
[146]	fMRI Resting state Auditory oddball	Preserved	Ļ	Ļ	
[99]	fMRI		Ļ	Ļ	 Reduced local efficiency Increased global efficiency
	Working memory				Processing in schizophrenia may be more concurrent across different brain regions
[137]	MEG Resting state	Preserved	\leftrightarrow	\leftrightarrow	U
Symbols					

Table 6.3 (continued)

↓: reduced when compared with controls

1: increased when compared with controls

 \leftrightarrow : no difference when compared with controls

N/A: not available/not reported

memory task [104]. Further, structural changes in gray matter regions are associated with functional changes from a different location, serving as a morphological substrate for changes in functional connectivity [105-107]. These results directly support the notion that functional consequences in schizophrenia are closely related to, or may even constitute a consequence of, aberrant gray matter structure. Multimodal approaches provide useful information on whether and how disparate pathophysiological features influence each other and how schizophrenia alters these relationships.

Pattern analysis primarily involves methods adopted from the machine learning community applied to neuroimaging data. The basic idea is that the relationship between multiple parameters from multiple neuroimaging modalities can be combined to identify key differences between groups which in turn aid individual classification. In this pattern-analytical approach, data are constructed into training and validation sets in various schemes of classifier algorithms, including support vector machines (SVM), Bayesian networks, and artificial neural networks. Compared with traditional general linear model (GLM)-based approaches for detecting group differences, high-dimensional machine learning techniques show promise for improving the characterization of individual-based classification and finding subtle but significance features undetectable by traditional approaches. A review of recent machine learning studies in schizophrenia revealed that classification accuracy ranging from 81 to 92 % could be achieved. These are higher than rates usually achieved using GLM-based methods [108].

The amount of data in these types of analysis is immense. For example, structural or functional MR data will result in parameters which can number up to hundreds of thousands of voxels. Multiplied by the number of different statistical data reduction and analysis approaches, the complexity makes the outcomes difficult to interpret, especially from one study to another. Therefore, while showing much promise, replication is needed to validate the findings from multimodal and pattern analysis studies.

Critical Questions

While advances in neuroimaging data collection and analysis greatly improve our understanding of the pathophysiology of schizophrenia at a group level, they do not yet translate into reliable neuroimaging biomarkers at an individual level useful to the clinician. New developments of neuroimaging technology will continue to push the boundary on how we understand in vivo brain structure, function, and the interactions of neuronal cells and complex neural networks in both time and space. Development of new technologies, including the complete description of the structural connections between elements of the brain—the Human Connectome Project [109]; the complete understanding of activities of the brain—the NIH BRAIN (Brain Research through Advancing Innovative Neurotechnologies, http://www.nih.gov/science/brain/) initiative; and the mapping of genetic vulnerabilities onto brain structure and function, may help us address the following critical questions:

- Can we achieve individualized diagnosis and treatment monitoring?
- Can we disentangle *effects of disease from effects of drug treatment and substance* use, and predict functional outcome?
- Can we reconcile discrepancies and characterize psychosis spectra which include schizophrenia and bipolar disorder?

For now, the field remains unsettled with regard to clinically relevant issues such as prodromal prediction of emerging symptoms, diagnosis of disease spectrum, and clinical course and specificity regarding overlapping symptoms with other related neuropsychiatric disorders. Despite its unproven clinical utility, neuroimaging as a tool for predicting disease course, treatment response, and identifying risk factors shows much promise [110]. It has served, and will continue to serve, as a validation tool for linking disease in the brain with clinical symptomatology and cognitive deficits. Thus, neuroimaging may soon become a biomarker for diagnosis, monitoring progression, and predicting treatment response in schizophrenia.

List of Key Facts

• Schizophrenia is associated with subtle but widespread abnormalities of brain structure, function, and connectivity.

- These abnormalities are part of dysfunctional large-scale brain networks which together give rise to the cognitive deficits observed in individuals with schizophrenia.
- For now, there remains much we do not understand about the neural mechanisms of schizophrenia.

Acknowledgements The authors would like to acknowledge the following NIH grants: 1R01 MH084803, 1 U01 MH097435, P50 MH071616, and R01 MH056584. The authors would like to thank Eva C. Alden, Katherine D. Blizinsky, and Daniel B. Stern for assistance on literature search.

References

- 1. Kraepelin E. Dementia Praecox. New York: Churchill Livingstone, Inc.; 1919.
- 2. Bleuler E. Dementia praecox or the group of schizophrenias. New York: International Universities Press; 1911.
- Johnstone EC, Crow TJ, Frith CD, Husband J, Kreel L. Cerebral ventricular size and cognitive impairment in chronic schizophrenia. Lancet. 1976;ii:924–6.
- 4. Buckley PF. Neuroimaging of schizophrenia: structural abnormalities and pathophysiological implications. Neuropsychiatr Dis Treat. 2005;1(3):193–204.
- 5. Moseley ME, Cohen Y, Kucharczyk J, et al. Diffusion-weighted MR imaging of anisotropic water diffusion in cat central nervous system. Radiology. 1990;176(2):439–45.
- Moseley ME, Kucharczyk J, Mintorovitch J, et al. Diffusion-weighted MR imaging of acute stroke: correlation with T2-weighted and magnetic susceptibility-enhanced MR imaging in cats. AJNR Am J Neuroradiol. 1990;11(3):423–9.
- Phelps ME, Hoffman EJ, Mullani NA, Ter-Pogossian MM. Application of annihilation coincidence detection to transaxial reconstruction tomography. J Nucl Med. 1975;16(3):210–24.
- Phelps ME, Huang SC, Hoffman EJ, Selin C, Sokoloff L, Kuhl DE. Tomographic measurement of local cerebral glucose metabolic rate in humans with (F-18)2-fluoro-2-deoxy-D-glucose: validation of method. Ann Neurol. 1979;6(5):371–88.
- Lauterbur PC. Image formation by induced local interactions: examples employing nuclear magnetic resonance. Nature. 1973;242(5394):190–1.
- Hawkes RC, Holland GN, Moore WS, Worthington BS. Nuclear magnetic resonance (NMR) tomography of the brain: a preliminary clinical assessment with demonstration of pathology. J Comput Assist Tomogr. 1980;4(5):577–86.
- 11. Holland GN, Hawkes RC, Moore WS. Nuclear magnetic resonance (NMR) tomography of the brain: coronal and sagittal sections. J Comput Assist Tomogr. 1980;4(4):429–33.
- 12. Ogawa S, Lee TM, Kay AR, Tank DW. Brain magnetic resonance imaging with contrast dependent on blood oxygenation. Proc Natl Acad Sci U S A. 1990;87(24):9868–72.
- Le Bihan D, Breton E, Lallemand D, Grenier P, Cabanis E, Laval-Jeantet M. MR imaging of intravoxel incoherent motions: application to diffusion and perfusion in neurologic disorders. Radiology. 1986;161(2):401–7.
- Wesbey GE, Moseley ME, Ehman RL. Translational molecular self-diffusion in magnetic resonance imaging. II. Measurement of the self-diffusion coefficient. Invest Radiol. 1984;19(6):491–8.
- Wesbey GE, Moseley ME, Ehman RL. Translational molecular self-diffusion in magnetic resonance imaging. I. Effects on observed spin-spin relaxation. Invest Radiol. 1984;19(6): 484–90.
- Filler A. The history, development and impact of computed imaging in neurological diagnosis and neurosurgery: CT, MRI, and DTI. Nature Precedings 2009.

- Ai T, Morelli JN, Hu X, et al. A historical overview of magnetic resonance imaging, focusing on technological innovations. Invest Radiol. 2012;47(12):725–41.
- Smith RC, Calderon M, Ravichandran GK, et al. Nuclear magnetic resonance in schizophrenia: a preliminary study. Psychiatry Res. 1984;12(2):137–47.
- 19. Geschwind N. Disconnexion syndromes in animals and man. I. Brain. 1965;88(2):237-94.
- 20. Friston KJ. The disconnection hypothesis. Schizophr Res. 1998;30(2):115-25.
- Friston KJ, Frith CD. Schizophrenia: a disconnection syndrome? Clin Neurosci. 1995;3(2):89–97.
- Cronenwett WJ, Csernansky J. Thalamic pathology in schizophrenia. In: Swerdlow NR, editor. Behavioral neurobiology of schizophrenia and its treatment. Berlin: Springer; 2010. p. 509–28.
- Pettersson-Yeo W, Allen P, Benetti S, McGuire P, Mechelli A. Dysconnectivity in schizophrenia: where are we now? Neurosci Biobehav Rev. 2011;35(5):1110–24.
- Pearlson GD, Marsh L. Structural brain imaging in schizophrenia: a selective review. Biol Psychiatry. 1999;46(5):627–49.
- McCarley RW, Wible CG, Frumin M, et al. MRI anatomy of schizophrenia. Biol Psychiatry. 1999;45(9):1099–119.
- Shenton ME, Dickey CC, Frumin M, McCarley RW. A review of MRI findings in schizophrenia. Schizophr Res. 2001;49(1–2):1–52.
- Shenton ME, Whitford TJ, Kubicki M. Structural neuroimaging in schizophrenia: from methods to insights to treatments. Dialogues Clin Neurosci. 2010;12(3):317–32.
- Karlsgodt KH, Jacobson SC, Seal M, Fusar-Poli P. The relationship of developmental changes in white matter to the onset of psychosis. Curr Pharm Des. 2012;18(4):422–33.
- 29. Thomason ME, Thompson PM. Diffusion imaging, white matter, and psychopathology. Annu Rev Clin Psychol. 2011;7:63–85.
- Waddington JL. Neuroimaging and other neurobiological indices in schizophrenia: relationship to measurement of functional outcome. Br J Psychiatry. 2007;50(Suppl):s52–7.
- McGuire P, Howes OD, Stone J, Fusar-Poli P. Functional neuroimaging in schizophrenia: diagnosis and drug discovery. Trends Pharmacol Sci. 2008;29(2):91–8.
- Mueller S, Keeser D, Reiser MF, Teipel S, Meindl T. Functional and structural MR imaging in neuropsychiatric disorders, part 2: application in schizophrenia and autism. AJNR Am J Neuroradiol. 2012;33(11):2033–7.
- Fornito A, Zalesky A, Pantelis C, Bullmore ET. Schizophrenia, neuroimaging and connectomics. Neuroimage. 2012;62(4):2296–314.
- 34. Andreasen N, Nasrallah HA, Dunn V, et al. Structural abnormalities in the frontal system in schizophrenia. A magnetic resonance imaging study. Arch Gen Psychiatry. 1986;43(2): 136–44.
- Lawrie SM, Abukmeil SS. Brain abnormality in schizophrenia. A systematic and quantitative review of volumetric magnetic resonance imaging studies. Br J Psychiatry. 1998;172:110–20.
- Schmitt A, Hasan A, Gruber O, Falkai P. Schizophrenia as a disorder of disconnectivity. Eur Arch Psychiatry Clin Neurosci. 2011;261 Suppl 2:S150–4.
- Pearlson GD, Calhoun V. Structural and functional magnetic resonance imaging in psychiatric disorders. Can J Psychiatry. 2007;52(3):158–66.
- 38. Pantelis C, Yucel M, Bora E, et al. Neurobiological markers of illness onset in psychosis and schizophrenia: the search for a moving target. Neuropsychol Rev. 2009;19(3):385–98.
- Levitt JJ, Bobrow L, Lucia D, Srinivasan P. A selective review of volumetric and morphometric imaging in schizophrenia. Curr Top Behav Neurosci. 2010;4:243–81.
- Palaniyappan L, Balain V, Liddle PF. The neuroanatomy of psychotic diathesis: a metaanalytic review. J Psychiatr Res. 2012;46(10):1249–56.
- Keshavan MS, Prasad KM, Pearlson G. Are brain structural abnormalities useful as endophenotypes in schizophrenia? Int Rev Psychiatry. 2007;19(4):397–406.
- Honea R, Crow TJ, Passingham D, Mackay CE. Regional deficits in brain volume in schizophrenia: a meta-analysis of voxel-based morphometry studies. Am J Psychiatry. 2005;162(12): 2233–45.

- Fornito A, Yucel M, Patti J, Wood SJ, Pantelis C. Mapping grey matter reductions in schizophrenia: an anatomical likelihood estimation analysis of voxel-based morphometry studies. Schizophr Res. 2009;108(1–3):104–13.
- Bora E, Fornito A, Radua J, et al. Neuroanatomical abnormalities in schizophrenia: a multimodal voxelwise meta-analysis and meta-regression analysis. Schizophr Res. 2011;127(1–3):46–57.
- Kindermann SS, Karimi A, Symonds L, Brown GG, Jeste DV. Review of functional magnetic resonance imaging in schizophrenia. Schizophr Res. 1997;27(2–3):143–56.
- 46. Gur RE. Functional brain-imaging studies in schizophrenia. In: Bloom FE, Kupfer DJ, editors. Psychopharmacology: the fourth generation of progress. New York: Raven; 1995.
- Renshaw PF, Yurgelun-Todd DA, Cohen BM. Greater hemodynamic response to photic stimulation in schizophrenic patients: an echo planar MRI study. Am J Psychiatry. 1994; 151(10):1493–5.
- Wenz F, Schad LR, Knopp MV, et al. Functional magnetic resonance imaging at 1.5 T: activation pattern in schizophrenic patients receiving neuroleptic medication. Magn Reson Imaging. 1994;12(7):975–82.
- Yurgelun-Todd DA, Waternaux CM, Cohen BM, Gruber SA, English CD, Renshaw PF. Functional magnetic resonance imaging of schizophrenic patients and comparison subjects during word production. Am J Psychiatry. 1996;153(2):200–5.
- Buchsbaum MS, Tang CY, Peled S, et al. MRI white matter diffusion anisotropy and PET metabolic rate in schizophrenia. Neuroreport. 1998;9(3):425–30.
- 51. Mccarley RW. Structural magnetic resonance imaging studies in schizophrenia. In: Davis KL, Charney D, Coyle JT, Nemeroff C, editors. Neuropsychopharmacology: the fifth generation of progress. Philadelphia, PA: Lippincott Williams & Wilkins; 2002.
- Lim KO, Hedehus M, Moseley M, de Crespigny A, Sullivan EV, Pfefferbaum A. Compromised white matter tract integrity in schizophrenia inferred from diffusion tensor imaging. Arch Gen Psychiatry. 1999;56(4):367–74.
- Kubicki M, Westin CF, Maier SE, et al. Uncinate fasciculus findings in schizophrenia: a magnetic resonance diffusion tensor imaging study. Am J Psychiatry. 2002;159(5):813–20.
- 54. Berman KF. Functional neuroimaging in schizophrenia. In: Davis KL, Charney D, Coyle JT, Nemeroff C, editors. Neuropsychopharmacology: the fifth generation of progress. Philadelphia, PA: Lippincott Williams & Wilkins; 2002.
- 55. Glahn DC, Ragland JD, Abramoff A, et al. Beyond hypofrontality: a quantitative metaanalysis of functional neuroimaging studies of working memory in schizophrenia. Hum Brain Mapp. 2005;25(1):60–9.
- Ragland JD, Laird AR, Ranganath C, Blumenfeld RS, Gonzales SM, Glahn DC. Prefrontal activation deficits during episodic memory in schizophrenia. Am J Psychiatry. 2009; 166(8):863–74.
- White T, Nelson M, Lim KO. Diffusion tensor imaging in psychiatric disorders. Top Magn Reson Imaging. 2008;19(2):97–109.
- Heinrichs RW, Zakzanis KK. Neurocognitive deficit in schizophrenia: a quantitative review of the evidence. Neuropsychology. 1998;12(3):426–45.
- Weinberger DR, Egan MF, Bertolino A, et al. Prefrontal neurons and the genetics of schizophrenia. Biol Psychiatry. 2001;50(11):825–44.
- Bressler SL, Menon V. Large-scale brain networks in cognition: emerging methods and principles. Trends Cogn Sci. 2010;14(6):277–90.
- Mesulam MM. Large-scale neurocognitive networks and distributed processing for attention, language, and memory. Ann Neurol. 1990;28(5):597–613.
- Goldman-Rakic PS. Topography of cognition: parallel distributed networks in primate association cortex. Annu Rev Neurosci. 1988;11:137–56.
- Sporns O, Chialvo DR, Kaiser M, Hilgetag CC. Organization, development and function of complex brain networks. Trends Cogn Sci. 2004;8(9):418–25.
- 64. McIntosh AR. Towards a network theory of cognition. Neural Netw. 2000;13(8–9):861–70.
- Volkow ND, Wolf AP, Brodie JD, et al. Brain interactions in chronic schizophrenics under resting and activation conditions. Schizophr Res. 1988;1(1):47–53.

- 66. Weinberger DR, Berman KF, Suddath R, Torrey EF. Evidence of dysfunction of a prefrontallimbic network in schizophrenia: a magnetic resonance imaging and regional cerebral blood flow study of discordant monozygotic twins. Am J Psychiatry. 1992;149(7):890–7.
- Bullmore ET, Frangou S, Murray RM. The dysplastic net hypothesis: an integration of developmental and dysconnectivity theories of schizophrenia. Schizophr Res. 1997;28(2–3):143–56.
- Meyer-Lindenberg A, Poline JB, Kohn PD, et al. Evidence for abnormal cortical functional connectivity during working memory in schizophrenia. Am J Psychiatry. 2001;158(11): 1809–17.
- Andreasen NC, Paradiso S, O'Leary DS. "Cognitive dysmetria" as an integrative theory of schizophrenia: a dysfunction in cortical-subcortical-cerebellar circuitry? Schizophr Bull. 1998;24(2):203–18.
- McGuire PK, Frith CD. Disordered functional connectivity in schizophrenia. Psychol Med. 1996;26(4):663–7.
- Fletcher P, McKenna PJ, Friston KJ, Frith CD, Dolan RJ. Abnormal cingulate modulation of fronto-temporal connectivity in schizophrenia. Neuroimage. 1999;9(3):337–42.
- 72. Friston KJ. Theoretical neurobiology and schizophrenia. Br Med Bull. 1996;52(3):644-55.
- 73. Liang M, Zhou Y, Jiang T, et al. Widespread functional disconnectivity in schizophrenia with resting-state functional magnetic resonance imaging. Neuroreport. 2006;17(2):209–13.
- 74. Benetti S, Mechelli A, Picchioni M, Broome M, Williams S, McGuire P. Functional integration between the posterior hippocampus and prefrontal cortex is impaired in both first episode schizophrenia and the at risk mental state. Brain. 2009;132(Pt 9):2426–36.
- Henseler I, Falkai P, Gruber O. Disturbed functional connectivity within brain networks subserving domain-specific subcomponents of working memory in schizophrenia: relation to performance and clinical symptoms. J Psychiatr Res. 2010;44(6):364–72.
- Lawrie SM, Buechel C, Whalley HC, Frith CD, Friston KJ, Johnstone EC. Reduced frontotemporal functional connectivity in schizophrenia associated with auditory hallucinations. Biol Psychiatry. 2002;51(12):1008–11.
- Wolf DH, Gur RC, Valdez JN, et al. Alterations of fronto-temporal connectivity during word encoding in schizophrenia. Psychiatry Res. 2007;154(3):221–32.
- Greicius M. Resting-state functional connectivity in neuropsychiatric disorders. Curr Opin Neurol. 2008;21(4):424–30.
- Garrity AG, Pearlson GD, McKiernan K, Lloyd D, Kiehl KA, Calhoun VD. Aberrant "default mode" functional connectivity in schizophrenia. Am J Psychiatry. 2007;164(3):450–7.
- Whitfield-Gabrieli S, Thermenos HW, Milanovic S, et al. Hyperactivity and hyperconnectivity of the default network in schizophrenia and in first-degree relatives of persons with schizophrenia. Proc Natl Acad Sci U S A. 2009;106(4):1279–84.
- Zhou Y, Liang M, Tian L, et al. Functional disintegration in paranoid schizophrenia using resting-state fMRI. Schizophr Res. 2007;97(1–3):194–205.
- Bluhm RL, Miller J, Lanius RA, et al. Spontaneous low-frequency fluctuations in the BOLD signal in schizophrenic patients: anomalies in the default network. Schizophr Bull. 2007;33(4):1004–12.
- Whitfield-Gabrieli S, Ford JM. Default mode network activity and connectivity in psychopathology. Annu Rev Clin Psychol. 2012;8:49–76.
- Broyd SJ, Demanuele C, Debener S, Helps SK, James CJ, Sonuga-Barke EJ. Default-mode brain dysfunction in mental disorders: a systematic review. Neurosci Biobehav Rev. 2009;33(3):279–96.
- Sakoglu U, Pearlson GD, Kiehl KA, Wang YM, Michael AM, Calhoun VD. A method for evaluating dynamic functional network connectivity and task-modulation: application to schizophrenia. MAGMA. 2010;23(5–6):351–66.
- Jafri MJ, Pearlson GD, Stevens M, Calhoun VD. A method for functional network connectivity among spatially independent resting-state components in schizophrenia. Neuroimage. 2008;39(4):1666–81.
- Honey GD, Pomarol-Clotet E, Corlett PR, et al. Functional dysconnectivity in schizophrenia associated with attentional modulation of motor function. Brain. 2005;128(Pt 11):2597–611.
- Kim DI, Manoach DS, Mathalon DH, et al. Dysregulation of working memory and defaultmode networks in schizophrenia using independent component analysis, an fBIRN and MCIC study. Hum Brain Mapp. 2009;30(11):3795–811.
- Liu Y, Liang M, Zhou Y, et al. Disrupted small-world networks in schizophrenia. Brain. 2008;131(Pt 4):945–61.
- 90. Tan HY, Sust S, Buckholtz JW, et al. Dysfunctional prefrontal regional specialization and compensation in schizophrenia. Am J Psychiatry. 2006;163(11):1969–77.
- Bassett DS, Bullmore E, Verchinski BA, Mattay VS, Weinberger DR, Meyer-Lindenberg A. Hierarchical organization of human cortical networks in health and schizophrenia. J Neurosci. 2008;28(37):9239–48.
- van den Heuvel MP, Mandl RC, Stam CJ, Kahn RS, Hulshoff Pol HE. Aberrant frontal and temporal complex network structure in schizophrenia: a graph theoretical analysis. J Neurosci. 2010;30(47):15915–26.
- Wang L, Metzak PD, Honer WG, Woodward TS. Impaired efficiency of functional networks underlying episodic memory-for-context in schizophrenia. J Neurosci. 2010;30(39):13171–9.
- 94. Yu Q, Sui J, Rachakonda S, et al. Altered topological properties of functional network connectivity in schizophrenia during resting state: a small-world brain network study. PLoS One. 2011;6(9):e25423.
- Zalesky A, Fornito A, Seal ML, et al. Disrupted axonal fiber connectivity in schizophrenia. Biol Psychiatry. 2011;69(1):80–9.
- 96. Wang Q, Su TP, Zhou Y, et al. Anatomical insights into disrupted small-world networks in schizophrenia. Neuroimage. 2012;59(2):1085–93.
- Rubinov M, Knock SA, Stam CJ, et al. Small-world properties of nonlinear brain activity in schizophrenia. Hum Brain Mapp. 2009;30(2):403–16.
- Yu Q, Plis SM, Erhardt EB, et al. Modular organization of functional network connectivity in healthy controls and patients with schizophrenia during the resting state. Front Syst Neurosci. 2011;5:103.
- 99. He H, Sui J, Yu Q, et al. Altered small-world brain networks in schizophrenia patients during working memory performance. PLoS One. 2012;7(6):e38195.
- 100. van den Berg D, Gong P, Breakspear M, van Leeuwen C. Fragmentation: loss of global coherence or breakdown of modularity in functional brain architecture? Front Syst Neurosci. 2012;6:20.
- Schultz CC, Fusar-Poli P, Wagner G, et al. Multimodal functional and structural imaging investigations in psychosis research. Eur Arch Psychiatry Clin Neurosci. 2012;262 Suppl 2:S97–S106.
- 102. Koch K, Schultz CC, Wagner G, et al. Disrupted white matter connectivity is associated with reduced cortical thickness in the cingulate cortex in schizophrenia. Cortex. 2013;49(3): 722–9.
- 103. Rasser PE, Johnston P, Lagopoulos J, et al. Functional MRI BOLD response to Tower of London performance of first-episode schizophrenia patients using cortical pattern matching. Neuroimage. 2005;26(3):941–51.
- 104. Schultz CC, Koch K, Wagner G, et al. Reduced anterior cingulate cognitive activation is associated with prefrontal-temporal cortical thinning in schizophrenia. Biol Psychiatry. 2012;71(2):146–53.
- 105. Calhoun VD, Adali T, Giuliani NR, Pekar JJ, Kiehl KA, Pearlson GD. Method for multimodal analysis of independent source differences in schizophrenia: combining gray matter structural and auditory oddball functional data. Hum Brain Mapp. 2006;27(1):47–62.
- 106. Hagmann P, Cammoun L, Gigandet X, et al. Mapping the structural core of human cerebral cortex. PLoS Biol. 2008;6(7):e159.
- 107. Skudlarski P, Jagannathan K, Calhoun VD, Hampson M, Skudlarska BA, Pearlson G. Measuring brain connectivity: diffusion tensor imaging validates resting state temporal correlations. Neuroimage. 2008;43(3):554–61.

- Orru G, Pettersson-Yeo W, Marquand AF, Sartori G, Mechelli A. Using Support Vector Machine to identify imaging biomarkers of neurological and psychiatric disease: a critical review. Neurosci Biobehav Rev. 2012;36(4):1140–52.
- Van Essen DC, Ugurbil K, Auerbach E, et al. The Human Connectome Project: a data acquisition perspective. Neuroimage. 2012;62(4):2222–31.
- Woolley J, McGuire P. Neuroimaging in schizophrenia: what does it tell the clinician? Adv Psychiatr Treat. 2005;11(3):195–202.
- 111. Thompson DAW. On growth and form. Cambridge: Cambridge University Press; 1917.
- 112. Grenander U, Miller MI. Computational anatomy: an emerging discipline. Q Appl Math. 1998;LVI(4):617–94.
- Miller MI, Qiu A. The emerging discipline of computational functional anatomy. Neuroimage. 2009;45 Suppl 1:S16–39.
- 114. Ashburner J, Csernansky JG, Davatzikos C, Fox NC, Frisoni GB, Thompson PM. Computerassisted imaging to assess brain structure in healthy and diseased brains. Lancet Neurol. 2003;2(2):79–88.
- Thompson PM, Miller MI, Ratnanather JT, Poldrack RA, Nichols TE. Preface to the special issue. NeuroImage 2004; 23 Suppl 1:S1. doi: 10.1016/j.neuroimage.2004.07.009.
- 116. Klein A, Andersson J, Ardekani BA, et al. Evaluation of 14 nonlinear deformation algorithms applied to human brain MRI registration. Neuroimage. 2009;46(3):786–802.
- 117. Csernansky JG, Wang L, Joshi SC, Ratnanather JT, Miller MI. Computational anatomy and neuropsychiatric disease: probabilistic assessment of variation and statistical inference of group difference, hemispheric asymmetry, and time-dependent change. Neuroimage. 2004;23 Suppl 1:S56–68.
- Friston KJ, Frith CD, Liddle PF, Frackowiak RS. Functional connectivity: the principalcomponent analysis of large (PET) data sets. J Cereb Blood Flow Metab. 1993;13(1):5–14.
- 119. Biswal B, Yetkin FZ, Haughton VM, Hyde JS. Functional connectivity in the motor cortex of resting human brain using echo-planar MRI. Magn Reson Med. 1995;34(4):537–41.
- 120. Bell AJ, Sejnowski TJ. An information-maximization approach to blind separation and blind deconvolution. Neural Comput. 1995;7(6):1129–59.
- 121. Calhoun VD, Adali T, McGinty VB, Pekar JJ, Watson TD, Pearlson GD. fMRI activation in a visual-perception task: network of areas detected using the general linear model and independent components analysis. Neuroimage. 2001;14(5):1080–8.
- Calhoun VD, Adali T, Pearlson GD, Pekar JJ. A method for making group inferences from functional MRI data using independent component analysis. Hum Brain Mapp. 2001; 14(3):140–51.
- 123. Calhoun VD, Adali T, Pearlson GD, Pekar JJ. Spatial and temporal independent component analysis of functional MRI data containing a pair of task-related waveforms. Hum Brain Mapp. 2001;13(1):43–53.
- 124. Calhoun VD, Adali T. Multisubject independent component analysis of fMRI: a decade of intrinsic networks, default mode, and neurodiagnostic discovery. IEEE Rev Biomed Eng. 2012;5:60–73.
- 125. Calhoun VD, Eichele T, Pearlson G. Functional brain networks in schizophrenia: a review. Front Hum Neurosci. 2009;3:17.
- 126. Lee MH, Smyser CD, Shimony JS. Resting-State fMRI: a review of methods and clinical applications. AJNR Am J Neuroradiol. 2013;34(10):1866–72.
- 127. Fox MD, Snyder AZ, Vincent JL, Corbetta M, Van Essen DC, Raichle ME. The human brain is intrinsically organized into dynamic, anticorrelated functional networks. Proc Natl Acad Sci U S A. 2005;102(27):9673–8.
- Watts DJ, Strogatz SH. Collective dynamics of 'small-world' networks. Nature. 1998; 393(6684):440–2.
- Latora V, Marchiori M. Efficient behavior of small-world networks. Phys Rev Lett. 2001; 87(19):198701.
- 130. Newman ME. Assortative mixing in networks. Phys Rev Lett. 2002;89(20):208701.
- 131. Milgram S. Small-world problem. Psychol Today. 1967;1(1):61-7.

- 132. Bullmore E, Sporns O. Complex brain networks: graph theoretical analysis of structural and functional systems. Nat Rev Neurosci. 2009;10(3):186–98.
- 133. Wen W, He Y, Sachdev P. Structural brain networks and neuropsychiatric disorders. Curr Opin Psychiatry. 2011;24(3):219–25.
- 134. Humphries MD, Gurney K, Prescott TJ. The brainstem reticular formation is a small-world, not scale-free, network. Proc Biol Sci. 2006;273(1585):503–11.
- 135. Micheloyannis S, Pachou E, Stam CJ, et al. Small-world networks and disturbed functional connectivity in schizophrenia. Schizophr Res. 2006;87(1–3):60–6.
- 136. Pachou E, Vourkas M, Simos P, et al. Working memory in schizophrenia: an EEG study using power spectrum and coherence analysis to estimate cortical activation and network behavior. Brain Topogr. 2008;21(2):128–37.
- 137. Rutter L, Nadar SR, Holroyd T, et al. Graph theoretical analysis of resting magnetoencephalographic functional connectivity networks. Front Comput Neurosci. 2013;7:93.
- 138. Breakspear M, Terry JR, Friston KJ, et al. A disturbance of nonlinear interdependence in scalp EEG of subjects with first episode schizophrenia. Neuroimage. 2003;20(1):466–78.
- 139. Minzenberg MJ, Laird AR, Thelen S, Carter CS, Glahn DC. Meta-analysis of 41 functional neuroimaging studies of executive function in schizophrenia. Arch Gen Psychiatry. 2009;66(8):811–22.
- Achim AM, Lepage M. Episodic memory-related activation in schizophrenia: meta-analysis. Br J Psychiatry. 2005;187:500–9.
- 141. Goghari VM, Sponheim SR, MacDonald 3rd AW. The functional neuroanatomy of symptom dimensions in schizophrenia: a qualitative and quantitative review of a persistent question. Neurosci Biobehav Rev. 2010;34(3):468–86.
- 142. Kubicki M, McCarley R, Westin CF, et al. A review of diffusion tensor imaging studies in schizophrenia. J Psychiatr Res. 2007;41(1–2):15–30.
- 143. van Veen V, Carter CS. The anterior cingulate as a conflict monitor: fMRI and ERP studies. Physiol Behav. 2002;77(4–5):477–82.
- 144. Shizukuishi T, Abe O, Aoki S. Diffusion tensor imaging analysis for psychiatric disorders. Magn Reson Med Sci. 2013;12(3):153–9.
- 145. Boksman K, Theberge J, Williamson P, et al. A 4.0-T fMRI study of brain connectivity during word fluency in first-episode schizophrenia. Schizophr Res. 2005;75(2–3):247–63.
- 146. Ma S, Calhoun VD, Eichele T, Du W, Adali T. Modulations of functional connectivity in the healthy and schizophrenia groups during task and rest. Neuroimage. 2012;62(3):1694–704.

Part III Medical Management of Schizophrenia

Chapter 7 Acute Management of Schizophrenia

Philip G. Janicak

Introduction

The largest database for the treatment of acute schizophrenia involves antipsychotic drug therapy, primarily to manage multiepisode patients. Ironically, management of high-risk, first-episode, and early-onset patients is less well studied but may be more critical to ultimately achieve recovery.

Strategies to optimize outcome at any phase of schizophrenia should consider the various levels of improvement. *Response* is typically defined as at least a 20–30 % reduction in symptom severity (e.g., positive, negative) based on various assessment metrics used in antipsychotic drug trials. In clinical settings, it typically implies sufficient improvement of symptoms in an adherent patient to reduce or eliminate the need for hospitalization. *Remission* refers to sustained (e.g., ≥ 6 months) maintenance of minimal symptoms in dimensions such as psychosis and disorganization. It typically requires a comprehensive strategy beyond medication including case management, psychosocial interventions, psychotherapy, and vocational training. Ideally, this should involve the patient, family and other caregivers. *Recovery* requires improvements in cognitive, social, and vocational functioning over sustained periods of time, allowing individuals the opportunity to realize their personal goals (e.g., friends, employment, marriage).

This chapter considers the data and clinical experience with various strategies for acute management of schizophrenia. Since inadequate response and treatmentresistance are common, augmentation of or alternatives to standard antipsychotics are also reviewed.

P.G. Janicak (🖂)

Department of Psychiatry, Rush University Medical Center, 2150 West Harrison Street, Chicago, IL 60612, USA e-mail: pjanicak@rush.edu

Choice and Dosing of an Antipsychotic

The choice and dosing of an antipsychotic are critical to maximizing benefit and take into account the quality and quantity of the evidence base, the clinical experience with a specific agent and the risk-to-benefit ratio [1]. The results from large "pragmatic trials" such as the Clinical Antipsychotic Trials of Intervention Effectiveness (CATIE) and the Cost Utility of the Latest Antipsychotic Drugs in Schizophrenia Study (CUtLASS) and expert consensus guidelines such as the Schizophrenia Patient Outcomes Research Team (PORT) project also help guide clinicians in this choice [2–4]. Important factors in choosing an antipsychotic may include personal or physician preference, history of treatment response or resistance, sensitivity to certain adverse effects, comorbid substance use and/or medical disorders, available formulations, long-term planning, and cost.

In time, pharmacogenetics and pharmacogenomics will increasingly clarify how inheritance and acquired genetic variations (e.g., epigenetics) alter a drug's effects, promising to facilitate:

- Identification of biomarkers to optimize treatment
- Avoidance of treatments likely to produce adverse effects
- Improved understanding of a drug's mechanism of action
- Development of new targets for drug therapy
- Prediction of *metabolic profiles* to help with the choice of drug, duration of treatment, and/or dose
- *Enhancement of medication adherence* and the subsequent reduction in relapse rates [5]

Pharmacogenetics identifies single nucleotide polymorphisms (SNPs) associated with a drug's action and then tests for an association to clinical response [6]. As an example, this strategy was used to study dopamine and serotonin transporter gene polymorphisms to assess for the potential response and the risk for EPS with clozapine (CLZ) [7, 8]. *Pharmacogenomics* uses a genome-wide association study (GWAS) approach to interrogate thousands of SNPs randomly distributed across the genome [9]. This may identify "profiles" of SNPs with additive predictive value for clinical efficacy. In one study, six genetic loci were associated with response to iloperidone [10]. More recently, however, their specificity was questioned, since two of these SNPs also appeared predictive of response to risperidone (see Chap. 9) [11].

Twenty-seven antipsychotics are available in the United States. They are grouped into typical or first-generation antipsychotics (FGAs) and atypical or secondgeneration antipsychotics (SGAs). The validity of such a classification, however, is debated since there are many similarities between drugs within these two groups. In this context, adverse effects may still help to distinguish them [12]. For example, neuromotor toxicity is more characteristic of high potency FGAs (e.g., haloperidol, fluphenazine, thiothixene), while weight gain and metabolic dysregulation are more characteristic of certain SGAs (e.g., CLZ, olanzapine, quetiapine). Given our present knowledge about the relative efficacy of the various antipsychotics, the initial choice of a specific agent is often based on safety and tolerability considerations. This is discussed in greater detail in Chaps. 10-12.

In the context of dosing, neuroimaging (e.g., positron emission tomography (PET), single photon emission computer tomography (SPECT)) and dose/plasma level response studies can provide guidance for certain agents. For example, a recent meta-analysis of neuroimaging studies supports a role for D_2 occupancy levels in achieving optimal dosing of antipsychotics with the exception of CLZ and quetiapine [13]. Studies also indicate plasma levels of certain agents such as haloperidol (e.g., 3–15 ng/mL range) are optimal and usually achieved at much lower doses than initially recommended [14]. Existing data also indicates that optimization of CLZ's efficacy may occur with levels of 350–800 ng/mL [15].

Table 7.1 lists the antipsychotics presently available in the United States with information about their recommended dosing range and alternate routes of administration. Further, although not recommended for routine clinical use, plasma level data may help guide dosing in certain circumstances (e.g., suspected adherence problems, potential for drug interactions, unexpected adverse effects, insufficient benefit with adequate dosing) and results of preliminary trials are provided [15–19].

Acute Treatment of Psychosis

High-Risk Individuals

Patients who develop schizophrenia may experience subtle changes in behavior and cognition for long periods prior to their first psychotic exacerbation. Early recognition (e.g., high-risk phase) and appropriate interventions can facilitate symptom reduction and possibly delay or prevent psychosis onset [20, 21]. The accurate identification of those who will transition from this "prepsychotic phase" to schizophrenia, however, is problematic [22, 23]. Thus, while individuals with potential prodromal symptoms are at an increased risk to transition to a psychotic disorder, less than 40 % will and some recent data indicates a decline in these transition rates [20]. Further, many individuals go on to develop other psychiatric disorders or their symptoms will resolve before experiencing a full psychotic episode. Complicating this issue is the potential for schizophrenia to develop later than the timeframes studied to date.

In this context, a number of centers are developing criteria to characterize this phase. For example, the ultra-high risk (UHR) criteria require the presence of one or more of the following:

- Attenuated psychotic symptoms
- Brief, limited, intermittent psychotic episodes
- Trait vulnerability (i.e., genetic risk)
- Psychosocial functioning decline
- Prodromal symptoms [24, 25]

	Oral dosing					
Name	range (mg)	Alternate formulations/putative therapeutic plasma levels ^a				
First-generation ag	gents					
Phenothiazines						
Chlorpromazine	100-1,000	Parenteral; rectal suppository				
Thioridazine	30-800	Oral concentrate; oral suspension				
Fluphenazine	5–40	Oral concentrate; acute parenteral; LAI (12.5–25 mg/2–3 week); plasma level of 1–3 ng/mL				
Trifluoperazine	2-60	Plasma level of 1–2.3 ng/mL				
Perphenazine	2-60	Plasma level of 1.5–3 mmoL/L				
Thioxanthenes						
Thiothixene	6-60	Oral concentrate; plasma level of 2-15 ng/mL				
Dibenzoxapines						
Loxapine	20–250	Acute inhalant (single dose of 5 or 10 mg)				
Butyrophenones Haloperidol	3–50	Oral concentrate; acute parenteral; LAI (50–200 mg/4 week); plasma level of 3–15 ng/mL				
Dihydroindolones						
Molindone	15-225	Oral concentrate				
Second-generation	agents					
Dibenzodiazepines						
Clozapine	100-900	Oral rapid dissolving; plasma level of 350–800 ng/mL				
Benzisoxazoles						
Risperidone	2-8	Oral: liquid and rapid dissolving; parenteral: LAI (150– 405 mg/2–4 week); plasma level of 20–60 ng/mL				
Paliperidone	3–12 Oral extended release; parenteral LAI (117 mg/4 week); plasma level of 20–52 ng/mL					
Iloperidone	12-24	-				
Lurasidone	40-80					
Thienobenzodiazer	oines					
Olanzapine	5-20	Oral rapid dissolving; parenteral LAI (150–405 mg/2–4 week); plasma level > 20 ng/mL				
Dibenzothiazepine	s	F				
Ouetiapine	75-800	Oral immediate and extended release				
Benzisothiazolvls						
Ziprasidone	40-160	Acute parenteral				
Ouinolinones		F				
Aripiprazole	5–30	Oral: liquid and rapid dissolving; parental: acute and LAI; plasma level 150–210 ng/mL				
Dibenzo-oxepinopyrrole						
Asenapine	10-20	Sublingual formulation only				
1		с ,				

 Table 7.1
 Commonly used first- and second-generation antipsychotics

^aBased on preliminary data

The North American Prodrome Longitudinal Study also identifies other potential predictors of psychosis, including high levels of unusual thought content, high suspicion/paranoia, and a history of substance abuse [26]. There is also preliminary data indicating presynaptic striatal dopamine abnormalities may precede the onset of

psychosis [27]. Various assessment tools are available, such as the Structured Interview for Prodromal Symptoms (SIPS), to help identify these individuals [28]. To qualify for high risk using this metric requires at least one fully positive psychotic symptom several times per week for at least 1 month; or for at least 1 day if the symptom causes significant disorganization or is dangerous.

In testament to the substantial progress made in this area over the past 20 years, the Diagnostic and Statistical Manual of Mental Illness (DSM)-5 will include the category of Attenuated Psychosis Syndrome in its Sect. 3 [29]. This means the category is of note but requires further study (see Chap. 2).

Treatment of High-Risk Individuals

There are preliminary data to guide clinicians in helping these individuals. Two reviews identified five controlled studies of focused treatments for those at high risk [30, 31]. These trials indicated that both pharmacological and psychological approaches may be beneficial. For example, both intensive community care with family psychoeducation and omega-3 polyunsaturated fatty acid (PUFA) supplementation reduced the transition to psychosis at 12 months [32]. Other "neuroprotective strategies" such as *N*-acetyl cysteine (NAC) are also possible candidate treatments for these individuals [33, 34]. In general, these strategies carry less risk of adverse effects than standard antipsychotics and might be first-line approaches if these early results are replicated [35, 36]. The use of antipsychotics should be reserved for more complicated or resistant individuals. In this context, a small (n=59) controlled trial combining low-dose risperidone (mean dose=1.3 mg) with cognitive behavioral therapy (CBT) reported a reduction in the risk of early transition to psychosis [21]. The dilemma, however, is that a small decrease in the conversion rate to psychosis involves a substantial risk of adverse effects associated with antipsychotics.

First-Episode and Early-Onset Patients

Recognizing the earliest signs of clear psychosis and introducing various interventions can reduce the duration of untreated symptoms and positively impact long-term outcome. While an average of 1 year elapses from the onset of psychotic symptoms to first treatment, two meta-analyses found that a shorter duration of untreated psychosis was associated with a better response and subsequent reduction in negative symptoms [37, 38]. Conversely, there is evidence demonstrating an association between longer duration of untreated psychosis in first-episode patients and worse outcomes in terms of psychotic symptoms, cognition, functioning, and quality of life [39–41].

In addition, imaging studies indicate that a subset of these patients demonstrate significant decreases in gray matter, white matter, and increases in cerebrospinal fluid during the early years after illness onset [42]. Complicating these observations are conflicting reports of an association between antipsychotic drug exposure and

either neuroprotective or neurotoxic effects. While these findings suggest a possible causal relationship, ethical issues preclude more definitive study designs and leave unclear whether such effects are deleterious or beneficial [43, 44].

Treatment of First-Episode and Early-Onset Patients

Several studies consider the benefit of earlier intervention in first-episode and earlyonset patients with schizophrenia and related psychotic disorders. In one intensive program (i.e., OPUS), 369 first-episode patients were followed for 2 years and 301 for 5 years [45]. In comparison with standard treatment, the addition of assertive community treatment, family involvement, and social skills training demonstrated better clinical outcomes at 2 but not 5 years (see Chap. 14). The investigational group, however, had fewer patients in supportive housing and experienced fewer days hospitalized over the 5-year period. A second study (n=281) found a significantly lower rate of serious suicidal behavior (i.e., p < 0.01) in first-episode patients participating in an early detection program which shortened the duration of untreated psychosis from 16 to 5 weeks compared with those not involved [46, 47]. A third program in Norway for first-episode patients (TIPS) followed 101 subjects for up to 10 years [48]. This approach initially achieved a reduction in the duration of untreated psychosis from a median of 26 to 5 weeks. In comparison with 73 usual care patients, the early detection group had a significantly greater recovery rate (p < 0.01), largely due to higher employment. The authors concluded that the early detection approach increased the chances of milder deficits and superior functioning over this time period. Recently, Wunderink et al. [49] reported that dose reduction/discontinuation (DR) in remitted first-episode patients (n=103) was superior to standard maintenance treatment in terms of recovery rates (i.e., 40 % versus 18 %). While the study confirmed an initial higher relapse rate in the DR group, surprisingly, this was not reflected in remission rates (equal between the two groups) or in the recovery rates (superior on the DR group; p < 0.01) after 7 years.

Other studies also consider the relative benefits of various antipsychotics in treating first-episode or early-onset schizophrenia. One trial (n=263) compared olanzapine with haloperidol for up to 2 years in first-episode patients [50]. Both agents demonstrated substantial and similar improvement in overall symptoms. Olanzapine, however, produced greater decreases in symptom severity based on the Positive and Negative Symptom Scale (PANSS) total, negative, and general psychopathology subscale scores and had a greater percentage of patients complete the 12-week acute phase. A controlled trial (n=160) conducted in China, found an initial benefit for CLZ versus chlorpromazine (CPZ) in treatment-naïve, first-episode patients after 12 weeks. This difference, however, disappeared at the 1-year and 9-year follow-up time points [51, 52]. Median doses at 1 year were 300 mg for CLZ and 400 mg for CPZ. Of note, patients receiving CLZ remitted significantly faster, remained in remission longer and experienced fewer adverse effects compared with CPZ. Another small, controlled trial (n=39) found that CLZ was superior to "high dose" olanzapine (i.e., up to 30 mg/day) in refractory, early-onset schizophrenia with

response rates of 66 % and 33 % (p < 0.04), respectively [53]. Both agents, however, were associated with significant weight gain and metabolic dysregulation.

A large, long-term study (n = 555) randomized first-episode patients to low-dose risperidone (mean modal dose=3.3 mg) or low-dose haloperidol (mean modal dose=2.9 mg [54]. While both groups demonstrated improvement in cognitive functioning over a 3-month period, risperidone improved more aspects (e.g., global functioning) to a greater degree than haloperidol. Further, the median time to relapse was 466 days with risperidone versus 205 days with haloperidol [55]. A second study (n = 104) compared haloperidol (mean dose = 2.7 mg/day), risperidone (mean dose = 3.9 mg/day), and olanzapine (mean dose = 10 mg/day) [56]. After 1 year, all three treatment groups demonstrated significant and similar improvements in cognition. In a 52-week, randomized, double-blind, flexible-dose, multicenter study (n=400), patients early in the course of their psychotic illness received olanzapine (mean modal dose = 11.7 mg), quetiapine (mean modal dose = 506 mg), or risperidone (mean modal dose 2.4 mg) [57]. Based on all-cause treatment discontinuation, these agents were similarly effective. All three agents also produced significant and similar modest improvements in neurocognition which were associated with improved functional outcomes [58].

The Treatment of Early-Onset Schizophrenia Spectrum Disorders Study (TEOSS) randomized 116 youth with schizophrenia or schizoaffective disorder to olanzapine (2.5-20 mg/day), risperidone (0.5-6 mg/day), or molindone (10-140 mg/day) plus benztropine (1 mg/day) [59]. While the degree of symptom reduction and response rates was comparable among the three agents, adverse effects differed. Indeed, olanzapine was discontinued due to its adverse effects in this population. Given their results, the authors questioned the nearly exclusive use of SGAs in this population in light of the associated greater risk for weight gain and metabolic issues. The European First-Episode Schizophrenia Trial (EUFEST) compared five antipsychotics in 498 patients (ages 18–40 years) with first-episode schizophrenia, schizoaffective disorder, or schizophreniform disorder [60]. In a 1-year, randomized, open-label, multicenter design, patients received haloperidol (1-4 mg/day), amisulpride (200-800 mg/day), olanzapine (5-20 mg/day), quetiapine (200-750 mg/day), or ziprasidone (40-160 mg/ day). Based on all-cause treatment discontinuation, the SGAs had a lower risk compared with haloperidol, while symptom improvement was similar for all drugs. In a post hoc analysis, the authors concluded that the results favor the SGAs given the importance of continuing treatment at this phase of the illness [61]. In contrast, a meta-analysis of 15 controlled trials (n=2,522) found no differences in discontinuation rates or symptom improvement between FGAs and SGAs in this patient population [62]. Adverse effects of SGAs (i.e., weight gain) and FGAs (i.e., EPS), however, clearly distinguished these two groups.

In summary, effective treatment is critical in the first 5 years after the onset of psychosis since maximal disability may occur during this period. Addressing known predictors of treatment nonadherence (e.g., depression; substance abuse) is also important to assure optimal outcomes during the recovery from a first episode [63]. When antipsychotics are prescribed, low doses of SGAs with less propensity to cause weight gain and metabolic complications are recommended first-line

approaches. In this context, the PORT recommendations relegate olanzapine and CLZ to second-line choices. In appropriate patients, long-acting injectable (LAI) formulations may be warranted. After two failed adequate trials, CLZ should be considered. The goal is to achieve remission (i.e., mild levels of key symptoms for at least 6 months) whenever possible [64].

Multiepisode Patients

For multiepisode patients responsive to prior treatment, common strategies for acute exacerbations include starting with the previously effective agent or certain generic FGAs (e.g., perphenazine) and SGAs (e.g., olanzapine, risperidone). Many of these antipsychotics also have alternate formulations (e.g., acute parenteral, oral liquid, disintegrating tablets) which may be used in emergencies for uncooperative patients or for those who cannot swallow tablets or capsules. Some also have LAI formulations, which makes the transition from oral medication easier if such a maintenance strategy is pursued.

For agitation and disruptive behaviors, non-pharmacological de-escalation techniques are the preferred first option. If this approach is insufficient, the short-term use of an adjunctive benzodiazepine (e.g., oral or acute parenteral lorazepam) can calm the patient and reduce the dose of antipsychotic. Alternate strategies include acute parenteral FGAs and SGAs, as well as acute loxapine inhalation therapy, all of which may be combined with a benzodiazepine [65, 66]. The goal is to optimize the intended antipsychotic effect with the minimally effective dose. Lessening longer-term exposure to polypharmacy is also a goal. This may be particularly true for benzodiazepines given the high rates of substance use disorders in these patients and whose extended exposure may be associated with increased mortality rates [67]. The hope is that such a strategy will substantially reduce adverse effects (e.g., acute EPS) and facilitate longer-term adherence.

Patients should demonstrate acceptable improvement during the first 2 weeks. If not, an alternate agent or augmentation of the initial but insufficient benefit is usually appropriate. Augmentation may include additional medications (e.g., combined antipsychotics), alternate nonpharmacologic, biological treatments (e.g., therapeutic neuromodulation), various types of psychotherapy (e.g., cognitive behavioral), and rehabilitation programs (e.g., social skills training).

After two or more failed adequate antipsychotic trials, CLZ is a reasonable next step [4]. Its advantages over other available antipsychotics include evidence for benefit in treatment-refractory, hostile, aggressive, violent, and suicidal patients. It may also lower the risk of death [68]. CLZ, however, has several disadvantages which require a careful risk-benefit assessment prior to its prescription. For example, its use requires ongoing blood count monitoring to minimize the risk of neutropenia and agranulocytosis. This increases the cost and complexity of managing patients on this agent (Table 7.2).

Table 7.2 Clozapine

Potential benefits	Potential risks ^a Box warnings	
Efficacy		
May benefit treatment-refractory patients	 Agranulocytosis 	
• May reduce suicidal, agressive, or violent behavior	Seizures	
May increase life expectancy	Myocarditis	
Adverse effects	Orthostasis	
Diminish extrapyramidal side effects	Increased mortality in dementia	
Minimize risk for or improve tardive dyskinesia	Serious adverse effects	
Minimize hyperprolactinemia	Weight/gain metabolic syndrome	
	Diabetic ketoacidosis	
	Gastrointestinal hypomotility	

^aSee Chap. 12 also

Conclusion

Early intervention in high-risk individuals and first-episode patients leading to shortened duration of untreated psychosis appears to favorably impact the course of schizophrenia. While many of these individuals do not transition to psychosis, they may still experience a variety of problems such as anxiety, depression, heightened suicidality, substance use disorders, negative symptoms, impaired academic performance, poor occupational functioning, and interpersonal problems. Even when psychotic symptoms do not emerge, these issues can be the focus of treatment which may improve quality of life. Nonpharmacological approaches (e.g., psychosocial interventions, psychotherapy) and neuroprotective agents (e.g., omega-3 fatty acids) are appropriate first-line interventions in high-risk groups given the adverse effect burden of antipsychotics and since a substantial proportion of these individuals will not go on to develop schizophrenia (at least in the immediate future). With clear psychotic behaviors, however, the introduction of antipsychotic medication conservatively dosed is warranted to shorten the duration of untreated psychosis [69]. In multiepisode patients, the focus should be on optimizing the dose of antipsychotic, improving adherence, establishing a therapeutic environment, and developing a constructive alliance with patients, families and other support groups.

Management of Treatment-Resistance

Given the multiple psychopathological dimensions of schizophrenia, drugs acting primarily through dopaminergic D_2 mechanisms are usually not sufficient to address this entire spectrum of symptoms. For example, while existing agents are relatively effective for positive symptoms, they are far less helpful for primary persistent negative, neurocognitive, and mood symptoms [70]. In addition, safety and tolerability

Potential benefits			Potential risks	
•	Facilitate switching antipsychotics	•	Lack of adequate evidence base	
•	Supplement long-acting agents	•	Increased mortality	
•	Improve <i>specific symptoms</i> (e.g., low-dose quetiapine for sedation)	•	Increase risk for <i>adverse effects</i> or <i>drug interactions</i>	
•	<i>Lower doses</i> of two agents to avoid safety or tolerability issues with standard dose of a single agent	•	Worsen <i>adherence issues</i> Increase cost	
•	Augment efficacy of primary antipsychotic			

Table 7.3 Combined antipsychotics

issues often prevent an adequate treatment course. Thus, when CLZ is not an option or proves intolerable or insufficient, there are a number of ongoing initiatives which may improve outcomes, including:

- · Refining existing agents to increase efficacy and/or decrease adverse effects
- Developing drugs with alternate mechanisms of action
- Augmenting standard antipsychotics with novel agents [71].

While there are many potential approaches reported in the literature, those discussed in this chapter have shown promising results, primarily from controlled trials. The quality and quantity of evidence, however, is presently limited and usually based on proof-of-concept study data.

Antipsychotic Polypharmacy

Combining antipsychotics is a frequently employed strategy in clinical practice. One rationale given for this approach is the ability to augment the effects of the first agent by tailoring the neuroreceptor profile of actions for a specific patient [72]. The most frequent combination studied to date involves CLZ polypharmacy for patients inadequately responsive to CLZ monotherapy [73, 74]. While the results from most of these controlled trials were negative, their quality was usually not adequate to definitively address the question. A meta-analysis of controlled trials comparing other antipsychotic combinations versus antipsychotic monotherapy also did not produce firm recommendations about the value of this approach [75]. In general, the paucity of data dictates the need for further studies. Table 7.3 summarizes the relative risks and benefits of this approach.

Agents Impacting Other Neurotransmitter Systems/Receptors

Dopamine

Existing antipsychotics directly impact the dopamine (DA) system, primarily through the D_2 receptor. Given an inadequate benefit in many patients and/or safety and tolerability issues, other ways of modulating the DA system (e.g., D_1 – D_5 receptors) are

being explored. D₁ receptors are abundant in the prefrontal cortex (PFC) and appear to play an important role in cognition (e.g., working memory) [76]. Several lines of evidence support a therapeutic potential by modulating these receptors. For example, hypofrontality is a common finding in schizophrenia, including medicationnaïve first-episode patients [77]. In this context, a D_1 agonist (e.g., dihydrexidine) was found to produce increased perfusion in prefrontal and nonprefrontal areas in these patients [78]. Presently, however, clinical data to support its efficacy in schizophrenia is insufficient [79, 80]. Both CLZ and asenapine appear to facilitate glutamate N-methyl-D-aspartate (NMDA) receptor-induced currents through modulation of prefrontal D_1 receptors [81]. Conversely, genetic variants of the D_1 receptor may predispose to adverse effects of antipsychotics such as tardive dyskinesia [82]. In addition, agents acting at the D₃ and D₄ receptors are also considered potential targets for treating schizophrenia [83, 84]. For example, *cariprazine* is a partial agonist at the D_2/D_3 (with preference for D_3) and 5-HT_{1A} receptors. Preclinical models predict procognitive effects and there are now positive Phase II and Phase III trial results [85]. Some evidence also supports selective D_3 agonism at presynaptic autoreceptors and antagonism at postsynaptic receptors as potential but relatively unexplored therapeutic approaches [86–88].

Norepinephrine

Norepinephrine (NE) can increase DA activity in the medial PFC. In turn, this may improve vigilance, cognition, and mood, as well as moderate stress reactions. Blockade of reuptake transporters, antagonism of presynaptic α_2 -NE receptors (preventing autoreceptor negative feedback on activity) and agonism of postsynaptic α_2 -NE receptors are also potentially therapeutic [89, 90]. Controlled trials have generated mixed results, however, regarding the cognitive benefit of NE reuptake inhibition with agents such as reboxetine or atomoxetine, usually as adjuncts to standard antipsychotics [91–94].

Serotonin

SGAs have multiple neuroreceptor interactions with the serotonin (5-HT) system. Thus, much interest is focused on developing agents which impact 5-HT receptor subtypes [95]. Evidence suggests that 5-HT_{1A} receptor agonism varies among available agents and may enhance the antipsychotic effects of D₂ receptor antagonists [96]. Newer compounds (e.g., cariprazine) possessing these two properties are now in development. Targeting other receptor subtypes may also improve psychosis, negative symptoms, and certain cognitive dimensions, while reducing the dose of standard antipsychotic and lessening their adverse effect burden. Examples include selective 5-HT_{2A} inverse agonists (e.g., pimavanserin), selective 5-HT_{2A} antagonists, selective 5-HT_{2C} agonists (e.g., vabicaserin), 5-HT₃ antagonists (e.g., ondansetron; tropisetron), 5-HT₄ receptor agonists, and 5-HT₆ (e.g., latrepirdine) and 5-HT₇ antagonists [97–100]. The most recent SGAs, (i.e., iloperidone, asenapine, and lurasidone) all demonstrate several of these properties [101–103].

Glutamate

Glutamate (Glu) is one of the most abundant free amino acids in the central nervous system (CNS) and the primary excitatory neurotransmitter. Glutamate contributes to the formation, maintenance, and plasticity of synaptic function, as well as facilitation of sensory processing, memory and executive functioning [104]. In schizophrenia, dysregulation of the glutamate system can compromise neuroplasticity and produce neuronal toxicity. The Glu NMDA ionotropic receptor may be critical to these processes [105]. For example, drugs which block this receptor (e.g., phencyclidine, ketamine) produce psychotic symptoms closely resembling schizophrenia. Some postulate, however, that it is the resultant overactivity of Glu, particularly at the α -amino-3-hydroxy-5-methyl-4-isoxazle propionic acid (AMPA) ionotropic receptor, which produces or worsens psychosis [105]. In this context, a recent, placebo-controlled trial (n=20) reported a rapid decrease in various symptoms of schizophrenia (i.e., positive, negative, depressive, anxious) after a single IV dose of nitroprusside (0.5 µg/kg/min) given over 4 h [106]. This agent is believed to modulate the NMDA receptor. Metabotropic Glu receptors are also implicated in the pathophysiology of schizophrenia and as such serve as potential therapeutic targets. Glu also interacts with other important systems (e.g., DA, GABA, Ach) which are modulated by and in turn also modulate its activity. As a result, there is great interest in a glutamatergic strategy to treat various psychopathological dimensions of schizophrenia. Since direct impact on Glu receptors may produce neuronal excitotoxicity, various indirect approaches are being explored.

Facilitation of the *glycine* receptor site colocated near the Glu site on the NMDA receptor may enhance channel efficiency [104]. A recent meta-analysis concluded that adjunctive glycine improved positive and total symptoms of schizophrenia when combined with non-CLZ antipsychotics, but worsened symptoms when added to CLZ [107]. Not all studies, however, support this strategy. The Cognitive and Negative Symptoms in Schizophrenia Trial (CONSIST) involved 157 patients with schizophrenia or schizoaffective disorder [108]. Patients were randomly assigned to 16 weeks of adjunctive glycine (60 g/day), the selective partial NMDA receptor agonist, D-cycloserine (50 mg/day), or placebo. The authors did not find benefit for these add-on agents when compared with placebo for the treatment of negative symptoms or cognitive impairment. A more recent, 16-week, placebo-controlled trial (n=195) involving patients with schizophrenia or schizoaffective, naturally occurring, allosteric modulator of the NMDA receptor, *D-serine* (2 g/day), for negative and cognitive symptoms [109].

Given the relatively poor CNS penetration and the resultant need for high doses of these amino acids, another strategy is to administer agents which act as glycine transport inhibitors, a strategy analogous to use of selective serotonin reuptake inhibitors [110]. A 6-week, double-blind trial (n=60) involving patients with schizophrenia found that 2 g/day of adjunctive *sarcosine* (N-methyl glycine), but not D-serine, was superior to placebo based on various clinical measures [111]. An earlier, controlled trial (n=20) also found scarcosine (2 g/day) was efficacious in drug-free patients with schizophrenia [112]. More recently, the glycine reuptake inhibitor bitopertin (RG 1678) at doses of 10 and 30 mg demonstrated efficacy

for negative symptoms in a placebo-controlled phase IIb trial (n=323) for schizophrenia [113]. Unfortunately, two recent Phase III trials found bitopertin was not beneficial for reducing predominant negative symptoms.

Another strategy is to modulate *metabotropic Glu 2/3 receptors*. Stimulation of presynaptic receptors inhibits further Glu release while postsynaptic receptor effects can further modify Glu activity. There was great initial excitement when *pomaglume-tad methionil* (a mGlu 2/3 receptor agonist) at doses of 40 mg BID demonstrated a benefit for positive and negative symptoms significantly better than placebo (p<0.001) and similar to olanzapine (15 mg/day) [114]. This was a 4-week, Phase II, controlled study (n=193) involving patients with schizophrenia. Subsequent trials, however, did not demonstrate efficacy for this agent compared with placebo, leading to a suspension of Phase III development [115, 116]. Positive allosteric modulation at other receptors (e.g., mGlu5), however, may be an alternate approach [105].

Other strategies include NAC which stimulates these receptors indirectly by enhancing formation of glutathione, a neuroprotective free radical scavenger which may counter oxidative stress [34]. In this context, a 24-week, controlled trial (n=140) found that adjunctive NAC (1 g BID) was superior to placebo in terms of the PANSS total (p<0.009), negative symptoms (p<0.02), and general psychopathology (p<0.04) scores, as well as the Clinical Global Impression-Severity (CGI-S) (p<0.04) and Clinical Global Impression-Improvement (CGI-I) (p<0.03) scores in patients with chronic schizophrenia [117]. A more recent controlled trial found that NAC (up to 2 g per day) improved scores on the PANSS total and the negative symptom subscale significantly more than placebo (p<0.006 and p<0.001, respectively) when used in combination with risperidone [118]. As noted earlier, NAC may also play a "protective" role in at-risk individuals, preventing the evolution into schizophrenia [33].

Strategies involving adjunctive anticonvulsants such as *lamotrigine* (LTG) (an inhibitor of glutamate release) plus CLZ, *memantine* (a weak nonselective NMDA receptor antagonist) plus CLZ, and *AMPA-receptor modulators* (e.g., ampakines such as CX 516) have also been considered. While preliminary data support each of these approaches, design limitations and difficulty with or lack of replication of these initial results leave unclear their potential roles [119–121].

Gamma-Aminobutyric Acid

Gamma-aminobutyric acid (GABA) is the primary inhibitory neurotransmitter in the CNS. There is evidence that cortical and subcortical GABA activity is disrupted in schizophrenia, including first-episode and antipsychotic-naïve patients [122]. Preclinical and clinical evidence support a potential role for agonists of this system to improve various symptoms. For example, activation of GABA interneurons may decrease hyperactivity of mesolimbic dopamine and/or glutamate, reducing positive symptoms.

One approach is to use agents with both DA antagonist and GABA agonist activity (e.g., BL-1020). In a placebo-controlled trial (n=363), this agent was superior to placebo (p<0.001) and comparable to risperidone in overall improvement (i.e., PANSS total score). Further, the 20–30 mg dose was superior to both placebo and risperidone (p<0.009 and p<0.019, respectively) on a measure of cognitive function [123].

Unfortunately, results from an interim analysis of a Phase II/III trial failed to meet the pre-specified primary cognitive efficacy endpoint and the trial was discontinued.

Another strategy to modulate this system involves the use of anticonvulsants which work in part through GABAergic mechanisms. While an initial 4-week study (n=249) generated encouraging results using adjunctive divalproex sodium (DVPX; 15–30 mg/kg/day), a subsequent 12-week, controlled trial (n=402) did not demonstrate efficacy for this agent at doses of 20–35 mg/kg/day when used as an adjunct to olanzapine or risperidone [124, 125]. This agent, however, did rapidly control hostility and anxiety. A small, 12-week, controlled study (n=24) also did not find benefit for either adjunctive DVPX or LTG versus placebo [126].

Selective agonists of GABA_A receptors represent an alternate strategy. Working memory deficits are associated with impairments in PFC function, possibly related to altered gamma band oscillations which reflect desynchronization of pyramidal cell activity. This process partially depends on GABAA receptor-mediated neurotransmission [127]. More recent data also indicates possible GABA_A dysfunction in high-risk and first-episode patients [128]. Using transcranial magnetic stimulation, this group found evidence of reduced GABA_A-mediated inhibition in these individuals (n=18) compared with healthy subjects (n=18). In addition, a recent magnetic resonance spectroscopy (MRS) study (n=54) observed 30 % elevations in glutamate-glutamine and GABA levels in the medial PFC of unmedicated patients compared with controls (p < 0.03) [129]. A positive, proof-of-concept, placebocontrolled trial (n=15) with MK-0777 (a relatively selective agonist of GABA_A receptors containing α_2/α_3 subunits) led to a larger, 4-week, placebo-controlled trial (n=60) [128, 130]. Even though the results of the second study were negative, the investigators opined that a more potent, partial agonist with greater intrinsic activity at this site might improve cognition.

Acetylcholine

The cholinergic system is implicated in sensory processing and cognition, modulated through both its nicotinic ionotropic and muscarinic metabotropic receptors. In this context, the use of perinatal choline appeared to enhance development of cerebral inhibition which is deficient in patients with schizophrenia and is related to poor sensory gating and attention [131]. This controlled trial compared dietary phosphatidylcholine (6,300 mg/day) supplementation with placebo starting in the second trimester. The results indicated choline had no adverse effects and produced a greater rate of intact P50 sensory gating at the 5th postnatal week (i.e., 76 % versus 43 %) but this difference was gone by the 13th week. Results from trials like this raise the possibility that dietary supplementation during pregnancy may decrease the long-term risk of schizophrenia. In multiepisode patients, the potential role of adjunctive acetylcholinesterase inhibitors (e.g., rivastigmine, donepezil, galantamine) to enhance cognition and improve negative symptoms in schizophrenia is not clearly established in adequately powered RCTs but remains a possible approach. One review and meta-analysis (including six open-label and 13 double-blind trials) concluded that the existing data suggested improvement in memory, and the motor speed and attention part of executive function, with these agents [132].

Heavy smoking patterns often seen in patients with schizophrenia may stimulate *nicotinic receptors*, temporarily improving sensory processing. Further, the α 7 nicotine ACh receptor gene (CHRNA7) is associated with schizophrenia, as well as related cognitive and sensory gating problems [133]. Agonists at this receptor may improve abnormal suppression of the P50 auditory evoked potential which is associated with certain cognitive deficits and perceptual impairments [134, 135]. In a controlled trial (n=40), nonsmoking patients with abnormal P50 rates (i.e., >0.5) were stabilized on risperidone (3–6 mg/day) and then randomized to *tropisetron* (an α 7 partial agonist and 5-HT₃ antagonist) or placebo [136]. All three tropisetron doses (5, 10, or 20 mg/day) improved P50 deficits after 10 days and this correlated with improvement in cognition. DMXB-A is an α 7 partial agonist which may work by improving the default network function in schizophrenia [137]. A 4-week, proof-of-concept, controlled trial (n=31)found that adjunctive DMXB-A (150 mg BID) was significantly better than placebo (p < 0.01) for negative symptoms (but not cognition) and generally well tolerated [138]. A more recent trial found that DMXB-A plus risperidone was superior to olanzapine monotherapy for improvement in neurocognitive function [139]. A controlled, exploratory trial (n=185) found that the addition of TC-5619 (another α 7 partial agonist) to quetiapine or risperidone was superior to placebo (p < 0.03) in terms of cognitive and negative symptom improvements [140]. Early signals for potential cognitive improvement and management of concurrent alcohol and nicotine dependence with adjunctive varenicline (a partial α_4 - β_2 and full α 7 agonist) must be balanced against its adverse effects and reports of behavioral toxicity [141-143]. Of note, galantamine has a more complex impact on the cholinergic system beyond anticholinesterase inhibition (i.e., also modulates α_7 and α_4 - β_2 receptors) [144].

Antagonism of *muscarinic receptors* (M_1-M_5) may disrupt cognitive function. A controlled, crossover trial (n = 20) found that monotherapy with xanomeline (75–225 mg/day) (a relatively selective M_1/M_4 cholinergic, orthosteric, receptor agonist) was significantly more effective than placebo (p < 0.05) on various measures including improvement in verbal learning and short-term memory [145]. Gastrointestinal effects were common with xanomeline, but did not lead to discontinuation from the trial. Studies utilizing selective allosteric activators of the M_1 or M_4 receptors are needed to clarify if this potential benefit is due to actions at either or both receptor sites [146–148]. Whether activation of M_2 , M_3 , or M_5 receptors might also be efficacious is not clear at this time.

Histamine

The activity at four identified histamine receptor subtypes contributes to the modulation of arousal and cognitive function. Thus, activity at autoreceptors may increase alertness and activity at heteroreceptors may modulate levels of neurotransmitters such as DA, NE, 5-HT, Glu, GABA, and Ach [149, 150]. Preliminary preclinical and clinical evidence support a therapeutic potential with histamine (e.g., H₃ receptor antagonists

and inverse agonists) for cognitive deficits associated with schizophrenia [151]. As an example, pitolisant, an H₃ receptor inverse agonist, demonstrated potential procognitive activity. Further, a pharmacokinetic study (n=129) in Chinese patients with schizophrenia indicated that histamine H₃ receptor polymorphisms may aid in predicting the efficacy of risperidone [152]. More recently, the H₂ receptor antagonist, famotidine, was reported to be superior to placebo based on the change in PANSS total, general psychopathology subscale, and CGI scores ($p \le 0.05$) [153].

Other Approaches

Antidepressants

Antidepressants are often used to manage depression in schizophrenia. They may, however, also benefit other psychopathological dimensions. For example, a 6-week, controlled trial (n=54) found amoxapine (150-250 mg/day) (an agent with both 5-HT₂ and D₂ receptor antagonism properties) was as effective as haloperidol (5-15 mg/day) in improving PANSS total and positive symptom subscale scores in patients with schizophrenia [154]. An earlier meta-analysis considering only SSRIs as add-on therapy to treat negative symptoms concluded there was no global support for benefit, but subsequent meta-analyses which included SSRIs plus other antidepressants found support for this approach [155, 156]. For example, a large, 12-week, controlled trial (n = 198) found that adjunctive citalopram (20 mg/day) was superior to place bo in improving depression (p < 0.002), negative symptoms (p < 0.05), mental functioning (p < 0.0001), and quality of life (p < 0.05) in middle-aged or older (i.e., 40 years of age) patients with schizophrenia or schizoaffective disorder [157]. In a 12-week, controlled trial (n=60), oral adjunctive selegiline (5 mg BID) was superior to placebo in improving negative symptoms (p < 0.009), Brief Psychiatric Rating Scale (BPRS) total scores (p < 0.02), and CGI-S scores (p < 0.002) [158]. A 16-week, controlled trial (n=33) reported that duloxetine (60 mg/day) augmentation of CLZ was superior to placebo in benefiting the PANSS negative (p < 0.004), general psychopathology (p < 0.001), and total scores (p < 0.001) [159]. The best evidence to date supports a potential role for *mirtazapine* to improve negative, cognitive, and depressive symptoms [160–164]. For example, a recent review found that five of six randomized trials supported the use of mirtazapine in treating negative symptoms of schizophrenia [160–164].

Glucocorticoids/Neurosteroids

The hypothalamic–pituitary axis (HPA) may play an important role in the pathophysiology of and stress vulnerability associated with schizophrenia. In this context, *mifepristone* (a glucocortoid receptor antagonist) has shown initial promise for improving cognition in schizophrenia [165]. *Pregnenolone* and *dehydroepiandrosterone* (DHEA) are neurosteroids whose various actions (e.g., neuroprotection, stress response, mood regulation, cognitive performance) predict the potential to improve symptoms of schizophrenia [166]. Promising results from preliminary trials demonstrated benefit as add-on therapies for positive, negative, and certain cognitive symptoms, as well as for relief of extrapyramidal side effects (EPS) [167]. For example, pregnenolone (30 mg/day) (but not DHEA) produced significant decreases in positive symptoms (p < 0.01); improved attention and working memory (p < 0.04); and reduced EPS in an 8-week, double-blind, placebo-controlled trial (n = 58) involving patients with chronic schizophrenia [168].

Anti-Inflammatory Agents

Inflammatory processes (e.g., abnormal levels of cytokines, autoantibodies, and lymphocytes in serum and CSF) are implicated in the pathophysiology of schizophrenia [169, 170]. A recent meta-analysis concluded that add-on, *nonsteroidal*, *anti-inflammatory drugs* (NSAIDs) such as *celecoxib* could reduce symptom severity, and noted a potential added benefit for *aspirin* to reduce the increased cardiac and cancer mortality in this population [171–173].

Minocycline is the second-generation tetracycline with anti-inflammatory, antimicrobial, and possible neuroprotective properties through modulation of Gluinduced excitotoxicity. Results of a 6-month, longitudinal, placebo-controlled trial (n=54) demonstrated an augmenting effect when minocycline (200 mg/day) was added to an antipsychotic, improving negative symptoms (p<0.01) and cognitive functioning (p<0.01) in early-onset schizophrenia [174]. A more recent study (n=94) found that minocycline plus TAU was significantly (p<0.001) more effective than TAU alone in improving negative symptoms in an early-onset population [175]. These results support earlier case reports and open-label trial data.

Cannabinoids

Central cannabinoid receptors are among the most abundant G protein-coupled receptors in the CNS, appear relevant to working memory, modulate both GABA and Glu release and are targets for delta-9-tetrahydrocannabinol (THC), the primary psychoactive substance in cannabis [176]. Stimulation of central cannabinoid receptors (e.g., CB₁) can produce transient psychotic symptoms, exacerbate preexisting psychotic disorders, and with early exposure may increase the risk of developing schizophrenia in vulnerable individuals [177]. Preliminary data suggest a positive cognitive effect (e.g., working memory) with CB₁ antagonists, through modulation of PFC GABA release [178].

Hormonal Agents

Decreases of *estrogen* during the postpartum and perimenopausal periods appear to increase the risk for a first episode or relapse of schizophrenia. Conversely, increases

in estrogen during pregnancy and certain phases of the menstrual cycle can ameliorate psychosis and reduce the risk of relapse [179]. The "estrogen protective hypothesis" is supported by preliminary results from controlled trials with add-on estradiol and possibly the selective estrogen modulator, raloxifene [180, 181]. A more recent, 4-week study (n=32) involving women of childbearing age with chronic schizophrenia compared conjugated estrogen (0.625 mg/day) with placebo to augment standard antipsychotics [182]. In contrast to placebo, those receiving estrogen experienced a significant decrease in positive (p<0.003), negative (p<0.001), general (p<0.001), and total PANSS (p<0.001) scores.

Oxytocin functions as both a hormone and peptide neurotransmitter and is linked to both social and reproductive behaviors [183, 184]. There is data from controlled trials indicating a potential role for oxytocin nasal spray as an adjunct to antipsychotics [185–187]. For example, an 8-week, controlled study (n=40) found oxytocin nasal spray (20–40 IU BID) superior to placebo based on change in the PANSS positive (p<0.001), negative (p<0.01), general psychopathology (p<0.02), and total (p<0.001) scores in patients with schizophrenia on a stable dose of risperidone [185]. More recent preliminary reports indicate oxytocin (40 IU) may also facilitate social cognition training (e.g., improve empathic accuracy) [188, 189].

Neutraceuticals

Omega-3 Fatty Acids

Given their role in neurodevelopment, neurodegeneration, and behavioral neurobiology, long-chain omega-3 PUFAs may play a role in schizophrenia. In a controlled trial (n=81), individuals at ultra-high risk of transitioning to a psychotic disorder received omega-3 PUFA (1.2 g/day) or placebo for 12 weeks and were then followed for an additional 40-weeks [32]. At 12 months, 2/41 (4.9 %) who initially received the active treatment had transitioned to a psychotic disorder versus 11/40 (27.5 %) who had initially received placebo (p<0.07). Further, the active treatment group experienced fewer positive symptoms (p<0.01), negative symptoms (p<0.02), and general symptoms (p<0.002) compared with placebo. By contrast, a subsequent meta-analysis of controlled trials considering omega-3, specifically eicosapentaenoic acid (EPA), found no benefit when used as an augmentation in patients with multiepisode schizophrenia [190].

Folate

Folate and other micronutrient deficiencies during fetal development may be risk factors for schizophrenia [191]. A recent, 16-week, controlled trial (n=140) compared folic acid (2 mg/day) plus vitamin B12 (400 µg/day) to placebo in patients with chronic schizophrenia [192]. Active treatment was superior to placebo in improving negative symptoms when genotype was taken into account but not when

excluded (p < 0.02). The authors concluded that folate plus B12 may improve negative symptoms, but the results were influenced by genetic variation in folate absorption. There is also data indicating that genetic and epigenetic factors which influence folic acid metabolism are associated with an increased risk of metabolic syndrome in patients taking SGAs [193].

Vitamin D

Nutritional deprivation increases the risk of schizophrenia [194]. Further, low vitamin D levels affect neuronal differentiation, axonal connectivity, dopamine ontogeny, and brain structure and function [195]. In this context, low prenatal concentrations of vitamin D are associated with neurobiological and neuropsychiatric disorders, such as schizophrenia [196]. Following up on the hypothesis that developmental vitamin D deficiency may contribute to the risk for schizophrenia, dried neonatal blood samples of vitamin D levels were compared between 424 individuals with schizophrenia and 424 matched controls [197]. Surprisingly, both low and high concentrations were associated with an increased risk. Such findings raise the possibility that vitamin D supplementation may prevent schizophrenia in some populations. This is supported by a Finnish birth cohort study (n=9,114) which observed a reduced risk of schizophrenia by age 31 years in males with vitamin D supplementation provided during the first year of life [198].

Cognitive and Psychosocial Therapies

Most studies of psychosocial and psychotherapeutic interventions for schizophrenia focus on the phases after acute symptoms are reasonably well controlled. In this context, there is good evidence that augmentation with cognitive behavioral therapy (CBT) may be as effective as polypharmacy in managing persistent positive symptoms [199]. There are, however, some data on their use to augment acute antipsychotic drug therapy, focusing primarily on behavioral and educational interventions. In one study (n=315), the use of adjunctive CBT to manage acute symptoms demonstrated that a 5-week course had transient benefits over treatment as usual (TAU) or supportive counseling in patients during their first or second hospital admission [200]. Further, at 18 months follow-up, both CBT and supportive counseling produced significantly greater symptom reduction versus TAU (p < 0.005) [201]. A second report (n=100) included patients who were experiencing an acute psychotic exacerbation which required hospitalization [202]. These patients were randomly assigned to a TAU plus CBT (up to 25 sessions) or a TAU control group. In the TAU plus CBT group, 60 % experienced a significantly better resolution of symptoms and improvement in social function over 12 months compared with 40 % in the TAU group (p < 0.04). At the 2-year follow-up, the TAU plus CBT continued to demonstrate significantly better improvement in negative symptoms (p < 0.004) and social functioning (p < 0.009) with no difference in cost [203]. A more recent study (n = 18)

Clinical Presentation

```
Strategy
```



Fig. 7.1 Strategy for treatment-resistant multiepisode schizophrenia

in hospitalized patients with schizophrenia found that both group CBT (16 sessions) and group psychoeducation (8 sessions) significantly improved quality of life measures after acute treatment and at 6 months follow-up [204]. Finally, adherence therapy (AT) utilizes motivational and CBT techniques to modify patient's beliefs about treatment to improve medication adherence [205]. In a European trial (n=161) comparing AT versus TAU, AT was superior to TAU in improving PANSS total scores 12 weeks after discharge from the hospital (p<0.05).

Clinical Strategy with Other Approaches

While there is insufficient data to support an FDA-approved indication for the novel approaches discussed, their relative safety and tolerability, ready availability and positive, preliminary, controlled trial data suggest they may benefit some patients. Figure 7.1 outlines a clinical strategy which incorporates some of these options.

Conclusion

The introduction of antipsychotic drug therapy for schizophrenia and related psychotic disorders has improved the lives of many patients. Some argue, however, that there has been little progress since the advent of CPZ and CLZ. As a result, the focus is increasingly on detection of high-risk individuals and utilization of appropriate treatments early in the course of illness. After a first episode of psychosis and during the critical period characterized as early-onset, strategies to improve efficacy, tolerability, and adherence are also important to prevent the deterioration seen during this phase. Initiatives involving genetics, neuroimaging, early identification, and psychosocial interventions to control acute symptoms, sustain remission, and ultimately achieve recovery are now being actively pursued. These topics are discussed elsewhere in this book and will hopefully lead to more specific, more effective, and better tolerated therapies.

References

- Janicak PG, Marder S, Pavuluri M. Principles and practice of psychopharmacotherapy. 5th ed. Philadelphia, PA: Lippincott Williams & Wilkins; 2011. p. 65–180.
- Lieberman JA, Stroup TS, McEvoy JP, Swartz MS, Rosenheck RA, Perkins DO, et al. Effectiveness of antipsychotic drugs in patients with chronic schizophrenia. N Engl J Med. 2005;353(12):1209–23.
- Jones PB, Barnes TR, Davies L, Dunn G, Lloyd H, Hayhurst KP, et al. Randomized controlled trial of the effect on quality of life of second- vs first-generation antipsychotic drugs in schizophrenia: cost utility of the latest antipsychotic drugs in schizophrenia study (CUtLASS 1). Arch Gen Psychiatry. 2006;63(10):1079–87.
- Buchanan RW, Kreyenbuhl J, Kelly DL, Noel JM, Boggs DL, Fischer BA, et al. The 2009 schizophrenia PORT psychopharmacological treatment recommendations and summary statements. Schizophr Bull. 2010;36(1):71–93.
- Wang L, McLeod HL, Weinshilbourn RM. Genomics and drug response. N Engl J Med. 2011;364(12):1144–53.
- 6. Malhotra AK. The state of pharmacogenetics: customizing treatments. Psychiatr Times. 2010;27(4):38–41.
- 7. Xu M, Xing Q, Li S, Zheng Y, Wu S, Gao R, et al. Pharmacogenetic effects of dopamine transporter gene polymorphisms on response to chlorpromazine and clozapine and on extra-

pyramidal syndrome in schizophrenia. Prog Neuropsychopharmacol Biol Psychiatry. 2010;34(6):1026–32.

- Kohlrausch FB, Salatino-Oliveira A, Gama CS, Lobato MI, Belmonte-de-Abreu P, Hutz MH. Influence of serotonin transporter gene polymorphisms on clozapine response in Brazilian schizophrenics. J Psychiatr Res. 2010;44(16):1158–62.
- Lencz T, Morgan TV, Athanasiou M, Dain B, Reed CR, Kane JM, et al. Converging evidence for a pseudoautosomal cytokine receptor gene locus in schizophrenia. Mol Psychiatry. 2007;12:572–80.
- Lavedan C, Licamele L, Volpi S, Hamilton J, Heaton C, Mack K, et al. Association of the NPAS3 gene and five other loci with response to the antipsychotic iloperidone identified in a whole genome association study. Mol Psychiatry. 2009;14(8):804–19.
- Fijal BA, Stauffer VL, Kinon BJ, et al. Analysis of gene variants previously associated with iloperidone response in patients with schizophrenia who are treated with risperidone. J Clin Psychiatry. 2012;73(3):367–71.
- Leucht S, Komossa K, Rummel-Kluge C, Corves C, Hunger H, Schmid F, et al. A metaanalysis of head-to-head comparisons of second-generation antipsychotics in the treatment of schizophrenia. Am J Psychiatry. 2009;166:152–63.
- Yilmaz Z, Zai CC, Hwang R, Mann S, Arenovich T, Remington G, et al. Antipsychotics, dopamine D₂ receptor occupancy and clinical improvement in schizophrenia: a meta-analysis. Schizophr Res. 2012;140(1–3):214–20.
- Janicak PG, Javaid JI, Leach A, Sharma RP, Dowd S, Davis JM. A two phase double-blind randomized study of three haloperidol plasma levels for acute psychosis with reassignment of initial non-responders. Acta Psychiatr Scand. 1997;95:343–50.
- Remington G, Agid O, Foussias G, Ferguson L, McDonald K, Powell V. Clozapine and therapeutic drug monitoring: is there sufficient evidence for an upper threshold? Psychopharmacology (Berl). 2013;225(3):505–18.
- 16. Lopez LV, Kane JM. Plasma levels of second-generation antipsychotics and clinical response in acute psychosis: a review of the literature. Schizophr Res. 2013;147(2–3):368–74.
- Sparshatt A, Taylor D, Patel MX, Kapur S. A systematic review of aripiprazole—dose, plasma concentration, receptor occupancy, and response: implications for therapeutic drug monitoring. J Clin Psychiatry. 2010;71(11):1447–56.
- Nazirizadeh Y, Vogel F, Bader W, et al. Serum concentrations of paliperidone versus risperidone and clinical effects. Eur J Clin Pharmacol. 2010;66(8):797–803.
- Bishara D, Olofinjana O, Sparshatt A, Kapur S, Taylor D, Patel MX. Olanzapine: a systematic review and meta-regression of the relationships between dose, plasma concentration, receptor occupancy, and response. J Clin Psychopharmacol. 2013;33:329–55.
- Cannon TD, Cadenhead K, Cornblatt B, Woods SW, Addington J, Walker E, et al. Prediction of psychosis in youth at high clinical risk: a multisite longitudinal study in North America. Arch Gen Psychiatry. 2008;65(1):25–7.
- McGorry PD, Yung AR, Phillips LJ, Yuen HP, Francey S, Cosgrave EM, et al. Randomized controlled trial of interventions designed to reduce the risk of progression to first-episode psychosis in a clinical sample with subthreshold symptoms. Arch Gen Psychiatry. 2002;59:921–8.
- Fusar-Poli P, Borgwardt S, Bechdolf A, Addington J, Riecher-Rossler A, Schultze-Lutter F, et al. The psychosis high-risk state: a comprehensive state-of-the-art review. JAMA Psychiatry. 2013;70(1):107–20.
- Zammit S, Kounali D, Cannon M, et al. Psychotic experiences and psychotic disorders at age 18 in relation to psychotic experiences at age 12 in a longitudinal population-based cohort study. Am J Psychiatry. 2013;170:742–50.
- Yung AR, Stanford C, Cosgrave E, Killackey E, Phillips L, Nelson B, et al. Testing the ultra high risk (prodromal) criteria for the prediction of psychosis in a clinical sample of young people. Schizophr Res. 2006;84(1):57–66.
- 25. Ruhrmann S, Schulte-Lutter F, Salokangas RK, Heinimaa M, Linszen D, Dingemans P, et al. Prediction of psychosis in adolescents and young adults at high risk: results from the prospective European prediction of psychosis study. Arch Gen Psychiatry. 2010;67(3):241–51.

- 26. Seidman LJ, Giuliano AJ, Meyer EC, Addington J, Cadenhead KS, Cannon TD, et al. Neuropsychology of the prodrome to psychosis in the NAPLS consortium: relationship to family history and conversion to psychosis. Arch Gen Psychiatry. 2010;67(6):578–88.
- Egerton A, Chaddock CA, Winton-Brown TT, et al. Presynaptic striatal dopamine dysfunction in people at ultra-high risk for psychosis: findings in a second cohort. Biol Psychiatry. 2013;74:106–12.
- Miller TJ, McGlashan TH, Rosen JL, Cadenhead K, Cannon T, Ventura J, et al. Prodromal assessment with the structured interview for prodromal syndromes and the scale of prodromal symptoms: predictive validity, interrater reliability, and training to reliability. Schizophr Bull. 2003;29(4):703–15.
- Tandon R. Schizophrenia spectrum and other psychotic disorders: DSM-5 revisions and their clinical implications. Psychopharm Rev. 2013;48(5):33–40.
- Preti A, Cella M. Randomized-controlled trials in people at ultra high risk of psychosis; a review of treatment effectiveness. Schizophr Res. 2010;123(1):30–6.
- McGorry PD, Nelson B, Amminger GP, Bechdolf A, Francey SM, Berger G, et al. Intervention in individuals at ultra-high risk for psychosis: a review and future directions. J Clin Psychiatry. 2009;70(9):1206–12.
- 32. Amminger GP, Schafer MR, Papageorgiu K, Klier CM, Cotton SM, Harrigan SM, et al. Long-chain omega-3 fatty acids for indicated prevention of psychotic disorders: a randomized, placebo-controlled trial. Arch Gen Psychiatry. 2010;67(2):146–54.
- Asevedo E, Cunha GR, Zugman A, Mansur RB, Brietzke E. N-acetyl cysteine as a potentially useful medication to prevent conversion to schizophrenia in at-risk individuals. Rev Neurosci. 2012;23(4):353–62.
- Dean O, Giorlando F, Berk M. N-acetyl cysteine in psychiatry: current therapeutic evidence and potential mechanisms of action. J Psychiatry Neurosci. 2011;36(2):78–86.
- 35. Liu CC, Chien YL, Hsieh MH, Hwang TJ, Hwu HG, Liu CM. Aripiprazole for drug-naïve or antipsychotic-short-exposure subjects with ultra-high risk state and first-episode psychosis: an open-label study. J Clin Psychopharmacol. 2013;33:18–23.
- McGorry PD, Nelson B, Phillips LJ, et al. Randomized controlled trial of interventions for young people at ultra-high risk of psychosis: twelve-month outcome. J Clin Psychiatry. 2013;74(4):349–56.
- Perkins DO, Gu H, Boteva K, Lieberman JA. Relationship between duration of untreated psychosis and outcome in first-episode schizophrenia: a critical review and meta-analysis. Am J Psychiatry. 2005;162:1785–804.
- Boonstra N, Klaassen R, Sytema S, Marshall M, De Haan L, Wunderink L, et al. Duration of untreated psychosis and negative symptoms—a systematic review and meta-analysis of individual patient data. Schizophr Res. 2012;142(1–3):12–9.
- Marshall M, Lewis S, Lockwood A, Drake R, Jones P, Croudace T. Association between duration of untreated psychosis and outcome in cohorts of first-episode patients: a systematic review. Arch Gen Psychiatry. 2005;62:975–83.
- 40. Perkins DO. Review: longer duration of untreated psychosis is associated with worse outcome in people with first episode psychosis. Evid Based Ment Health. 2006;9(2):36.
- 41. Cuesta MJ, Garcia de Jalon E, Campos MS, Ibanez B, Sanchez-Torres AM, Peralata V. Duration of untreated negative and positive symptoms of psychosis and cognitive impairment in first episode psychosis. Schizophr Res. 2012;141(2–3):222–7.
- 42. Andreasen NC, Nopoulos P, Magnotta V, Pierson R, Ziebell S, Ho BC. Progressive brain change in schizophrenia: a prospective longitudinal study of first-episode schizophrenia. Biol Psychiatry. 2011;70:672–9.
- Lewis DA. Antipsychotic medications and brain volume: do we have cause for concern? Arch Gen Psychiatry. 2011;68(2):126–7.
- 44. Ho B-C, Andreason NC, Ziebell S, Pierson R, Magnotta V. Long-term antipsychotic treatment and brain volumes: a longitudinal study of first-episode schizophrenia. Arch Gen Psychiatry. 2011;68(2):128–37.

- 45. Bertelsen M, Jeppesen P, Peterson L, Thorup A, Øhlenschlaeger J, le Quach P, et al. Five-year follow-up of a randomized multicenter trial of intensive early intervention vs standard treatment for patients with a first episode of psychotic illness: the OPUS trial. Arch Gen Psychiatry. 2008;65(7):762–71.
- 46. Melle I, Larsen TK, Haahr U, Friis S, Johannessen JO, Ojordsmoen S, et al. Reducing the duration of untreated first-episode psychosis: effects on clinical presentation. Arch Gen Psychiatry. 2004;61:143–50.
- 47. Melle I, Johannesen JO, Friis S, Haahr U, Joa I, Larsen TK, et al. Early detection of the first episode of schizophrenia and suicidal behavior. Am J Psychiatry. 2006;163:800–4.
- 48. ten Velden Hegelstad W, Larsen TK, Auestad B, Evensen J, Haahr U, Joa I, et al. Long-term follow-up of the TIPS early detection in psychosis study: effects on 10-year outcome. Am J Psychiatry. 2012;169:374–80.
- 49. Wunderink L, Nierboer RM, Wiersma D, Systema S, Nienhuis FJ. Recovery in remitted first-episode psychosis at 7 years of follow-up of an early dose reduction/discontinuation or maintenance treatment strategy: long-term follow-up of a 2-year randomized clinical trial. JAMA Psychiatry. 2013;70(9):913–20.
- Lieberman JA, Tollefson G, Tohen M, Green AI, Gur RE, Kahn R, et al. Comparative efficacy and safety of atypical and conventional antipsychotic drugs in first-episode psychosis: a randomized, double-blind trial of olanzapine versus haloperidol. Am J Psychiatry. 2003; 160:1396–404.
- 51. Lieberman JA, Phillips M, Gu H, Stroup S, Zhang P, Kong L, et al. Atypical and conventional antipsychotic drugs in treatment-naïve first-episode schizophrenia: a 52-week randomized trial of clozapine vs. chlorpromazine. Neuropsychopharmacology. 2003;28:995–1003.
- Girgis RR, Philips MR, Xiaodong L, Kejin L, Jiang H, Wu C, et al. Clozapine v. chlorpromazine in treatment-naive, first-episode schizophrenia: 9-year outcomes of a randomised clinical trial. Br J Psychiatry. 2011;199:281–8.
- 53. Kumra S, Kranzler H, Gerbino-Rosen G, Kester HM, DeThomas C, Kafantaris V, et al. Clozapine and "high-dose" olanzapine in refractory early-onset schizophrenia: a 12-week randomized and double-blind comparison. Biol Psychiatry. 2008;63:524–9.
- Harvey PD, Rabinowitz J, Eerdekens M, Davidson M. Treatment of cognitive impairment in early psychosis: a comparison of risperidone and haloperidol in a large long-term trial. Am J Psychiatry. 2005;162:1888–95.
- Schooler N, Rabinowitz J, Davidson M, Emsley R, Harvey PD, Kopala L, et al. Risperidone and haloperidol in first-episode psychosis: a long-term randomized trial. Am J Psychiatry. 2005;162:947–53.
- 56. Crespo-Facorro B, Rodriguez-Sanchez J, Perez-Iglesias R, Mata I, Ayesa R, Ramirez-Bonilla M, et al. Neurocognitive effectiveness of haloperidol, risperidone, and olanzapine in first-episode psychosis: a randomized, controlled one-year follow-up comparison. J Clin Psychiatry. 2009;70(5):717–29.
- 57. McEvoy JP, Lieberman JA, Perkins DO, Hamer RM, Gu H, Lazarus A, et al. Efficacy and tolerability of olanzapine, quetiapine, and risperidone in the treatment of early psychosis: a randomized, double-blind 52-week comparison. Am J Psychiatry. 2007;164:1050–60.
- Keefe RSE, Sweeney JA, Gu H, Gu H, Hamer RM, Perkins DO, et al. Effects on olanzapine, quetiapine, and risperidone on neurocognitive function in early psychosis: a randomized, double-blind 52-week comparison. Am J Psychiatry. 2007;164:1061–71.
- 59. Sikich L, Frazier JA, McClellan J, Findling RL, Vitiello B, Ritz L, et al. Double-blind comparison of first- and second-generation antipsychotics in early-onset schizophrenia and schizoaffective disorder: Findings from the treatment of early-onset schizophrenia spectrum disorders (TEOSS) study. Am J Psychiatry. 2008;165:1420–31.
- 60. Kahn R, Fleischhacker WW, Boter H, Davidson M, Vergouwe Y, Keet IP, et al. Effectiveness of antipsychotic drugs in first episode schizophrenia and schizophreniform disorder: an open randomised clinical trial. Lancet. 2008;371(9618):1085–97.
- Fleischhacker WW, Uchida H. Critical review of antipsychotic polypharmacy in the treatment of schizophrenia. Int J Neuropsychopharmacol. 2012;2:1–11.

- 7 Acute Management of Schizophrenia
 - 62. Crossley NA, Constante M, McGuire P, Power P. Efficacy of atypical v. typical antipsychotic in the treatment of early psychosis: meta-analysis. Br J Psychiatry. 2010;196:434–9.
 - 63. Perkins DO, Gu H, Weiden PJ, McEvoy JP, Hamer RM, Lieberman JA, et al. Predictors of treatment discontinuation and medication nonadherence in patients recovering from a first episode of schizophrenia, schizophreniform disorder, or schizoaffective disorder: a randomized, double-blind, flexible-dose, multicenter study. J Clin Psychiatry. 2008;69(1):106–13.
 - Emsley R, Rabinowitz J, Medori R, Early Psychosis Global Working Group. Remission in early psychosis: rates, predictors, and clinical and functional outcome correlates. Schizophr Res. 2007;89(1–3):129–39.
 - 65. Allen MH, Feifel D, Lesem MD, Zimbroff DL, Ross R, Munzar P, et al. Efficacy and safety of loxapine for inhalation in the treatment of agitation in patients with schizophrenia: a randomized, double-blind, placebo-controlled trial. J Clin Psychiatry. 2011;72(10):1313–21.
 - 66. Lesem MD, Tran-Johnson TK, Riesenberg RA, et al. Rapid acute treatment of agitation in individuals with schizophrenia: multicentre, randomised, placebo-controlled study of inhaled loxapine. Br J Psychiatry. 2011;198(1):51–8.
 - Tiihonen J, Suokas JT, Suvisaari JM, Haukka J, Korhonen P. Polypharmacy with antipsychotics, antidepressants, or benzodiazepines and mortality in schizophrenia. Arch Gen Psychiatry. 2012;69(5):476–83.
 - Fakra E, Azorin JM. Clozapine for the treatment of schizophrenia. Expert Opin Pharmacother. 2012;13(13):1923–35.
 - 69. Liffick E, Breier A. Pharmacotherapy of first-episode schizophrenia. Psychopharm Rev. 2010;45(6):41-8.
 - 70. Kirkpatrick B. Understanding the physiology of schizophrenia. J Clin Psychiatry. 2013;74(3):e05.
 - Miyamoto S, Miyake N, Jarskog LF, Fleischhacker WW, Lieberman JA. Pharmacological treatment of schizophrenia: a critical review of the pharmacology and clinical effects of current and future therapeutic agents. Mol Psychiatry. 2012;17(12):1206–27.
 - Lochmann van Bennekom MW, Gijsman HJ, Zitman FG. Antipsychotic polypharmacy in psychotic disorders: a critical review of neurobiology, efficacy, tolerability and cost effectiveness. J Psychopharmacol. 2013;27(4):327–36.
 - Cipriani A, Boso M, Barbui C. Clozapine combined with different antipsychotic drugs for treatment resistant schizophrenia. Cochrane Database Syst Rev. 2009 Jul 8;(3):CD006324.
 - 74. Barbui C, Accordini S, Nose M, Stroup S, Purgato M, Girlanda F, et al. Aripiprazole versus haloperidol in combination with clozapine for treatment-resistant schizophrenia in routine clinical care: a randomized, controlled trial. J Clin Pharmacol. 2011;31(3):266–73.
 - Correll CU, Rummel-Kluge C, Corves C, Kane JM, Leucht S. Antipsychotic combinations vs. monotherapy in schizophrenia: a meta-analysis of randomized controlled trials. Schizophr Bull. 2009;35(2):443–57.
 - Goldman-Rakic PS, Castner SA, Svensson TH, Siever LJ, Williams GV. Targeting the dopamine D₁ receptor in schizophrenia: insights for cognitive dysfunction. Psychopharmacology (Berl). 2004;174(1):3–16.
 - Snitz BE, MacDonald III A, Cohen JD, Cho RY, Becker T, Carter CS. Lateral and medial hypofrontality in first-episode schizophrenia: functional activity in a medication-naïve state and effects of short-term atypical antipsychotic treatment. Am J Psychiatry. 2005;162(12): 2322–9.
 - Mu Q, Johnson K, Morgan PS, Grenesko EL, Molnar CE, Anderson B, et al. A single 20 mg dose of the full D1 dopamine agonist dihydrexidine (DAR-0100) increases prefrontal perfusion in schizophrenia. Schizophr Res. 2007;94(1–3):333–41.
 - 79. George MS, Molnar CE, Grenesko EL, et al. A single 20 mg dose of dihydrexidine (DAR-0100), a full dopamine D1 agonist, is safe and tolerated in patients with schizophrenia. Schizophr Res. 2007;93(1–3):42–50.
 - Siever LJ, Zaluda LC, McClure MM, et al. Clinical testing of a D1 agonist for cognitive enhancement in the schizophrenia spectrum. Presented at: 14th International Congress on Schizophrenia Research, April 21–25, 2013; Grand Lakes, FL.

- Jardemark K, Marcus MM, Shahid M, Svensson TH. Effects of asenapine on prefrontal *N*-methyl-D-aspartate receptor-mediated transmission: involvement of dopamine D₁ receptors. Synapse. 2010;64(11):870–4.
- Lai C, Mo GH, Chen ML, Wang YC, Chen JY, Liao DL. Analysis of genetic variations in the dopamine D1 receptor (DRD1) gene and antipsychotics-induced tardive dyskinesia in schizophrenia. Eur J Pharmacol. 2011;67(4):383–4.
- Hwang R, Zai C, Tiwari A, Muller DJ, Arranz MJ, Morris AG, et al. Effect of dopamine D3 receptor gene polymorphisms and clozapine treatment response: exploratory analysis of nine polymorphisms and meta-analysis of the Ser9Gly variant. Pharmacogenomics J. 2010; 10(3):200–18.
- Jardemark K, Wadenberg ML, Grillner P, Svensson TH. Dopamine D3 and D4 receptor antagonists in the treatment of schizophrenia. Curr Opin Investig Drugs. 2002;3(1):101–5.
- Grunder G. Cariprazine, an orally active D₂/D₃ receptor antagonist, for the potential treatment of schizophrenia, bipolar mania and depression. Curr Opin Investig Drugs. 2010;11(7):823–32.
- 86. Gross G, Drescher K. The role of dopamine D(3) receptors in antipsychotic activity and cognitive functions. Handb Exp Pharmacol. 2012;213:167–210.
- 87. Kelleher JP, Centorrino F, Huxley NA, et al. Pilot randomized, controlled trial of pramipexole to augment antipsychotic treatment. Eur Neuropsychopharmacol. 2012;22:415–8.
- Gross G, Wicke K, Drescher KU. Dopamine D₃ receptor antagonism—still a therapeutic option for the treatment of schizophrenia. Naunyn Schmiedebergs Arch Pharmacol. 2013; 386(2):155–66.
- Yamamoto K, Hornykiewicz O. Proposal for a noradrenaline hypothesis of schizophrenia. Prog NeuroPsychopharmacol Biol Psychiatry. 2004;28:913–22.
- Masana M, Bortolozzi A, Artigas F. Selective enhancement of mesocortical dopaminergic transmission by noradrenergic drugs: therapeutic opportunities in schizophrenia. Int J Neuropsychopharmacol. 2011;14(1):53–68.
- Kelly DL, Buchanan RW, Boggs DL, McMahon RP, Dickenson D, Nelson M, et al. A randomized double-blind trial of atomoxetine for cognitive impairments in 32 people with schizophrenia. J Clin Psychiatry. 2009;70(4):518–25.
- 92. Poyurovsky M, Faragian S, Fuchs C, Pashinian A. Effect of the selective norepinephrine reuptake inhibitor reboxetine on cognitive dysfunction in schizophrenia patients: add-on, double-blind placebo-controlled study. Isr J Psychiatry Relat Sci. 2009;46(3):213–20.
- Marcus MM, Jardemark K, Malmerfelt A, Bjorkholm C, Svensson TH. Reboxetine enhances the olanzapine-induced antipsychotic-like effect, cortical dopamine outflow and NMDA receptor-mediated transmission. Neuropsychopharmacology. 2010;35(9):1952–61.
- 94. Bjorkholm C, Jardemark K, Marcus MM, Malmerfelt A, Nyberg S, Schilstrom B, et al. Role of concomitant inhibition of the norepinephrine transporter for the antipsychotic effect of quetiapine. Eur Neuropsychopharmacol. 2013;23(7):709–20.
- 95. Meltzer HY, Massey BW, Horiguchi M. Serotonin receptors as targets for drugs useful to treat psychosis and cognitive impairment in schizophrenia. Curr Pharm Biotechnol. 2012;13(8):1572–86.
- 96. Lerond J, Lothe A, Ryviln P, et al. Effects of aripiprazole, risperidone, and olanzapine on 5-HT_{1A} receptors in patients with schizophrenia. J Clin Psychopharmacol. 2013;33:84–9.
- 97. Bennett AC, Vila TM. The role of ondansetron in the treatment of schizophrenia. Ann Pharmacother. 2010;44(7–8):1301–6.
- 98. Meltzer HY, Elkis H, Vanover K, Weiner DM, van Kammen DP, Peters P, et al. Primavanserin, a selective serotonin (5-HT)2A-inverse agonist, enhances the efficacy and safety of risperidone, 2 mg/day, but does not enhance efficacy of haloperidol, 2 mg/day: comparison with reference dose risperidone, 6 mg/day. Schizophr Res. 2012;141(2–3):144–52.
- Rosenzweig-Lipson S, Comery TA, Marquis KL, Gross J, Dunlop J. 5-HT(2C) agonists as therapeutics for the treatment of schizophrenia. Handb Exp Pharmacol. 2012;213:147–65.
- 100. Morozova MA, Beniashvili AG, Lepilkina TA, Rupchev GE. Double-blind placebocontrolled randomized efficacy and safety trial of add-on treatment of dimebon plus risperidone in schizophrenic patients during transition from acute psychotic episode to remission. Psychiatr Danub. 2012;24(2):159–66.

- 101. Kongsamut S, Roehr JE, Cai J, Hartman HB, Weissensee P, Kerman LL, et al. Iloperidone bingind to human and rat dopamine and 5-HT receptors. Eur J Pharmacol. 1996;317(2–3): 417–23.
- 102. Shadid M, Walker GB, Zorn SH, Wong EH. Asenapine: a novel psychopharmacologic agent with a unique human receptor signature. J Psychopharmacol. 2009;23(1):65–73.
- 103. McIntyre RS, Cha DS, Alsuwaidan M, McIntosh D, Powell AM, Jerrell JM. A review of published evidence reporting on the efficacy and pharmacology of lurasidone. Expert Opin Pharmacother. 2012;13(11):1653–9.
- 104. Konradi C, Heckers S. Molecular aspects of glutamate dysregulation: implications for schizophrenia and its treatment. Pharmacol Ther. 2003;97(2):153–79.
- 105. Moghaddam B, Javitt D. From revolution to evolution: the glutamate hypothesis of schizophrenia and its implication for treatment. Neuropsychopharmacology. 2012;37(1):4–15.
- 106. Hallak JE, Maia-de-Oliveria JP, Abrao J, et al. Rapid improvement of acute schizophrenia symptoms after intravenous sodium nitroprusside: a randomized, double-blind, placebocontrolled trial. JAMA Psychiatry. 2013;70(7):668–76.
- 107. Singh SP, Singh V. Meta-analysis of the efficacy of adjunctive NMDA receptor modulators in chronic schizophrenia. CNS Drugs. 2011;25(10):859–85.
- Buchanan RW, Javitt DC, Marder SR, Schooler NR, Gold JM, McMahon RP, et al. The cognitive and negative symptoms in schizophrenia trial (CONSIST): the efficacy of glutamatergic agents for negative symptoms and cognitive impairments. Am J Psychiatry. 2007; 164(10):1593–602.
- 109. Weiser M, Heresco-Levy U, Davidson M, Javitt DC, Werbeloff N, Gershon AA, et al. A multicenter, add-on randomized controlled trial of low-dose d-serine for negative and cognitive symptoms of schizophrenia. J Clin Psychiatry. 2012;73(6):e728–34.
- 110. Chue P. Glycine reuptake inhibition as a new therapeutic approach in schizophrenia: focus on the glycine transporter 1 (GlyT1). Curr Pharm Des. 2013;19(7):1311–20.
- 111. Lane HY, Lin CH, Huang YJ, Liao CH, Chang YC, Tsai GE, et al. A randomized, doubleblind, placebo-controlled comparison study of sarcosine (*N*-methyl glycine) and D-serine add-on treatment for schizophrenia. Int J Neuropsychopharmacol. 2010;13(4):451–60.
- 112. Lane HY, Liu YC, Huang CL, Chang YC, Liau CH, Perng CH, et al. Sarcosine (*N*-methyl glycine) treatment for acute schizophrenia: a randomized, double-blind study. Biol Psychiatry. 2008;63(1):9–12.
- 113. Roche Global Web Site [Homepage on the Internet]. Phase II study with first-in-class investigational drug demonstrates improvement in negative symptoms in patients with schizophrenia. [Basel, December 6, 201]. http://www.roche.com/investors/ir_update/inv-update-2010-12-06b.htm.
- 114. Patil ST, Zhang L, Martenyi F, Lowe SL, Jackson KA, Andreev BV, et al. Activation of nGlu2/3 receptors as new approach to treat schizophrenia: a randomized Phase 2 clinical trial. Nat Med. 2007;13(9):1102–7.
- 115. Kinon BJ, Zhang L, Millen BA, Osuntokun OO, Williams JE, Kollack-Walker S, et al. A ulticenter, inpatient, phase 2, double-blind, placebo-controlled, dose-ranging study of LY2140023 monohydrate in patients with DSM-IV schizophrenia. J Clin Psychopharmacol. 2011;31(3):349–55.
- Kinon BJ, Gomez JC. Clinical development of pomaglumetad methionil: a non-dopaminergic treatment for schizophrenia. Neuropharmacology. 2013;66:82–6.
- 117. Berk M, Copolov D, Dean O, Lu K, Jeavons S, Schapkaitz I, et al. *N*-acetyl cysteine as a glutathione precursor for schizophrenia—a double-blind, randomized, placebo-controlled trial. Biol Psychiatry. 2008;64:361–8.
- 118. Farokhnia M, Azarkolah A, Adinehfar F, Khodaie-Ardakani MR, Hosseini SM, Yekehtaz H, et al. *N*-acetyl cysteine as an adjunct to risperidone for treatment of negative symptoms in patients with chronic schizophrenia: a randomized, double-blind, placebo-controlled study. Clin Neuropharmacol. 2013;36(6):185–92.
- 119. Goff DC, Leahy L, Berman I, Posever T, Herz L, Leon AC, et al. A placebo-controlled pilot study of the ampakine CX516 added to clozapine in schizophrenia. J Clin Psychopharmacol. 2001;21(5):484–7.

- 120. Tiihonen J, Wahlbeck K, Kiviniemi V. The efficacy of lamotrigine in clozapine-resistant schizophrenia: a systematic review and meta-analysis. Schizophr Res. 2009;109(1–3):10–4.
- 121. deLucena D, Fernandes BS, Berk M, Dodd S, Medeiros DW, Pedrini M, et al. Improvement of negative and positive symptoms in treatment-refractory schizophrenia: a double-blind, randomized, placebo-controlled trial with memantine as add-on therapy to clozapine. J Clin Psychiatry. 2009;70(10):1416–23.
- 122. Wassef A, Baker J, Kochan LD. GABA and schizophrenia: a review of basic science and clinical studies. J Clin Psychopharmacol. 2003;23(6):601–40.
- 123. Geffen Y, Keefe R, Rabinowitz J, Anand R, Davidson M. Bl-1020, a new y-aminobutyric acid-enhanced antipsychotic: results of 6-week, randomized, double-blind, controlled, efficacy and safety study. J Clin Psychiatry. 2012;73(9):e1168–74.
- 124. Casey DE, Daniel DG, Wassef AA, Tracy KA, Wozniak P, Sommerville KW. Effect of divaloproex combined with olanzapine or risperidone in patients with an acute exacerbation of schizophrenia. Neuropsychopharmacology. 2003;28(1):182–92.
- 125. Casey DE, Daniel DG, Tamminga C, Kane JM, Tran-Johnson T, Wozniak P, et al. Divalproex ER combined with olanzapine or risperidone for treatment of acute exacerbations of schizophrenia. Neuropsychopharmacology. 2009;34(5):1330–8.
- 126. Glick ID, Bosch J, Casey DE. A double-blind randomized trial of mood stabilizer augmentation using lamotrigine and valproate for patients with schizophrenia who are stabilized and partially responsive. J Clin Psychopharmacol. 2009;29(3):267–71.
- 127. Lewis DA, Cho RY, Carter CS, Eklund K, Forster S, Kelly MA, et al. Subunit-selective modulation of GABA type A receptor neurotransmission and cognition in schizophrenia. Am J Psychiatry. 2008;165:1585–93.
- 128. Hasan A, Wobrock T, Grefkes C, Labusga M, Levold K, Schneider-Axmann T, et al. Deficient inhibitory cortical networks in antipsychotic-naïve subjects at risk of developing first-episode psychosis and first-episode schizophrenia patients: a cross sectional study. Biol Psychiatry. 2012;72:744–51.
- 129. Kegeles LS, Mao X, Stanford AD, Girgis R, Ojeil N, Xu X, et al. Elevated prefrontal cortex γ-aminobutric acid and glutamate-glutamine levels in schizophrenia measured in vivo with proton magnetic resonance spectroscopy. Arch Gen Psychiatry. 2012;69(5):449–59.
- 130. Buchanan RW, Keefe RS, Lieberman JA, Barch DM, Csernansky JG, Goff DC, et al. A randomized clinical trial of MK-0777 for the treatment of cognitive impairments in people with schizophrenia. Biol Psychiatry. 2011;69(5):442–9.
- Ross RG, Hunter SK, McCarthy L, Beuler J, Hutchinson AK, Wagner BD, et al. Perinatal choline effects on neonatal pathophysiology related to later schizophrenia risk. Am J Psychiatry. 2013;170(3):290–8.
- 132. Ribeiz SR, Bassitt DP, Arrais JA, Avila R, Steffens DC, Bottino CM. Cholinesterase inhibitors as adjunctive therapy in patients with schizophrenia and schizoaffective disorder: a review and meta-analysis of the literature. CNS Drugs. 2010;24(4):303–17.
- 133. Dempster EL, Toulopoulou T, McDonald C, Bramon E, Walshe M, Wickham H, et al. Episodic memory performance predicted by the 2b deletion in exon 6 of the "alpha 7-like" nicotinic receptor subunit gene. Am J Psychiatry. 2006;163(10):1832–4.
- 134. Ishikawa M, Hashimoto K. α 7 nicotinic acetylcholine receptor as a potential therapeutic target for schizophrenia. Curr Pharm Des. 2011;17(2):121–9.
- Wallace TL, Bertrand D. Alpha7 neuronal nicotinic receptors as a drug target in schizophrenia. Expert Opin Ther Targets. 2013;17(2):139–55.
- 136. Zhang XY, Liu L, Liu S, Hong X, Chen DC, Xiu MH, et al. Short-term tropisetron treatment and cognitive and P50 auditory gating deficits in schizophrenia. Am J Psychiatry. 2012;169:974–81.
- 137. Tregellas JR, Tanabe J, Rojas DC, Shatti S, Olincy A, Johnson L, et al. Effects of an alpha 7-nicotinic agonist on default network activity in schizophrenia. Biol Psychiatry. 2011;69(1):7–11.
- 138. Freedman R, Olincy A, Buchanan RW, Harris JG, Gold JM, Johnson L, et al. Initial phase 2 trial of a nicotinic agonist in schizophrenia. Am J Psychiatry. 2008;165(8):1040–7.

- 139. Freedman R, Olincy A, Harris JG, et al. An alpha7-nicotinic receptor agonist in combination with risperidone versus olanzapine. Presented at: 14th International Congress on Schizophrenia Research; April 21–25, 2013; Grand Lakes, FL.
- 140. Lieberman JA, Dunbar G, Segreti AC, Girgis SF, Beaver JS, et al. A randomized exploratory trial of an alpha-7 nicotinic receptor agonist (TC-5619) for cognitive enhancement in schizophrenia. Neuropsychopharmacology. 2013;38(6):968–75.
- 141. Shim JC, Jung DU, Jung SS, Seo YS, Cho DM, Lee JH, et al. Adjunctive varenicline treatment with antipsychotic medications for cognitive impairments in people with schizophrenia: a randomized double-blind placebo-controlled trial. Neuropsychopharmacology. 2012;37:660–8.
- 142. Meszaros ZS, Abdul-Malak Y, Dimmock JA, Wang D, Ajagbe TO, Batki SL. Varenicline treatment of concurrent alcohol and nicotine dependence in schizophrenia: a randomized, placebo-controlled pilot trial. J Clin Psychopharmacol. 2013;33:243–7.
- 143. Ahmed AIA, Ali ANA, Kramers C, Harmark LVD, Burger DM, Verhoeven WMA. Neuropsychiatric adverse events of varenicline: a systematic review of published reports. J Clin Psychopharmacol. 2013;33:55–62.
- 144. Conley RR, Boggs DL, Kelly DL, et al. The effects of galantamine on psychopathology in chronic stable schizophrenia. Clin Neuropharmacol. 2009;32(2):69–74.
- 145. Shekhar A, Potter WZ, Lightfoot J, Lienemann J, Dubé S, Mallinckrodt C, et al. Selective muscarinic receptor agonist xanomeline as a novel treatment approach for schizophrenia. Am J Psychiatry. 2008;165(8):1033–9.
- 146. Bridges TM, LeBois EP, Hopkins CR, Wood MR, Jones CK, Conn PJ, et al. The antipsychotic potential of muscarinic allosteric modulation. Drug News Perspect. 2010;23(4): 229–40.
- 147. Bolbecker AR, Shekhar A. Muscarinic agonists and antagonists in schizophrenia: recent therapeutic advances and future directions. Handb Exp Pharmacol. 2012;208:167–90.
- 148. Dean B. Selective activation of muscarinic acetylcholine receptors for the treatment of schizophrenia. Curr Pharm Biotechnol. 2012;13(8):1563–71.
- 149. Lebois EP, Jones CK, Lindsley CW. The evolution of histamine H₃ antagonists/inverse agonists. Curr Top Med Chem. 2011;11(6):648–60.
- Schwartz JC. The histamine H₃ receptor: from discovery to clinical trials with pitolisant. Br J Pharmacol. 2011;163(4):713–21.
- 151. Ito C. Histamine H3-receptor inverse agonists as novel antipsychotics. Cent Nerv Syst Agents Med Chem. 2009;9(2):132–6.
- 152. Wei Z, Wang L, Zhang M, Xuan J, Wang Y, Liu Y, et al. A pharmacogenetic study of risperidone on histamine H3 receptor gene (HRH3) in Chinese Han schizophrenia patients. JPsychopharmacol. 2012;26(6):813–8.
- 153. Meskanen K, Ekelund H, Laitinen J, et al. A randomized clinical trial of histamine 2 receptor antagonism in treatment-resistant schizophrenia. J Clin Psychopharmacol. 2013;33:472–8.
- Chaudhry IB, Husain N, Khan S, Badshah S, Deakin B, Kapur S. Amoxapine as an antipsychotic. J Clin Psychopharmacol. 2007;27:575–81.
- 155. Sepehry AA, Potvin S, Elie R, Stip E, et al. Selective serotonin reuptake inhibitor (SSRI) add-on therapy for the negative symptoms of schizophrenia: a meta-analysis. J Clin Psychiatry. 2007;68(4):604–10.
- 156. Singh SF, Singh V, Kar N, Chan K. Efficacy of antidepressant in treating the negative symptoms of chronic schizophrenia: meta-analysis. Br J Psychiatry. 2010;197(3):174–9.
- 157. Zisook S, Kasckow JW, Golshan S, Fellows I, Solorzano E, Lehman D, et al. Citalopram augmentation for subsyndromal symptoms of depression in middle-aged and older outpatients with schizophrenia and schizoaffective disorder: a randomized controlled trial. J Clin Psychiatry. 2009;70(4):562–71.
- 158. Bodkin JA, Siris SG, Bermanzohn PC, Hennen J, Cole JO. Double-blind, placebo-controlled, multicenter trial of selegiline augmentation of antipsychotic medication to treat negative symptoms in outpatients with schizophrenia. Am J Psychiatry. 2005;162:388–90.

- 159. Mico U, Bruno A, Pandolfo G, Romeo VM, Mallamace D, D'Arrigo C, et al. Duloxetine as adjunctive treatment to clozapine in patients with schizophrenia: a randomized, placebocontrolled trial. Int Clin Psychopharmacol. 2011;26(6):303–10.
- Phan SV, Kreys TJ. Adjunct mirtazapine for negative symptoms of schizophrenia. Pharmacotherapy. 2011;31(10):1017–30.
- 161. Cho SJ, Yook K, Kim B, Choi TK, Lee S, Kim YW, et al. Mirtazapine augmentation enhances cognitive and reduces negative symptoms in schizophrenia patients treated with risperidone: a randomized controlled trial. Prog Neuropsychopharmacol Biol Psychiatry. 2011;35(1):208–11.
- 162. Terevnikov V, Stenberg JH, Tiihonen J, Joffe M, Burkin M, Tchoukhine E, et al. Add-on mirtazapine improves depressive symptoms in schizophrenia: a double-blind randomized placebo-controlled study with an open-label extension phase. Hum Psychopharmacol. 2011;26(3):188–93.
- 163. Stenberg JH, Terevnikoz V, Joffe M, Tiihonen J, Tchouhine E, Burkin M, et al. More evidence on proneurocognitive effects of add-on mirtazapine in schizophrenia. Prog Neuropsychopharmacol Biol Psychiatry. 2011;35(4):1080–6.
- 164. Caforio G, Di Giorgio A, Rampino A, Rizzo M, Romano R, Raurisano R, et al. Mirtazapine add-on improves olanzapine effect on negative symptoms of schizophrenia. J Clin Psychopharmacol. 2013;33(6):810–2.
- Gallagher P, Watson S, Smith MS, Ferrier IN, Young AH. Effect of adjunctive mifepristone (RU-486) administration on neurocognitive function and symptoms in schizophrenia. Biol Psychiatry. 2005;57(2):155–61.
- Ritsner MS. The clinical and therapeutic potentials of dehydroepiandrosterone and pregnenolone in schizophrenia. Neuroscience. 2011;191:91–100.
- 167. Marx CE, Bradford DW, Hamer RM, Naylor JC, Allen TB, Lieberman JA, et al. Pregnenolone as a novel therapeutic candidate in schizophrenia: emerging preclinical and clinical evidence. Neuroscience. 2011;191:78–90.
- 168. Ritsner MS, Gibel A, Shleifer T, Boguslavsky I, Zayed A, Maayan R, et al. Pregnenolone and dehydroepiandrosterone as an adjunctive treatment in schizophrenia and schizoaffective disorder: an 8-week, double-blind, randomized, controlled, 2-center, parallel-group trial. J Clin Psychiatry. 2010;71(10):1351–62.
- 169. Fineberg AM, Ellman LM. Inflammatory cytokines and neurological and neurocognitive alterations in the course of schizophrenia. Biol Psychiatry. 2013;73:951–66.
- 170. Miller BJ, Gassama B, Sebastian D, Buckley P, Mellor A. Meta-analysis of lymphocytes in schizophrenia: clinical status and antipsychotic effects. Biol Psychiatry. 2013;73:993–9.
- 171. Muller N, Riedel M, Scheppach C, Brandstatter B, Sokullu S, Krampe K, et al. Beneficial antipsychotic effects of celecoxib add-on therapy compared to risperidone alone in schizo-phrenia. Am J Psychiatry. 2002;159:1029–34.
- 172. Laan W, Grobbee DE, Selten JP, Heijnen CJ, Kahn RS, Burger H. Adjuvant aspirin therapy reduces symptoms of schizophrenia spectrum disorders: results from a randomized. Doubleblind, placebo-controlled trial. J Clin Psychiatry. 2010;71(5):520–7.
- 173. Sommer IE, de Witt L, Begemann M, Kahn RS, et al. Nonsteroidal anti-inflammatory drugs in schizophrenia: ready for practice of a good start? A meta-analysis. J Clin Psychiatry. 2012;73(4):414–9.
- 174. Levkovitz Y, Mendolovich S, Riwkes S, Braw Y, Levkovitch-Verbin H, Gal G, et al. A double-blind, randomized study of minocycline for the treatment of negative and cognitive symptoms in early-phase schizophrenia. J Clin Psychiatry. 2010;71(2):138–49.
- 175. Chaudhry IB, Hallak J, Husain N, Minhas F, Stirling J, Richardson P, et al. Minocycline benefits negative symptoms in early schizophrenia: a randomised double-blind placebocontrolled clinical trial in patients on standard treatment. J Psychopharmacol. 2013;26(9):1185–93.
- Skosnik PD, Ranganathan M, D'Souza DC. Cannabinoids, working memory, and schizophrenia. Biol Psychiatry. 2012;71:662–3.

- 177. Evins AE, Green AI, Kane JM, Murray RM. Does using marijuana increase the risk for developing schizophrenia? J Clin Psychiatry. 2013;74(4):e08.
- 178. Skosnik PD, D'Souza DC, Steinmetz AB, Edwards CR, Vollmer JM, Hetrick WP, et al. The effect of cannabinoids on broadband EEG neural oscillations in humans. Neuropsychopharmacology. 2012;37(10):2184–93.
- 179. Kulkarni J, de Castella A, Fitzgerald B, Gurvich CT, Bailey M, Batholomeusz C, et al. Estrogen in severe mental illness: a potential new treatment approach. Arch Gen Psychiatry. 2008;65(8):955–60.
- 180. Usall J, Heurta-Ramos E, Iniesta R, Cobo J, Araya S, Roca M, et al. Raloxifene as an adjunctive treatment for post menopausal women with schizophrenia: a double-blind, randomized, placebo-controlled trial. J Clin Psychiatry. 2011;72(11):1552–7.
- Begemann MJ, Dekker CF, van Lunenburg M, Sommer IE. Estrogen augmentation in schizophrenia: a quantative review of current evidence. Schizophr Res. 2012;141(2–3):179–84.
- 182. Ghafari E, Fararouie M, Shirazi HG, Farhangfar A, Ghaderi F, Mohammadi A. Combination of estrogen and antipsychotics in the treatment of women with chronic schizophrenia. Clinical Schizophr Relat Psychoses. 2013;6(4):172–6.
- MacDonald K, Feifel D. Oxytocin in schizophrenia: a review of evidence for its therapeutic effects. Acta Neuropsychiatr. 2012;24:130–46.
- Weisman O, Feldman R. Oxytocin effects on the human brain: findings, questions, and future directions. Biol Psychiatry. 2013;74:158–9.
- 185. Modabbernia A, Rezaei F, Salehi B, Jafarinia M, Ashrafi M, Tabrizi M, et al. Intranasal oxytocin as an adjunct to risperidone in patients with schizophrenia: an 8-week, randomized, double-blind, placebo-controlled study. CNS Drugs. 2013;27(1):57–65.
- 186. Lee MR, Wehring HJ, McMahon RP, et al. Effects of adjunctive intranasal oxytocin on olfactory identification and clinical symptoms in schizophrenia: results from a randomized doubleblind placebo controlled pilot study. Schizophr Res. 2013;145(1–3):110–5.
- 187. Woolley J, Chuang B, Lam O, et al. The effects of oxytocin on social cognition and olfaction in patients with schizophrenia and healthy controls. Presented at: 14th International Congress on Schizophrenia Research; April 21–25, 2013; Grand Lakes, FL.
- 188. Marder SR. Oxytocin and social cognition training in schizophrenia. Presented at: 4th International Congress on Schizophrenia Research; April 21–25, 2013; Grand Lakes, FL.
- 189. Davis MD, Lee J, Horan WP, et al. A pilot study on the effects of single dose intranasal oxytocin on social cognition in schizophrenia. Presented at: 14th International Congress on Schizophrenia Research; April 21–25, 2013; Grand lakes, FL.
- 190. Fusar-Poli P, Berger G. Eicosapentaenoic acid interventions in schizophrenia: meta-analysis of randomized, placebo-controlled studies. J Clin Psychopharmacol. 2012;32:179–85.
- 191. Gunawardana L, Smith GD, Zammit S, Whitley E, Gunnell D, Lewis S, et al. Pre-conception inter-pregnancy interval and risk of schizophrenia. Br J Psychiatry. 2011;199(4):338–9.
- 192. Roffman JL, Lamberti JS, Achtyes E, Macklin EA, Galendez GC, Raeke LH, et al. Randomized multicenter investigation of folate plus vitamin B12 supplementation in schizophrenia. JAMA Psychiatry. 2013;6:1–9.
- 193. Burghardt KJ, Ellingrod VL. Detection of metabolic syndrome in schizophrenia and implications for antipsychotic therapy: is there a role for folate? Mol Diagn Ther. 2013;17(1):21–30.
- McGrath J, Brown A, St Clair D. Prevention and schizophrenia—the role of dietary factors. Schizophr Bull. 2011;37(2):272–83.
- 195. Eyles DW, Burne TH, McGrath JJ. Vitamin D, effects on brain development, adult brain functions and the links between low levels of vitamin D and neuropsychiatric diseases. Front Neuroendocrinol. 2013;34(1):47–64.
- 196. McGrath JJ, Burne TH, Feron F, Mackay-Sim A, Eyles DW. Developmental vitamin D deficiency and risk of schizophrenia: a 10-year update. Schizophr Bull. 2010;36(6):1073–8.
- 197. McGrath JJ, Eyles DW, Pedersen CB, Anderson C, Ko P, Burne TH, et al. Neonatal vitamin D status and risk of schizophrenia: a population-based case–control study. Arch Gen Psychiatry. 2010;67(9):889–94.

- 198. McGrath JJ, Saari K, Hakko H, Jokelainen J, Jones P, Jarvelin MR, et al. Vitamin D supplementation during the first year of life and risk of schizophrenia: a Finnish birth cohort study. Schizophr Res. 2004;67(2–3):237–45.
- 199. Shean GD. Empirically based psychosocial therapies for schizophrenia: the disconnection between science and practice. Schizophr Res Treatment. 2013;2013:792769.
- 200. Lewis S, Tarrier N, Haddock G, Bentail R, Kinderman P, Kingdon D, et al. Randomised controlled trial of cognitive-behavioural therapy in early schizophrenia: acute-phase outcomes. Br J Psychiatry. 2002;43(Suppl):s91–7.
- 201. Tarrier N, Lewis S, Haddock G, Bentall R, Drake R, Kinderman P, et al. Cognitive-behavioural therapy in first-episode and early schizophrenia: 18-month follow-up of a randomised controlled trial. Br J Psychiatry. 2004;184:231–9.
- 202. Startup M, Jackson MC, Bendix S. North Wales randomized controlled trial of cognitive behavior therapy for acute schizophrenia spectrum disorders: outcomes at 6 and 12 months. Psychol Med. 2004;34(3):413–22.
- 203. Startup M, Jackson MC, Evans KE, Bendix S. North Wales randomized controlled trial of cognitive behavior therapy for acute schizophrenia spectrum disorders: two-year follow-up and economic evaluation. Psychol Med. 2005;35(9):1307–16.
- 204. Bechdolf A, Knost B, Nelson B, Schneider N, Veith V, Yung AR, et al. Randomized comparison of group cognitive behavior therapy and group psychoeducation in acute patients with schizophrenia: effects on subjective quality of life. Aust N Z J Psychiatry. 2010; 44(2):144–50.
- 205. Schulz M, Gray R, Spiekermann AC, Behrens J, Driessen M. Adherence therapy following an acute episode of schizophrenia: a multi-centre randomised controlled trial. Schizophr Res. 2013;146(1–3):59–63.
Chapter 8 Therapeutic Neuromodulation for Treatment of Schizophrenia

Jeffrey T. Rado and Edgar I. Hernandez

Introduction

Antipsychotics are the primary pharmacologic treatment for schizophrenia, yet many patients do not achieve a clinically meaningful response. Approximately 25–30 % are refractory to these medications and up to 40 % of previously unresponsive patients also fail a trial of clozapine [1, 2]. As a result, nonpharmacologic approaches are increasingly being considered as potentially safe and effective adjunctive strategies. Therapeutic neuromodulation is a growing field of study for refractory auditory hallucinations (AHs), the deficit syndrome, and other symptoms such as catatonia. This chapter discusses its emerging role in the treatment of schizophrenia.

Positive Symptoms

Electroconvulsive Therapy

Electroconvulsive therapy (ECT) was first introduced in 1938 by Ugo Cerletti who applied electrical current to the head to elicit seizures as a treatment for various mental illnesses, including schizophrenia [3]. With the introduction of antipsychotics

J.T. Rado (🖂)

Department of Psychiatry, Rush University Medical Center, Chicago, IL 60612, USA

Department of Internal Medicine, Rush University Medical Center, Chicago, IL 60612, USA e-mail: jeffrey_rado@rush.edu

E.I. Hernandez Department of Psychiatry, Rush University Medical Center, Chicago, IL 60612, USA e-mail: edgar i hernandez@rush.edu

in the 1960s, ECT fell out of favor. Given the substantial subset of patients who still did not achieve sufficient benefit from medication, however, interest in ECT resurfaced in the 1980s [4].

Mechanism of Action

ECT's proposed mechanisms of action include modulation of monamines; change in neurotrophic factors (e.g., increased BDNF); and changes in GABA transmission and receptor antagonism [5]. Interestingly, in a review by Bolwig on the prevailing theories, the author states that "ECT is effective in various illnesses such as depression, mania, schizophrenia, and catatonia, but it remains an unresolved issue whether ECT exerts differential effects, or whether these obviously different disorders have common pathophysiological bases" [6].

ECT for Positive Symptoms

While direct comparisons between ECT and antipsychotic drugs for treating schizophrenia are limited, numerous studies have evaluated ECT as an adjunct to antipsychotics. For example, Goswani et al. randomized 25 chlorpromazine-treated adults with schizophrenia to either active ECT or a control group (anesthesia was induced in these patients but no ECT treatments were given) [7]. The authors reported that weekly Brief Psychiatric Rating Scale (BPRS) scores significantly improved (p < 0.002) from baseline in the active ECT-treated patients (n = 15) after at least six treatments. Kupchik and colleagues reviewed 36 cases of patients treated with ECT plus clozapine [8]. Of these cases, 23 patients were diagnosed with schizophrenia or schizoaffective disorder, while the rest were diagnosed with mood disorders. Clozapine was added to ECT in 18 patients, while in 17, ECT was introduced after the initiation of clozapine. The ECT electrode placements, number of treatments, and stimulus intensity were not reported. This combination was highly effective in 24 patients with 67 % demonstrating marked improvement from baseline. The authors point to a possible synergistic effect by enhancing brain electrophysiological changes. Chanpattana et al. found benefit for refractory patients who received 20 bilateral ECT treatments (three times per week) in addition to flupenthixol [9]. Refractory patients were defined as those who failed two separate trials of antipsychotic treatment from two different classes for a minimal duration of 6 weeks. Flupenthixol was started simultaneously with ECT after other psychotropics were discontinued. Patients received 12 mg daily during the first week and then the dose was increased to 24 mg daily (depending on tolerability). Those achieving a BPRS score of ≤ 25 entered a 3-week stabilization period where patients received three ECT treatments in the first week followed by once weekly treatments during the subsequent 2 weeks. Patients were considered responders if they completed this period with no worsening in their scores. Response was achieved in 160 of 293 (54.6 %) participants. In particular, five symptoms demonstrated marked improvement after ECT: conceptual disorganization (effect size (ES) = 0.94); tension (ES = 0.89); mannerisms and posturing (ES = 1.06); hostility (ES = 0.83); and unusual thought content (ES = 1.02). By contrast, motor retardation, blunted affect, and disorientation worsened. In summary, the ECT–flupenthixol combination resulted in marked improvement for specific positive symptoms, an intermediate improvement for affective symptoms, and no benefit for negative symptoms.

Nothdurfter et al. conducted a retrospective analysis of 455 patients receiving ECT combined with first- (FGAs) or second-generation antipsychotics (SGAs) versus ECT monotherapy [10]. Patients with schizophrenia (n = 170) received 11.59 (±5.3) ECT treatments (76 % bilateral electrode placement). The authors suggest that the ECT-antipsychotic combination improves electrophysiological indices, particularly the postictal suppression index (PSI), which may predict treatment response to ECT. The PSI demonstrates how fast and complete the EEG amplitude flattens directly after a seizure, with a correlation between high percent suppression and a more robust therapeutic response [11]. The outcome measure was assessed by the Clinical Global Impression Scale (CGI-S). The authors report that the PSI of treatments done with SGAs was significantly higher than the PSI of treatments done with low potency FGAs (p < 0.0001; actual values not reported), supporting the notion that the index appears to be a reliable parameter of response. Compared with FGAs, CGI score improvements were superior for SGAs and this correlated to a higher percent suppression ($p \le 0.0001$; actual values not reported). Finally, they found no differences in adverse effects among the antipsychotics.

Reviews and Meta-analyses of ECT for Positive Symptoms

Tharyan et al. conducted a Cochrane review of 26 studies comparing active versus sham ECT, ECT combined with antipsychotics, and ECT versus other interventions [12]. Improvement was measured by various outcomes, and the authors describe the definition of a "clinically meaningful benefit" based on the individual study's scale. The data was then made binary to calculate a relative risk (RR) when possible. Overall, greater improvement occurred in patients treated with active versus sham ECT (n=392; 10 RCTs; RR=0.71 [CI=0.59–0.98]; NNT=6 [CI 4–12]). Although results favor the medication group when ECT is compared directly with antipsychotics, there is evidence to suggest that the *combination* treatment results in greater improvement of symptoms in the short term (n=203; 7 RCTs; RR 0.76 [CI 0.52–1.10]). Further, the authors found data supporting the ECT–antipsychotic combination as a viable option for adult patients with schizophrenia, particularly when rapid global improvement and reduction of symptoms are necessary (n=40; 1 RCT, Weighted Mean Differences BPRS –3.9 [CI –2.28 to –5.52]). The review noted that these results are less relevant for those with schizophreniform disorder or unspecified psychosis.

A review and meta-analysis conducted by Painuly et al. found the combination of ECT plus antipsychotic to be more efficacious than antipsychotic monotherapy [13]. For the review, 11 open and controlled trials were selected (n=651), while the meta-analysis only included four RCTs (n=113). The benefit was seen primarily in

the first 3–6 weeks when ECT appeared to cause an "acceleration of treatment response" during an acute exacerbation as measured by changes in BPRS scores. The effect of combined treatment was most pronounced on affective and behavioral disturbances, yielding an additional five-point reduction in BPRS scores with combination treatment versus an antipsychotic alone.

Matheson et al. conducted a meta-analysis which found that adjunctive ECT provided a small but significant improvement in global symptoms compared with a sham procedure (RR=0.76) [14]. Further, ECT plus antipsychotic demonstrated a moderate benefit over sham ECT plus antipsychotic based on BPRS scores (d=-0.75) for up to 6 weeks posttreatment. Gazdag et al., however, point to the heterogeneity of study samples due to changes in the diagnostic criteria for schizo-phrenia over the years and variability in study designs as limitations in assessing ECT's effectiveness to treat schizophrenia [15]. A more recent review article by Zervas et al. concluded that ECT is most effective in treating catatonia, paranoid delusions, and other positive symptoms [16].

Despite decades of utilizing ECT for the treatment of various psychiatric diseases, the literature reports inconsistencies when it comes to its role in schizophrenia. This may be due to the lack of well-controlled study designs, heterogeneous samples, and small sample sizes. Despite these acknowledged limitations, the evidence supports the use of ECT in the treatment of schizophrenia, particularly if combined with antipsychotics in patients who demonstrated limited response to an antipsychotic alone. As the authors of the Cochrane review declared, "The efficacy of ECT, when schizophrenia is diagnosed using operationally defined criteria, indicates a specific effect on schizophrenic symptoms, rather than causing an improvement in those with affective disorder misdiagnosed as schizophrenia" [12].

ECT for Catatonia

Several case reports support ECT's efficacy and safety for patients with catatonia, especially if symptoms do not improve with benzodiazepines [17-19]. In a retrospective study (n=22), Rohland et al. reported no difference in the effectiveness of ECT for the resolution of catatonic symptoms in patients with mood disorders versus schizophrenia [20]. By contrast, it has also been reported that patients with catatonia secondary to mood disorders obtain a greater improvement compared to those with catatonia associated with schizophrenia [21].

In a randomized, controlled study of 14 non-affective, catatonic patients (n=9 schizophrenia; n=5 psychosis NOS), ECT plus oral placebo was compared with sham ECT plus risperidone (4–6 mg) in nonresponders to lorazepam [22]. ECT demonstrated significant improvement over risperidone on both outcome measurements (i.e., the Bush-Francis Catatonia Rating Scale ($p \le 0.035$) and Positive and Negative Symptom Scale (PANSS) ($p \le 0.04$)). The study concluded that ECT may be safely used as a monotherapy for catatonia. The Cochrane review by Tharyan et al., however, concluded that "there is no clear evidence to support or refute the use of ECT in catatonia given the lack of randomized controlled trials" [12].

Variable	Male	Female
Age	Younger	Younger
Duration of current episode	Short duration	Short duration
Symptom profiles	Delusions, blunted affect, suspiciousness	Anxiety, depressed mood, hallucinations
Diagnostic subtype	Paranoid	-
Psychiatric admissions	More frequent admissions	-
Duration of illness	-	Short duration
Family history	-	Negative family history

 Table
 8.1
 Predictors of response to adjunctive ECT in treatment-resistant schizophrenia: interaction between clinical variables and gender

Source: Data from Chanpattana et al. [25]

ECT for Adolescents

In a prospective study of seven antipsychotic-treated patients with intractable first-episode schizophrenia or schizophreniform disorder (age 15–34 years), Suzuki et al. reported improvement in symptoms when bilateral ECT was administered two or three times per week (total number of sessions was limited to 20) [23]. BPRS and Global Assessment of Functioning (GAF) scores improved significantly after the acute treatment course ($p \le 0.018$ and $p \le 0.018$, respectively). Based on Chanpattana et al.'s criterion of a final BPRS score ≤ 25 after 1 week of an acute ECT treatment, all seven patients were considered responders $(10.7 \pm 8.0 \text{ after the first ECT course})$ versus 44.4 ± 5.3 before the course) [9]. Further, ECT was safe and well tolerated. Similarly, in a retrospective study of 13 adolescents (ages 13–17), Baeza et al. demonstrated that an acute course of bilateral ECT (mean number of sessions = 13.9 (±4.3)) combined with an antipsychotic was effective and safe in treating schizophrenia spectrum disorders [24]. In addition, the mean endpoint PANSS scores after 6 months of completing the first acute ECT course demonstrated significant improvement in the positive and general subscale scores ($p \le 0.004$), while negative symptoms were unchanged.

Predictors of Response

In a prospective study of 253 patients, Chanpattana et al. considered factors which may predict response to ECT plus flupenthixol in treatment-resistant schizophrenia (see Table 8.1) [25]. The authors report that frequent psychiatric admissions and paranoid subtype were associated with better response on the BPRS (primary outcome) in males, while shorter duration of illness and a negative family history for schizophrenia were associated with better response in females. Finally, greater severity of positive or affective symptoms compared with more severe negative symptoms also predicted greater response as measured by the BPRS.

ECT for Maintenance Treatment

In a retrospective chart review of 19 patients with schizophrenia or schizoaffective disorder, Levy-Rueff et al. considered maintenance ECT (M-ECT) plus medication as a treatment strategy in refractory patients [26]. The authors note a "substantial, although moderate" efficacy for M-ECT on symptom intensity (although no standardized measures were provided); duration of hospitalizations (80 % reduction); and quality of life. No factors predicted relapse. Similarly, in a chart review (n=79)conducted by Kristensen et al., the authors report that no predictive factors were identified, but most patients treated with bilateral ECT (acute treatment 2-26 sessions; M-ECT duration 3 months to 12 years; once every 1-3 weeks) experienced good or moderate outcomes (n=66 and n=8, respectively) with only five patients experiencing a poor outcome [27]. The outcomes were defined by assessing severity of illness and degree of improvement by categorizing the wording in each patient's chart. The authors note that these categories correlated well with the CGI Scale (Item 3.1). Specifically, of the 18 patients treated with M-ECT, 16 (88 %) showed excellent or good response based on the degree of relief from psychosis and disruptive behavior, while two demonstrated only a moderate benefit.

ECT Administration

Important issues when using ECT to treat schizophrenia include optimal electrode placement; stimulus intensity and frequency; and number of ECT sessions. While successful treatment with both bilateral and unilateral electrode placement is documented, bilateral ECT may result in a quicker response. The existing data, however, does not conclusively support the use of either electrode placement. Thus far, the stimulus intensity requirement for patients with schizophrenia has not differed from those with major depression. Chanpattana et al., however, proposed that doses 2–4 times the seizure threshold are more likely to speed recovery with bilateral ECT [28]. They also support the use of 20 ECT treatments to elicit full response. Further, in line with studies in major depression [29, 30], the authors concluded patients receiving thrice weekly ECT treatments responded faster than those on twice weekly schedules based on the improvement in BPRS and GAF scores [31].

ECT Guidelines for Treating Schizophrenia

There is no general consensus regarding the clinical use of ECT in schizophrenia. In a review by Pompili et al., the authors conclude that the most common indication for ECT in schizophrenia was to augment pharmacotherapy (particularly in treatment-resistant patients), while the most common symptoms treated were catatonia, aggression, and suicidal ideation [32]. Further, the American Psychiatric Association Guidelines (2001) propose that patients with acute onset of predominantly depressive or catatonic symptoms would benefit most from ECT, implying that

Table 8.2 Parameters used with adjunctive ECT in schizophrenia	Electrode placement	Bilateral ^a
	Stimulus intensity	1.5-4 times seizure threshold
	Number of treatments	6–20
	Dosing frequency	3 times per week ^b
	Adjunct medication	Second-generation antipsychotics ^c
	^a Noted to result in faster response	
	^b Faster onset of improvement	
	^c Clozapine in treatment-resistant cases	

predominant delusions or hallucinations are less likely to benefit [33]. The review by Zervas et al. concludes, however, "these studies have indicated that one should not exclude patients with refractory schizophrenia from a trial of ECT only on the basis of lack of affective symptoms, especially if the so called 'positive' symptoms of psychosis are present" [16]. Finally, the 2009 Schizophrenia Patient Outcomes Research Team (PORT) review concluded that "there is insufficient evidence to support a recommendation for the use of ECT for the core symptoms of schizophrenia in treatment-resistant individuals" [34]. The authors, nonetheless, acknowledge the role of ECT in treating acute positive psychotic symptoms.

Conclusion

Adjunctive ECT may be an effective therapy for positive symptoms such as refractory auditory hallucinations (AHs) or catatonia; when rapid reduction of symptoms is needed; as both an acute and maintenance treatment; and in both adults and adolescents. Although more research is necessary to clarify its mechanisms of action, predictors of response, and optimal application, ECT remains an important treatment option in clinical practice. Table 8.2 provides a general summary of parameters which were successful in the treatment of schizophrenia.

Transcranial Magnetic Stimulation

Transcranial magnetic stimulation (TMS) delivers intense, intermittent, magnetic pulses generated by an electrical charge into a ferromagnetic coil. Applied to the scalp, rapid repetitive TMS (≥ 1 Hz) causes neuronal depolarization in a localized area under the coil, as well as distal effects [35]. In contrast, slow repetitive TMS (<1 Hz) inhibits neuronal depolarization. TMS is generally a safe and well-tolerated procedure, with application site discomfort and mild headaches being the most common adverse effects. While inadvertent seizure occurrence is more serious, its incidence is far less than 1 %. This novel device-based therapy is approved for the treatment of depression, but has demonstrated benefit for other neuropsychiatric disorders, including positive and negative symptoms in schizophrenia. While high

frequency (≥ 1 Hz) TMS has FDA approval for depression and may also benefit negative symptoms, it is suggested that low frequencies (<1 Hz) with extended duration of treatment (e.g., >15 min) may benefit treatment-resistant auditory hallucinations (AHs).

Mechanism of Action

Functional magnetic resonance imaging (fMRI) studies implicate left temporoparietal cortex (TPC) hyperactivity in the etiology of hallucinations [36]. Further, it is postulated that low frequency TMS may attenuate excitability in this region, while high frequency stimulation generates the opposite effect. Thus, most treatment studies with TMS for AHs stimulated with low frequency over the region between T3 and P3 as described in the International 10-20 EEG System of Electrode Placement [37].

TMS for Positive Symptoms

Early studies by Geller et al. and Feinsod et al. employed low frequency TMS (<1 Hz) in patients with depression or schizophrenia [38, 39]. The authors demonstrated improvement in mood and anxiety but not in the core symptoms of schizophrenia. Neither report looked at the effects of TMS specifically for AHs.

To investigate TMS for the treatment of medication-resistant AHs, Hoffman et al. conducted a double-blind, crossover study (n=3) with low frequency TMS at 80 % motor threshold (MT) over the left TPC given on 4 days during a 2-week period [40]. Stimulation was initiated at 4 min on day 1 with increases of 4 min each day to a maximum of 16 min duration on day 4. Active versus sham TMS stimulation took place on separate weeks with 2–3 days off between trials. The authors reported a significant decrease in hallucination severity based on an individualized rating scale for each patient which corresponded to a composite score incorporating loudness, frequency, content, and level of distress. Following active but not sham stimulation, all three patients reportedly experienced significant improvement in AHs (*p* value not reported).

Subsequently, Hoffman et al. conducted a second, double-blind, crossover trial (n=12) using the same design as in their first study [41]. Again, hallucination severity was significantly decreased with active versus sham stimulation $(p \le 0.006)$. This group then published additional data in two separate reports which included a total of 50 patients randomized to low frequency, active TMS (90 % MT; 132 min of total stimulation) or sham TMS over the left TPC [42, 43]. Patients received 8 min of continuous stimulation on day 1, 12 min on day 2, and 16 min daily for the next 7 days. Responders had a Hallucination Change Scale (HCS) score of 5 or lower at the end of treatment. Fourteen patients in the active group (51.9 %) versus four in the sham group (17.4 %) met this criterion ($p \le 0.01$). In addition, the frequency quality item of the Auditory Hallucination Rating Scale (AHRS) demonstrated a significant linear decrease over time in the active versus sham group ($p \le 0.0001$). There were no

differences in other AHRS variables or PANSS scores. Of note, Fitzgerald et al. were unable to replicate Hoffman's results using the same parameters in a randomized, controlled study with 33 patients [44]. The authors, however, did find that eight patients in the active group met response criteria compared with only four in the sham group (p < 0.05) using the HCS scores as the primary outcome measure. In addition, except for a significant reduction in the volume of hallucinations in the active versus sham group (p < 0.01), no significant differences were observed in change scores on the Psychotic Symptom Rating Scale (PSYRATS). Finally, a randomized, double-blind controlled study (n=83) by Hoffman et al. recently investigated the use of TMS over Wernicke's area anterior to the T3/P3 site and its right homologous site to treat refractory auditory/verbal hallucinations (AVHs) [45]. This was based on fMRI studies linking AVHs with activation at these sites which are involved in the processing of external speech and acoustic processing. The TMS parameters used for this study included 1 Hz; 90 % MT; 16 min per session; and 960 pulses over a 15-day course. The stimulations were administered during five weekday sessions at one site and then switched to the opposite hemisphere for another five sessions. A final five-session block was delivered to the site associated with greater initial improvement in AVHs based on change in scores in the HCS. Hallucination severity at endpoint did not differ between the active or sham groups (p < 0.09), but the hallucination frequency difference (i.e., baseline minus endpoint score) was significantly less in the active treatment group (-1.32 active versus -0.26 m)sham), as were the CGI scores ($p \le 0.005$ and $p \le 0.045$, respectively). Interestingly, by eliminating patients for whom the MT could not be consistently detected, there was a statistically significant improvement in the active condition (p < 0.044) attributed to a reduction in sham response. The authors concluded that a bilateral site-optimization protocol using active TMS reduced AVHs.

Other groups have also pursued the use of TMS for positive symptoms. McIntosh et al. conducted a placebo-controlled, crossover study of TMS (1 Hz; 80 % MT) over the left TPC in 16 patients given on 4 days over a 2-week period [46]. Stimulation was initiated at 4 min on day 1 with increases of 4 min each day to a maximum of 16 min duration on day 4. Of note, a 15 s pause was allowed between each continuous minute of treatment. The authors found a trend for improvement on the PANSS hallucination subscale score after the second week of stimulation, suggesting that a longer period of treatment may be more effective. Poulet et al. conducted a double-blind, crossover study (n=10) of active (1 Hz; 90 % MT; 1,000 stimulations per day) versus sham TMS delivered twice daily over the left TPC on 5 days [47]. All seven items of the AHRS were significantly reduced during active versus sham treatment ($p \le 0.008$). They also found that 50 % of the actively treated patients achieved response (i.e., improvement of ≥ 20 % compared to baseline severity) after 8 weeks. Of note, instead of delivering sham treatment with an active coil angled at 45 or 90° away from the skull, this study was the first to use a sham coil which produced the same sound but no superficial scalp stimulation, making it less distinguishable from the active coil.

Chibarro et al. conducted a randomized, controlled study in 16 schizophrenia patients treated with SGAs [48]. Low frequency TMS (1 Hz; 90 % MT; one

continuous 15-min period; 900 total stimulations) over the left TPC was administered on four consecutive days. The Scale for the Assessment of Positive Symptoms (SAPS) and Severity of Auditory Hallucinations (SAH) scores were significantly decreased in both the active ($p \le 0.04$ and $p \le 0.001$, respectively) and sham groups (p < 0.02 and p < 0.01, respectively). The reduction in AHs in the active group, however, was sustained 6 weeks after treatment completion, whereas the sham group returned to baseline.

Saba et al. conducted a double-blind study (n=18) of 10 daily active (1 Hz; 80 % MT; five trains; 1 min; 1 min intertrain interval (ITI)) or sham TMS treatments over the left TPC [49]. All patients demonstrated the same pattern of improvement with no difference between active and sham treatment on the PANSS AH items. Lee et al. conducted a double-blind study (left, right, or sham TMS) administering 10 daily treatments (1 Hz; 100 % MT; 20 continuous minutes; 1,600 stimulations) applied bilaterally over the TPC in 39 patients [50]. The study found significant changes in frequency of AHs ($p \le 0.001$) and in PANSS positive symptom scores ($p \le 0.001$) over the course of treatment. The greatest improvement was noted in frequency and attentional salience. The study was also the first to observe that TMS over the right side may be beneficial in improving global symptoms based on the CGI-I Scale ($p \le 0.002$), implying that its effects may spread to the opposite hemisphere.

Jandl et al. compared the effects of TMS (1 Hz; 100 % MT; one continuous 15 min stimulation; 900 stimuli per session) over the left or right TPC for 5 days in a randomized, three-armed, sham-controlled study (n=16) [51]. While PSYRATS change scores were not significantly different between the groups, only the five patients who received active left TPC stimulation achieved complete or partial response. The authors note that the number needed to treat (NNT) (i.e., three) at the end of the left TPC stimulation period falls within the range seen for other psychiatric treatments. The NNT after 4-week follow-up, however, increased to 8.

In a double-blind trial (n=24), Brunelin et al. delivered twice daily active (1 Hz; 90 % MT; 10 sessions; 1,000 stimulations) or sham TMS over the left TPC for 5 days [52]. The authors reported that AHRS scores in the active group were significantly improved versus the sham group ($p \le 0.002$). Positive symptoms as measured by the SAPS, however, did not differ between groups. Finally, the authors were the first to report a trend, noting that as AHs improved, source monitoring performance (e.g., discriminating between internally generated thoughts and externally generated events) also improved.

Rosa et al. conducted a randomized, controlled study in 11 clozapine-resistant patients [53]. TMS (1 Hz; 90 % MT; one continuous 16-min period; 9,600 pulses) was delivered over the left TPC for 10 days. Reality and attentional salience scores on the AHRS significantly decreased at endpoint and at 6 weeks of follow-up in the active TMS group ($p \le 0.049$ and $p \le 0.036$, respectively). By contrast, the sham group's initial linear decrease over time was not sustained after 2 weeks (suggesting a possible placebo effect).

Vercammen et al. randomly assigned 38 patients to TMS (1 Hz; 90 % MT; 20 min of continuous stimulation; 12 sessions with a minimum of 5 h delay between sessions; 14,400 pulses) on six consecutive work days [54]. Patients received

bilateral TPC, left TPC, or a sham condition. The authors reported no difference in total AHRS change scores between the stimulation conditions; however, the frequency subscale of AHs was significantly reduced in the left and bilateral TPC groups only after the 1-week follow-up assessment (p < 0.043 and p < 0.043, respectively).

In a double-blind, crossover trial (n=18), Loo et al. assessed the effects of TMS over the right and left temporal cortex versus the vertex (control site) using low frequency stimulation (1 Hz; 110 % MT) [55]. Participants received three consecutive days of TMS over each site (with a gap of 4 days in between stimulation conditions) for 4 min on days 1–2 and 8 min on day 3. There was a nonsignificant trend for decreasing the quality of distress level on the AHRS in the active versus sham group (p<0.087). There was no difference between stimulation over the left or right temporal cortex.

de Jesus et al. conducted a randomized study in 17 patients with clozapinerefractory schizophrenia who received active TMS (1 Hz; 90 % MT; 20 sessions over 4 weeks) over the left TPC, while the sham procedure angled the coil at 45° and reduced intensity to 80 % MT [56]. Patients received one continuous 8-min stimulation period on day 1, 16 min on day 2, and 20 min daily for the next 18 days. Changes in BPRS scores from baseline to endpoint were significantly greater in the active versus sham group ($p \le 0.002$). This study also found a positive effect in general psychopathology.

Meta-analyses of TMS for Positive Symptoms

Four meta-analyses summarized these results, reporting moderate to high mean ES ranging from 0.54 to 1.0 [57–60]. In their most recent meta-analysis, Slotema et al. included 15 randomized controlled studies utilizing TMS over the left TPC (n=337) [60]. The mean weighted ES was 0.44 (i.e., moderate effect). In two additional studies (whose primary site of stimulation was not the left TPC; n=122), a mean weighted ES of 0.33 was obtained. This analysis also looked into the long-term effects of low frequency TMS for AHs 1 month after the end of treatment. Five studies (n=118) were included and produced a mean weighted ES of 0.40. All of these studies applied TMS over the left TPC.

A recent review of five meta-analyses considering TMS for treatment of schizophrenia was published by Hovington et al. [61]. The authors report that the data for AVHs support the efficacy of active versus sham TMS when applied over the left TPC (i.e., ES ranged from 0.54 to 1.04). When considering global positive symptoms, the ESs were more modest and varied from 0.17 to 0.51. Of note, when the SANS was the primary outcome measure (versus the PANSS), it appeared to be more sensitive in detecting changes in negative symptoms (i.e., ES=0.73 and E=0.35, respectively). The authors concluded that TMS over the left TPC was the most common site of stimulation for AVHs, using low frequencies and treatment durations lasting between 4 and 15 days (pulses per session ranged from 120 to 2,000). Finally, the authors note that given the limited number of RCTs for positive symptoms, "there does not appear to be a dose-dependent increase in effect sizes with more treatment sessions." Overall, the procedure was well tolerated and safe.

TMS for Adolescents

In a case series of childhood-onset schizophrenia (COS), ten adolescents underwent twice daily TMS treatment (1 Hz; 90 % MT; 1,200 continuous pulses) applied over the left TPC for five consecutive days [62]. AHRS scores significantly decreased from baseline to acute endpoint and from baseline to 1-month posttreatment (i.e., $p \le 0.007$ and $p \le 0.004$, respectively). GAF scores were also significantly improved at both time points (i.e., $p \le 0.002$ and $p \le 0.009$, respectively). The authors concluded that larger controlled trials are needed to validate these results.

TMS for Maintenance Treatment

There is limited evidence regarding the use of maintenance TMS in schizophrenia. Thirthalli et al. described a patient with clozapine-treated refractory AHs [63]. This patient underwent an acute course of TMS over the left TPC (1 Hz; 100 % MT; once daily five times a week; 900 continuous stimulation) followed by once weekly maintenance treatment for 6 weeks, then once every 2 weeks for 3 months, and then once every 4 weeks for 3 months. The patient reported almost total remission of AHs after week 4 with frequency, intensity, duration, intrusiveness, and overall severity all improved.

TMS Administration

While many TMS studies demonstrate efficacy for treating schizophrenia, there are several limitations. For example, in the Fitzgerald, McIntosh, and Loo studies, the authors suggest that the pause within the stimulation trains may contribute to a reduction in efficacy compared with other protocols which did not employ these breaks [44, 46, 55]. In addition, the Hoffman and Vercammen studies included patients who were on anticonvulsants which may attenuate the effects of TMS [41, 54]. In this context, Poulet et al. noted that none of their seven responders received anticonvulsant drugs [47]. A challenge to device-based research is that the putative neuroanatomical substrate for hallucinations may differ between subjects, making one stimulation target impractical for all patients. It may also be that only a subgroup of patients with schizophrenia benefit from this approach. Another problem is the difficulty in creating an adequate sham procedure.

Conclusion

Despite the limitations noted, low frequency TMS over the TPC holds promise as a potential treatment for refractory AHs and possibly other positive symptoms. While it is difficult to identify a definitive set of ideal treatment parameters, Table 8.3 provides a general summary of parameters which were successful in the treatment of auditory hallucinations.

Table 8.3 Parameters used with adjunctive TMS for auditory hallucinations	Frequency Stimulation location Duration Number of pulses per session	1 Hz Left temporoparietal cortex 4–15 days 12–2,000
	Number of pulses per session	12–2,000
	Motor threshold	80–90 %

Negative Symptoms

Electroconvulsive Therapy

Few trials with ECT report a formal outcome measure for negative symptoms (e.g., the SANS). In an open-label, add-on trial of 8–20 ECT sessions in 15 patients with schizophrenia on a stable antipsychotic regimen, Tang et al. reported a significant improvement in SANS scores ($p \le 0.05$) [64]. By contrast, no improvement in negative symptoms was observed in 253 patients with treatment-resistant schizophrenia who received a combination of ECT and flupenthixol [9]. Matheson et al. analyzed five systematic reviews conducted since 2000 [14]. They concluded the evidence supported a "medium size" benefit for overall clinical improvement as measured by the BPRS in schizophrenia, but the authors did not comment on ECT's specific effect for negative symptoms.

Transcranial Magnetic Stimulation

Mechanism of Action

Since several neuroimaging studies indicate an association between negative symptoms and frontal cortical hypoactivity, the potential for TMS to excite the dorsolateral prefrontal cortex (DLPFC) is the rationale for its treatment of these symptoms [65].

Open-Label Studies

Treatment with TMS for negative symptoms was first reported in six inpatients with schizophrenia on a stable antipsychotic dose for 3 months [66]. Ten daily TMS treatments (20 stimulation trains; 2 s duration; 20 Hz; 80 % MT; 58 s ITI) over the prefrontal cortex (side not defined) decreased PANSS negative subscale scores significantly ($p \le 0.02$). Subsequently, an open-label study administered TMS (20 stimulation trains; 3.5 s duration; 10 Hz; 100 % MT; 10 s ITI) over the left DLPFC to 10 patients with schizophrenia and prominent negative symptoms [67]. Mean baseline SANS scores decreased from 49.0 (±10.7) to 44.7 (±11.8) posttreatment ($p \le 0.001$). In another study, Sachdev et al. administered daily TMS (24 stimulation trains; 5 s duration; 15 Hz; 80 % MT; 25 s ITI) over the left DLPFC for 20 days to four patients with

schizophrenia [68]. PANSS negative subscale scores decreased from a mean of 29.25 at baseline to 19.5 following treatment (*p* value not reported). In 27 patients, Jin et al. provided 2 weeks of daily bilateral TMS (20 stimulation trains; 2 s duration; 80 % MT; 58 s ITI) over the DLPFC at 3 Hz, 20 Hz, or a stimulation set individually at each subject's EEG-derived alpha frequency [69]. A subject's alpha wave frequency refers to any rhythmic activity detected between 8 and 12 Hz on an EEG. The alpha frequency-derived stimulus produced the greatest reduction in PANSS negative symptom subscale scores (i.e., 29.6 %) versus the other two groups (i.e., <9 %; $p \le 0.007$).

Controlled Studies

The majority of the published, sham-controlled studies applied TMS over the left DLPFC, since this brain region is believed to play a role in negative symptoms.

Holi et al. randomized 22 medicated inpatients with schizophrenia in doubleblind fashion to either ten daily TMS treatments (20 stimulation trains; 5-s duration; 10 Hz; 100 % MT; 30 s ITI) or a sham procedure [70]. No significant improvement in PANSS negative subscale scores was detected in the active versus sham groups. Hajak et al. employed identical TMS treatment parameters (except 110 % MT) and duration in a similar study of 20 inpatients with schizophrenia or schizoaffective disorder on stable doses of antipsychotics [71]. While significant improvement in the PANSS negative subscale scores was reported for the active versus sham group, neither raw scores nor p values were provided.

Novak et al. applied 10 days of higher frequency TMS (40 stimulation trains; 2.5 s duration; 20 Hz; 90 % MT; 30 s ITI) to 16 patients with schizophrenia [72]. Neither active nor sham TMS produced any benefit. Mogg et al. randomized 17 subjects with schizophrenia in a double-blind fashion to either 10 daily active (20 stimulation trains; 10 s duration; 10 Hz; 110 % MT; 50 s ITI) or sham TMS treatments [73]. Again, no significant difference was detected between the active or sham groups for the primary outcome measure (i.e., PANSS negative subscale score).

When Prikryl et al. administered a greater number of treatments, however, benefit for negative symptoms was observed with active TMS [74]. Twenty-two patients with schizophrenia were randomized to active (15 stimulation trains; 10 s duration; 10 Hz; 110 % MT; 30 s ITI) or sham TMS for 15 consecutive daily treatments. Active TMS was associated with significant improvements (i.e., 29 % reduction in the PANSS negative subscale score and a 50 % reduction in the SANS score $(p \le 0.01$ and $p \le 0.01$, respectively)).

Goyal et al. randomized 10 patients with schizophrenia to 10 days of active (10 stimulation trains; 4.9 s duration; 10 Hz; 110 % MT; 30 s ITI) or sham TMS [75]. An ANOVA between-groups analysis demonstrated significant improvement in the PANSS negative symptom scale scores with active versus sham TMS ($p \le 0.008$). Unique to this study is that the authors included depression assessments (Calgary Depression Scale Schizophrenia (CDSS)) at baseline and follow-up to evaluate the possible confounding effect of mood on improvements in negative symptoms. Changes in negative symptoms scores, however, did not correlate with changes in the CDSS.

In one of the largest and longest trials to date, Schneider et al. randomized 51 patients with schizophrenia to active (20 stimulation trains; 5 s duration; 1 or 10 Hz; 110 % MT; 15 s ITI) or sham TMS [76]. After 20 daily treatments, a significantly greater improvement on the SANS was observed in the 10 Hz group compared with the 1 Hz and sham groups ($p \le 0.031$) which persisted 4 weeks posttreatment. No significant changes were noted on the HAM-D, again providing support that the observed changes were not attributable to improvements in depression.

Other investigators have employed bilateral and right-sided TMS for negative symptoms, but neither was found efficacious [77, 78]. Finally, Levkovitz applied deep TMS (20 Hz; 120 % MT) over the lateral prefrontal cortex in 15 patients and observed a mean SANS score decrease of 16.83 % after 20 daily treatments [79].

Meta-analyses of TMS for Negative Symptoms

TMS for the treatment of negative symptoms was evaluated in two meta-analyses. Freitas and colleagues included only those trials employing greater than one session of high frequency stimulation over the left DLPFC [80]. The final analysis included 107 patients. The authors reported an ES of 0.58 using a random effects model to analyze only the active arms of all studies, while a mixed effect model revealed an ES of 0.49. They concluded that high frequency TMS over the left DLPFC provided a "modest to moderate" reduction in negative symptoms. When only considering sham-controlled data, however, the pooled weighted ES was 0.27 with no significant difference in improvement for negative symptoms observed in patients receiving active versus sham stimulation. Of note, lack of sufficient data led to two positive trials being excluded from this analysis. Dlabac-de Lange et al. included two previously unpublished trials in a more recent meta-analysis [81]. Thus, the authors analyzed nine controlled trials involving 213 patients. Inclusion of only those studies utilizing high frequency (>10 Hz) TMS generated a mean ES of 0.63 (96 % CI 0.11–1.15) with the authors concluding that TMS for negative symptoms warrants further study.

Conclusion

As with TMS for positive symptoms, the preliminary data are inconsistent but promising and await more definitive studies which are adequately powered and well designed.

Cognitive Symptoms

Transcranial Magnetic Stimulation

Cognitive symptoms in schizophrenia impair function, adversely impact quality of life, and substantially contribute to overall disability. Further, antipsychotic medications are usually insufficient to control these symptoms. In this context, TMS has been

considered as a possible option. Fitzgerald et al., however, did not detect cognitive improvement after 10 sessions of daily TMS over the left temporal-parietal region [44]. A review including three studies evaluating cognition in schizophrenia treated with TMS also found no improvement in cognitive speed, Mini Mental State Exam scores, digit span, or verbal fluency [82]. A study primarily focused on treating negative symptoms (Mogg et al., described earlier) did report significant improvement in verbal learning [73]. In the Levkovitz et al. study described earlier, the application of open-label, bilateral deep TMS (20 Hz; 120 % MT) in 15 subjects also resulted in improvements in executive function (i.e., spatial working memory ($p \le 0.001$); sustained attention ($p \le 0.01$)) [79].

In more recent studies, Guse et al. applied TMS (10 Hz; 110 % MT) over the left posterior middle frontal gyrus daily for 15 days [83]. No improvement was noted in the verbal letter 2-back task, a measure of working memory. Further, Barr et al. randomized 27 patients on antipsychotics to 20 daily active (25 stimulation trains; 1.5 s duration; 20 Hz; 90 % MT; 30 s ITI; 1,500 pulses/session) or sham TMS sessions bilaterally over the DLPFC [84]. Compared with the sham procedure, patients receiving active TMS significantly improved on the verbal working n-back task (Cohen's d=0.92).

Conclusion

Future studies need to clarify the ideal neuroanatomic targets for TMS which may produce improvement in various aspects of cognition.

Other Neuromodulation Approaches

Transcranial Direct Current Stimulation

Transcranial direct current stimulation (tDCS) produces a weak current flow through the cerebral cortex via two scalp electrodes: the anode which increases neuronal activity and the cathode which reduces activity [85]. The technique was established in the 1950s, primarily in animal studies. Since the 1960s, tDCS has been studied for its effects on mood and depressive symptoms in humans with promising results.

tDCS for Acute Psychotic Symptoms

tDCS for treatment of schizophrenia is based on the premise that excitatory stimulation delivered via the anodal electrode over the left DLPFC may improve negative symptoms. Conversely, inhibitory stimulation delivered via the cathodal electrode over the left TPC could reduce AHs. In a report of two patients, Brunelin et al. utilized tDCS to treat refractory symptoms of schizophrenia [86]. The authors hypothesize that with tDCS, cortical excitability will be increased in the vicinity of the anodal electrode (analogous to high frequency TMS) and decreased near the cathodal electrode (analogous to low frequency TMS). Stimulations were delivered at an intensity of 2 mA for 20 min twice a day (with a 3 h break in between) for five consecutive days. A significant decrease in AHRS scores and a slight improvement in the total PANSS scores were noted (no p value reported). Further, clinical benefit was maintained over 3 months.

Brunelin et al. expanded their initial study by investigating the efficacy of tDCS in reducing the severity of AH and negative symptoms in a randomized, double-blind study (n=30) [87]. Twice daily stimulations (2 mA; 20 min; 3 h break in between) were also conducted on five consecutive days. Active tDCS resulted in a significant improvement in AHRS scores versus the sham procedure ($p \le 0.001$). Further, improvements remained significant at 1 month ($p \le 0.001$) and 3 months ($p \le 0.001$) posttreatment. A significant ES was observed for the PANSS negative subscale score (d=1.07), while the positive and depressive subscale scores demonstrated medium ESs (i.e., d=0.64 and d=0.61, respectively). The treatment was well tolerated with the most common adverse events being a transient mild tingling or itching sensation at the beginning of stimulation.

tDCS for Maintenance Treatment

In terms of maintenance treatment using tDCS, there is one case report of a clozapine-refractory patient with severe AHs whose symptoms improved after 1–2 daily stimulations (1–3 mA; 30 min) over 3 years of treatment at the patient's residence with frequency determined by her day-to-day needs [88]. Improvements in cognitive, psychosocial, and occupational functioning were maintained over this period. Adequately powered maintenance studies are needed to confirm this observation.

Conclusion

While initial pilot data is promising, large, well-designed, controlled studies are needed to clarify the potential role of tDCS in treating schizophrenia.

Deep Brain Stimulation

Deep brain stimulation (DBS) involves the implantation of stimulating electrodes into brain sites such as the thalamus for essential tremor or the globus pallidus internus or the subthalamic nucleus for Parkinson's disease and intractable primary dystonia [89]. Depending on the brain target site, DBS can exert excitatory (by facilitating neuronal activity/conduction) or inhibitory (by blocking neuronal activity/conduction) effects. To our knowledge, its only use in schizophrenia involves a case report of a woman whose comorbid OCD symptoms improved after receiving DBS, with no worsening of psychosis [90].

Conclusion

ECT is currently used to treat refractory positive and mood symptoms in schizophrenia, as well as catatonia. While large definitive studies are lacking, its usefulness for these indications has been demonstrated over many years in clinical practice. TMS may benefit both positive and negative symptoms, but presently remains investigational. Further studies with larger samples, optimized parameters, and adequate number of treatments are needed. The use of tDCS also appears promising in treating refractory symptoms, but further studies with larger samples are needed to confirm these pre-liminary results. Finally, there is an inadequate database to suggest a therapeutic role for DBS or other device-based approaches in treating schizophrenia.

References

- 1. Shergill SS, Murray RM, McGuire PK. Auditory hallucinations: a review of psychological treatments. Schizophr Res. 1998;32(3):137–50.
- Kane JM. Clinical efficacy of clozapine in treatment-refractory schizophrenia: an overview. Br J Psychiatry. 1992;160(17):41–5.
- 3. Cerletti U. Old and new information about electroshock. Am J Psychiatry. 1950;107(2):87-94.
- Haskett RF, Loo C. Role of psychotropic medications during ECT in the treatment of depression, mania and schizophrenia. J ECT. 2010;26(3):196–201.
- 5. Taylor S. Electroconvulsive therapy: a review of history, patient selection, technique, and medication management. South Med J. 2007;100(5):494–8.
- Bolwig TG. How does electroconvulsive therapy work? Theories on its mechanism. Can J Psychiatry. 2011;56(1):13–8.
- 7. Goswani U, Kumar U, Singh B. Efficacy of electroconvulsive therapy in treatment resistant schizophrenia: a double-blind study. Indian J Psychiatry. 2003;45(1):26–9.
- Kupchik M, Spivak B, Mester R, et al. Combined electroconvulsive-clozapine therapy. Clin Neuropharmacol. 2000;23:14–6.
- 9. Chanpattana W, Somchai Chakrabhand ML. Combined ECT and neuroleptic therapy in treatment-refractory schizophrenia: prediction of outcome. Psychiatry Res. 2001;105:107–15.
- Nothdurfter C, Eser D, Schule C, et al. The influence of concomitant neuroleptic medication on safety, tolerability and clinical effectiveness of electroconvulsive therapy. World J Biol Psychiatry. 2006;7(3):162–70.
- 11. Suppes T, Webb A, Carmody T, et al. Is postictal electrical silence a predictor of response to electroconvulsive therapy? J Affect Disord. 1996;41:55–8.
- 12. Tharyan P, Adams CE. Electroconvulsive therapy for schizophrenia. Cochrane Database Syst Rev. 2005;2:CD000076
- Painuly N, Chakrabarti S. Combined use of electroconvulsive therapy and antipsychotics in schizophrenia: the Indian evidence. A review and a meta-analysis. J ECT. 2006;22:59–66.

- Matheson SL, Green MJ, Loo C, Carr VJ. Quality assessment and comparison of evidence for electroconvulsive therapy and repetitive transcranial magnetic stimulation for schizophrenia: a systematic meta-review. Schizophr Res. 2010;118:201–10.
- 15. Gazdag G, Ungvari GS. Non-pharmacological biological therapies in schizophrenia. Neuropsychopharmacol Hung. 2011;13(4):233–8.
- Zervas IM, Theleritis C, Soldatos CR. Using ECT in schizophrenia: a review from a clinical perspective. World J Biol Psychiatry. 2012;13:96–105.
- 17. Rosebush PI, Mazurek MF. Catatonia and its treatment. Schizophr Bull. 2009;36(2):239-42.
- Makhinson M, Furst BA, Shuff MK, Kwon GE. Successful treatment of co-occurring catatonia and obsessive-compulsive disorder with concurrent electroconvulsive therapy and benzodiazepine administration. J ECT. 2012;28(3):e35–6.
- Häßler F, Reis O, Weirich S, Hoppner J, Pohl B, Buchmann J. A case of catatonia in a 14-yearold girl with schizophrenia treated with electroconvulsive therapy. Z Kinder Jugendpscychiatr Psychother. 2013;41(1):69–74.
- Rohland BM, Caroll BT, Jacoby RG. ECT in the treatment of the catatonic syndrome. J Affect Disord. 1993;29(4):255–61.
- Escobar R, Rios A, Montoya ID, et al. Clinical and cerebral blood flow changes in catatonic patients treated with ECT. J Psychosom Res. 2000;49:423–9.
- Girish K, Gill NS. Electroconvulsive therapy in lorazepam non-responsive catatonia. Indian J Psychiatry. 2003;45(1):21–5.
- Suzuki K, Awata S, Takano T, et al. Improvement of psychiatric symptoms after electroconvulsive therapy in young adults with intractable first-episode schizophrenia and schizophreniform disorder. J Exp Med. 2006;210:213–20.
- Baeza I, Flamarique I, Garrido JM, et al. Clinical experience using electroconvulsive therapy in adolescents with schizophrenia spectrum disorders. J Child Adolesc Psychopharmacol. 2010;20(3):205–9.
- 25. Chanpattana W, Sackeim HA. Electroconvulsive therapy in treatment-resistant schizophrenia. J ECT. 2010;26(4):289–98.
- Levy-Rueff M, Gourevitch R, Loo H, Olie JP, Amado I. Maintenance electroconvulsive therapy: an alternative treatment for refractory schizophrenia and schizoaffective disorders. Psychiatry Res. 2010;175:280–3.
- Kristensen D, Bauer J, Hageman I, Jorgensen MB. Electroconvulsive therapy for treating schizophrenia: a chart review of patients from two catchment areas. Eur Arch Psychiatry Clin Neurosci. 2011;261:425–32.
- Chanpattana W, Andrade C. ECT for treatment-resistant schizophrenia: a response from the far east to the UK nice report. J ECT. 2006;22:4–12.
- Lerer B, Shapira B, Calev A, et al. Antidepressant and cognitive effects of twice-versus threetimes-weekly ECT. Am J Psychiatry. 1995;152(4):564–70.
- 30. Charlson F, Siskind D, Doi SAR, et al. ECT efficacy and treatment course: a systematic review and meta-analysis of twice vs thrice weekly schedules. J Affect Disord. 2012;138:1–8.
- Chanpattana W, Charabhand MLS, Kitaroonchai W, et al. Effects of twice- versus thriceweekly electroconvulsive therapy in schizophrenia. J Med Assoc Thai. 1999;82:477–83.
- Pompili M, Lester D, Dominici G, et al. Indications for electroconvulsive treatment in schizophrenia: a systematic review. Schizophr Res. 2013;146(1–3):1–9.
- 33. American Psychiatric Association Committee on Electroconvulsive Therapy. The practice of electroconvulsive therapy: recommendations for treatment, training, and privileging (A task force report of the American Psychiatric Association). 2nd ed. Washington, DC: American Psychiatric Association; 2001.
- Buchannan RW, Kreyenbuhl J, Kelly DL, et al. The 2009 schizophrenia PORT psychopharmacological treatment recommendations and summary statements. Schizophr Bull. 2010;36:71–93.
- 35. Post A, Keck ME. Transcranial magnetic stimulation as a therapeutic tool in psychiatry: what do we know about the neurobiological mechanism? J Psychiatr Res. 2001;35:193–215.

- 36. Slotema CW, Blom JD, Hoek HW, Sommer IEC. Should we expand the toolbox of psychiatric treatment methods to include repetitive transcranial magnetic stimulation? A meta-analysis of the efficacy of rTMS in psychiatric disorders. J Clin Psychiatry. 2010;71(7):873–84.
- Jasper HH. The ten-twenty electrode system of the International Federation. Electroencephalogr Clin Neurophysiol. 1958;66:376–82.
- Geller V, Grisaru N, Abarbanel JM, Lemberg T, Belmaker RH. Slow magnetic stimulation of prefrontal cortex in depression and schizophrenia. Prog Neuropsychopharmacol Biol Psychiatry. 1997;21:105–10.
- Feinsod M, Kreinin B, Chistyakov A, Klein E. Preliminary evidence for a beneficial effect of low-frequency, repetitive transcranial magnetic stimulation in patients with major depression and schizophrenia. Depress Anxiety. 1998;7:65–8.
- Hoffman RE, Boutros NN, Berman RM, et al. Transcranial magnetic stimulation of left temporoparietal cortex in three patients reporting hallucinated "voices". Biol Psychiatry. 1999;46:130–2.
- Hoffman RE, Boutros NN, Hu S, Berman RM, Krystal JH, Charney DS. Transcranial magnetic stimulation and auditory hallucinations in schizophrenia. Lancet. 2000;355:1073–5.
- Hoffman RE, Hawkins KA, Gueorguieva R, et al. Transcranial magnetic stimulation of left temporoparietal cortex and medication-resistant auditory hallucinations. Arch Gen Psychiatry. 2003;60:49–56.
- Hoffman RE, Gueorguieva R, Hawkins KA, et al. Temporoparietal transcranial magnetic stimulation for auditory hallucinations: safety, efficacy and moderators in a fifty patient sample. Biol Psychiatry. 2005;58:97–104.
- 44. Fitgerald PB, Benitez J, Daskalakis JZ, et al. A double-blind sham-controlled trial of repetitive transcranial magnetic stimulation in the treatment of refractory auditory hallucinations. J Clin Psychopharmacol. 2005;25(4):358–62.
- 45. Hoffman RE, Wu K, Pittman B, et al. Transcranial magnetic stimulation of Wernicke's and right homologous sites to curtail "voices": a randomized trial. Biol Psychiatry. 2013;73:1008–14.
- 46. McIntosh AM, Semple D, Tasker K, et al. Transcranial magnetic stimulation for auditory hallucinations in schizophrenia. Psychiatry Res. 2004;127:9–17.
- 47. Poulet E, Brunelin J, Bediou B, et al. Slow transcranial magnetic stimulation can rapidly reduce resistant auditory hallucinations in schizophrenia. Biol Psychiatry. 2005;57:188–91.
- 48. Chibarro G, Daniele M, Alagona G, et al. Repetitive transcranial magnetic stimulation in schizophrenic patients reporting auditory hallucinations. Neurosci Lett. 2005;383:54–7.
- 49. Saba G, Verdon CM, Kalalou K, et al. Transcranial magnetic stimulation in the treatment of schizophrenic symptoms: a double blind sham controlled study. J Psychiatr Res. 2006;40: 147–52.
- 50. Lee SH, Kim W, Chung YC, et al. A double blind study showing that two weeks of daily repetitive TMS over the left or right temporoparietal cortex reduces symptoms in patients with schizophrenia who are having treatment-refractory auditory hallucinations. Neurosci Lett. 2005;376:177–81.
- Jandl M, Steyer J, Weber M, et al. Treating auditory hallucinations by transcranial magnetic stimulation: a randomized controlled cross-over trial. Neuropsychobiology. 2006;53:63–9.
- Brunelin J, Poulet E, Bediou B, et al. Low frequency repetitive transcranial magnetic stimulation improves source monitoring deficit in hallucinating patients with schizophrenia. Schizophr Res. 2006;81:41–5.
- Rosa MO, Gattaz WF, Rosa MA, et al. Effects of repetitive transcranial magnetic stimulation on auditory hallucinations refractory to clozapine. J Clin Psychiatry. 2007;68(10):1528–32.
- 54. Vercammen A, Knegtering H, Bruggeman R, et al. Effects of bilateral repetitive transcranial magnetic stimulation on treatment resistant auditory-verbal hallucinations in schizophrenia: a randomized controlled trial. Schizophr Res. 2009;114:172–9.
- Loo CK, Sainsbury K, Mitchell P, Hadzi-Pavlovic D, Sachdev PS. A sham controlled trial of left and right temporal rTMS for the treatment of auditory hallucinations. Psychol Med. 2010;40:541–6.

- 56. de Jesus DR, Gil A, Barbosa L, et al. A pilot double-blind sham-controlled trial of repetitive transcranial magnetic stimulation for patients with refractory schizophrenia treated with clozapine. Psychiatry Res. 2011;188:203–7.
- Aleman A, Sommer IE, Khan RS. Efficacy of slow repetitive transcranial magnetic stimulation in the treatment of resistant auditory hallucinations in schizophrenia: a meta-analysis. J Clin Psychiatry. 2007;68(3):416–21.
- Tranulis C, Sepehry AA, Galinowski A, Stip E. Should we treat auditory hallucinations with repetitive transcranial magnetic stimulation? A meta-analysis. Can J Psychiatry. 2008;53(9):577–86.
- 59. Frietas C, Fregni F, Pascual-Leone A. Meta-analysis of the effects of repetitive transcranial magnetic stimulation on negative and positive symptoms in schizophrenia. Schizophr Res. 2009;108:11–24.
- 60. Slotema CW, Aleman A, Daskalakis ZJ, Sommer IE. Meta-analysis of repetitive transcranial magnetic stimulation in the treatment of auditory verbal hallucinations: update and effects after one month. Schizophr Res. 2012;42:40–5.
- Hovington CL, McGirr A, Lepage M, Berlim MT. Repetitive transcranial magnetic stimulation (rTMS) for treating major depression and schizophrenia: a systematic review of recent metaanalyses. Ann Med. 2013;45(4):308–21.
- 62. Jardri R, Bubrovszky M, Demeulemeester M, et al. Repetitive transcranial magnetic stimulation to treat early-onset auditory hallucinations. J Am Acad Child Adolesc Psychiatry. 2012;51(9):947–9.
- 63. Thirthalli J, Bharadway B, Kulkarni S, Gangadhar BN, Kharawala S, Andrade C. Successful use of maintenance rTMS for 8 months in a patient with antipsychotic-refractory auditory hallucinations. Schizophr Res. 2008;100:351–2.
- Tang WK, Ungvari GS. Efficacy of electroconvulsive therapy combined with antipsychotic medication in treatment-resistant schizophrenia: a prospective, open trial. J ECT. 2002;18(2):90–4.
- 65. Rado JT. Management treatment of negative symptoms in schizophrenia. Psychopharm Rev. 2011;46(5):33–40.
- 66. Cohen E, Bernardo M, Mansana J, et al. Repetitive transcranial magnetic stimulation in the treatment of chronic negative schizophrenia: a pilot study. J Neurol Neurosurg Psychiatry. 1999;67(1):129–30.
- 67. Jandl M, Bittner R, Sack A, et al. Changes in negative symptoms and EEG in schizophrenic patients after repetitive transcranial magnetic stimulation (rTMS): an open-label pilot study. J Neural Transm. 2005;112:955–67.
- Sachdev P, Loo C, Mitchell P, Malhi G. Transcranial magnetic stimulation for the deficit syndrome of schizophrenia: a pilot investigation. Psychiatry Clin Neurosci. 2005;59(3):354–7.
- 69. Jin Y, Potkin SG, Kemp AS, et al. Therapeutic effects of individualized alpha frequency transcranial magnetic stimulation (αTMS) on the negative symptoms of schizophrenia. Schizophr Bull. 2006;32(3):556–61.
- Holi MM, Eronen M, Toivonev K, Toivonen P, Marttunen M, Naukkarinen H. Left prefrontal repetitive transcranial magnetic stimulation in schizophrenia. Schizophr Bull. 2004;30(2): 429–34.
- Hajak G, Marienhagen J, Langguth B, Werner S, Binder H, Eichhammer P. High-frequency repetitive transcranial magnetic stimulation in schizophrenia: a combined treatment and neuroimaging study. Psychol Med. 2004;34:1157–63.
- Novak T, Horacek J, Mohr P, et al. The double-blind sham-controlled study of high-frequency rTMS (20 Hz) for negative symptoms in schizophrenia: negative results. Neuro Endocrinol Lett. 2006;27(1–2):209–13.
- 73. Mogg A, Purvis R, Eranti S, et al. Repetitive transcranial magnetic stimulation for negative symptoms of schizophrenia: a randomized controlled pilot study. Schizophr Res. 2007;93: 221–8.
- 74. Prikryl R, Kasparek T, Skotakova S, Ustohal L, Kucerova H, Ceskova E. Treatment of negative symptoms of schizophrenia using repetitive transcranial magnetic stimulation in a doubleblind, randomized controlled study. Schizophr Res. 2007;95:151–7.

- 75. Goyal N, Nizamie SH, Desarkar P. Efficacy of adjuvant high frequency repetitive transcranial magnetic stimulation on negative and positive symptoms of schizophrenia: preliminary results of a double-blind sham-controlled study. J Neuropsychiatry Clin Neurosci. 2007;19:464–7.
- 76. Schneider AL, Schneider TL, Stark H. Repetitive transcranial magnetic stimulation (rTMS) as an augmentation treatment for the negative symptoms of schizophrenia: a 4-week randomized placebo controlled study. Brain Stimul. 2008;1:106–11.
- Klein E, Kolsky Y, Puyerovsky M, Koren D, Chistyakov A, Feinsod M. Right prefrontal slow repetitive transcranial magnetic stimulation in schizophrenia: a double-blind sham-controlled pilot study. Biol Psychiatry. 1999;46:1451–4.
- 78. Fitzgerald PB, Herring S, Hoy K, et al. A study of the effectiveness of bilateral transcranial magnetic stimulation in the treatment of the negative symptoms of schizophrenia. Brain Stimul. 2008;1:27–32.
- Levkovitz Y, Rabany L, Harel EV, Zangen A. Deep transcranial magnetic stimulation add-on for treatment of negative symptoms and cognitive deficits of schizophrenia: a feasibility study. Int J Neuropsychopharmacol. 2011;14(7):991–6.
- Freitas C, Fregni F, Pascual-Leone A. Meta-analysis of the effects of repetitive transcranial magnetic stimulation (rTMS) on negative and positive symptoms in schizophrenia. Schizophr Res. 2009;108:11–24.
- Blabac-de Lange JJ, Knegtering R, Aleman A. Repetitive transcranial magnetic stimulation for negative symptoms of schizophrenia: review and meta-analysis. J Clin Psychiatry. 2010;71(4):411–8.
- 82. Guse B, Falkai P, Wobrock T. Cognitive effects of high-frequency repetitive transcranial magnetic stimulation: a systematic review. J Neural Transm. 2010;117(1):105–22.
- 83. Guse B, Falkai P, Gruber O, et al. The effect of long-term high frequency repetitive transcranial magnetic stimulation on working memory in schizophrenia and healthy controls—a randomized placebo-controlled, double-blind fMRI study. Behav Brain Res. 2013;237:300–7.
- 84. Barr MS, Farzan F, Rajji TK, et al. Can repetitive magnetic stimulation improve cognition in schizophrenia? Pilot data from a randomized controlled trial. Biol Psychiatry. 2013; 73(6):510–7.
- Nitsche MA, Baggio PS, Fregni F, Pascual-Leone A. Treatment of depression with transcranial direct current stimulation (tDCS): a review. Exp Neurol. 2009;219:14–9.
- Brunelin J, Mondino M, Haesebaert F, et al. Efficacy and safety of bifocal tDCS as an interventional treatment for refractory schizophrenia. Brain Stimul. 2011;5(3):431–2.
- Brunelin J, Mondino M, Gassab L, et al. Examining transcranial direct-current stimulation (tDCS) as a treatment for hallucinations in schizophrenia. Am J Psychiatry. 2012;169(7): 719–24.
- Andrade C. Once- to twice-daily, 3 year domiciliary maintenance transcranial direct current stimulation for severe, disabling, clozapine-refractory continuous auditory hallucinations in schizophrenia. J ECT. 2013;29(3):239–42.
- 89. Coffey RJ. Deep brain stimulation devices: a brief technical history and review. Artif Organs. 2008;33(3):208–20.
- Plewnia C, Schober F, Rilk A, et al. Sustained improvement of obsessive-compulsive disorder by deep brain stimulation in a woman with residual schizophrenia. Int J Neuropsychopharmacol. 2008;11:1181–3.

Chapter 9 Pharmacogenetics in the Treatment of Schizophrenia

Bill W. Massey, Jiang Li, and Herbert Y. Meltzer

Introduction

Pharmacogenetics/pharmacogenomics is the science of interindividual genetic differences which contribute to variation in drug efficacy and adverse effects [1]. Understanding the role of genetic and epigenetic variation in drug response is key to achieving the goal of personalized medicine for mental disorders. As yet, there are no commonly available laboratory methods to establish diagnosis. For any psychiatric diagnosis, diverse pharmacologic and non-pharmacologic treatments are available. Further, with the emphasis on diagnosis in decline despite the 2013 publication of the American Psychiatric Association Diagnostic and Statistical Manual of Mental Disorders—5th edition, the treatment focus is shifting to specific domains of psychopathology and behaviors (e.g., delusions, negative symptoms, specific types of cognitive impairment, suicide) which often are relevant to various diagnostic categories (see Chap. 2).

Even though psychotropic drugs are usually approved by regulatory authorities (e.g., the US Food and Drug Administration) for specific diagnoses, they are often prescribed off-label based on symptoms rather than diagnosis. For example, second-generation antipsychotics (SGAs) are more widely used for indications other than schizophrenia, and only recently is their labeling being extended to some of these other indications. There is also great heterogeneity in response to psychotropic drugs. Since biological variation is largely due to individual genetic and epigenetic factors, the identification and use of genetic biomarkers offer great promise for providing an effective means to identify the most appropriate drug for a given patient.

e-mail: bill.massey@northwestern.edu; jiang.li@northwestern.edu; h-meltzer@northwestern.edu

B.W. Massey (🖂) • J. Li • H.Y. Meltzer

Department of Psychiatry and Behavioral Sciences, Northwestern University, Feinberg School of Medicine, Chicago, IL, USA

Identification of these markers for APD response utilizes candidate gene, genome wide association studies (GWAS), whole exome, and theoretically, whole genome sequencing. In addition, there is an array of databases available to validate the biological significance of putative genetic markers. These include BrainCloud (braincloud.jhmi.edu), Brainspan (www.brainspan.org), ENCODE (encodeproject. org/ENCODE/), and other bioinformatics. Reductions in the cost of such testing make it economically feasible to intensively investigate biomarkers for psychotropic drugs. Current commercial chips have as many as 5,000,000 single-nucleotide polymorphisms (SNPs) and millions of additional mutations can be imputed from these SNPs through use of linkage disequilibrium.

Presently, the genetic biomarker discovery process is best approached with a GWAS (see Chap. 5 for further discussion and other methodologic approaches). A GWAS examines many common genetic variants in a group of individuals (hundreds to thousands) seeking variants (usually SNPs) for associations with a trait, including response to a specific drug, resistance to treatment, or adverse events. These studies normally compare the DNA of those with and those without the trait of interest. A GWAS is an economically viable means to identify novel genetic associations. Follow-up studies with the SNPs identified can then be used to verify the findings from the GWAS. Candidate gene studies are also widely utilized to identify possible biomarkers but are limited by our current understanding of the biology of schizophrenia and APD mechanisms of action. The clinical data available for testing associations between genetic variation and drug response using GWAS or candidate genes must be accurate and all important covariates (e.g., ethnicity, drug dose, concomitant medications) must be taken into account. Both the discovery and replication samples must be sufficiently large to provide enough statistical power to find relevant biomarkers. Additionally, since most drug responses are the result of a cluster of pharmacodynamic and pharmacokinetic influences, polygenetic methods of analysis are critical for finding clinically useful markers [1].

In this context, we examine methodological issues in identifying gene variants of clinical utility in psychiatry, review the most clinically useful genetic biomarkers currently available, address the practical implementation of pharmacogenetics and pharmacogenomics into practice, and speculate about their future promise for psychiatry.

Genetic Biomarkers of Drug Response

The known targets for psychotropic drugs are mainly proteins (e.g., monoamine receptors such as DA D_2 receptors; transporters of neurotransmitters such as the serotonin (5-HT) transporter; or enzymes which inactivate neurotransmitters such as monoamine oxidase inhibitors). The degree of expression of a target protein is under genetic control which affects its overall level of activity and resulting physiological impact. Genetic variants also result in changes in protein function (e.g., differences in receptor activation) which also impact drug effectiveness. Likewise, pharmacogenetic information predictive of response to drugs which share

a given pharmacophore (i.e., the essential features of the drug which bind it to the protein site which produces its effect) can guide drug selection prospectively. This can increase the probability of a therapeutic response, reduce treatment failures, and decrease associated suffering.

Synaptic Vesicle (SV2C)

Synaptic vesicle 2 proteins (SV2) which include SV2A, SV2B, and SV2C are localized on the surface of synaptic vesicles in all neurons [2-4]. SV2C protein is expressed in the ventral tegmental area, nucleus accumbens, olfactory bulb, and olfactory tubercle which comprise most of the mesolimbic dopaminergic system (i.e., the region of the brain believed to be the basis for the role of DA in psychosis). SV2C is important for exocytosis in glutamatergic and GABAergic neurons [5]. SV2C may contribute to the regulation of DA release in limbic and striatal terminal regions and for glutamate and gamma amino butyric acid (GABA) release in many areas of the brain. It may also be important for cholinergic striatal interneurons which play a role in extrapyramidal side effects (EPS) and cognition [6]. SV2C knockout mice exhibit large increases in tyrosine hydroxylase mRNA production, indicating SV2C decreases tyrosine hydroxylase expression [7]. Tyrosine hydroxylase is the rate limiting enzyme in the biosynthesis of DA and norepinephrine. Thus, SV2C may modulate the availability of DA which is involved in psychosis, cognition, motor function, and hormone secretion; and norepinephrine which is involved in psychosis, mood, anxiety, and hormone secretion [6, 7].

SV2C also appears to be involved in GABA–glutamate interactions in the prefrontal cortex, preferentially co-localizing with GABA, but not glutamate vesicles [5]. Instead, glutamate vesicles show enrichment of SV2B protein levels [5]. Down-regulation of mRNAs encoding glutamic acid decarboxylase 67 and reelin decreases the co-expression of cognate proteins in prefrontal cortex GABAergic neurons in schizophrenia and bipolar disorder, perhaps through the release of reelin from GABAergic neurons [8]. Reelin can interact with integrin receptors on cortical pyramidal neurons and regulate mRNA translation [8]. Such down-regulation of cognate proteins may be associated with the hypoplasticity of the prefrontal cortex in schizophrenia and bipolar disorder.

SV2C and Schizophrenia

Analysis of data from the Clinical Antipsychotic Trials of Intervention Effectiveness (CATIE) study revealed that two SNPs of SV2C were associated with inadequate response to olanzapine and quetiapine, but not to risperidone and ziprasidone [9]. Thus, patients possessing these SV2C variants experienced a worsening of their symptoms when receiving olanzapine or quetiapine [9]. Of note, these findings have not yet been replicated.

Sulfotransferase Family (SULT) 4A1

Another effort to use pharmacogenetics to guide treatment is based upon sulfotransferase 4A1 (SULT4A1), a unique member of the sulfotransferase family. Most members of this family are not located in the brain and are involved in detoxification. While, the function of SULT4A1 in the brain is unknown, a number of neurotransmitters (e.g., norepinephrine, DA, 5-HT) which are central to brain function are sulfonated by one or more other cytosolic sulfotransferases, leading to their inactivation and renal excretion [10, 11]. SULT4A1, however, is not involved in the inactivation of any known neurotransmitter or neuromodulator, differing from the other human sulfotransferases, in that it shares less than 36 % amino acid sequence homology with other known cytosolic sulfotransferases [12, 13]. Nevertheless, SULT4A1 is highly conserved between species (i.e., human, mouse, and rat isoforms sharing 97 % sequence homology), suggesting it plays an important physiological role. In both human and rodent brains, it is highly expressed in areas thought to be involved in the etiology of psychosis (e.g., cerebral cortex, cerebellum, pituitary brainstem) [14]. Although no coding variants are identified in man, there is genetic variation in the region that regulates expression which is in line with much evidence that gene expression rather than variation in gene product is relevant to functional effects of genetic variation [15]. Additional research to identify the biological function of SULT4A1 is needed.

SULT4A1 and Schizophrenia

An association between SULT4A1 and susceptibility to schizophrenia was initially established in a family-based study [16]. A SULT4A1 SNP was associated with greater impairment in working memory and executive function in schizophrenia patients [17]. Subsequently, a common haplotype, SULT4A1-1, was associated with greater improvement in total psychopathology during treatment with olanzapine in the CATIE study. SULT4A1-1 was significantly correlated with higher baseline psychopathology measured by the total Positive and Negative Symptom Scale (PANSS) score. Further, SULT4A1-1(+) patients treated with olanzapine had greater improvement in psychopathology compared with olanzapine-treated SULT4A1-1(-) and quetiapine- and risperidone-treated SULT4A1-1(+) patients [18]. Thus, SULT4A1-1(+) patients receiving olanzapine were three times more likely to exhibit a clinically significant therapeutic response than olanzapine-treated SULT4A1-1(-) subjects. Indeed, these patients had the best response to any treatment in the CATIE study. These results were corroborated in a retrospective analysis of an independent study of 160 patients with schizophrenia or bipolar disorder randomized to treatment with olanzapine or risperidone. In that trial, SULT4A1-1(+) patients had a superior response compared with olanzapine-treated SULT4A1-1(-) and risperidone-treated SULT4A1-1(+) patients [18]. SULT4A1(+) and (-) patients did not differ in response to risperidone, similar to the results in the CATIE study. A subsequent follow-up retrospective analysis of the CATIE sample showed that SULT4A1-1(+) patients receiving olanzapine had an approximate 80 % reduction in re-hospitalization events during CATIE compared with SULT4A1-1(-) patients receiving olanzapine [19].

Prospective studies of the ability of SULT4A1-1 to guide treatment assignment in schizophrenia and bipolar patients are needed to confirm these interesting results. If verified, they might help to determine the risk–benefit ratio of prescribing olanzapine which produces significant weight gain, since it was superior in efficacy only in SULT4A1-1(+) patients.

The association of SULT4A1-1(+) with olanzapine response was strengthened by an expression quantitative trait loci (eQTL) analysis. This analysis utilized the previously cited gene expression database, BrainCloud, for a relevant disease tissue [i.e., dorsal lateral prefrontal cortex (DLPFC)]. Two SNPs, rs470089 and rs138079, in linkage disequilibrium with the SULT4A1 haplotype which predicted response to olanzapine, had a significant cis-association with gene expression of a splicing variant of SULT4A1. This suggests that the common variants associated with olanzapine response could be an eQTL for SULT4A1 (Li et al., in preparation). The BrainCloud database suggests this isoform is uniquely overexpressed at the prenatal stage, implying an important role in neurodevelopment. According to the National Center for Biotechnology Information (NCBI) AceView database, this alternative splicing variant was identified specifically in cortex, but not in subcortical regions such as the hippocampus, striatum, thalamus, and cerebellum, indicating a unique importance for cortex. This finding was confirmed by microarray data from the Sestan Lab Human Brain Atlas.

In summary, the existing data suggests that the previously identified genetic markers associated with treatment response to olanzapine are DNA sequences in the vicinity of the structural portion of the SULT4A1 gene. These sequences may be required for gene expression (cis-acting) and code for a specific alternative splicing variant.

Genetic Biomarkers of Drug Metabolism

Most drugs are subject to enzymatic processes which alter their chemical structure (i.e., metabolism), thus changing their biological activity. These changes result in metabolites with altered biological activity, usually more hydrophilic and more easily eliminated from the body. Metabolic reactions are categorized into two broad classes: *Phase I reactions* which add polar groups to the parent compound by oxidation, reduction, or hydrolysis; and *Phase II conjugation reactions* (i.e., acetylation, glucuronidation, sulfation, or methylation) via transferases, resulting in polar conjugates.

Genetic variants affect the function and expression level of individual metabolic enzymes, impacting the rate and extent of their activity. Of clinical importance, the rate of metabolism determines the duration and extent of a drug's pharmacological activity by affecting the blood levels of the parent compound and metabolites. Dosing to achieve drug and/or metabolite levels within the therapeutic window (i.e., the range of concentrations between the lowest efficacious level and the level at which toxicity emerges) is critical to achieving a successful outcome. Since drug effects and toxicity are concentration dependent, the direction of change in efficacy or safety depends upon the nature (active or inactive; toxic) and concentration (elevated or decreased) of the parent compound and its metabolites. When capacity of a particular enzyme is decreased, resultant metabolite levels are decreased and the concentration of the parent compound is elevated compared with normal activity. Conversely, when the metabolic capacity of a particular enzyme is increased, resultant metabolite concentrations are increased and the parent compound concentrations are decreased. By including pharmacogenetic information in the decision process for drug and dose selection, the prescriber can choose doses with increased precision, resulting in concentrations within the therapeutic window and improved outcomes.

Of the currently available pharmacogenetic tests for metabolism of psychotropic drugs, the most relevant for psychiatry assesses the functional status of cytochrome P450 isozymes. Clinical guidance surrounding specific drug and dose recommendations based on genotype is published by the Royal Dutch Pharmacists Association–Pharmacogenetics Working Group and can be found at www.pharmgkb.org. In the following sections, we discuss the most clinically relevant genes which encode metabolic enzymes of the cytochrome P450 family and the antipsychotics which are metabolized by these isoenzymes.

Cytochrome P450 2D6 (CYP2D6)

CYP2D6 metabolizes a number of commonly used antipsychotic and antidepressant drugs and is primarily expressed in the liver [20, 21]. CYP2D6 metabolizes both risperidone and aripiprazole, as well as several first-generation APDs, including haloperidol and perphenazine. A number of functional variants of the gene CYP2D6 are identified and included in most commercially available pharmacogenetic tests. Patients classified as poor metabolizers (PMs) do not process these drugs well, often requiring alternate metabolic routes for biotransformation and achieving higher than expected steady-state concentrations of the parent compound. Thus, patients have a greatly elevated risk for serious adverse events [21]. Conversely, ultra metabolizers (UMs) rapidly metabolize drugs compared with the general population, leading to lower-than-expected blood concentrations of the parent drug. Concomitant use of drugs that act as CYP2D6 inhibitors or inducers can produce a metabolic state similar to either PM or UM status via drug-drug interactions. It is also important to note that the administration of multiple drugs, which use the same metabolic enzymatic pathways, can lead to lowered metabolism of some of these drugs due to competition for the isoenzyme.

Cytochrome P450 1A2

Cytochrome P450 1A2 (CYP1A2) is an isoenzyme responsible for the metabolism of a number of drugs (e.g., clozapine is metabolized to norclozapine by CYP1A2). Genetic variations which produce PMs or UMs appear to be very rare [21]. Cigarette smoking induces the expression of CYP1A2, leading to a more rapid clearance of drugs metabolized by this isoenzyme [21]. Since a large number of psychotic patients smoke, it is important to consider this when prescribing a CYP1A2-metabolized drug as lower-than-expected blood concentrations may result, especially if there is also a genetic propensity for rapid CYP1A2 metabolism.

Clozapine, olanzapine, and promazine are commonly used antipsychotics metabolized by the CYP1A2 isozyme.

Cytochrome P450 3A4/5 (CYP3A4/5)

CYP3A4 and CYP3A5 are isozymes responsible for the metabolism of a number of antipsychotic, anxiolytic, and antidepressant drugs, with both having similar activities. CYP3A4 is the predominant isoform in Caucasians and CYP3A5 the predominant isoform in African Americans.

Quetiapine and ziprasidone are commonly used antipsychotics metabolized by CYP3A4/5 isoenzymes.

Genetic Biomarkers of Antipsychotic-Related Adverse Effects

Antipsychotic-Induced Weight Gain

Weight gain and the associated metabolic syndrome are some of the most common adverse effects associated with APDs, especially clozapine, olanzapine, and quetiapine [22]. Antipsychotics target multiple neurotransmitter systems and brain regions involved in the regulation of food intake, metabolic rate, and body weight. Extensive pharmacogenetic investigations indicate an SNP in HTR2C, the gene that encodes for the 5-HT_{2C} receptor, may play an important role in antipsychotic-induced weight gain (AIWG) [23–25]. Another gene which impacts the 5-HT_{2C} receptor, MC4R, is also strongly implicated in AIWG. In studies of four independent populations, an SNP near MC4R (the gene that encodes for the melanocortin 4 receptor) was associated with AIWG with risk allele homozygotes gaining twice the weight as other patients [26–28]. In addition, MC4R regulates 5-HT_{2C} receptors, further implicating them in AIWG [28]. The strength of the associations with HTR2C and MC4R indicates that genetic testing for these variants may prove

clinically useful. For example, patients may be genotyped prior to initiation of APD treatment to see if they carry a risk allele for drug-induced weight gain. If so, antipsychotics which are less weight-inducing should be considered along with non-pharmacological interventions to prevent or attenuate weight gain (see Chaps. 10 and 12) [28].

Agranulocytosis

Clozapine is the most efficacious agent for treatment-resistant and suicidal schizophrenia or schizoaffective disorder patients [29, 30]. Despite superior efficacy, its use is limited due to potentially fatal drug-induced agranulocytosis which occurs in approximately 0.5–1 % of patients (see Chap. 12). Clozapine-induced agranulocytosis (CIA) is defined as a decrease in the absolute neutrophil count to less than 500 cells/mm³. This neutrophil deficiency greatly increases the risk of infection and possible fatality unless the drug is stopped and treatment with growth factors and antibiotics started. If these interventions are initiated, the fatality rate is 1:10,000. Current treatment guidelines restrict clozapine use to those patients who failed two previous trials of other antipsychotic drugs (APDs) or have made a serious suicide attempt. Patients are also required to have periodic blood cell monitoring. If this is done, several studies with clozapine found this agent reduced overall mortality due to its profound effect on suicide. There is no biomarker for the ability of clozapine to reduce the risk of suicide despite attempts to identify one.

Pharmacogenetic studies indicate that CIA is associated with a variation in genes that encode for the human leukocyte antigen (HLA) [31]. In an initial study of HLA gene variants in Ashkenazi Jews, a strong association was found between a specific HLA haplotype and CIA, despite the small study population [32]. Another study analyzed the whole exome sequence of a small sample of Finnish schizophrenia patients suffering from CIA (25 cases and 27 controls) and found a nominal association between multiple coding variants in the HLA-C gene and CIA [33]. Although the study was underpowered, carriers of an allelic variation in the HLA-B/HLA-C region were approximately twofold less likely to develop CIA, indicating that HLA genes warrant further investigation [34]. In this context, a recent study of HLA genes found that a single SNP, 6672G>C, was associated with an extremely elevated risk for CIA (i.e., the odds were 16.9 times greater in patients who carried the marker as compared with those who did not) [34].

Tardive Dyskinesia

Tardive dyskinesia (TD) is a group of delayed-onset, iatrogenic movement disorders (most notably oral-bucco-lingual stereotypy) associated with APD treatment (see Chap. 12) [35]. TD is caused by inherent vulnerability and chronic DA receptor blockade. First-generation antipsychotics (FGAs) with high DA D_2 receptor occupancy are associated with a higher risk of TD than SGAs which produce lower levels of D_2 receptor occupancy when used appropriately. TD is reported to occur in approximately 20 % of patients treated chronically with FGAs, especially older females [36]. While the use of SGAs greatly reduces the risk, even these drugs can cause TD, but at much lower rates than with the FGAs [37]. Indeed, the greatly reduced risk of TD with the SGAs is a major reason for their adoption as preferred first-line therapy. The severe morbidity and increased mortality associated with TD, however, provide a rationale for developing genetic tests to detect susceptibility.

Familial studies showing a hereditary association with TD provide compelling evidence for genetic susceptibility in its etiology [38]. Promising pharmacogenetic associations are reported for genetic variants of CYP2D6, CYP1A2, the dopamine receptor genes DRD2 and DRD3, and manganese superoxide dismutase (MnSOD) [39–41]. The DRD2 A1 allele at rsl800497 is associated with a 40 % reduction in striatal D₂ receptor density and appears to be protective against TD [42]. Meta-analyses based on overlapping sets of studies, however, strongly support increased rates of TD in carriers of the DRD2 A2 allele [43, 44]. These results indicate a 30 % increase in risk for TD per A2 allele (i.e., A2/A2 homozygotes are nearly 80 % more likely to develop TD as A1/A1 homozygotes). While genetic variants of DRD3 (the gene that encodes for the DA D₃ receptor) also showed an association with TD risk, this did not hold up to replication in all studies [42].

Oxidative stress is hypothesized as contributing to the development of TD [45]. Enzymes which scavenge free radicals (e.g., the mitochondrial enzyme MnSOD) are associated with TD. The gene encoding MnSOD possesses one polymorphism which results in its less efficient transport [44]. Pharmacogenetic studies, however, produced mixed results with one meta-analysis actually showing a protective association between this MnSOD polymorphism and TD, with the genotype associated with less efficient MnSOD exhibiting an approximate 50 % risk reduction [44].

Since many antipsychotics are metabolized by CYP2D6 and CYP1A2 and many adverse effects are concentration dependent, variants of the genes encoding for these isoenzymes which result in metabolic deficiencies may be useful in determining TD risk [42]. In this context, a meta-analysis of eight studies demonstrated that CYP2D6 PMs had a 64 % greater risk of developing TD [46]. Investigations into genetic variants of CYP1A2 related to isoenzyme induction have thus far provided inconsistent results [47].

Clinical Use of Pharmacogenetics in Psychotropic Drug Selection

Although pharmacogenetic testing is commercially available and offered by a growing number of laboratories, it has not been widely adopted in psychiatric practice. Guidance regarding the use of these tests must be enhanced before more clinicians will utilize them. The experience in implementation obtained by early adopters needs to be carefully evaluated and reported in peer reviewed journals. This is no substitute, however, for prospective effectiveness trials in a variety of clinical settings. These studies must compare the outcome on a wide range of measures including time to discontinuation, change in psychopathology, and adverse effect burden from treatment as usual versus the choice of medication based upon genotypic data [28]. Other clinical trial methodological issues specific to schizophrenia (e.g., poor adherence) should also be considered in the design of such studies. In addition, improvements in pharmacogenetic clinical trial methodologies are needed to overcome various shortcomings such as confounding factors (e.g., race, ethnicity, and environment).

Ethnicity and Pharmacogenomics

Ethnicity and race are used to categorize groups of people with shared ancestry and inherited physical traits [48]. Ideally, pharmacogenetic biomarkers should apply across an entire population to maximize clinical utility. There is great racial and ethnic variability, however, in the incidence of pharmacogenetic biomarkers, including the expression of cytochrome P450 isoforms and their variants [21]. Self-reported race/ethnicity identification corresponds well to major genetic clusters, suggesting that self-identification of race is of considerable value in dividing populations for the development and utilization of genetic biomarkers [49]. There are errors, however, perhaps in the 5–10 % range in this regard. The introduction of race/ethnicity categorization into the study of pharmacogenetics is not without confounding limitations. For example, in addition to recognized group physical characteristics, ethnicity and race often include shared behavioral, dietary, cultural, linguistic, or religious characteristics [48]. Thus, ethnicity and race encompass both genetic and environment influences. Since an individual's biological status is the result of these complicated and intertwined influences, the exact contribution of each to one's biological status is complex and difficult to determine. Thus, these complex interactions confound many studies which try to identify the specific connections between pharmacogenetics and race/ethnicity.

The limitations of race/ethnicity categorizations are illustrated in the findings of a study by Wilson et al. which investigated the interactions of genetic variation, drug response, and race [50]. In this study, pharmacogenetic associations were investigated by assigning people to different groups depending on shared genetic regions and comparing to self-reported race/ethnicity. They found that microsatellite analysis-based genetic groupings were more informative regarding differences in CYP2D6 drug metabolism than the groupings based on self-reported race/ethnicity [50]. Whether such genetic grouping categorization techniques could generally improve the results of pharmacogenetic studies is unknown. There are significant obstacles to including such techniques, as they increase the complexity and cost of conducting such studies. While racial and ethnic categorization in pharmacogenetic studies facilitates the discovery of associations to drug response, it also limits the applicability of these findings as they are race/ethnicity dependent.

Conclusion

The treatment of patients with schizophrenia will surely benefit from advances in pharmacogenetics and pharmacogenomics. The large variation in clinical presentation and response to APDs seen in patients with schizophrenia engenders an opportunity for these approaches to meaningfully improve therapeutic outcomes. Current use of genetic testing is limited but seems likely to grow as prescribers become more familiar with the benefits of incorporating drug metabolism information into their choices of drugs and dosages. As with many new technologies, there are barriers which slow adoption (e.g., lack of familiarity among physicians; lack of prospective outcome studies to confirm utility; uncertainty on how to interpret results). The convergence of many factors (e.g., more accurate and faster genotyping technologies; reductions in cost of testing; and confirmation of the clinical utility of pharmacogenetic biomarkers) will help to facilitate more extensive clinical implementation.

Disclosures Bill W. Massey is Chief Scientific Officer for MyGenesRx and a Shareholder at SureGene and MyGenesRx. He is a previous consultant for Neurotherapeutics and Mallinckrodt. **Herbert Y. Meltzer** reports the following financial relationships: (1) ownership/investment interests in Suregene, ACADIA, and GlaxoSmithKline; (2) industry activities (such as speaking, advising, consulting, and providing educational programs) with Janssen Pharmaceuticals, Lundbeck Inc., Sunovion Pharmaceuticals Inc. (Dainippon Sumitomo Pharma Co., Ltd.), TEVA, BiolineRx, BI (Boehringer Ingelheim Pharma GmbH & Co. KG), Envivo, and Companion Diagnostics; (3) expert testimony with Janssen; and (4) current grants: Sunovion Novartis, Dainippon Sumitomo, Envivo, NeuroTherapeutics Pharma, Inc., Otsuka, Takeda, Alkermes, Naurex, Inc., and Astellas Research Institute of America.

Jiang Li has no disclosures.

References

- 1. Ma Q, Lu AY. Pharmacogenetics, pharmacogenomics, and individualized medicine. Pharmacol Rev. 2011;63(2):437–59.
- Bajjalieh SM, Peterson K, Shinghal R, Scheller RH. SV2, a brain synaptic vesicle protein homologous to bacterial transporters. Science. 1992;257(5074):1271–3.
- 3. Bindra PS, Knowles R, Buckley KM. Conservation of the amino acid sequence of SV2, a transmembrane transporter in synaptic vesicles and endocrine cells. Gene. 1993;137(2): 299–302.
- 4. Tao-Cheng JH. Ultrastructural localization of active zone and synaptic vesicle proteins in a preassembled multi-vesicle transport aggregate. Neuroscience. 2007;150(3):575–84.
- 5. Gronborg M, Pavlos NJ, Brunk I, et al. Quantitative comparison of glutamatergic and GABAergic synaptic vesicles unveils selectivity for few proteins including MAL2, a novel synaptic vesicle protein. J Neurosci. 2010;30(1):2–12.
- Dardou D, Dassesse D, Cuvelier L, Deprez T, De RM, Schiffmann SN. Distribution of SV2C mRNA and protein expression in the mouse brain with a particular emphasis on the basal ganglia system. Brain Res. 2011;1367:130–45.
- Dardou D, Monlezun S, Foerch P, et al. A role for Sv2c in basal ganglia functions. Brain Res. 2013;1507:61–73.

- 8. Guidotti A, Auta J, Davis JM, et al. GABAergic dysfunction in schizophrenia: new treatment strategies on the horizon. Psychopharmacology (Berl). 2005;180(2):191–205.
- Ramsey TL, Liu Q, Massey BW, Brennan MD. Genotypic variation in the SV2C gene impacts response to atypical 2 antipsychotics in the CATIE Study. Schizophr Res. 2013;149(1–3): 21–5.
- Mitchell DJ, Minchin RF. Cytosolic Aryl sulfotransferase 4A1 interacts with the peptidyl prolyl cis-trans isomerase Pin1. Mol Pharmacol. 2009;76(2):388–95.
- 11. Gamage N, Barnett A, Hempel N, et al. Human sulfotransferases and their role in chemical metabolism. Toxicol Sci. 2006;90:5–22.
- 12. Allali-Hassani A, Pan PW, Dombrovski L, et al. Structural and chemical profiling of the human cytosolic sulfotransferases. PLoS Biol. 2007;5(5):e97.
- Falany CN, Xie X, Wang J, et al. Molecular cloning and expression of novel sulphotransferaselike cDNAs from human and rat brain. Biochem J. 2000;346(3):857–64.
- 14. Liyou NE, Buller KM, Tresillian MJ, et al. Localization of a brain sulfotransferase, SULT4A1, in the human and rat brain: an immunohistochemical study. J Histochem Cytochem. 2003;51(12):1655–64.
- Lewis AG, Minchin RF. Lack of exonic sulfotransferase 4A1 mutations in controls and schizophrenia cases. Psychiatr Genet. 2009;19(1):53–5.
- Brennan MD, Condra J. Transmission disequilibrium suggests a role for the sulfotransferase-4A1 gene in schizophrenia. Am J Med Genet B Neuropsychiatr Genet. 2005;139B(1):69–72.
- Meltzer HY, Brennan MD, Woodward ND, Jayathilake K. Association of Sult4A1 SNPs with psychopathology and cognition in patients with schizophrenia or schizoaffective disorder. Schizophr Res. 2008;106(2–3):258–64.
- Ramsey TL, Meltzer HY, Brock GN, Mehrotra B, Jayathilake K, Bobo WV, Brennan MD. Evidence for a SULT4A1 haplotype correlating with baseline psychopathology and atypical antipsychotic response. Pharmacogenomics. 2011;12(4):471–80.
- Liu Q, Ramsey TL, Meltzer HY, Massey BW, Padmanabhan S, Brennan MD. Sulfotransferase 4A1 haplotype (*SULT4A1-1*) is associated with decreased hospitalization events in antipsychotictreated patients with schizophrenia. Prim Care Companion CNS Disord. 2012;14(3):1–6.
- Ingelman-Sundberg M. Genetic polymorphisms of cytochrome P450 2D6 (CYP2D6): clinical consequences, evolutionary aspects, and functionary diversity. Pharmacogenomics J. 2005; 5:6–13.
- 21. de Leon J. Incorporating pharmacogenetics into clinical practice: reality of a new tool in psychiatry. Current issues in clinical implementation. CNS Spectr. 2006;11(3 Suppl 3):8–12.
- 22. Pramyothin P, Khaodhiar L. Metabolic syndrome with the atypical antipsychotics. Curr Opin Endocrinol Diabetes Obes. 2010;17(5):460–6.
- Reynolds GP, Zhang ZJ, Zhang XB. Association of antipsychotic drug-induced weight gain with a 5-HT2C receptor gene polymorphism. Lancet. 2002;359(9323):2086–7.
- Sicard MN, Zai CC, Tiwari AK, et al. Polymorphisms of the HTR2C gene and antipsychotic induced weight gain: an update and meta-analysis. Pharmacogenomics. 2010;11(11):1561–71.
- Müller DJ, Chowdhury NI, Zai CC. The pharmacogenetics of antipsychotic-induced adverse events. Curr Opin Psychiatry. 2013;26(2):144–50.
- Malhotra AK, Correll CU, Chowdhury NI, et al. Association between common variants near the melanocortin 4 receptor gene and severe antipsychotic drug-induced weight gain. Arch Gen Psychiatry. 2012;69(9):904–12.
- Lett TA, Wallace TJ, Chowdhury NI, Tiwari AK, Kennedy JL, Müller DJ. Pharmacogenetics of antipsychotic-induced weight gain: review and clinical implications. Mol Psychiatry. 2012;17(3):242–66.
- Zhang JP, Malhotra AK. Pharmacogenetics of antipsychotics: recent progress and methodological issues. Expert Opin Drug Metab Toxicol. 2013;9(2):183–91.
- 29. Meltzer HY. Treatment-resistant schizophrenia—the role of clozapine. Curr Med Res Opin. 1997;14(1):1–20.

9 Pharmacogenetics in the Treatment of Schizophrenia

- 30. McEvoy JP, Lieberman JA, Stroup TS, et al. Effectiveness of clozapine versus olanzapine, quetiapine, and risperidone in patients with chronic schizophrenia who did not respond to prior atypical antipsychotic treatment. Am J Psychiatry. 2006;163(4):600–10.
- Chowdhury NI, Remington G, Kennedy JL. Genetics of antipsychotic-induced side effects and agranulocytosis. Curr Psychiatry Rep. 2011;13(2):156–65.
- 32. Lieberman JA, Yunis J, Egea E, Canoso RT, Kane JM, Yunis EJ. HLA-B38, DR4, DQw3 and clozapine-induced agranulocytosis in Jewish patients with schizophrenia. Arch Gen Psychiatry. 1990;47(10):945–8.
- 33. Tiwari AK, Need A, Knight J, et al. Sequencing analysis of exomes of Finnish patients with clozapine-induced agranulocytosis. Mol Psychiatry. 2013;11.
- Athanasiou MC, Dettling M, Cascorbi I, et al. Candidate gene analysis identifies a polymorphism in HLA-DQB1 associated with clozapine-induced agranulocytosis. J Clin Psychiatry. 2011;72(4):458–63.
- 35. Waln O, Jankovic J. An update on tardive dyskinesia: from phenomenology to treatment. Tremor Other Hyperkinet Mov. 2013;3. pii: tre-03-161-4138-1.
- Kane JM, Woerner M, Lieberman J. Tardive dyskinesia: prevalence, incidence, and risk factors. J Clin Psychopharmacol. 1988;8(4 Suppl):52S–6.
- Pena MS, Yaltho TC, Jankovic J. Tardive dyskinesia and other movement disorders secondary to aripiprazole. Mov Disord. 2011;26:147–52.
- Müller DJ, Schulze TG, Knapp M, et al. Familial occurrence of tardive dyskinesia. Acta Psychiatr Scand. 2001;104(5):375–9.
- Thelma B, Srivastava V, Tiwari AK. Generic underpinnings of tardive dyskinesia: passing the baton to pharmacogenetics. Pharmacogenomics. 2008;9(9):1285–306.
- Müller DJ, Shinkai T, De Luca V, Kennedy JL. Clinical implications of pharmacogenomics for tardive dyskinesia. Pharmacogenomics J. 2004;4(2):77–87.
- 41. Mackenzie B, Souza RP, Likhodi O, et al. Pharmacogenetics of antipsychotic treatment response and side effects. Therapy. 2010;7(2):191–8.
- 42. Lencz T, Malhotra AK. Pharmacogenetics of antipsychotic-induced side effects. Dialogues Clin Neurosci. 2009;11(4):405–15.
- 43. Zai CC, De Luca V, Hwang RW, et al. Meta-analysis of two dopamine D2 receptor gene polymorphisms with tardive dyskinesia in schizophrenia patients. Mol Psychiatry. 2007;12:794–5.
- Bakker PR, van Harten PN, van Os J. Antipsychotic-induced tardive dyskinesia and polymorphic variations in COMT, DRD2, CYP1 A2 and MnSOD genes: a meta-analysis of pharmacogenetic interactions. Mol Psychiatry. 2008;13:544–56.
- 45. Tsai G, Goff DC, Chang RW, Flood J, Baer L, Coyle JT. Markers of glutamatergic neurotransmission and oxidative stress associated with tardive dyskinesia. Am J Psychiatry. 1998;155:1207–13.
- 46. Patsopoulos NA, Ntzani EE, Zintzaras E, Ioannidis JP. CYP2D6 polymorphisms and the risk of tardive dyskinesia in schizophrenia: a meta-analysis. Pharmacogenet Genomics. 2005; 15:151–8.
- 47. Bakker PR, Bakker E, Amin N, van Duijn CM, van Os J, van Harten PN. Candidate genebased association study of antipsychotic-induced movement disorders in long-stay psychiatric patients: a prospective study. PLoS One. 2012;7(5):e36561.
- 48. Sankar P, Cho MK, Mountain J. Genetic research and health disparities. JAMA. 2004; 291:2985–9.
- 49. Tang H, Quertermous T, Rodriguez B, et al. Genetic structure, self-identified race/ethnicity, and confounding in case-control association studies. Am J Hum Genet. 2005;76:268–75.
- Wilson JF, Weale ME, Smith AC, et al. Population genetic structure of variable drug response. Nat Genet. 2001;29:265–9.

Chapter 10 Management of Comorbid Medical Conditions in Schizophrenia

Jeffrey T. Rado

Scope of the Problem

Patients with schizophrenia are at greater risk than the general population for developing a variety of diseases, including asthma, chronic obstructive pulmonary disease (COPD), diabetes, hepatitis C, and congestive heart failure [1]. For example, a 2-year, prospective study of 602 patients with schizophrenia in the UK found increased rates of angina and respiratory symptoms [2]. A study of Swedish patients with schizophrenia (n=8,277, drawn from a database of more than six million adults) followed for 7 years found increased rates of COPD (women: hazard ratio=2.16, CI=1.80–2.34; men: hazard ratio=1.53, CI 1.32–1.76) and diabetes (women: hazard ratio=2.15, CI=1.96–2.36; men: hazard ratio=1.68, CI=1.54–1.83) [3]. A third study of nearly 7,000 Taiwanese patients with schizophrenia found elevated rates of COPD, diabetes, and stroke [4].

Mortality

Numerous epidemiological studies document an increased mortality rate in patients with schizophrenia. One of the earliest and longest of these studies assessed 510 patients admitted to the University of Iowa Psychiatric Hospital between 1934 and 1944 and then followed them for 40 years [5]. The standardized mortality ratio (SMR, ratio of observed to expected deaths) in the third and fourth decades of follow-up was 4.21 for males and 7.89 for females, indicating a mortality rate 4 to nearly 8 times higher than the general population. Another prospective cohort study

J.T. Rado (🖂)

Department of Psychiatry, Rush University Medical Center, Chicago, IL 60612, USA

Department of Internal Medicine, Rush University Medical Center, Chicago, IL 60612, USA e-mail: jeffrey_rado@rush.edu


Fig. 10.1 Factors contributing to increased mortality in schizophrenia

included 370 patients followed for 13 years in England [6]. The SMR for all causes was 2.98 (CI=2.36-2.72) and also elevated for specific disease states including circulatory (2.49, CI=1.64-3.63), respiratory (3.17, CI=1.16-6.90), and endocrine (11.66, CI=3.79-27.21). Enger et al. conducted a retrospective cohort review of 1,920 antipsychotic-treated patients in a health plan general membership database compared with 9,600 patients without schizophrenia [7]. The mortality rate in these patients was four times the general population and the rate of new-onset diabetes was more than two times higher (Fig. 10.1).

More recent cohort studies draw similar conclusions. Ran et al. conducted a 10-year follow-up of 510 patients drawn from a larger population of 123,572 rural Chinese [8]. The calculated all-cause SMR for the entire cohort was 4.0 (CI=2.4–5.8). For men it was 4.9 (CI=2.8–8.1) and for women it was 3.3 (CI=1.9–6.1). Fors et al. followed 255 Swedes with schizophrenia over a 10-year period and compared them to 1,275 age- and sex-matched patients from a national population registry [9]. The mortality rate (i.e., 23.0 %) was double that of the comparator group (i.e., 11.2 %) with the largest portion of excess mortality attributable to cardiovascular disease. A large Swedish cohort study found elevated all-cause mortality in both women (adjusted hazard ratio=2.75; CI=2.52–3.00) and men (adjusted hazard ratio=2.44; CI=2.25–2.64) with the leading causes of death being coronary artery disease (CAD) and cancer [3].

A meta-analysis of 18 studies involving more than 66,000 schizophrenia patients followed for 2–30 years found an aggregate SMR of 1.51 (1.48–1.54) [10]. A second meta-analysis of 20 schizophrenia studies found an SMR of 2.2 for all-cause mortality [11].

Reduced Life Expectancy

As a result of this increased mortality, patients with schizophrenia have a 20 % shortened life expectancy compared with those without the disease (i.e., from 76 to 61 years) [12]. The majority of this excess mortality is attributed to CAD, with suicide accounting for approximately 10–15 %. More premature deaths are attributable to heart disease than any other cause. This is due, in part, to an overabundance of poorly controlled cardiac risk factors such as hyperlipidemia, metabolic syndrome, and obesity.

Medical Care of Patients with Schizophrenia

Before reviewing the specific medical conditions prevalent in this population, it is important to examine their quality of and access to medical care, since several epidemiological studies attribute the elevated mortality rates to these issues. In a 9-year prospective study of 253 institutionalized patients (80 % schizophrenia) in Finland, the elevated all-cause mortality risk (SMR=1.9 for males; 3.2 for females) was attributed partially to suboptimal care of physical illnesses [13]. A prospective analysis of 73 Irish patients with schizophrenia followed for 7.5 years found a twofold elevated mortality risk with the authors postulating that disengagement from health services is an important contributing factor [14].

Access to Care

Access to health care is the focus of several studies in schizophrenia. Utilizing data from the National Health Interview Survey (NHIS), Bradford et al. compared individuals with psychotic disorders, bipolar disorder, or major depression to persons without mental disorders [15]. Those with psychotic disorders were less likely to have a primary care physician and reported greater difficulties accessing care. Retrospective analysis of medical services obtained by 175,653 patients in the Veterans Administration system found that those with diabetes or hypertension and comorbid schizophrenia, bipolar disorder, or anxiety had substantially fewer medical visits than those without these disorders [16]. In another study, 60 % of veterans with serious mental illness (including schizophrenia) perceived barriers to accessing medical care [17]. Among these barriers were the lack of recognition of health complaints by providers, reluctance of non-psychiatrists to provide comprehensive care to psychiatric patients, screening procedures not provided to these patients, and lack of continuity of care due to unstable domiciles common in this population [18]. Even when patients were screened for common ailments such as hypertension or hyperlipidemia, psychiatric providers struggled to obtain referrals for appropriate care or were unable to provide general medical care on site [19].

Underdiagnosis

Even when care is accessible, underdiagnosis of common conditions may occur. Briskman et al. evaluated blood pressure, lipid levels, and fasting blood sugar in 200 hospitalized psychiatric patients (primarily schizophrenia) compared with a nonpsychiatric group [20]. Diabetes, hypertension, and dyslipidemia were all underdiagnosed in the psychiatric patients. In a 7-year cohort study carried out in more than 6,000 Swedes with schizophrenia, the authors reported elevated mortality due to cancer and ischemic heart disease compared with the general population [3]. Despite having twice as many contacts with the health care system, these patients were less likely than those without schizophrenia to have had their cancer (73.9 % versus 83.3 %) or heart disease (26.3 % versus 43.7 %) previously diagnosed.

Undertreatment

The Clinical Antipsychotic Trials of Intervention Effectiveness (CATIE) Study (n=1,460) attempted to mimic naturalistic antipsychotic treatment patterns in schizophrenia. Baseline point prevalence of diabetes, hypertension, and hyperlipidemia was measured with rates of non-treatment being 30.2 %, 62.4 %, and 88 %, respectively [30]. Of note, these patients were engaged in regular, ongoing psychiatric care. Vahia et al. conducted a cross-sectional study of 198 community-dwelling patients (age 55 or older) [22]. Compared with nonpsychiatric patients, those with schizophrenia were less likely to receive treatment for their hypertension (75 % versus 93 %), heart disease (28 % versus 75 %), or gastric ulcers (53 % versus 100 %) and were more likely to have their diabetes treated (86 % versus 81 %). A retrospective analysis of 113,653 patients (age 65 or older) from a national database examined those hospitalized for acute myocardial infarction [19]. Patients with a comorbid mental disorder (n=5,365; 41 % schizophrenia) were significantly less likely to undergo percutaneous transluminal angioplasty (11.8 % versus 16.8 %, p < 0.001) or coronary artery bypass graft (CABG) surgery (8.3 % versus 12.6 %, p<0.001). A comparative meta-analysis involving 22 studies (10 in patients with schizophrenia), including a total of 825,754 patients, examined inequalities in coronary procedures for patients diagnosed with a psychiatric disorder [23]. Those with schizophrenia were less likely to receive invasive coronary interventions (relative risk (RR) of CABG=0.69, CI=0.55–0.85; RR of cardiac catheterization=0.50, CI=0.34–0.75). Not all studies, however, find disparities in care. For example, Whyte et al. reviewed the electronic charts of 481 UK general medicine practices and did not find differences in quality of diabetes care for those with schizophrenia [24].

Screening

Recommended screening may also occur at lower rates. Data from a free psychiatric clinic in San Diego followed 47 patients (age 45 or older) for 1 year [25]. Compared with age- and gender-matched patients from the same clinic, those with schizophrenia had fewer medical visits and were less likely to receive a detailed physical exam, lipid screening, or colon cancer screening. In another study, Martens et al. utilized the Population Health Research Data Repository in Manitoba, Canada, to determine factors associated with obtaining the papanicolaou (PAP) smear, a standard screening procedure for early detection of cervical cancer [26]. Compared with nonpsychiatric patients, women with schizophrenia were less likely to be screened with a PAP smear (58.8 % versus 67.8 %, p<0.001).

Medical Adverse Events

Acute medical and surgical hospitalizations among 1,746 patients with schizophrenia were compared to 732,158 non-schizophrenia patients in a Maryland database [27]. Diagnosis of schizophrenia was associated with increased risk of postoperative respiratory failure (OR 2.08, CI=1.41–3.06), postoperative deep venous thrombosis (OR=1.96, CI=1.18–3.26), postoperative sepsis (OR 2.29, CI=1.49–3.51), and an increased risk of ICU admission and death. Outcomes data from the Taiwanese National Health Insurance Research Database (NHIRD) described postoperative outcomes in 8,967 patients with schizophrenia [4]. Compared with the general population, patients had higher 30-day postoperative complications, including stroke (2.6 % versus 1.6 %, p<0.0001), bleeding (1.6 % versus 1.2 %, p<0.002), pneumonia (1.3 % versus 0.4 %, p<0.0005), sepsis (0.8 % versus 0.3 %, p<0.0001), and acute renal failure (0.3 % versus 0.1 %, p<0.0005). Overall, the postoperative complication rate was higher (6.2 % versus 3.9 %, p<0.0001) and perhaps most concerning, there was a greater risk of dying 30 days after surgery (i.e., 1.1 %) versus those without schizophrenia (i.e., 0.4 %, p<0.0001).

In summary, the quality of care that patients with schizophrenia receive is often subpar. Specifically, access to medical care is limited, rates of screening for common medical conditions are low, and patients are too often underdiagnosed. Even when a diagnosis is made, these patients frequently do not receive appropriate treatment. Finally, patients with schizophrenia experience higher rates of medical adverse events compared to the general population.

Coronary Artery Disease and Its Risk Factors

As noted earlier, numerous studies report a higher mortality rate in patients with schizophrenia, much of which is attributed to cardiovascular disease. Not surprisingly, patients experience cardiac risk factors, such as hypertension and hyperlipidemia, at higher rates than the general population.

Hypertension

Hypertension (HTN) is classified by the World Health Organization (WHO) as one of the most important global mortality risk factors [28]. Kilbourne et al. found that hypertension in patients with schizophrenia conferred a nearly 40 % increase in mortality (hazard ratio = 1.38 (1.32-1.46)) [29]. A high prevalence (i.e., 33 %) of hypertension was found in the CATIE Study, while the Comparison of Atypicals in First Episode of Psychosis (CAFÉ) Study found up to 25 % of patients at baseline had mild hypertension (systolic BP>130) [30, 31]. After 1 year of treatment with olanzapine, risperidone, or quetiapine, rates of mild hypertension further increased to 41 %. De Hert also found similar elevated rates of hypertension (i.e., 25.9 %) in a retrospective chart review of patients with schizophrenia which increased after 3 years of antipsychotic drug treatment (i.e., 41.7 %) [32]. Not all studies, however, find increased rates of hypertension. A chart review of 125 patients with schizophrenia did not find elevated rates of hypertension compared with 1,721 age-matched, nonpsychiatric primary care patients [33]. Further, 6 months of treatment with antipsychotics was not associated with an increased prevalence in 130 outpatients with schizophrenia or schizoaffective disorder [34]. For additional discussion of this issue, see Chap. 12.

Treatment of hypertension is based on the 2014 Evidence-Based Guideline for the Management of High Blood Pressure in Adults [35]. For persons age ≥ 60 years, the goal blood pressure is <150/90 mmHg. The goal for those under age 60 is <140/90 mmHg. Thiazide-type diuretics, angiotensin converting enzyme (ACE) inhibitors, angiotension receptor blockers (ARBs) or calcium channel blockers alone or in combination are among the recommended initial treatment options for nonblack individuals. Thiazide-type diuretics or calcium channel blockers alone or in combination are the initial recommended treatment for blacks.

In summary, hypertension is common in schizophrenia. Lifestyle changes and the addition of antihypertensive medications, if needed, are the standard treatment approach.

Hyperlipidemia

The WHO reported that hyperlipidemia accounts for 4.5 % of global deaths. Rates of hyperlipidemia are higher in schizophrenia with elevated triglycerides occurring in 47.3 % of CATIE participants [30]. A large population-based study of 766,427 Taiwanese found that individuals with schizophrenia had a small but significant increase in hyperlipidemia prevalence (i.e., 8.15 % versus 8.10 % in the general population) [36]. Worse lipid profiles were noted in patients with acute phase schizophrenia, while community-dwelling clozapine-treated patients had hyperlipidemia rates of 2–4.6 % [37, 38]. The CAFÉ Study found rates of hyperlipidemia between 22 and 28 % at baseline [31]. Insurance claims data from 1,631 schizophrenia patients was

 Table 10.1
 2013
 ACC/AHA
 Guideline on the Treatment of Blood
 Cholesterol to Reduce

 Atherosclerotic Cardiovascular Risk in Adults
 Image: Cardiovascular Risk in Adults

Groups who are will benefit from statin therapy:

- 1. Individuals with known clinical atherosclerotic cardiovascular disease (ASCVD).
- 2. Individuals with primary elevations of LDL-C³190 mg/dL.
- 3. Individuals 40-75 years of age with diabetes with LDL-C 70-189 mg/dL.
- 4. Individuals without clinical atherosclerotic cardiovascular disease or diabetes who are 40–75 years of age with LDL-C 70–189 mg/dL and an estimated 10-year ASCVD risk of 7.5 % or higher (risk calculator: http://tools.cardiosource.org/ASCVD-Risk-Estimator/).

Source: [147]

reviewed and a slight but significant increase was found in hyperlipidemia with second-generation antipsychotics (SGAs) (HR = 1.41, CI = 1.09-1.83) but not with first-generation antipsychotics (FGAs) [39]. Of note, some studies in this population have not found baseline-elevated cholesterol levels [40].

The 2013 ACC/AHA Guidelines on the Treatment of Blood Cholesterol to Reduce Atherosclerotic Cardiovascular Risk in Adult established revised guidelines for the management of hyperlipidemia. Treatment involves a combination of low-fat diet counseling (e.g., reducing intake of fried foods, red meat, and other animal fat) and/or statin medications. Statins (e.g., atorvastatin, simvastatin) inhibit HMG CoA reductase, the main enzyme that produces cholesterol in the liver. The most common adverse effect is myalgia, which can be managed by lowering the statin dose or changing to a lower risk agent such as pravastatin. Four groups of patients have been identified who are recommended to be treated with statin therapy (see Table 10.1). In contrast to prior guidelines, statin therapy is no longer tied to a specific low-density lipoprotein (LDL) goal.

In summary, patients with schizophrenia are at greater risk of hyperlipidemia. Treatment usually involves a combination of lifestyle changes and statin medications.

Diabetes Mellitus

Diabetes is such a strong risk factor for CAD that it is considered a "coronary artery disease equivalent." Diabetes is diagnosed by one of the following:

- *Hemoglobin* $A_{1C} \ge 6.5 \%$.
- Fasting plasma glucose \geq 126 mg/dL.
- Two-hour plasma glucose \geq 200 mg/dL during an *oral glucose tolerance test*.
- *Random plasma glucose* ≥ 200 mg/dL in a patient with symptoms of hyperglycemia.

In addition to physical activity, initial treatment recommendations include weight loss with either low-carbohydrate, low-fat, calorie-restricted or Mediterranean diets. Metformin is the preferred initial pharmacologic agent. Schizophrenia is associated with high rates of impaired glucose tolerance and diabetes. Even first-episode, drug-naïve patients exhibit impaired fasting glucose tolerance and increased insulin resistance [41]. This finding, however, is not entirely consistent [42]. Bresee et al. studied a population-based cohort of 28,755 patients with schizophrenia using an administrative database in Alberta, Canada [43]. Compared with non-affected controls, diabetes (age 30–39; 3.8 % versus 1.4 %) and CAD occurred at higher rates (i.e., 27 % versus 17 %) in patients with schizophrenia. A retrospective cohort study in a Saskatchewan health database compared 3,022 subjects with schizophrenia with an age- and sex-matched control group without psychiatric illness. Rates of diabetes were more than doubled (OR=2.1, CI=1.8–2.4) in those with schizophrenia [44].

Lifestyle Factors

Obesity is a major risk factor for diabetes and rates are higher in schizophrenia [45]. The reason for this is multifactorial. Obesity is associated with low socioeconomic status and low education, both of which are more prevalent among the seriously mentally ill [46]. Further, negative symptoms and cognitive impairment are associated with low motivation, suboptimal self-care, and poor dietary habits, all of which impact body mass index (BMI) and increase the risk for metabolic derangements. Sedation from antipsychotics or other psychotropics may also decrease motivation to exercise. Most clinicians who work with these patients are familiar with their unhealthy eating habits. Thus, those with schizophrenia are less likely to be physically active and do not eat adequate amounts of fruits and vegetables [47]. In their study of the dietary intake of 88 chronic, mentally ill patients in an urban mental health clinic, however, Henderson et al. found diets lower in calories, fat, carbohydrates, and sodium, suggesting that poor dietary habits are not universal [48].

In summary, unhealthy lifestyle is common in schizophrenia and negatively impacts cardiometabolic risk.

Metabolic Syndrome

The presence of metabolic syndrome greatly increases the risk for developing diabetes and CAD. According to the National Cholesterol Education Program Adult Treatment Panel III Guidelines, the diagnosis of metabolic syndrome requires three of the five criteria:

- *Waist circumference* \geq 40 in. (men) or \geq 35 in. (women).
- Fasting triglycerides \geq 150 mg/dL.
- *High density lipoprotein* (HDL) \leq 40 mg/dL (men) or \leq 50 mg/dL (women).
- Blood pressure \geq 135/85 mmHg.
- Fasting blood glucose \geq 110 mg/dL.

Studies in schizophrenia report higher rates of metabolic syndrome, which significantly increases the risk of CAD, myocardial infarction, and stroke [48, 49].

Antipsychotic-Induced Weight Gain

A meta-analysis of 80 randomized controlled trials involving SGAs found that after 10 weeks of treatment, patients taking clozapine gained an average of 4.5 kg, olanzapine 4.2 kg, risperidone 2.1 kg, haloperidol 1.0 kg, and fluphenazine 0.4 kg [50]. A more recent meta-analysis of 21 placebo-controlled trials examined short-term (<12 weeks; n=1,742) and long-term (1 year; n=1,649) outcomes in subjects receiving amisulpride, haloperidol, olanzapine, risperidone, and ziprasidone [51]. With short-term treatment (i.e., 4–12 weeks), significant weight gain was associated with risperidone (17 %) and olanzapine (48 %; both p < 0.05 versus placebo) but not with haloperidol (12 %) or ziprasidone (9 %, both NS). Studies of 1-year treatment duration demonstrated clinically significant weight gain (defined as >7 % change from baseline) in those taking risperidone (39 %) and olanzapine (57 %, both p < 0.05 versus placebo) but not ziprasidone (17 %, NS). The CATIE Study demonstrated weight gain of >7 % from baseline in 30 % of those on olanzapine, 16 % on quetiapine, 14 % on risperidone, 12 % on perphenazine, and 7 % on ziprasidone [52]. Similar risk differentials appear to exist for hyperlipidemia, diabetes, and CAD. While many studies indicate that SGAs pose a higher risk compared with FGAs, the accuracy of this finding is debated in the scientific literature [53].

In summary, antipsychotics are associated with varying degrees of weight gain. For additional discussion of this issue, see Chap. 12.

Stroke

Higher rates of hypertension, hyperlipidemia, and diabetes lead to a higher risk for stroke. Tsai et al. examined 80,569 Taiwanese patients with schizophrenia over a 4-year period using a national health database [54]. Compared with 241,707 age- and sex-matched controls, stroke incidence (1.71 %) was higher than in the comparison population (1.22 %; no p value reported). After controlling for demographic variables and comorbid conditions, this still resulted in a small but significant increased risk (adjusted hazard ratio=1.13, CI=1.05–1.22). Similarly, a retrospective cohort study of 3,022 schizophrenia patients in Canada revealed adjusted relative risk for stroke of 1.5 (CI=1.2–2.0). While older age is a strong risk factor, younger age groups were also studied. A database review of 5,001 Taiwanese patients under 45 years of age hospitalized for schizophrenia found a stroke incidence of 2.46 % versus 0.96 % in the comparison group [55]. By contrast, Kang et al. found a reduced mortality rate (hazard ratio=0.35, CI=0.21–0.57) in schizophrenia patients hospitalized for stroke compared with non-schizophrenia patients, even when adjusting for age, sex, stroke type, and comorbidity [56].

Management of Cardiometabolic Disorders

There are three main approaches to address the outsize burden of cardiometabolic disease in these patients. The first two are preventive in nature and involve improved screening for common medical illnesses and increased utilization of strategies for risk factor modification. The third, discussed later in the chapter, is to improve the access to and quality of medical care.

Screening

The American Psychiatric Association and the American Diabetes Association issued joint guidelines for regular, systematic screening and monitoring of weight, cholesterol, blood sugar, blood pressure, and waist circumference for patients treated with antipsychotics; adherence to these guidelines, however, is suboptimal [57, 58]. These guidelines include monitoring for:

- Personal and family at the start of treatment and then yearly.
- Weight (BMI) at the start of treatment, monthly for 3 months, and then every 3 months.
- Waist circumference at the start of treatment and then yearly.
- Blood pressure at the start of treatment, at 3 months, and then yearly.
- Fasting blood glucose at the start of treatment, at 3 months, and then yearly.
- Fasting lipids at the start of treatment, at 3 months, and then every 5 years [57].

Psychiatric providers who prescribe these medications are typically the ones most in contact with the patient and need to be more cognizant of these guidelines. In this context, the advent of electronic medical records and the potential for computer-based reminders may help.

The authors also highlighted the obesity and metabolic risk differential among the various SGAs. Clozapine and olanzapine were considered highest risk, risperidone and quetiapine middle risk, and aripiprazole and ziprasidone lowest risk. A recent meta-analysis included 56 (mostly short-term) trials of the newer SGAs (i.e., asenapine n=9; iloperidone n=11; lurasidone n=8; paliperidone n=28) in schizophrenia (n=19,299) and bipolar disorder (n=2,392) [59]. The overall risk of weight gain was low. Among these agents, the greatest risk of clinically meaningful weight gain was associated with asenapine (RR=4.09, CI 2.25–7.43) followed by iloperidone (RR=3.13, CI 2.08–4.70) and paliperidone (RR=2.17, CI 1.64–2.86). Subjects on lurasidone did not demonstrate statistically significant weight gain (RR=1.42, CI 0.87–2.29).

In summary, mental health providers must be more proactive in following national guidelines for monitoring of metabolic parameters.

Risk Factor Modification

Approaches to reduce cardiometabolic burden include switching to a lower risk antipsychotic, adjunctive use of metformin or topiramate, and behavioral or nutritional counseling.

Several studies examined the benefit of switching antipsychotics, usually from a higher risk agent such as olanzapine to lower risk agents such as ziprasidone or aripiprazole. In one study, patients taking FGAs (n=108), olanzapine (n=104), or risperidone (n=58) were switched to ziprasidone (titrated up to 160 mg daily) for 6 weeks [60]. Mean weight decreased from 93.4 to 91.6 kg (p<0.001) for those originally on olanzapine. Weight loss was also noted in those switched from risperidone (i.e., 87.3–86.5 kg; p<0.05) but not from FGAs. In another study, patients with schizophrenia or schizoaffective disorder (n=84) experiencing glucose intolerance, dyslipidemia, type 2 diabetes, or weight gain with their current antipsychotic were switched to ziprasidone (mean dose=120 mg) [61]. Mean weight decreased by 5.1 kg after 6 months (p<0.0001). Reductions were also seen in serum glucose, cholesterol, and triglyceride levels (p<0.0001).

Weight reduction also occurred when switching patients to aripiprazole. Two large, multicenter trials reported reductions in weight and other metabolic parameters. In one study (n=173), switching from the current antipsychotic to aripiprazole resulted in a mean weight reduction of 1.3–1.7 kg depending on the switch strategy (significance not reported) [62]. A second study enrolled 173 olanzapine-treated subjects with schizophrenia or schizoaffective disorder randomized to continue olanzapine monotherapy or switch to aripiprazole monotherapy [63]. Over 16 weeks, the aripiprazole group lost 1.8 kg versus a gain of 1.4 kg in the olanzapine group (p < 0.001).

In summary, these studies indicate that it is important whenever possible to consider switching patients to antipsychotics with a lower cardiometabolic risk profile.

Pharmacologic Approaches

Metformin

Metformin is a first-line treatment for diabetes mellitus. It is associated with weight loss, as well as lower blood glucose and hemoglobin A_{1C} levels. It was studied for both prevention and reversal of antipsychotic-induced metabolic derangements.

The best evidence for metformin to treat *antipsychotic-induced weight gain* (*AIWG*) comes from three Chinese studies. The first involved patients with schizophrenia (n=128) who had gained ≥ 10 % of their baseline weight within 1 year of beginning an antipsychotic. Patients were randomized to placebo, metformin

(750 mg daily), metformin (750 mg daily) plus lifestyle intervention, or lifestyle intervention alone for 12 weeks [64]. Lifestyle intervention consisted of psychoeducation (e.g., roles of eating and activity in weight management, behavioral techniques); the American Heart Association Step 2 diet; and exercise sessions led by a physiologist. The lifestyle plus metformin group was superior to the metformin alone and lifestyle plus placebo for decreases in weight (4.7 kg (CI=3.4-5.7) versus)3.2 kg (CI=2.5-3.9) and 1.4 kg (CI=0.7-2.0), respectively; p < 0.05) and BMI (1.8) (CI=1.3-2.3) versus 1.2 (CI=0.9-1.5) and 0.5 (CI=0.3-0.8), respectively; p < 0.05). Similar trends were found for the insulin resistance index (a measure of glucose intolerance). All treatment groups had better outcomes in the above measures compared with placebo. A second Chinese study involved 84 first-episode women with schizophrenia randomized to 6 months of metformin (1.000 mg daily)or placebo in addition to their antipsychotic [65]. The metformin group experienced a mean weight loss of 2.3 kg while the placebo-treated patients gained a mean of 2.1 kg (p < 0.01). Finally, 72 first-episode patients who had gained >7 % of their baseline weight on an antipsychotic were randomized to metformin (1,000 mg daily) or placebo for 12 weeks [66]. Metformin led to a mean 3.3 kg weight loss while the placebo group gained a mean of 2.5 kg (p < 0.001). The insulin resistance index was also significantly decreased in the metformin group (p < 0.001). Of note, all patients in these studies had a normal baseline BMI (<25) which may not be representative of typical patients in the United States who tend to be overweight (BMI>25) or obese (BMI>30.0). A meta-analysis of metformin for AIWG included seven randomized, placebo-controlled studies (n=398) [67]. Metformin was associated with a significant body weight reduction in adults (4.8 %, CI=1.6-8.0). The most common adverse effects associated with metformin included gastrointestinal upset and diarrhea.

The best evidence supporting metformin for the *prevention* of antipsychoticassociated weight gain was conducted by Wu et al. [68]. Forty, never-medicated, first-episode patients were randomized to 12 weeks of olanzapine (15 mg daily) plus metformin (750 mg daily) or olanzapine (15 mg daily) plus placebo. The active treatment group gained 1.90 (± 2.72) kg, compared with 6.87 (± 4.23) kg in the placebo group (p < 0.02). Although this is an off-label use, metformin should be considered in any patient requiring an antipsychotic with a higher metabolic risk profile (e.g., olanzapine).

Topiramate

Topiramate is an antiepileptic associated with weight loss. Following several promising open-label studies, Narula et al. randomized 72 drug-naïve patients initiating treatment with olanzapine to also receive topiramate (100 mg daily) or placebo for 12 weeks [69]. Patients on topiramate lost an average of 1.27 (\pm 2.28) kg versus a 6.03 (\pm 3.16) kg weight gain with placebo (p<0.001). A second study randomized 43 female patients (diagnosis not reported) who had gained weight on olanzapine to

also receive topiramate (250 mg daily) or placebo for 10 weeks [70]. Weight loss was more significant in the topiramate group, though specific changes in weight were only reported in graph form. An 18-month, open-label extension of this study continued in the same 43 subjects [71]. Patients in the former placebo group received no adjunctive medication, while the former topiramate group continued on open-label medication. Mean weight in the active treatment group decreased from 82.8 (\pm 11.3) kg to 77.2 (\pm 3.9) kg, while the placebo group increased from 86.2 (\pm 10) kg to 90.6 (\pm 8.8) kg. Finally, a 12-week placebo-controlled trial randomized 66 hospitalized patients with first-episode schizophrenia to topiramate (100 mg or 200 mg daily) or placebo for 12 weeks [72]. A significant decrease in body weight (-5.35 kg) or placebo (-0.4 kg) (p<0.01).

In summary, although not FDA-approved, the best evidence for adjunctive pharmacotherapy to attenuate AIWG supports metformin. Topiramate may be a reasonable alternate choice.

Nonpharmacological Approaches

Nonpharmacologic interventions are also useful for weight loss and cardiometabolic risk reduction. These interventions involve cognitive behavioral therapy (CBT), nutritional counseling, and nutritional counseling combined with exercise interventions conducted in both individual and group formats. A meta-analysis of 10 randomized controlled studies of non-pharmacological interventions for AIWG included a total of 482 patients, 75 % with a schizophrenia spectrum diagnosis [73]. Six of the 10 studies targeted weight loss and four targeted prevention of weight gain. The length of the studies ranged from 8 weeks to 6 months with a 2- to 3-month follow-up. All but one study involved outpatients. The authors found a statistically significant reduction in mean body weight for those in the active treatment group versus a treatment as usual (TAU) condition (weighted mean difference=-2.5 kg; CI=-3.20 to -1.90).

Caemmerer et al. conducted a more recent meta-analysis, adding an additional six, nonrandomized studies to examine the effect of nonpharmacologic interventions for antipsychotic-induced metabolic abnormalities [74]. Their analysis included 810 patients with schizophrenia, schizoaffective disorder, schizophreniform disorder, unipolar depression, or bipolar disorder (42 % unknown). Compared with the control conditions, CBT and nutritional and/or exercise interventions resulted in significantly less weight gain over 12 months (weighted mean difference = -3.48 kg, CI = -6.37 to -0.58). In addition, there were significant reductions in percent body fat, insulin levels, glucose, total cholesterol, LDL-cholesterol, and triglycerides.

In summary, evidence supports nonpharmacologic intervention for AIWG. Implementation of these interventions in real world settings, however, is limited by lack of access to services, the relative paucity of appropriately trained providers, and lack of reimbursement for such services. Nonetheless, clinicians should counsel their patients on healthy diet and exercise.

Infectious Diseases

The rates of human immunodeficiency virus (HIV) and hepatitis B and C are higher in schizophrenia than the general population. For example, Rosenberg et al. studied 931 inpatients and outpatients with severe mental illness (50 % schizophrenia, 20 % schizoaffective disorder, 17 % bipolar disorder) [75]. They found an HIV infection rate of 3.1 % which was eight times higher than the general population rate at that time. Further, there was a 23.4 % hepatitis B infection rate and a 19.6 % hepatitis C infection rate, both 5–11 times above general population rates. Studies confirm that the increased prevalence of high risk behaviors (e.g., intravenous drug use; unprotected sexual intercourse) in the mentally ill leads to higher infection rates [76]. A more recent review of 595 Belgian patients with schizophrenia, however, did not find a higher prevalence of hepatitis C compared with the general population, though HIV rates were elevated (i.e., 0.5 % or double the rate in the general population) [77].

Human Immunodeficiency Virus

HIV is a retrovirus that causes acquired immunodeficiency syndrome (AIDS). This syndrome is characterized by a weakening of the immune system leading to life-threatening opportunistic infections and cancers. A cross-sectional study of Medicaid recipients in Philadelphia found an HIV prevalence of 1.8 % among those with a schizophrenia spectrum disorder compared with 0.6 % in those without a psychiatric illness (p<0.001) [78]. A retrospective study from a national VA-based registry of HIV patients (n=9,003) found that having a chronic mental illness such as schizophrenia or bipolar disorder increased HIV-related all-cause mortality [79]. HIV infection is treated with highly active antiretroviral therapy (HAART).

Hepatitis C

Hepatitis C is a chronic liver infection which may lead to cirrhosis and hepatic failure. Studies describe higher rates of hepatitis C in chronic mentally ill patients. For example, in one study of patients (n=1,556) in a psychiatric hospital, 8.5 % were seroprevalent for hepatitis C versus 1.8 % in the general population [80]. Sockalingham et al. surveyed 110 patients in a Canadian clozapine clinic and found that 2.7 % were positive for hepatitis C, compared with 0.8 % in the general population [81]. HIV and hepatitis laboratory evaluations were conducted in 668 patients in four public sector clinics for the chronic mentally ill [82]. Eighteen percent (n=122) were positive for hepatitis C (of these, 53 had only hepatitis C; 56 had hepatitis B and C; and 13 had hepatitis C and HIV). More than 20 % of those had injected drugs, 14 % had shared needles, and more than 20 % used crack cocaine, all of which are associated with an increased risk of hepatitis C. Similarly, higher

than normal rates of hepatitis C (prevalence=9.1 %) were described in hospitalized patients with schizophrenia and comorbid psychoactive substance abuse (n=1,193) in Japan [83].

Compared with controls, schizophrenia patients in a large VA database were two times more likely to get tested for hepatitis C and also two times more likely to be infected [84]. Of note, patients in this study were as likely to receive treatment compared with controls. Butterfield et al. found that hepatitis C in the chronic mentally ill occurred twice as often in men versus women [85]. Higher rates of needle sharing and crack cocaine use occurred in men while women were more likely to report unprotected intercourse. In contrast to Osher et al., 2 years after a screening for hepatitis C in a cohort of 98 patients with schizophrenia none of the patients (n=8) who tested positive had received treatment, again underscoring the inadequate access to care in this population [86].

Hepatitis B

Hepatitis B viral infection is spread via contact with infected blood and body fluids. Most patients develop subclinical or anicteric (no jaundice) hepatitis. In adults, most patients recover and do not develop chronic hepatitis B. Nonspecific constitutional symptoms consisting of fever, fatigue, headache, cough, nausea, and vomiting may be present [87]. Jaundice, commonly associated with hepatic diseases, develops in the next 7–14 days.

Management

There are recommendations for public sector mental health systems to better address infectious diseases in this population. The Centers for Disease Control (CDC) published guidelines addressing the increased risk for hepatitis and HIV [88]. All patients with a chronic mental illness should be screened for sharing of drug use paraphernalia and high risk sexual activity such as unprotected intercourse with a high risk person. Patients with any of these risk factors should be tested for the HIV-1 antibody, the hepatitis C antibody, and the hepatitis B surface and core antibodies and hepatitis B surface antigen [89]. The CDC recently expanded their recommendations, stating that all patients between the ages of 15 and 65, regardless of risk status, should be tested for the hepatitis C antibody.

A brief intervention to increase screening and treatment in chronic mentally ill patients was undertaken by Rosenberg et al. in four community-based mental health clinics [90]. The investigators recruited 236 patients to the intervention, entitled Screening, Testing for HIV and Hepatitis, Immunization for Hepatitis A and B, Risk Reduction Counseling and Medical Treatment and Referral and

Support (STIRR), in the mental health clinic. The treatment group was more likely to be tested for hepatitis B and C, increase their knowledge about hepatitis, and reduce their substance use. High risk behaviors, however, did not decrease; patients were not more likely to be referred for specialty care and HIV knowledge did not increase [90].

Counseling patients on risk reduction—avoiding unprotected sexual intercourse and sharing of drug paraphernalia—is essential. While CBT-based group interventions for HIV risk reduction are effective for the chronic mentally ill, they are not widely available [91]. At a minimum, psychiatrists should educate patients about the risks and complications of these infectious diseases in an effort to motivate them to change associated behaviors.

Chronic hepatitis C infection is treated with ribavirin and pegylated interferon. Pegylated interferon is associated with various neuropsychiatric symptoms, including depression, suicide, and fatigue. This led to hesitation in providing these treatments for patients with schizophrenia. Huckans et al., however, reviewed a VA database of patients with schizophrenia who received hepatitis C antiviral treatment (n=30) and compared them with hepatitis C patients without schizophrenia from the same database [92]. Patients with schizophrenia were equally likely to reach end of treatment response (ETR) and sustained viral response (SVR), indicating no differential in outcome of adverse effects. Further, patients with schizophrenia in a VA database (n=30) who also received antiviral treatment for hepatitis C were followed over an 8-year period. Compared to controls, these patients did not experience higher rates of symptoms of schizophrenia, depression, or mania [93]. While interferon is not contraindicated in these patients, close monitoring is necessary for early detection of treatment-emergent depressive symptoms. Psychiatric symptom control should be optimized prior to starting treatment for hepatitis C.

The hepatitis B vaccine is delivered in a series of three injections. Anyone engaging in unsafe sex practices or sharing drug paraphernalia should receive the vaccine. In addition, the CDC recommends the hepatitis B vaccine for any sexually active person not engaged in a long-term monogamous relationship and men who have sex with men. Immunization and testing should be integrated into regular ongoing mental health care. Patients who test positive for any of these illnesses should be referred for treatment. Current treatment of chronic viral hepatitis B includes pegylated interferon, entecavir, or tenofovir [87].

Cancer

Studies in schizophrenia provide conflicting results with some finding a *lower* cancer incidence. For instance, an 8-year cohort of 59,257 patients with schizophrenia in the Taiwanese NIHRD was compared with 178,156 age- and gender-matched controls without schizophrenia [94]. Overall, cancer incidence was lower in those with schizophrenia (OR=0.81, CI=0.74–0.88). Lower rates were found for all types except breast and cervical/uterine cancer. Mortality, however, increased (hazard

ratio=1.36, CI 1.24–1.50) which may reflect issues of quality and access to care. Similarly, a linkage study connecting national cancer and psychiatric databases found reduced cancer rates for both schizophrenia patients and their parents [95].

Other studies, however, found cancer rates similar to or greater than the general population. In this context, Goldacre et al. employed a cohort analysis of cancer incidence in schizophrenia compared with a reference population reporting that the rate ratio for cancer overall was 1.02 (CI=0.90-1.08) [96]. Except for a significantly lower rate ratio for skin cancer (0.56, CI=0.36-0.83), risk of individual cancers was not reduced. A population-based, nested, case–control study found that patients with schizophrenia experienced a 190 % increased risk of colon cancer (OR=2.90, CI=1.85-4.57) and a 52 % increased risk for breast cancer (OR=1.52, CI=1.20-2.11) [97]. Of note, the increased colon cancer risk was most pronounced in patients taking antipsychotics (OR=4.08, CI=2.43-6.08). Finally, a recent study calculated standard incidence ratios (SIRs) of total and site-specific cancers in more than 3,000 Maryland Medicaid beneficiaries over a 10-year period [98]. For those with schizophrenia, the SIR for all cancer sites was 2.6 (CI=2.2-3.0) with the highest being for lung cancer (SIR=4.7, CI=3.1-6.8), though the authors did not control for tobacco use.

A meta-analysis of 16 studies involving patients with schizophrenia and their first-degree family members found that the effect of the illness on cancer rates depended on the type of cancer [99]. Thus, while the overall cancer incidence was not significantly increased (SIR = 1.05, CI = 0.95-1.15), the rate of lung cancer was (SIR = 1.31, CI = 1.01 - 1.71). This, however, no longer reached significance once the authors controlled for tobacco use. Breast cancer was also more common in schizophrenia (SIR=1.02, CI=1.02-1.23). Several cancers including colorectal, malignant melanoma, and prostate had a reduced incidence. Cancer incidence may also vary by age of onset. Lin et al. examined a nationwide cohort of 102,292 patients in the Taiwanese NHIRD during a 7-year period [100]. Overall, the SIR of any cancer declined with age (i.e., for ages 20-29, SIR=1.97 (CI 1.85-2.33); for ages 60-69, SIR = 0.68 (CI 0.65–0.45)). Further, SIRs for cancers that develop at an older age were lower for prostate cancer (i.e., 0.35 (CI=0.29-0.58); stomach cancer 0.62 (CI=0.57-0.80); and pancreatic cancer 0.49 (CI=0.39-0.84)). By contrast, cancers with a younger age of onset were more common (i.e., for nasopharyngeal cancer SIR = 1.18 (CI = 1.08–1.49); breast cancer SIR = 1.49 (CI 1.44–1.66); uterine cancer SIR = 2.15 (CI = 1.98 - 2.74)).

Cancer Mortality

Studies demonstrate that cancer is also a major contributor to the increased overall mortality in schizophrenia [3, 99]. Crump et al. followed 8,277 Swedes with schizophrenia over a 7-year period [3]. Despite frequent contact with the health care system, men died 15 years earlier and women 12 years earlier compared with the general population, suggesting suboptimal disease management. The two main causes of death

were heart disease and cancer. The authors concluded that improved screening and risk factor modification were needed. Capasso et al. described 319 Minnesota residents with schizophrenia followed for over 23 years [101]. Cancer was the number two cause of mortality (19 %) in this group which was significantly higher than the general population. A prospective cohort study included more than 3,000 French patients with schizophrenia followed for 11 years [102]. Mortality rates were four times higher than in nonpsychiatric patients, with cancer being the second most common cause. Compared with the general population, the global SMR for any type of cancer was 1.5 (CI=1.2–1.9).

Less is known about cancer rates in schizoaffective disorder. One linkage study using a database in the Jewish-Israeli general population calculated SIRs for all types of cancer in patients with schizoaffective disorder and found no increased risk (SIR = 1.11, CI = 0.48-1.73) [103].

Breast Cancer

A systematic review of 13 studies concluded there was a higher incidence of breast cancer in schizophrenia [104]. Only 6 of the 13 studies, however, reported elevated incidence rates which varied from a 52 % increase to a 40 % decrease. Another review of breast cancer in schizophrenia identified the following relevant risk factors: obesity, elevated prolactin, low rates of screening with mammography, low parity, low rates of breastfeeding, high levels of smoking and alcohol use, and sed-entary lifestyle [105].

Breast cancer treatment in schizophrenia is the subject of only a small number of studies. Using case examples, Cole and Padmanabhan described the challenges of treating breast cancer in women with schizophrenia [106]. Understanding their illness, medication adherence and dealing with psychiatric exacerbations during cancer treatment were among the issues cited by the authors. Nonetheless, a longitudinal study of 37 women with schizophrenia and breast cancer found no impact of the psychiatric illness for either treatment delivery or outcomes [107].

Lung Cancer

Most studies find a higher incidence of lung cancer in schizophrenia. Lichtermann conducted a large record linkage study in 27,000 Finn patients, reporting an elevated SIR of 2.17 (CI=1.78–2.60) [108]. A large study (n=147,973) in Danish patients reported an incidence rate ratio of 3.03 [109]. Neither of these studies, however, controlled for smoking. This is important, since the increased incidence typically disappears when studies control for tobacco use. A retrospective study of lung cancer in 29 patients with schizophrenia did not find a disparity in care, defined as less aggressive therapy prescribed for a potentially curable cancer [110].

Breast cancer	Recommends biennial screening mammography for women aged 50-74 years
Colorectal cancer	Recommends screening using fecal occult blood testing, sigmoidoscopy, or colonoscopy in adults beginning at age 50 continuing until age 75
Cervical cancer	Recommends screening women age 21–65 with PAP smear every 3 years or for women age 30–65 with combination of PAP and human papillomavirus testing every 5 years
Prostate cancer	Recommends against screening with PSA blood test
Lung cancer	Recommends annual screening with low-dose computed tomography in adults ages 55–80 years who have a 30 pack-year smoking history and currently smoke or have quit within the past 15 years
Ovarian cancer	Recommends against screening
Pancreatic cancer	Recommends against screening
Testicular cancer	Recommends against screening
Source: United Sta	ates Preventive Service Task Force [148]

 Table 10.2
 Screening recommendations for selected cancers (abbreviated)

Management

Evidence indicates that patients with schizophrenia receive inadequate cancer screening [111]. Thus, all patients should receive age- and risk- appropriate screening for common cancers. Table 10.2 shows selected cancer screening recommendations from the United States Preventive Services Taskforce. Colorectal cancer screening should begin at age 50 and continue to age 75. Bienniel mammography is recommended for women between the ages of 50 and 75. Prostate cancer screening with prostate-specific antigen (PSA) testing is not currently recommended, though there is some controversy surrounding this. Screening for lung cancer with low-dose CT scan is recommended in adults ages 55 to 80 years who have a 30 pack-year smoking history and currently smoke or have quit within the 15 years. Psychiatric providers can play a proactive role in counseling patients to obtain age-appropriate cancer screening. In addition, bupropion is safe and effective for tobacco cessation in these patients.

In summary, studies find both elevated and decreased cancer rates in schizophrenia. Cancer-related mortality, however, appears to be consistently elevated in schizophrenia. The reasons for this are unclear, though delayed screening and diagnosis, inadequate risk factor modification, and suboptimal or incomplete treatment are possible reasons.

Substance Abuse

A full discussion of substance abuse in schizophrenia is beyond the scope of this chapter. It is important, however, to understand its health impact on patients. According to the Epidemiologic Catchment Area Study, nearly 50 % of patients with

schizophrenia or schizophreniform disorder have a diagnosis of substance abuse or dependence [112]. Studies report the following prevalences: nicotine 50–90 %; alcohol 20–60 %; cannabis 12–42 %; and cocaine 15–50 % [113]. Risk factors for substance use disorder include younger age, male gender, early age of schizophrenia onset, and lower levels of negative symptoms [114]. Substance abuse is associated with greater morbidity, greater deterioration in function, and reduced concentration [115]. Further, it is associated with increased risk for infectious hepatitis and HIV. Cannabis or cocaine use may lead to higher risk of tardive dyskinesia, while nicotine use increases the risk for lung cancer and CAD.

Management

All patients should be screened for substance use. Urine and blood toxicology may aid in detection. Development of a strong therapeutic alliance is important to increase awareness of the problem and help motivate the patient to change. Integrated treatment (by the same clinician or team of clinicians) of both schizophrenia and the substance use disorder is the most effective approach in this population. SGAs may be more beneficial in dual diagnosis patients, with clozapine in particular showing greater benefits for decreasing nicotine and alcohol use [116, 117]. Nicotine replacement, CBT, and bupropion are all successful in achieving tobacco cessation. For example, a 12-week trial of bupropion 300 mg daily or placebo added to transdermal nicotine patch 21 mg/24 h, nicotine gum, and CBT was conducted in 51 adults with schizophrenia. Bupropion achieved greater rates of 50–100 % smoking reduction (60 % versus 31 %, p=0.036) compared with placebo [118]. Relapse rates were high, however, upon discontinuation. No worsening of psychosis was noted with bupropion. Naltrexone is effective for reducing alcohol intake. In this context, naltrexone 100–150 mg per day directly administered to 19 patients with schizophrenia for 8 weeks resulted in a significant reduction in number of drinks per week, number of drinks per drinking day, and alcohol craving [119].

Other Medical Conditions

Chronic Obstructive Pulmonary Disease and Pneumonia

Population studies demonstrate an increased risk of lung diseases such as COPD and pneumonia in patients with schizophrenia. Using a national Taiwanese database of nearly 800,000 people, researchers compared rates of COPD in schizophrenia versus the general population [120]. COPD was more prevalent in schizophrenia (3.8 % versus 2.9 %) with an OR of 1.66 (1.42–1.94) and the annual incidence of COPD was 2.21 % versus 1.43 % in the general population. Among veterans who

died while in the hospital, schizophrenia was associated with a diagnosis of COPD or pneumonia in the last year of life [121]. Similarly, a record linkage study of patient databases in England found that schizophrenia increased the likelihood of developing lobar pneumonia, pneumococcal pneumonia, or pneumococcal sepsis [122]. Increased rates of COPD are most likely linked to higher rates of nicotine use in this population.

There is no screening recommendation for asymptomatic individuals at risk for COPD. Addressing nicotine use, as already discussed, is essential. Patients with chronic medical conditions (i.e., heart or lung disease; cancer) and all patients over the age of 65 should receive the pneumonia vaccine.

Autoimmune Disorders

Immune dysregulation is implicated in the pathophysiology of schizophrenia. A cohort study linking two large Danish population-based registers examined the relationship between atopic disorders such as asthma and schizophrenia [123]. Of note, asthma appeared to increase the risk of having schizophrenia (RR = 1.59; 1.31–1.9). The authors, however, used a sibling comparator group instead of a general population comparison group. In contrast, a Taiwanese research group examined a large health insurance database and compared 44,187 individuals with schizophrenia with 132,561 matched nonpsychiatric controls [124]. Patients had a 30 % greater risk of developing asthma (RR = 1.3; 1.24–1.39). Interestingly, they were not at greater risk for other allergic disorders such as rhinitis or urticaria.

A linkage study in a Danish national patient register and a database of 7,704 patients with schizophrenia also found a 50 % higher prevalence of other autoimmune disorders compared with the general population [125]. Acquired hemolytic anemia, intestinal malabsorption, celiac disease, thyrotoxicosis, Sjogren's syndrome, interstitial cystitis, alopecia areata, and polymyalgia rheumatica were among the disorders found to occur more frequently in schizophrenia (p < 0.05). Shared disease-specific or non-disease-specific genetic loci are proposed as the putative cause. Thyroid dysfunction in antipsychotic-treated patients is also reported in several studies [126].

In contrast, rheumatoid arthritis (RA) is consistently shown to have a negative relationship with schizophrenia. A large review of 14 studies conducted between 1934 and 1985 found a negative association between schizophrenia and rheumatoid arthritis [127]. Twelve of the 14 studies found a reduced risk of rheumatoid arthritis in patients with schizophrenia, mirroring findings of other studies [128, 129]. Alterations in platelet activating factor (PAF) may explain this relationship [130]. PAF is pro-inflammatory and may predispose to RA, yet it is also responsible for neuronal migration and synaptic connectivity, so reduced bioavailability may predispose to schizophrenia.

Epilepsy

Psychosis is common among patients with epilepsy and may present with hallucinations and delusions, psychomotor retardation, and conceptual disorganization [131, 132]. In particular, patients with temporal lobe epilepsy are more likely to develop psychosis resembling schizophrenia [133]. In turn, epilepsy is more common in patients with schizophrenia [131]. Wotton et al. conducted a retrospective cohort study linking records within a national hospital statistics database in the United Kingdom [134]. Patients hospitalized for schizophrenia were twice as likely to have epilepsy while those admitted for epilepsy were five times more likely to have schizophrenia. Another study used the Finnish Hospital Discharge Register to follow parents and children born between 1947 and 1960 [135]. The authors found that individuals with a parental history of epilepsy had twice the risk of developing psychosis and those with parental history of psychosis had nearly three times the risk of developing epilepsy.

The co-occurrence of these disorders led some to propose a possible link based on shared genetic or neuropathologic etiologies. Ventricular enlargement in both conditions points to a shared neuropathologic etiology; left temporal lobe pathology may also be a shared putative factor [136]. Leucine-rich glioma-inactivated (LGI) family gene loci were also studied as potential causative factors [137]. Increased risk of seizure due to antipsychotics further complicates the issue.

Future Needs: Integrated Delivery of Medical Care

There is a need to improve screening of patients with schizophrenia for common chronic medical conditions such as diabetes and hypertension. Early interventions need to emphasize exercise and healthy diet and address nicotine use [138]. Improving quality and access to care is challenging and requires integrated models to better address patients' physical health [139]. Interventions to improve the integration of medical and psychiatric care are increasingly a focus of research. Co-located models of care may bring internists into psychiatric settings. Alternatively, mental health providers may be brought into medical settings to identify and treat psychiatric disorders, though these models more commonly target the primary care management of depression [140]. Finally, practicing psychiatrists may be trained in primary management of common medical conditions. This is a recent focus of various professional organizations. Of note, however, psychiatrists have been slow to implement metabolic monitoring guidelines for patients on antipsychotics [58].

Collaborative models involving a co-located primary care doctor or a consulting internist are one of the better studied approaches. Druss et al. randomized 120 patients in the VA system to a primary care medical clinic (contiguous to a mental health clinic) or care as usual (referral to general medical clinic in an adjacent building) [141]. The medical clinic was staffed with an on-site primary care physician, a nurse practitioner, and a nurse care manager. After 1 year, 15 of 17 measures of

quality of medical care were significantly better than the control group. Inpatient settings are also a focus of study. For example, 130 hospitalized patients with chronic mental illness were randomized to either medical care provided by a consulting internist (who conducted medical history and physical exam and communicated with the outpatient primary care provider) or medical care as usual provided by psychiatric house staff [142]. Twelve of 17 quality measures were improved in the intervention group and the cost between the two was similar (i.e., \$8,558 versus \$8,527). A variation on this approach is a model which employs a non-physician manager in a mental health clinic to facilitate medical care for chronic mentally ill patients [143]. In one study, patients with serious mental illness in an urban mental health clinic (n=407) were randomized to medical care management or TAU for 1 year. Participants in the intervention group received 58.7 % of recommended preventive services versus only 21.8 % in the control group.

A systematic review of interventions for the chronic mentally ill concluded that regardless of whether services were co-located, the most important element was integration of the treatment team, as this has demonstrated improved quality of general medical and psychiatric care [144, 145]. Scharf et al. describe the early experiences of the 56 Primary and Behavioral Health Care Integration grantees, all of whom designed models for combined medical and psychiatric care within their current systems [146]. Data collection on the effect of these interventions is ongoing and should inform providers and mental health systems on how best to integrate medical and psychiatric care to enhance patient outcomes.

Conclusion

Patients with schizophrenia suffer from several medical disorders at higher rates and experience a shortened life-span than the general population. Causes are multifactorial and include lifestyle factors, delivery of medical care, and medication adverse effects. Thus, mental health providers need to play an increasingly proactive role in the physical health of psychiatric patients—including screening for chronic medical conditions and substance abuse and addressing modifiable risk factors. Finally, there is a need for innovative delivery approaches of medical care for patients with serious mental illness.

References

- Carney CP, Jones L, Woolson RF. Medical comorbidity in women and men with schizophrenia: a population-based controlled study. J Gen Intern Med. 2006;21(11):1133–7.
- Filik R, Sipos A, Kehoe PG, et al. The cardiovascular and respiratory health of people with schizophrenia. Acta Psychiatr Scand. 2006;113:298–305.
- Crump C, Winkleby MA, Sundquist K, Sundquist J. Comorbidities and mortality in persons with schizophrenia: a Swedish national cohort study. Am J Psychiatry. 2013;170:324–33.

- 4. Liao CC, Shen WW, Chang CC, Chang H, Chen TL. Surgical adverse outcomes in patients with schizophrenia: a population-based study. Ann Surg. 2013;257:433–8.
- Simpson JC, Tsuang MT. Mortality among patients with schizophrenia. Schizophr Bull. 1996;22(3):485–99.
- Brown S, Inskip H, Barraclough B. Causes of the excess mortality of schizophrenia. Br J Psychiatry. 2000;177:212–7.
- Enger C, Weatherby L, Reynolds RF, Glasser DB, Walker AM. Serious cardiovascular events and mortality among patients with schizophrenia. J Nerv Ment Dis. 2004;192:19–27.
- 8. Ran MS, Chen EYH, Conwell Y, et al. Mortality in people with schizophrenia in rural China: a 10-year cohort study. Br J Psychiatry. 2007;190:237–42.
- 9. Fors BM, Isacson D, Bingefors K, Widerlov B. Mortality among persons with schizophrenia in Sweden: an epidemiological study. Nord J Psychiatry. 2007;61:252–9.
- 10. Brown S. Excess mortality of schizophrenia. A meta-analysis. Br J Psychiatry. 1997;171(12):502-8.
- 11. Harris EC, Barraclough B. Excess mortality of mental disorder. Br J Psychiatry. 1998;173(7):11–53.
- 12. Hennekens CH, Hennekens AR, Hollar D, Casey DE. Schizophrenia and increased risks of cardiovascular disease. Am Heart J. 2005;150:1115–21.
- Rasanen S, Hakko H, Viilo K, Meyer-Rochow VB, Moring J. Avoidable mortality in longstay psychiatric patients of Northern Finland. Nord J Psychiatry. 2005;59:103–8.
- 14. Morgan MG, Scully PJ, Youssef HA, Kinsella A, Owens JM, Waddington JL. Prospective analysis of premature mortality in schizophrenia in relation to health service engagement: a 7.5-year study within an epidemiologically complete, homogeneous population in rural Ireland. Psychiatry Res. 2003;117:127–35.
- Bradford DW, Kim MM, Braxton LE, Marx CE, Butterfield M, Elbogen EB. Access to medical care among persons with psychotic and major affective disorders. Psychiatr Serv. 2008;59:847–52.
- Cradock-O'Leary J, Young AS, Yano EM, Wang M, Lee ML. Use of general medical services by VA patients with psychiatric disorders. Psychiatr Serv. 2002;53:874–8.
- Drapalski AL, Milford J, Goldberg RW, Brown CH, Dixon LB. Perceived barriers to medical care and mental health care among veterans with serious mental illness. Psychiatr Serv. 2008;59:921–4.
- Muir-Cochrane E. Medical co-morbidity risk factors and barriers to care for people with schizophrenia. J Psychiatr Ment Health Nurs. 2006;13(4):447–52.
- Druss BG, Bradford DW, Rosenheck RA, Radford MJ, Krumholz HM. Mental disorders and use of cardiovascular procedures after myocardial infarction. JAMA. 2000;283(4):506–11.
- Briskman I, Bar G, Boaz M, Shargorodsky M. Impact of co-morbid mental illness on the diagnosis and management of patients hospitalized for medical conditions in a general hospital. Int J Psychiatry Med. 2012;43(4):339–48.
- Nasrallah HA. Neurologic comorbidities in schizophrenia. J Clin Psychiatry. 2005;66 Suppl 6:34–46.
- Vahia IV, Diwan S, Bankole AO, et al. Adequacy of medical treatment among older persons with schizophrenia. Psychiatr Serv. 2008;59:853–9.
- Mitchell AJ, Lawrence D. Revascularization and mortality rates following acute coronary syndromes in people with severe mental illness: comparative meta-analysis. Br J Psychiatry. 2011;198:424–41.
- Whyte S, Penny C, Phelan M, Hippsley-Cox J, Majeed A. Clinical care and delivery quality of diabetes care in patients with schizophrenia and bipolar disorder: cross-sectional study. Diabet Med. 2007;24:1442–8.
- Folsom DP, McCahill M, Bartels SJ, Lindamer LA, Ganiats TG, Jeste DV. Medical comorbidity and receipt of medical care by older homeless people with schizophrenia or depression. Psychiatr Serv. 2002;53:1456–60.
- Martens PJ, Chochinov HM, Prior HJ, Fransoo R, Burland E; Need To Know Team. Are cervical cancer screening rates different for women with schizophrenia? A Manitoba population-based study. Schizophr Res. 2009;113(1):101–6.

- Daumit GL, Pronovost PJ, Anthony CB, Guallar E, Steinwachs DM, Ford DE. Adverse events during medical and surgical hospitalizations for persons with schizophrenia. Arch Gen Psychiatry. 2006;63:267–72.
- World Health Organization. High blood pressure-country experiences and effective interventions utilized across the European Region. WHO Regional Office for Europe. 2009:1–23. http://www.euro.who.int/__data/assets/pdf_file/0008/185903/e96816.pdf. Accessed 31 July 2013.
- Kilbourne AM, Morden NE, Austen K, et al. Excess heart-disease-related mortality in a national study of patients with mental disorders: identifying modifiable risk factors. Gen Hosp Psychiatry. 2009;31(6):555–63.
- Nasrallah HA, Meyer JM, Goff DC, et al. Low rates of treatment for hypertension, dyslipidemia and diabetes in schizophrenia: data from the CATIE schizophrenia trial sample at baseline. Schizophr Res. 2006;86:15–22.
- McEvoy JP, Lieberman JA, Perkins DO, et al. Efficacy and tolerability of olanzapine, quetiapine, and risperidone in the treatment of early psychosis: a randomized, double-blind 52-week comparison. Am J Psychiatry. 2007;164:1050–60.
- 32. De Hert M, Schreurs V, Sweers K, et al. Typical and atypical antipsychotics differentially affect long-term incidence rates of the metabolic syndrome in first-episode patients with schizophrenia: a retrospective chart review. Schizophr Res. 2008;101(1–3):295–303.
- 33. Ferreira L, Belo A, Abreu-Lima C; RICAVA Study Group. A case-control study of cardiovascular risk factors and cardiovascular risk among patients with schizophrenia in a country in the low cardiovascular risk region of Europe. Rev Port Cardiol. 2010;29(10):1481–93.
- Saddichha S, Vishnuvardhan G, Akhtar S. Obesity, diabetes and hypertension associated with antipsychotic use in remitted schizophrenia. Int J Risk Saf Med. 2011;23(3):181–5.
- 35. James PA, Oparil S, Carter BL, et al. Evidence-based guideline for the management of high blood pressure in adults: report from the panel members appointed to the eighth joint national committee (JNC 8). JAMA. 2014;311(5):507–20.
- 36. Hsu JH, Chien IC, Lin CH, Chou YJ, Chou P. Hyperlipidemia in patients with schizophrenia: a national population-based study. Gen Hosp Psychiatry. 2012;34(4):360–7.
- Huang TL, Chen JF. Serum lipid profiles and schizophrenia: effects of conventional or atypical antipsychotic drugs in Taiwan. Schizophr Res. 2005;80(1):55–9.
- Lund BC, Perry PJ, Brooks JM, Arndt S. Clozapine use in patients with schizophrenia and the risk of diabetes, hyperlipidemia, and hypertension: a claims-based approach. Arch Gen Psychiatry. 2001;58(12):1172–6.
- 39. Liao CH, Chang CS, Wei WC, et al. Schizophrenia patients at higher risk of diabetes, hypertension and hyperlipidemia: a population-based study. Schizophr Res. 2011; 126:110–6.
- Spelman LM, Walsh PI, Sharifi N, Collins P, Thakore JH. Impaired glucose tolerance in firstepisode drug-naïve patients with schizophrenia. Diabet Med. 2007;24:481–5.
- Ryan MCM, Collins P, Thakore JH. Impaired fasting glucose tolerance in first-episode, drugnaïve patients with schizophrenia. Am J Psychiatry. 2003;160:284–9.
- 42. Sengupta S, Parrilla-Escobar MA, Klink RA, et al. Are metabolic indices different between drug-naïve first-episode psychosis patients and healthy controls? Schizophr Res. 2008;102:329–36.
- Bresee LC, Majumdar SR, Patten SB, Johnson JA. Prevalence of cardiovascular risk factors and disease in people with schizophrenia: a population-based study. Schizophr Res. 2010;117(10):75–82.
- 44. Curkendall SM, Mo J, Glasser DB, Stang RM, Jones JK. Cardiovascular disease in patients with schizophrenia in Saskatchewan, Canada. J Clin Psychiatry. 2004;65(5):715–20.
- 45. Homel P, Casey D, Allison DB. Changes in body mass index for individuals with and without schizophrenia. Schizophr Res. 2002;55(3):277–84.
- 46. McCreadie RG on behalf of the Scottish Schizophrenia Lifestyle Group. Diet, smoking and cardiovascular risk in people with schizophrenia: descriptive study. Br J Psychiatry. 2003;183:534–9.

- Henderson DC, Borba CP, Daley TB, et al. Dietary intake profile of patients with schizophrenia. Ann Clin Psychiatry. 2006;18(2):99–105.
- Kato MM, Currier MB, Gomez CM, et al. Prevalence of metabolic syndrome in hispanic and non-hispanic patients with schizophrenia. Prim Care Companion J Clin Psychiatry. 2004;6:74–7.
- 49. Isomma B, Almgren P, Tuomi T, et al. Cardiovascular morbidity and mortality associated with the metabolic syndrome. Diabetes Care. 2001;24(4):683–9.
- Allison DB, Mentore JL, Heo M, et al. Antipsychotic-induced weight gain: a comprehensive research synthesis. Am J Psychiatry. 1999;156:1686–96.
- 51. Parsons B, Allison DB, Loebel A, et al. Weight effects associated with antipsychotics: a comprehensive database analysis. Schizophr Res. 2009;110(1–3):103–10.
- Lieberman JA, Stroup S, McEvoy JP, et al. Effectiveness of antipsychotic drugs in patients with chronic schizophrenia. N Engl J Med. 2005;353:1209–23.
- Citrome LL, Holt RL, Zachry WM, et al. Risk of treatment-emergent diabetes mellitus in patients receiving antipsychotics. Ann Pharmacother. 2007;41(10):1593–603.
- 54. Tsai KY, Lee CC, Chou YM, Su CY, Chou FH. The incidence and relative risk of stroke in patients with schizophrenia: a five-year follow-up study. Schizophr Res. 2012;138(1):41–7.
- 55. Lin HC, Hsiao FH, Pfeiffer S, Hwang YT, Lee HC. An increased risk of stroke among young schizophrenia patients. Schizophr Res. 2008;101:234–41.
- 56. Kang JH, Xirasagar S, Lin HC. Lower mortality among stroke patients with schizophrenia: a nationwide population-based study. Psychosom Med. 2011;73(1):106–11.
- American Diabetes Association, American Psychiatric Association, American Association of Clinical Endocrinologists, North American Association for the Study of Obesity. Consensus development conference on antipsychotic drugs and obesity and diabetes. Diabetes Care. 2004;27(2):596–601.
- Morrato EH, Newcomer JW, Kamat S, Baser O, Harnett J, Cuffel B. Metabolic screening after the American Diabetes Association's consensus statement on antipsychotic drugs and diabetes. Diabetes Care. 2009;32(6):1037–42.
- 59. De Hert M, Yu W, Detraux J, et al. Body weight and metabolic adverse effects of asenapine, iloperidone, lurasidone and paliperidone in the treatment of schizophrenia and bipolar disorder: a systematic review and exploratory meta-analysis. CNS Drugs. 2012;26(9):733–59.
- Weiden PJ, Daniel DG, Simpson G, Romano SJ. Improvement in indices of health status in outpatients with schizophrenia switched to ziprasidone. J Clin Psychopharmacol. 2003;23(6): 595–600.
- Montes JM, Rodriguez JL, Balbo E, et al. Improvement in antipsychotic-related metabolic disturbances in patients with schizophrenia switched to ziprasidone. Prog Neuropsychopharmacol Biol Psychiatry. 2007;31:383–8.
- Casey DE, Carson WH, Saha AR, et al. Switching patients to aripiprazole from other antipsychotic agents: a multicenter randomized study. Psychopharmacology (Berl). 2003;166: 391–9.
- 63. Newcomer JW, Campos JA, Marcus RN, et al. A multicenter, randomized, double-blind study of the effects of aripiprazole in overweight subjects with schizophrenia or schizoaffective disorder switched from olanzapine. J Clin Psychiatry. 2008;69:1046–56.
- Wu R, Zhao J, Jin H, et al. Lifestyle intervention and metformin for treatment of antipsychoticinduced weight gain: a randomized controlled trial. JAMA. 2008;299(1):185–93.
- 65. Wu RR, Jin H, Gao K, et al. Metformin for treatment of antipsychotic-induced amenorrhea and weight gain in women with first-episode schizophrenia: a double-blind, randomized, placebo-controlled study. Am J Psychiatry. 2012;169:813–21.
- 66. Wang M, Tong JH, Zhu G, Liang GM, Yan HF, Wang XZ. Metformin for treatment of antipsychotic-induced weight gain: a randomized, placebo-controlled study. Schizophr Res. 2012;138(1):54–7.
- Bjorkhem-Bergman L, Asplund AB, Lindh JD. Metformin for weight reduction in non-diabetic patients on antipsychotic drugs: a systematic review and meta-analysis. J Psychopharmacol. 2011;25(3):299–305.

- Wu RR, Zhao JP, Guo XF, et al. Metformin addition attenuates olanzapine-induced weight gain in drug-naïve first-episode schizophrenia patients: a double-blind, placebo-controlled study. Am J Psychiatry. 2008;165:352–8.
- 69. Narula PK, Rehan HS, Unni KE, Gupta N. Topiramate for prevention of olanzapine associated weight gain and metabolic dysfunction in schizophrenia: a double-blind, placebocontrolled trial. Schizophr Res. 2010;118(1–3):218–23.
- Nickel MK, Nickel C, Muehlbacher M, et al. Influence of topiramate on olanzapine-related adiposity in women: a random, double-blind, placebo-controlled study. J Clin Psychopharmacol. 2005;25(3):211–7.
- Egger CM, Muehlbacher M, Schatz M, Nickel M. Influence of topiramate on olanzapinerelated weight gain in women: an 18-month follow-up observation. J Clin Psychopharmacol. 2007;27(5):475–8.
- Ko YH, Joe SH, Jung IK, Kim SH. Topiramate as an adjuvant treatment with atypical antipsychotics in schizophrenic patients experiencing weight gain. Clin Neuropharmacol. 2005;28(4):169–75.
- 73. Alvarez-Jimenez M, Hetrick SE, Gonzalez-Blanch C, Gleeson JF, McGorry PD, et al. Nonpharmacological management of antipsychotic-induced weight gain: systematic review and meta-analysis of randomised controlled trials. Br J Psychiatry. 2008;193:101–7.
- 74. Caemmerer J, Correll CU, Maayan L. Acute and maintenance effects of nonpsychopharmacologic interventions for antipsychotic associated weight gain and metabolic abnormalities: a meta-analytic comparison of randomized controlled trials. Schizophr Res. 2012;140(1–3):159–68.
- 75. Rosenberg SD, Goodman LA, Osher FC, et al. Prevalence of HIV, hepatitis B, and hepatitis C in people with severe mental illness. Am J Public Health. 2001;91:31–7.
- 76. Rosenberg SD, Drake RE, Brunette RE, et al. Hepatitis C virus and HIV co-infection in people with severe mental illness and substance use disorders. AIDS. 2005;19:S26–33.
- 77. De Hert M, Wampers M, van Eyck D, et al. Prevalence of HIV and hepatitis C infection among patients with schizophrenia. Schizophr Res. 2009;108(1–3):307–8.
- Blank MB, Mandell DS, Aiken L, Hadley TR. Co-occurrence of HIV and serious mental illness among Medicaid recipients. Psychiatr Serv. 2002;53(7):868–73.
- Nurutdinova D, Chrusciel T, Zeringue A, et al. Mental health disorders and the risk of AIDSdefining illness and death in HIV-infected veterans. AIDS. 2012;26(2):229–34.
- 80. Dinwiddie SH, Shicker L, Newman T. Prevalence of hepatitis C among psychiatric patients in the public sector. Am J Psychiatry. 2003;160(1):172–4.
- Sockalingam S, Shammi C, Powell V, Barker L, Remington G. Determining rates of hepatitis C in a clozapine treated cohort. Schizophr Res. 2010;124:86–90.
- Osher FG, Goldberg RW, McNary SW, et al. Substance abuse and the transmission of hepatitis C among persons with severe mental illness. Psychiatr Serv. 2003;54(6):842–7.
- 83. Nakamura Y, Koh M, Miyoski E, et al. High prevalence of the hepatitis C virus infection among the inpatients of schizophrenia and psychoactive substance abuse in Japan. Prog Neuropsychopharmacol Biol Psychiatry. 2004;24:591–7.
- Huckans MS, Blackwell AD, Harms TA, Hauser P. Management of hepatitis C disease among VA patients with schizophrenia and substance use disorders. Psychiatr Serv. 2006; 57(3):403–6.
- 85. Butterfield MI, Bosworth HB, Meador KG, et al. Gender differences in hepatitis C infection and risks among persons with severe mental illness. Psychiatr Serv. 2003;54(3):848–53.
- Freudenreich O, Gandhi RT, Walsh JP, Henderson DC, Goff DC. Hepatitis C in schizophrenia: screening experience in a community-dwelling clozapine cohort. Psychosomatics. 2007; 48(5):405–11.
- 87. Lok A, McMahon B. Chronic hepatitis B: update 2009. Hepatology. 2009;50(3):661-2.
- 88. http://www.cdc.gov/mmwr/preview/mmwrhtml/rr5514a1.htm. Accessed 15 June 2013.
- Brunette MF, Drake RE, Marsh BJ, Torrey WC, Rosenberg SD; Five-Site Health and Risk Study Research Committee. Responding to blood-borne infections among persons with severe mental illness. Psychiatr Serv. 2003;54(6):860–5.

- Rosenberg S, Brunette M, Oxman T, et al. The STIRR model of best practices for bloodborne diseases among clients with serious mental illness. Psychiatr Serv. 2010;55(6):660–4.
- Johnson-Masotti AP, Pinkerton SD, Kelly JA, Stevenson LY. Cost-effectiveness of an HIV risk reduction intervention for adults with severe mental illness. AIDS Care. 2000;12(3):321–32.
- Huckans M, Mitchell A, Ruimy S, Loftis J, Hauser P. Antiviral therapy completion and response rates among hepatitis C patients with and without schizophrenia. Schizophr Bull. 2010;36(1):165–72.
- Huckans M, Mitchell A, Pavawalla S, et al. The influence of antiviral therapy on psychiatric symptoms among patients with hepatitis C and schizophrenia. Antivir Ther. 2010;15(1):111–9.
- 94. Chou FH, Tsai KY, Su CY, Lee CC. The incidence and relative risk factors for developing cancer among patients with schizophrenia: a nine-year follow-up study. Schizophr Res. 2011;129(2–3):97–103.
- 95. Levav I, Lipshitz I, Novikov I, et al. Cancer risk among parents and siblings of patients with schizophrenia. Br J Psychiatry. 2007;190:156–61.
- 96. Goldacre MJ, Kurina LM, Wotton CJ, Yeates D, Seagroat V. Schizophrenia and cancer: an epidemiological study. Br J Psychiatry. 2005;187:334–8.
- Hippisley-Cox J, Vinogradova Y, Coupland C, Parker C. Risk of malignancy in patients with schizophrenia or bipolar disorder: nested case-control study. Arch Gen Psychiatry. 2007;64(12):1368–76.
- 98. McGinty EE, Zhang Y, Guallar E, et al. Cancer incidence in a sample of Maryland residents with serious mental illness. Psychiatr Serv. 2012;63(7):714–7.
- 99. Catts SV, O'Toole BI, Frost AD. Cancer incidence in patients with schizophrenia and their first-degree relatives—a meta-analysis. Acta Psychiatr Scand. 2008;117(5):323–36.
- 100. Lin CY, Lane HY, Chen TT, Wu YH, Cy W, Wu VY. Inverse association between cancer risks and age in schizophrenic patients; a 12-year nationwide cohort study. Inverse association between cancer risks and age in schizophrenic patients: a 12-year nationwide cohort study. Cancer Sci. 2013;104(3):383–90.
- 101. Capasso RM, Lineberry TW, Bostwick JM, St Sauver J. Mortality in schizophrenia and schizoaffective disorder: county, Minnesota cohort: 1950-2005. Schizophr Res. 2008; 98(1–3):287–94.
- Tran E, Rouillon F, Loze JY, et al. Cancer mortality in patients with schizophrenia: an 11-year prospective cohort study. Cancer. 2009;115:3555–62.
- 103. Levav I, Kohn R, Barchana M, et al. The risk for cancer among patients with schizoaffective disorders. J Affect Disord. 2009;114(1–3):316–20.
- 104. Bushe CJ, Bradley AJ, Wildgust HJ, Hodgson RE. Schizophrenia and breast cancer incidence: a systematic review of clinical studies. Schizophr Res. 2009;114(1–3):6–16.
- Seeman MV. Preventing breast cancer in women with schizophrenia. Acta Psychiatr Scand. 2011;123(2):107–17.
- Cole M, Padmanabhan A. Breast cancer treatment of women with schizophrenia and bipolar disorder from Philadelphia, PA: lessons learned and suggestions for improvement. J Cancer Educ. 2012;27(4):774–9.
- 107. Sharma A, Ngan S, Nandoskar A, et al. Schizophrenia does not adversely affect the treatment of women with breast cancer: a cohort study. Breast. 2010;19(5):410–20.
- 108. Lichtermann D, Ekelund J, Pukkala E, Tanskanen A, Lonnqvist J. Incidence of cancer among persons with schizophrenia and their relatives. Arch Gen Psychiatry. 2001;58:573–8.
- 109. Dalton SO, Schuz J, Engholm G, et al. Social inequality in incidence of and survival from cancer in a population-based study in Denmark, 1994-2003: summary of findings. Eur J Cancer. 2008;44(14):2074–85.
- 110. Mateen FJ, Jatoi A, Lineberry TW, et al. Do patients with schizophrenia receive state-of-theart lung cancer therapy? A brief report. Psychooncology. 2008;17(7):721–5.
- 111. Howard LM, Barley EA, Davis E, et al. Cancer diagnosis in people with severe mental illness: practical and ethical issues. Lancet Oncol. 2010;11:797–804.

- 112. Regier DA, Farmer ME, Rae DS, et al. Comorbidity of mental disorders with alcohol and other drug abuse. Results from the Epidemiologic Catchment Area (ECA) Study. JAMA. 1990;264(19):2511–8.
- 113. De Leon J, Diaz FJ. A meta-analysis of world-wide studies demonstrating an association between schizophrenia and tobacco smoking behaviors. Schizophr Res. 2005;76:135–57.
- Gregg L, Barrowclough C, Haddock G. Reasons for increased substance abuse in psychosis. Clin Psychol Rev. 2007;27:494–510.
- 115. Salyers MP, Mueser KT. Social functioning, psychopathology, and medication side effects in relation to substance use and abuse in schizophrenia. Schizophr Res. 2001;48:109–23.
- 116. McEvoy JP, Freudenrich O, Wilson WH. Smoking and therapeutic response to clozapine in patients with schizophrenia. Biol Psychiatry. 1999;46:125–9.
- 117. Zimmet SV, Strous RD, Burgess ES, et al. Effects of clozapine on substance use in patients with schizophrenia and schizoaffective disorder: a retrospective survey. J Clin Psychiatry. 2000;20:94–8.
- 118. Evins AD, Cather C, Culhane MA, et al. A 12-week double-blind placebo-controlled study of bupropion SR added to high dose dual nicotine replacement therapy for smoking cessation or reduction in schizophrenia. J Clin Psychopharmacol. 2007;27(4):380–6.
- 119. Batki SL, Dimmock AJ, Wade M, et al. Monitored naltrexone without counseling for alcohol abuse/dependence in schizophrenia-spectrum disorders. Am J Addict. 2007;16:253–9.
- Hsu JH, Chien IC, Lin CH, et al. Increased risk of chronic obstructive pulmonary disease in patients with schizophrenia: a population-based study. Psychosomatics. 2013;54(4):345–51.
- 121. Copeland LA, Mortensen EM, Zeber JE, Pugh MJ, Restrepo MI, Dalack GW. Pulmonary disease among inpatient decedents: impact of schizophrenia. Prog Neuropsychopharmacol Biol Psychiatry. 2007;31:720–6.
- 122. Seminog OO, Goldacre MJ. Risk of pneumonia and pneumococcal disease in people with severe mental illness: English record linkage studies. Thorax. 2013;68(2):171–6.
- 123. Pedersen MS, Benros ME, Agerbo E, Borglum AD, Mortensen PB. Schizophrenia in patients with atopic disorders with particular emphasis on asthma: a Danish population-based study. Schizophr Res. 2012;138(1):58–62.
- 124. Chen YH, Lee HC, Lin HC. Prevalence and risk of atopic disorders among schizophrenia patients: a nationwide population based study. Schizophr Res. 2009;108(1–3):191–6.
- 125. Eaton WW, Byrne M, Ewald H, et al. Association of schizophrenia and autoimmune diseases: linkage of Danish national registers. Am J Psychiatry. 2006;163(3):521–8.
- 126. Kelly DL, Conley BB. Thyroid function in treatment-resistant schizophrenia patients treated with quetiapine, risperidone or fluphenazine. J Clin Psychiatry. 2005;66(1):80–4.
- 127. Eaton WW, Hayward C, Ram R. Schizophrenia and rheumatoid arthritis: a review. Schizophr Res. 1992;6(3):181–92.
- 128. Allebeck P, Rodvall Y, Wistedt B. Incidence of rheumatoid arthritis among patients with schizophrenia, affective psychosis or neurosis. Acta Psychiatr Scand. 1985;71(6):615–9.
- 129. Mors O, Mortensen PB, Ewald H. A population-based register study of the association between schizophrenia and rheumatoid arthritis. Schizophr Res. 1999;40(1):67–74.
- 130. Oken RJ, Schulzer M. At issue: Schizophrenia and rheumatoid arthritis: the negative association revisited. Schizophr Bull. 1999;25(4):625–38.
- 131. Chang YT, Chen PC, Tsai IJ, et al. Bidirectional relation between schizophrenia and epilepsy: a population-based retrospective cohort study. Epilepsia. 2011;52(11):2036–42.
- Marsh L, Rao V. Psychiatric complications in patients with epilepsy: a review. Epilepsy Res. 2002;49(1):11–33.
- 133. Adachi N, Onuma T, Nishiwaki S, et al. Inter-ictal and post-ictal psychoses in frontal lobe epilepsy: a retrospective comparison with psychoses in temporal lobe epilepsy. Seizure. 2000;9(5):328–35.
- Wotton CJ, Goldacre MJ. Coexistence of schizophrenia and epilepsy: record-linkage studies. Epilepsia. 2012;53:e71–4.

- 135. Clarke MO, Tanskanen A, Huttunen MO, Clancy M, Cotter DR, Cannon M. Evidence for shared susceptibility to epilepsy and psychosis: a population-based family study. Biol Psychiatry. 2012;71(9):836–9.
- 136. Marsh L, Sullivan EV, Morrell M, Lim KO, Pfefferbaum A. Structural brain abnormalities in patients with schizophrenia, epilepsy, and epilepsy with chronic interictal psychosis. Psychiatry Res. 2001;108(1):1–15.
- Cascella NG, Schretlen DJ, Sawa A. Schizophrenia and epilepsy: is there a shared susceptibility? Neurosci Res. 2009;63(4):227–35.
- 138. Beary M, Hodgson R, Wildgust HJ. A critical review of major mortality risk factors for all-cause mortality in first-episode schizophrenia: clinical and research implications. J Psychopharmacol. 2012;26(5 Suppl):52–61.
- 139. Butler M, Lane RL, McApine D, et al. Integration of mental health/substance abuse and primary care. Evidence reports/technology assessments, No. 173. 2008. Agency for Healthcare Research and Quality U.S. Department of Health and Human Services. http://www.ahrq.gov/research/ findings/evidence-based-reports/mhsapc-evidence-report.pdf. Accessed 31 July 2013.
- 140. Katon WJ, Lin EHB, Korff MV, et al. Collaborative care for patients with depression and chronic illness. N Engl J Med. 2010;363(27):2611–20.
- 141. Druss BG, Rohrbaugh RM, Levinson CM, et al. Integrated medical care for patients with serious psychiatric illness: a randomized trial. Arch Gen Psychiatry. 2001;58(9):861–8.
- 142. Rubin AS, Littenberg B, Ross R, Wehry S, Jones M. Effects on processes and costs of care associated with the addition of an internist to an inpatient psychiatry team. Psychiatr Serv. 2005;56(4):463–7.
- 143. Druss BG, von Esenwein SA, Compton MT, Rask KJ, Zhao L, Parker RM. A randomized trial of medical care management for community mental health settings: the primary care access, referral, and evaluation (PCAARE) study. Am J Psychiatry. 2010;167:151–9.
- 144. Druss BG, von Esenwein SA. Improving general and medical care for persons with mental and addictive disorders; systematic review. Gen Hosp Psychiatry. 2006;28:145–53.
- 145. Smith TE, Sederer LI. A new kind of homelessness for individuals with serious mental illness? The need for a "mental health home". Psychiatr Serv. 2009;60:528–33.
- 146. Scharf DM, Eberhart NK, Schmidt N, et al. Integrating primary care into community behavioral health settings: programs and early implementation experiences. Psychiatr Serv. 2013;64(7):660–5.
- 147. Stone NJ, Robinson J, Lichtenstein AH, et al. ACC/AHA guideline on the treatment of blood cholesterol to reduce atherosclerotic cardiovascular risk in adults: a report of the American College of Cardiology/American Heart Association task force on practice guidelines. Circulation. 2013. Epub ahead of print. https://circ.ahajournals.org/content/ early/2013/11/11/01.cir.0000437738.63853.7a.full.pdf. Accessed 17 March 2014.
- 148. United States Preventive Service Task Force. www.uspreventiveservicestaskforce.org. Accessed 21 July 2013.

Chapter 11 Management of Water Imbalance in Schizophrenia

Morris Goldman and Pichai Ittasakul

Introduction

Historical Overview

Since the 1930s, unexplained increases in water intake, impairments in water excretion which vary with severity of psychosis, and reports of water intoxication were linked to chronic psychotic disease [1–3]. In the following decades, hyponatremia was often unnoticed and patients were frequently thought to have a primary seizure disorder [4]. Beginning in the 1950s, the characterization of the syndrome of inappropriate antidiuretic hormone secretion (SIADH), the increased interest in distinguishing diabetes insipidus from primary polydipsia, and the development of a sensitive radioimmunoassay for the antidiuretic hormone, arginine vasopressin (AVP), all contributed to a renewed interest in water balance in psychiatric patients [5, 6].

Initial efforts to characterize the mechanism of chronic moderate hyponatremia and episodic water intoxication in schizophrenia were obscured by the different etiologies of the hyponatremia, as well as the complexities of antidiuretic function [7]. In the absence of a deficiency of salt or diminished blood volume, hyponatremia occurs when fluid intake overtakes renal water excretion thereby diluting the amount of sodium in the body. Early studies did not distinguish patients with idiopathic hyponatremia from those with iatrogenic hyponatremia (typically attributable to

P. Ittasakul

M. Goldman (🖂)

Department of Psychiatry and Behavioral Sciences, Northwestern University, East Ontario 446, Chicago, IL 60611, USA e-mail: m-goldman@northwestern.edu

Department of Psychiatry, Ramathibodi Hospital, Mahidol University, 270 Rama VI Road, Ratchathewi, Bangkok 10400, Thailand e-mail: pichail18@gmail.com

thiazide diuretics, carbamazepine, or antidepressants) or other recognized causes such as hypothyroidism or severe alcoholism [8, 9].

Iatrogenic hyponatremia and hyponatremia due to other recognized causes typically involved minimal or no excess fluid intake but fixed and relatively severe deficits in water excretion. In contrast, idiopathic cases typically involved marked polydipsia and transient, relatively mild, impairments in water excretion that were difficult to detect. While some reports indicated that patients with unexplained hyponatremia exhibited impaired water excretion or elevated AVP levels, others found normal water excretion [5–7]. Some asserted the hyponatremia in these patients was solely due to polydipsia, and labeled these patients as having psychogenic polydipsia and "self-induced water intoxication." Even when impaired excretion was apparent, this was often considered a consequence of hyponatremia (e.g., increased antidiuretic hormone secretion from seizures, vomiting), rather than a contributor [9].

Hariprasad et al. in 1980 were the first to reconcile the conflicting evidence regarding the role of impaired water excretion in patients with idiopathic hyponatremia [10]. Renal water excretion is primarily determined by the actions of AVP on the collecting ducts of the nephron. Plasma AVP normally increases linearly as plasma osmolality increases, and diminishes to undetectable levels as plasma osmolality drops (Fig. 11.1; normal). As AVP rises, urine concentration rises to five times that of plasma (~1,500 mOsm/kg), while undetectable AVP levels are associated with urine concentrations that are one-tenth of plasma (~30 mOsm/kg). At moderate or high AVP levels any amount of ongoing fluid intake will overwhelm renal water excretion, while with undetectable AVP levels renal water excretion can easily exceed ~20 L/day.

The plasma osmolality at which AVP becomes undetectable (and hence water excretion reaches maximal levels) is called the set point or osmotic threshold and is generally about 3–5 % below the actual plasma osmolality (normal 280–300 mOsm/kg) or plasma sodium (135–145 mEq/L). Hariprasad measured plasma osmolality and concurrent renal water excretion during enforced fluid restriction in a group with schizophrenia and idiopathic hyponatremia. At extremely low plasma osmolalities, these patients' water excretion appeared to be normal (i.e., \geq 20 L/day), but as plasma osmolality began to rise their water excretion was diminished from normal. Hence, depending on the concurrent plasma osmolality, water excretion could appear impaired or normal. The findings implicated a condition called reset osmostat in which the relationship between AVP and plasma osmolality/sodium is maintained but shifted leftward (Fig. 11.1; type C). Several groups reproduced these findings and confirmed they were attributable to a shift in the set point for AVP, while other modulators of renal water excretion remained intact [10–13].

In an effort to distinguish psychotic patients with idiopathic hyponatremia attributable to reset osmostat from those whose hyponatremia was due to recognized factors (Fig. 11.1; type A or type B), Vieweg coined the term "psychosis, intermittent hyponatremia, polydipsia" (PIP) syndrome [14]. This initial clarification of the different mechanisms of hyponatremia provided the foundation for diagnosis and management of the disorders discussed below, as well as for subsequent research into the relationship of reset osmostat to the underlying mental illness. In contrast, little is still known about the mechanism of primary polydipsia.



Fig. 11.1 Normal and abnormal regulation of arginine vasopressin (AVP). AVP is secreted from the brain into the peripheral circulation and binds receptors in the kidney which diminish renal water excretion. AVP is primarily determined by concurrent levels of plasma osmolality (and sodium). Increases in osmolality cause a linear rise in AVP concentration in plasma, while decreases diminish AVP to undetectable levels. The linear relationship is normally very precise $(r \sim 0.9)$ and is characterized by its gain (slope of line) and set point (osmotic threshold). The white triangles are an example of a normal relationship. In most cases of abnormal osmoregulation attributable to medication or other unrecognized medical disorders, plasma AVP is unresponsive to plasma osmolality (i.e., syndrome of inappropriate antidiuretic hormone (SIADH; Type A) or only partially inhibited (Type B), producing relatively fixed and severe deficits in water excretion. In contrast, water excretion in patients with idiopathic hyponatremia/ PIP syndrome depends on the level of plasma osmolality (Type C=reset osmostat). At very low levels (i.e., near the set point) it will appear normal, while at slightly higher levels it will appear impaired. Moreover, the resetting itself varies with the acuity of the psychosis. Gray shading represents the normal range of levels, hence concurrent measures of AVP and plasma osmolality can help establish impaired water excretion which otherwise eludes detection

Epidemiology, Risk Factors, and Sequalae

Primary and Secondary Polydipsia

In individuals with chronic mental illness, 15–25 % have a "*primary*" polydipsia (i.e., oral intake >4 L/day, normal <3.0 L/day), exhibiting increased intake which is not attributable to impaired concentrating capacity (e.g., not nephrogenic diabetes insipidus, diabetes mellitus, or renal failure) [15, 16]. Genetic factors may contribute to this primary polydipsia [17, 18]. Patients also exhibit an increased incidence of alcoholism and smoking which appear to precede the mental illness and may further characterize the phenotype [16, 19]. *Secondary polydipsia* is commonly seen with nephrogenic diabetes insipidus attributable to lithium treatment (in about 30 % of those receiving lithium), and may occur more commonly in patients receiving serotonergic antidepressants [20]. The polydipsia is secondary to the increased

Primary polydipsia
Increasing age
Heavy smoking
Alcoholism
Polypharmacy
Chronic psychosis
Medications (e.g., use of diuretic, SSRI, TCA, venlafaxine, bupropion, carbamazepine, and calcium antagonist)
Medical conditions that decrease water excretion (e.g., diabetes mellitus, syndrome of inappropriate antidiuresis from lung cancer kidney disease, heart failure, cirrhosis, hypothyroidism, adrenal insufficiency)
Source: Modified from Ittasakul P, Goldman MB. Hyponatremia in Psychosis. In: Simon EE, edi
tor Hypenetromics Evoluction and Treatment New York, Springer 2012 with normission from

Table 11.1 Risk factors for hyponatremia

Source: Modified from Ittasakul P, Goldman MB. Hyponatremia in Psychosis. In: Simon EE, editor. Hyponatremia: Evaluation and Treatment. New York: Springer; 2012, with permission from Springer Science+Business Media

SSRI selective serotonin reuptake inhibitor, TCA tricyclic antidepressants

water excretion which causes thirst. In the absence of lithium toxicity, irreversible nephrogenic diabetes insipidus and impaired glomerular filtration (e.g., elevated plasma creatinine) normally occur only after years of treatment. Recent findings indicate the lithium-induced tuberointerstitial nephritis which accompanies chronic polyuria may be directly responsible for this progressive renal failure [21]. Thus, polyuria is a risk factor for renal failure with lithium therapy, and physicians should consider treating it or switching medications (see Treatment). Secondary polydipsia does not predispose to hyponatremia, since the polydipsia is due to increased water excretion. Primary polydipsia in a patient with lithium-induced renal failure, however, would place the patient at markedly increased risk of hyponatremia.

Regardless of the etiology, over time excessive amounts of water flowing through the body may lead to systemic "plumbing" problems (i.e., dilated bowel, bladder, and kidneys as well as associated disorders such as obstruction, incontinence, urinary infection, and renal failure) [16]. Further, the increased intake and output seems to leech calcium from bones predisposing to osteoporosis and pathologic fractures [22].

Chronic Moderate Hyponatremia and Episodic Water Intoxication

Psychiatric disorders are associated with a higher risk of chronic hyponatremia and episodic water intoxication. The increased risk is primarily attributable to the prescribing of medications which impair water excretion, the elevated frequency of primary polydipsia, and the PIP syndrome (Table 11.1) [23–30]. Carbamazepine, thiazide diuretics, and antidepressants most commonly impair water excretion and are responsible iatrogenic hyponatremia [27]. Many of these patients may also have

primary polydipsia. Serotonin reuptake inhibitors (SSRIs) have a risk that is four times higher than other antidepressants [23, 31]. Elderly women during the first few weeks of therapy appear to be at highest risk [31]. It is likely that any antidepressant which enhances serotonin activity (e.g., trazodone, clomipramine) may increase the risk of hyponatremia, though reports have implicated all agents, including mirtazapine and buproprion [23–27, 29]. Other serotinergic agents, including street drugs such as methylenedioxymethamphetamine (MDMA: ecstasy), are also associated with hyponatremia, perhaps by directly enhancing vasopressin secretion [32]. While many psychotropic medications induce transient nausea and orthostatic hypotension (i.e., recognized AVP stimuli), these are not routinely implicated as causes for hyponatremia in this population.

The incidence of iatrogenic hyponatremia in polydipsic psychotic patients is not clearly established but is likely about the same or higher based on population surveys [33, 35]. About one in five patients with schizophrenia and primary polydipsia (i.e., about 3 % of all chronic patients) experiences intermittent hyponatremia due to the PIP syndrome [15, 33, 34]. In rare cases, hyponatremia may occur from primary polydipsia alone (i.e., "self-induced"), though this has never been conclusively demonstrated.

Other reported risk factors associated with hyponatremia (Table 11.1) include smoking which is a recognized stimulus for AVP release and contributed to impaired water excretion in case reports [36]. Smoking is extremely common in these patients (~70 %), especially in the hyponatremic subset [34]. Of note, this also places patients at increased risk for SIADH which is associated with small cell lung cancer. Alcoholism is also common in psychiatric patients and the associated malnutrition or cirrhosis in severe cases can predispose polydipsic patients to hyponatremia. Further, alcoholism is independently associated with polydipsia in schizophrenia [14, 37]. Finally, polypharmacy appears to be a risk, and may simply be due to the additive effects of two or more medications associated with impaired water excretion [38].

Clinical Presentation

Polydipsia is frequently undetected, though many patients are observed to have a cup at all times or may spend excessive time in proximity to water sources (potable or nonpotable). While moderate chronic hyponatremia may appear asymptomatic, most patients have impaired cognition characterized by deficits in attention, learning, memory, and executive function which may be difficult to distinguish from their underlying mental illness [30, 39, 40]. These patients also exhibit impaired fine motor skills and are at greater risk of falls and fractures [41–44].

Signs and symptoms of water intoxication are dependent on the severity and rapidity of its development [45]. Generally, the clinical presentation of symptomatic hyponatremia not only resembles that of nonpsychiatric patients (vomiting, nausea, ataxia, confusion, lethargy, seizures, and coma) but also includes aggravation of the underlying psychiatric illness (e.g., increased paranoia and aggression). Conversely, patients may appear stable one moment and begin seizing the next [30, 46]. Neurological symptoms usually do not occur until the serum sodium concentration falls below 120 mEq/L or even lower with chronic hyponatremia [47]. Water intoxication is primarily attributable to acute cerebral edema (i.e., a water-saturated brain encountering a rigid cranium). Patients with chronic hyponatremia adapt to the excess brain water by excreting organic osmolytes (i.e., idiogenic osmoles) but will experience water intoxication if their sodium levels drop further (e.g., ~110 mEq/L) [48]. Hyponatremia may present with rhabdomyolysis (with or without compartment syndrome) and with neuroleptic malignant syndrome [49–52]. Rhabdomyolysis may also occur in patients whose hyponatremia was rapidly corrected [53].

Over the past 30 years there are many reports of death in severely mentally ill patients due to water intoxication [54, 55]. Once largely confined to medicated patients on extended stay units in public psychiatric hospitals, episodic water intoxication is now increasingly seen in unmedicated patients living in the community [56].

The PIP Syndrome

Primary polydipsia typically appears about 5 years after the onset of psychiatric illness, and hyponatremia appears about 5 years after that in patients with the PIP syndrome [14, 57]. Rarely, however, unexplained polydipsia and water intoxication can occur concurrently with the first psychotic break [58]. PIP patients usually have a severe unrelenting psychosis with marked social deficits [59]. Patients often hide their drinking, and symptomatic hyponatremia may be suspected only after someone notes a striking diuresis in a post-ictal patient. Because the impairment in water excretion is relatively minor and may even be transient, water accumulates over the course of the day and is excreted at night. Thus, hyponatremia is most marked in the mid-afternoon and is frequently not present when blood samples are obtained in the morning. This pattern of drinking and of water retention led to some unique clinical observations (i.e., institutionalized PIP patients in the winter tend to congregate at midday around heaters due to massive water drinking lowering their body temperature (so-called afternoon radiator sitting syndrome)) [60].

PIP patients and their first degree relatives differ from others with schizophrenia [18, 19, 37, 61] in that they are more severely debilitated by their illness and exhibit an increase in primary sensory deficits [62]. Hawken et al. recently examined the long-term effects of polydipsia and hyponatremia on mortality [63]. The median age at death was 57 years for hyponatremic polydipsia, 60 years for normonatremic polydipsia, and 68 years for matched non-polydipsia patients. Hyponatremic patients had a 74 % greater chance of dying before non-polydipsic patients. How much of the enhanced mortality is attributable to the water imbalance or other aspects of their schizophrenia (e.g., increased smoking) is not known.

Diagnosis

Polydipsia

Polydipsia of any origin typically presents with nocturia and incontinence. Since urine output normally matches up well with oral intake and the rate of solute excretion in the urine is fairly constant, urine tonicity provides a reasonable index of oral intake [64]. Despite the absence of fluid intake over the course of the night, morning urine tonicity may be diminished due to a concentrating defect (to be distinguished from the impaired excretion) attributable to the polydipsia (i.e., secondary to "renal medullary washout") [65]. Thus, afternoon measures are clearly more sensitive and hence preferred. Polydipsia is diagnosed by demonstrating diminished urine specific gravity (<1.008; normal 1.015–1.030) or diminished urine osmolality (<150 mOsm/kg; normal 500–1,400 mOsm/kg) on two of three spot urine samples taken over a week or more (Table 11.2).

PIP Syndrome

Because hyponatremia in PIP patients is intermittent and most apparent in the afternoon, morning serum sodium levels do not make the diagnosis [14]. The amount of retained water required to induce dilutional hyponatremia is significant. Because accumulation begins again each morning, patients at risk of water intoxication can be easily identified by obtaining diurnal measures of body weight. Acute water intoxication is commonly associated with a marked increase in body weight (i.e., 20 % decrease in plasma sodium typically accompanied by a 8 kg increase in body weight in the typical 70 kg person) [14], but on a typical day, patients gain 2–3 kg. Thus, determining the difference between morning and afternoon body weights is a reliable means of diagnosing the PIP syndrome, since diurnal weight rarely varies more than 1.5 kg in normals (Table 11.2). Care must be taken to use the same scale and make certain the patient is wearing approximately the same amount of clothing. Diagnosis of hyponatremia is confirmed by obtaining a concurrent sodium concentration or plasma osmolality with the afternoon weight (Table 11.2).

Concurrent plasma and urine concentration measures may or may not implicate reset osmostat in the hyponatremia. Given the transient nature of reset osmostat in this population, it is difficult to establish the diagnosis unless serial measures of plasma sodium and urine osmolality samples are obtained following water loading or fluid restriction when the patient is hyponatremic.

Medication-Induced Hyponatremia

Medication-induced hyponatremia is typically more stable than that seen in PIP patients, because of the more limited role of polydipsia and the more severe and
Disorder	Causes	Diagnosis	Criteria	Clues	Management
Primary polydipsia	Hippocampal dysfunction	2–3 urine osmolalities or specific gravities	Two or more <1.008 or Uosm <150 (i.e., same criteria)	Worse in afternoon than AM	Usually none
Secondary polydipsia	Lithium	in AM and PM		Minimal variation within or between days	Lower dose, amiloride, thiazide
Medication-induced hyponatremia	Thiazide, carbamazepine,	AM and PM weights for 1 week, two AM	Na+ <130 mEq/L; Urine osmolality >220 mOsm/kg	Minimal diurnal variation in weight	Discontinue medication
PIP syndrome	antipsychofic Hippocampal dysfunction	serum sodium with urine osmolalities	Urine osmolality <220 mOsm/kg; PM weight 3 kg >AM weight	AM sodium levels may be normal	Target weight, clozapine

•	ſeni
•	ndo
:	schize
	Ξ
	oalance
	Ē
	water
¢	ot
	management
	and
•	agnosis
;	9
	Iterential
(ี่วี
	ble 11.2
	G

fixed impairment in water excretion. For these reasons, a single concurrent measure of urine and plasma osmolality at any time of the day is adequate to make the diagnosis. The role of one of the above noted agents is strongly implicated if treatment was initiated in the past 30 days and concurrent urine osmolality is greater than plasma osmolality (i.e., >300 mOsm/kg) (Table 11.2). Such levels are rarely if ever seen in PIP patients. Depending on the half-life of the medication (e.g., the half-life of fluoxetine is 2–7 days), these cases should rapidly resolve with discontinuation. In the absence of a rapid reversal, other factors noted in Table 11.1 should be considered.

Antipsychotic medication presents a particular challenge, since discontinuing them in PIP patients frequently worsens both the psychiatric disorder and the water imbalance [52, 66]. While antipsychotics may modestly enhance renal sensitivity to AVP in a dose-related manner, this effect, per se, does not appear to contribute to the hyponatremia [65, 67]. The mechanism of antipsychotic-induced hyponatremia is unknown but like other drug-induced cases, antipsychotic-induced hyponatremia is more stable than that seen in PIP patients and is attributable to severe impairments in water excretion [52]. A recent review of published cases found that persons with antipsychotic-induced hyponatremia tended to have concurrent urine osmolality levels >220 mOsm/kg. By contrast, levels are lower than this with the PIP syndrome. Persons with concentrated urine meet Naranjo Criteria (score based on a series of questions about the relationship of an adverse outcome to medication usage) for drug-induced disorder seven times as often as those with dilute urine. Further, duration of antipsychotic treatment was one-eighth as long, and recurrence of hyponatremia following antipsychotic rechallenge occurred exclusively in those with concentrated urine [66, 68–70] (Atsariyasing and Goldman, unpublished).

Pathophysiology of PIP Syndrome

Mechanism of Primary Polydipsia

The mechanism of primary polydipsia is unknown. Polydipsia does not seem attributable to increased thirst or to hydrophilic delusions [71, 72]. Most patients say they do not drink because of particular health beliefs but simply because it makes them feel better [73]. The osmotic set point for desiring water, like that for AVP, appears to be reset downward in polydipsic patients with or without hyponatremia [65]. This suggests that thirst regulation is generally intact but altered by an extrinsic factor [11, 65]. Initial studies indicated that the reset osmostat and primary polydipsia were independent of each other. More recent data, however, indicates abnormalities in fluid intake and excretion may both be related to hippocampal dysfunction [74]. Hippocampal lesions enhance the development of polydipsia and stereotypic behaviors in animal models [75], and polydipsic patients exhibit an increased incidence of stereotypic behaviors compared with other schizophrenia patients [76, 77]. Some evidence suggests that polydipsia (and perhaps other stereotypes) reduces arousal, constituting an adaptive response to the enhanced stress reactivity arising from the hippocampal dysfunction. This enhanced stress reactivity, in turn, underlies the reset osmostat and enhanced stress hormone responses as discussed below [78]. Another possibility, not inconsistent with that stated above, is that hippocampal dysfunction enhances the rewarding properties of water [79]. Further research into the mechanism of polydipsia may provide insights into the behavioral abnormalities associated with schizophrenia.

Mechanism of Reset Osmostat in PIP Syndrome

In addition to reset osmostat, PIP patients also exhibit enhanced renal sensitivity to low levels of AVP [11, 65]. Together with primary polydipsia they may account for the moderate chronic hyponatremia but are insufficient to produce episodic water intoxication. Previous observations suggested that impaired water excretion worsens during acute psychosis [5, 6], and one study specifically found that reset osmostat worsened during a psychotic relapse [80]. The mechanism of this association was addressed by giving a psychotomimetic to a group of clinically stabilized hyponatremic and normonatremic polydipsic subjects. Following the psychotomimetic, psychotic symptoms increased similarly in the two groups, but plasma AVP increased more in those with hyponatremia while the desire to drink did not change in either group [80]. The peak AVP response was predicted by basal positive psychotic symptoms in both groups. Moreover, it was proportional to concurrent plasma osmolality in those with hyponatremia (Fig. 11.2). This latter observation indicates that psychosis may further lower the set point in the PIP syndrome to a level capable of inducing water intoxication. Concurrent measures of recognized and putative physiologic factors did not explain why the set point shifted.

The anterior hippocampus normally restrains AVP and hormone responses to psychological stress [81, 82]. Further, the hippocampal segment is smaller [83, 84], stress hormone activity is elevated [85], and the reset osmostat normalizes with clinical stabilization [74, 80] in PIP patients relative to others with schizophrenia. This led to the hypothesis that hippocampal pathology might induce an enhanced AVP response to stress in PIP patients [74]. This hypothesis was supported by subsequent studies. One study demonstrated that both AVP and stress hormone responses to a psychological, but not a physical stimulus, were enhanced in PIP patients relative to healthy controls and nonpolydipsic patients [86]. Since AVP response to the psychological stressor was predicted by concurrent plasma osmolality, the hypothesis that a heightened response to stress lowers the AVP set point was supported. A second study reported that hippocampal-mediated negative feedback which contributes to braking the stress response was nearly absent in PIP patients, further linking the findings to hippocampal dysfunction [87]. A third study found these neuroendocrine findings were proportional to deformations on the surface of the hippocampus overlying the segment (i.e., anterior lateral) which projects to the anterior hypothalamus and normally restrains neuroendocrine release [88].



Fig. 11.2 Effect of acute psychosis on AVP regulation in polydipsic patients. An infusion of the psychotomimetic, methylphenidate, was given to hyponatremic and normonatremic polydipsic patients to induce a brief psychotic exacerbation. Peak AVP levels were higher in the hyponatremic patients and moreover were predicted by their concurrent plasma osmolality. In contrast, there was no significant relationship between peak AVP and plasma osmolality in the normonatremic patients (not shown). The mean linear relationships between AVP and plasma osmolality for the two groups when clinically stabilized is also shown, demonstrating how the set point (*x*-intercept) shifts in the hyponatremic group with acute psychosis without changing the slope of the relationship. The drop in the set point during acute psychosis (i.e., 242 mOsm/kg) is sufficient to produce water intoxication. The mean linear relationship of nonpolydipsic patients is also included, and highlights how "medullary washout" from polydipsia blunts the relationship between plasma osmolality and P_{AVP}

The hippocampal findings summarized above are also proportional to deformations on the amygdala and anterior hypothalamus [88]. These two structures, like the hippocampus, modulate both neuroendocrine activity to stress and are implicated in the pathophysiology of schizophrenia. In particular, the same hippocampal region that restrains these neuroendocrine responses also restrains dopamine release in the ventral striatum which many believe underlies acute psychosis and behavioral response to stress and is essential for coping efforts to buffer the impact of stress [79, 89]. Direct evidence linking hippocampal dysfunction to mental illness also comes from studies of plasma oxytocin, and the effects of intranasal oxytocin treatment on these patients' marked deficits in social functioning [59]. Taken as a whole, the data support the view that there are disruptions in both peripheral and central neuroendocrine function in polydipsic patients which are attributable to hippocampal dysfunction and contribute to a stress diathesis that accounts for the core features of their illness [74, 88].

Unlike the transient changes in the set point for AVP secretion, little is known about the mechanism of the enhanced renal sensitivity to AVP action and its relationship, if any, to its altered secretion or psychosis. One possibility is the effects of chronic antipsychotic treatment [90]. Polydipsia does not appear to be responsible, although the possibility there is an idiosyncratic response (perhaps due to longstanding polydipsia) in PIP patients cannot be excluded. One potential explanation which has not been explored is variations in the genetic and functional properties of the renal AVP (V2) receptors.

Treatment and Prevention

Polydipsia

Primary polydipsia is typically not treated (Table 11.2). Efforts to restrict fluid intake, even in supervised settings, are rarely successful. In previous years, ongoing group and behavioral therapies were reported effective for inpatients on long-term units [20, 91]. Many putative therapeutic agents (e.g., beta-blockers, clonidine, ACE inhibitors, naltrexone) have been tried but their efficacy remains uncertain [92]. Clozapine's salutary effects on water imbalance in PIP patients may be due to its reduction of primary polydipsia, but using clozapine solely for this reason may not be appropriate due to its associated risks [93].

Because of the increased risk of renal failure, lithium-induced polydipsia must not be ignored. Lowering the lithium dose or changing to a single daily dose may help. Addition of a thiazide (e.g., 25 mg of hydrochlorthiazide) or potassium-sparing diuretic (e.g., 5 mg of amiloride) is generally effective in reducing polydipsia, though it is unclear whether this reduces the risk of renal disease (Table 11.2) [94]. If adding a thiazide diuretic, the lithium dose should first be reduced by about one third as more lithium will be retained by the kidney. Lithium levels should initially be followed at more frequent intervals to assure they remain therapeutic and safe. It is important to realize that these agents place patients at an increased risk of lithium intoxication (e.g., dehydration and salt loss during hot dry weather) which is also a risk factor for renal failure [28]. Theoretically, amiloride which blocks lithium uptake into renal tubule cells should lower the risk of nephrotoxicity [29]. In any case, patients need to maintain fluid and salt intake during heat spells or periods of intense exercise. Also, if polyuria is due to primary polydipsia rather than lithium treatment, a thiazide diuretic will predispose the patient to water intoxication.

Hyponatremia in the PIP Syndrome and with Recognized Causes

Following an acute episode of water intoxication, the appropriate response to most PIP patients is close observation and fluid restriction. This conservative approach is predicated on the relatively minor contribution of the impairment in water excretion to the hyponatremia, the rapid shifts in fluid balance and the increased risk of overcorrection due to the concurrent concentrating defect. Thus, patients who have seized and regained consciousness or who are exhibiting symptoms of impending water intoxication (tremors, ataxia, vomiting) can be fluid restricted for 3–4 h during which time they will excrete large amounts of the retained water. Patients must be closely monitored as they may continue to absorb ingested water from their gastrointestinal tract and (re)develop water intoxication. Monitoring changes in body weight is an effective means of assessing the efficacy of fluid restriction [14].

Prolonged fluid restriction should be avoided because of the concentrating defect and subsequent risk of overcorrection leading to central pontine myelinolysis (CPM). The risk of CPM in these patients may be, however, somewhat overstated because there is not adequate time for the patient to adapt to the lower sodium level (which is critical to CPM) [95]. Particular caution should be taken, however, in patients with alcoholism, malnutrition, medication-induced hyponatremia, or a subacute course of hyponatremia where the risk of CPM is high [47, 48]. While the risk of overcorrection or rapid correction producing CPM in PIP patients appears low, it does appear to induce rhabdomyolysis [53].

If the patient does not rapidly regain consciousness, standard measures of restoring water balance (e.g., saline infusion) should be considered. In the absence of reversible causes, preventive measures are appropriate following resolution of the acute episode.

Prevention

Targeted fluid restriction is an effective means of preventing water intoxication in PIP syndrome in an inpatient or nursing home setting and enables the patient and nursing staff to avoid the intensity of constant monitoring (Table 11.2) [14, 96]. The procedure relies on the fact that water retention leads to easily detected gains in body weight, and the "target weight" is based on an estimate of the weight gain at which the patient's hyponatremia is more severe than usual but not yet at a level likely to induce water intoxication. The procedure requires weighing patients twice a day (morning and afternoon) and whenever latent signs (light headedness, dizziness, bloating, or nausea) of water intoxication are observed. Initially, the target can be set at 7–10 lb above the average morning water weight, and a sodium level obtained when this target is exceeded. If the target is too conservative, it can be increased to a level more likely associated with symptomatic hyponatremia. Thus, it may take a few days to find a target weight that conforms to a sodium level which warrants fluid restriction. Fluid restriction for 2–6 h is usually sufficient to return a patient to the morning weight. Again, overcorrection should be avoided.

Pharmacologic Treatment

Until the introduction of clozapine and the vaptans, pharmacologic options were limited. While many agents were tested, with the possible exception of demeclocyline, they were not effective [72].

Clozapine

Many investigators report that clozapine reduces the risk of hyponatremia and water intoxication, even though there are no double-blind placebo-controlled studies [93]. The mechanism is unclear, but clozapine may lower water intake rather than enhance water excretion. The drug often normalizes sodium levels within several weeks, enabling patients to leave restricted settings and participate in therapeutic programming. Clozapine may be effective in low doses (approximately 100 mg/day), and there is no evidence that going above 300 mg/day will produce further improvements in water balance [93]. Water intoxication and other sequelae of polydipsia have, however, occurred in patients on clozapine [97]. In addition, clozapine requires monitoring not available to all physicians, and is associated with an impressive array of life-threatening adverse effects. Thus, the risk/benefit ratio must be examined, particularly in patients without a history of frank water intoxication, and whose chronic moderate hyponatremia does not appear to overtly impair cognition or contribute to altered balance and associated fractures.

Vaptans

A double-blind study documented the marked efficacy of tolvaptan which is a competitive vasopressin receptor 2 antagonist [98]. Nineteen PIP subjects were randomly assigned to receive placebo (n = 12) or tolvaptan (n = 7) once daily for 30 days at a dose of 15-60 mg, based on serum sodium changes. Baseline sodium levels were 130 mEq/L in both groups, and normalized (>135 mEq/L) within 24 h of treatment with tolvaptan while they did not change with placebo (p < 0.005 at all samples after day 4). The salutary effects were apparent throughout the 30-day treatment period and subjects on tolvaptan returned to previous hyponatremic levels after treatment was stopped. Two subjects receiving active drug (28.6 %) became dehydrated and experienced hypotension, while five subjects receiving placebo (41.7 %) experienced symptoms associated with latent or overt water intoxication. The study included an open-label extension arm during which the salutary effects appeared to be maintained (Goldman and Josiassen; unpublished data). As of this writing, in the United States these medications are limited to acute inpatient settings but presumably will become integrated into routine outpatient care as well. Careful dose adjustment is needed to prevent dehydration, particularly in patients with concurrent medical problems (e.g., renal insufficiency).

Conclusion

Despite the clinical significance of the water imbalance and the effective interventions that are available, hyponatremia is frequently undetected in schizophrenia, making its recognition and proper management an important and neglected aspect of patient care. Chronic moderate hyponatremia produces significant morbidity, while episodic water intoxication has caused many deaths in those with severe mental illness. Episodic water intoxication frequently occurs in the PIP syndrome and is a consequence of a marked primary polydipsia and transient mild impairments in water excretion that frequently coincide with psychotic exacerbations. A series of investigations have shown the impaired water excretion is attributable to resetting of the osmostat for AVP release and potentially characterizes a distinct subset of schizophrenia patients with enhanced stress reactivity.

Moderate chronic hyponatremia is frequently a consequence of medicationinduced impairments in water excretion, which may or may not be compounded by primary polydipsia. The impairment in water excretion in these cases tends to be more marked and stable, resembling classic SIADH. When hyponatremia occurs outside of these scenarios other causes must be vigorously pursued.

Prevention of water intoxication in the PIP syndrome includes targeted fluid restriction, clozapine, and vasopressin antagonism. For medication-induced hyponatremia the offending agent should be discontinued and replaced. Antipsychotic medication is rarely responsible for hyponatremia in severe mental illness, and routine discontinuation may aggravate the water imbalance.

References

- 1. Hoskins RG, Sleeper FH. Organic functions in schizophrenia. Arch Neurol Psychiatry. 1933;30:123-40.
- Targowla R. Des troubles fonctionnel du rein dans les maladies mentales. L'excretion del'eau (Kidney malfunction and mental illness: water excretion). Bull Mem Soc Med Hop Paris. 1923;47:1711–5.
- 3. Barahal HS. Water intoxication in a mental case. Psychiatry Q. 1938;12:767-71.
- 4. Jos CJ. Generalized seizures from self-induced water intoxication. Psychosomatics. 1984;25:153–7.
- 5. Barlow ED, De Wardener HE. Compulsive water drinking. QJM. 1959;28:235-58.
- 6. Hobson JA, English JT. Self-induced Water Intoxication. Ann Intern Med. 1963;58:324-32.
- Riggs A, Dysken M, Kim S, Opsahl J. A review of disorders of water homeostasis in psychiatric patients. Psychosomatics. 1991;32:133–48.
- Cronin RE. Psychogenic polydipsia with hyponatremia: report of eleven cases. Am J Kidney Dis. 1987;9:410–6.
- Smith WO, Clark ML. Self-induced water intoxication in schizophrenic patients. Am J Psychiatry. 1980;137:1055–60.
- 10. Hariprasad MK, Eisinger RP, Nadler IM, Padmanabhan CS, Nidus BD. Hyponatremia in psychogenic polydipsia. Arch Intern Med. 1980;140:1639–42.
- Goldman MB, Luchins DJ, Robertson GL. Mechanisms of altered water metabolism in psychotic patients with polydipsia and hyponatremia. N Engl J Med. 1988;318:397–403.
- Kishimoto T, Hirai M, Ohsawa H, Terada M, Matsuoka I, Ikawa G. Manners of arginine vasopressin secretion in schizophrenic patients-with reference to the mechanism of water intoxication. Jpn J Psychiatry Neurol. 1989;43:161–9.
- 13. Ragavan V, Verbalis J., Woods M. Psychogenic polydipsia and hyponatremia: evidence for a reset osmostat. In: Seventh international congress of endocrinology, Quebec, Canada; 1984.
- Vieweg WV. Treatment strategies in the polydipsia-hyponatremia syndrome. J Clin Psychiatry. 1994;55:154–60.

- de Leon J, Verghese C, Tracy JI, Josiassen RC, Simpson GM. Polydipsia and water intoxication in psychiatric patients: A review of the epidemiological literature. Biol Psychiatry. 1994; 35:408–19.
- 16. Mercier-Guidez E, Loas G. Polydipsia and water intoxication in 353 psychiatric inpatients: an epidemiological and psychopathological study. Eur Psychiatry. 2000;15:306–11.
- Meerabux J, Iwayama Y, Sakurai T, et al. Association of an orexin 1 receptor 408Val variant with polydipsia-hyponatremia in schizophrenic subjects. Biol Psychiatry. 2005;58:401–7.
- 18. Shinkai T, Ohmori O, Hori H, Nakamura J. Genetic approaches to polydipsia in schizophrenia: a preliminary report of a family study and an association study of an angiotensin-converting enzyme gene polymorphism. Am J Med Genet B Neuropsychiatr Genet. 2003;119B:7–12.
- 19. Ahmed AG, Heigh LM, Ramachandran KV. Polydipsia, psychosis, and familial psychopathology. Can J Psychiatry. 2001;46:522–7.
- 20. Movig KL, Baumgarten R, Leufkens HG, van Laarhoven JH, Egberts AC. Risk factors for the development of lithium-induced polyuria. Br J Psychiatry. 2003;182:319–23.
- 21. Presne C, Fakhouri F, Noel LH, et al. Lithium-induced nephropathy: Rate of progression and prognostic factors. Kidney Int. 2003;64:585–92.
- Delva NJ, Crammer JL, Jarzylo SV, et al. Osteopenia, pathological fractures, and increased urinary calcium excretion in schizophrenic patients with polydipsia. Biol Psychiatry. 1989;26:781–93.
- 23. Jacob S, Spinler SA. Hyponatremia associated with selective serotonin-reuptake inhibitors in older adults. Ann Pharmacother. 2006;40:1618–22.
- 24. Kim CS, Choi JS, Bae EH, Kim SW. Hyponatremia associated with bupropion. Electrolyte Blood Press. 2011;9:23–6.
- 25. Kirby D, Harrigan S, Ames D. Hyponatraemia in elderly psychiatric patients treated with Selective Serotonin Reuptake Inhibitors and venlafaxine: a retrospective controlled study in an inpatient unit. Int J Geriatr Psychiatry. 2002;17:231–7.
- 26. Letmaier M, Painold A, Holl AK, et al. Hyponatraemia during psychopharmacological treatment: results of a drug surveillance programme. Int J Neuropsychopharmacol. 2012;15(6):739–48.
- Palmer BF, Gates JR, Lader M. Causes and management of hyponatremia. Ann Pharmacother. 2003;37:1694–702.
- 28. Siegler EL, Tamres D, Berlin JA, Allen-Taylor L, Strom BL. Risk factors for the development of hyponatremia in psychiatric inpatients. Arch Intern Med. 1995;155:953–7.
- 29. Van Amelsvoort T, Bakshi R, Devaux CB, Schwabe S. Hyponatremia associated with carbamazepine and oxcarbazepine therapy: a review. Epilepsia. 1994;35:181–8.
- 30. Siegel AJ. Hyponatremia in psychiatric patients: update on evaluation and management. Harv Rev Psychiatry. 2008;16:13–24.
- 31. Movig KLL, Leufkens HGM, Lenderink AW, et al. Association between antidepressant drug use and hyponatraemia: a case–control study. Br J Clin Pharmacol. 2002;53:363–9.
- Budisavljevic MN, Stewart L, Sahn SA, Ploth DW. Hyponatremia associated with 3,4-Methylenedioxymethylamphetamine ("ecstasy") abuse. Am J Med Sci. 2003;326:89–93.
- de Leon J. Polydipsia-a study in a long-term psychiatric unit. Eur Arch Psychiatry Clin Neurosci. 2003;253:37–9.
- 34. de Leon J, Dadvand M, Canuso C, Odom-White A, Stanilla J, Simpson GM. Polydipsia and water intoxication in a long-term psychiatric hospital. Biol Psychiatry. 1996;40:28–34.
- Ohsawa H, Kishimoto T, Hirai M, et al. An epidemiological study on hyponatremia in psychiatric patients in mental hospitals in Nara Prefecture. Jpn J Psychiatry Neurol. 1992;46:883–9.
- Allon M, Allen HM, Deck LV, Clark ML. Role of cigarette use in hyponatremia in schizophrenic patients. Am J Psychiatry. 1990;147:1075–7.
- 37. Poirier S, Legris G, Tremblay P, et al. Schizophrenia patients with polydipsia and water intoxication are characterized by greater severity of psychotic illness and a more frequent history of alcohol abuse. Schizophr Res. 2010;118:285–91.
- Lacarta GL, Chiappetta VI, Peluffo I. Hyponatremia associated with psychotropic drugs: a side effect to consider. Vertex. 2008;19:364–70.

- 11 Management of Water Imbalance in Schizophrenia
- 39. Arieff AI, Llach F, Massry SG. Neurological manifestations and morbidity of hyponatremia: correlation with brain water and electrolytes. Medicine (Baltimore). 1976;55:121–9.
- 40. Webb Jr WL, Gehi M. Electrolyte and fluid imbalance: neuropsychiatric manifestations. Psychosomatics. 1981;22:199–203.
- Geboy AG, Filmyer DM, Josiassen RC. Motor deficits associated with mild, chronic hyponatremia: a factor analytic study. J Mot Behav. 2012;44:255–9.
- 42. Bun S, Serby MJ, Friedmann P. Psychotropic Medication Use, Hyponatremia, and Falls in an Inpatient Population: A Retrospective Study. J Clin Psychopharmacol. 2011;31:395–7.
- 43. Hoorn EJ, Rivadeneira F, van Meurs JBJ, et al. Mild hyponatremia as a risk factor for fractures: The rotterdam study. J Bone Miner Res. 2011;26:1822–8.
- 44. Renneboog B, Musch W, Vandemergel X, Manto MU, Decaux G. Mild chronic hyponatremia is associated with falls, unsteadiness, and attention deficits. Am J Med. 2006;119:71.e1–e8.
- Koczapski AB, Millson RC. Individual differences in serum sodium levels in schizophrenic men with self-induced water intoxication. Am J Psychiatry. 1989;146:1614–5.
- 46. Adrogué HJ, Madias NE. Hyponatremia. N Engl J Med. 2000;342:1581-9.
- 47. Reddy P, Mooradian AD. Diagnosis and management of hyponatraemia in hospitalised patients. Int J Clin Pract. 2009;63:1494–508.
- Tanneau RS, Henry A, Rouhart F, et al. High incidence of neurologic complications following rapid correction of severe hyponatremia in polydipsic patients. J Clin Psychiatry. 1994;55: 349–54.
- Adler S. Hyponatremia and rhabdomyolysis: a possible relationship. South Med J. 1980;73:511–2.
- 50. Lara Aguayo P, de la Fuente Martos C, Moran Fernandez E, Soriano Rodriguez F, Rojas Amezcua M, Aguilar Alonso E. Rhabdomyolysis secondary to hyponatraemia. Nefrologia 2011;31:500–2.
- Maiocchi L, Bernardi E. Acute anterior compartment syndrome associated with psychogenic polydipsia. Australas Psychiatry. 2012;20:159–61.
- Meulendijks D, Mannesse CK, Jansen PAF, van Marum RJ, Egberts TCG. Antipsychoticinduced hyponatraemia: a systematic review of the published evidence. Drug Saf. 2010;33:101–14.
- 53. Morita S, Inokuchi S, Yamamoto R, et al. Risk factors for rhabdomyolysis in self-induced water intoxication (SIWI) patients. J Emerg Med. 2010;38:293–6.
- Vieweg WV, David JJ, Rowe WT, Wampler GJ, Burns WJ, Spradlin WW. Death from selfinduced water intoxication among patients with schizophrenic disorders. J Nerv Ment Dis. 1985;173:161–5.
- 55. Jose CJ, Perez-Cruet J. Incidence and morbidity of self-induced water intoxication in state mental hospital patients. Am J Psychiatry. 1979;136:221–2.
- 56. Williams ST, Kores RC. Psychogenic polydipsia: comparison of a community sample with an institutionalized population. Psychiatry Res. 2011;187:310–1.
- Vieweg V, Rowe W, David J, Spradlin W. Hyposthenuria as a marker for self-induced water intoxication and schizophrenic disorders. Am J Psychiatry. 1984;141:1258–60.
- Zilles D, Hasan A, Gruber O, Degner D. Acute polydipsia and water intoxication in first episode schizophrenia. Aust N Z J Psychiatry. 2010;44:489.
- Bralet MC, Ton T, Falissard B. Schizophrenic patients with polydipsia and water intoxication more often have a form of schizophrenia first described by Kraepelin. Psychiatry Res. 2007;152:267–71.
- Koczapski AB, Ashby YT, Ibraheem S, Paredes J, Jones BD, Ancill R. 'Afternoon radiator sitting syndrome': hyperthermia and early diagnosis of self induced water intoxication. Br J Psychiatry. 1987;151:133–4.
- 61. Fukunaka Y, Shinkai T, Hwang R, et al. The Orexin 1 Receptor (HCRTR1) Gene as a Susceptibility Gene Contributing to Polydipsia-Hyponatremia in Schizophrenia. Neuromolecular Med 2007:292–7.
- Kopala LC, Good KP, Koczapski AB, Honer WG. Olfactory deficits in patients with schizophrenia and severe polydipsia. Biol Psychiatry. 1998;43:497–502.

- 63. Hawken ER, Crookall JM, Reddick D, Millson RC, Milev R, Delva N. Mortality over a 20-year period in patients with primary polydipsia associated with schizophrenia: a retrospective study. Schizophr Res. 2009;107:128–33.
- Abbasi QA, Carbonell FE, Koczapski AB, Vieweg WV. Measuring and estimating daily urine volume in psychiatric patients: strengths and weaknesses. Schizophr Res. 1997;28:87–93.
- Goldman MB, Robertson GL, Luchins DJ, Hedeker D. The influence of polydipsia on water excretion in hyponatremic, polydipsic, schizophrenic patients. J Clin Endocrinol Metab. 1996;81:1465–70.
- 66. Raskind MA, Courtney N, Murburg MM, et al. Antipsychotic drugs and plasma vasopressin in normals and acute schizophrenic patients. Biol Psychiatry. 1987;22:453–62.
- Alvelos M, Ferreira A, Bettencourt P, et al. Effect of Saline Load and Metoclopramide on the Renal Dopaminergic System in Patients with Heart Failure and Healthy Controls. J Cardiovasc Pharmacol. 2005;45:197–203.
- Ajlouni K, Kern MW, Tures JF, Theil GB, Hagen TC. Thiothixene-Induced Hyponatremia. Arch Intern Med. 1974;134:1103–5.
- 69. Madhusoodanan S, Bogunovic OJ, Moise D, Brenner R, Markowitz S, Sotelo J. Hyponatraemia associated with psychotropic medications: a review of the literature and spontaneous reports. Adverse Drug React Toxicol Rev. 2002;21:17–29.
- 70. Mannesse CK, van Puijenbroek EP, Jansen PAF, van Marum RJ, Souverein PC, Egberts TCG. Hyponatraemia as an adverse drug reaction of antipsychotic drugs: a case–control study in VigiBase. Drug Saf. 2010;33:569–78.
- McKinley MJ, Cairns MJ, Denton DA, et al. Physiological and pathophysiological influences on thirst. Physiol Behav. 2004;81:795–803.
- 72. Goldman MB. The assessment and treatment of water imbalance in patients with psychosis. Clin Schizophr Relat Psychoses. 2010;4:115–23.
- 73. Millson RC, Koczapski AB, Cook MI, Daszkiewicz M. A survey of patient attitudes toward self-induced water intoxication. Can J Psychiatry. 1992;37:46–7.
- 74. Goldman MB. The mechanism of life-threatening water imbalance in schizophrenia and its relationship to the underlying psychiatric illness. Brain Res Rev. 2009;61:210–20.
- Luchins DJ. A possible role of hippocampal dysfunction in schizophrenic symptomatology. Biol Psychiatry. 1990;28:87–91.
- Shutty Jr MS, McCulley K, Pigott B. Association between stereotypic behavior and polydipsia in chronic schizophrenic patients. J Behav Ther Exp Psychiatry. 1995;26:339–43.
- Luchins DJ, Goldman MB, Lieb M, Hanrahan P. Repetitive behaviors in chronically institutionalized schizophrenic patients. Schizophr Res. 1992;8:119–23.
- 78. Mittleman G, Jones GH, Robbins TW. The relationship between schedule-induced polydipsia and pituitary-adrenal activity: pharmacological and behavioral manipulations. Behav Brain Res. 1988;28:315–24.
- Lodge DJ, Grace AA. Developmental pathology, dopamine, stress and schizophrenia. Int J Dev Neurosci. 2011;29:207–13.
- Goldman MB, Robertson GL, Luchins DJ, Hedeker D, Pandey GN. Psychotic exacerbations and enhanced vasopressin secretion in schizophrenic patients with hyponatremia and polydipsia. Arch Gen Psychiatry. 1997;54:443–9.
- Herman JP, Dolgas CM, Carlson SL. Ventral subiculum regulates hypothalamo-pituitary-adrenocortical and behavioural responses to cognitive stressors. Neuroscience. 1998;86:449–59.
- Nettles KW, Pesold C, Goldman MB. Influence of the ventral hippocampal formation on plasma vasopressin, hypothalamic-pituitary-adrenal axis, and behavioral responses to novel acoustic stress. Brain Res. 2000;858:181–90.
- Luchins DJ, Nettles KW, Goldman MB. Anterior medial temporal lobe volumes in polydipsic schizophrenic patients with and without hypo-osmolemia: a pilot study. Biol Psychiatry. 1997;42:767–70.
- Goldman MB, Torres IJ, Keedy S, Marlow-O'Connor M, Beenken B, Pilla R. Reduced anterior hippocampal formation volume in hyponatremic schizophrenic patients. Hippocampus. 2007;17:554–62.

- 11 Management of Water Imbalance in Schizophrenia
- Goldman MB, Blake L, Marks RC, Hedeker D, Luchins DJ. Association of nonsuppression of cortisol on the DST with primary polydipsia in chronic schizophrenia. Am J Psychiatry. 1993;150:653–5.
- Goldman MB, Gnerlich J, Hussain N. Neuroendocrine responses to a cold pressor stimulus in polydipsic hyponatremic and in matched schizophrenic patients. Neuropsychopharmacology. 2006;32:1611–21.
- Goldman MB, Wood G, Goldman MB, et al. Diminished glucocorticoid negative feedback in polydipsic hyponatremic schizophrenic patients. J Clin Endocrinol Metab. 2007;92:698–704.
- Goldman MB, Wang L, Wachi C, et al. Structural pathology underlying neuroendocrine dysfunction in schizophrenia. Behav Brain Res. 2011;218:106–13.
- Rowland LM. Who Is Resilient to Depression? Multimodal Imaging of the Hippocampus in Preclinical Chronic Mild Stress Model May Provide Clues. Biol Psychiatry. 2011;70:406–7.
- Goldman MB, Robertson GL, Hedeker D. Oropharyngeal regulation of water balance in polydipsic schizophrenics. Clin Endocrinol (Oxf). 1996;44:31–7.
- 91. Costanzo ES, Antes LM, Christensen AJ. Behavioral and medical treatment of chronic polydipsia in a patient with schizophrenia and diabetes insipidus. Psychosom Med. 2004;66:283–6.
- 92. Brookes G, Ahmed AG. Pharmacological treatments for psychosis-related polydipsia. Cochrane Database Syst Rev 2006:CD003544.
- Canuso CM, Goldman MB. Clozapine restores water balance in schizophrenic patients with polydipsia-hyponatremia syndrome. J Neuropsychiatry Clin Neurosci. 1999;11:86–90.
- 94. Timmer RT, Sands JM. Lithium intoxication. J Am Soc Nephrol. 1999;10:666-74.
- Cheng JC, Zikos D, Skopicki HA, Peterson DR, Fisher KA. Long-term neurologic outcome in psychogenic water drinkers with severe symptomatic hyponatremia: the effect of rapid correction. Am J Med. 1990;88:561–6.
- Goldman MB, Luchins DJ. Prevention of episodic water intoxication with target weight procedure. Am J Psychiatry. 1987;144:365–6.
- Tenyi T, Voros V. Successful switch to olanzapine after rhabdomyolysis caused by water intoxication and clozapine use. Pharmacopsychiatry. 2006;39:157–8.
- Josiassen RC, Goldman M, Jessani M, et al. Double-blind, placebo-controlled, multicenter trial of a vasopressin V2-receptor antagonist in patients with schizophrenia and hyponatremia. Biol Psychiatry. 2008;64:1097–100.

Chapter 12 Management of Medication-Related Adverse Effects

Wanlop Atsariyasing and Morris Goldman

Introduction

The first part of this chapter reviews the major adverse effects (AEs) which typically antipsychotic selection: Sudden death. Movement influence disorders. Hyperprolactinemia, Sexual dysfunction, and Metabolic syndrome. For each of these AEs we divide the discussion into three parts: (1) definition, significance, patient risk factors, and mechanisms; (2) risks of different antipsychotics; and (3) guidelines for prevention and treatment. Since metabolic syndrome and other risk factors relevant to cardiovascular disease are discussed in detail elsewhere (Chap. 10), we restrict ourselves to those issues most relevant to antipsychotic therapy. Next, we consider advanced age because of specific concerns for this group of patients, especially those with dementia. There are many elderly patients who receive antipsychotics who do not have a history of schizophrenia, and most of the data reported here is not specific to the schizophrenic patient. Clozapine is also considered separately because its risk-benefit ratio differs from other agents. Finally, we consider less common but severe AEs which require immediate attention including neuroleptic malignant syndrome (NMS), laryngeal dystonia, thromboembolism, and respiratory distress in the newborn.

M. Goldman

W. Atsariyasing (🖂)

Department of Psychiatry, Faculty of Medicine Siriraj Hospital, Mahidol University, 2 Prannok Road, Chalermprakiat Building, 8th Floor, Bangkok, 0700, Thailand e-mail: wanlop.atr@mahidol.ac.th

Department of Psychiatry and Behavioral Sciences, Northwestern University, East Ontario 446, Chicago, IL 60611, USA e-mail: m-goldman@northwestern.edu

Sudden Death

Definition, Significance, Patient Risk Factors, and Mechanisms

Since the 1960s, antipsychotics have been associated with an increased risk of sudden death (e.g., thioridazine) [1]. This is the unexpected death of an individual which occurs within 1 h of the onset of acute symptoms if there are witnesses, or the unexpected death of an individual who is known to have been medically stable less than 24 h previously if there are no witnesses [2]. The risk of sudden death in patients receiving antipsychotics is reported to be 2.4 times higher than those who did not receive these medications, and appears to increase with dose [3–7]. The incidence of sudden death is 2.9 events per 1,000 patient-years which is substantially higher than other recognized antipsychotic-related safety concerns (e.g., the rate of death from clozapine due to agranulocytosis is approximately 0.2 per 1,000 patient-years) [7, 8].

In addition to medication, other risk factors for sudden death in psychotic patients include female gender, hypertension, diabetes mellitus, increased age, cardiovascular disease, and electrolyte imbalance [1-3, 9]. Stroke, aortic dissection, and pulmonary embolism (PE) are also potential consequences of long-term antipsychotic therapy and may contribute to sudden death. The increased incidence, however, is largely attributable to the metabolic syndrome discussed here and in Chap. 10. Further, assessing the role of antipsychotics in sudden death is complicated by the approximate fourfold higher mortality risk of having schizophrenia, which at least over the typical 8- to 16-week, double-blind, placebo-controlled study time frame is similar in treated and untreated patients [10].

Antipsychotics likely contribute to some cases of sudden death due to their acute arrhythmogenic potential [2]. In most cases, other risk factors are typically present (Tables 12.1, 12.2, and 12.3). Because life-threatening arrhythmias are rarely observed, and their etiology even more rarely deduced, their role (i.e., primary or secondary) in sudden death is difficult to determine. Instead, primary arrhythmogenic risk is inferred from measures of the QT interval on an electrocardiogram (ECG). Increases in the QTc interval may trigger torsade de pointes (TdP), an arrhythmia which can cause ventricular fibrillation and death [1, 11–13].

The QT interval starts at the QRS complex and ends at the T wave (Fig. 12.1). It represents the time between the onset of ventricular depolarization and the end of repolarization. The QT interval varies with the heart rate (i.e., the lower the heart rate, the longer the QT interval). Various formulas correct for heart rate (QTc), the most commonly used being Bazett's formula (QTc=QT interval in seconds divided by the square root of the RR interval in seconds, QT/RR^{1/2}). QT measurement may be influenced by a number of factors including the shape of the U wave, ECG paper speeds, ECG lead placement, gender, time of day, and menstrual cycle [12]. QTc interval is considered prolonged if it is greater than 450 ms for men and 470 ms for women [14]. Most, but not all evidence, supports the inference that QTc interval prolongation increases the risks of TdP and sudden cardiac death [1, 12, 15–18].

Table 12.1 Risk factors of QTc prolongation and TdP [1, 12, 13, 17, 22, 242–245]^a

Table 12.2 Other drugs associated with QT prolongation [2, 12, 17, 19, 244, 246–249]

Cardiovascular disease Congenital long QT syndrome Bradycardia Ischemic heart disease Myocarditis Myocardial infarction Congestive heart failure Left ventricular hypertrophy Electrolyte imbalance Hypokalemia Hypomagnesemia Hypocalcemia Pharmacokinetic factors Poor metabolizer Inhibition of specific cytochrome P450 enzymes Competition of specific cytochrome P450 enzymes Others Older age Female gender Extreme physical exertion Restraint and psychological stress Anorexia nervosa Renal and hepatic impairment Obesity (BMI>25)

QTc corrected QT interval, *TdT* Torsades de Points ^aSee Table 12.2 for additional medications which prolong the QT interval

Psychotropics	Antiarrhythmics
Bupropion	Amiodarone
Citalopram	Bretylium
Escitalopram	Disopyramide
Fluoxetine	Dofetilide
Lithium	Ibutilide
Sertraline	Procainamide
Trazodone	Quinidine
Tricyclic antidepressants	Sotalol
Venlafaxine	Others
Antibiotics	Amantadine
Clarithromycin	Cyclosporin
Co-trimoxazole	Diphenhydramine
Erythromycin	Domperidone
Fluoroquinolones	Hydroxyzine
Antimalarials	Methadone
Chloroquine	Nicardipine
Mefloquine	Ondansetron
Quinine	Tamoxifen
	Terfenadine

Cytochrome		
P(CYP)450 enzyme	Substrate	Inhibitor
CYP1A2	Clozapine	Ciprofloxacin
	Haloperidol	Fluvoxamine
		Grapefruit juice
CYP2D6	Butyrophenones (e.g. haloperidol)	Beta-blockers (e.g. propanolol)
	Phenothiazines (e.g. chlorpromazine, thioridazine)	Bupropion
	Thioxanthines (e.g. flupentixol)	Fluoxetine
	Clozapine	Haloperidol
	Risperidone	Paroxetine
	Sertindole	Phenothiazines
	Zuclopenthixol	Quinidine
		Tricyclic antidepressants
CYP3A4	Risperidone	Grapefruit juice
	Quetiapine	HIV proteases inhibitors (e.g. indinavir, ritonavir)
	Pimozide	Itraconazole
		Ketoconazole
		Macrolide antibiotics (e.g.
		clarithromycin)
		Nefazodone

 Table 12.3
 Antipsychotic medications and commonly used enzyme inhibitors [2, 12, 244, 250]



Fig. 12.1 QT interval



Fig. 12.2 Mean change from baseline in QTc for antipsychotics. Bazett's correction. Study 054 as presented to FDA; 2000 [257]. Proposed USPI data for aripiprazole. Data on file Bristol-Myers Squibb

The evidence is strongest for QTc durations greater than 500 ms and for QTc increases from baseline greater than 60 ms [12, 13].

Antipsychotics prolong the QT interval by blocking the HERG potassium ion channels in cardiomyocytes [13, 19–22]. TdP is a ventricular arrhythmia characterized by a gradual change in the amplitude and twisting of the QRS complex around the isoelectric line. Although usually spontaneously reversible, it can cause palpitations, dizziness, syncope, and sudden death [1, 13].

Risks of Different Antipsychotics

All antipsychotics can increase the QT interval. Thioridazine, pimozide, ziprasidone, droperidol, and intravenous haloperidol pose the greatest risks [19, 21, 23–25]. Increases in QTc duration with thioridazine and ziprasidone are approximately 25 and 10 ms more than other antipsychotics (Fig. 12.2). Nearly all cases of clinically significant intravenous haloperidol-related QTc prolongation occur in intensive care settings, with concomitant risk factors and with cumulative doses more than 2 mg. Such effects with haloperidol may be seen for up to 8 h after the last dose [16, 26, 27].

The Clinical Antipsychotic Trials of Intervention Effectiveness (CATIE) study found that 3 % of patients receiving risperidone and quetiapine developed QTc prolongation [28]. In a study of 61 patients treated with clozapine, QTc duration over 500 ms was seen in two patients and occurred in a dose-dependent fashion [29]. Paliperidone may cause minor prolongation in QTc, approximately 7 and 12 ms more than placebo at doses of 4 and 8 mg/day, respectively [30–32]. Iloperidone can increase QTc approximately 9 ms at doses of 8 and 12 mg twice/day and 15 ms at a dose of 24 mg/day [33]. Asenapine and lurasidone may cause smaller increases in QTc, approximately 2–5 ms

for asenapine (5–20 mg/day) and 5 ms for lurasidone (40, 120 mg/day) [34, 35]. Aripipazole generally has even less of an effect on QTc. Hence, all these agents are associated with a lower QT prolongation than ziprasidone.

Guidelines for Prevention and Treatment

Given the risk of sudden death from antipsychotic medications, physicians should screen for risk factors associated with QTc prolongation, TdP, and sudden cardiac death before prescribing antipsychotics (see Table 12.1). Inquiry should be made about a personal or family (e.g., long QT syndrome) history of cardiac disease, recurrent syncope, or sudden cardiac death [1, 2, 17]. An ECG should be obtained at baseline and after initiating treatment if a patient has any of these risk factors. Combining antipsychotics with medications which also prolong the QTc interval or inhibit cytochrome P450 enzymes should be avoided in these patients (Tables 12.2 and 12.3) [1, 22].

Once treatment is initiated, an ECG should be repeated in patients who develop syncope or dizziness, especially in the context of diarrhea, emesis, electrolyte imbalance, antidiuretic medications, or concurrent medications that prolong QTc or inhibit specific cytochrome P450 enzymes [1, 17, 36]. If QTc prolongation or cardiac arrhythmia develop, dose reduction or switching to lower risk medications and consultation with a cardiologist are appropriate.

Movement Disorders

Definition, Significance, Patient Risk Factors, and Mechanisms

Antipsychotics cause a variety of movement disorders or extrapyramidal symptoms (EPS) which negatively influence quality of life in patients with schizophrenia [37]. There are four major movement disorders associated with antipsychotic use: acute dystonia, akathisia, parkinsonism, and tardive dyskinesia (TD).

Acute dystonia is characterized by sustained contraction of voluntary muscles leading to abnormal postures and pain. The contraction may last from minutes to hours [38]. The most commonly affected site is the head and neck area, although any body part can be involved. Presentations include oculogyric crisis, laryngeal dystonia, blepharospasm, trismus, and torticollis [39]. Ninety-five percent of all cases appear within the first 4 days of starting or increasing the dose of antipsychotic [40]. The prevalence is widely variable, ranging from 2.3 to 94 % [39]. Patient risk factors include younger age, male sex, use of cocaine, and a previous history of acute dystonia [40].

Akathisia consists of a subjective and an objective component. Subjectively, patients complain of inner restlessness. Objectively, akathisia manifests as increased

Table 12.4 Differential diagnoses of akathisia [41, 207, 251]	Agitation related to psychotic symptoms or mood symptoms Anxiety
	Restless legs syndrome
	Tardive dyskinesia
	Antipsychotic dysphoria
	Drug withdrawal syndromes (e.g., opiates, cannabis)
	Neurological disorders (e.g., Parkinson's disease, Huntington's disease)
	Organic disorders (e.g., delirium, head injury, hypoglycemia, encephalitis)

motor activity described as complex, semipurposeful, repetitive movements reflecting their urge to move [39, 41]. For example, patients may constantly pace up and down, rock from foot to foot, frequently shift their body position in the chair, or repeatedly stand and sit. Akathisia is divided into subtypes depending on timing of onset. *Acute akathisia* develops soon after starting or increasing the dose of antipsychotic. It typically appears within the first 2 weeks and 90 % of cases occur within the first 10 weeks of initiating therapy [39]. *Tardive akathisia* develops after longer term use of an antipsychotic. *Withdrawal akathisia* refers to symptoms which occur after discontinuation or dose decrease usually within 6 weeks. The diagnosis of akathisia may be difficult to make because it can be caused by a number of disorders (Table 12.4). The incidence of akathisia varies from 21 to 75 % and its prevalence from 20 to 35 % [41]. There appear to be no patient specific risk factors for akathisia.

Parkinsonism consists of bradykinesia, rigidity, tremor, and postural instability. It is characterized by slowness and interruption of the normal flow of movement [38, 39]. Features include decreased facial expression, decreased arm swing, reduced volume and articulation of speech, and difficulty initiating movement. *Rigidity* is characterized by an increase in resting muscle tone and manifests as cogwheel and lead-pipe rigidity. *Tremor* appears at rest and resolves with movement. The tremor has a low frequency, wide amplitude, and typically affects the hands. *Postural instability* refers to impairment of righting reflexes and manifests as swaying backwards when rising from a chair, standing, or turning. Parkinsonism usually develops within the first 72 h of treatment with an antipsychotic and usually resolves after discontinuation of the offending drug. Its natural course is unknown if the drug is continued [39]. The prevalence of antipsychotic-induced parkinsonism during treatment with first generation antipsychotics (FGAs) varies between 50 and 75 % [42]. Risk factors include older age, female sex, history of antipsychotic-induced EPS, and Alzheimer's disease [42].

Tardive dyskinesia (TD) is characterized by involuntary writhing or purposeless, irregular movements. The most common type is *orofacial* and involves the oralbuccal areas. It consists of involuntary movements of the tongue, jaw, lips or face, such as tongue protrusion, lip smacking, and chewing [43]. Other less common presentations include *limb-truncal* movements, such as writhing movements of the fingers;

Table 12.5 Differential diagnoses of TD [43]	Ill-fitting dentures or dental problems Stereotypies and mannerisms in schizophrenia Tics
	Akathisia Neurological disorders (e.g., Wilson's disease, Huntington's disease, Syndenham's chorea)

head nodding and pelvis thrusting; and *respiratory* movements of the muscles and/ or the diaphragm, such as fast irregular breathing pattern and dyskinetic movement of the abdomen. TD develops after months to years of treatment with an antipsychotic. It is aggravated by anxiety and volitional motor activity and disappears during sleep. TD is potentially irreversible. The mortality rates in patients with TD are approximately 2.5 times higher than patients without TD [44]. Differential diagnosis of TD includes a number of disorders (Table 12.5). The annualized incidence of TD is 3.9 % for second generation antipsychotics (SGAs) and 5.5 % for FGAs. Prevalence is 13.1 % for SGAs and 32.4 % for FGAs [45]. The incidence increases linearly at about 5 % per year for the first 5 years of FGA exposure [46, 47].

The main risk factors are older age, history of antipsychotic-induced EPS, and nonwhite race [48]. Other reported risk factors include brain damage or dementia, major affective disorder, treatment with anticholinergics, LAI antipsychotic formulations, prolactin-related sexual disturbances, and substance abuse [47–51]. There is evidence both for and against female sex, diabetes mellitus, antipsychotic dose, positive symptoms, and negative symptoms acting as additional risk factors [47–49, 51, 52].

Common Mechanisms

Efforts to explain EPS focus on the observed lower incidence with SGAs. One hypothesis is the high ratio of serotonergic $(5-HT_{2A})$ to dopaminergic (D_2) blockade. Since serotonin inhibits dopamine transmission, blocking serotonin receptors might increase dopamine release to an extent that some of the D_2 blockade is reversed, thereby reducing EPS [53]. SGAs block $5-HT_{2A}$ in addition to D_2 receptors. Neuroimaging studies show that clinical response occurs with D_2 occupancy rates around 65 % and EPS with occupancy rates around 78 %, indicating that, for a given antipsychotic, dose is important [54]. Other models include the "fast dissociation hypothesis" which postulates that SGAs bind to the D_2 receptor long enough to cause antipsychotic effects, but not long enough to cause EPS [55].

The pathophysiology of TD is uncertain, but two hypotheses have held sway for many years. The "dopamine supersensitivity hypothesis" states that long-term D_2 receptor blockade in the striatal area induces an up-regulation of receptor activity which causes dyskinetic symptoms [43, 53]. The "neurotoxicity hypothesis" states that antipsychotics increase dopamine metabolism, thereby releasing neurotoxic free radicals leading to damage of its receptors [53].

Risks of Different Antipsychotics

High potency FGAs (e.g. haloperidol) induce more EPS than low potency FGAs (e.g. chlorpromazine). SGAs induce fewer EPS than high potency FGAs, even when the latter are prescribed at low doses. Some SGAs (clozapine, olanzapine, and risperidone) have fewer EPS than low potency FGAs [56]. Among the SGAs, risperidone induces more EPS; quetiapine induces fewer EPS; and olanzapine induces fewer EPS than ziprasidone and aripiprazole [57].

Guidelines for Prevention and Treatment

Management of antipsychotic-induced EPS depends on their type and severity. In patients at high risk for **acute dystonia** (e.g., young males), prophylaxis with oral anticholinergic medications such as benztropine (1 mg two or three times daily) or trihexyphenidyl (5 mg one to two times daily) for the first week of treatment can reduce the incidence [39]. Anticholinergics are the treatment of choice for acute dystonia and are given orally or parenterally depending on the setting and severity of symptoms. Parenteral treatment such as diphenhydramine (50 mg) or benztropine (1–2 mg) can be repeated in 30 min if response is not sufficient. Other diagnoses (e.g., catatonia, tardive dystonia, or hypocalcemia) should be considered if the dystonia does not respond to repeated injections. Parenteral treatment is typically followed by oral anticholinergics for at least a few days to prevent recurrence.

Management of **acute akathisia** includes switching or decreasing the dose of antipsychotic. Pharmacological treatment for akathisia includes propranolol (starting at 10 mg three times daily and increasing every few days to a maximum of 90–120 mg/day) [41]. Blood pressure and heart rate should be closely monitored during dose titration. Limited evidence shows that benzodiazepines are also useful for treatment of akathisia [58]. There are a few double-blind, randomized controlled trials supporting the use of anticholinergics for akathisia. The quality of these trials is limited, however, due to their small sample sizes and concomitant benzodiazepine treatment [59–62]. Several agents, including mirtazapine, mianserin, cyprohepadine, zolmitriptan, trazodone, clonidine, and vitamin B6 are reported to be helpful in small, double-blind trials (n < 30) [63–75]. Presently the most convincing evidence is for mirtazapine (15 mg/day).

In patients at high risk for antipsychotic-induced or preexisting **parkinsonism**, starting or switching to low potency FGAs or SGAs is appropriate when clinically feasible. The use of anticholinergics to prevent EPS is controversial. Anticholinergic prophylaxis may benefit certain high-risk patients such as younger males receiving treatment with high potency antipsychotics and a previous history of EPS. Generally, however, they are not beneficial for routine prophylactic use in most patients [76, 77]. There are several strategies to manage antipsychotic-induced parkinsonism. These include dose reduction, switching to a medication with lower EPS risk or adding antiparkinsonian agents. Dose reduction and medication switching should be considered before using an antiparkinsonian drug, because these medications have their own adverse effects. The most common medications used to treat

antipsychotic-induced parkinsonism are anticholinergics such as benztropine (1–6 mg/day) or trihexyphenidyl (5–15 mg/day). These agents, however, can cause a number of adverse effects including cognitive impairment, delirium, blurred vision, dry mouth, tachycardia, urinary retention, and constipation. There is also a potential for anticholinergic abuse, a worsening of TD, and drug-induced psychosis [42, 78]. These effects are particularly concerning in the elderly. If prescribed for this age group, their use should be reviewed every 3 months and slowly tapered (to avoid cholinergic rebound and re-emergence of EPS) when patients are psychiatrically stable and have undetectable EPS [77, 78]. Limited evidence shows that amantadine (100–400 mg/day) is comparable to anticholinergic medications but has fewer cognitive and peripheral adverse effects. It may, however, carry a greater risk for aggravating psychosis [79–81].

To minimize the risk of tardive dyskinesia, antipsychotics should be used at the lowest effective dose and discontinued when clinically appropriate. Although there are more than 500 randomized controlled trials evaluating over 90 different interventions to treat TD, they have not produced sufficient evidence to recommend a definite treatment [82]. If discontinuation of antipsychotics is possible, clinicians should be aware that TD may worsen initially, with the severity returning to the level before discontinuation after 6-12 weeks [43]. Discontinuation can lead to complete resolution of TD, particularly when the duration of TD is short and patients are younger than 50-60 years [43]. There are, however, no randomized controlled trials evaluating antipsychotic cessation to treat TD [83]. If discontinuation is not possible, other treatment options include dose reduction, switching to an alternate antipsychotic, and adding other agents. There are two small, double-blind, randomized controlled trials evaluating the effects of 50 and 90 % reduction in dose compared with the continuation of the current dose [83-85]. The results indicated that dose reduction was not associated with either improvement or worsening of TD [83–85]. Switching to clozapine or quetiapine may be effective [43]. Small trials demonstrate that tetrabenazine diminishes TD; however, its use is limited by cost and adverse effects such as depression, parkinsonism, and somnolence [86]. Vitamin E may protect against further deterioration but does not appear to improve symptoms of TD [87]. There is no compelling evidence supporting cholinergic medications, benzodiazepine, or calcium channel blockers for TD or for adding or withdrawing anticholinergic medications [88–91]. Finally, bilateral deep brain stimulation of the globus pallidus may be effective and relatively safe for patients with severe, disabling, treatment-resistant TD [92-94].

Hyperprolactinemia and Sexual Dysfunction

These two adverse effects are considered under a single heading because hyperprolactinemia is one of the major contributors to sexual dysfunction. It is also associated with other adverse effects, while other pharmacologic effects may also contribute to sexual dysfunction.

Hyperprolactinemia

Definition, Significance, Patient Risk Factors, and Mechanisms

Hyperprolactinemia is a common but frequently undetected adverse effect of antipsychotics. While normal serum prolactin levels vary between different laboratories, they are usually below 20 for men and 25 μ g/L for women [95]. Higher levels are considered diagnostic, though during pregnancy and breast-feeding the levels can increase 10–20 times above the nonpregnant values [96, 97]. Prolactin is secreted by the lactotroph cells in the pituitary and its main physiological function is to induce milk production in mammary glands. Dopamine, via its tuberoinfundibular pathway, is the predominant inhibitory factor for prolactin secretion. Blockade of D₂ receptors on lactotroph cells removes the inhibitory effect leading to hyperprolactinemia. Antipsychotic-induced hyperprolactinemia is often, but not always, dose-related [98, 99]. Levels begin to increase a few hours after initiating antipsychotics, persist throughout treatment and, following discontinuation, return to normal range within 2–3 weeks after oral treatment and 6 months after intramuscular treatment [100, 101].

The prevalence of antipsychotic-induced hyperprolactinemia among psychiatric patients is 42 % for men and 66 % for women [102]. Direct action of prolactin causes gynecomastia in men and galactorrhea in both sexes. Gynecomastia, however, is relatively rare, occurring in 2 % of male patients [103]. High prolactin levels inhibit the hypothalamic-pituitary-gonadal axis at several levels resulting in a reduction of gonadal hormone levels (estrogen in women and testosterone in men). Symptoms of secondary hypogonadism include sexual dysfunction, menstrual abnormalities, and infertility. Forty to 50 percent of female patients with hyperprolactinemia experience menstrual abnormalities with an association between increasing prolactin level and increasing risk [102]. Long-term hypogonadism can lead to decreased bone mineral density (BMD). In one study of premenopausal women with schizophrenia who took antipsychotics for approximately 8 years, hyperprolactinemia was associated with low BMD (i.e., 62 % of those with hyperprolactinemia had a low BMD versus only 11 % of those without hyperprolactinemia) [104]. The relationship between antipsychotic use and increased risk of breast cancer is controversial. A recent well-designed study, however, concluded that women receiving prolactin-raising antipsychotics have a modest but significantly increased risk [96, 105].

Patient risk factors for hyperprolactinemia include female sex, the postpartum period, and younger age [96, 98, 102, 106]. In addition, renal and hepatic failure, hypothyroidism and pituitary tumors can increase this risk (Table 12.6).

Risks of Different Antipsychotics

FGAs and some SGAs (e.g., risperidone and paliperidone) significantly elevate prolactin levels [31, 107]. Other SGAs including clozapine, olanzapine, quetiapine,

Physiological conditions	Medications
Pregnancy	Antidepressants (TCAs, MAOIs, SSRIs)
Lactation	Antihypertensive (verapamil, methyldopa, reserpine)
Sleep	Antiemetics (metoclopramide, domperidone)
Stress	H2 receptor blockers (ranitidine, cimetidine)
Exercise	Hormones (estrogens, oral contraceptives, protirelin, antiandrogens)
Sexual activity	Others (opiates, cocaine, protease inhibitors)
Breast stimulation	
Pituitary diseases	Endocrine diseases
Prolactinoma	Cushing's disease
Adenomas	Polycystic ovarian disease
Empty sella syndrome	Acromegaly
Pituitary stalk section	Primary hypothyroidism
Lymphoid hypophysitis	
Hypothalamic diseases	Miscellaneous
Hypothalamic tumors	Cirrhosis
Hypothalamic sarcoidosis	Chronic renal failure
Postencephalitis	Chest wall lesions (trauma, tumors, herpes zoster)
	Ectopic production of prolactin (small-cell bronchial carcinoma)

Table 12.6 Differential diagnoses of hyperprolactinemia [96, 97, 100, 106, 107, 252]

ziprasidone, and aripiprazole are rarely implicated [107]. A recent review indicates that aripiprazole can lower prolactin levels an average of 74.3 %, even in psychotic patients with prolactinoma [108].

Guidelines for Prevention and Treatment

Screening for hyperprolactinemia is not routinely done [100]. A serum prolactin level should be obtained, however, if a patient presents with related symptoms. A single measurement of serum prolactin at any time of the day without excessive venipuncture stress is sufficient to establish the diagnosis. A level above the upper limit of normal (e.g., $20 \ \mu g/L$ for men and $25 \ \mu g/L$ for women; $1 \ \mu g/L = 21.2 \ mIU/L$) confirms the diagnosis [95, 97]. A prolactin level above 250 $\mu g/L$ usually indicates a prolactinoma and above 500 $\mu g/L$ indicates a macroprolactinoma, though risperidone and some phenothiazines are linked to elevations above 200 $\mu g/L$ in patients without evidence of adenoma [97]. If highly elevated prolactin levels do not return to normal following discontinuation, magnetic resonance imaging (MRI), or computed tomography (CT) of the hypothalamic-pituitary area should be obtained to exclude a mass lesion [107].

The first step in treatment of antipsychotic-induced hyperprolactinemia is to determine if the patient is significantly symptomatic. Reassurance may be sufficient for a woman with normal menses who has only some non-bothersome galactorrhea, whereas more active treatment is required when significant symptoms (e.g., bothersome galactorrhea, decreased libido, amenorrhea, and osteoporosis) occur [97, 107]. If a patient is symptomatic and requires continuation of an antipsychotic,

switching to aripiprazole can lower prolactin levels and resolve symptoms of antipsychotic-induced hyperprolactinemia [108]. Switching to another prolactinsparing agent is also effective [96, 97, 107]. In the unlikely event that none of these are clinically effective, adjunctive treatment with a D_2 receptor agonist such as bromocriptine (5.0–7.5 mg/day) or cabergoline (0.125–0.250 mg/day) may help [96, 107]. Bromocriptine, however, can cause a psychotic relapse, postural hypotension, gastrointestinal symptoms, and vasospasm of the fingers and toes. A recent meta-analysis found that cabergoline was superior in normalizing prolactin levels and restoring gonadal function; better tolerated with fewer adverse events compared with bromocriptine; and effective in patients resistant to bromocriptine [97, 109]. If estrogen or testosterone levels are decreased, these hormones can be supplemented. With osteoporosis, a bisphosphonate should be considered [107].

Sexual Dysfunction

Definition, Significance, Patient Risk Factors, and Mechanisms

Sexual dysfunction is a distressing adverse effect, not surprisingly associated with poorer adherence [110]. A large, international study found that sexual dysfunction affected approximately 50 % of both male and female antipsychotic-medicated patients [111]. Another study reported that the rates of sexual dysfunction in an outpatient schizophrenia clinic resembled those in a specialized sexual dysfunction clinic [112]. Women appear 15 times more likely and men 4 times more likely to complain of sexual dysfunction compared with gender-matched healthy controls [112]. Despite the high prevalence, sexual dysfunction is typically discounted by psychiatrists [111, 113].

Other factors which contribute to sexual dysfunction in patients with schizophrenia include their psychopathology; social sequelae of the illness; medical illnesses; and alcohol abuse [96]. Antipsychotic dose is correlated with sexual dysfunction, while other risk factors include low potency FGAs, polypharmacy, severity of illness, and depot FGAs in women [113, 114].

Antipsychotics may affect sexual function via dopaminergic, histaminergic, cholinergic, and α -adrenergic receptor actions leading to diminished motivation and reward, increased sedation, and reduced performance [96]. The role of prolactin in sexual dysfunction is not entirely clear, though its incidence largely parallels the classification into prolactin-raising and prolactin-sparing antipsychotics [112, 115, 116].

Risks of Different Antipsychotics

Quetiapine, ziprasidone, perphenazine, and aripiprazole are associated with relatively low total sexual dysfunction rates (16–27 %), whereas olanzapine, risperidone, haloperidol, clozapine, and thioridazine are associated with relatively high rates (40–60 %) [117]. Aripiprazole is associated with the lowest rates for each

aspect of sexual dysfunction. The liability of antipsychotics for phase-specific sexual dysfunction, including *desire*, *arousal*, *and orgasmic dysfunction* are about the same as for total sexual dysfunction [117]. More variability is demonstrated by clozapine, however, which is associated with relatively low rates of arousal and orgasmic dysfunction, and haloperidol which is associated with relatively low rates of orgasmic dysfunction [117].

Guidelines for Prevention and Treatment

Management of antipsychotic-induced sexual dysfunction includes decreasing the dose, switching to a medication with lower risk, and adding medication to specifically target these symptoms. Dose reduction should be beneficial, but has not been clearly established. Switching to aripiprazole is the most studied strategy [118]. Aripiprazole can improve sexual function and lower prolactin levels compared with olanzapine, quetiapine, or risperidone [119]. A randomized, placebo-controlled trial in male patients with antipsychotic-induced erectile dysfunction reported that sildenafil (25 or 50 mg/day) significantly improved the number of adequate erections, satisfactory sexual intercourse, and the duration of erections [120]. Another study of sildenafil for treatment of psychotropic-induced sexual dysfunction in psychiatric patients reported significant improvements in all domains for both men and women [121].

Metabolic Syndrome

The metabolic syndrome is discussed in greater detail in Chap. 10. We focus primarily on the contribution of antipsychotic medications.

Definition, Significance, Patient Risk Factors, and Mechanisms

The metabolic syndrome is a cluster of risk factors which increase the likelihood of developing type 2 diabetes mellitus and cardiovascular disease (Table 12.7). As a result, the average life expectancy for patients with schizophrenia is 10–25 years shorter than the general population. Overall, the prevalence of and the mortality from cardiovascular disease is two- to three-fold higher for patients with schizophrenia than the general population [122, 123].

In addition to generic disease-based factors, personal and family history of obesity, diabetes, dyslipidemia, hypertension, or cardiovascular disease contribute to the risk of developing or aggravating preexisting metabolic syndrome. Some patient risk factors, however, appear counter-intuitive. Thus, a significant predictor of antipsychotic-induced weight gain appears to be lower baseline body mass index

Risk factor	IDF (central obesity plus ≥ 2 other factors)	ATP III (≥ 3 factors)
Waist circumference (cm)		
Men	≥94 for Europids	>102 for Europids
	≥90 for South Asians and Chinese	
Women	≥80 for Europids	>88 for Europids
	≥80 for South Asians and Chinese	
Triglycerides (mg/dL)	≥150	≥150
HDL cholesterol (mg/dL)		
Men	<40	<40
Women	<50	<50
Blood pressure (mmHg)	≥130/85	≥130/85
Fasting plasma glucose (mg/dL)	≥100	≥100 ^a

 Table 12.7
 Metabolic syndrome criteria [253–255]

IDF International Diabetes Federation, *ATP* adult treatment protocol

^aOriginally ≥ 110

(BMI) [124]. Other predictors include younger age, nonwhite ethnic background, tendency to overeat in time of stress, cannabis use, and first episode psychosis [125].

Antipsychotics increase the risk of the metabolic syndrome through various mechanisms [125, 126]. They cause weight gain via activity at histamine H₁, serotonin 5-HT_{2C}, serotonin 5-HT_{2A}, muscarinic M₃, and adrenergic receptors [127]. They alter insulin and glucose metabolism by decreasing pancreatic *B*-cell responsiveness via effects on the 5-HT_{1A} receptor and suppressing skeletal muscle glucose uptake via 5-HT_{2A} receptor effects [128]. In addition, some antipsychotics may act on adipocytes to alter insulin action, lipogenesis, and lipolysis leading to lipid accumulation [129].

Risks of Different Antipsychotics

The incidence of metabolic syndrome and the individual contributory risk factors vary markedly across agents. Specifically, the risk of the metabolic syndrome is highest with clozapine and olanzapine; moderate with quetiapine, risperidone paliperidone, iloperidone, and asenapine; and low with aripiprazole, ziprasidone, lurasidone, haloperidol, and perphenazine [130, 131]. Related adverse effects are commonly seen in the first 3 months of exposure to olanzapine [132]. The development of metabolic risk factors is predicted by clozapine and olanzapine plasma concentrations [133]. Patients, however, may present early in the course of treatment with life-threatening diabetic ketoacidosis (DKA) even before other symptoms such as weight gain occur (Chap. 10) [134].

Weight gain is a major risk factor and occurs most with clozapine and olanzapine and least with aripiprazole and ziprasidone [124, 130, 135, 136]. Weight gain appears to be dose-related with clozapine and olanzapine, averaging about 1.5 lbs per month, particularly in the first months of treatment. The time period over which

weight continues to accumulate varies among drugs. For example, weight gain continues beyond the first year with clozapine but stabilizes at about 9 months with olanzapine and haloperidol [124].

Clozapine and olanzapine are associated with the highest risk of **insulin resistance**, whereas ziprasidone and amisulpride are associated with the lowest risk [28, 137–140]. Quetiapine, risperidone, haloperidol, and perphenazine also have negative effects on glucose regulation [28, 137, 138]. Changes in glucose regulation are usually seen within 6 weeks of initiating treatment with antipsychotics [138]. An average change in blood glucose of about 14 mg/dL and glycosylated hemoglobin of about 0.40 % occurs in patients treated with olanzapine within 18 months [28]. In contrast, ziprasidone and amisulpride can improve glycemic parameters, including glycosylated hemoglobin and fasting blood glucose over 6 months [28, 139, 140].

The risk of antipsychotic-induced **dyslipidemia** is greatest with olanzapine and least with ziprasidone and aripiprazole [28, 137, 140–147]. Clozapine, quetiapine, haloperidol, and perphenazine may induce dyslipidemia, whereas an association between risperidone and dyslipidemia is controversial [28, 137, 143, 148]. Dyslipidemia associated with olanzapine tends to be dose-related [149]. Olanzapine can increase total cholesterol an average of 10–30 mg/dL and triglycerides an average of approximately 30 mg/dL within 8 weeks of initiating treatment [137, 141, 149]. In this context, ziprasidone and aripiprazole can improve lipid parameters, including total cholesterol, triglycerides, low-density lipoprotein, and high-density lipoprotein [28, 140, 142, 145–147].

Guidelines for Prevention and Treatment

Given the serious health risks associated with the metabolic syndrome, patients taking antipsychotics should receive baseline screening and ongoing monitoring [150]. Consensus guidelines developed jointly by the American Diabetes Association, American Psychiatric Association, American Association of Clinical Endocrinologists, and North American Association for the Study of Obesity recommend screening for and monitoring of metabolic risk factors as follows:

- Personal and family history of obesity, diabetes, dyslipidemia, hypertension, or cardiovascular disease at baseline
- Weight and height (BMI for adults and BMI percentile for children and adolescents) at baseline, every 4 weeks during the first 12 weeks and quarterly thereafter
- Waist circumference (at the level of the umbilicus) at baseline and then annually
- Blood pressure at baseline, at 12 weeks and then annually
- Fasting plasma glucose at baseline, at 12 weeks and then annually
- Fasting lipid profile at baseline, at 12 weeks and then every 5 years

These guidelines also emphasize educating patients, family members, and caregivers about the signs and symptoms of diabetes and DKA, and referring patients with metabolic dysregulation to appropriate healthcare professionals [150]. Management of patients who gain weight or develop metabolic dysregulation during treatment with antipsychotic medications includes non-pharmacological and pharmacological interventions. *Non-pharmacological interventions* are effective in preventing and treating antipsychotic-induced weight gain and other metabolic risk factors for metabolic syndrome [151, 152]. The interventions include dietary counseling, increasing energy-consuming activities, behavioral therapy, and smoking cessation [153–156]. *Pharmacological interventions* may include switching to another agent with a lower risk or which can reverse weight gain and metabolic dysregulation [136, 150]. There is strong evidence for switching to aripiprazole and ziprasidone, while adding aripiprazole to clozapine also reverses weight gain [145, 157–163]. Current best evidence indicates metformin at doses of 750 and 1,000 mg in divided doses for 12 weeks to prevent and reverse antipsychotic-induced weight gain and insulin resistance (see Chap. 10 for additional information about management) [164–168].

Advanced Age

While the risk of most adverse effects reviewed above increases with age, there are particular concerns for the elderly, especially those with dementia. Interestingly, these risks do not seem to vary as much across different antipsychotics for elderly patients with and without schizophrenia and hence risks of different medications are integrated into the first section below. In some cases, risks are clearly elevated in those with dementia versus age-matched controls, but this question remains unresolved for many adverse effects.

Definition, Significance, Patient Risk Factors, and Mechanisms

An older (or elderly) person is defined as a chronological age of 65 years or more, particularly in Western countries [169]. Antipsychotics are commonly used to treat both behavioral and psychological symptoms in elderly patients, particularly those with dementia. In the USA, an average of 620,000 elderly patients receive an antipsychotic annually [170]. Frequently reported diagnoses among those patients are dementia (26 %), anxiety (20 %), and schizophrenia (7 %) [170].

Antipsychotics increase the risk of *death* (1.2–1.6 times) in elderly patients with dementia versus those without dementia [171]. The risk is similar when comparing FGAs and SGAs, and no specific medication is proven safer [171]. Patient risk factors for death include older age, male sex, severity of dementia, and functional impairment [171]. Recent meta-analyses reported that the risk of *cerebrovascular events* is 1.3–2 times higher in those with dementia and the risk is similar for FGAs and SGAs, particularly at higher doses. Risk factors include older age, vascular

dementia, atrial fibrillation, hypertension, a history of previous stroke, and concurrent use of anticoagulants [171, 172]. As noted above, the elderly receiving antipsychotics are at increased risk for *cardiac arrhythmias* and the association with dementia is not known [17]. Antipsychotics increase the risk of *falls and hip fractures* in the elderly [173–176]. Evidence about the differential risk between FGAs and SGAs is inconclusive [177]. The risk increases within 1 week after exposure to FGAs or SGAs but stabilizes at a higher level with exposure beyond 12 weeks [176]. An antipsychotic increases the risks of both hospital- and community-acquired *pneumonia* in the elderly with or without dementia and is highest during the first week [176, 178–180]. Data comparing the liability between FGAs and SGAs are conflictual [176, 178–180]. A few studies also report an association between antipsychotics and an increased risk of *venous thromboembolism* (VTE) in elder patients either with or without dementia; however, a recent study could not replicate the findings [181–183].

The Clinical Antipsychotic Trials of Intervention Effectiveness-Alzheimer's Disease (CATIE-AD) study found that SGAs had an adverse effect on weight, but surprisingly no effect on glucose levels, total cholesterol, or triglyceride in patients with schizophrenia and dementia [184]. Other studies confirm no association between antipsychotics and diabetes in the elderly [185–188]. Of note, the risk of developing antipsychotic-induced movement disorders in patients with dementia is similar for FGAs and SGAs [189, 190].

Mechanisms regarding increased sensitivity to antipsychotics in the elderly are not fully understood. They may involve age-related changes in peripheral and central pharmacokinetics and pharmacodynamics. Specifically, plasma levels in elderly patients are higher for a given dose; permeability of the blood–brain barrier increases with aging; and age-related decreases in endogenous dopamine levels and receptors result in higher susceptibility to clinical/adverse effects for a given level of occupancy [191].

Guidelines for Prevention and Treatment

Given the increased mortality and morbidity among elderly patients treated with antipsychotics, clinicians should consider non-pharmacological approaches (e.g., by modifying the environment, improving communication, and optimizing social contacts) before prescribing these agents [192]. Antipsychotics should be especially avoided in elderly patients with risk factors for cerebrovascular adverse effects including vascular dementia, hypertension, atrial fibrillation, or a history of stroke [171, 172]. If necessary, these medications should be started at a low dose, increased slowly, and maintained at the minimally effective dose [193]. The need for continued medications should be reviewed regularly to avoid unnecessary exposure and patients should be monitored regularly for hypotension, sedation, and EPS [193].

Clozapine

Clozapine is a very effective antipsychotic, frequently the treatment of choice for patients who are treatment-resistant or suicidal [194]. It is, however, associated with a number of serious adverse effects (i.e., the United States Food and Drug Administration (FDA) insert includes five boxed warnings). We review this agent's serious adverse effects, more common but less serious adverse effects and their management by organ system. In addition, we review evidence about the safety of rechallenge after an adverse effect has occurred.

Hematologic Adverse Effects

Clozapine has a range of adverse effects on blood components, particularly neutropenia and agranulocytosis [195]. Neutropenia is defined as a neutrophil count (ANC) less than 1,500/mm³ or a white blood cell (WBC) count less than 3,000/ mm³. Agranulocytosis is defined as ANC less than 500/mm³ or WBC less than 1,000 mm³. The risk of developing neutropenia and agranulocytosis from clozapine is 3 % and 0.8 %, respectively. The mortality rate due to clozapine-induced agranulocytosis is 3-4% of identified cases under the blood monitoring system with clozapine [196]. The risk of developing agranulocytosis is higher during the first 18 weeks of treatment and decreases thereafter (Table 12.8). Guidelines require the WBC to be monitored weekly for the first 18 weeks in the UK (6 months in the USA), then every other week to the end of the first year and every 4 weeks thereafter. The blood monitoring system is highly effective in reducing both the incidence of clozapine-related agranulocytosis and its associated mortality [197]. Agranulocytosis related to clozapine is an idiosyncratic (type B) reaction which is not dose-related [198]. Patient risk factors include older age, female sex, and its combination with other medications known to cause neutropenia or agranulocytosis (e.g., carbamazepine). If a patient develops an abnormal WBC or ANC during treatment, the recommendation according to the US clozapine monitoring system is shown in Table 12.9.

A recent review (n=112) reported that 70 % of patients who developed neutropenia were successfully rechallenged with clozapine. Of the remaining 30 %, 44 % of those patients went on to develop agranulocytosis during their second exposure

Duration of clozapine treatment (weeks)	Incidence (per 1,000 patient-years)
0–18	7.77
18–52	0.83
>52	0.37

 Table 12.8
 Incidence of clozapine-induced agranulocytosis [196]

Values	Actions
WBC \geq 3,500/mm ³ and ANC \geq 2,000/mm ³	Initiate therapy
WBC<3,500/mm ³ and/or ANC<2,000/mm ³	Monitor twice weekly until WBC>3,500/mm ³ and ANC>2,000/mm ³
WBC < 3,000/mm ³ and/or	1. Temporarily discontinue therapy
ANC < 1,500/mm ³	2. Monitor daily until WBC>3,000/mm ³ and ANC>1,500/mm ³
	3. After that monitor twice weekly until WBC>3,500/mm ³ and ANC>2,000/mm ³ and then rechallenge may be considered
	4. If rechallenged, monitor weekly for 1 year before returning to the usual monitoring schedule of 2 weeks for 6 months and then every 4 weeks thereafter
WBC<2,000/mm ³ and/or	1. Permanently discontinue therapy
ANC < 1,000/mm ³	2. Monitor until normal and for at least 4 weeks from day of discontinuation as follows
	 Daily until WBC>3,000/mm³ and ANC>1,500/mm³
	- Twice weekly until WBC>3,500/mm ³ and ANC>2,000/mm ³
	 Weekly after WBC>3,500/mm³

Table 12.9 Clozapine monitoring guideline [256]

to clozapine [199]. Of the 15 patients who developed agranulocytosis, 80 % failed rechallenge [199]. Therefore, clozapine rechallenge should not be considered in patients with previous agranulocytosis, and it should be undertaken carefully in patients with neutropenia only when other treatment options are not viable [194, 195, 199, 200].

Lithium, filgrastim, granulocyte colony-stimulating factor (G-CSF), and granulocyte-macrophage colony-stimulating factor (GM-CSF) may be useful in managing neutropenia and agranulocystosis [201]. Lithium can increase WBC counts approximately 2.0×10^{9} /L with a serum concentration of at least 0.4 mmol/L [202]. Lithium may also increase WBC counts for both patients with low baseline counts and those who develop neutropenia after treated with clozapine [195]. In addition, GM-CSF and G-CSF may shorten the duration of clozapine-induced agranulocytosis [202].

Cardiovascular Adverse Effects

Common cardiovascular effects include tachycardia (25 %), postural hypotension (9 %), and hypertension (4 %) [194]. Tachycardia is attributable to the anticholinergic activity of clozapine, elevated plasma norepinephrine levels, and hypotension [197]. Tachycardia may be transient and related to rapidity of titration [197]. Reducing the dose or adding a beta-blocker can help if it persists [194]. Postural hypotension is related to the alpha-adrenergic antagonistic properties of clozapine, usually transient and occurs at the beginning of treatment [197]. It is also influenced by the rate and magnitude of clozapine dose titration. While tolerance often develops, it can persist and limit dose escalation. Isolated episodes of hypertension may also occur early in the course of treatment and are related to the rapidity of titration.

Clozapine-related myocarditis and cardiomyopathy are less common but potentially fatal. Myocarditis usually develops within 1 month of clozapine initiation and presents with fever (48 %), dyspnea (35 %), "flu-like illness" (30 %), chest pain (22 %), and fatigue (17 %) [203, 204]. The mechanism of clozapine-induced myocarditis is unclear; however, it may be related to an IgE-mediated hypersensitivity and a direct toxic effect on cardiac muscle followed by inflammatory infiltrates [205]. Factors associated with an increased risk of clozapine-induced myocarditis include high cumulative doses of clozapine (more than 920 mg) during days 1–9; concomitant use of sodium valproate; and increasing age [206]. Cardiomyopathy arises less acutely [194]. The most common type is dilated cardiomyopathy which often presents with symptoms of congestive heart failure [203].

Given that major cardiovascular adverse effects can occur within the first month, it is reasonable to perform an ECG at baseline and 2 and 4 weeks after starting treatment [203]. Clozapine should be titrated slowly and concomitant sodium valproate avoided [206]. If myocarditis is suspected, clozapine should be discontinued immediately. Clozapine rechallenge after myocarditis may be successful with a slower dose titration [206, 207].

Neurological Adverse Effects

Common neurological effects of clozapine include sedation (39 %), dizziness (19 %), and syncope (6 %) with tolerance usually developing [194]. Sedation is often mild, transient, and restricted to the initial treatment period [197]. It can be minimized by giving the majority of the dose at bedtime and avoiding concomitant central nervous system depressants. Nevertheless, since chronic sedation is an important issue, stimulants and stimulant-like drugs (e.g., dextroamphetamine, monafidil, methylphenidate, L-dopa) can potentially ameliorate it in some cases. There is limited, conflicting evidence, however, supporting their efficacy and they may worsen psychoses or induce movement disorders [208]. Two small, double-blind controlled trials reported no benefit with modanifil for treatment of clozapine-induced sedation [209, 210].

Clozapine can lower the seizure threshold and induce various types of episodes. The most common type is a generalized, tonic-clonic seizure, while myoclonic and atonic seizures occur less frequently [211]. Clozapine-induced seizures are dose-related with a 1 % risk at doses less than 300 mg/day; 2.7 % at 300–600 mg/day; and 4.4 % at more than 600 mg/day [211]. Plasma clozapine levels and screening EEGs are of limited value in predicting and diagnosing seizures [211]. Treatment for the first clozapine-induced seizure includes reducing the dose and eliminating other risk factors. If a second seizure occurs, an anticonvulsant should be initiated. Recommended anticonvulsants are valproate, lamotrigine, and gabapentin, while carbamazepine should be avoided because like clozapine it may cause bone marrow suppression [211].

Gastrointestinal Adverse Effects

The most common gastrointestinal effect of clozapine is constipation (14 %) [194]. Other less common effects include nausea, vomiting, bloating, heartburn, and abdominal pain. The mechanisms of these adverse effects may relate to the anticholinergic effects and 5-HT₃ receptors antagonism of clozapine [197, 212]. These effects can be managed with a high fiber diet, adequate hydration, and the use of bulk laxatives and stool softeners [197]. Since clozapine may induce gastrointestinal hypomotility (including dysphagia, ileus, intestinal obstruction, bowel ischemia, and megacolon) which may lead to a fatal outcome, prophylactic treatment with a stool softener is important.

Hypersalivation

Hypersalivation (31 %) is common, occurs early in treatment and is more profound during sleep [194]. Hypersalivation may result in parotitis, skin irritation, skin infection, aspiration pneumonia, and sleep disturbances. Further, it can be stigmatizing and lead to discontinuation of therapy [213, 214]. The pathophysiology of clozapine-induced hypersalivation is poorly understood with proposed mechanisms including an increase in salivary flow via adrenergic alpha₂ antagonism and muscarinic M_4 agonism, a decrease in laryngeal peristalsis or inhibition of the swallowing reflex [214]. Antimuscarinic agents are widely used to treat clozapine-induced hypersalivation but the data supporting their use is weak. Glycopyrrolate demonstrated promise in an initial study [215]. Lowering the dose and a slower dose titration may also help.

Thermoregulatory Adverse Effects

Fever occurs in 4–13 % of patients treated with clozapine [194]. This may be due to host defense mechanisms mediated by cytokines [208]. It occurs early in treatment and is usually benign (seldom rises above 38 °C or 100 °F), transient (a few days), and responds to antipyretics [194, 208]. If fever is high or persistent, other serious conditions including underlying infection, myocarditis, agranulocytosis, and NMS should be ruled out.

Other Serious Adverse Effects

The following adverse effects are relatively rare but serious and therefore are discussed separately.

Neuroleptic Malignant Syndrome

NMS is an uncommon, idiosyncratic, life-threatening adverse effect of antipsychotics characterized by fever, severe muscle rigidity, autonomic instability, and mental status changes. Although its incidence has decreased from 3 % in the past to 0.01–0.02 % at present, it can still cause significant morbidity and mortality [216]. An international consensus study and DSM-5 diagnostic features emphasize the following criteria [217, 218]:

- Recent dopamine antagonist exposure
- Hyperthermia (*T*>100.4 °F or >38.0 °C on at least two occasions measured orally)
- Generalized muscle rigidity (except clozapine)
- Mental status alterations (delirium, altered consciousness ranging from stupor to coma)
- Creatine kinase elevation (\geq 4 times the upper limit of normal)
- Autonomic nervous system instability manifested by tachycardia (rate >25 % above baseline), blood pressure elevation (systolic or diastolic ≥25 % above baseline) or fluctuation (≥20 mmHg diastolic change or ≥25 % mmHg systolic change within 24 h), and tachypnea (rate >50 % above baseline)
- · A negative work-up for other causes

NMS usually occurs 1 to 2 weeks after initiation or a dose increase of antipsychotic. Fever usually exceeds 38 °C but occasionally can exceed 41 °C [219]. The clinical profile of SGA-induced NMS is similar to FGA-induced NMS with the exception of clozapine which presents with less rigidity [220].

Possible risk factors include higher doses and more rapid rates of antipsychotic dose titration; acute IM antipsychotic exposure; psychomotor agitation; dehydration; physical restraints; iron deficiency; history of NMS; and concomitant lithium administration [216, 219]. NMS can occur with FGAs and SGAs, and any difference in liability between the two classes is not established [221]. High potency FGAs are, however, associated with a greater risk compared with low potency FGAs [222]. The pathophysiology of NMS remains unknown. Two popular hypotheses include a central hypodopaminergic state resulting from dopamine D2 receptor antagonism and sympathoadrenal hyperactivity [216].

The most important management step is discontinuation of the antipsychotic. NMS is a self-limiting condition which usually resolves within 1–2 weeks after discontinuation of the causative agent and supportive care [219]. Supportive care includes fluid replacement; fever reduction; monitoring and correction of electrolyte abnormalities; and monitoring for complications including cardio-respiratory failure, renal failure, aspiration pneumonia, and coagulopathies [216]. Pharmacological treatment of NMS includes benzodiazepines; dopaminergic agents such as bromocriptine and amantadine; and dantrolene. Lorazepam (1–2 mg IM or IV every 4–6 h) may ameliorate symptoms and reduce time to recovery in patients with NMS, particularly in those with milder and primarily catatonic symptoms [216]. Bromocriptine (2.5–5 mg orally two or three times daily) and amantadine (200–400 mg orally in divided dose)

may reverse the parkinsonism in NMS, as well as reduce time to recovery and mortality rates. Bromocriptine can, however, worsen psychosis and produce hypotension [216]. Dantrolene (1–2.5 mg/kg followed by 1 mg/kg every 6 h) may be useful in NMS presenting with extremely high fever, rigidity, and hypermetabolism. Although these medications are widely used to treat NMS, there is no definitive evidence supporting their efficacy [216, 223]. Electroconvulsive therapy (ECT) (six to ten treatments with bilateral electrode placement) is another option for NMS which is relatively safe and appears effective, even after failed pharmacological trials [224]. It can improve the underlying psychiatric disorder as well. A review of published cases reported a 10 % mortality rate in the patients with NMS who received ECT or those who received specific drug treatments (including amantadine, bromocriptine, L-dopa, and dantrolene) compared with 21 % in those who received no specific treatment [225]. ECT is considered the preferred treatment in severe NMS, for patients presenting with underlying psychotic depression or catatonia and for patients in which lethal catatonia cannot be ruled out. Data supporting its efficacy, however, are limited to a few reviews of published cases [224, 225].

Rechallenge with an antipsychotic after NMS is associated with 30–50 % risk of recurrence [219]. If an antipsychotic is required, low potency FGAs or SGAs are preferred and should be titrated slowly with the lowest possible dose starting no earlier than 2 weeks after NMS resolution.

Venous Thromboembolism

VTE, manifesting as deep vein thrombosis (DVT) and pulmonary embolism (PE), is a serious and potentially fatal medical condition [226]. Patients with DVT usually complain of calf or thigh pain, leg swelling or redness whereas patients with PE present with dyspnea, chest pain, and syncope [226]. Treatment with antipsychotics is associated with an increased risk of VTE [227]. The use of SGAs (particularly clozapine) and low potency FGAs is associated with a higher risk compared with high potency FGAs, with the risk higher among new users [227–230]. The underlying mechanisms for antipsychotic-related VTE are unknown. Proposed mechanisms include antipsychotic-induced weight gain and immobility which are recognized risk factors of VTE; enhanced platelet aggregation by FGAs; increased antiphospholipid antibodies; and hyperhomocysteinemia [227, 231]. The impact of changing antipsychotics to prevent VTE has not been investigated. Antipsychotics with lower risk of VTE [231].

Acute Laryngeal Dystonia

While acute laryngeal dystonia is a rare reaction to antipsychotics, immediate attention is required to prevent death. Laryngeal dystonia usually occurs at the beginning of antipsychotic treatment or a few days after an increase in dose. It presents with
dyspnea, laryngeal stridor, extreme distress, and dystonic reactions in other parts of the body [232]. The patient's hands characteristically grab his/her throat. Immediate attention is frequently required to save the patient's life. Differential diagnosis includes acute anaphylaxis, tardive laryngeal dystonia, airway obstruction, and respiratory dyskinesia [232]. The risk factors for this dystonia include male sex and younger age [232]. All FGAs can cause acute laryngeal dystonia [232]. Only a few cases with SGAs are reported and all were receiving ziprasidone [233, 234]. Acute laryngeal dystonia responds well to parenteral anticholinergics.

Hepatotoxicity

Up to 50 % of patients treated with SGAs have idiosyncratic, asymptomatic liver enzyme elevations [235]. Hepatotoxicity usually occurs in the early weeks of treatment and is reversible [236, 237]. Some reported risk factors include high daily doses, high plasma concentrations, older age, alcoholism, obesity, and antecedent hepatic disorders like Gilbert syndrome [237, 238]. Elevation in liver enzymes is more frequent with clozapine and olanzapine than with risperidone and quetiapine [236, 239]. There is insufficient data about the degree of liver enzyme elevation which should lead to discontinuation or reduction in dose. Discontinuation is clearly indicated, however, if patients have clinical symptoms related to the hepatotoxicity or if enzyme levels exceed three times the upper limit of normal [239, 240].

EPS and Respiratory Distress in Newborns

All antipsychotics may cause EPS and withdrawal symptoms in newborns whose mothers are treated with these agents during the third trimester of pregnancy. A search of the FDA Adverse Event Reporting System database identified 69 cases of neonatal EPS or withdrawal from antipsychotics [241]. Symptoms may include agitation, hypertonia, hypotonia, tremor, somnolence, respiratory distress, and feeding disorder which may occur anytime from birth to 1 month postpartum. The severity of symptoms varies with some neonates recovering within hours or days without specific treatment, while others require intensive care unit support and prolonged hospitalization. Phenobarbital and benzodiazepines may help in treating these withdrawal symptoms.

Conclusion

Apart from EPS which is more common with FGAs, all antipsychotics have similar adverse effects. The severity of the adverse effects, however, varies widely within each class. Adverse effects have a considerable negative impact on quality of life and undermine adherence to treatment. Because efficacy does not differ substantially among these agents with the exception of clozapine, selection of a specific antipsychotic should involve consideration of tolerability and safety, patient risk factors and preferences. Subsequent to starting medication, regular monitoring for adverse effects is crucial to prevent or ameliorate their negative impact and to a reevaluation of the risk–benefit ratio should they arise.

References

- 1. Titier K, Girodet PO, Verdoux H, Molimard M, Begaud B, Haverkamp W, et al. Atypical antipsychotics: from potassium channels to torsade de pointes and sudden death. Drug Saf. 2005;28(1):35–51.
- 2. Abdelmawla N, Mitchell AJ. Sudden cardiac death and antipsychotics. Part1: risk factors and mechanisms. Adv Psychiatr Treat. 2006;12:35–44.
- 3. Ray WA, Meredith S, Thapa PB, Meador KG, Hall K, Murray KT. Antipsychotics and the risk of sudden cardiac death. Arch Gen Psychiatry. 2001;58(12):1161–7.
- Hennessy S, Bilker WB, Knauss JS, Margolis DJ, Kimmel SE, Reynolds RF, et al. Cardiac arrest and ventricular arrhythmia in patients taking antipsychotic drugs: cohort study using administrative data. BMJ. 2002;325(7372):1070.
- 5. Straus SMJM, Bleumink GS, Dieleman JP, van der Lei J, t Jong GW, Kingma JH, et al. Antipsychotics and the risk of sudden cardiac death. Arch Intern Med. 2004;164(12):1293–7 [Erratum appears in Arch Intern Med. 2004 Sep 27;164(17):1839].
- Liperoti R, Gambassi G, Lapane KL, Chiang C, Pedone C, Mor V, et al. Conventional and atypical antipsychotics and the risk of hospitalization for ventricular arrhythmias or cardiac arrest. Arch Intern Med. 2005;165(6):696–701.
- Ray WA, Chung CP, Murray KT, Hall K, Stein CM. Atypical antipsychotic drugs and the risk of sudden cardiac death. N Engl J Med. 2009;360(3):225–35 [Erratum appears in N engl J Med. 2009 Oct 29;361(18):1814].
- Schneeweiss S, Avorn J. Antipsychotic agents and sudden cardiac death—how should we manage the risk? N Engl J Med. 2009;360(3):294–6.
- Reilly JG, Ayis SA, Ferrier IN, Jones SJ, Thomas SHL. Thioridazine and sudden unexplained death in psychiatric in-patients. Br J Psychiatry. 2002;180:515–22.
- Khan A, Faucett J, Morrison S, Brown WA. Comparative mortality risk in adult patients with schizophrenia, depression, bipolar disorder, anxiety disorders, and attention-deficit/hyperactivity disorder participating in psychopharmacology clinical trials. JAMA Psychiatry. 2013;70(10):1091–9.
- 11. Roden DM. Drug-induced prolongation of the QT interval. N Engl J Med. 2004;350(10): 1013–22.
- 12. Taylor DM. Antipsychotics and QT prolongation. Acta Psychiatr Scand. 2003;107(2):85-95.
- 13. Haddad PM, Anderson IM. Antipsychotic-related QTc prolongation, torsade de pointes and sudden death. Drugs. 2002;62(11):1649–71.
- Goldenberg I, Moss AJ, Zareba W. QT interval: how to measure it and what is "normal". J Cardiovasc Electrophysiol. 2006;17(3):333–6.
- Moss AJ. QTc prolongation and sudden cardiac death: the association is in the detail. J Am Coll Cardiol. 2006;47(2):368–9.
- 16. Straus SMJM, Kors JA, De Bruin ML, van der Hooft CS, Hofman A, Heeringa J, et al. Prolonged QTc interval and risk of sudden cardiac death in a population of older adults. J Am Coll Cardiol. 2006;47(2):362–7.
- Vieweg WVR, Wood MA, Fernandez A, Beatty-Brooks M, Hasnain M, Pandurangi AK. Proarrhythmic risk with antipsychotic and antidepressant drugs: implications in the elderly. Drugs Aging. 2009;26(12):997–1012.

- 12 Management of Medication-Related Adverse Effects
 - Montanez A, Ruskin JN, Hebert PR, Lamas GA, Hennekens CH. Prolonged QTc interval and risks of total and cardiovascular mortality and sudden death in the general population: a review and qualitative overview of the prospective cohort studies. Arch Intern Med. 2004;164(9):943–8.
 - Zemrak WR, Kenna GA. Association of antipsychotic and antidepressant drugs with Q-T interval prolongation. Am J Health Syst Pharm. 2008;65(11):1029–38.
- Kongsamut S, Kang J, Chen X-L, Roehr J, Rampe D. A comparison of the receptor binding and HERG channel affinities for a series of antipsychotic drugs. Eur J Pharmacol. 2002;450(1):37–41.
- Harrigan EP, Miceli JJ, Anziano R, Watsky E, Reeves KR, Cutler NR, et al. A randomized evaluation of the effects of six antipsychotic agents on QTc, in the absence and presence of metabolic inhibition. J Clin Psychopharmacol. 2004;24(1):62–9.
- Reilly JG, Ayis SA, Ferrier IN, Jones SJ, Thomas SH. QTc-interval abnormalities and psychotropic drug therapy in psychiatric patients. Lancet. 2000;355(9209):1048–52.
- 23. US Food and Drug Administration Advisory Committee. Zeldox capsules (ziprasidone): summary of efficacy and safety and overall benefit risk relationship. Bethesda, MD: US Food and Drug Administration; 2000.
- Inapsine (Droperidol) Dear Healthcare Professional Letter Dec 2001. U.S. Food and Drug Administration. 2001. http://www.fda.gov/safety/medwatch/safetyinformation/safetyalertsforhumanmedicalproducts/ucm173778.htm. Updated 27 July 2009; Accessed 6 Sept 2013.
- 25. Charbit B, Alvarez JC, Dasque E, Abe E, Demolis JL, Funck-Brentano C. Droperidol and ondansetron-induced QT interval prolongation: a clinical drug interaction study. Anesthesiology. 2008;109(2):206–12.
- Muzyk AJ, Rayfield A, Revollo JY, Heinz H, Gagliardi JP. Examination of baseline risk factors for QTc interval prolongation in patients prescribed intravenous haloperidol. Drug Saf. 2012;35(7):547–53.
- Meyer-Massetti C, Cheng CM, Sharpe BA, Meier CR, Guglielmo BJ. The FDA extended warning for intravenous haloperidol and torsades de pointes: how should institutions respond? J Hosp Med. 2010;5(4):E8–16.
- Lieberman JA, Stroup TS, McEvoy JP, Swartz MS, Rosenheck RA, Perkins DO, et al. Effectiveness of antipsychotic drugs in patients with chronic schizophrenia. N Engl J Med. 2005;353(12):1209–23 [Erratum appears in N engl J Med. 2010 Sep 9;363(11):1092-3].
- Kang UG, Kwon JS, Ahn YM, Chung SJ, Ha JH, Koo YJ, et al. Electrocardiographic abnormalities in patients treated with clozapine. J Clin Psychiatry. 2000;61(6):441–6.
- 30. Chue P, Chue J. A review of paliperidone palmitate. Expert Rev Neurother. 2012;12(12): 1383–97.
- Wang S-M, Han C, Lee S-J, Patkar AA, Pae C-U, Fleischhacker WW. Paliperidone: a review of clinical trial data and clinical implications. Clin Drug Investig. 2012;32(8):497–512.
- INVEGA Prescribing Information. Janssen Pharmaceuticals. 2011. http://www.invega.com/ prescribing-information. Updated June 2011; Accessed 12 Aug 2013.
- Potkin SG, Preskorn S, Hochfeld M, Meng X. A thorough QTc study of 3 doses of iloperidone including metabolic inhibition via CYP2D6 and/or CYP3A4 and a comparison to quetiapine and ziprasidone. J Clin Psychopharmacol. 2013;33(1):3–10.
- 34. Chapel S, Hutmacher MM, Haig G, Bockbrader H, de Greef R, Preskorn SH, et al. Exposureresponse analysis in patients with schizophrenia to assess the effect of asenapine on QTc prolongation. J Clin Pharmacol. 2009;49(11):1297–308.
- Meltzer HY, Cucchiaro J, Silva R, Ogasa M, Phillips D, Xu J, et al. Lurasidone in the treatment of schizophrenia: a randomized, double-blind, placebo- and olanzapine-controlled study. Am J Psychiatry. 2011;168(9):957–67.
- Zareba W, Lin DA. Antipsychotic drugs and QT interval prolongation. Psychiatr Q. 2003;74(3):291–306.
- Yamauchi K, Aki H, Tomotake M, Iga J, Numata S, Motoki I, et al. Predictors of subjective and objective quality of life in outpatients with schizophrenia. Psychiatry Clin Neurosci. 2008;62(4):404–11.

- Haddad PM, Dursun SM. Neurological complications of psychiatric drugs: clinical features and management. Hum Psychopharmacol. 2008;23 Suppl 1:15–26.
- Dayalu P, Chou KL. Antipsychotic-induced extrapyramidal symptoms and their management. Expert Opin Pharmacother. 2008;9(9):1451–62.
- 40. van Harten PN, Hoek HW, Kahn RS. Acute dystonia induced by drug treatment. BMJ. 1999;319(7210):623-6.
- 41. Kane JM, Fleischhacker WW, Hansen L, Perlis R, Pikalov 3rd A, Assuncao-Talbott S. Akathisia: an updated review focusing on second-generation antipsychotics. J Clin Psychiatry. 2009;70(5):627–43.
- Mamo DC, Sweet RA, Keshavan MS. Managing antipsychotic-induced parkinsonism. Drug Saf. 1999;20(3):269–75.
- van Harten PN, Tenback DE. Tardive dyskinesia: clinical presentation and treatment. Int Rev Neurobiol. 2011;98:187–210.
- 44. Chong S-A, Tay JAM, Subramaniam M, Pek E, Machin D. Mortality rates among patients with schizophrenia and tardive dyskinesia. J Clin Psychopharmacol. 2009;29(1):5–8.
- Correll CU, Schenk EM. Tardive dyskinesia and new antipsychotics. Curr Opin Psychiatry. 2008;21(2):151–6.
- Kane JM, Woerner M, Borenstein M, Wegner J, Lieberman J. Integrating incidence and prevalence of tardive dyskinesia. Psychopharmacol Bull. 1986;22(1):254–8.
- 47. Novick D, Haro JM, Bertsch J, Haddad PM. Incidence of extrapyramidal symptoms and tardive dyskinesia in schizophrenia: thirty-six-month results from the European schizophrenia outpatient health outcomes study. J Clin Psychopharmacol. 2010;30(5):531–40.
- 48. Tarsy D, Lungu C, Baldessarini RJ. Epidemiology of tardive dyskinesia before and during the era of modern antipsychotic drugs. Handb Clin Neurol. 2011;100:601–16.
- Miller DD, McEvoy JP, Davis SM, Caroff SN, Saltz BL, Chakos MH, et al. Clinical correlates of tardive dyskinesia in schizophrenia: baseline data from the CATIE schizophrenia trial. Schizophr Res. 2005;80(1):33–43.
- Tenback DE, van Harten PN. Epidemiology and risk factors for (tardive) dyskinesia. Int Rev Neurobiol. 2011;98:211–30.
- Morgenstern H, Glazer WM. Identifying risk factors for tardive dyskinesia among long-term outpatients maintained with neuroleptic medications. Results of the Yale Tardive Dyskinesia Study. Arch Gen Psychiatry. 1993;50(9):723–33.
- 52. Tenback DE, van Harten PN, van Os J. Non-therapeutic risk factors for onset of tardive dyskinesia in schizophrenia: a meta-analysis. Mov Disord. 2009;24(16):2309–15.
- Casey DE. Pathophysiology of antipsychotic drug-induced movement disorders. J Clin Psychiatry. 2004;65 Suppl 9:25–8.
- 54. Kapur S, Zipursky R, Jones C, Remington G, Houle S. Relationship between dopamine D(2) occupancy, clinical response, and side effects: a double-blind PET study of first-episode schizophrenia. Am J Psychiatry. 2000;157(4):514–20.
- 55. Kapur S, Seeman P. Does fast dissociation from the dopamine d(2) receptor explain the action of atypical antipsychotics? A new hypothesis. Am J Psychiatry. 2001;158(3):360–9.
- Leucht S, Corves C, Arbter D, Engel RR, Li C, Davis JM. Second-generation versus first-generation antipsychotic drugs for schizophrenia: a meta-analysis. Lancet. 2009; 373(9657):31–41.
- 57. Rummel-Kluge C, Komossa K, Schwarz S, Hunger H, Schmid F, Kissling W, et al. Secondgeneration antipsychotic drugs and extrapyramidal side effects: a systematic review and meta-analysis of head-to-head comparisons. Schizophr Bull. 2012;38(1):167–77.
- Lima AR, Soares-Weiser K, Bacaltchuk J, Barnes TR. Benzodiazepines for neurolepticinduced acute akathisia. Cochrane Database Syst Rev. 2002;(1):CD001950.
- 59. Rathbone J, Soares-Weiser K. Anticholinergics for neuroleptic-induced acute akathisia. Cochrane Database Syst Rev. 2006;(4):CD003727.
- 60. Friis T, Christensen TR, Gerlach J. Sodium valproate and biperiden in neuroleptic-induced akathisia, parkinsonism and hyperkinesia. A double-blind cross-over study with placebo. Acta Psychiatr Scand. 1983;67(3):178–87.

- 61. Adler LA, Peselow E, Rosenthal M, Angrist B. A controlled comparison of the effects of propranolol, benztropine, and placebo on akathisia: an interim analysis. Psychopharmacol Bull. 1993;29(2):283–6.
- 62. Sachdev P, Loneragan C. Intravenous benztropine and propranolol challenges in acute neuroleptic-induced akathisia. Clin Neuropharmacol. 1993;16(4):324–31.
- Hieber R, Dellenbaugh T, Nelson LA. Role of mirtazapine in the treatment of antipsychotic-induced akathisia. Ann Pharmacother. 2008;42(6):841–6.
- 64. Miodownik C, Lerner V, Statsenko N, Dwolatzky T, Nemets B, Berzak E, et al. Vitamin B6 versus mianserin and placebo in acute neuroleptic-induced akathisia: a randomized, double-blind, controlled study. Clin Neuropharmacol. 2006;29(2):68–72.
- Poyurovsky M, Shardorodsky M, Fuchs C, Schneidman M, Weizman A. Treatment of neuroleptic-induced akathisia with the 5-HT2 antagonist mianserin. Double-blind, placebocontrolled study. Br J Psychiatry. 1999;174:238–42.
- Poyurovsky M, Fuchs C, Weizman A. Low-dose mianserin in treatment of acute neurolepticinduced akathisia. J Clin Psychopharmacol. 1998;18(3):253–4.
- 67. Fischel T, Hermesh H, Aizenberg D, Zemishlany Z, Munitz H, Benjamini Y, et al. Cyproheptadine versus propranolol for the treatment of acute neuroleptic-induced akathisia: a comparative double-blind study. J Clin Psychopharmacol. 2001;21(6):612–5.
- 68. Weiss D, Aizenberg D, Hermesh H, Zemishlany Z, Munitz H, Radwan M, et al. Cyproheptadine treatment in neuroleptic-induced akathisia. Br J Psychiatry. 1995;167(4):483–6.
- 69. Avital A, Gross-Isseroff R, Stryjer R, Hermesh H, Weizman A, Shiloh R. Zolmitriptan compared to propranolol in the treatment of acute neuroleptic-induced akathisia: a comparative double-blind study. Eur Neuropsychopharmacol. 2009;19(7):476–82.
- Gross-Isseroff R, Magen A, Shiloh R, Hermesh H, Weizman A. The 5-HT1D receptor agonist zolmitriptan for neuroleptic-induced akathisia: an open label preliminary study. Int Clin Psychopharmacol. 2005;20(1):23–5.
- Stryjer R, Rosenzcwaig S, Bar F, Ulman AM, Weizman A, Spivak B. Trazodone for the treatment of neuroleptic-induced acute akathisia: a placebo-controlled, double-blind, crossover study. Clin Neuropharmacol. 2010;33(5):219–22.
- Stryjer R, Strous RD, Bar F, Poyurovsky M, Weizman A, Kotler M. Treatment of neurolepticinduced akathisia with the 5-HT2A antagonist trazodone. Clin Neuropharmacol. 2003; 26(3):137–41.
- Zubenko GS, Cohen BM, Lipinski Jr JF, Jonas JM. Use of clonidine in treating neurolepticinduced akathisia. Psychiatry Res. 1984;13(3):253–9.
- Adler LA, Angrist B, Peselow E, Reitano J, Rotrosen J. Clonidine in neuroleptic-induced akathisia. Am J Psychiatry. 1987;144(2):235–6.
- Lerner V, Bergman J, Statsenko N, Miodownik C. Vitamin B6 treatment in acute neurolepticinduced akathisia: a randomized, double-blind, placebo-controlled study. J Clin Psychiatry. 2004;65(11):1550–4.
- Abas MA, Acuda SW, Broadhead JC, Chagwedera IV, Chikara FB, Piachaud JM, et al. WHO consensus statement. Br J Psychiatry. 1990;157:619–20.
- Burgyone K, Aduri K, Ananth J, Parameswaran S. The use of antiparkinsonian agents in the management of drug-induced extrapyramidal symptoms. Curr Pharm Des. 2004;10(18): 2239–48.
- Desmarais JE, Beauclair L, Margolese HC. Anticholinergics in the era of atypical antipsychotics: short-term or long-term treatment? J Psychopharmacol. 2012;26(9):1167–74.
- Allen RM. Role of amantadine in the management of neuroleptic-induced extrapyramidal syndromes: overview and pharmacology. Clin Neuropharmacol. 1983;6 Suppl 1:S64–73.
- Borison RL. Amantadine in the management of extrapyramidal side effects. Clin Neuropharmacol. 1983;6 Suppl 1:S57–63.
- Konig P, Chwatal K, Havelec L, Riedl F, Schubert H, Schultes H. Amantadine versus biperiden: a double-blind study of treatment efficacy in neuroleptic extrapyramidal movement disorders. Neuropsychobiology. 1996;33(2):80–4.
- 82. Soares-Weiser K, Fernandez HH. Tardive dyskinesia. Semin Neurol. 2007;27(2):159-69.

- 83. Soares-Weiser K, Rathbone J. Neuroleptic reduction and/or cessation and neuroleptics as specific treatments for tardive dyskinesia. Cochrane Database Syst Rev. 2006;25(1).
- Kane JM, Rifkin A, Woerner M, Reardon G, Sarantakos S, Schiebel D, et al. Low-dose neuroleptic treatment of outpatient schizophrenics. I. Preliminary results for relapse rates. Arch Gen Psychiatry. 1983;40(8):893–6.
- 85. Cookson IB. The effects of a 50 % reduction of cis(z)-flupenthixol decanoate in chronic schizophrenic patients maintained on a high dose regime. Int Clin Psychopharmacol. 1987;2(2):141–9.
- 86. Leung JG, Breden EL. Tetrabenazine for the treatment of tardive dyskinesia. Ann Pharmacother. 2011;45(4):525–31.
- Soares-Weiser K, Maayan N, McGrath J. Vitamin E for neuroleptic-induced tardive dyskinesia. Cochrane Database Syst Rev. 2011;(2):CD000209.
- Soares KV, McGrath JJ. Anticholinergic medication for neuroleptic-induced tardive dyskinesia. Cochrane Database Syst Rev. 2000;2.
- Tammenmaa IA, Sailas E, McGrath JJ, Soares-Weiser K, Wahlbeck K. Systematic review of cholinergic drugs for neuroleptic-induced tardive dyskinesia: a meta-analysis of randomized controlled trials. Prog Neuropsychopharmacol Biol Psychiatry. 2004;28(7):1099–107.
- Essali A, Deirawan H, Soares-Weiser K, Adams CE. Calcium channel blockers for neuroleptic-induced tardive dyskinesia. Cochrane Database Syst Rev. 2011;(11):CD000206.
- Bhoopathi PS, Soares-Weiser K. Benzodiazepines for neuroleptic-induced tardive dyskinesia. Cochrane Database Syst Rev. 2006;19(3).
- Mentzel CL, Tenback DE, Tijssen MAJ, Visser-Vandewalle VERM, van Harten PN. Efficacy and safety of deep brain stimulation in patients with medication-induced tardive dyskinesia and/or dystonia: a systematic review. J Clin Psychiatry. 2012;73(11):1434–8.
- Damier P, Thobois S, Witjas T, Cuny E, Derost P, Raoul S, et al. Bilateral deep brain stimulation of the globus pallidus to treat tardive dyskinesia. Arch Gen Psychiatry. 2007;64(2): 170–6.
- 94. Gruber D, Trottenberg T, Kivi A, Schoenecker T, Kopp UA, Hoffmann KT, et al. Long-term effects of pallidal deep brain stimulation in tardive dystonia. Neurology. 2009;73(1):53–8.
- Holt RIG. Medical causes and consequences of hyperprolactinaemia. A context for psychiatrists. J Psychopharmacol. 2008;22(2 Suppl):28–37.
- 96. Haddad PM, Wieck A. Antipsychotic-induced hyperprolactinaemia: mechanisms, clinical features and management. Drugs. 2004;64(20):2291–314.
- Melmed S, Casanueva FF, Hoffman AR, Kleinberg DL, Montori VM, Schlechte JA, et al. Diagnosis and treatment of hyperprolactinemia: an endocrine society clinical practice guideline. J Clin Endocrinol Metab. 2011;96(2):273–88.
- Smith S, Wheeler MJ, Murray R, O'Keane V. The effects of antipsychotic-induced hyperprolactinaemia on the hypothalamic-pituitary-gonadal axis. J Clin Psychopharmacol. 2002;22(2):109–14.
- 99. Montgomery J, Winterbottom E, Jessani M, Kohegyi E, Fulmer J, Seamonds B, et al. Prevalence of hyperprolactinemia in schizophrenia: association with typical and atypical antipsychotic treatment. J Clin Psychiatry. 2004;65(11):1491–8.
- Madhusoodanan S, Parida S, Jimenez C. Hyperprolactinemia associated with psychotropics—a review. Hum Psychopharmacol. 2010;25(4):281–97.
- 101. Wieck A, Haddad P. Hyperprolactinaemia caused by antipsychotic drugs. BMJ. 2002; 324(7332):250-2.
- 102. Kinon BJ, Gilmore JA, Liu H, Halbreich UM. Prevalence of hyperprolactinemia in schizophrenic patients treated with conventional antipsychotic medications or risperidone. Psychoneuroendocrinology. 2003;28 Suppl 2:55–68.
- 103. Kaneda Y, Fujii A, Yamaoka T, Morimoto T, Nagamine I. Neither gynecomastia nor galactorrhea is a common side effect of neuroleptics in male patients. Neuroendocrinol Lett. 2000;21(6):447–51. Epub 2001/05/04.
- 104. O'Keane V, Meaney AM. Antipsychotic drugs: a new risk factor for osteoporosis in young women with schizophrenia? J Clin Psychopharmacol. 2005;25(1):26–31.

- 105. Wang PS, Walker AM, Tsuang MT, Orav EJ, Glynn RJ, Levin R, et al. Dopamine antagonists and the development of breast cancer. Arch Gen Psychiatry. 2002;59(12):1147–54.
- 106. Halbreich U, Kinon BJ, Gilmore JA, Kahn LS. Elevated prolactin levels in patients with schizophrenia: mechanisms and related adverse effects. Psychoneuroendocrinology. 2003;28 Suppl 1:53–67.
- 107. Molitch ME. Drugs and prolactin. Pituitary. 2008;11(2):209-18.
- Hoffer ZS, Roth RL, Mathews M. Evidence for the partial dopamine-receptor agonist aripiprazole as a first-line treatment of psychosis in patients with iatrogenic or tumorogenic hyperprolactinemia. Psychosomatics. 2009;50(4):317–24.
- 109. dos Santos NV, El Dib R, Boguszewski CL, Nogueira CR. Cabergoline versus bromocriptine in the treatment of hyperprolactinemia: a systematic review of randomized controlled trials and meta-analysis. Pituitary. 2011;14(3):259–65.
- 110. Lambert M, Conus P, Eide P, Mass R, Karow A, Moritz S, et al. Impact of present and past antipsychotic side effects on attitude toward typical antipsychotic treatment and adherence. Eur Psychiatry. 2004;19(7):415–22.
- 111. Dossenbach M, Hodge A, Anders M, Molnar B, Peciukaitiene D, Krupka-Matuszczyk I, et al. Prevalence of sexual dysfunction in patients with schizophrenia: international variation and underestimation. Int J Neuropsychopharmacol. 2005;8(2):195–201. Epub 2005/01/06.
- 112. Howes OD, Wheeler MJ, Pilowsky LS, Landau S, Murray RM, Smith S. Sexual function and gonadal hormones in patients taking antipsychotic treatment for schizophrenia or schizoaffective disorder. J Clin Psychiatry. 2007;68(3):361–7. Epub 2007/03/29.
- 113. Montejo AL, Majadas S, Rico-Villademoros F, Llorca G, De La Gandara J, Franco M, et al. Frequency of sexual dysfunction in patients with a psychotic disorder receiving antipsychotics. J Sex Med. 2010;7(10):3404–13.
- 114. Ucok A, Incesu C, Aker T, Erkoc S. Sexual dysfunction in patients with schizophrenia on antipsychotic medication. Eur Psychiatry. 2007;22(5):328–33.
- Smith SM, O'Keane V, Murray R. Sexual dysfunction in patients taking conventional antipsychotic medication. Br J Psychiatry. 2002;181:49–55.
- 116. Rettenbacher MA, Hofer A, Ebenbichler C, Baumgartner S, Edlinger M, Engl J, et al. Prolactin levels and sexual adverse effects in patients with schizophrenia during antipsychotic treatment. J Clin Psychopharmacol. 2010;30(6):711–5. Epub 2010/11/26.
- 117. Serretti A, Chiesa A. A meta-analysis of sexual dysfunction in psychiatric patients taking antipsychotics. Int Clin Psychopharmacol. 2011;26(3):130–40.
- 118. Nunes LVA, Moreira HC, Razzouk D, Nunes SOV, Mari JDJ. Strategies for the treatment of antipsychotic-induced sexual dysfunction and/or hyperprolactinemia among patients of the schizophrenia spectrum: a review. J Sex Marital Ther. 2012;38(3):281–301.
- 119. Hanssens L, L'Italien G, Loze J-Y, Marcus RN, Pans M, Kerselaers W. The effect of antipsychotic medication on sexual function and serum prolactin levels in community-treated schizophrenic patients: results from the Schizophrenia Trial of Aripiprazole (STAR) study (NCT00237913). BMC Psychiatry. 2008;8:95.
- 120. Gopalakrishnan R, Jacob KS, Kuruvilla A, Vasantharaj B, John JK. Sildenafil in the treatment of antipsychotic-induced erectile dysfunction: a randomized, double-blind, placebocontrolled, flexible-dose, two-way crossover trial. Am J Psychiatry. 2006;163(3):494–9.
- 121. Salerian AJ, Deibler WE, Vittone BJ, Geyer SP, Drell L, Mirmirani N, et al. Sildenafil for psychotropic-induced sexual dysfunction in 31 women and 61 men. J Sex Marital Ther. 2000;26(2):133–40.
- 122. Laursen TM, Munk-Olsen T, Vestergaard M. Life expectancy and cardiovascular mortality in persons with schizophrenia. Curr Opin Psychiatry. 2012;25(2):83–8.
- 123. Hennekens CH, Hennekens AR, Hollar D, Casey DE. Schizophrenia and increased risks of cardiovascular disease. Am Heart J. 2005;150(6):1115–21.
- 124. Haddad P. Weight change with atypical antipsychotics in the treatment of schizophrenia. J Psychopharmacol. 2005;19(6 Suppl):16–27.
- 125. Holt RIG, Peveler RC. Obesity, serious mental illness and antipsychotic drugs. Diabetes Obes Metab. 2009;11(7):665–79.

- 126. Hasnain M, Vieweg WVR, Fredrickson SK, Beatty-Brooks M, Fernandez A, Pandurangi AK. Clinical monitoring and management of the metabolic syndrome in patients receiving atypical antipsychotic medications. Prim Care Diabetes. 2009;3(1):5–15.
- 127. Roerig JL, Steffen KJ, Mitchell JE. Atypical antipsychotic-induced weight gain: insights into mechanisms of action. CNS Drugs. 2011;25(12):1035–59.
- 128. Nasrallah HA. Atypical antipsychotic-induced metabolic side effects: insights from receptorbinding profiles. Mol Psychiatry. 2008;13(1):27–35.
- Vestri HS, Maianu L, Moellering DR, Garvey WT. Atypical antipsychotic drugs directly impair insulin action in adipocytes: effects on glucose transport, lipogenesis, and antilipolysis. Neuropsychopharmacology. 2007;32(4):765–72.
- 130. Hasnain M, Fredrickson SK, Vieweg WVR, Pandurangi AK. Metabolic syndrome associated with schizophrenia and atypical antipsychotics. Curr Diab Rep. 2010;10(3):209–16.
- 131. De Hert M, Yu W, Detraux J, Sweers K, van Winkel R, Correll CU. Body weight and metabolic adverse effects of asenapine, iloperidone, lurasidone and paliperidone in the treatment of schizophrenia and bipolar disorder: a systematic review and exploratory meta-analysis. CNS Drugs. 2012;26(9):733–59.
- 132. Meyer JM, Davis VG, Goff DC, McEvoy JP, Nasrallah HA, Davis SM, et al. Change in metabolic syndrome parameters with antipsychotic treatment in the CATIE Schizophrenia Trial: prospective data from phase 1. Schizophr Res. 2008;101(1–3):273–86.
- Simon V, van Winkel R, De Hert M. Are weight gain and metabolic side effects of atypical antipsychotics dose dependent? A literature review. J Clin Psychiatry. 2009;70(7):1041–50.
- 134. Guenette MD, Hahn M, Cohn TA, Teo C, Remington GJ. Atypical antipsychotics and diabetic ketoacidosis: a review. Psychopharmacology (Berl). 2013;226(1):1–12.
- 135. Gentile S. Long-term treatment with atypical antipsychotics and the risk of weight gain : a literature analysis. Drug Saf. 2006;29(4):303–19.
- 136. Newcomer JW, Haupt DW. The metabolic effects of antipsychotic medications. Can J Psychiatry. 2006;51(8):480–91.
- 137. Wu R-R, Zhao J-P, Liu Z-N, Zhai J-G, Guo X-F, Guo W-B, et al. Effects of typical and atypical antipsychotics on glucose-insulin homeostasis and lipid metabolism in first-episode schizophrenia. Psychopharmacology (Berl). 2006;186(4):572–8.
- 138. Saddichha S, Manjunatha N, Ameen S, Akhtar S. Diabetes and schizophrenia—effect of disease or drug? Results from a randomized, double-blind, controlled prospective study in first-episode schizophrenia. Acta Psychiatr Scand. 2008;117(5):342–7.
- Peuskens J, De Hert M, Mortimer A, Group SS. Metabolic control in patients with schizophrenia treated with amisulpride or olanzapine. Int Clin Psychopharmacol. 2007;22(3):145–52.
- 140. Kinon BJ, Lipkovich I, Edwards SB, Adams DH, Ascher-Svanum H, Siris SG. A 24-week randomized study of olanzapine versus ziprasidone in the treatment of schizophrenia or schizoaffective disorder in patients with prominent depressive symptoms. J Clin Psychopharmacol. 2006;26(2):157–62.
- 141. Lindenmayer J-P, Czobor P, Volavka J, Citrome L, Sheitman B, McEvoy JP, et al. Changes in glucose and cholesterol levels in patients with schizophrenia treated with typical or atypical antipsychotics. Am J Psychiatry. 2003;160(2):290–6.
- 142. Simpson GM, Weiden P, Pigott T, Murray S, Siu CO, Romano SJ. Six-month, blinded, multicenter continuation study of ziprasidone versus olanzapine in schizophrenia. Am J Psychiatry. 2005;162(8):1535–8.
- 143. Perez-Iglesias R, Crespo-Facorro B, Amado JA, Garcia-Unzueta MT, Ramirez-Bonilla ML, Gonzalez-Blanch C, et al. A 12-week randomized clinical trial to evaluate metabolic changes in drug-naive, first-episode psychosis patients treated with haloperidol, olanzapine, or risperidone. J Clin Psychiatry. 2007;68(11):1733–40.
- 144. Kerwin R, Millet B, Herman E, Banki CM, Lublin H, Pans M, et al. A multicentre, randomized, naturalistic, open-label study between aripiprazole and standard of care in the management of community-treated schizophrenic patients Schizophrenia Trial of Aripiprazole: (STAR) study. Eur Psychiatry. 2007;22(7):433–43.

- 145. Newcomer JW, Campos JA, Marcus RN, Breder C, Berman RM, Kerselaers W, et al. A multicenter, randomized, double-blind study of the effects of aripiprazole in overweight subjects with schizophrenia or schizoaffective disorder switched from olanzapine. J Clin Psychiatry. 2008;69(7):1046–56.
- 146. Weiden PJ, Newcomer JW, Loebel AD, Yang R, Lebovitz HE. Long-term changes in weight and plasma lipids during maintenance treatment with ziprasidone. Neuropsychopharmacology. 2008;33(5):985–94.
- 147. Chrzanowski WK, Marcus RN, Torbeyns A, Nyilas M, McQuade RD. Effectiveness of longterm aripiprazole therapy in patients with acutely relapsing or chronic, stable schizophrenia: a 52-week, open-label comparison with olanzapine. Psychopharmacology (Berl). 2006;189(2):259–66.
- 148. Deberdt WG, Dysken MW, Rappaport SA, Feldman PD, Young CA, Hay DP, et al. Comparison of olanzapine and risperidone in the treatment of psychosis and associated behavioral disturbances in patients with dementia. Am J Geriatr Psychiatry. 2005; 13(8):722–30.
- 149. Lauriello J, Lambert T, Andersen S, Lin D, Taylor CC, McDonnell D. An 8-week, doubleblind, randomized, placebo-controlled study of olanzapine long-acting injection in acutely ill patients with schizophrenia. J Clin Psychiatry. 2008;69(5):790–9 [Erratum appears in J Clin Psychiatry. 2011 Aug;72(8):1157].
- 150. American Diabetes Association, American Psychiatric Association, American Association of Clinical Endocrinologists, North American Association for the Study of Obesity. Consensus development conference on antipsychotic drugs and obesity and diabetes. Diabetes Care. 2004;27(2):596–601.
- 151. Gabriele JM, Dubbert PM, Reeves RR. Efficacy of behavioural interventions in managing atypical antipsychotic weight gain. Obes Rev. 2009;10(4):442–55.
- 152. Alvarez-Jimenez M, Hetrick SE, Gonzalez-Blanch C, Gleeson JF, McGorry PD. Nonpharmacological management of antipsychotic-induced weight gain: systematic review and meta-analysis of randomised controlled trials. Br J Psychiatry. 2008;193(2):101–7.
- 153. Poulin M-J, Chaput J-P, Simard V, Vincent P, Bernier J, Gauthier Y, et al. Management of antipsychotic-induced weight gain: prospective naturalistic study of the effectiveness of a supervised exercise programme. Aust N Z J Psychiatry. 2007;41(12):980–9.
- 154. Wu R-R, Zhao J-P, Jin H, Shao P, Fang M-S, Guo X-F, et al. Lifestyle intervention and metformin for treatment of antipsychotic-induced weight gain: a randomized controlled trial. JAMA. 2008;299(2):185–93.
- 155. Wu M-K, Wang C-K, Bai Y-M, Huang C-Y, Lee S-D. Outcomes of obese, clozapine-treated inpatients with schizophrenia placed on a six-month diet and physical activity program. Psychiatr Serv. 2007;58(4):544–50.
- 156. Knoops KT, de Groot LC, Kromhout D, Perrin AE, Moreiras-Varela O, Menotti A, et al. Mediterranean diet, lifestyle factors, and 10-year mortality in elderly European men and women: the HALE project. JAMA. 2004;292(12):1433–9.
- 157. Casey DE, Carson WH, Saha AR, Liebeskind A, Ali MW, Jody D, et al. Switching patients to aripiprazole from other antipsychotic agents: a multicenter randomized study. Psychopharmacology (Berl). 2003;166(4):391–9.
- 158. De Hert M, Hanssens L, van Winkel R, Wampers M, Van Eyck D, Scheen A, et al. A case series: evaluation of the metabolic safety of aripiprazole. Schizophr Bull. 2007;33(3):823–30.
- 159. Kim SH, Ivanova O, Abbasi FA, Lamendola CA, Reaven GM, Glick ID. Metabolic impact of switching antipsychotic therapy to aripiprazole after weight gain: a pilot study. J Clin Psychopharmacol. 2007;27(4):365–8.
- Weiden PJ, Daniel DG, Simpson G, Romano SJ. Improvement in indices of health status in outpatients with schizophrenia switched to ziprasidone. J Clin Psychopharmacol. 2003;23(6):595–600.
- 161. Montes JM, Rodriguez JL, Balbo E, Sopelana P, Martin E, Soto JA, et al. Improvement in antipsychotic-related metabolic disturbances in patients with schizophrenia switched to ziprasidone. Prog Neuropsychopharmacol Biol Psychiatry. 2007;31(2):383–8.

- Englisch S, Zink M. Combined antipsychotic treatment involving clozapine and aripiprazole. Prog Neuropsychopharmacol Biol Psychiatry. 2008;32(6):1386–92.
- 163. Schorr SG, Slooff CJ, Postema R, Van Oven W, Schilthuis M, Bruggeman R, et al. A 12-month follow-up study of treating overweight schizophrenic patients with aripiprazole. Acta Psychiatr Scand. 2008;118(3):246–50.
- 164. Hasnain M, Vieweg WVR, Fredrickson SK. Metformin for atypical antipsychotic-induced weight gain and glucose metabolism dysregulation: review of the literature and clinical suggestions. CNS Drugs. 2010;24(3):193–206.
- 165. Khan AY, Macaluso M, McHale RJ, Dahmen MM, Girrens K, Ali F. The adjunctive use of metformin to treat or prevent atypical antipsychotic-induced weight gain: a review. J Psychiatr Pract. 2010;16(5):289–96.
- 166. Lee YJ, Jeong JH. A systematic review of metformin to limit weight-gain with atypical antipsychotics. J Clin Pharm Ther. 2011;36(5):537–45.
- 167. Wang M, Tong J-H, Zhu G, Liang G-M, Yan H-F, Wang X-Z. Metformin for treatment of antipsychotic-induced weight gain: a randomized, placebo-controlled study. Schizophr Res. 2012;138(1):54–7.
- Ellinger LK, Ipema HJ, Stachnik JM. Efficacy of metformin and topiramate in prevention and treatment of second-generation antipsychotic-induced weight gain. Ann Pharmacother. 2010;44(4):668–79.
- 169. World Health Organization. Definition of an older or elderly person. 2013. http://www.who. int/healthinfo/survey/ageingdefnolder/en/index.html. Accessed 12 Aug 2013.
- 170. Jano E, Johnson M, Chen H, Aparasu RR. Determinants of atypical antipsychotic use among antipsychotic users in community-dwelling elderly, 1996-2004. Curr Med Res Opin. 2008;24(3):709–16.
- 171. Mittal V, Kurup L, Williamson D, Muralee S, Tampi RR. Risk of cerebrovascular adverse events and death in elderly patients with dementia when treated with antipsychotic medications: a literature review of evidence. Am J Alzheimers Dis Other Demen. 2011;26(1):10–28.
- 172. Sacchetti E, Turrina C, Valsecchi P. Cerebrovascular accidents in elderly people treated with antipsychotic drugs: a systematic review. Drug Saf. 2010;33(4):273–88.
- 173. Hartikainen S, Lonnroos E, Louhivuori K. Medication as a risk factor for falls: critical systematic review. J Gerontol A Biol Sci Med Sci. 2007;62(10):1172–81.
- 174. Hugenholtz GWK, Heerdink ER, van Staa TP, Nolen WA, Egberts ACG. Risk of hip/femur fractures in patients using antipsychotics. Bone. 2005;37(6):864–70.
- 175. Kolanowski A, Fick D, Waller JL, Ahern F. Outcomes of antipsychotic drug use in community-dwelling elders with dementia. Arch Psychiatr Nurs. 2006;20(5):217–25.
- 176. Pratt N, Roughead EE, Ramsay E, Salter A, Ryan P. Risk of hospitalization for hip fracture and pneumonia associated with antipsychotic prescribing in the elderly: a self-controlled case-series analysis in an Australian health care claims database. Drug Saf. 2011;34(7):567–75.
- 177. Trifiro G, Spina E, Gambassi G. Use of antipsychotics in elderly patients with dementia: do atypical and conventional agents have a similar safety profile? Pharmacol Res. 2009;59(1):1–12.
- 178. Trifiro G, Gambassi G, Sen EF, Caputi AP, Bagnardi V, Brea J, et al. Association of community-acquired pneumonia with antipsychotic drug use in elderly patients: a nested case-control study. Ann Intern Med. 2010;152(7):418–25, W139–40.
- 179. Knol W, van Marum RJ, Jansen PAF, Souverein PC, Schobben AFAM, Egberts ACG. Antipsychotic drug use and risk of pneumonia in elderly people. J Am Geriatr Soc. 2008;56(4):661–6.
- Barnett MJ, Perry PJ, Alexander B, Kaboli PJ. Risk of mortality associated with antipsychotic and other neuropsychiatric drugs in pneumonia patients. J Clin Psychopharmacol. 2006;26(2):182–7.
- 181. Liperoti R, Pedone C, Lapane KL, Mor V, Bernabei R, Gambassi G. Venous thromboembolism among elderly patients treated with atypical and conventional antipsychotic agents. Arch Intern Med. 2005;165(22):2677–82.

- 182. Lacut K, Le Gal G, Couturaud F, Cornily G, Leroyer C, Mottier D, et al. Association between antipsychotic drugs, antidepressant drugs and venous thromboembolism: results from the EDITH case-control study. Fundam Clin Pharmacol. 2007;21(6):643–50.
- Kleijer BC, Heerdink ER, Egberts TCG, Jansen PAF, van Marum RJ. Antipsychotic drug use and the risk of venous thromboembolism in elderly patients. J Clin Psychopharmacol. 2010;30(5):526–30.
- 184. Schneider LS, Tariot PN, Dagerman KS, Davis SM, Hsiao JK, Ismail MS, et al. Effectiveness of atypical antipsychotic drugs in patients with Alzheimer's disease. N Engl J Med. 2006;355(15):1525–38.
- 185. Albert SG, Grossberg GT, Thaipisuttikul PJ, Scouby J, Green E. Atypical antipsychotics and the risk of diabetes in an elderly population in long-term care: a retrospective nursing home chart review study. J Am Med Dir Assoc. 2009;10(2):115–9.
- 186. Hammerman A, Dreiher J, Klang SH, Munitz H, Cohen AD, Goldfracht M. Antipsychotics and diabetes: an age-related association. Ann Pharmacother. 2008;42(9):1316–22.
- 187. Klebovich A, Hanko B, Orban K, Zelko R. Antipsychotic treatment and the prevalence of diabetes among elderly patients in psychiatric rehabilitation. Arch Gerontol Geriatr. 2009;48(1):19–21.
- 188. Rondanelli M, Sarra S, Antoniello N, Mansi V, Govoni S, Falvo F, et al. No effect of atypical antipsychotic drugs on weight gain and risk of developing type II diabetes or lipid abnormalities among nursing home elderly patients with Alzheimer's disease. Minerva Med. 2006;97(2):147–51.
- 189. Lee PE, Sykora K, Gill SS, Mamdani M, Marras C, Anderson G, et al. Antipsychotic medications and drug-induced movement disorders other than parkinsonism: a population-based cohort study in older adults. J Am Geriatr Soc. 2005;53(8):1374–9.
- Rochon PA, Stukel TA, Sykora K, Gill S, Garfinkel S, Anderson GM, et al. Atypical antipsychotics and parkinsonism. Arch Intern Med. 2005;165(16):1882–8.
- 191. Uchida H, Mamo DC, Mulsant BH, Pollock BG, Kapur S. Increased antipsychotic sensitivity in elderly patients: evidence and mechanisms. J Clin Psychiatry. 2009;70(3):397–405.
- 192. Chahine LM, Acar D, Chemali Z. The elderly safety imperative and antipsychotic usage. Harv Rev Psychiatry. 2010;18(3):158–72.
- 193. Alexopoulos GS, Streim J, Carpenter D, Docherty JP, Expert Consensus Panel for Using Antipsychotic Drugs in Older Patients. Using antipsychotic agents in older patients. J Clin Psychiatry. 2004;65 Suppl 2:5–99. discussion 100–2; quiz 3–4.
- Fakra E, Azorin JM. Clozapine for the treatment of schizophrenia. Expert Opin Pharmacother. 2012;13(13):1923–35.
- 195. Flanagan RJ, Dunk L. Haematological toxicity of drugs used in psychiatry. Hum Psychopharmacol. 2008;23 Suppl 1:27–41.
- 196. Gerson SL. G-CSF and the management of clozapine-induced agranulocytosis. J Clin Psychiatry. 1994;55(Suppl B):139–42.
- 197. Miller DD. Review and management of clozapine side effects. J Clin Psychiatry. 2000;61 Suppl 8:14–7. discussion 8-9.
- 198. Munro J, O'Sullivan D, Andrews C, Arana A, Mortimer A, Kerwin R. Active monitoring of 12,760 clozapine recipients in the UK and Ireland. Beyond pharmacovigilance. Br J Psychiatry. 1999;175:576–80.
- 199. Manu P, Sarpal D, Muir O, Kane JM, Correll CU. When can patients with potentially lifethreatening adverse effects be rechallenged with clozapine? A systematic review of the published literature. Schizophr Res. 2012;134(2–3):180–6.
- Dunk LR, Annan LJ, Andrews CD. Rechallenge with clozapine following leucopenia or neutropenia during previous therapy. Br J Psychiatry. 2006;188:255–63.
- Nielsen J, Damkier P, Lublin H, Taylor D. Optimizing clozapine treatment. Acta Psychiatr Scand. 2011;123(6):411–22.
- Whiskey E, Taylor D. Restarting clozapine after neutropenia: evaluating the possibilities and practicalities. CNS Drugs. 2007;21(1):25–35.

- 203. Merrill DB, Dec GW, Goff DC. Adverse cardiac effects associated with clozapine. J Clin Psychopharmacol. 2005;25(1):32–41.
- 204. Haas SJ, Hill R, Krum H, Liew D, Tonkin A, Demos L, et al. Clozapine-associated myocarditis: a review of 116 cases of suspected myocarditis associated with the use of clozapine in Australia during 1993-2003. Drug Saf. 2007;30(1):47–57.
- 205. Kilian JG, Kerr K, Lawrence C, Celermajer DS. Myocarditis and cardiomyopathy associated with clozapine. Lancet. 1999;354(9193):1841–5.
- 206. Ronaldson KJ, Fitzgerald PB, Taylor AJ, Topliss DJ, Wolfe R, McNeil JJ. Rapid clozapine dose titration and concomitant sodium valproate increase the risk of myocarditis with clozapine: a case-control study. Schizophr Res. 2012;141(2–3):173–8.
- 207. Ittasakul P, Archer A, Kezman J, Atsariyasing W, Goldman MB. Rapid re-challenge with clozapine following pronounced myocarditis in a treatment-resistance schizophrenia patient. Clin Schizophr Relat Psychoses. 2013;18:1–11.
- 208. Fitzsimons J, Berk M, Lambert T, Bourin M, Dodd S. A review of clozapine safety. Expert Opin Drug Saf. 2005;4(4):731–44.
- Freudenreich O, Henderson DC, Macklin EA, Evins AE, Fan X, Cather C, et al. Modafinil for clozapine-treated schizophrenia patients: a double-blind, placebo-controlled pilot trial. J Clin Psychiatry. 2009;70(12):1674–80.
- 210. Sevy S, Rosenthal MH, Alvir J, Meyer S, Visweswaraiah H, Gunduz-Bruce H, et al. Doubleblind, placebo-controlled study of modafinil for fatigue and cognition in schizophrenia patients treated with psychotropic medications. J Clin Psychiatry. 2005;66(7):839–43.
- Wong J, Delva N. Clozapine-induced seizures: recognition and treatment. Can J Psychiatry. 2007;52(7):457–63.
- 212. Palmer SE, McLean RM, Ellis PM, Harrison-Woolrych M. Life-threatening clozapineinduced gastrointestinal hypomotility: an analysis of 102 cases. J Clin Psychiatry. 2008;69(5):759–68.
- 213. Sockalingam S, Shammi C, Remington G. Clozapine-induced hypersalivation: a review of treatment strategies. Can J Psychiatry. 2007;52(6):377–84.
- Praharaj SK, Arora M, Gandotra S. Clozapine-induced sialorrhea: pathophysiology and management strategies. Psychopharmacology (Berl). 2006;185(3):265–73.
- Bird AM, Smith TL, Walton AE. Current treatment strategies for clozapine-induced sialorrhea. Ann Pharmacother. 2011;45(5):667–75.
- 216. Strawn JR, Keck Jr PE, Caroff SN. Neuroleptic malignant syndrome. Am J Psychiatry. 2007;164(6):870–6.
- 217. Gurrera RJ, Caroff SN, Cohen A, Carroll BT, DeRoos F, Francis A, et al. An international consensus study of neuroleptic malignant syndrome diagnostic criteria using the Delphi method. J Clin Psychiatry. 2011;72(9):1222–8.
- Neuroleptic Malignant Syndrome. American psychiatric association: diagnostic and statistical manual of mental disorders. 5th ed. Arlington, VA: American Psychiatric Association; 2013. p. 709–11.
- 219. Perry PJ, Wilborn CA. Serotonin syndrome vs neuroleptic malignant syndrome: a contrast of causes, diagnoses, and management. Ann Clin Psychiatry. 2012;24(2):155–62.
- Trollor JN, Chen X, Chitty K, Sachdev PS. Comparison of neuroleptic malignant syndrome induced by first- and second-generation antipsychotics. Br J Psychiatry. 2012;201(1):52–6.
- 221. Trollor JN, Chen X, Sachdev PS. Neuroleptic malignant syndrome associated with atypical antipsychotic drugs. CNS Drugs. 2009;23(6):477–92.
- 222. Caroff SN, Mann SC. Neuroleptic malignant syndrome. Med Clin North Am. 1993;77(1): 185–202.
- 223. Reulbach U, Dutsch C, Biermann T, Sperling W, Thuerauf N, Kornhuber J, et al. Managing an effective treatment for neuroleptic malignant syndrome. Crit Care. 2007;11(1):R4.
- 224. Trollor JN, Sachdev PS. Electroconvulsive treatment of neuroleptic malignant syndrome: a review and report of cases. Aust N Z J Psychiatry. 1999;33(5):650–9.
- 225. Davis JM, Janicak PG, Sakkas P, Gilmore C, Wang Z. Electroconvulsive therapy in the treatment of the neuroleptic malignant syndrome. Convuls Ther. 1991;7(2):111–20. Epub 1991/01/01.

- 226. Hogg K, Wells PS, Gandara E. The diagnosis of venous thromboembolism. Semin Thromb Hemost. 2012;38(7):691–701.
- 227. Jonsson AK, Spigset O, Hagg S. Venous thromboembolism in recipients of antipsychotics: incidence, mechanisms and management. CNS Drugs. 2012;26(8):649–62.
- 228. Parker C, Coupland C, Hippisley-Cox J. Antipsychotic drugs and risk of venous thromboembolism: nested case-control study. BMJ. 2010;341:c4245.
- 229. Zhang R, Dong L, Shao F, Tan X, Ying K. Antipsychotics and venous thromboembolism risk: a meta-analysis. Pharmacopsychiatry. 2011;44(5):183–8.
- 230. Hagg S, Bate A, Stahl M, Spigset O. Associations between venous thromboembolism and antipsychotics. A study of the WHO database of adverse drug reactions. Drug Saf. 2008;31(8):685–94.
- Tromeur C, Couturaud F. Antipsychotic drugs and venous thromboembolism. Thromb Res. 2012;130 Suppl 1:S29–31.
- 232. Christodoulou C, Kalaitzi C. Antipsychotic drug-induced acute laryngeal dystonia: two case reports and a mini review. J Psychopharmacol. 2005;19(3):307–11.
- Duggal HS. Ziprasidone-induced acute laryngeal dystonia. Prog Neuropsychopharmacol Biol Psychiatry. 2007;31(4):970. author reply 1.
- 234. Mellacheruvu S, Norton JW, Schweinfurth J. Atypical antipsychotic drug-induced acute laryngeal dystonia: 2 case reports. J Clin Psychopharmacol. 2007;27(2):206–7.
- 235. Ozcanli T, Erdogan A, Ozdemir S, Onen B, Ozmen M, Doksat K, et al. Severe liver enzyme elevations after three years of olanzapine treatment: a case report and review of olanzapine associated hepatotoxicity. Prog Neuropsychopharmacol Biol Psychiatry. 2006;30(6): 1163–6.
- 236. Pae C-U, Lim H-K, Kim T-S, Kim J-J, Lee C-U, Lee S-J, et al. Naturalistic observation on the hepatic enzyme changes in patients treated with either risperidone or olanzapine alone. Int Clin Psychopharmacol. 2005;20(3):173–6.
- 237. Dumortier G, Cabaret W, Stamatiadis L, Saba G, Benadhira R, Rocamora JF, et al. Hepatic tolerance of atypical antipsychotic drugs. Encéphale. 2002;28(6 Pt 1):542–51.
- 238. Stine JG, Sateesh P, Lewis JH. Drug-induced liver injury in the elderly. Curr Gastroenterol Rep. 2013;15(1):299.
- 239. Manceaux P, Constant E, Zdanowicz N, Jacques D, Reynaert C. Management of marked liver enzyme increase during olanzapine treatment: a case report and review of the literature. Psychiatr Danub. 2011;23 Suppl 1:S15–7.
- 240. Erdogan A, Kocabasoglu N, Yalug I, Ozbay G, Senturk H. Management of marked liver enzyme increase during clozapine treatment: a case report and review of the literature. Int J Psychiatry Med. 2004;34(1):83–9.
- 241. FDA Drug Safety Communication. Antipsychotic drug labels updated on use during pregnancy and risk of abnormal muscle move [updated March 29, 2011; cited 2013 August 12]; ments and withdrawal symptoms in newborns. U.S. Food and Drug Administration; 2011. http://www.fda.gov/Drugs/DrugSafety/ucm243903.htm
- Lin C-H, Chen M-C, Wang S-Y, Lin C-Y. Predictive factors for QTc prolongation in schizophrenic patients taking antipsychotics. J Formos Med Assoc. 2004;103(6):437–41.
- 243. Zeltser D, Justo D, Halkin A, Prokhorov V, Heller K, Viskin S. Torsade de pointes due to noncardiac drugs: most patients have easily identifiable risk factors. Medicine. 2003;82(4):282–90.
- Crouch MA, Limon L, Cassano AT. Clinical relevance and management of drug-related QT interval prolongation. Pharmacotherapy. 2003;23(7):881–908.
- 245. Kannankeril PJ, Norris KJ, Carter S, Roden DM. Factors affecting the degree of QT prolongation with drug challenge in a large cohort of normal volunteers. Heart Rhythm. 2011;8(10):1530–4.
- 246. Sicouri S, Antzelevitch C. Sudden cardiac death secondary to antidepressant and antipsychotic drugs. Expert Opin Drug Saf. 2008;7(2):181–94.
- 247. Taira CA, Opezzo JAW, Mayer MA, Hocht C. Cardiovascular drugs inducing QT prolongation: facts and evidence. Curr Drug Saf. 2010;5(1):65–72.

- 248. Briasoulis A, Agarwal V, Pierce WJ. QT prolongation and torsade de pointes induced by fluoroquinolones: infrequent side effects from commonly used medications. Cardiology. 2011;120(2):103–10.
- 249. Woosley RL. Drugs that prolong the qt interval and/or induce Torsades de Pointes. ARIZONA CERT Center for Educationa and Research on Therapeutics; 2013. http://www.azcert.org/medical-pros/drug-lists/printable-drug-list.cfm. Updated 12 July 2013; Accessed 12 Aug 2013.
- 250. Drug Development and Drug Interactions. Table of substrates, inhibitors and inducers. U.S. Food and Drug Administration; 2011. http://www.fda.gov/Drugs/DevelopmentApprovalProcess/ DevelopmentResources/DrugInteractionsLabeling/ucm093664.htm. Updated 16 Sept 2011; Accessed 12 Aug 2013.
- 251. Miller CH, Fleischhacker WW. Managing antipsychotic-induced acute and chronic akathisia. Drug Saf. 2000;22(1):73–81.
- 252. Cortet-Rudelli C, Sapin R, Bonneville JF, Brue T. Etiological diagnosis of hyperprolactinemia. Ann Endocrinol. 2007;68(2-3):98–105.
- 253. Alberti KGMM, Zimmet P, Shaw J. Metabolic syndrome—a new world-wide definition. A consensus statement from the International Diabetes Federation. Diabet Med. 2006; 23(5):469–80.
- 254. Expert Panel on Detection, Evaluation, Treatment of High Blood Cholesterol in Adults. Executive summary of the Third Report of the National Cholesterol Education Program (NCEP) Expert Panel on Detection, Evaluation, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III). JAMA. 2001;285(19):2486–97.
- 255. Grundy SM, Brewer Jr HB, Cleeman JI, Smith Jr SC, Lenfant C, American Heart A, et al. Definition of metabolic syndrome: report of the National Heart, Lung, and Blood Institute/ American Heart Association conference on scientific issues related to definition. Circulation. 2004;109(3):433–8.
- 256. Teva Clozapine Monitoring Guidelines. Teva Pharmaceuticals. 2008. http://www.clozapineregistry.com/resuming_treatment_after_interruption.pdf.ashx. Accessed 12 Aug 2013.
- 257. Advisory Committee Briefing Document for Zeldox Capsules (Ziprasidone HCl). U.S. Food and Drug Administration. 2000. http://www.fda.gov/ohrms/dockets/ac/00/backgrd/3619b1a. pdf. Updated 19 July 2000; Accessed 12 Aug 2013.

Part IV Management of Schizophrenia and Its Long-Term Complications

Chapter 13 Long-Term Pharmacological Management of Schizophrenia

Stephen R. Marder

Introduction

Patients with schizophrenia generally reach a stable clinical state after treatment for an acute episode. For some individuals, this is characterized by the elimination of all symptoms. For most patients, however, some positive symptoms will persist along with impaired cognition and possibly negative symptoms. Patients may also be burdened by impaired mood—particularly depression—and anxiety. Problems with substance use may also interfere with a full recovery.

Until recently, most clinicians viewed relapse prevention as the main goal of pharmacology during long-term treatment. Effective relapse prevention reduced the need for psychiatric hospitalization and allowed patients to live in their communities. Preventing relapse, however, was often insufficient as a treatment goal. Long-term studies suggested that unrelapsed patients were usually unable to meet their educational and vocational goals or to fulfill their family responsibilities. This observation, as well as the views of schizophrenia patients and their advocates, led to a reevaluation of the role of long-term treatment and the concept of recovery.

The term recovery can have different meanings in an illness such as schizophrenia. We use the definition from the Presidents New Freedom Commission on Mental Health. They define it as "... the process in which people are able to live, work, learn, and participate fully in their communities. For some individuals, recovery is the ability to live a fulfilling and productive life despite a disability." It should be noted that recovery is a process rather than a well-defined state. Its conceptualization also emphasizes the importance of the patient setting his or her own goals. Recoveryoriented programs are built on individuals' strengths and support their ability to

S.R. Marder (🖂)

UCLA Semel Institute, Los Angeles, CA 90073, USA e-mail: marder@ucla.edu

manage the challenges in their lives. In this chapter and Chap. 14, we emphasize the roles for both drugs and non-pharmacological treatments to help patients meet their own recovery goals.

Antipsychotic Medications and Long-Term Treatment

Antipsychotic medications are effective in decreasing the risk of psychotic relapse. This is demonstrated in a large number of clinical trials in which patients who are stable on their medication either stay on an antipsychotic or are switched to placebo. Taken together, these studies find that those who continue on medication are less likely to have a relapse of their psychosis [1, 2]. In the first year following drug discontinuation (i.e., those switched to placebo), about two-thirds of patients relapsed in contrast to only about a quarter who remained on an antipsychotic. There is also a suggestion from the Leucht et al. meta-analysis that the second-generation antipsychotics (SGAs) may have a small advantage over older medications [2]. This may be attributable to better adherence due to the milder adverse effects of these newer drugs.

Antipsychotics do more than just prevent psychotic relapse. Thus, when relapses occur in patients receiving an antipsychotic they tend to be less severe than those occurring in patients off drugs. For example, those on an antipsychotic are less likely to require involuntary hospitalization and less likely to make suicide attempts [3].

These findings support the clinical recommendation that patients with schizophrenia should remain on their medication after they recover from an episode of acute psychosis. Patients who are stable for years with minimal symptoms will often ask whether there is a need to continue on their antipsychotic. Studies which attempted to discontinue medications in these patients, however, found that they had similar rates of relapse as other stable phase patients [4]. In a sense, these individuals may be deriving the greatest benefit from an antipsychotic.

Relapse Prevention and Recovery

Although preventing relapse in schizophrenia with antipsychotic medication does not by itself lead to recovery, there is evidence that it is an important component. This was noted by Hogarty and coworkers who compared long-acting and oral fluphenazine combined with a type of social therapy or a control condition [5]. They found that patients who received long-acting fluphenazine derived the most benefit from social therapy. Thus, it appears that remaining stable (or unrelapsed) is a necessary condition for deriving benefit from such treatments. A subsequent study found that patients who were more symptomatic tended to have greater impaired work skills and derived less benefit from vocational programs [6]. This area of research supports the importance of continuing antipsychotic medication for patients involved in psychosocial treatments or rehabilitation.

Medication Adherence During Long-Term Treatment

Medication nonadherence is relatively common and the most frequent cause of relapse in schizophrenia [7]. A Veterans Affairs study found that 61 % of patients had adherence problems at some time during a 4-year period [8]. There is also evidence that nonadherence is associated with a worse functional outcome [9]. As a result, managing nonadherence and improving partial adherence are important components of longterm treatment.

Managing nonadherence requires being aware of when it is occurring. There is evidence that clinicians tend to underestimate the levels of nonadherence in their patients which may result from the reluctance of many to admit to their provider that they are not following directions. Barry Blackwell suggests an approach to this dilemma [10]. Asking patients if they are taking their medications may elicit a defensive response if they believe the provider will be critical. By contrast, inquiring about the difficulties of taking medication once or more a day in a manner that is not judgmental may provide an opportunity for patients to discuss their attitudes about taking antipsychotics. This may provide further opportunities for clinicians to educate about the role of these drugs in stable patients.

Long-Acting Injectable Antipsychotics

Long-acting injectable (LAI) antipsychotics are administered intramuscularly every 2–4 weeks. Possible advantages include:

- Assuring drug delivery in patients who are unreliable pill takers or are drug reluctant
- Providing a mechanism for monitoring adherence with injections
- Treating patients with *lower and more stable plasma concentrations* than oral medications

Regarding the last point, LAIs have a different pharmacokinetic profile compared with oral compounds. Patients who receive an oral drug usually reach peak plasma concentration shortly after ingestion which is then followed by a gradual elimination. In contrast, patients who receive an LAI agent will usually have a relatively steady release of the drug from the injection site [11]. There may, however, be some exceptions. For example, paliperidone palmitate will usually be released shortly after an injection [12]. Nevertheless, LAIs generally have a lower peak plasma concentration which can reduce adverse effects.

Effectiveness of LAI Versus Oral Antipsychotics

A number of studies comparing the effectiveness of LAI and oral antipsychotics have generated inconsistent results. Large naturalistic trials with diverse populations usually find advantages for LAIs in relapse prevention. For example, a recent study from Finland followed 2,588 consecutive patients hospitalized for a first episode of schizophrenia [13]. LAI antipsychotics were associated with a lower risk of re-hospitalization compared with oral antipsychotics (HR=0.36,95 % CI=0.17–0.75). Controlled trials, however, are less clear with a recent meta-analysis (ten trials; 1,700 patients) finding only modest advantages for LAIs [14]. The difference between the naturalistic and controlled trials may be explained by the types of individuals who are included. Thus, patients who are willing to cooperate with a controlled trial which involves a long-acting condition may not be the subjects most likely to derive a differential benefit from a treatment that assures drug delivery.

Certain patients are more likely to benefit from LAIs. They include those who are unreliable in taking pills, as well as those who have a history of repeated psychotic episodes which occur after medication nonadherence. As mentioned earlier, there is also evidence that patients receiving LAIs may derive greater benefit from psychosocial treatments. Clinicians often reserve LAIs for patients who have an established history of nonadherence and are reluctant to prescribe these medications early in the illness. There is evidence, however, that LAIs may help patients who are recovering from an initial episode because even brief periods of nonadherence can have negative consequences [15]. In a study of recent onset patients, those treated with LAI risperidone had better cognitive outcomes and, notably, avoided the losses in white matter associated with oral risperidone [16].

Switching from an Oral to an LAI Antipsychotic

Clinicians who switch patients from an oral antipsychotic should consider a number of important characteristics of LAIs [17]. The most important is that LAIs are 100 % bioavailable, whereas the availability of an oral antipsychotic is usually limited by drug metabolism, absorption in the gut, and variable adherence by the patient. As a result, simple conversion formulas for calculating the LAI dose requirements based on the oral drug dose may be inaccurate. Therefore, LAIs should be prescribed at relatively conservative doses. In addition, it often takes three or more months for patients to reach a steady state plasma level after starting an LAI agent. This problem is addressed by continuing patients on an oral drug for the first several weeks after starting the LAI. This may not be necessary, however, for patients who are treated with paliperidone palmitate or olanzapine pamoate, since there is a more rapid release of these agents during the first days of treatment.

Selecting an LAI Antipsychotic

There is no data indicating that the available LAIs differ in their effectiveness or their tendency to cause extrapyramidal side effects (EPS). As noted in Table 13.1, there are differences in the intervals between injections and the convenience of

	Dose	Injection		
Agent	range (mg)	interval	Available doses	Comments
Fluphenazine decanoate	12.5–100	2–4 weeks	25 mg/mL vials	z-Track injection in gluteal or deltoid
Haloperidol decanoate	50-200	4 weeks	50 or 100 mg/ mL vials	z-Track injection in gluteal or deltoid
Risperidone microspheres	12.5–50	2 weeks	12.5, 25, 37.5 or 50 mg	Requires storage in a refrigerator
Paliperidone palmitate	39–234	4 weeks	39, 78, 117, 156, or 234 mg	Higher plasma concentrations with deltoid injections
Olanzapine pamoate	150-405	2–4 weeks	210, 300 or 405 mg	Patients should be observed for 3 h after each injection for postinjection delirium sedation. Requires gluteal injection
Aripipirazole monohydrate	300 or 400	4 weeks	300 or 400 mg	Requires suspension in sterile water prior to injection

Table 13.1 Long-acting injectable antipsychotic medications

giving injections. The adverse effects of each LAI agent, however, are similar to their corresponding oral agent. In addition, the newer LAI agents are likely to be more costly.

Olanzapine Pamoate

This agent is a microcrystalline salt of pamoic acid and olanzapine suspended in an aqueous solution [18]. Once injected the constituents slowly dissociate into their separate components. The peak plasma concentration is reached 3 or 4 days following a single injection. This indicates there may be a sufficient release of olanzapine during the first days to provide therapeutic plasma levels. As a result, patients with acute schizophrenia may not need oral supplementation. The half-life of olanzapine pamoate (OP) is about 30 days and steady state is reached in about 12 weeks [19].

OP is associated with a risk of postinjection delirium/sedation syndrome (PDSS) which consists of severe sedation, delirium, and confusion, with some individuals having lost consciousness as a result. Most of these symptoms occur within an hour of the OP injection and resemble an overdose of olanzapine. A case analysis based on all eight olanzapine LAI clinical trials identified this syndrome in approximately 0.07 % of injections or 1.4 % of patients [20]. Because of the concerns about this syndrome, OP is only available through a system which requires registration of the patient, the prescriber, the facility, and the pharmacy. Further, patients who receive OP must be observed for a PDSS in a medical setting by a healthcare professional for 3 h after each injection. In the absence of direct comparison between LAI agents, it is unclear if this risk differs among the various agents.

The efficacy of OP for acute schizophrenia was evaluated in a placebo-controlled, 8-week trial which compared 210 mg every 2 weeks; 300 mg every 2 weeks; 405 mg every 4 weeks; and placebo. All three active treatment groups showed greater improvement on the PANSS total score compared with the placebo group (all *p* values ≤ 0.001) [21]. Similar to the oral olanzapine, there was clinically significant weight gain and changes on some lipid parameters. The effectiveness of OP in preventing relapse was also evaluated in a 24-week trial [22]. This study randomized patients on a stable oral dose of olanzapine to the LAI formulation at 150 mg every 2 weeks, 300 mg every 2 weeks, 45 mg every 4 weeks, 405 mg every 4 weeks, or to remain on the stabilized oral dose. The LAI 45 mg dose every 4 weeks was considered as a subtherapeutic control arm. Based on the changes in the PANSS total score, all three standard doses of OP were superior to the 45 mg dose (p < 0.001) and similar to the oral condition in protecting patients against psychotic exacerbations. There were no clinically significant differences in general safety between the LAI doses and oral formulation. Two patients experienced the PDSS.

Paliperidone Palmitate

This agent is an ester in an aqueous suspension [12]. Given the very low solubility, it is slowly absorbed at the injection site where it is hydrolyzed by muscle esterases, releasing paliperidone into the circulation. The release begins on the first day, but the maximum plasma concentration is not reached for about 2 weeks. As a result, this formulation has antipsychotic activity on the first day following administration. A higher maximum plasma concentration is reached when the drug is injected into the deltoid rather than the gluteal muscle. Therefore, if a clinician is starting a patient on paliperidone palmitate (PP) without a cross titration from an oral antipsychotic, deltoid administration may be preferred. On the other hand, if minimal adverse effects are a priority, gluteal administration may have advantages.

The effectiveness of PP for treating acute schizophrenia was evaluated in a placebocontrolled trial which compared injections of two doses of PP (50 or 100 mg equivalents) or placebo on days 1, 8, and 36 [23]. Both active drug doses were more effective than placebo ($p \le 0.001$). Safety measures indicated PP was well tolerated. In a second 13-week trial, three fixed doses of PP (25, 50, and 100 mg equivalents) were compared with placebo [24]. Based on change in the PANSS total score, all PP doses were superior to placebo (25 and 50 mg, $p \le 0.02$; 100 mg, $p \le 0.001$). The active drug was generally well tolerated both locally and systemically. The effectiveness of PP for maintenance therapy was demonstrated in a trial which randomized stable patients to continue on active drug or a placebo injection [25]. PP was more effective than placebo for delaying relapse ($p \le 0.001$). A Cochrane review noted that PP was more effective than placebo, comparable to risperidone LAI and has similar adverse effects to oral risperidone and paliperidone including weight gain, elevated prolactin and tachycardia [26].

Risperidone Microspheres

This agent is an aqueous suspension of risperidone contained in microspheres of a biodegradable copolymer [27]. After injections, the copolymer is degraded within the muscle, releasing risperidone. This begins about 3 weeks after the first injection and steady state is usually reached after the fourth injection. Because of the delayed release of risperidone, patients should continue to receive the oral formulation or another oral antipsychotic during the transition. In their review, Harrison and Goa concluded that risperidone microspheres (RM) (25 or 50 mg every 2 weeks) was superior to placebo and comparable to oral risperidone (2–6 mg/day) in two, 12-week, double-blind trials [27]. Further, symptom improvement in stable patients was significant in two non-comparative multicentral trials (25 or 50 mg over 12 months [28]; or 25, 37.5 of 50 mg over 12 weeks [29]). There was generally a low incidence of injection site pain and a similar adverse effect profile to oral risperidone.

Aripiprazole Monohydrate

This agent consists of lyophilized particles containing aripiprazole which are suspended in sterile water just prior to injection. The aripiprazole particles are slowly absorbed with a time to maximum plasma concentration of about 5–7 days and a mean terminal elimination half-life of 29.9 days for the 300 mg dose and 46.5 days for the 400 mg dose. Since steady state is reached after four injections, the patient's oral antipsychotic should be continued for at least 2 weeks [30].

The effectiveness of aripiprazole monohydrate (AM) was demonstrated in a 52-week, Phase 3, multicenter, randomized, double-blind study [31, 32]. Stable patients (n=403) were randomly assigned to either 400 mg of LAI aripiprazole or placebo. The time to impending relapse was significantly delayed compared with placebo in both the interim and final analysis ($p \le 0.001$). The most common adverse effects with AM were insomnia, tremor, and headache. The discontinuation rates were less in the AM group compared with placebo (i.e., 24.9 % versus 54.5 %).

Haloperidol and Fluphenazine Decanoate

These formulations are produced by combining the antipsychotic with a fatty acid (decanoic acid) administered in sesame oil. The esterified antipsychotic is gradually absorbed by the oil and hydrolyzed by tissue esterases. Since the drug release for both is delayed, neither decanoate is appropriate for acute treatment. The effectiveness of both LAIs for relapse prevention is supported by multiple trials and meta-analyses [33–37].

Conclusion

Antipsychotic medications play a critical but limited role in helping patients to achieve recovery. They are effective in decreasing the likelihood that patients who are stable will experience a relapse of their psychosis, as well as minimizing the severity if a relapse does occur. This is an essential component, since remaining stable is a necessary element if patients are to benefit from the psychosocial treatment and rehabilitation methods described in Chap. 14.

References

- Davis JM, Chen N. Choice of maintenance medication for schizophrenia. J Clin Psychiatry. 2003;64 Suppl 16:24–33.
- Leucht S, Barnes TR, Kissling W, Engel RR, Correll C, Kane JM. Relapse prevention in schizophrenia with new-generation antipsychotics: a systematic review and exploratory metaanalysis of randomized, controlled trials. Am J Psychiatry. 2003;160(7):1209–22.
- Johnson DAW, Pasterski JM, Ludlow JM, et al. The discontinuance of maintenance neuroleptic therapy in chronic schizophrenic patients: drug and social consequences. Acta Psychiatr Scand. 1983;67:339–52.
- Hogarty GE, Ulrich RF, Mussare F, Aristigueta N. Drug discontinuation among long term, successfully maintained schizophrenic outpatients. Dis Nerv Syst. 1976;37:494–500.
- 5. Hogarty GE, Schooler NR, et al. Fluphenazine and social therapy in the aftercare of schizophrenic patients: relapse analysis of two year controlled study of fluphenazine decanoate and fluphenazine hydrochloride. Arch Gen Psychiatry. 1979;36:1283–94.
- Anthony WA, Rogers ES, Cohen M, Davies RR. Relationships between psychiatric symptomatology, work skills, and future vocational performance. Psychiatr Serv. 1995;46(4):353–8.
- Weiden PJ. Understanding and addressing adherence issues in schizophrenia: from theory to practice. J Clin Psychiatry. 2007;68 Suppl 14:14–9.
- Valenstein M, Ganoczy D, McCarthy JF, Myra Kim H, Lee TA, Blow FC. Antipsychotic adherence over time among patients receiving treatment for schizophrenia: a retrospective review. J Clin Psychiatry. 2006;67(10):1542–50.
- Ascher-Svanum H, Faries DE, Zhu B, Ernst FR, Swartz MS, Swanson JW. Medication adherence and long-term functional outcomes in the treatment of schizophrenia in usual care. J Clin Psychiatry. 2006;67(3):453–60.
- 10. Blackwell B. Treatment adherence. Br J Psychiatry. 1976;129:513-31.
- Midha KK, Hubbard JW, Marder SR, Marshall BD, Van Putten T. Impact of clinical pharmacokinetics on neuroleptic therapy in patients with schizophrenia. J Psychiatry Neurosci. 1994;19(4):254–64.
- Gilday E, Nasrallah HA. Clinical pharmacology of paliperidone palmitate a parenteral longacting formulation for the treatment of schizophrenia. Rev Recent Clin Trials. 2012;7(1): 2–9.
- Tiihonen J, Haukka J, Taylor M, Haddad PM, Patel MX, Korhonen P. A nationwide cohort study of oral and depot antipsychotics after first hospitalization for schizophrenia. Am J Psychiatry. 2011;168(6):603–9.
- Leucht C, Heres S, Kane JM, Kissling W, Davis JM, Leucht S. Oral versus depot antipsychotic drugs for schizophrenia—a critical systematic review and meta-analysis of randomised longterm trials. Schizophr Res. 2011;127(1–3):83–92.
- 15. Subotnik KL, Nuechterlein KH, Ventura J, et al. Risperidone nonadherence and return of positive symptoms in the early course of schizophrenia. Am J Psychiatry. 2011;168(3):286–92.

- Bartzokis G, Lu PH, Amar CP, Raven EP, Detore NR, Altshuler LL, Mintz J, et al. Long acting injection versus oral risperidone in first-episode schizophrenia: differential impact on white matter myelination trajectory. Schizophr Res. 2011;132(1):35–41.
- Marder SR, Hubbard JW, Van Putten T, Midha KK. Pharmacokinetics of long-acting injectable neuroleptic drugs: clinical implications. Psychopharmacology (Berl). 1989; 98(4):433–9.
- 18. Chue P, Chue J. A review of olanzapine pamoate. Expert Opin Pharmacother. 2012;13(11): 1661–70.
- Di Lorenzo R, Brogli A. Profile of olanzapine long-acting injection for the maintenance treatment of adult patients with schizophrenia. Neuropsychiatr Dis Treat. 2010;6:573–81.
- Novakovic V, Adel T, Peselow E, Lindenmayer JP. Long-acting injectable antipsychotics and the development of postinjection delirium/sedation syndrome (PDSS). Clin Neuropharmacol. 2013;36(2):59–62.
- Lauriello J, Lambert T, Andersen S, Lin D, Taylor CC, McDonnell D. An 8-week, doubleblind, randomized, placebo-controlled study of olanzapine long-acting injection in acutely ill patients with schizophrenia. J Clin Psychiatry. 2008;69(5):790–9.
- Kane JM, Detke HC, Naber D, et al. Olanzapine long-acting injection: a 24-week, randomized, double-blind trial of maintenance treatment in patients with schizophrenia. Am J Psychiatry. 2010;167(2):181–9.
- Kramer M, Litman R, Hough D, Lane R, Lim P, Liu Y, et al. Paliperidone palmitate, a potential long-acting treatment for patients with schizophrenia. Results of a randomized, double-blind, placebo-controlled efficacy and safety study. Int J Neuropsychopharmacol. 2010; 13(5):635–47.
- 24. Nasrallah HA, Gopal S, Gassmann-Mayer C, Quiroz JA, Lim P, Eerdekens M, et al. A controlled, evidence-based trial of paliperidone palmitate, a long-acting injectable antipsychotic, in schizophrenia. Neuropsychopharmacology. 2010;35(10):2072–82.
- 25. Hough D, Gopal S, Vijapurkar U, Lim P, Morozova M, Eerdekens M. Paliperidone palmitate maintenance treatment in delaying the time-to-relapse in patients with schizophrenia: a randomized, double-blind, placebo-controlled study. Schizophr Res. 2010;116(2–3):107–17.
- Nussbaum AM, Stroup TS. Paliperidone palmitate for schizophrenia. Cochrane Database Syst Rev. 2012;(6):CD008296.
- Harrison TS, Goa KL. Long-acting risperidone: a review of its use in schizophrenia. CNS Drugs. 2004;18(2):113–32.
- Fleischhacker WW, Eerdekens M, Karcher K, Remington G, Llorca PM, Chrzanowski W, et al. Treatment of schizophrenia with long-acting injectable risperidone: a 12-month openlabel trial of the first long-acting second-generation antipsychotic. J Clin Psychiatry. 2003;64(10):1250–7.
- Kane JM, Eerdekens M, Lindenmayer JP, Keith SJ, Lesem M, Karcher K. Long-acting injectable risperidone: efficacy and safety of the first long-acting atypical antipsychotic. Am J Psychiatry. 2003;160(6):1125–32.
- Spanarello S, La Ferla T. The pharmacokinetics of long-acting antipsychotic medications. Curr Clin Pharmacol. 2013. [Epub ahead of print].
- 31. Kane JM, Sanchez R, Perry PP, Jin N, Johnson BR, Forbes RA, et al. Aripiprazole intramuscular depot as maintenance treatment in patients with schizophrenia: a 52-week, multicenter, randomized, double-blind, placebo-controlled study. J Clin Psychiatry. 2012;73(5):617–24.
- 32. Gopalakrishna G, Aggarwal A, Lauriello J. Long-acting injectable aripiprazole: how might it fit in our tool box? Clin Schizophr Relat Psychoses. 2013;87–92. www.clinicalschizophrenia. net. Accessed 22 July 2013.
- Kane JM, Davis JM, Schooler N, et al. A multidose study of haloperidol decanoate in the maintenance treatment of schizoprenia. Am J Psychiatry. 2002;159(4):554–60.
- Chouinard G, Annable L, Campbell W. A randomized clinical trial of haloperidol decanoate and fluphenazine decanoate in the outpatient treatment of schizophrenia. J Clin Psychopharmacol. 1989;9(4):247–53.

- Hemstrom CA, Evans RL, Lobeck FG. Haloperidol decanoate: a depot antipsychotic. Drug Intell Clin Pharm. 1988;22(4):290–5.
- 36. Purgato M, Adams CE. Bromperidol decanoate (depot) for schizophrenia. Cochrane Database Syst Rev. 2012;(11):CD0017919.
- 37. David A, Adams CE, Eisenbruch M, Quraishi S, Rathbone J. Depot fluphenazine decanoate and enanthate for schizophrenia. Cochrane Database Syst Rev. 2005;(1):CD000307.

Chapter 14 Psychosocial Rehabilitation and Psychotherapy Approaches

Robert S. Kern, William P. Horan, Shirley M. Glynn, L. Felice Reddy, Jason Holden, Eric Granholm, Luana R. Turner, Mary E. Sullivan, and William D. Spaulding

Abbreviations

ACT	Assertive Community Treatment
BFT	Behavioral family therapy
CAT	Cognitive adaptation training

R.S. Kern (🖂)

VA Greater Los Angeles Healthcare System, University of California - Los Angeles, Semel Institute for Neuroscience and Human Behavior, David Geffen School of Medicine, MIRECC 210A, 11301 Wilshire Boulevard, Building 210, Room 116, Los Angeles, CA 90073, USA e-mail: rkern@ucla.edu

W.P. Horan

VA Greater Los Angeles Healthcare System, University of California - Los Angeles, Semel Institute for Neuroscience and Human Behavior, David Geffen School of Medicine, MIRECC 210A, Los Angeles, CA, USA

S.M. Glynn

VA Greater Los Angeles Healthcare System, University of California - Los Angeles, Semel Institute for Neuroscience and Human Behavior, David Geffen School of Medicine, Los Angeles, CA, USA

L.F. Reddy VA Greater Los Angeles Healthcare System, USA

J. Holden University of California, San Diego and Veterans Medical Research Foundation, USA

E. Granholm VA San Diego Healthcare System and University of California, San Diego, USA

L.R. Turner University of California - Los Angeles, Semel Institute for Neuroscience and Human Behavior, David Geffen School of Medicine, Los Angeles, CA, USA

M.E. Sullivan • W.D. Spaulding Department of Psychology, University of Nebraska - Lincoln, USA

Cognitive behavioral therapy
Cognitive behavioral therapy for psychosis
Cognitive remediation
Cognitive remediation therapy
Intensive Case Management
Individual placement and support
Integrated psychological therapy
Multifamily group treatment
Neurocognitive enhancement therapy
Patient outcomes research team
Randomized controlled trials
Substance abuse and mental health services administration
Social cognition interaction training
Social cognition skills training
Supported employment
Severe mental illness
Training in affect recognition
Treatment as usual
Training in community living
Theory of mind
Veterans administration

Introduction

Despite advances in pharmacological approaches to the treatment of people with schizophrenia, it has become clear that medications alone are not sufficient to adaptively function in the community. Even with optimal pharmacological treatment, many affected individuals do not achieve a full remission of psychotic symptoms. In addition to the long-standing challenges associated with treating psychotic symptoms, the treatment of schizophrenia has recently shifted fundamentally from a focus on management and stabilization of psychotic symptoms to the much broader and more ambitious goal of achieving functional recovery. The burning question for the field is whether or not having a severe mental illness, with or without a complete remission of symptoms, necessarily places a limit on the ability to attain a personally meaningful and productive life.

A recovery orientation to psychiatric illness holds that individuals are more than the sum of their symptoms and that recovery involves "a redefinition of one's illness as only one aspect of a multidimensional sense of self, capable of identifying, choosing, and pursuing personally meaningful goals and aspirations." [1] All formal definitions of recovery include criteria to address symptom stability or freedom from psychiatric hospitalization plus some criteria for normalization of social and work/school functioning over a prescribed period of time (e.g., 2–5 years) [2–5]. Recovery-oriented treatments in schizophrenia require identifying and addressing the determinants of poor functioning. Although psychotic symptoms are generally poor predictors of functioning in community dwelling outpatients (although the topic is open to debate), other factors, particularly the cognitive impairments that are so common among people with schizophrenia, are strongly associated with functional outcome and are increasingly addressed in psychosocial treatments [6]. Indeed, psychosocial treatments that enable people with schizophrenia to cope with the disabling aspects of their illness and achieve personal goals are widely regarded as a necessary complement to somatic treatments.

In this chapter, we provide an overview of evidence-based approaches to the psychosocial treatment of people with schizophrenia. Available approaches can be broadly categorized into *rehabilitation*, which includes skills training, assertive community treatment (ACT), cognitive remediation (CR), social cognition training, supported employment (SE), and peer-implemented services, and *psychotherapy*, which includes family-based therapy and cognitive behavioral therapy (CBT). Some of these approaches have existed for many decades (e.g., ACT, skills training, family-based therapy) whereas others have emerged in the past decade or so (e.g., social cognition training). As summarized in Table 14.1, the eight treatment approaches covered in this review also differ on several other dimensions, including treatment targets, format, and duration. For each approach, we provide a brief background on its development, describe the main intervention methods, and summarize results on efficacy. We conclude by discussing the current state of psychosocial treatments and discuss implementation and dissemination obstacles that impede broader service delivery.

Rehabilitation

Skills Training

Background

The idea of *skills training* as an approach to psychiatric or psychological treatment has been integral to the evolution of modern CBT in general, and to psychiatric rehabilitation for schizophrenia in particular (e.g., [7, 8]). In the 1960s "assertive-ness" was first identified as a particular set of behavioral abilities, whose acquisition through "assertiveness training" reduces the emotional and behavioral difficulties associated with anxiety, depression, and related clinical presentations [9]. American psychologist Andrew Salter and South African psychiatrist Joseph Wolpe are generally credited for introducing "assertiveness training" as a psychotherapy technique. Perhaps because both were Pavlovian behaviorists, "skills training" did not take on its contemporary meaning until the 1970s, after the cognitive revolution transformed psychology, and *social learning theory* achieved paradigmatic status [10].

TIME 14-1 AMPT	пагу от рууспозостат и саписит арр	Udulies			
	Treatment target(s)	Format	Duration	Training and set up	PORT recommended
Rehabilitation Skills training	Social skills, symptom and medication management skills, recreation and leisure skills, basic conversation skills, social problem solving ability, workplace	Individual + group	3-6 month modules	Training can take up to 6 months; treatment can be provided by any mental health team member; treatment manuals for specific treatment targets widely available	Yes
Assertive community treatment	Prevention of relapse and rehospitalizations; lengthening community tenure	Multidisciplinary treatment team	12 months or longer	Team members often have necessary skill sets (e.g., case management) prior to implementing the shared team infrastructure, which can take 6	Yes
Supported employment	Competitive employment	Employment specialist	Indefinite	Approximately 6 months to train employment specialists and achieve "good" fidelity levels in most proorgams	Yes
Cognitive remediation	Neurocognitive abilities (attention, memory, reasoning and problem solving, auditory and visual perception)	Individual or group	Dosing and hours of training varies considerably across programs; 3–12 months	Varies between approaches; for strictly computer-based training, requirements are minimal; more integrated programs require longer training for group	Needing further empirical support
Social cognition training	Social cognitive abilities (emotion processing, social cue perception, attributional bias, empathy, Theory of Mind)	Group	3 months	anoging comons 3-6 months of training for group leaders	Not yet considered

Table 14.1 Summary of psychosocial treatment approaches

Peer-implemented services	Weight management, substance use prevention, and illness management have been targeted in peer-delivered educational services; employment rate and job tenure are targets in supported employment	Individual or group	Tied to services implemented	Varies widely depending on role	Needing further empirical support
Psychotherapy Family-based therapy	Prevention of relapse through improving family skills and recovery attitudes	Single family or multifamily groups	6–24 months	6 months including initial training and consultation on going cases	Yes
Cognitive behavioral therapy for psychosis	Treatment-resistant psychotic symptoms, negative symptoms	Individual or group	6–12 months	6-9 months including initial training and consultation on ongoing cases	Yes

In social learning theory, a "skill" became any assemblage of behavioral, cognitive, psychophysiological and/or psychomotor processes or abilities, whether elemental or complex, that is acquired, organized, activated, and performed by a person for an instrumental purpose. In this climate, "assertiveness skills" quickly expanded to include "social skills," and the skill concept soon became applicable to virtually every domain of personal and social functioning. Two landmark applications appeared in 1975, one the original assertiveness concept for neurosis and the other a broader "personal effectiveness" social skills concept for schizophrenia [11, 12]. Today, the versatility of the skills training concept complements the functional, pragmatic imperatives of psychiatric rehabilitation. Virtually any deficit or functional failure associated with schizophrenia could potentially be targeted as a measurable "skill to be learned." Psychiatric disorders may compromise normal developmental acquisition of skills, and may cause skills to be lost, but they do not prohibit acquisition or reacquisition under special skills training conditions.

Description of Treatment

In psychiatric applications, "skills training" is nearly coterminous with "social skills training." This makes sense conceptually—in social learning theory, all skills are "social" skills. However, this terminology can cause confusion. For pragmatic if not conceptual reasons, "social skills" have differentiated into major subcategories of interpersonal, independent living, and illness/wellness management and recovery skills.

Interpersonal skills training focuses on dyadic interactions, from the relatively basic skills involved in simple conversation to more advanced relationship building and intimacy. It relies heavily on structured role-play and practice, conducted in small groups, to acquire (or reacquire) skills needed for a wide variety of common interpersonal situations. A variety of comprehensive clinical manuals (e.g., a guide for social skills training) and packaged materials for training basic to advanced skills (e.g., basic conversational skills, symptom management, workplace fundamentals) are commercially available [13, 14]. It can be administered in either a closed format involving a specific number of group sessions (usually 2–3 per week for 2–3 months) or an open format involving rotation of participants in and out without a uniform "start" or "end" point. The former appears most in controlled research, while the latter is more practical in real-world settings.

Independent living skills training includes basic personal hygiene and self-care, wardrobe maintenance, housekeeping, meal planning and preparation, personal financial management, and use of public resources (e.g., for transportation, recreation). Living skill modalities project basic social skills training techniques into natural settings, usually in a "personal life coach" format suitable for use by (specifically trained) case managers and community support personnel [15–17].

Illness/wellness/recovery skills are what a person uses to manage the illness process itself and generally promote health. They range from simple and concrete, e.g., adhering to a medication regimen, to more complex, e.g., stress management, psychophysiological self-regulation, and the skills used by many people to maintain general health and reduce stress, from yoga to good nutrition [18]. *Coping skills*

is also a highly structured and evidence-based skills training modality within the illnesss/wellness/recovery subcategory [19]. As the recovery concept has evolved, abilities for identifying and systematically pursuing personal desires, life goals, frustrations, and barriers are increasingly addressed in skills training packages.

Tailoring skills training for targeted subgroups has been an important area of advancement. Targeted subgroups include older people with schizophrenia and people early in the course of the illness [20–23]. Skills training is used broadly, crossing national and ethnic boundaries [14, 17, 22, 24–27].

Results on Efficacy

In outcome research on skills training for schizophrenia, "social skills training" has been by far the most common rubric, but pragmatically any modality can only address a subset of skills. Two main issues arise in interpreting treatment outcome research in this area. First, the methodological rigor varies substantially across studies. As in the early years of psychotherapy research, it was initially difficult to compare studies that used different versions of "social skills training." Similarly, overuse of "Treatment As Usual" (TAU, usually case management and medication) as a control condition limited interpretation of differential outcome. These methodological challenges were eventually met through more active comparison conditions (e.g., supportive or occupational therapy) and by the sheer volume of studies. The second issue is that the outcome domain that would be expected to change as a result of participation in social skills training programs has been a topic of some debate [28, 29]. Although the focus of treatment in social skills training programs is developing social and independent living skills, a number of studies, particularly from the earlier literature, examined symptom exacerbation and relapse as the primary outcome. These areas represent a more distal target and can be influenced by a number of factors.

Findings indicate that people with schizophrenia can acquire a variety of precisely defined social and independent living skills through the type of structured behavioral training used in skills training. In the Kurtz and Mueser meta-analysis of 22 social skills training studies that included 1,521 individual subjects, the results revealed a large effect size for acquisition of social skills knowledge (d=1.20), a moderate effect size on social and daily living skills performance-based assessments in the clinic (d=0.52), a moderate effect size for community functioning (d=0.52), and a small effect size for reduction of symptoms and relapse (d=0.23) [30]. The few studies that have included follow-up data have shown retention of acquired skills for up to 1-year post-training.

Efforts to facilitate generalization of acquired skills in the clinic to community functioning are an important and necessary focus of this rehabilitation approach. Do the skills acquired and demonstrated under the structure of a classroom setting under a trainer's supervision and guidance generalize to other settings with other people? Traditionally, homework assignments have been used to facilitate generalization. More recent strategies include having trainers work directly with consumers in the community where they can practice newly learned skills in real-world settings (e.g.,

coffee shop) or the enlistment of family members to help facilitate the transfer of skills. Skills training programs that have used a more direct approach have shown better generalization compared to those using homework assignments [15, 16, 31, 32].

Summary and Future Directions

Skills training approaches have a well-established history in the psychosocial treatment of schizophrenia and have been extensively employed internationally. Several recent trends and advances in skills training are expected to continue in the coming years. First, efforts are underway to integrate skills training with other psychosocial interventions and psychotherapies, such as CR and CBT [33, 34]. Second, there is interest in extending skills training beyond conventional targets to such areas as emotion regulation and work-related skills [35]. Money management is an oft-neglected skill area, but one critical to independent living for adults with schizophrenia, and may have particular relevance for homeless individuals with severe mental illness. This area may be another target for future efforts [36, 37]. Finally, an important new treatment area, social cognition training, is largely based on a skills training model. Given its putative importance for facilitating recovery along with the rapid growth of investigative interest in social cognition and schizophrenia, this training approach is treated separately and appears later in this chapter.

Assertive Community Treatment

Background

With the advent of effective antipsychotic medication and the deinstitutionalization movement in the 1960s and 1970s, it was hoped that most individuals with schizophrenia and other debilitating psychiatric illnesses would be able to live satisfying lives in the community and that the need for psychiatric hospitalizations would be reduced. The US Community Mental Health Centers Act of 1964 established funding for local outpatient facilities where consumers, even those with limited financial resources, could access outpatient psychiatric treatment, case management, and counseling. However, it was not immediately clear how to best offer and coordinate these office-based services to support community tenure. While it was considered possible that many individuals would only need low intensity services, it also appeared likely that more intensive outpatient services might be needed by those who had high levels of persisting symptoms, had a history of long or frequent hospitalizations, or were currently exacerbated.

In the early 1970s, a group of mental health clinicians in Madison, Wisconsin developed an intensive outpatient mental healthcare program, the *Training in Community Living* (TCL), which focused on providing team-based community-housed mental health services as an alternative for individuals seeking an inpatient hospital admission. TCL served as the foundation for ACT. It was grounded in

several principles that have now become core ACT tenets, including meeting consumers' physical and material needs as well as their mental health needs, teaching coping skills, promoting consumer motivation, avoiding dependent relationships, supporting and educating community members who care for or interact with consumers, providing community-based services, focusing on treatment engagement and retention, and assuring consumers do not fall through "service gaps" if additional services are needed. In a 14-month randomized controlled trial (RCT) comparing TCL to regular care in 135 individuals presenting for hospitalization, TCL participants spent less time in the hospital, more time in sheltered employment (rather than being unemployed), had more trusting relationships and higher self-esteem, and reported less symptomatology than those in the traditional care group [38]. However, most of these advantages were lost once TCL support ended, highlighting the importance of long-term community support. The TCL program findings attracted national notice and served as the catalyst for the further development of intensive community support programs such as ACT, which were refined throughout the 1980s and early 1990s [39]. ACT was manualized and is now a component of the SAMHSA evidence-based toolkit initiative (http://store.samhsa.gov/product/Assertive-Community-Treatment-ACT-Evidence-Based-Practices-EBP-KIT/SMA08-434) [40].

Description of Treatment

In many ways, ACT can be understood more as a way to organize interdisciplinary services rather than as a unique psychosocial intervention. The specific content guiding interactions with consumers is less defined, and a concern has been raised that learning principles and specific evidence-based practices have not been fully integrated into the model [16]. Many organizational features are central to how ACT teams are comprised and function, and there are now several ACT fidelity scales [41-43]. Core program features, as operationalized on the ACT fidelity scale in the SAMHSA kit, include the team comprising interdisciplinary members with a small caseload (consumer: provider ratio of 10 or less) which functions as a group (rather than several independent clinicians) and meets frequently (usually daily) to review cases [43]. The team should be stable and fully staffed, overseen by a supervisor who also provides direct services, and include at least one psychiatrist and two nurses for 100 consumers, and substance abuse and vocational specialists. The caseload across the team should be large enough to support the full complement of diverse clinicians necessary to meet the comprehensive needs of consumers. The team should have well-articulated admission criteria, and have an entry pace that assures consumers are easily incorporated. ACT staff should be involved in decisions to hospitalize and discharge consumers when they may need more intensive inpatient services, and should offer 24-h crisis services in house. Services should be time unlimited and both the intensity and frequency of weekly contact should be high between staff members and consumers.

ACT programs are widely available in the USA, especially for targeted populations such as high service users. Much of the current scientific work in this area involves integrating more recovery-oriented techniques and philosophies into the ACT model [44]. ACT was developed prior to the wider acceptance of the principles of the consumer recovery movement such as person-driven care [45]. The program has been criticized for being intrusive, coercive, and grounded in an outdated medical model of psychiatric illness [46]. More recently, several efforts have been made to incorporate more recovery-oriented practices into ACT, such as including consumer (peer) providers and broadening the relevant outcome domains to include ones of particular value to consumers (e.g., graduating out of the ACT program) but not integral to the initial conceptualization of the model [47, 48].

Results on Efficacy

Extensive controlled research on the ACT model has been conducted over the past 35 years; it has been summarized in several review articles and meta-analyses. Mueser et al.'s early review of 32 RCTs and 36 quasi-experimental studies on ACT and Intensive Care Management (ICM) found that these models reduced time in the hospital and improved housing stability, especially among consumers who are high service users, but had moderate effects on improving symptomatology and quality of life [49]. Most studies in their review suggested little effect of ACT and ICM on social functioning, arrests and time spent in jail, or vocational functioning.

A Cochrane meta-analysis supported the Mueser et al. conclusions and found that ACT participants were also more likely to remain in contact with services [50]. ACT reduced the cost of hospital care, but did not have a clear advantage over standard care when other costs were taken into account. A subsequent meta-analysis by Ziguras and Stuart of 44 controlled studies on ACT and clinical case management found that traditional clinical and assertive types of case management were more effective than usual care with regard to family burden, family satisfaction with services, and cost of care [51]. The total number of admissions and the proportion of consumers hospitalized were reduced in ACT compared to traditional case management in the number of hospital days.

Coldwell and Bender conducted a meta-analysis on RCTs and observational studies investigating the benefits of ACT for *homeless* individuals with severe mental illness [52]. They found that those receiving ACT achieved greater reductions in homelessness and greater improvements in psychiatric symptom severity compared with standard case management treatments. In contrast to the findings described above, hospitalization outcomes were not significantly different between the two groups.

Summary and Future Directions

ACT has permitted many individuals with the most severe course of schizophrenia to remain living in the community. Two sets of recent ACT findings have drawn notice. First, the primary established benefit of participation in ACT has been a reduction in hospitalization. However, with the decline in psychiatric hospitalization overall, this benefit has become increasingly limited to individuals with a high rate of baseline hospitalizations and not the larger consumer population [53, 54].

Thus, more recent work in the field has been directed at assessing the benefits of ACT in specialized populations, such as individuals with severe mental illness and concurrent forensic issues, and consumers in a first episode [55–57]. Second, the benefits of ACT in comparison to standard care have been harder to establish in Europe, although there have been some positive findings [58–60]. Several reasons for the contrasting results in the USA and Europe include variations in how the ACT model has been operationalized, differences in basic medical and social services systems (e.g., a greater availability of home-based care in Europe) that may duplicate or obscure condition differences, or the fact that the European studies have been conducted in the context of decreasing rates of psychiatric hospitalization [61]. With changing patterns of psychiatric hospitalizations, and the broader availability of community support in general mental health, it seems clear that further work clarifying for whom the more intensive ACT program is best suited will be especially valuable. Adaptations of the model to incorporate recovery principles and consumer empowerment would be especially timely.

Supported Employment

Background

The benefits and significance of employment for people with severe mental illness are well established. Work provides opportunities for socialization and independence, is associated with reduced levels of psychotic symptoms and re-hospitalization, and is associated with enhanced quality of life and social functioning [62]. Unfortunately, obtaining and holding a job remains a substantial challenge for many people with severe mental illness [63, 64]. Estimates of unemployment range from 65 to 90 % with the highest rates associated with schizophrenia. These figures are particularly striking in light of evidence that approximately 70 % of individuals with severe mental illness report that they would like to have a job.

A number of factors can interfere with efforts to obtain and maintain employment [65]. These include cognitive impairments that interfere with performance of work tasks, positive symptom exacerbations resulting in hospitalization and job absence, social interaction difficulties with co-workers and supervisors, motivational difficulties, stigma from employers, and the fear of losing disability benefits. A variety of vocational rehabilitation approaches have been tested over the past several decades. Initially, sheltered employment or job clubs were commonly utilized, but these services did not typically provide or lead to competitive employment. Subsequently, there was debate about the superiority of two main approaches-the "train-place" model, which moves through a stepwise process involving prejob screening, assessment of job-readiness, training, transitional job placement, and ultimately placement in competitive employment, or the "placetrain" model. Research now consistently supports the latter approach. An example of the place-train model is supported employment (SE), which involves rapid job search with ongoing follow-along vocational and mental health support but bypasses extensive prevocational training [66].
Description of Treatment

SE programs provide people with disabilities support in finding and maintaining employment. The most widely used model of SE, Individual Placement and Support (IPS), was first introduced in the mid-1990s and is now implemented in over 250 programs throughout the USA and Europe [67]. Two key elements of IPS include rapid job search and integration of SE services with delivery of clinical services provided by the consumer's mental health treatment team [68]. A major shift from previous employment-related interventions was the focus on job placement regardless of "job readiness."

The IPS model is based on the following set of principles: (a) zero exclusion criteria (i.e., IPS is appropriate for anyone wanting to work), (b) including work rehabilitation as an integral component of mental health treatment, (c) competitive employment as the primary goal, (d) rapid job search that is based on consumer preferences, and (e) continuous follow-along support which is community-based and provided by an employment specialist [67]. The focus is on obtaining communitybased competitive jobs that pay at least minimum wage and are available to anyone who wishes to apply for them. A strong emphasis is placed on matching the job with the consumer's abilities and interests. Employment specialists actively facilitate job acquisition, establishing working relationships with community employers to optimize a good job match.

SE programs are found in a wide variety of service contexts in the USA and internationally. Common settings include community mental health centers, community rehabilitation programs, clubhouses, and psychiatric rehabilitation centers. IPS has also been evaluated in some consumer subgroups including those receiving Social Security Disability Insurance benefits, younger adults with a recent onset of psychosis, and middle-aged and older adults [69–71].

One recent trend in research settings has been to augment IPS programs with psychosocial interventions that target key barriers to job placement and retention. For example, motivational interviewing has been used as a complementary intervention to help engage consumers who are in early stages of commitment to employment [71]. A few studies have also integrated elements of social skills training and cognitive rehabilitation within the context of IPS [72]. In a new area of research, there have been efforts to extend IPS to address educational goals, particularly those with a relatively recent onset [73, 74].

Results on Efficacy

Research clearly supports the efficacy of SE. Five major reviews and meta-analyses all found that SE led to significantly greater employment rates than alternative approaches with effect sizes ranging from 0.43 to 0.79 [75–79]. A recent meta-analysis of 11 RCTs including high-fidelity IPS programs found that 61 % of IPS participants obtained a competitive job compared with 23 % of participants in alternative vocational rehabilitation conditions [79]. About two-thirds of IPS

participants who obtained competitive employment worked 20 h or more per week; the average time to employment was 138 days for IPS participants compared with 206 days for controls. These findings are similar to those from a multisite RCT involving 312 consumers from six European centers [80]. Studies augmenting standard IPS with additional interventions have yielded promising results as well. Initial studies suggest that consumers receiving SE plus either skills training or cognitive rehabilitation showed better outcomes than those receiving SE alone [72].

Despite clear support for the efficacy of SE, a few caveats from these reviews should be noted. Among consumers who enroll in high-fidelity SE programs, approximately 1/3 are unsuccessful in finding a competitive job. These numbers increase to approximately 55 % when examining IPS programs more broadly. For those who do obtain employment, job tenure is typically brief lasting approximately 6–8 months (11–12 months in high-fidelity programs). Also, earnings are typically quite modest, and unsatisfactory job terminations are not uncommon. Consumer variables have been examined to help understand who is most likely to benefit from SE. Better work history predicted higher job attainment (though not retention) and receiving disability benefits predicted lower job retention (though not attainment) [81]. A large range of clinical and demographic variables, including substance abuse, educational background, disability income, hospitalization history, and symptomatology, showed no relation to employment outcome. Hence, benefits from IPS appear to extend to a broad range of consumers.

Summary and Future Directions

Employment is a critical element of recovery and there is now substantial support for the efficacy of SE on job acquisition. However, the findings on job tenure are more modest, particularly when one considers the median length of employment for SMI individuals compared to national averages of individuals in comparable jobs at the same age level (3–5 years; US Bureau of Labor Statistics) [82]. Efforts to integrate IPS with other psychosocial interventions may target job tenure more directly (e.g., augmenting IPS with cognitive rehabilitation and/or social cognition training). Lastly, an area likely to see continued growth is supported education.

Cognitive Remediation

Background

In the past 40 years numerous CR programs have been developed and evaluated. This approach is motivated by the substantial evidence that neurocognitive impairments are core features of schizophrenia [83]. Schizophrenia is associated with a range of cognitive impairments, which include attention, learning and memory, working memory, speed of processing, and reasoning and problem

solving among others. Furthermore, neurocognitive deficits show consistent relations to community functioning, instrumental living skills, and rehabilitation outcome [83, 84].

Description of Treatment

Recent reviews indicate that there are as many as 14 CR approaches, although arguably they have more similarities than differences. Across modalities, the aim of CR is to improve community functioning, although some argue that improvement in cognition alone is a worthwhile goal and treatment endpoint [85]. Among the various CR approaches, a broad distinction can be made between cognition-enhancing and compensatory approaches. Cognition-enhancing approaches aim to improve cognitive functioning through stimulation of impaired areas of cognition (e.g., working memory). Originally, this approach was based on the assumption that cognitive impairment could be restored through activation and repetitive exercise. More recently, this approach is based on discoveries that support the brain's neuroplasticity, referring to the brain's lifelong capacity for physical and functional change. These approaches directly target cognitive impairment as a means by which to improve functioning. In contrast, compensatory approaches aim to bypass or "compensate" for cognitive impairments by devising training methods that emphasize recruitment of relatively intact cognitive processes or reduce the burden on affected ones often by establishing supports or prosthetic aids in the environment. These approaches directly target functional deficits but with consideration of the cognitive impairments which limit or impede the ability to carry out those functions.

Among cognition-enhancing approaches, the most common method is "drill and practice" which involves the use of repetitive practice or training in increasingly difficult cognitive exercises. Exercises may be titrated to the individual's level, so that they are neither too challenging nor too easy. Difficulty level is increased in a stepwise manner as performance improves. Drill-and-practice exercises are frequently computerized although they can be administered by using paper-and-pencil exercises. These interventions have been used in isolation or in the context of multifaceted packages that include complementary psychosocial interventions.

There are two primary approaches to drill-and-practice CR. "Bottom-up" approaches are based on the assumption that there is a basic deficit in the ability to discriminate signal from noise in auditory and visual processing. This basic deficit is theorized to be the cause of higher-level cognitive impairments (i.e., planning, problem solving, memory, attention), and is targeted through high intensity training exercises. Software developed by PositScience is one commonly used approach that targets auditory and visual processing in computerized training sessions, typically administered daily over a period of several weeks totaling 50–100 h [86, 87]. "Top-down" approaches target higher-order cognitive functions (i.e., attention, memory, and executive functioning) through complex, consciously effortful exercises that place demands on multiple cognitive abilities. Top-down approaches involving

computer-based training share similar methods with bottom-up approaches. The primary difference is the target of training. Both involve standardized computerbased curricula that utilize intense and repetitive practice and a lengthy duration of training with defined levels of mastery. Examples of top-down approaches include Neuropsychological Educational Approach to Remediation (NEAR), Neurocognitive Enhancement Therapy (NET) and CogPack [33, 88, 89]. These comprehensive programs involve training exercises across a broad range of cognitive functions, including attention and concentration, psychomotor speed, learning and memory, and executive functions. Both of these programs have been used in the context of multifaceted treatment programs that incorporate vocational rehabilitation.

In addition to drill and practice, strategy coaching has been used to teach individuals to apply new cognitive techniques learned through training exercises to areas of real-world functioning. For example, cognitive remediation therapy (CRT) is an individualized 40-h program that uses drill and practice to enhance working memory, planning, and problem solving along with an emphasis on strategy coaching [90]. The strategy coaching component relies on the therapist's emphasis on and discussion of the relevance of new cognitive skills to community functioning. For example, a therapist may provide coaching about how to use recently acquired memory (e.g., list learning) and categorization skills (e.g., sorting by taxonomic category) while grocery shopping. Therapists continuously encourage participants to reflect on how the trained cognitive skills might be used in a number of real-world settings [91].

In contrast to cognition-enhancing approaches, compensatory approaches focus on compensatory strategies to reduce overall cognitive load and minimize the functional impairment caused by specific cognitive deficits [92, 93]. The most common compensatory approaches are ones that manipulate environmental demands and errorless learning. Environmental adaptation involves using cues to direct the consumer's attention and thereby augment performance on goal-oriented tasks. An example of this type of approach is Cognitive Adaptation Training (CAT), a manualized treatment in which a caregiver or mental health provider tailors environmental supports to the needs of each individual to increase behaviors such as medication adherence and self-care skills [93]. Commonly used environmental supports in the consumer's home environment may include pill containers with alarms to facilitate medication adherence, and activity checklists to provide reminders about clinic appointments [94].

Errorless learning is a separate approach that compensates for difficulties associated with error commission and concomitant filtering of relevant from irrelevant information observed in schizophrenia. Training involves preventing (or minimizing to the extent possible) the commission of errors during the learning of a new skill or behavior. Training proceeds stepwise in a hierarchical manner beginning with the most basic, simple task components to the eventual integration of all complex aspects, enabling the participant to master the task without making mistakes in the process [95–98]. In errorless learning, the need to self-correct is bypassed, and processing burden is believed to predominantly shift from explicit to implicit memory processes, of which selected components are relatively intact in schizophrenia. Errorless learning has been used in a variety of settings to facilitate acquisition of work and social skills.

Results on Efficacy

Several meta-analyses and review papers have summarized CR approaches [29, 99–105]. Recently, Wykes and colleagues conducted a comprehensive meta-analysis including 40 RCTs of drill-and-practice or drill plus strategy coaching interventions [105]. Results were wholly consistent with a previously published meta-analysis finding small to medium mean effect sizes for improvements on cognitive measures that were not directly involved in training exercises [104]. Recent evidence also documents changes in brain systems targeted by drill-and-practice interventions [106].

With regard to generalization to improvements in functional outcome, an important factor is whether CR is administered alone or in conjunction with other psychosocial rehabilitation approaches. Research consistently shows the effects of drill-and-practice interventions on functional outcome are largely moderated by whether they are administered along with adjunctive psychosocial rehabilitation. Across 26 studies involving 1,151 consumers the mean effect size for improvement in functional outcome was 0.36 [104]. However, when only studies with adjunctive psychosocial rehabilitation were included, the effect size rose to 0.47; studies that did not include additional psychosocial rehabilitation interventions had a mean effect size of 0.05.

Although substantially fewer randomized clinical studies have evaluated compensatory interventions, they have also been found to show significant treatment benefits. A series of studies indicate that CAT improves medication adherence and community functioning, and decreases symptom relapse rates [93, 107]. Similarly, errorless learning is associated with improvements in performance on entry-level job tasks, social problem-solving ability, and assigned job tasks at a community mental health setting, with gains maintained up to 3 months later without further intervention [96, 108, 109].

Summary and Future Directions

Research on CR has substantially expanded in concert with growing awareness of the impact of neurocognitive impairments on functioning. Evidence indicates that drill-and-practice interventions lead to improvements in functioning when implemented in conjunction with more comprehensive psychosocial rehabilitation approaches (e.g., skills training, SE). Also, strategy coaching appears to lead to better community functioning than drill-and-practice alone. Compensatory approaches, though less extensively studied, also show potential for improving specifically targeted areas of functioning. Medalia and Saperstein recently provided suggestions for continued development and evaluation of CR [110]. Suggestions included (a) measuring outcomes in terms of consumers' personal goals (as opposed to standard competency and capacity measures) and (b) expanding CR to include problem-solving and planning skills to facilitate the use of improved cognitive abilities in the service of daily living activities.

Social Cognition Training

Background

Research into the social aspect of cognition in schizophrenia is relatively new but has expanded rapidly over the past decade. Social cognition refers to cognitive processes that are involved in perceiving and understanding social situations, particularly the behaviors, emotions, and intentions of other people. Schizophrenia is associated with impairment in several domains of social cognition, with a mean overall effect size of 0.69 compared to healthy controls [111, 112]. Importantly, social cognition deficits show stronger relations to community functioning than neurocognition or clinical symptoms [113–115]. These findings have generated interest in intervening at the level of social cognition as a means of improving functioning. Early evidence indicates that social cognition impairments are amenable to psychosocial intervention.

Description of Treatment

Social cognition domains include emotion perception (facial affect identification), social perception, social knowledge, Theory of Mind (ToM), attributional bias, and empathy [116]. Social cognition interventions commonly follow the skills training model. Training is conducted in group format, and incorporates didactic skills training and experiential role-play and rehearsal. Standard behavioral techniques, such as modeling, shaping, providing behavioral cues/prompts, and repetition, under a rich schedule of positive social reinforcement, are intrinsic to social cognition training. Participants are trained on perceiving, identifying, interpreting, and responding to social cues and affective information. Training proceeds from simple to complex exercises, with later stages incorporating ambiguous and emotionally evocative stimuli to mimic real-life interpersonal situations (e.g., beginning with happy and sad faces and progressing to fear, anger, disgust, and shame-inducing situations). A distinction can be made between targeted treatments, which focus on one or more social cognitive domains, and broad-based treatments, which include social cognitive training exercises within multifaceted treatment programs.

Early efforts involving targeted treatments focused on emotion perception. There were several short-term interventions (1–5 sessions) that trained participants to use effective face-scanning techniques and repetition and coaching to identify the forehead, eye, and mouth movements that are characteristic of various emotions [117–120]. An example of an emotion-focused intervention is Training in Affect Recognition (TAR), a 12-session manualized approach consisting of three segments: identifying prototypical components of basic emotions, integrating facial cues to form quick decisions about affect, and applying the learned information to the processing of ambiguous facial expressions [121–123]. Training consists of practice with computerized stimuli as well as learning compensatory strategies such as verbalization and self-instruction.

As research in this area progressed, more comprehensive programs that address multiple social cognition domains were developed. These programs included 20 or more sessions and targeted emotion perception, ToM, and social perception, with some addressing higher-order social cognitive processes such as attributional style and deception as well. For example, Social Cognition Interaction Training (SCIT) is a manualized, group-based, 20–24 session treatment, targeting affect recognition, attributional style, and ToM [124]. It is divided into three discrete sections: identifying and understanding basic emotions and paranoia, learning strategies to avoid the tendency to "jump to conclusions" in social situations, and applying newly acquired social cognitive skills to real-life situations. Another approach, Social Cognition Skills Training (SCST), combines and augments elements of SCIT and TAR to address emotion recognition, social cue perception, attributional bias, and ToM (e.g., distinguishing among sarcasm, white lies, and sincerity) [125, 126].

Social cognition has also been included as a target of intervention in broad-based treatments that concomitantly address other areas, such as nonsocial cognition and employment. For example, cognitive enhancement therapy (CET) uses a group skills training format to address social information processing, emphasizing accurate social context appreciation and perspective-taking, in addition to a rigorous program of computerized neurocognition training administered to consumer dyads [127]. Similarly, integrated psychological therapy (IPT) includes a 21-session social and emotion perception module that is embedded within a broader group-based neurocognition training intervention [128]. IPT contains five subprograms, administered sequentially, totaling approximately 8–12 months of biweekly group sessions that target neurocognition training as a necessary prerequisite to training on higher cognitive functions such as social cognition and social skills.

Results on Efficacy

Early evidence supports the efficacy of social cognition training, at least on selected, most commonly lower level areas of processing [129–131]. Horan and colleagues concluded that social cognition is responsive to systematic intervention, but stressed the importance of expanding beyond emotion perception and developing interventions for more complex areas such as conveying empathy and detecting sarcasm and hostility. Kurtz and Richardson's meta-analysis included 19 RCTs and found medium to large treatment effect sizes for performance on measures of affect recognition, medium effect sizes for ToM, and no clinically meaningful effects for social perception or attributional bias. Six studies assessed generalization to functional or institutional outcome and revealed an overall mean effect size of 0.78.

Although the early results of social cognition training efforts are promising, caveats include the relatively small samples in most studies, inclusion of inpatient and outpatient studies (effects were much larger in inpatients), large variability in intervention methods and outcome measures, and minimal consideration of the durability of training effects. In addition, a challenge in interpreting these studies has been the difficulty in determining how much, if any, treatment benefits are

associated with social cognition exercises per se rather than general improvements in nonsocial cognition, particularly in broad-based treatments that include CR. The few studies that have directly addressed this issue suggest that gains in social cognition are not necessarily dependent on improvements in neurocognition, challenging the notion that neurocognition training is a prerequisite building block for achieving gains in social cognition [126, 132].

Summary and Future Directions

Although social cognition is a relatively new area for treatment development, emerging evidence suggests efficacy of training for selected areas of social cognition (typically lower-order skills), and there is some evidence that these gains may confer improvements in real-world functioning. Much work remains to be done in further refining these training intervention programs, identifying the most potent and efficient training methods, improving their effects on higher-order social cognitive domains, and demonstrating both durability and generalization to community functioning. A separate issue particularly germane to social cognition is the need to develop or refine psychometrically sound assessment measures that can be used across treatment trials. Such efforts are currently underway [133].

Peer-Implemented Services

Background

Peer support and peer-provided services are a rapidly expanding component of the psychosocial recovery and rehabilitation toolbox. Peer providers are individuals who have achieved functional recovery from a mental disorder, continue to receive mental health services themselves, and are employed by mental health programs to work alongside professionally prepared staff members [134]. The peer movement is based on the premise that persons with lived experience can offer support, respect, shared responsibility, guidance, and wisdom that cannot be offered by a professional [135]. The peer support approach to psychosocial rehabilitation includes consumer-run programs as well as the employment of peers within traditional mental health services. The peer-services approach has its roots in the recovery movement, becoming more widely accepted and supported by the Supreme Court and the President's New Freedom Commission on Mental Health in the late 1990s [136].

Description of Treatment

Consumer-run programs are prevalent across the USA, although the majority of these are support groups and self-help organizations. A 2002 national study found that of the 7,467 groups, organizations, and services run by and for mental health

consumers, 1,133 (15.2 %) were identified as consumer-operated services [136]. Consumer-operated services include drop-in centers, mutual support groups, peer educator and advocacy programs, multi-service agencies with benefits counseling and case-management, peer phone services, and specialized supportive services focusing on crisis respite, employment, and housing [137].

Within traditional mental health services, roles for peer providers have also grown exponentially in the past 10 years. Peer provider roles include case manager aides, community aides that connect hospitalized consumers to continuing outpatient services, counselors and advocates, outreach workers, providers of self-help educational services, and vocational counselors [138]. Formal training for peer specialists is a central component of effective peer-provided services. Certified Peer Specialists (also called Peer Support Specialists) are employed in numerous capacities, which most frequently include serving on ACT teams, in peer-support programs, as counselors and advocates in psychiatric inpatient facilities, and as group leaders or providers of one-on-one services in community mental health systems of care [139].

Results on Efficacy

The majority of published research has focused on the roles of peers in traditional mental health service settings. Qualitative and non-controlled studies on the use of peers as Peer Support Specialists and case management aides indicate comparable benefits for individuals with SMI receiving services from peer vs. non-peer providers [140–143]. Research also indicates working as a Peer Support Specialist confers benefits for the peer-provider, with improvements in job satisfaction as well as decreased need for mental health services [144-146]. The few RCT published in this area have shown positive results. Consumers who received services from peers serving as assistants to case managers resulted in increased service engagement and improvements in the level of social participation and quality of life [147, 148]. Furthermore, recent studies show that peer-led educational training on mental illness self-management led to reduced levels of depression and anxiety, increased self-perceived recovery, and increased empowerment, self-esteem, and self-advocacy-assertiveness [149–151]. These results suggest that not only do objective measures of clinical symptoms and autonomy improve as a result of peer-led services, so too do more subjective measures of empowerment and self-efficacy.

To illustrate an example of peers as implementers of rehabilitation services, we describe a recently conducted community-based study of IPS SE in which three peers were trained to assume the role of employment specialists [134]. As noted earlier in this chapter, IPS emphasizes an integrative, collaborative working relationship among clinicians, employment specialists, employers, and consumers with the aim of helping the consumer attain his/her vocational goals. The duties of an employment specialist under this model are multifaceted, requiring comprehensive vocational assessments of each consumer, initiating a rapid job or school search, and continued monitoring, support and problem solving after job or school placement. The results from this study showed that peers could be trained to implement

evidence-based practices with a modest degree of fidelity. Importantly, the peerimplemented program yielded employment rates over the span of the study comparable to other IPS SE programs in the region. Of note, all three peer employment specialists received promotions within the county department of mental health and remain with the program 5 years later.

Summary and Future Directions

Peer-providers are an important and promising addition to recovery-oriented services for adults with schizophrenia and other forms of SMI. The three main roles for peers are leading self-help/support groups, running consumer-operated programs, and working as trained peer-providers in traditional services. A new role deserving further examination is the use of peers as implementers of evidence-based practices. Peer-provided services are not only cost-effective—they also create roles and responsibilities for people in recovery from mental illness, and may augment the benefits attained through traditional rehabilitation services. Although the involvement of peers in service delivery has progressed substantially in recent years, there remain few RCTs to document the efficacy of their involvement in treatment and rehabilitation services.

Psychotherapy

Family-Based Services

Background

There is a large and robust literature supporting the benefits of family therapy in reducing relapses and hospital readmissions for adolescents and adults with schizophrenia [152, 153]. The impetus for much of this work was the finding that the family environment played a significant role in influencing outcomes after a hospitalization, with consumers whose relatives expressed high levels of critical comments and/or report high levels of self-sacrificing behavior at the time of a symptom exacerbation (i.e., who are "high" in expressed emotion) having a significantly greater likelihood of relapse within the subsequent 9 months [154, 155]. An additional rationale for these interventions emanates from the recognition that schizophrenia can be especially debilitating, and relatives often assume responsibility for caring for their ill loved ones for extended periods of time [156, 157]. Interventions designed to increase coping skills of consumers and supportive behaviors of relatives simultaneously can be particularly efficient in improving outcomes.

Family involvement in mental health services for those individuals with SMI is associated with greater treatment retention and increased consumer satisfaction [158]. Increasing evidence confirms that many SMI individuals *want* their family

involved in their care [159, 160]. In fact, family involvement in care to support recovery from SMI is a core principle in the President's New Freedom Commission on Mental Health, which states that "services and treatments must be consumer and family centered." Furthermore, the SAMHSA working definition of recovery articulates two guiding principles of special relevance to consumers in mental health treatment and their loved ones: "Recovery is supported through relationships and social networks," and "Recovery involves individual, family, and community strengths and responsibility."

Description of Treatment

A range of family-based interventions has been found to be effective in reducing relapse in schizophrenia. Most of these interventions are cognitive behavioral; they share a number of core components, including illness education, development of a relapse prevention plan, and formal problem-solving skills training, but they vary on several dimensions including duration, the presence of consumers in the family sessions, single vs. multifamily group modality, and relative emphasis on formal behavioral rehearsal as a key component of the treatment [161]. The interventions are designed to be embedded in a comprehensive treatment program and are open to a wide range of relatives, including partners, parents, adult children, siblings, and non-kin supporters. Although in practice these interventions are used with individuals in almost any phase of illness, in the research trials described below the interventions were usually begun either during or immediately after a symptomatic exacerbation, with the outcome of interest being delay or elimination of subsequent relapse.

The two most widely implemented and studied family programs in schizophrenia are Behavioral Family Therapy (BFT) and Multifamily Group (MFG) Treatment [162–164]. BFT is offered to single families; typically, 20–25 one-hour conjoint sessions are conducted across 9-12 months on a declining contact basis, with monthly boosters offered in year two. After orientation and an assessment with each participant, the active intervention begins with education on the psychiatric illness-symptoms, causes, etiology, prognosis, and treatment-to promote recovery attitudes and attainable expectations in both the consumer and relatives. Following this education, participants are provided with several sessions of instruction on communication skills and problems solving. BFT incorporates many behavioral techniques-modeling, shaping, prompting, positive reinforcement, and programmed generalization-to help relatives compensate for consumer attentional difficulties and to assure that family members together can successfully address stressors and achieve goals to improve outcomes with the illness. BFT was developed as a home-based program, but a subsequent trial comparing BFT to customary care in 41 military veterans with schizophrenia substantiated the efficacy of offering BFT in the clinic [162, 165].

There has been considerable interest in determining whether family interventions conducted with *groups* of relatives may confer more benefits than those conducted

with single families, because groups offer more opportunities to obtain social support and normalize responses to the development of a psychiatric illness in a loved one. MFG includes many of the components of the single family intervention discussed above, including active engagement, illness education (typically, an extended educational workshop), and problem solving. In contrast to the BFT, MFG deemphasizes formal behavioral rehearsal and programmed generalization and does not include a unique communication skills component; significant attention is directed at strengthening social bonds among participants. Both consumers and relatives typically attend biweekly MFG meetings. A typical course of treatment would be at least a year.

Results on Efficacy

Several meta-analyses on the impact of family interventions sharing the features described above for schizophrenia have been conducted, with positive results. Pitschel-Walz et al. compared the impact of family interventions to customary care on symptom relapse and reported a mean effect size of 0.19 for first year outcomes and 0.25 for second year outcomes, with treatments lasting less than 3 months having smaller mean effect sizes than longer treatments (0.14 vs. 0.30) [166]. Individual and group treatments had comparable results. Pfammatter et al. compared family interventions to standard care and reported an effect size of 0.42 for relapse rates and 0.22 for rehospitalizations from 6 to 12 months post-study entry, and 0.51 for a reduction in rehospitalizations at 18–24 months [167]. Other meta-analyses on 1-year results reported similar positive findings [168, 169].

Matching interventions to the specific characteristics of consumers and their families is an understudied area. Many published family intervention studies are insufficiently powered to test whether subgroups of participants benefit differentially. There are a few studies that have attempted to examine the issue of differential benefit more directly [170]. Unfortunately, the guidance accruing from these studies to assist consumers, families, and professionals in family-based treatment planning is limited and sometimes contradictory. At this point, it seems the most prudent course of action would be to allow consumer and relative preference drive the choice of modality.

Summary and Future Directions

Several decades of research document the efficacy of family interventions for reducing symptom relapses and rehospitalizations. The benefits of involving relatives in care and providing education and problem-solving training are widely supported in psychosocial intervention studies for schizophrenia. Further, the President's New Freedom Commission on Mental Health highlights the importance of services and treatments being family centered. Unfortunately, implementation is the significant challenge, and innovative strategies are needed to improve access and uptake. To address these

challenges, recent efforts have moved to explore the feasibility and efficacy of Internet-based services. Making family-based therapies more personalized and tailored to the characteristics of individual consumers and their family members is an area requiring further investigation.

Cognitive Behavior Therapy

Background

The premise of CBT is that cognition, the process of acquiring information and forming beliefs, can influence feelings and behavior. Through CBT, individuals learn to identify and modify thoughts, misinterpretations of experiences, and thinking patterns that contribute to maladaptive behavior or distressing feelings. CBT for psychosis (CBTp) developed out of work by Aaron T. Beck as early as the 1950s. Since then, CBT has been widely used to treat mood and anxiety disorders, but the use of CBT to treat psychotic disorders was not as readily accepted, due to skepticism that SMI could be successfully treated through psychotherapy [171]. Although Beck published a case report describing the use of CBT to treat a consumer with a paranoid delusion in 1952 [172], followed by additional reports in the 1970s [173, 174], it was not until the 1990s that controlled trials of CBTp started to emerge, primarily in the United Kingdom.

Description of Treatment

CBTp includes a range of approaches with a variety of treatment elements [175, 176]. Core elements of CBTp include the collaborative identification of target problems or symptoms, development of specific cognitive and behavioral strategies to cope with these problems or symptoms, and testing of key beliefs that maintain delusional thinking and distressing beliefs about voices. Other elements may include a collaborative understanding of the nature of the illness, identification of factors exacerbating symptoms, and problem solving to reduce relapse and achieve recovery goals.

CBTp assumes: (a) hallucinations and delusions are not fixed and can be modified as a target for intervention; (b) situational factors can exacerbate or ameliorate psychotic symptoms; and (c) the content of psychotic symptoms can be meaningful when placed in the context of the individual's prior experiences and beliefs [177]. A primary goal of CBTp is to help consumers become more cognitively flexible, change the way they evaluate their symptoms and thoughts, and consider alternative explanations for psychotic experiences (e.g., objective distancing and reappraisal of psychotic symptoms by examining evidence). Exercises and homework assignments (e.g., thought records and experiments) are used to examine the evidence for and against thoughts and interpretations of events and symptoms. For example, an experiment involving an activity that exacerbates (e.g., imagining a stressor) or ameliorates (e.g., humming) hallucinations might be conducted to challenge beliefs about the power or control of voices to reduce distress or compliance with command hallucinations. By re-evaluating evidence and considering alternative explanations for psychotic experiences, positive symptom severity, distress, and dysfunction may be reduced in CBTp.

The majority of CBTp approaches have focused on reducing positive symptom severity in people with schizophrenia who have persistent psychotic symptoms despite adequate pharmacotherapy. Although positive symptoms remain the most common treatment target, more recent trials have targeted other outcomes. For example, CBTp interventions have been bundled with other approaches, like social skills training for functioning [34], anxiety interventions for social phobia [178, 179], and motivational interviewing for substance use [180]. CBTp has also been modified to specifically target negative symptoms with some success [181, 182].

Results on Efficacy

Significant improvements in symptoms and other outcomes have been consistently found in clinical trials of CBTp in schizophrenia, with medium or large effects found relative to standard care and small effects found relative to other active psychosocial treatments. Benefits have been found for severity of positive, negative, and overall symptoms, as well as depression, suicidality, social functioning, and social anxiety [171, 175, 183–187]. In a meta-analysis by Wykes et al. [175], effect sizes for a variety of outcomes were remarkably similar to the effect size for positive symptoms (typically the primary treatment target; d=0.37), including social anxiety (d=0.35), negative symptoms (d=0.44), mood (d=0.36), and functioning (d=0.38). There is also preliminary evidence that CBTp is effective as part of an early intervention approach in recent onset schizophrenia [188, 189], as well as a possible approach to preventing psychosis in prodromal/high-risk populations [190], with more than 50 % reduction in risk of developing psychosis found in a meta-analysis of seven trials of people who were receiving CBTp and not taking antipsychotic medication [191].

Some meta-analyses, however, have found minimal or no benefit for CBTp when only highly controlled trials (e.g., blinded) with active psychosocial control conditions were included [192–194]. As is generally true in clinical trials, smaller effect sizes have been found for trials with more rigorous controls and active comparison groups [175, 193]. Importantly, greater benefits have been found at 6-month to 5-year follow-ups than at the end of treatment [183, 184, 187, 195] indicating that the benefits of CBTp continue to accrue over time. This finding is in contrast to many other psychotherapies and psychosocial rehabilitation interventions where there is concern over durability of treatment effects. Therefore, meta-analyses focused only on effects at the end of treatment in well-controlled trials with active control interventions fail to capture the lasting benefits associated with CBTp.

Only a few studies have examined factors that predict outcome in CBTp and results have thus far been mixed. Regarding consumer characteristics, some studies

report that better positive symptom outcomes are associated with lower delusional conviction [196, 197], greater insight [198, 199], less severe symptoms [200] (but see [196, 201]), shorter duration of illness [200–203] (but see [34, 204]), female relative to male consumers [197], and white relative to non-white consumers [205]. Greater severity of neurocognitive impairment has not been found to moderate outcome in CBT for schizophrenia, indicating that impaired neurocognition does not preclude treatment benefits [196, 206, 207].

Regarding therapy format characteristics, a meta-analysis by Wykes et al. [175] found better outcomes in CBT programs with greater behavioral emphasis, suggesting behavioral components (e.g., experiment activities to test beliefs, taking action toward recovery goal steps) are at least as important as cognitive interventions (e.g., development of a shared conceptualization of the nature of illness). They also reported no difference in outcome between studies that used individual vs. group formats. A recent meta-analysis found that therapies that were 20 sessions or longer had better outcomes than shorter approaches [184] (but see [183]). Finally, there is evidence from effectiveness trials that CBTp delivered in community mental health settings improves outcomes [201, 208, 209], and community therapists can deliver CBTp with adequate fidelity [210].

Summary and Future Directions

CBT has proven effective for a number of mental disorders, including schizophrenia. Increasingly, best practice guidelines and nationalized healthcare systems are recommending or mandating CBTp [211]. Importantly, CBT has been shown to be effective at improving a number of outcome areas with benefits observed in psychiatric symptoms, mood, anxiety, and social functioning. It also has shown strong durability with continued benefits observed long after completion of the active therapeutic intervention. Not surprisingly, the behavioral element of CBT appears to be an important element in determining treatment success. Despite its efficacy and recommended use, CBTp is largely unavailable in the USA and implementation barriers in other countries limit access to CBTp. The dissemination and implementation of evidence-based practices like CBTp into real-world clinical practice is an important focus looking forward. Additional remaining challenges include treating serious comorbidities, such as substance abuse and other conditions that limit recovery, and treating special populations, such as racial minorities, older consumers, and those at high-risk for the development of psychosis [186].

Conclusion

As the treatment of schizophrenia has shifted toward the ambitious goal of functional recovery it has become clear that psychosocial interventions are an essential component of contemporary treatment. In this chapter we reviewed eight leading rehabilitation and psychotherapy approaches. Some are supported by extensive efficacy data whereas other more recent approaches, although promising, require further development and evaluation. In this concluding section we briefly summarize the status of these interventions as empirically based treatments recommended for best clinical practices, and consider implementation (what it takes to deliver the service) and dissemination (what it takes to deliver the service to vast numbers of consumers) issues.

Regarding the evidence base on efficacious treatments, the 2009 report from the Schizophrenia Patient Outcomes Research Team (PORT [176]) comprehensively reviewed the treatment outcome literature and, using research-based standards, recommended the following treatments: ACT, SE, CBT, skills training, and family-based services among others not reviewed in this chapter (token economy, interventions for alcohol and substance use disorders, interventions for weight management). CR and peer-delivered services were seen as areas of interest but were not recommended treatments because of lack of evidence. Social cognition training was not reviewed at that time but likely will be considered in future PORT updates given the rapid expansion of research in this area.

Unfortunately, despite the existence of well-documented evidence-based psychosocial treatments, implementation issues are a nontrivial concern for most of these interventions. They vary considerably in terms of the certification requirements for service providers, duration of treatment, and human resources needed to implement the treatments with high levels of fidelity (see Table 14.1). For example, treatments like CBT require considerable training to reach professional standards. In contrast, skills training can be led by any mental health team member and requires less extensive training to reliably implement. A separate consideration is that duration of treatment can be lengthy for many of these programs (ranging from 3 to 12 months or longer). Without sufficient staffing, lengthy treatment duration ties up staff supplying these services and limits availability to other "wait-listed" consumers.

Dissemination is an even larger problem. Surveys indicate that evidence-based practices are not broadly disseminated in the USA. Epidemiological data from the National Comorbidity Study in the 1990s found that 60 % of SMI individuals received no treatment, 25 % received inadequate treatment, and only 15 % received minimally adequate treatment [212]. Data indicate that as many as 95 % of individuals with schizophrenia receive no care or less than adequate care [213]. Furthermore, some studies indicate that quality of care may be worsening despite the development of better treatments [214–216].

What obstacles stand in the way? First, state and federal government policies play an enormous role in promoting (or curtailing) which treatment services are vs. are not made available to individuals with severe mental illness through the availability of funds. Second, in recent years smaller mental health budgets at the state and county level have created a forced prioritization of clinical and rehabilitation services with some services dropped due to lack of funds to support resources needed for these programs. Third, a related financial problem is the priority on providing services that can be billed and reimbursed (e.g., meeting Medicaid regulations for a reimbursable clinical service). On the positive side, there is recognition of the value and need to implement evidence-based practices by state mental health authorities in most states. Interventions such as ACT, CBT, skills training, and SE can be found in most regions. Also, the Veterans Administration has sponsored an initiative to disseminate SE in VA hospitals throughout the USA. Despite these advances, consideration of methods by which to improve dissemination clearly deserves continued attention and some frameworks have been proposed. According to Drake et al. [213], "Public mental health systems need better alignment between evidencebased practices and payments, sufficient funding to create a sustainable and professional workforce, electronic medical records to monitor process and outcomes, and a systemic commitment to quality."

To conclude, our understanding of the determinants of poor functional outcome has advanced considerably in recent years. These findings have been useful in guiding the development of a number of promising new rehabilitation and psychotherapeutic approaches to address these determinants. As a whole, the field is developing therapeutic tools that show considerable promise for enabling people with schizophrenia to achieve personally meaningful goals and achieve recovery despite the challenges posed by living with a severe mental illness. As these treatment development efforts continue to move forward, we are clearly at a point where making these treatments available to all who can benefit from them needs to become a priority.

References

- Davidson L. Recovery, self management and the expert patient—changing the culture of mental health from a UK perspective. J Ment Health. 2005;14:25–35.
- Harding CM, Brooks GW, Ashikaga T, Strauss JS, Breier A. The Vermont longitudinal study of persons with severe mental illness: II. Long-term outcome of subjects who retrospectively met DSM-III criteria for schizophrenia. Am J Psychiatry. 1987;144:727–35.
- 3. Liberman RP, Kopelowicz A, Ventura J, Gutkind D. Operational criteria and factors related to recovery from schizophrenia. Int Rev Psychiatry. 2002;14:256–72.
- Torgalsboen AK, Rund BR. Lessons learned from three studies of recovery form schizophrenia. Int Rev Psychiatry. 2002;14:312–7.
- Whitehorn D, Brown J, Richard J, Rui Q, Kopala L. Multiple dimensions of recovery in early psychosis. Int Rev Psychiatry. 2002;14:273–83.
- 6. Green MF, Horan WP. Social cognition in schizophrenia. Curr Dir Psychol Sci. 2010;19:243-8.
- 7. Liberman R. Recovery from disability: manual of psychiatric rehabilitation. Washington, DC: American Psychiatric Press; 2008.
- Spaulding WD, Sullivan M, Poland J. Treatment and rehabilitation of severe mental illness. New York, NY: Guilford; 2003.
- 9. Wolpe J, Lazarus A. Behavior therapy techniques. New York, NY: Pergamon; 1966.
- 10. Bandura A. Social learning theory. New York, NY: General Learning Press; 1977.
- 11. Smith M. When I, say no I feel guilty. New York, NY: Bantam Books; 1975.
- Liberman RP, King LW, Derisi WJ. Personal effectiveness: guiding patients to improve their social skills. Champaign, IL: Research Press; 1975.
- Bellack A, Mueser K, Gingerich S, Agresta J. Social skills training for schizophrenia: a stepby-step guide. 2nd ed. New York, NY: Guilford Press; 2004.

14 Psychosocial Rehabilitation and Psychotherapy Approaches

- 14. Liberman R. Dissemination and adoption of social skills training: social validation of an evidence-based treatment for the mentally disabled. J Ment Health. 2007;16:595–623.
- Kopelowicz A, Liberman R, Zarate R. Recent advances in social skills training for schizophrenia. Schizophr Bull. 2006;32 Suppl 1:S12–23.
- Glynn SM, Marder SR, Liberman RP, et al. Supplementing clinic-based skills training with manual-based community support sessions: effects on social adjustment of patients with schizophrenia. Am J Psychiatry. 2002;159:829–37.
- van Meijel B, Megens Y, Koekkoek B, de Vogel W, Kruitwagen C, Grypdonck M. Effective interaction with patients with schizophrenia: qualitative evaluation of the interaction skills training programme. Perspect Psychiatr Care. 2009;45:254–61.
- Lukoff D, Wallace CJ, Liberman R, Burke K. A holistic program for chronic schizophrenic patients. Schizophr Bull. 1986;12:274–82.
- 19. Mueser K, Corrigan P, Hilton D, et al. Illness management and recovery: a review of the research. Psychiatr Serv. 2002;53:1272–84.
- Granholm E, McQuaid JR, McClure FS, et al. Randomized controlled trial of cognitive behavioral social skills training for older people with schizophrenia: 12-month follow-up. J Clin Psychiatry. 2007;68:730–7.
- Patterson TL, Mausbach BT, McKibbin C, Goldman S, Bucardo J, Jeste DV. Functional adaptation skills training (FAST): a randomized trial of a psychosocial intervention for middleaged and older patients with chronic psychotic disorders. Schizophr Res. 2006;86:291–9.
- 22. Vesterager L, Christensen TO, Olsen BB, et al. Cognitive training plus a comprehensive psychosocial programme (OPUS) versus the comprehensive psychosocial programme alone for patients with first-episode schizophrenia (the NEUROCOM trial): a study protocol for a centrally randomised, observer-blinded multi-centre clinical trial. Trials. 2012;12:35.
- Waldheter EJ, Penn DL, Perkins DO, Mueser KT, Owens LW, Cook E. The graduated recovery intervention program for first episode psychosis: treatment development and preliminary data. Community Ment Health J. 2008;44:443–55.
- Cui Y, Yang W, Weng Y. Effectiveness of social skills trainign in patients with chronic schizophrenia. Chin Ment Health J. 2004;18:799–804.
- 25. Jung SH, Kim HJ. Perceived stigma and quality of life of individuals diagnosed with schizophrenia and receiving psychiatric rehabilitation services: a comparison between the clubhouse model and a rehabilitation skills training model in South Korea. Psychiatr Rehabil J. 2012;35:460–5.
- Valencia M, Rascon ML, Juarez F, Murow E. A psychosocial skills training approach in Mexican out-patients with schizophrenia. Psychol Med. 2007;37:1393–402.
- Yildiz M, Veznedaroglu B, Eryavuz A, Kayahan B. Psychosocial skills training on social functioning and quality of life in the treatment of schizophrenia: a controlled study in Turkey. Int J Psychiatry Clin Pract. 2004;8:219–25.
- Benton MK, Schroeder HE. Social skills training with schizophrenics: a meta-analytic evaluation. J Consult Clin Psychol. 1990;58:741–7.
- Pilling S, Bebbington P, Kuipers E. Psychological treatments in schizophrenia: II. Meta-analyses of randomized controlled trials of social skills training and cognitive remediation. Psychol Med. 2002;32:783–91.
- Kurtz MM, Mueser KT. A meta-analysis of controlled research on social skills training for schizophrenia. J Consult Clin Psychol. 2008;76:491–504.
- Liberman RP, Glynn S, Blair KE, Ross D, Marder SR. In vivo amplified skills training: promoting generalization of independent living skills for clients with schizophrenia. Psychiatry. 2002;65:137–55.
- Kurzban S, Davis L, Brekke JS. Vocational, social, and cognitive rehabilitation for individuals diagnosed with schizophrenia: a review of recent research and trends. Curr Psychiatry Rep. 2010;12:345–55.
- Bell MD, Zito W, Greig T, Wexler BE. Neurocognitive enhancement therapy with vocational services: work outcomes at two-year follow-up. Schizophr Res. 2008;105:18–29.

- 34. Granholm E, McQuaid JR, McClure FS, et al. A randomized, controlled trial of cognitive behavioral social skills training for middle-aged and older outpatients with chronic schizophrenia. Am J Psychiatry. 2005;162:520–9.
- Linehan M. Cognitive-behavioral treatment of borderline personality disorder. New York, NY: Guilford; 1993.
- 36. Elbogen E, Tiegreen J, Vaughan C, Bradford D. Money management, mental health, and psychiatric disability: a recovery-oriented model for improving financial skills. Psychiatr Rehabil J. 2011;34:223–31.
- Marson D, Savage R, Phillips J. Financial capacity in persons with schizophrenia and serious mental illness: clinical and research ethics aspects. Schizophr Bull. 2006;32:81–91.
- 38. Stein LI, Test MA. Alternative to mental hospital treatment. I. Conceptual model, treatment program, and clinical evaluation. Arch Gen Psychiatry. 1980;37:392–7.
- Dixon L. Assertive community treatment: twenty-five years of gold. Psychiatr Serv. 2000;51:759–65.
- 40. Santos A, Stein L. Assertive community treatment of persons with severe mental illness. New York, NY: Norton; 1998.
- McGrew JH, Bond GR, Dietzen L, Salyers M. Measuring the fidelity of implementation of a mental health program model. J Consult Clin Psychol. 1994;62:670–8.
- 42. Cuddeback GS, Morrissey JP, Domino ME, Monroe-DeVita M, Teague GB, Moser LL. Fidelity to recovery-oriented ACT practices and consumer outcomes. Psychiatr Serv. 2013;64:318–23.
- Teague GB, Bond GR, Drake RE. Program fidelity in assertive community treatment: development and use of a measure. Am J Orthopsychiatry. 1998;68:216–32.
- 44. Salyers MP, Stull LG, Rollins AL, Hopper K. The work of recovery on two assertive community treatment teams. Adm Policy Ment Health. 2011;38:169–80.
- 45. Bellack AS. Scientific and consumer models of recovery in schizophrenia: concordance, contrasts, and implications. Schizophr Bull. 2006;32:432–42.
- Gomory T. A critique of the effectiveness of assertive community treatment. Psychiatr Serv. 2001;52:1394–5.
- van Vugt MD, Kroon H, Delespaul PA, Mulder CL. Consumer-providers in assertive community treatment programs: associations with client outcomes. Psychiatr Serv. 2012;63: 477–81.
- 48. Stull LG, McGrew JH, Salyers MP. Staff and consumer perspectives on defining treatment success and failure in assertive community treatment. Psychiatr Serv. 2010;61:929–32.
- 49. Mueser KT, Bond GR, Drake RE, Resnick SG. Models of community care for severe mental illness: a review of research on case management. Schizophr Bull. 1998;24:37–74.
- 50. Marshall M, Lockwood A. Assertive community treatment for people with severe mental disorders. Cochrane Database Syst Rev. 2000;CD001089.
- Ziguras SJ, Stuart GW. A meta-analysis of the effectiveness of mental health case management over 20 years. Psychiatr Serv. 2000;51:1410–21.
- Coldwell CM, Bender WS. The effectiveness of assertive community treatment for homeless populations with severe mental illness: a meta-analysis. Am J Psychiatry. 2007;164:393–9.
- Slade EP, McCarthy JF, Valenstein M, Visnic S, Dixon LB. Cost savings from assertive community treatment services in an era of declining psychiatric inpatient use. Health Serv Res. 2013;48:195–217.
- Burns T, Catty J, Dash M, Roberts C, Lockwood A, Marshall M. Use of intensive case management to reduce time in hospital in people with severe mental illness: systematic review and meta-regression. BMJ. 2007;335:336.
- 55. Beach C, Dykema LR, Appelbaum PS, et al. Forensic and nonforensic clients in assertive community treatment: a longitudinal study. Psychiatr Serv. 2013;64:437–44.
- 56. Lamberti JS, Weisman R, Faden DI. Forensic assertive community treatment: preventing incarceration of adults with severe mental illness. Psychiatr Serv. 2004;55:1285–93.
- Tempier R, Balbuena L, Garety P, Craig TJ. Does assertive community outreach improve social support? Results from the Lambeth Study of early-episode psychosis. Psychiatr Serv. 2012;63:216–22.

14 Psychosocial Rehabilitation and Psychotherapy Approaches

- Boden R, Sundstrom J, Lindstrom E, Wieselgren IM, Lindstrom L. Five-year outcome of first-episode psychosis before and after the implementation of a modified assertive community treatment programme. Soc Psychiatry Psychiatr Epidemiol. 2010;45:665–74.
- Burns T. The UK700 trial of intensive case management: an overview and discussion. World Psychiatry. 2002;1:175–8.
- 60. Karow A, Reimer J, Konig HH, et al. Cost-effectiveness of 12-month therapeutic assertive community treatment as part of integrated care versus standard care in patients with schizophrenia treated with quetiapine immediate release (ACCESS trial). J Clin Psychiatry. 2012;73:e402–8.
- Burns T, Fioritti A, Holloway F, Malm U, Rossler W. Case management and assertive community treatment in Europe. Psychiatr Serv. 2001;52:631–6.
- 62. Provencher HGR, Crawford S, Mueser K. The role of work in the recovery of persons with psychiatric disabilities. Psychiatr Rehabil J. 2002;26:132–44.
- Bond GR, Drake RE. Predictors of competitive employment among patients with schizophrenia. Curr Opin Psychiatry. 2008;21:362–9.
- Marwaha S, Johnson S. Schizophrenia and employment. Soc Psychiatry Psychiatr Epidemiol. 2004;39:337–49.
- 65. Rosenheck R, Leslie D, Keefe R, et al. Barriers to employment for people with schizophrenia. Am J Psychiatry. 2006;163:411–7.
- 66. Bond GR. Supported employment: evidence for an evidence-based practice. Psychaitr Rehabil J. 2004;27:345–59.
- Becker DR, Drake RE. A working life for people with severe mental illness. New York, NY: Oxford Press; 2003.
- Drake RE, McHugo GJ, Becker DR, Anthony WA, Clark RE. The New Hampshire study of supported employment for people with severe mental illness. J Consult Clin Psychol. 1996;64:391–9.
- 69. Frey W, Azrin S, Goldman HH. The mental health treatment study. Psychiatr Rehabil J. 2008;31:306–12.
- Twamley EW, Narvaez JM, Becker DR, Bartels SJ, Jeste DV. Supported employment for middle-aged and older people with schizophrenia. Am J Psychiatr Rehabil. 2008; 11:76–89.
- Larson JE, Barr LK, Corrigan P. Perspectives on the benefits and costs of work from individuals with psychiatric disorders. J Vocat Rehabil. 2007;26:71–7.
- Boycott N, Schneider J, McMurran M. Additional interventions to enhance the effectiveness of individual placement and support: a rapid evidence assessment. Rehabil Res Pract. 2012;2012:382420.
- Kidd SA, Kaur Bajwa J, McKenzie KJ, Ganguli R, Haji Khamneh B. Cognitive remediation for individuals with psychosis in a supported education setting: a pilot study. Rehabil Res Pract. 2012;2012:715176.
- Nuechterlein KH, Subotnik KL, Turner LR, Ventura J, Becker DR, Drake RE. Individual placement and support for individuals with recent-onset schizophrenia: integrating supported education and supported employment. Psychiatr Rehabil J. 2008;31:340–9.
- 75. Crowther RE, Marshall M, Bond GR, Huxley P. Helping people with severe mental illness to obtain work: systematic review. Br Med J. 2001;322:204–8.
- 76. Burns T, Catty J, Becker T. The effectiveness of supported employment for people with severe mental illness: a randomised controlled trial. Lancet. 2007;270:1146–52.
- Bellack A, Morrison R, Mueser K. Social problem solving in schizohrenia. Schizophr Bull. 1989;15:101–16.
- Twamley EW, Jeste DV, Lehman AF. Vocational rehabilitation in schizophrenia and other psychotic disorders: a literature review and meta-analysis of randomized controlled trials. J Nerv Ment Dis. 2003;191:515–23.
- 79. Bond GR, Drake RE, Becker DR. An update on randomized controlled trials for evidence based supported employment. Psychiatr Rehabil J. 2008;31:280–90.

- Cook JA, Copeland ME, Jonikas JA, et al. Results of a randomized controlled trial of mental illness self-management using wellness recovery action planning. Schizophr Bull. 2012;38: 881–91.
- Campbell K, Bond GR, Drake RE. Who benefits from supported employment: a Metaanalytic Study. Schizophr Bull. 2011;37:370–80.
- 82. U.S. Department of Labor BoLS. Unemployment rates by county. 2011, Annual. http://data. bls.gov/map/MapToolServlet
- 83. Green MF, Kern RS, Braff DL, Mintz J. Neurocognitive deficits and functional outcome in schizophrenia: are we measuring the 'right stuff'? Schizophr Bull. 2000;26:119–36.
- Green MF, Kern RS, Heaton RK. Longitudinal studies of cognition and functional outcome in schizophrenia: implications for MATRICS. Schizophr Res. 2004;72:41–51.
- Medalia A, Choi J. Cognitive remediation in schizophrenia. Neuropsychol Rev. 2009;19:353–64.
- Fisher M, Holland C, Merzenich MM, Vinogradov S. Using neuroplasticity-based auditory training to improve verbal memory in schizophrenia. Am J Psychiatry. 2009;166:805–11.
- Fisher M, Holland C, Subramaniam K, Vinogradov S. Neuroplasticity-based cognitive training in schizophrenia: an interim report on the effects 6 months later. Schizophr Bull. 2010;36: 869–79.
- McGurk SR, Mueser KT, Feldman K, Wolfe R, Pascaris A. Cognitive training for supported employment: 2-3 year outcomes of a randomized controlled trial. Am J Psychiatry. 2007;164:437–41.
- Medalia A, Freilich B. The Neuropsychological Educational Approach to Cognitive Remediation (NEAR) Model: Practice Principles and Outcome Studies. Am J Psychiatr Rehabil. 2008;11:123–143.
- Wykes T, Reeder C. Cognitive remediation therapy for schizophrenia: theory and practice. London: Routledge; 2005.
- Wykes T, Reeder C, Landau S, et al. Cognitive remediation therapy in schizophrenia: randomised controlled trial. Br J Psychiatry. 2007;190:421–7.
- Twamley EW, Savla GN, Zurhellen CH, Heaton RK, Jeste DV. Development and pilot testing of a novel compensatory cognitive training intervention for people with psychosis. Am J Psychiatr Rehabil. 2008;11:144–63.
- Velligan DI, Bow-Thomas CC, Huntzinger C, et al. Randomized controlled trial of the use of compensatory strategies to enhance adaptive functioning in outpatients with schizophrenia. Am J Psychiatry. 2000;157:1317–23.
- Velligan DI, Diamond PM, Mintz J, et al. The use of individually tailored environmental supports to improve medication adherence and outcomes in schizophrenia. Schizophr Bull. 2008;34:483–93.
- Kern RS, Green MF, Mintz J, Liberman RP. Does 'errorless learning' compensate for neurocognitive impairments in the work rehabilitation of persons with schizophrenia? Psychol Med. 2003;33:433–42.
- Kern RS, Green MF, Mitchell S, Kopelowicz A, Mintz J, Liberman RP. Extensions of errorless learning for social problem-solving deficits in schizophrenia. Am J Psychiatry. 2005;162:513–9.
- 97. Kern RS, Glynn SM, Horan WP, Marder SR. Psychosocial treatments to promote functional recovery in schizophrenia. Schizophr Bull. 2009;35:347–61.
- Pope JW, Kern RS. An 'Errorful' learning deficit in schizophrenia? J Clin Exp Neuropsychol. 2006;28:101–10.
- Grynszpan O, Perbal S, Pelissolo A, et al. Efficacy and specificity of computer-assisted cognitive remediation in schizophrenia: a meta-analytical study. Psychol Med. 2011;41:163–73.
- 100. Hayes RL, McGrath JJ. Cognitive rehabilitation for people with schizophrenia and related conditions. Cochrane Database Syst Rev. 2000;(3).
- Krabbendam L, Aleman A. Cognitive rehabilitation in schizophrenia: a quantitative analysis of controlled studies. Psychopharmacology (Berl). 2003;169:376–82.

- 102. Kurtz MM, Moberg PJ, Gur RC, Gur RE. Approaches to cognitive remediation of neuropsychological deficits in schizophrenia: a review and meta-analysis. Neuropsychol Rev. 2001;11:197–210.
- Twamley EW, Jeste DV, Bellack AS. A review of cognitive training in schizophrenia. Schizophr Bull. 2003;29:359–82.
- McGurk SR, Twamley EW, Sitzer DI, McHugo GJ, Mueser KT. A meta-analysis of cognitive remediation in schizophrenia. Am J Psychiatry. 2007;164:1791–802.
- Wykes T, Huddy V, Cellard C, McGurk SR, Czobor P. A meta-analysis of cognitive remediation for schizophrenia: methodology and effect sizes. Am J Psychiatry. 2011;168:472–85.
- 106. Subramaniam K, Luks TL, Fisher M, Simpson GV, Nagarajan S, Vinogradov S. Computerized cognitive training restores neural activity within the reality monitoring network in schizophrenia. Neuron. 2012;23:842–53.
- Velligan DI, Prihoda TJ, Ritch JL, Maples N, Bow-Thomas CC, Dassori A. A randomized single-blind pilot study of compensatory strategies in schizophrenia outpatients. Schizophr Bull. 2002;28:283–92.
- Kern RS, Liberman RP, Kopelowicz A, Mintz J, Green MF. Applications of errorless learning for improving work performance in persons with schizophrenia. Am J Psychiatry. 2002;159:1921–6.
- Kern RS, Liberman RP, Becker DR, Drake RE, Sugar CA, Green MF. Errorless learning for training individuals with schizophrenia at a community mental health setting providing work experience. Schizophr Bull. 2009;35:807–15.
- Medalia A, Saperstein AM. Does cognitive remediation for schizophrenia improve functional outcomes? Curr Opin Psychiatry. 2013;26:151–7.
- Penn D, Sanna L, Roberts D. Social cognition in schizophrenia: an overview. Schizophr Bull. 2008;34:408–11.
- 112. Savla GN, Vella L, Armstrong CC, Penn DL, Twamley EW. Deficits in domains of social cognition in schizophrenia: a meta-analysis of the empirical evidence. Schizophr Bull. 2012;4:4.
- Couture SM, Penn DL, Roberts DL. The functional significance of social cognition in schizophrenia: a review. Schizophr Bull. 2006;32:S44–63.
- 114. Fett A-KJ, Viechtbauer W, Dominguez M-D-G, Penn DL, van Os J, Krabbendam L. The relationship between neurocognition and social cognition with functional outcomes in schizophrenia: a meta-analysis. Neurosci Biobehav Rev. 2011;35:573–88.
- 115. Mancuso F, Horan WP, Kern RS, Green MF. Social cognition in psychosis: multidimensional structure, clinical correlates, and relationship with functional outcome. Schizophr Res. 2011;125:143–51.
- 116. Green MF, Penn DL, Bentall R, et al. Social cognition in schizophrenia: an NIMH workshop on definitions, assessment, and research opportunities. Schizophr Bull. 2008;34:1211–20.
- 117. Combs DR, Tosheva A, Penn DL, Basso MR, Wanner JL, Laib K. Attentional-shaping as a means to improve emotion perception deficits in schizophrenia. Schizophr Res. 2008;105:68–77.
- 118. Combs DR, Chapman D, Waguspack J, Basso MR, Penn DL. Attention shaping as a means to improve emotion perception deficits in outpatients with schizophrenia and impaired controls. Schizophr Res. 2011;127:151–6.
- 119. Marsh PJ, Green MJ, Russell TA, McGuire J, Harris A, Coltheart M. Remediation of facial emotion recognition in schizophrenia: functional predictors, generalizability, and durability. Am J Psychiatr Rehabil. 2010;13:143–70.
- Russell TA, Chu E, Phillips ML. A pilot study to investigate the effectiveness of emotion recognition remediation in schizophrenia using the micro-expression training tool. Br J Clin Psychol. 2006;45:579–83.
- 121. Frommann N, Streit M, Wölwer W. Remediation of facial affect recognition impairments in patients with schizophrenia: a new training program. Psychiatry Res. 2003;117:281–4.
- 122. Wolwer W, Frommann N. Social-cognitive remediation in schizophrenia: generalization of effects of the Training of Affect Recognition (TAR). Schizophr Bull. 2011;37:S63–70.

- 123. Sachs G, Winklbaur B, Jagsch R, et al. Training of affect recognition (TAR) in schizophrenia—impact on functional outcome. Schizophr Res. 2012;138:262–7.
- 124. Penn DL, Roberts DL, Combs D, Sterne A. The development of the social cognition and interaction training program for schizophrenia spectrum disorders. Psychiatr Serv. 2007;58:449–51.
- 125. Horan WP, Kern RS, Shokat-Fadai K, Sergi MJ, Wynn JK, Green MF. Social cognitive skills training in schizophrenia: an initial efficacy study of stabilized outpatients. Schizophr Res. 2009;107:47–54.
- 126. Horan WP, Kern RS, Tripp C, et al. Efficacy and specificity of social cognitive skills training for outpatients with psychotic disorders. J Psychiatr Res. 2011;45:1113–22.
- 127. Hogarty GE, Flesher S. Practice principles of cognitive enhancement therapy for schizophrenia. Schizophr Bull. 1999;25:693–708.
- 128. Brenner HD, Roder V, Hodel B, Kienzle N, Reed D, Liberman RP. Integrated psychological therapy for schizophrenic patients (IPT). Ashland, OH: Hogrefe & Huber Publishers; 1994.
- 129. Horan WP, Kern RS, Green MF, Penn DL. Social cognition training for individuals with schizophrenia: emerging evidence. Am J Psychiatr Rehabil. 2008;11:205–52.
- 130. Kurtz MM, Richardson CL. Social cognitive training for schizophrenia: a meta-analytic investigation of controlled research. Schizophr Bull. 2012;38:1092–104.
- 131. Fiszdon JM, Reddy LF. Review of social cognitive treatments for psychosis. Clin Psychol Rev. 2012;32:724–40.
- 132. Wölwer WFN, Haufmann S, Piaszek A, Streit M, Gaebel W. Remediation of impairments in facial affect recognition in schizophrenia: efficacy and specificity of a new training program. Schizophr Res. 2005;80:295–303.
- 133. Pinkham AE, Penn DL, Green MF, Buck B, Healey K, Harvey PD. The social cognition psychometric evaluation study: results of the expert survey and RAND panel. Schizophr Bull (in press).
- 134. Kern R, Zarate R, Glynn S, et al. A demonstration project involving peers as providers of evidence-based, supported employment services. J Psychiatr Rehabil. 2013;36(2):99–107.
- 135. Mead S, Hilton D, Curtis L. Peer support: a theoretical perspective. Psychiatr Rehabil J. 2001;25:134–41.
- 136. Goldstrom ID, Campbell J, Rogers JA, et al. National estimates for mental health mutual support groups, self-help organizations, and consumer-operated services. Adm Policy Ment Health. 2006;33:92–103.
- 137. Campbell J. The historical and philosophical development of peer-run support programs. In: Clay S, Schell B, Corrigan PW, Ralph RO, editors. On our own together: peer programs for people with mental illness. Nashville, TN: Vanderbilt Press; 2005. p. 17–64.
- Sledge WH, Lawless M, Sells D, Wieland M, O'Connell MJ, Davidson L. Effectiveness of peer support in reducing readmissions of persons with multiple psychiatric hospitalizations. Psychiatr Serv. 2011;62:541–4.
- 139. Fricks L. Building a foundation for recovery: a community education guide on establishing medicaid-funded peer support services and a trained peer workforce, DHHS. Rockville, MD: Center for Mental Health Services, Substance Abuse and Mental Health Services Administration; 2005.
- Davidson L, Chinman M, Kloos B, Weingarten R, Stayner D, Tebes JK. Peer support among individuals with severe mental illness: a review of the evidence. Clin Psychol. 1999;6:165–87.
- 141. Davidson L, Chinman M, Sells D, Rowe M. Peer support among adults with serious mental illness: a report from the field. Schizophr Bull. 2006;32:443–50.
- 142. O'Donnell M, Parker G, Proberts M, et al. A study of client-focused case management and consumer advocacy: the Community and Consumer Service Project. Aust N Z J Psychiatry. 1999;33:684–93.
- 143. Schmidt LT, Gill KJ, Pratt CW, Solomon P. Comparison of service outcomes of case management teams with and without a consumer provider. Am J Psychiatr Rehabil. 2008;11:310–29.

- 144. Grant EA, Reinhart C, Wituk S, Meissen G. An examination of the integration of certified peer specialists into community mental health centers. Community Ment Health J. 2012; 48:477–81.
- Sherman PS, Porter R. Mental health consumers as case management aides. Hosp Community Psychiatry. 1991;42:494–8.
- 146. Solomon P, Draine J. The efficacy of a consumer case management team: 2-year outcomes of a randomized trial. J Ment Health Adm. 1995;22:135–46.
- 147. Felton CJ, Stastny P, Shern DL, et al. Consumers as peer specialists on intensive case management teams: impact on client outcomes. Psychiatr Serv. 1995;46:1037–44.
- 148. Craig T, Doherty I, Jamieson-Craig R, Boocock A, Attafua G. The consumer-employee as a member of a Mental Health Assertive Outreach Team. I. Clinical and social outcomes. J Ment Health Adm. 2004;13:59–69.
- 149. Cook JA, Steigman P, Pickett S, et al. Randomized controlled trial of peer-led recovery education using Building Recovery of Individual Dreams and Goals Through Education And Support (BRIDGES). Schizophr Res. 2012;136:36–42.
- 150. Druss BG, Zhao L, von Esenwein SA, et al. The health and recovery peer (HARP) program: a peer-led intervention to improve medical self-management for persons with serious mental illness. Schizophr Res. 2010;118:264–70.
- 151. Pickett SA, Diehl SM, Steigman PJ, et al. Consumer empowerment and self-advocacy outcomes in a randomized study of peer-led education. Community Ment Health J. 2012;48:420–30.
- Justo LP, Soares BG, Calil HM. Family interventions for bipolar disorder. Cochrane Database Syst Rev. 2007;CD005167.
- Lucksted A, McFarlane W, Downing D, Dixon L. Recent developments in family psychoeducation as an evidence-based practice. J Marital Fam Ther. 2012;38:101–21.
- 154. Butzlaff RL, Hooley JM. Expressed emotion and psychiatric relapse: a meta-analysis. Arch Gen Psychiatry. 1998;55:547–52.
- 155. Hooley JM. Expressed emotion and relapse of psychopathology. Annu Rev Clin Psychol. 2007;3:329–52.
- 156. Magliano L, Fiorillo A, De Rosa C, Malangone C, Maj M. Family burden in long-term diseases: a comparative study in schizophrenia vs. physical disorders. Soc Sci Med. 2005;61:313–22.
- 157. Baronet A-M. Factors associated with caregiver burden in mental illness: a critical review of the research literature. Clin Psychol Rev. 1999;19:819–41.
- Prince JD. Family involvement and satisfaction with community mental health care of individuals with schizophrenia. Community Ment Health J. 2005;41:419–30.
- Cohen AN, Drapalski AL, Glynn SM, Medoff D, Fang LJ, Dixon LB. Preferences for family involvement in care among consumers with serious mental illness. Psychiatr Serv. 2013; 64:257–63.
- 160. Murray-Swank A, Glynn S, Cohen AN, et al. Family contact, experience of family relationships, and views about family involvement in treatment among VA consumers with serious mental illness. J Rehabil Res Dev. 2007;44:801–11.
- 161. Glynn SM, Cohen AN, Dixon LB, Niv N. The potential impact of the recovery movement on family interventions for schizophrenia: opportunities and obstacles. Schizophr Bull. 2006;32:451–63.
- 162. Falloon IR, Boyd JL, McGill CW. Family care of schizophrenia: a problem-solving approach to the treatment of mental illness. New York, NY: The Guilford Press; 1984.
- 163. McFarlane WR, Lukens E, Link B, et al. Multiple-family groups and psychoeducation in the treatment of schizophrenia. Arch Gen Psychiatry. 1995;52:679–87.
- 164. McFarlane WR, Link B, Dushay R, Marchal J, Crilly J. Psychoeducational multiple family groups: four-year relapse outcome in schizophrenia. Fam Process. 1995;34:127–44.
- 165. Randolph ET, Eth S, Glynn SM, et al. Behavioural family management in schizophrenia. Outcome of a clinic-based intervention. Br J Psychiatry. 1994;164:501–6.

- 166. Pitschel-Walz G, Leucht S, Bauml J, Kissling W, Engel RR. The effect of family interventions on relapse and rehospitalization in schizophrenia—a meta-analysis. Schizophr Bull. 2001;21:73–92.
- 167. Pfammatter M, Junghan UM, Brenner HD. Efficacy of psychological therapy in schizophrenia: conclusions from meta-analyses. Schizophr Bull. 2006;32:S64–80.
- 168. Pharoah F, Mari J, Rathbone J, Wong W. Family intervention for schizophrenia. Cochrane Database Syst Rev. 2010;CD000088.
- 169. Pilling S, Bebbington P, Kuipers E, et al. Psychological treatments in schizophrenia: I. Meta-analysis of family intervention and cognitive behaviour therapy. Psychol Med. 2002;32:763–82.
- 170. Montero I, Hernandez I, Asencio A, Bellver F, LaCruz M, Masanet MJ. Do all people with schizophrenia receive the same benefit from different family intervention programs? Psychiatry Res. 2005;133:187–95.
- 171. Turkington D, Kingdon D, Weiden PJ. Cognitive behavior therapy for schizophrenia. Am J Psychiatry. 2006;163:365–73.
- 172. Beck AT. Successful outpatient psychotherapy of a chronic schizophrenic with a delusion based on borrowed guilt. Psychiatry 1952;15(SRC—GoogleScholar):305–12.
- 173. Hole RW, Rush AJ, Beck AT. A cognitive investigation of schizophrenic delusions. Psychiatry. 1979;42:312–9.
- 174. Watts FN, Powell GE, Austin SV. The modification of abnormal beliefs. Br J Med Psychol. 1973;46:359–63.
- 175. Wykes T, Steel C, Everitt B, Tarrier N. Cognitive behavior therapy for schizophrenia: effect sizes, clinical models, and methodological rigor. Schizophr Bull. 2008;34:523–37.
- 176. Dixon LB, Dickerson F, Bellack AS, et al. The 2009 schizophrenia PORT psychosocial treatment recommendations and summary statements. Schizophr Bull. 2010;36:48–70.
- 177. Beck AT, Rector NA. Cognitive therapy of schizophrenia: a new therapy for the new millennium. Am J Psychother. 2000;54:291–300.
- 178. Kingsep P, Nathan P, Castle D. Cognitive behavioural group treatment for social anxiety in schizophrenia. Schizophr Res. 2003;63:121–9.
- 179. McEvoy PM. Effectiveness of cognitive behavioural group therapy for social phobia in a community clinic: a benchmarking study. Behav Res Ther. 2010;45:3030–40.
- 180. Barrowclough C, Haddock G, Tarrier N, et al. Randomized controlled trial of motivational interviewing, cognitive behavior therapy, and family intervention for patients with comorbid schizophrenia and substance use disorders. Am J Psychiatry. 2001;158:1706–13.
- 181. Grant PM, Huh GA, Perivoliotis D, Stolar NM, Beck AT. Randomized trial to evaluate the efficacy of cognitive therapy for low-functioning patients with schizophrenia. Arch Gen Psychiatry. 2012;69:121–7.
- 182. Klingberg S, Wölwer W, Engel C, et al. Negative symptoms of schizophrenia as primary target of cognitive behavioral therapy: results of the randomized clinical TONES study. Schizophr Bull. 2011;37 Suppl 2:S98–110.
- 183. Gould RA, Mueser KT, Bolton E, Mays V, Goff D. Cognitive therapy for psychosis in schizophrenia: an effect size analysis. Schizophr Res. 2001;48:335–42.
- 184. Sarin F, Wallin L, Widerlöv B. Cognitive behavior therapy for schizophrenia: a meta-analytical review of randomized controlled trials. Nord J Psychiatry. 2011;65:162–74.
- Rector NA, Beck AT. Cognitive behavioral therapy for schizophrenia: an empirical review. J Nerv Ment Dis. 2001;189:278–87.
- 186. Tarrier N. Cognitive behavior therapy for schizophrenia and psychosis: current status and future directions. Clin Schizophr Relat Psychoses. 2010;4:176–84.
- 187. Zimmermann G, Favrod J, Trieu VH, Pomini V. The effect of cognitive behavioral treatment on the positive symptoms of schizophrenia spectrum disorders: a meta-analysis. Schizophr Res. 2005;77:1–9.
- 188. Birchwood M, Todd P, Jackson C. Early intervention in psychosis. The critical period hypothesis. Br J Psychiatry Suppl. 1998;172:53–9.
- Marshall NB, Swain SL. Cytotoxic CD4 T cells in antiviral immunity. J Biomed Biotechnol. 2011;2011:954602.

- 190. Addington J, Marshall C, French P. Cognitive behavioral therapy in prodromal psychosis. Curr Pharm Des. 2012;18:558–65.
- 191. Hutton P, Taylor PJ. Cognitive behavioural therapy for psychosis prevention: a systematic review and meta-analysis. Psychol Med. 2014;44(3):449–68.
- 192. Jones C, Hacker D, Cormac I, Meaden A, Irving CB. Cognitive behaviour therapy versus other psychosocial treatments for schizophrenia. Cochrane Database of Syst Rev. 2012;(4). doi:1010021465CD008712pub2 1858.
- 193. Lynch D, Laws KR, McKenna PJ. Cognitive behavioural therapy for major psychiatric disorder: does it really work? A meta-analytical review of well-controlled trials. Psychol Med. 2010;40:9–24.
- 194. Newton-Howes G, Wood R. Cognitive behavioural therapy and the psychopathology of schizophrenia: systematic review and meta-analysis. Psychol Psychother. 2013;86:127–38.
- 195. Turkington D, Sensky T, Scott J, et al. A randomized controlled trial of cognitive-behavior therapy for persistent symptoms in schizophrenia: a five-year follow-up. Schizophr Res. 2008;98:1–7.
- 196. Garety P, Fowler D, Kuipers E, et al. London-East Anglia randomised controlled trial of cognitive-behavioural therapy for psychosis. II: Predictors of outcome. Br J Psychiatry. 1997;171:420–6.
- 197. Brabban A, Tai S, Turkington D. Predictors of outcome in brief cognitive behavior therapy for schizophrenia. Schizophr Bull. 2009;35:859–64.
- 198. Naeem F, Kingdon D, Turkington D. Predictors of response to cognitive behavior therapy in the treatment of schizophrenia: A comparison of brief and standard interventions. Cognit Ther Res. 2008;32(SRC—GoogleScholar):651–6.
- 199. Perivoliotis D, Grant PM, Peters ER, Ison R, Kuipers E, Beck AT. Cognitive insight predicts favorable outcome in cognitive behavioral therapy for psychosis. Psychosis Psychol Soc Integr Approaches. 2010;2(SRC—GoogleScholar):23–33.
- 200. Tarrier N, Yusupoff L, Kinney C, et al. Randomised controlled trial of intensive cognitive behaviour therapy for patients with chronic schizophrenia. BMJ. 1998;317:303–7.
- 201. Morrison AP, Renton JC, Williams S, et al. Delivering cognitive therapy to people with psychosis in a community mental health setting: an effectiveness study. Acta Psychiatr Scand. 2004;110:36–44.
- Drury V, Birchwood M, Cochrane R, Macmillan F. Cognitive therapy and recovery from acute psychosis: a controlled trial. II. Impact on recovery time. Br J Psychiatry. 1996;169:602–7.
- 203. Morrison AP, Turkington D, Wardle M, et al. A preliminary exploration of predictors of outcome and cognitive mechanisms of change in cognitive behaviour therapy for psychosis in people not taking antipsychotic medication. Behav Res Ther. 2012;50:163–7.
- 204. Granholm E, Holden J, Link PC, McQuaid JR, Jeste DV. Randomized controlled trial of cognitive behavioral social skills training for older consumers with schizophrenia: defeatist performance attitudes and functional outcome. Am J Geriatr Psychiatry. 2013;21:251–62.
- 205. Rathod S, Kingdon D, Smith P, Turkington D. Insight into schizophrenia: the effects of cognitive behavioral therapy on the components of insight and association with sociodemographics—data on a previously published randomised controlled trial. Schizophr Res. 2005;74:211–9.
- 206. Kurtz MM. Neurocognition as a predictor of response to evidence-based psychosocial interventions in schizophrenia: what is the state of the evidence? Clin Psychol Rev. 2011;31:663–72.
- 207. Granholm E, McQuaid JR, Link PC, Fish S, Patterson T, Jeste DV. Neuropsychological predictors of functional outcome in Cognitive Behavioral Social Skills Training for older people with schizophrenia. Schizophr Res. 2008;100:133–43.
- Lincoln TM, Ziegler M, Mehl S, et al. Moving from efficacy to effectiveness in cognitive behavioral therapy for psychosis: a randomized clinical practice trial. J Consult Clin Psychol. 2012;80:674–86.
- Pinninti NR, Fisher J, Thompson K, Steer R. Feasibility and usefulness of training assertive community treatment team in cognitive behavioral therapy. Community Ment Health J. 2010;46:337–41.

- 210. Granholm E, Loh C, Link PC, Jeste DV. Feasibility of implementing cognitive behavioral therapy for psychosis on assertive community treatment teams: a controlled pilot study. Int J Cogn Ther 2010;3(SRC—GoogleScholar):294–302.
- Gaebel W, Weinmann S, Sartorius N, Rutz W, McIntyre JS. Schizophrenia practice guidelines: international survey and comparison. Br J Psychiatry. 2005;187:248–55.
- 212. Wang PS, Demler O, Kessler RC. Adequacy of treatment for serious mental illness in the United States. Am J Public Health. 2002;92:92–8.
- 213. Drake RE, Bond GR, Essock SM. Implementing evidence-based practices for people with schizophrenia. Schizophr Bull. 2009;35:704–13.
- 214. NAMI. Grading the states: a report on America's health care system for serious mental illness. Arlington, VA: National Alliance on Mental Illness; 2006.
- Cunningham P, McKenzie K, Taylor EF. The struggle to provide community-based care to low-income people with serious mental illness. Health Aff (Millwood). 2006;25:694–705.
- Young GJ, Mohr DC, Meterko M, Siebert M, McGlynn G. Psychiatrists' self reported adherence to evidence-based prescribing practices in the treatment of schizophrenia. Psychiatr Serv. 2006;57:130–2.

Part V Future Directions and Implications

Chapter 15 Conclusion

Philip G. Janicak, Stephen R. Marder, Rajiv Tandon, and Morris Goldman

Overview

The modern conceptualization of schizophrenia began in the late nineteenth century and continues to evolve. Two major initiatives exemplify this process. First, the most recent editions of the Diagnostic and Statistical Manual of Mental Disorders (DSM) and International Classification of Diseases (ICD) incorporate harmonized approaches to diagnosis which differ substantially from their predecessors. Second, major advances in our understanding of the biological basis of this disorder promise to guide future refinements in diagnosis and treatment.

This book updated the reader in four major sections:

• An overview which emphasized the evolving *nosology of schizophrenia* to better differentiate patients based on core symptom dimensions and thus improve response with more personalized treatment regimens (Chap. 2).

P.G. Janicak (🖂)

S.R. Marder UCLA Semel Institute, Los Angeles, CA, USA e-mail: marder@ucla.edu

M. Goldman Department of Psychiatry and Behavioral Sciences, Northwestern University, East Ontario 446, Chicago, IL 60611, USA e-mail: m-goldman@northwestern.edu

Department of Psychiatry, Rush University Medical Center, 2150 West Harrison Street, Chicago, IL 60612, USA e-mail: pjanicak@rush.edu

R. Tandon College of Medicine, University of Florida, P.O. Box 100256, Gainesville, FL 32610-0256, USA e-mail: tandon@ufl.edu

- Updates on recent research into the *pathoetiology of schizophrenia*, focusing on barriers and progress in the identification of biological markers (e.g., genetic, neuroimaging) which will also enhance diagnosis and treatment (Chaps. 3–6).
- The *medical management of schizophrenia* which considered optimal biological strategies and raised awareness about the need to address concurrent complications due to treatment- and medical comorbidity-related complications (Chaps. 7–12).
- The *long-term management of schizophrenia* which increasingly involves the integration of both biological and psychosocial modalities. The ultimate aim is to improve symptoms and enhance functioning so that patients can achieve important personal goals and experience a better quality of life (Chaps. 13 and 14).

Diagnostic Issues in Schizophrenia

The diagnosis of schizophrenia is based on the presence of a characteristic set of symptoms associated with impairment in functioning for a period of at least 6 months and the exclusion of other causes of psychosis. Refinements in the description and criteria of schizophrenia in the DSM-5 should assist clinicians in making a more reliable diagnosis and more clearly differentiating schizophrenia from schizoaffective disorder [1]. The provision in DSM-5 of a simple scale to quantify different dimensions of psychopathology should enable measurement-based and targeted individualized treatment [2]. The dimensional approach to the heterogeneous psychopathology of schizophrenia should facilitate a better bridge to a pathoetiological diagnosis with laboratory markers in the future [3]. These diagnostic issues and the clinical implications of the changes in DSM-5 are considered in Chap. 2.

Pathophysiology of Schizophrenia

While Chaps. 3–6 summarize our current knowledge regarding the neurobiology of schizophrenia, they are largely devoted to understanding some of the limitations of previous research, as well as how current approaches address these challenges [4]. All four chapters underscore the need to clarify mechanisms to better identify biomarkers and intermediate phenotypes which serve to focus future research and treatment efforts. Chapter 3 emphasizes the general need to characterize mechanisms of brain function to better distinguish adaptive from pathological processes. Chapter 4 provides a framework for integrating diverse findings, as well as laying out some of the barriers which are now being addressed [5]. Chapter 5 provides a historical review of genetic findings in schizophrenia and summarizes the logic, strengths, and weaknesses of evolving approaches. In particular, efforts to incorporate downstream analyses and other "omics" approaches are addressed. Chapter 6 focuses on recent advances in neuroimaging and like Chap. 5 illustrates efforts to incorporate multiple modalities to better characterize the functional significance of findings and provide biomarkers of the illnesses [6].

Medical Management of Schizophrenia

The recognition and management of the earliest phases (e.g., high risk, first-episode, early-onset) of schizophrenia are critically important to favorably alter its long-term outcome. Thus, there is an increasing effort dedicated to improving the identification of at-risk individuals. This is exemplified by inclusion of attenuated psychosis syndrome in the DSM-5 as a condition which requires greater scrutiny and further elaboration [7]. Once reliably indentified, the next crucial question is which interventions are both reasonable (i.e., constitute a favorable risk/benefit ratio) and increase the chances of producing a clinically meaningful impact on an individual's long-term prognosis. In this context, treatment approaches include more intense monitoring for symptom evolution; managing associated symptoms (e.g., anxiety, depression, substance abuse); introducing various psychotherapeutic and psychosocial strategies; and when appropriate, employing various standard (e.g., antipsychotics) and novel (e.g., N-acetylcysteine, omega-3 FAs) biological therapies. In multi-episode patients, the use of antipsychotic agents remains a critical component for reducing symptoms, and allowing patients to more effectively benefit from other interventions (e.g., cognitive remediation). Optimal use of these agents during acute exacerbations and to reduce episode relapse and recurrence requires a better appreciation of drug choice and dosing strategies. Regarding the latter, we are progressing from dosing based on data from controlled trials, to plasma-level data, to imaging of dopamine receptor binding, and to the anticipated use of pharmacogenetics/pharmacogenomics to help guide clinicians in their medication strategies. Unfortunately, many obstacles preclude an optimal trial with existing antipsychotics. Thus, even when patients adequately adhere to an appropriate medication regimen their symptoms often persist, impeding recovery.

As a result, a number of strategies are presently applied to enhance the benefit of standard agents. These include:

- · An adequate trial of clozapine when feasible
- · Antipsychotic combinations and development of newer agents
- · Novel monotherapy and adjunctive approaches

The last option often involves the repurposing of existing agents as monotherapies or augmentation strategies. The most promising of these include drugs with a novel spectrum of neuroreceptor actions including various dopamine (e.g., cariprazine), serotonin (e.g., ondansetron), glutamatergic (e.g., bitopertin), gabaergic (e.g., divalproex sodium), and cholinergic (e.g., phosphatidylcholine) agents. In addition, other promising approaches which may eventually benefit patients include the use of certain antidepressants (e.g., mirtazapine) to better manage depressive, negative, and cognitive symptoms; anti-inflammatory agents (e.g., minocycline) to improve negative and cognitive symptoms in early-onset patients; hormonal agents (e.g., estrogen compounds), especially during heightened levels of vulnerability such as the postpartum and perimenopausal periods; and neutraceuticals (e.g., omega-3 FAs, folate) to aid in normal neurodevelopment during pregnancy.

Adjunctive therapeutic modulation may also play a role in managing some of the most severe exacerbations (e.g., electroconvulsive therapy) or for resistant positive and negative symptoms (e.g., transcranial magnetic stimulation).

Medical Comorbidities

There is increasing awareness that medical comorbidities play a major role in determining the overall well-being and functionality of patients with schizophrenia. Their impact includes a substantial decrease in life expectancy due to various factors (e.g., inattention to health issues by both patients and their treatment teams). One example is the psychosis intermittent hyponatremia (PIP) syndrome associated with episodic water intoxication. This occurs in up to 10 % of chronically-ill mental patients. PIP appears related to the illness itself (e.g., schizophrenia) but hyponatremia also frequently arises from other factors (e.g., treatment with certain psychotropics such as carbamazepine). Unfortunately, these issues often go unrecognized and may lead to subtle neurological symptoms, renal and bladder problems, or even death due to severe hyponatremia. When properly diagnosed, however, recent studies support the use of clozapine or competitive vasopressin receptor 2 antagonists (i.e., vaptans) to manage this condition [8].

Medication Adverse Effects

Related to the general health issues of patients are the adverse effects associated with antipsychotics and other psychotropic agents frequently used to manage schizophrenia. In particular, antipsychotic-induced weight gain and its subsequent complications (e.g., metabolic syndrome, new-onset diabetes) contribute to much higher rates of cardiovascular disease in patients on these drugs compared with the general population. It is important for clinicians who manage these patients to monitor the modifiable risk factors for cardiovascular disease including weight, blood pressure, and lipids. In this context, recent studies support the use of metformin to prevent or reverse the weight-gain, as well as to improve BMI and lower triglyceride and hemoglobin A_{1c} levels [9]. Since this drug's impact appears to be modest, however, various psychosocial interventions (e.g., education about nutrition, caloric expenditure, portion control), behavioral self-management (e.g., goal setting, regular weigh-ins), and increased physical activity should be attempted before a trial of this agent [10].

Long-Term Management of Schizophrenia

Improving the long-term outcome in schizophrenia entails:

- Preventing recurrences of psychosis
- Improving functioning and quality of life
- · Promoting improved physical health

Antipsychotic medications are effective for preventing episodes of psychosis [11]. Many patients, however, resist taking their medications and others are unreliable pill takers. As a result, successful treatment often requires strategies for assuring adequate adherence such as educating patients and family members, addressing medication side effects, and using long-acting injectable (LAI) formulations. The recent introduction of several second-generation LAIs provides new choices for patients and clinicians and should lead to greater use of this method for administering antipsychotics.

In addition to medication management, this volume emphasizes an approach to long-term management which accepts the principles of recovery [12]. According to these principles, treatment should focus on the goals of the patient. Although optimal treatment may not completely alleviate the burden of the illness, patients can often learn strategies which compensate for their impairments. This approach almost always includes combining psychosocial and pharmacological treatments. Effective psychosocial treatments include patient and family education, cognitive behavioral treatment for psychosis, social skills training, supported employment, and cognitive remediation [13].

Clinicians who manage patients with schizophrenia should also address the substantial physical health problems associated with the illness, as well as comorbidities such as substance abuse. Psychosocial treatments are effective for helping patients adapt healthier life styles with improved diets and exercise, while they also help patients address nicotine and other addictions [13]. Despite the effectiveness of these treatments, it is often difficult for patients and their families to access them. Thus, the need for resources to provide these effective and available therapies is an important social–political issue.

In the Future

Although we have made considerable progress in improving treatments and outcomes for persons with schizophrenia, substantial needs remain. Whereas a plethora of neurobiological findings in schizophrenia are described, their precise meaning is unclear and their relevance to clinical practice is limited. This is likely to change in the near future because of both methodological and conceptual advances and because of improvements in nosology, including refinement of DSM criteria and incorporation of objective measures. As neuroimaging, neurochemical and electrophysiological technologies mature, dysfunctions will become more apparent. Better integration of these data, as well as the combined use of biomarkers, may be more useful than individual markers used in isolation. Multimodal approaches (e.g., combining fMRI with ERP data) may be more useful than individual ones. Better delineation of endophenotypic markers and susceptibility genes are likely to yield better animal models for further hypothesis testing. Larger, multisite studies will also improve statistical power to confirm/refute the tantalizing observations currently limited by small sample sizes. The future holds much promise for more effective, better individualized treatments with greater opportunities for recovery in individuals with schizophrenia.

References

- 1. Tandon R, Carpenter WT. The DSM-5 status of psychotic disorders: 1 year pre-publication. Schizophr Bull. 2012;38:369–70.
- Tandon R, Targum SD, Nasrallah HA, Ross RA. Strategies for maximizing clinical effectiveness in the treatment of schizophrenia. J Psychiatr Pract. 2006;12:348–63.
- Tandon R. The nosology of schizophrenia: towards DSM-5 and ICD-11. Psychiatr Clin North Am. 2012;35:555–69.
- 4. Insel TR. Rethinking schizophrenia. Nature. 2010;468(7321):187-93.
- Keshavan MS, Clementz B, Pearlson GD, Sweeney J, Tamminga C. Reimagining psychoses: an agnostic approach to diagnosis. Schizophr Res. 2013;146(1–3):10–6.
- Schultz CC, Fusar-Poli P, Wagner G. Multimodal functional and structural imaging investigations in psychosis research. Eur Arch Psychiatry Clin Neurosci. 2012;262 Suppl 2:S97–106.
- Tsuang MT, Van Os J, Tandon R, et al. Attenuated psychosis syndrome in DSM-5. Schizophr Res. 2013;150(1):31–5.
- Josiassen RC, Filmyer DM, Geboy AG, Shaughnessy RA. Reconsidering chronic hyponatremia in psychosis. J Clin Psychiatry. 2013;74(3):278–9.
- Jarskog LF, Hamer RM, Catellier DJ, et al. Metformin for weight loss and metabolic control in overweight outpatients with schizophrenia and schizoaffective disorder. Am J Psychiatry. 2013;170:1032–404.
- Dixon LB, Dickerson F, Bellack AS, Schizophrenia Patient Outcomes Research Team (PORT). The 2009 schizophrenia PORT psychosocial treatment recommendations and summary statements schizophrenia. Schizophr Bull. 2010;36(1):94–103.
- 11. Davis JM. Maintenance therapy and the natural course of schizophrenia. J Clin Psychiatry. 1985;46(11 Pt 2):18–21.
- Lysaker PH, Glynn SM, Wilkniss SM, Silverstein SM. Psychotherapy and recovery from schizophrenia: a review of potential applications and need for future study. Psychol Serv. 2010;7(2):75–91.
- Dixon LB, Dickerson F, Bellack AS, Bennett M, Dickinson D, Goldberg RW, et al. The 2009 schizophrenia PORT psychosocial treatment recommendations and summary statements. Schizophr Bull. 2010;36(1):48–70.

Index

A

Access to care common ailments, 177 NHIS, 177 Veterans Administration system, 177 ACE. See Angiotensin converting enzyme (ACE) Acetylcholine, 120-121 Acquired immunodeficiency syndrome (AIDS), 188 ACT. See Assertive community treatment (ACT) Acute dystonia, 230, 233, 248-249 Acute treatment options, 317 Advanced age antipsychotics sensitivity, 242 CATIE-AD study, 242 definition. 241 patient risk factors, 241–242 prevention and treatment, 242 Adverse effects (AEs) acute larvngeal dystonia, 248-249 advanced age, 241-242 clozapine, 243-246 description, 225 hepatotoxicity, 249 hyperprolactinemia, 235-237 metabolic syndrome, 238-241 movement disorders, 230-234 negative impact, 249-250 NMS, 247-248 sexual dysfunction, 237-238 sudden death, 226–230 VTE, 248 AEs. See Adverse effects (AEs) Agranulocytosis, 168

AHRS. See Auditory Hallucination Rating Scale (AHRS) AIDS. See Acquired immunodeficiency syndrome (AIDS) Akathisia, 230-231, 233 AM. See Aripiprazole monohydrate (AM) Angiotensin converting enzyme (ACE), 180 Angiotension receptor blockers (ARBs), 180 Antidepressants, 122 Anti-inflammatory agents, 123 Antipsychotic drugs, 162, 167, 168 Antipsychotic-induced weight gain (AIWG) CATIE Study, 183 description, 183 meta-analysis, 183 HTR2C and MC4R, 167-168 Antipsychotics AM, 271 dosing range, 109, 110 FGAs. 108 haloperidol and fluphenazine decanoate, 271 medications, 268, 269 neuroimaging studies, 109 OP, 269-270 pharmacogenetics, 108 PP, 270 "pragmatic trials", 108 RM. 271 SGAs, 108 ARBs. See Angiotension receptor blockers (ARBs) Arginine vasopressin (AVP) anterior hippocampus, 214 antipsychotic medication, 213 plasma osmolality, 206, 207 renal water excretion, 206

Aripiprazole monohydrate (AM), 271 Assertive community treatment (ACT) efficacy, 284 TCL, 282, 283 treatment, 283–284 Attenuated Psychosis Syndrome, 16, 19 Auditory Hallucination Rating Scale (AHRS), 146–149, 155 AVP. See Arginine vasopressin (AVP)

B

Behavioral family therapy (BFT), 296, 297 Behavioral interventions, 296 BFT. See Behavioral family therapy (BFT) Biological markers, 316 Biomarkers. See also Neuroimaging diagnostic and predictive, 50 physiological and molecular domains, 50 Bipolar disorder (BP), 65 BP. See Bipolar disorder (BP) Brain network disorganization clinical symptoms, 82 cognitive behaviors, 82 ICA components, 84 regional structural and functional impairments, 83 technologies and characterization, 83-84

С

CABG surgery. See Coronary artery bypass graft (CABG) surgery CAD. See Coronary artery disease (CAD) CAFÉ Study. See Comparison of Atypicals in First Episode of Psychosis (CAFÉ) Study Cancer breast, 192 lower cancer incidence, 190 lung, 192 management, 193 meta-analysis, 191 mortality, 191-192 schizophrenia, 191 SIRs, 191 types, 191 Cannabinoids, 123 Cardiovascular adverse effects, 244-245 Cardiovascular disease ACC/AHA Guideline, 181 mortality, 176

CATIE-AD study. See The Clinical Antipsychotic Trials of Intervention Effectiveness-Alzheimer's Disease (CATIE-AD) study CATIE Study. See Clinical Antipsychotic Trials of Intervention Effectiveness (CATIE) Study CBT. See Cognitive behavior therapy (CBT) Challenges treating breast cancer, women with schizophrenia, 192 treating psychotic symptoms, 276 treating serious comorbidities, 300 Chronic obstructive pulmonary disease (COPD), 175, 194-195 The Clinical Antipsychotic Trials of Intervention Effectiveness-Alzheimer's Disease (CATIE-AD) study, 242 Clinical Antipsychotic Trials of Intervention Effectiveness (CATIE) Study, 178, 180.183 Clozapine adverse effects cardiovascular, 244-245 hematologic, 243-244 neurological, 245-246 thermoregulatory, 246 agranulocytosis, 168 ECT. 140 hypersalivation, 246 hyponatremia and water intoxication, 218 insulin resistance, 240 potential benefits and risks, 114, 115 CNVs. See Copy number variants (CNVs) Cognitive behavior therapy (CBT) development, 298 efficacy, 299-300 treatment, 298-299 Cognitive remediation (CR) development and evaluation, 290 efficacy, 290 schizophrenia, 287 treatment, 288-289 Cognitive symptoms, 153-154 Comparison of Atypicals in First Episode of Psychosis (CAFÉ) Study, 180 Complex disorders, 63, 65 Computational anatomy manual tracing, 75 methods and software tools, 76 structural shape and cortical thickness, 75 COPD. See Chronic obstructive pulmonary disease (COPD)
Index

Copy number variants (CNVs) BP, 65 genotype:phenotype correlations, 64 GWAS, 63 pathophysiology, 64 and trait-associated SNPs, 64 Coronary artery bypass graft (CABG) surgery, 178 Coronary artery disease (CAD), 176, 181, 182 CR. *See* Cognitive remediation (CR) Cytochrome P450 CYP1A2, 167 CYP3A4, 167 CYP3A5, 167 CYP2D6, 166

D

DBS. See Deep brain stimulation (DBS) Deep brain stimulation (DBS), 155-156 Device-based therapies, 145, 150 Diabetes mellitus diagnose, 181 retrospective cohort study, 182 schizophrenia, 182 Diffusion DTI, 7, 37, 76 **DWI.** 74 imaging techniques, 76 Diffusion tensor imaging (DTI), 37, 76 Dimensions, psychopathological, 18 Disconnection functional, 38 hypothesis, 74 Disrupted regional networks frontal-visual networks, 85 functional connectivity, 85-87 hyper-connectivity, 85 prefronto-temporal functional connectivity, 85 DLPFC. See Dorsolateral prefrontal cortex (DLPFC) Dopamine (DA) CLZ and asenapine, 117 D1-D5 receptors, 116-117 dysfunction, 46 existing antipsychotics, 116 hypothesis, 44-45 mesolimbic hyperactivity and ameliorating positive symptoms, 6 Dorsolateral prefrontal cortex (DLPFC), 151-154 DRD2, 169 DSM-5 and ICD-11

addition of psychopathological dimensions, 18 Attenuated Psychosis Syndrome, 16, 19 characteristic symptoms of schizophrenia, 16–17 "positive symptom", 17 psychotic disorders, dimensions, 18 Research Domain Criteria (RDoC) project, 18 "special hallucination", 17 subtypes in DSM-5 and ICD-11, 18 subtypes' stability, 16 DTI. See Diffusion tensor imaging (DTI)

Е

Early intervention, 112, 115 Early-onset of psychosis, 5 Electroconvulsive therapy (ECT) negative symptoms, 151 positive symptoms action mechanism, 140 administration, 144 adolescents, 143 Brief Psychiatric Rating Scale (BPRS) scores, 140 catatonia, 142 Clinical Global Impression Scale (CGI-S), 141 clozapine, 140 description, 139-140 flupenthixol, 140 guidelines, 144-145 maintenance treatment, 144 meta-analyses, 141-142 parameters, 145 postictal suppression index (PSI), 141 predict response, 143 Electrophysiology biomarker domains, 50 clinical and research tool, 39 neural processes, 41 and structural imaging, 44 End of treatment response (ETR), 190 Endophenotypes abnormal sleep with COMT mutations, 8 biomarkers, 8 GABA inhibitory interneurons, 8 structural and functional dysregulation, 8 Epigenetics, 7 ETR. See End of treatment response (ETR) Evidence-based treatments, 277, 301 Exome sequencing, 65 Extended major histocompatibility complex (xMHC), 63, 64

F

FGAs. See First-generation antipsychotics (FGAs) First-episode and early-onset patients, 111–114, 317 First-generation antipsychotics (FGAs), 108, 113, 114, 181, 183, 185 Function brain, 38–39

neuroimaging biomarkers (see Structure)

G

GABA. See Gamma aminobutyric acid (GABA) Gamma aminobutyric acid (GABA), 45, 119-120 G-CSF. See Granulocyte colony-stimulating factor (G-CSF) Genetics antipsychotics, 7 BP. 65 clinical applications, 65-66 DNA re-sequencing, 60 epidemiological approaches, 61 epigenetic variations, 7 genome-wide association studies (GWAS), 6 Human Genome Project, 60 human neuropsychiatric disorders, 60 molecular approaches candidate gene association, 62-63 GWAS, 63-65 linkage, 61-62 pharmacogenomics and pharmacogenetics, 65 single-nucleotide polymorphisms (SNPs), 7 statistical analyses, 66 Genome-wide association studies (GWAS) CNVs. 63 exonic de novo mutations, 65 genetic variants, 162 GWAS SNP arrays, 63 pharmacogenomics, 108 statistical associations, 64 transcriptomics (mRNA expression) analysis, 64 Glucocorticoids, 122-123 Glutamate (Glu), 45–46, 118–119 GM-CSF. See Granulocyte-macrophage colony-stimulating factor (GM-CSF) Granulocyte colony-stimulating factor (G-CSF), 244

Granulocyte-macrophage colony-stimulating factor (GM-CSF), 244 Gray matter antipsychotic exposure, 37 first-generation antipsychotic (FGA) exposure, 37 nonpsychotic relatives, 37 prodromal symptoms, 37 regional reductions, 37 structural brain alterations, 36 GWAS. *See* Genome-wide association studies (GWAS)

Н

HAART. See Highly active antiretroviral therapy (HAART) Haloperidol and fluphenazine decanoate, 271 Hematologic adverse effects agranulocytosis, 243 GM-CSF and G-CSF, 244 lithium. 244 monitoring guideline, 243, 244 neutropenia, 243 Hepatotoxicity, 249 Heritability, 61 Highly active antiretroviral therapy (HAART), 188 High risk (HR), study, 9, 37, 190, 317 Histamine, 121-122 HLA. See Human leukocyte antigen (HLA) Hormonal agents, 123-124 HTN. See Hypertension (HTN) Human Genome Project, 60 Human leukocyte antigen (HLA), 168 Human neuropsychiatric disorders, 60 Hyperlipidemia ACC/AHA Guidelines, 181 access to care, 177 CAFÉ Study, 180-181 schizophrenia patients, 180-181 statins, 181 WHO report, 180 Hyperprolactinemia antipsychotics risks, 235-236 bone mineral density (BMD), 235 definition, 235 diagnoses, 235, 236 gynecomastia, 235 patient risk factors, 235 prevention and treatment, 236-237 Hypersalivation, 246 Hypertension (HTN) antipsychotic drug, 180

Index

CATIE Study, 180 nonpsychiatric patients, 178 schizophrenia, 180 Thiazide-type diuretics, 180 treatment, 180 Hyponatremia alcoholism, 209 antidepressants, 209 lung cancer, 209 polydipsia, 208–209 psychiatric disorders, 208 risk factors, 208, 209 SSRIs, 209

I

IL-3 receptor (IL3RB), 6 Imaging animal models and longitudinal brain, 49 functional and molecular imaging, 38–39 neurochemical imaging, 43–44 and structural imaging, 44 Independent component analysis (ICA), 83 Infectious disease hepatitis B, 189 hepatitis C, 188–189 HIV, 188 management, 189–190 International Classification of Disease (ICD-6), 13

L

LAI. See Long-acting injectable (LAI) Large-scale networks brain network organization findings, 92-93 "fragmentation of brain pathologies", 92 graph-theoretic application, 88 graph theory network analysis, 88-91 organizations, 87 types, 88 Leucine-rich glioma-inactivated (LGI), 196 LGI. See Leucine-rich glioma-inactivated (LGI) Long-acting injectable (LAI) adequate adherence, 319 advantages, 267 antipsychotics (see Antipsychotics) formulations, 319 vs. oral antipsychotics, 267-268 Long-term management antipsychotic medications, 266, 319 complications and treatment cognitive behavioral therapy, 10

pharmacological treatment, 9 psychotic relapse, prevention, 9 LAI (*see* Long-acting injectable (LAI)) medication adherence, 267 psychosocial treatments, 319 relapse prevention and recovery, 265, 266 social-political issue, 319

M

Machine learning, 95 MC4R, 167-168 Measurement-based care. See DSM-5 and ICD-11 Medical care, patients with schizophrenia access to care, 177 epidemiological studies, 177 medical adverse events, 179 prospective analysis, 177 screening, 179 underdiagnosis, 178 undertreatment, 178 Medical complications adverse effects, 318 comorbidities, 318 monotherapies/augmentation strategies, 317 pharmacogenetics/pharmacogenomics, 317 psychosis syndrome, 317 therapeutic modulation, 317 Metabolic syndrome antipsychotics risks clozapine and olanzapine, 239 dyslipidemia, 240 insulin resistance, 240 weight gain, 239-240 definition, 238-239 prevention and treatment, 240-241 risk factors, 238-239 MFG treatment. See Multifamily group (MFG) treatment Mismatch negativity (MMN), 41 MMN. See Mismatch negativity (MMN) Mortality cancer, 191-192 comparator group, 176 HIV. 188 meta-analysis, 176 retrospective cohort review, 176 schizophrenia, 176 standardized mortality ratio (SMR), 175, 176 Movement disorders acute dystonia, 230 akathisia, 230-231 definition, 230

Movement disorders (cont.) mechanisms antipsychotics risks, 233 "dopamine supersensitivity hypothesis", 232 prevention and treatment, 233-234 serotonin inhibits. 232 parkinsonism, 231 TD, 231-232 Multifamily group (MFG) treatment, 279, 296, 297 Multimodal imaging classifier algorithms, 95 dysfunctional neural networks, 92 functional connectivity, 95 imaging modality, 92 reduced functional activity, 92 replication, 96

N

National Health Interview Survey (NHIS), 177 Negative symptoms ECT, 151 TMS, 151-153 Neurobiology categorizing illnesses, 28 "collateral damage", 30 compensatory responses, 31 differences, with and without schizophrenia, 5 endophenotypes abnormal sleep with COMT mutations, 8 biomarkers, 8 GABA inhibitory interneurons, 8 structural and functional dysregulation, 8 endophenotypic markers, 32 genetics, 7 heterogeneity, 30-31 heterogeneous disorder, 28 identification, 29 integrate pathologic findings, 27-28 multimodal imaging, 31 neuroimaging, 6-7 neurotransmission, 6 regulation, 29 replications, 28 "Rethinking Schizophrenia", 27 Neuroimaging biomarkers abnormal structure and function, 74-82 behavioral functions, 74 data collection and analysis, 96

disorganization, brain networks, 82-84 disrupted large-scale global networks, 87-92 disrupted regional networks, 85-87 multimodal and pattern-analytical, 92-96 "connectomics", 6 pathoetiology determination, 6 presynaptic/postsynaptic activity, 7 Neuroleptic malignant syndrome (NMS). 247-248 Neurological adverse effects, 245-246 Neuropathology, 46-47 Neurophysiology cortical oscillations and neural synchrony, 41 "endophenotypes", 41 MMN, 41 P300, 39-40 P50 auditory-evoked potential suppression, 40 prepulse inhibition of startle response (PPI), 40 Neuroplasticity, 47 Neuroscience-informed nosology, 49-50 Neurosteroids, 122-123 Neurotransmission dopamine (DA), 6 IL-3 receptor (IL3RB), 6 pathoetiology, 6 polymorphisms of neuregulin (NRG1), 6 Neurotransmitter systems dopamine hypothesis, 44-45 dopaminergic dysfunction, 46 **GABA**, 45 glutamate, 45-46 neurochemical imaging, 43-44 Neutraceuticals folate, 124-125 omega-3 fatty acids, 124 vitamin D. 125 NHIRD. See Taiwanese National Health Insurance Research Database (NHIRD) NHIS. See National Health Interview Survey (NHIS) NMS. See Neuroleptic malignant syndrome (NMS) Norepinephrine (NE), 117

0

Obesity, 182 Olanzapine pamoate (OP) efficacy, 269

and pamoic acid, 269 and PDSS, 269 OP. See Olanzapine pamoate (OP) Organization characterization, 88 mammalian brain networks, 87 neuroimaging modalities, 90 Osmostat, PIP syndrome acute psychosis, 214, 215 AVP. 214, 215 chronic antipsychotic treatment, 215 hippocampal pathology, 214 methylphenidate, 214, 215 neuroendocrine findings, 214 oxytocin treatment, 215 plasma osmolality, 214, 215 schizophrenia, 215

Р

Paliperidone palmitate (PP) antipsychotic activity, 270 olanzapine, 268 schizophrenia, 270 PAP. See Papanicolaou (PAP) Papanicolaou (PAP), 179 Parkinsonism, 231, 233-234 Pathophysiology abnormal limbic activation, 39 animal models and longitudinal brain imaging, 49 anti-inflammatory agents, 51 biomarkers and intermediate phenotypes, 316 brain structure, 36-38 diagnostic and predictive biomarkers, 50 dimensions of schizophrenia, 14 functional and molecular imaging, 38-39 GABA-ergic systems, 48 glutamate-mediated processes, 48 immunological theories, 42 longitudinal brain imaging, 49 molecular stratification, disease, 50 neurobiological basis, 35 neurodevelopmental disorder, 47 neuroendocrine research, 43 neuropathology, 46-47 neurophysiology, 39-41 neuroplasticity, 47 neuroscience-based classification, psychoses, 49-50 neurotransmitter systems, 43-46 oxidative stress, 42 pathways, treatment discovery, 49

perinatal/prenatal abnormalities, 48 synaptic/axonal pruning, 48 theoretical models, 35, 36 PDSS. See Postiniection delirium/sedation syndrome (PDSS) Peer-implemented services efficacy, 294-295 treatment, 293-294 Pharmacogenetics clinical guidance, 65 clinical use, 169-170 description, 161 and ethnicity, 170 genetic biomarkers antipsychotic-related adverse effects, 167-169 drug metabolism, 165-167 drug response, 162-165 genetic testing, 171 **GWAS**, 162 Pharmacogenomics. See Pharmacogenetics Pharmacologic treatment clozapine, 218 vaptans, 218 PIP syndrome. See Psychosis intermittent hyponatremia (PIP) syndrome Polydipsia amiloride, 216 chronic mental illness, 207 Clozapine's salutary effects, 216 lithium levels, 216 nephrogenic diabetes, 207 nephrotoxicity, 216 osteoporosis, 208 pathologic fractures, 208 polyuria, 208 thiazide diuretic, 216 tuberointerstitial nephritis, 208 water imbalance, 211, 212 Polygenic, 63, 65 Polymorphisms of neuregulin (NRG1), 6 Positive allosteric modulation, 119 Positive symptoms ECT, 139-145 TMS, 145-151 Postinjection delirium/sedation syndrome (PDSS), 269, 270 PP. See Paliperidone palmitate (PP) Prostate-specific antigen (PSA), 193 PSA. See Prostate-specific antigen (PSA) Psychopathological dimensions, 316 Psychosis acute treatment first-episode and early-onset patients imaging studies, 111

Psychosis acute treatment (cont.) meta-analyses, 111 treatment, 112-114 high-risk individuals, 109-111 multiepisode patients, 114-115 neuroprotective agents, 115 nonpharmacological approaches, 115 Psychosis intermittent hyponatremia (PIP) syndrome hyponatremia, 216-217 mortality, 210 primary polydipsia, 210, 213-214 radiator sitting syndrome, 210 reset osmostat, 214-216 sensory deficits, 210 water excretion, 210 Psychosocial rehabilitation approaches, 290 skills training approaches, 282 treatment approaches, 277-279 Psychotherapy CBT, 298-300 family-based services efficacy, 297 SMI, 295-296 treatment, 296-297

R

RA. See Rheumatoid arthritis (RA) Recovery antipsychotic medications, 272 earlier disease detection and interventions. 9 individualized treatments, 319 medication. 266 relapse prevention, 266 stage of illness, 4 Rehabilitation ACT. 282-285 CR, 287-290 peer-implemented services, 293-295 SE (see Supported employment (SE)) skills training CR and CBT. 282 efficacy, 281-282 money management, 282 psychosocial treatment approaches, 277-279 social learning theory, 280 treatment, 280-281 social cognition training, 291-293 Relapse prevention, 265, 266

Remission inter-episode, 4 and recovery, 9 Resting state networks (RSN), 84 Rheumatoid arthritis (RA), 195 Risk factors, epidemiology/genetic/ environmental cannabis, early onset of psychosis, 5 lifetime prevalence, 4 mean incidence of schizophrenia, 4 obstetric-related hypoxic events, 5 prenatal/postnatal environmental deficiencies, 5 Risperidone microspheres (RM), 269, 271 RM. See Risperidone microspheres (RM) RSN. See Resting state networks (RSN)

S

Schizophrenia ACT. 284 antidiuretic function, 205 antipsychotic, 108-110 AVP, 206, 207 CBTp, 299 chronic psychotic disease, 205 clinical presentation neuroleptic malignant syndrome, 210 PIP syndrome, 210 polydipsia, 209 rhabdomyolysis, 210 water intoxication, 209 course first psychotic episode, 4 premorbid phase, 4 prodromal phase, 4 data and clinical experience, 107 definition, 3 diagnosis hyponatremia, 211-213 PIP syndrome, 211 polydipsia, 211 diagnostic issues, 316 etiological and pathophysiological factors, 31 first psychotic episode, 4 genetics (see Genetics) hyponatremia, 208-209 idiopathic hyponatremia, 205 inter-episode remission, 4 management acute and long-term, 9 deficits in verbal fluency and memory functioning, 9

high risk (HR) states for psychosis, 9 ultrahigh risk states or first-episode psychosis, 9-10 medical management, 316-318 multiple psychopathological dimensions, 4 neurobiological findings, 319 neuroimaging findings, 31 nosology, 315 osmotic threshold, 206, 207 pathophysiology (see Pathophysiology) pathophysiology, PIP syndrome, 213 - 216pharmacogenetics (see Pharmacogenetics) pharmacologic treatment, 217-218 plasma osmolality, 206, 207 polydipsia (see Polydipsia) psychosis (see Psychosis acute treatment) psychosocial treatment, 277 psychotic symptoms, 3 recovery, 4 rehabilitation, 277 repeated episodes of psychosis, 4 and social cognition, 282 social skills training, 281 stable phase or plateau, 4 targeted fluid restriction, 217 19th century to DSM-IV and ICD-10 definitions, 15 delusions and hallucinations, 15 dementia praecox, 14 impairment of empathic communication, 15 manic-depressive insanity, 14 Schneiderian first-rank symptoms, 15 therapeutic neuromodulation (see Therapeutic neuromodulation) treatment-resistance agents impacting, neurotransmitter systems/receptors, 116-122 antidepressants, 122 anti-inflammatory agents, 123 antipsychotic polypharmacy, 116 cannabinoids, 123 clinical strategy, 127 cognitive and psychosocial therapies, 125 - 126dopaminergic D2 mechanisms, 115 glucocorticoids/neurosteroids, 122-123 hormonal agents, 123-124 neutraceuticals, 124-125 water excretion, 206 water intoxication, 208-209 SCIT. See Social cognition interaction training (SCIT)

SCST. See Social cognition skills training (SCST) SE. See Supported employment (SE) Second-generation antipsychotics (SGAs). 108, 113, 114, 181, 183, 184 Serotonin, 117 Serotonin reuptake inhibitors (SSRIs), 209 Severe mental illness (SMI) and ACT, 284, 285 family involvement, 295 and schizophrenia, 295 Sexual dysfunction antipsychotics risks, 237-238 definition, 237 prevention and treatment, 238 SGAs. See Second-generation antipsychotics (SGAs) Single nucleotide polymorphism (SNP) genetic markers, 63 trait-associated, 64 xMHC, 63 SMI. See Severe mental illness (SMI) Social cognition interaction training (SCIT), 292 Social cognition skills training (SCST), 292 Social cognition training efficacy, 292-293 schizophrenia, 291 treatment, 291-292 Somatic therapies, 277 SSRIs. See Serotonin reuptake inhibitors (SSRIs) Structure brain gray matter, 36-37 integration, structural data, 38 neurodevelopmental timeline, 38 white matter, 37-38 and function, neuroimaging biomarkers brain structural abnormalities, 74-75 computational anatomy, 75-76 distributed and heterogeneous pattern, 77 - 80DTI, 76, 81, 82 fMRI activation abnormalities, 81 gray and white matter abnormalities, 75,77 meta-analysis tools, 75 ventricular spaces, 74 volumetric measurement, ROIs, 74 Subtypes. See DSM-5 and ICD-11 Sudden death antipsychotics risks, 229-230 definition, 226

Sudden death (cont.) drugs, 226, 227 enzyme inhibitors, 226, 228 prevention and treatment, 230 primary arrhythmogenic risk, 226 OT interval, 226, 228, 229 risk factors, 226, 227 Sulfotransferase 4A1 (SULT4A1) description, 164 neurotransmitter/neuromodulator, 164 and schizophrenia, 164-165 SULT4A1. See Sulfotransferase 4A1 (SULT4A1) Supported employment (SE) efficacy, 286-287 "train-place" model, 285 treatment, 286 Sustained viral response (SVR), 190 SV2C. See Synaptic vesicle (SV2C) SVR. See Sustained viral response (SVR) Synaptic vesicle (SV2C), 163

Т

Taiwanese National Health Insurance Research Database (NHIRD), 179, 191 Tardive dyskinesia (TD), 168-169, 231-232, 234 Target fluid restriction, 217, 219 TCL. See Training in community living (TCL) TD. See Tardive dyskinesia (TD) tDCS. See Transcranial direct current stimulation (tDCS) Therapeutic neuromodulation cognitive symptoms, 153-154 DBS, 155-156 negative symptoms ECT, 151 TMS. 151–153 positive symptoms ECT. 139-145 TMS, 145-151 tDCS, 154-155 Thermoregulatory adverse effects, 246 TMS. See Transcranial magnetic stimulation (TMS) Training in community living (TCL), 282, 283 Transcranial direct current stimulation (tDCS) acute psychotic symptoms, 154-155 description, 154 maintenance treatment, 155 pilot data, 155

Transcranial magnetic stimulation (TMS) cognitive symptoms, 153-154 negative symptoms action mechanism, 151 controlled studies, 152-153 meta-analyses, 153 open-label studies, 151-152 positive symptoms action mechanism, 146 administration, 150 adolescents, 150 AHRS, 146-147 cognitive symptoms, 153-154 description, 145-146 double-blind study, 146, 148 Hallucination Change Scale (HCS) score, 146, 147 maintenance treatment, 150 meta-analyses, 149 parameters, 150, 151 placebo-controlled crossover study, 147 randomized controlled study, 147-148 Treatment ACT, 283-284 CBT. 298-299 coping skills, 280-281 CR, 288-289 family-based services BFT, 296, 297 MFG, 296, 297 illness/wellness/recovery skills, 280 implications of current conceptualization, 20 independent living skills training, 280 interpersonal skills training, 280 peer-implemented services, 293-294 SE. 286 social cognition training, 291-292 tailoring skills training, 281 therapeutic approaches, 4 Treatment-resistance agents impacting, neurotransmitter systems/receptors, 116-122 antidepressants, 122 anti-inflammatory agents, 123 antipsychotic polypharmacy, 116 cannabinoids, 123 clinical strategy, 126, 127 cognitive and psychosocial therapies, 125, 126 dopaminergic D2 mechanisms, 115 glucocorticoids/neurosteroids, 122-123 hormonal agents, 123-124 neutraceuticals, 124-125

Index

U

Ultrahigh risk or first-episode psychosis, 9-10

v

Vaptans, 217, 218 Vasopressin, 209, 218 Venous thromboembolism (VTE), 248 VTE. See Venous thromboembolism (VTE)

W

Water intoxication acute cerebral edema, 210 clozapine, 218 hyponatremia, 208–209 schizophrenia, 205 targeted fluid restriction, 217 thiazide diuretic, 216 WGS. *See* Whole-genome sequencing (WGS) White matter DTI, 37 FA reductions, 37 syndrome, functional disconnection, 38 Whole-genome sequencing (WGS), 65, 66

Х

xMHC. See Extended major histocompatibility complex (xMHC)